

ZyWALL 110/310/1100 Series

VPN Firewall

Version 3.10 Edition 4, 01/2014

User's Guide

Default Login Details		
LAN Port IP Address	https://192.168.1.1	
User Name	admin	
Password	1234	

IMPORTANT!

READ CAREFULLY BEFORE USE.

KEEP THIS GUIDE FOR FUTURE REFERENCE.

This is a User's Guide for a series of products. Not all products support all firmware features. Screenshots and graphics in this book may differ slightly from your product due to differences in your product firmware or your computer operating system. Every effort has been made to ensure that the information in this manual is accurate.

Screen Identification Syntax Convention

The > symbol is used to identify a mouse click in a path to access a screen in the web configurator. For example, **Configuration** > **Network** > **Interface** > **Ethernet** means first you click the **Configuration** icon in the navigation panel, then click the **Network** menu item, then the **Interface** submenu and finally the **Ethernet** tab in order to access the Ethernet interface screen.

Related Documentation

• Quick Start Guide

The Quick Start Guide shows how to connect the ZyWALL and access the Web Configurator wizards. (See the wizard real time help for information on configuring each screen.) It also contains a connection diagram and package contents list.

CLI Reference Guide

The CLI Reference Guide explains how to use the Command-Line Interface (CLI) to configure the ZyWALL.

Note: It is recommended you use the Web Configurator to configure the ZyWALL.

• Web Configurator Online Help

Click the help icon in any screen for help in configuring that screen and supplementary information.

Chapter 1	
Introduction	
1.1 Overview	
1.2 Management Overview	
1.3 Web Configurator	21
1.3.1 Web Configurator Access	21
1.3.2 Web Configurator Screens Overview	
1.3.3 Navigation Panel	
1.3.4 Tables and Lists	
Chapter 2	
Installation Setup Wizard	33
2.1 Installation Setup Wizard Screens	
2.1.1 Internet Access Setup - WAN Interface	
2.1.2 Internet Access: Ethernet	
2.1.3 Internet Access: PPPoE	
2.1.4 Internet Access: PPTP	
2.1.5 ISP Parameters	
2.1.6 Internet Access - Finish	
Chapter 3 Hardware Introduction	37
3.1 Default Zones, Interfaces, and Ports	
3.2 Stopping the ZyWALL	
3.3 Rack-mounting	
3.3 Rack-mounting	
3.3 Rack-mounting 3.4 Wall-mounting 3.5 Front Panel LEDs	
3.3 Rack-mounting 3.4 Wall-mounting 3.5 Front Panel LEDs 3.5.1 Rear Panels	
3.3 Rack-mounting 3.4 Wall-mounting 3.5 Front Panel LEDs 3.5.1 Rear Panels Chapter 4	
3.3 Rack-mounting 3.4 Wall-mounting 3.5 Front Panel LEDs 3.5.1 Rear Panels Chapter 4 Quick Setup Wizards	
3.3 Rack-mounting 3.4 Wall-mounting 3.5 Front Panel LEDs 3.5.1 Rear Panels Chapter 4 Quick Setup Wizards 4.1 Quick Setup Overview	
3.3 Rack-mounting	
3.3 Rack-mounting 3.4 Wall-mounting 3.5 Front Panel LEDs 3.5.1 Rear Panels Chapter 4 Quick Setup Wizards 4.1 Quick Setup Overview 4.2 WAN Interface Quick Setup 4.2.1 Choose an Ethernet Interface	
3.3 Rack-mounting	38 39 39 41 41 42 42 42 42 43 43
3.3 Rack-mounting	38 39 39 41 42 42 42 42 43 43 43 43
 3.3 Rack-mounting 3.4 Wall-mounting 3.5 Front Panel LEDs 3.5.1 Rear Panels Chapter 4 Quick Setup Wizards 4.1 Quick Setup Overview 4.2 WAN Interface Quick Setup 4.2.1 Choose an Ethernet Interface 4.2.2 Select WAN Type 4.2.3 Configure WAN Settings 4.2.4 WAN and ISP Connection Settings 	38 39 39 41 41 42 42 42 42 43 43 43 44
 3.3 Rack-mounting 3.4 Wall-mounting 3.5 Front Panel LEDs 3.5.1 Rear Panels Chapter 4 Quick Setup Wizards 4.1 Quick Setup Overview 4.2 WAN Interface Quick Setup 4.2.1 Choose an Ethernet Interface 4.2.2 Select WAN Type 4.2.3 Configure WAN Settings 4.2.4 WAN and ISP Connection Settings 4.2.5 Quick Setup Interface Wizard: Summary 	38 39 39 41 42 42 42 42 42 43 43 43 43 44 44
 3.3 Rack-mounting 3.4 Wall-mounting 3.5 Front Panel LEDs 3.5.1 Rear Panels Chapter 4 Quick Setup Wizards 4.1 Quick Setup Overview 4.2 WAN Interface Quick Setup 4.2.1 Choose an Ethernet Interface 4.2.2 Select WAN Type 4.2.3 Configure WAN Settings 4.2.4 WAN and ISP Connection Settings 4.2.5 Quick Setup Interface Wizard: Summary 4.3 VPN Setup Wizard 	38 39 39 41 41 42 42 42 42 43 43 43 43 44 44 44 44
 3.3 Rack-mounting 3.4 Wall-mounting 3.5 Front Panel LEDs 3.5.1 Rear Panels Chapter 4 Quick Setup Wizards 4.1 Quick Setup Overview 4.2 WAN Interface Quick Setup 4.2.1 Choose an Ethernet Interface 4.2.2 Select WAN Type 4.2.3 Configure WAN Settings 4.2.4 WAN and ISP Connection Settings 4.2.5 Quick Setup Interface Wizard: Summary 4.3 VPN Setup Wizard 4.3.1 Welcome 	38 39 39 41 42 42 42 42 43 43 43 43 43 43 43 43 43 43 43 44 44

4.3.3 VPN Express Wizard - Scenario	49
4.3.4 VPN Express Wizard - Configuration	50
4.3.5 VPN Express Wizard - Summary	50
4.3.6 VPN Express Wizard - Finish	51
4.3.7 VPN Advanced Wizard - Scenario	52
4.3.8 VPN Advanced Wizard - Phase 1 Settings	53
4.3.9 VPN Advanced Wizard - Phase 2	55
4.3.10 VPN Advanced Wizard - Summary	56
4.3.11 VPN Advanced Wizard - Finish	56
4.4 VPN Settings for Configuration Provisioning Wizard: Wizard Type	57
4.4.1 Configuration Provisioning Express Wizard - VPN Settings	58
4.4.2 Configuration Provisioning VPN Express Wizard - Configuration	59
4.4.3 VPN Settings for Configuration Provisioning Express Wizard - Summary	60
4.4.4 VPN Settings for Configuration Provisioning Express Wizard - Finish	61
4.4.5 VPN Settings for Configuration Provisioning Advanced Wizard - Scenario	62
4.4.6 VPN Settings for Configuration Provisioning Advanced Wizard - Phase 1 Settings	63
4.4.7 VPN Settings for Configuration Provisioning Advanced Wizard - Phase 2	64
4.4.8 VPN Settings for Configuration Provisioning Advanced Wizard - Summary	65
4.4.9 VPN Settings for Configuration Provisioning Advanced Wizard- Finish	65
Chanter 5	
Dashboard	67
5.1 Overview	67

5.1 Overview	67
5.1.1 What You Can Do in this Chapter	67
5.2 The Dashboard Screen	67
5.2.1 The CPU Usage Screen	72
5.2.2 The Memory Usage Screen	73
5.2.3 The Active Sessions Screen	73
5.2.4 The VPN Status Screen	74
5.2.5 The DHCP Table Screen	75
5.2.6 The Number of Login Users Screen	76

Part II: Technical Reference	77
------------------------------	----

Chapter 6

79
79
80
81
82

6.4 The Traffic Statistics Screen	86
6.5 The Session Monitor Screen	
6.6 The DDNS Status Screen	91
6.7 IP/MAC Binding Monitor	91
6.8 The Login Users Screen	
6.9 Cellular Status Screen	
6.9.1 More Information	
6.10 USB Storage Screen	
6.11 The IPSec Monitor Screen	
6.11.1 Regular Expressions in Searching IPSec SAs	
6.12 The SSL Connection Monitor Screen	
6.13 The L2TP over IPSec Session Monitor Screen	
6.14 Log Screen	
Chapter 7 Interfaces	103

7.1 Interface Overview	
7.1.1 What You Can Do in this Chapter	
7.1.2 What You Need to Know	
7.1.3 What You Need to Do First	
7.2 Port Role Screen	
7.3 Ethernet Summary Screen	
7.3.1 Ethernet Edit	
7.3.2 Object References	
7.3.3 Add/Edit DHCPv6 Request/Release Options	
7.3.4 Add/Edit DHCP Extended Options	
7.4 PPP Interfaces	
7.4.1 PPP Interface Summary	
7.4.2 PPP Interface Add or Edit	127
7.5 Cellular Configuration Screen (3G)	132
7.5.1 Cellular Add/Edit Screen	134
7.6 Tunnel Interfaces	140
7.6.1 Configuring a Tunnel	142
7.6.2 Tunnel Add or Edit Screen	143
7.7 VLAN Interfaces	147
7.7.1 VLAN Summary Screen	148
7.7.2 VLAN Add/Edit	150
7.8 Bridge Interfaces	159
7.8.1 Bridge Summary	160
7.8.2 Bridge Add/Edit	
7.9 Virtual Interfaces	170
7.9.1 Virtual Interfaces Add/Edit	171
7.10 Interface Technical Reference	

Chapter 8 Trunk	
8.1 Overview	176
8.1.1 What You Can Do in this Chapter	176
8 1 2 What You Need to Know	176
8 2 The Trunk Summary Screen	179
8 2 1 Configuring a User-Defined Trunk	180
8.2.2 Configuring the System Default Trunk	
Chapter 9 Policy and Static Routes	
9.1 Policy and Static Routes Overview	
9.1.1 What You Can Do in this Chapter	
9.1.2 What You Need to Know	
9.2 Policy Route Screen	
9.2.1 Policy Route Edit Screen	
9.3 IP Static Route Screen	
9.3.1 Static Route Add/Edit Screen	
9.4 Policy Routing Technical Reference	195
Chapter 10 Routing Protocols	197
10.1 Routing Protocols Overview	
10.1.1 What You Can Do in this Chapter	197
10.1.2 What You Need to Know	
10.2 The RIP Screen	197
10.3 The OSPF Screen	
10.3.1 Configuring the OSPF Screen	
10.3.2 OSPF Area Add/Edit Screen	
10.3.3 Virtual Link Add/Edit Screen	
10.4 Routing Protocol Technical Reference	
Chapter 11 Zones	
11.1 Zones Overview	
11.1.1 What You Can Do in this Chapter	
11.1.2 What You Need to Know	
11.2 The Zone Screen	
11.3 Zone Edit	
Chapter 12 DDNS	212
12.1 DDNS Overview	

	010
12.1.1 What You Can Do In this Chapter	
12.1.2 What You Need to Know	
12.2 The DDNS Screen	
12.2.1 The Dynamic DNS Add/Edit Screen	
Chapter 13 NAT	
	0.17
13.1 NAT OVERVIEW	
13.1.1 What You Can Do in this Chapter	
12.2 The NAT Screen	
12.2.1 The NAT Add/Edit Screen	
13.3 NAT Technical Reference	
Chapter 14	
HTTP Redirect	
14.1 Overview	
14.1.1 What You Can Do in this Chapter	
14.1.2 What You Need to Know	
14.2 The HTTP Redirect Screen	
14.2.1 The HTTP Redirect Edit Screen	
Chapter 15	
ALG	
15.1 ALG Overview	
15.1.1 What You Can Do in this Chapter	
15.1.2 What You Need to Know	
15.1.3 Before You Begin	231
15.2 The ALG Screen	231
15.3 ALG Technical Reference	
Chapter 16	
IP/MAC Binding	235
16.1 IP/MAC Binding Overview	235
16.1.1 What You Can Do in this Chapter	235
16.1.2 What You Need to Know	
16.2 IP/MAC Binding Summary	
16.2.1 IP/MAC Binding Edit	
16.2.2 Static DHCP Edit	
16.3 IP/MAC Binding Exempt List	
Chanter 47	
Unapter 17 Inhound Load Balancing	240
mound Load Dalahong	

17.1 Inbound Load Balancing Overview	240
17.1.1 What You Can Do in this Chapter	240
17.2 The Inbound LB Screen	241
17.2.1 The Inbound LB Add/Edit Screen	242
17.2.2 The Inbound LB Member Add/Edit Screen	244
Chapter 18 Authentication Policy	246
18.1 Overview	246
18.1.1 What You Can Do in this Chapter	
18.1.2 What You Need to Know	246
18.2 Authentication Policy Screen	247
18.2.1 Creating/Editing an Authentication Policy	249
18.3 User-aware Access Control Example	251
18.3.1 Set Up User Accounts	251
18.3.2 Set Up User Groups	
18.3.3 Set Up User Authentication Using the RADIUS Server	
18.3.4 User Group Authentication Using the RADIUS Server	254
Chapter 19 Firewall	
19.1 Overview	
19.1.1 What You Can Do In this Chapter	
10.2 The Firewall Sereen	
19.2 The Filewall Screen	
19.2.1 Conliguing the Firewall Screen	
10.2 The Section Limit Screen	
19.3 The Session Limit Add/Edit Screen	
19.3.1 The Session Limit Add/Edit Screen	
19.5 Firewall Rule Example Applications	
Chapter 20 IPSec VPN	
20.1 Virtual Private Networks (VPN) Overview	272
20.1.1 What You Can Do in this Chanter	
20.1.2 What You Need to Know	274
20.1.3 Before You Begin	274 975
20.2 The VPN Connection Screen	276
20.2.1 The VPN Connection Add/Edit (IKE) Screen	
20.2.2 The VPN Connection Add/Edit Manual Key Screen	283
20.3 The VPN Gateway Screen	
20.3.1 The VPN Gateway Add/Edit Screen	

	20.4 VPN Concentrator	
	20.4.1 VPN Concentrator Requirements and Suggestions	
	20.4.2 VPN Concentrator Screen	
	20.4.3 The VPN Concentrator Add/Edit Screen	
	20.5 ZyWALL IPSec VPN Client Configuration Provisioning	
	20.6 IPSec VPN Background Information	
Chap	pter 21	
SSL	VPN	
	21.1 Overview	
	21.1.1 What You Can Do in this Chapter	
	21.1.2 What You Need to Know	
	21.2 The SSL Access Privilege Screen	
	21.2.1 The SSL Access Policy Add/Edit Screen	
	21.3 The SSL Global Setting Screen	
	21.3.1 How to Upload a Custom Logo	
	21.4 SSL VPN Example	
Char	pter 22	
SSL	User Screens	
	22.1 Overview	
	22.1.1 What You Need to Know	
	22.2 Remote SSL User Login	
	22.3 The SSL VPN User Screens	
	22.4 Bookmarking the ZyWALL	
	22.5 Logging Out of the SSL VPN User Screens	
	22.6 SSL User Application Screen	
	22.7 SSL User File Sharing	
	22.7.1 The Main File Sharing Screen	
	22.7.2 Opening a File or Folder	
	22.7.3 Downloading a File	
	22.7.4 Saving a File	
	22.7.5 Creating a New Folder	
	22.7.6 Renaming a File or Folder	
	22.7.7 Deleting a File or Folder	
	22.7.8 Uploading a File	329
Char	pter 23	
ZyW	ALL SecuExtender	331
	23.1 The ZyWALL SecuExtender Icon	
	23.2 Status	
	23.3 View Log	
	23.4 Suspend and Resume the Connection	
	the second se	

23.6 Uninstalling the ZyWALL SecuExtender 333 Chapter 24 335 L2TP VPN 335 24.1 Overview 335 24.1.1 What You Can Do in this Chapter 335 24.1.2 What You Need to Know 335 24.2 L2TP VPN Screen 337 Chapter 25 339 Bandwidth Management. 339 25.1 Overview 339 25.1 Overview 339 25.1 The Nat You Can Do in this Chapter 339 25.2 The Bandwidth Management Screen 343 25.2.1 The Bandwidth Management Add/Edit Screen 349 26.1 Overview 349 26.1.2 What You Need to Know 349 26.1.3 Before You Begin 350 26.3 The Active-Passive Mode Device HA 353 2		23.5 Stop the Connection	333
Chapter 24 335 L2TP VPN		23.6 Uninstalling the ZyWALL SecuExtender	
Chapter 24 335 24.1 Overview 335 24.1 2 What You Can Do in this Chapter 335 24.1 2 What You Need to Know 335 24.1 2 What You Need to Know 335 24.1 2 TP VPN Screen 337 Chapter 25 Bandwidth Management 339 25.1 Overview 339 25.1 Overview 339 25.2 The Bandwidth Management Screen 339 345 349 25.2 The Bandwidth Management Add/Edit Screen 349 349 26.1 Overview 349 349 340 26.1 Overview 349 349 356 349 26.1 Overview 349 361.1 What You Can Do in this Chapter 349 349 26.1 Overview 349 361.1 What You Can Do in this Chapter 349 356.2 Device HA 350 26.1 Overview 349 361.3 Before You Begin 350 350 350 26.3 The Active-Passive Mode Screen 351 263.1 Configuring Active-Passive Mode Device HA 353 356 356 356 356.3 Chapter 27 361 27.1 What You Can Do in this Chapter 356 356 356.4	Chanta		
24.1 Overview 335 24.1.1 What You Can Do in this Chapter 335 24.1.2 What You Need to Know 335 24.1.2 What You Need to Know 335 24.1.2 What You Need to Know 337 Chapter 25 Bandwidth Management. 339 25.1 Overview 339 25.1.1 What You Can Do in this Chapter 339 25.1.2 What You Need to Know 339 25.1.2 What You Need to Know 339 25.2 The Bandwidth Management Screen 343 25.2.1 The Bandwidth Management Add/Edit Screen 349 26.1.0 Verview 349 26.1.1 What You Can Do in this Chapter 349 26.1.1 What You Can Do in this Chapter 349 349 26.1.2 What You Need to Know 349 341 26.1.2 What You Need to Know 349 351 26.1.3 Defore You Begin 350 350 26.3 The Active-Passive Mode Device HA 353 352 26.4 Configuring Active-Passive Mode Device HA 353 352 356 26.5 Device HA Technical Reference 356 26.3 The Active-Passive Mode Device HA 353 352 356 356 356 356 356 <	L2TP V	er 24 /PN	
241 Overview			
24.1.1 What You Can Do in this Chapter		24.1 Overview	
24.12 What You Need to Know 335 24.2 L2TP VPN Screen 337 Chapter 25 339 25.1 Overview 339 25.1.1 What You Can Do in this Chapter 339 25.1.2 What You Need to Know 339 25.2.1 The Bandwidth Management Screen 343 25.2.1 The Bandwidth Management Add/Edit Screen 349 25.2.1 The Bandwidth Management Add/Edit Screen 349 26.1 Overview 349 26.1.1 What You Can Do in this Chapter 349 26.1.1 What You Can Do in this Chapter 349 26.1.2 What You Need to Know 349 26.1.3 What You Need to Know 349 26.1.4 What You Can Do in this Chapter 349 26.1.2 What You Need to Know 349 26.3.1 Configuring Active-Passive Mode Device HA 350 26.4 Configuring an Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 356 Chapter 27 361 27.1 Overview		24.1.1 What You Can Do in this Chapter	
24.2 L2TP VPN Screen 337 Chapter 25 Bandwidth Management 339 25.1 Overview 339 25.1.1 What You Can Do in this Chapter 339 25.2 The Bandwidth Management Screen 343 25.2.1 The Bandwidth Management Screen 343 25.2.1 The Bandwidth Management Add/Edit Screen 349 26.1 Overview 349 26.1 Overview 349 26.1 Overview 349 26.1.2 What You Can Do in this Chapter 349 26.1 Overview 349 26.1.3 Before You Begin 350 26.3 The Active-Passive Mode Screen 351 26.3 The Active-Passive Mode Device HA 353 26.4 Configuring Active-Passive Mode Device HA 353 26.5 Device HA Ceneral 355 26.5 Device HA Technical Reference 356 Chapter 27 361 27.1 Overview 361 27.1.1 What You Can Do in this Chapter 361 27.1.2 What You Need To Know 361 27.1.2 What You Can Do in this Chapter 361 27.1 Overview 361 27.1.1 What You Can Do in this Chapter <td< td=""><td></td><td>24.1.2 What You Need to Know</td><td></td></td<>		24.1.2 What You Need to Know	
Chapter 25 339 25.1 Overview 339 25.1.1 What You Can Do in this Chapter 339 25.1.2 What You Can Do in this Chapter 339 25.1.2 What You Need to Know 339 25.2 The Bandwidth Management Screen 343 25.2.1 The Bandwidth Management Screen 343 25.2.1 The Bandwidth Management Add/Edit Screen 349 26.1 Overview 349 26.1.1 What You Can Do in this Chapter 349 26.1.2 What You Need to Know 349 26.1.3 Before You Begin 350 26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.4 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Device HA 353 26.5 Device HA Technical Reference 356 Chapter 27 361 27.1 Overview 361 27.1 Overview 361 27.1 Overview 361 27.1 Overview 361 27.1 User Summary Screen 363 27.2 User Summary Screen 363 <t< td=""><td></td><td>24.2 L2TP VPN Screen</td><td></td></t<>		24.2 L2TP VPN Screen	
Bandwidth Management. 339 25.1 Overview 339 25.1.1 What You Can Do in this Chapter 339 25.1.2 What You Need to Know 339 25.2 The Bandwidth Management Screen 343 25.2.1 The Bandwidth Management Add/Edit Screen 345 Chapter 26 Device HA 349 26.1 Overview 349 26.1.1 What You Can Do in this Chapter 349 26.1.2 What You Need to Know 349 26.1.3 Before You Begin 350 26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.3 The Active-Passive Mode Screen 353 26.4 Configuring an Active-Passive Mode Device HA 353 26.5 Device HA Technical Reference 355 26.5 Device HA Technical Reference 356 27.1 Overview 361 27.1 Overview 361 27.1 Overview 361 27.2.1 User Add/Edit Screen 363 27.2.2 User Summary Screen 363 27.2.3 User Group Summary Screen 364 <t< td=""><td>Chapte</td><td>er 25</td><td></td></t<>	Chapte	er 25	
25.1 Overview 339 25.1.1 What You Can Do in this Chapter 339 25.2 The Bandwidth Management Screen 343 25.2 The Bandwidth Management Screen 343 25.2.1 The Bandwidth Management Add/Edit Screen 349 25.2.1 The Bandwidth Management Add/Edit Screen 349 26.1.1 What You Can Do in this Chapter 349 26.1.1 What You Can Do in this Chapter 349 26.1.2 What You Need to Know 349 26.1.2 What You Need to Know 349 26.1.3 Before You Begin 350 26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.3 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 User/Group 361 27.1 Overview 361 371.1 What You Can Do in this Chapter 361 27.1 Overview 361 371.2 What You Need To Know 361 27.2 User Summary Screen 363 361 372.2 User Add/Edit Screen 363 27.3 User Group Summary Screen 366 36	Band	width Management	
25.1.1 What You Can Do in this Chapter 339 25.1.2 What You Need to Know 339 25.2 The Bandwidth Management Screen 343 25.2.1 The Bandwidth Management Add/Edit Screen 345 Chapter 26 Device HA 349 26.1 Overview 349 26.1.1 What You Can Do in this Chapter 349 26.1.2 What You Need to Know 349 26.1.3 Before You Begin 350 26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 User/Group 361 27.1 Overview 361 27.1 Overview 361 27.2.1 User Summary Screen 363 27.2.1 User Group Summary Screen 363 27.3.1 Group Add/Edit Screen 366 27.3.1 Group Add/Edit Screen 367 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.1 Default User Authentication		25.1 Overview	
25.1.2 What You Need to Know 339 25.2 The Bandwidth Management Screen 343 25.1 The Bandwidth Management Add/Edit Screen 345 Chapter 26 Device HA 349 26.1 Overview 26.1.1 What You Can Do in this Chapter 349 26.1.2 26.1.2 What You Need to Know 349 26.1.3 26.1.3 Before You Begin 350 26.2 26.3.1 Chapter Passive Mode Screen 351 26.3.1 26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 26.5 Device HA Technical Reference 356 26.5 Chapter 27 User/Group 361 27.1.1 What You Can Do in this Chapter 361 27.1.1 27.1.2 What You Can Do in this Chapter 361 27.2.1 27.1.1 What You Can Do in this Chapter 361 27.1.2 27.1.1 What You Can Do in this Chapter <t< td=""><td></td><td>25.1.1 What You Can Do in this Chapter</td><td></td></t<>		25.1.1 What You Can Do in this Chapter	
25.2 The Bandwidth Management Screen		25.1.2 What You Need to Know	
25.2.1 The Bandwidth Management Add/Edit Screen		25.2 The Bandwidth Management Screen	
Chapter 26 349 26.1 Overview 349 26.1.1 What You Can Do in this Chapter 349 26.1.2 What You Need to Know 349 26.1.3 Before You Begin 350 26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 User/Group 361 27.1 Overview 361 361 27.2 User Summary Screen 363 363 27.2.1 User Add/Edit Screen 364 363 27.3.1 Group Add/Edit Screen 366 364 27.3.1 Group Add/Edit Screen 366 363 27.4.1 Default User Authentication Timeout Settings Edit Screens 366 27.4.1 Default User Authentication Timeout Settings Edit Screens 366 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372		25.2.1 The Bandwidth Management Add/Edit Screen	
Orapier 26 Device HA 349 26.1 Overview 349 26.1.1 What You Can Do in this Chapter 349 26.1.2 What You Need to Know 349 26.1.3 Before You Begin 350 26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 User/Group 361 27.1 Overview 361 27.1.1 What You Can Do in this Chapter 361 27.2.1 User Add/Edit Screen 364 27.3.1 User Group Summary Screen 366 27.3.1 Group Add/Edit Screen 366 27.3.1 Group Add/Edit Screen 366 27.3.1 Group Add/Edit Screen 366 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.1 Default User Authentication Timeout Settings Edit Screens 371 27.5.1 User /Group Technical Reference 372	Chante	or 26	
26.1 Overview 349 26.1.1 What You Can Do in this Chapter 349 26.1.2 What You Need to Know 349 26.1.3 Before You Begin 350 26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Device HA 353 26.5 Device HA Technical Reference 356 Chapter 27 User/Group User/Group 361 27.1 Overview 361 27.2.1 What You Can Do in this Chapter 361 27.2.1 User Add/Edit Screen 363 27.2.1 User Add/Edit Screen 363 27.3.1 Group Add/Edit Screen 364 27.3.1 Group Setting Screen 366 27.3.1 Group Add/Edit Screen 366 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372	Device	er zo e HA	
26.1 Overview 349 26.1.1 What You Can Do in this Chapter 349 26.1.2 What You Need to Know 349 26.1.3 Before You Begin 350 26.1.3 Before You Begin 350 26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 361 27.1 Overview 361 27.1.2 What You Can Do in this Chapter 361 27.2.1 User Add/Edit Screen 363 27.2.1 User Add/Edit Screen 363 27.3.1 Group Summary Screen 363 27.4 The User/Group Setting Screen 366 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372			0.40
26.1.1 What You Can Do in this Chapter		26.1 Overview	
26.1.2 What You Need to Know 349 26.1.3 Before You Begin 350 26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 361 27.1 Overview 361 27.1.1 What You Can Do in this Chapter 361 27.2.1 User Summary Screen 363 27.2.1 User Add/Edit Screen 363 27.3.1 Group Summary Screen 366 27.3.1 Group Add/Edit Screen 366 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372		26.1.1 What You Can Do in this Chapter	
26.1.3 Before You Begin 350 26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 361 27.1 Overview 361 27.1.1 What You Can Do in this Chapter 361 27.2.1 User Summary Screen 363 27.2.1 User Add/Edit Screen 363 27.3.1 Group Add/Edit Screen 366 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372		26.1.2 What You Need to Know	
26.2 Device HA General 350 26.3 The Active-Passive Mode Screen 351 26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 361 27.1 Overview 361 27.1.1 What You Can Do in this Chapter 361 27.2 User Summary Screen 363 27.3.1 Group Add/Edit Screen 364 27.3.1 Group Add/Edit Screen 366 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372		26.1.3 Before You Begin	
26.3 The Active-Passive Mode Screen 351 26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 User/Group 361 27.1 Overview 27.1 Overview 361 27.1.1 What You Can Do in this Chapter 361 27.2.1 What You Need To Know 361 27.2.2 User Summary Screen 363 27.3.1 Group Add/Edit Screen 364 27.3.1 Group Add/Edit Screen 366 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372		26.2 Device HA General	
26.3.1 Configuring Active-Passive Mode Device HA 353 26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 361 27.1 Overview 361 27.1.1 What You Can Do in this Chapter 361 27.2.2 What You Need To Know 361 27.2.1 User Summary Screen 363 27.3.1 Group Summary Screen 364 27.3.1 Group Add/Edit Screen 367 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372		26.3 The Active-Passive Mode Screen	
26.4 Configuring an Active-Passive Mode Monitored Interface 355 26.5 Device HA Technical Reference 356 Chapter 27 361 27.1 Overview 361 27.1.1 What You Can Do in this Chapter 361 27.1.2 What You Need To Know 361 27.2.1 User Summary Screen 363 27.3.1 Group Add/Edit Screen 366 27.3.1 Group Add/Edit Screen 367 27.4 The User/Group Setting Screen 368 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372		26.3.1 Configuring Active-Passive Mode Device HA	
26.5 Device HA Technical Reference 356 Chapter 27 361 27.1 Overview 361 27.1.1 What You Can Do in this Chapter 361 27.1.2 What You Need To Know 361 27.2 User Summary Screen 363 27.3 User Group Summary Screen 364 27.3.1 Group Add/Edit Screen 366 27.4 The User/Group Setting Screen 368 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372		26.4 Configuring an Active-Passive Mode Monitored Interface	
Chapter 27 User/Group36127.1 Overview36127.1.1 What You Can Do in this Chapter36127.1.2 What You Need To Know36127.2 User Summary Screen36327.2.1 User Add/Edit Screen36427.3 User Group Summary Screen36627.3.1 Group Add/Edit Screen36727.4 The User/Group Setting Screen36827.4.1 Default User Authentication Timeout Settings Edit Screens37027.5 User /Group Technical Reference372		26.5 Device HA Technical Reference	
User/Group36127.1 Overview36127.1.1 What You Can Do in this Chapter36127.1.2 What You Need To Know36127.2 User Summary Screen36327.2.1 User Add/Edit Screen36427.3 User Group Summary Screen36627.3.1 Group Add/Edit Screen36727.4 The User/Group Setting Screen36827.4.1 Default User Authentication Timeout Settings Edit Screens37027.4.2 User Aware Login Example37127.5 User /Group Technical Reference372	Chapte	er 27	
27.1 Overview36127.1.1 What You Can Do in this Chapter36127.1.2 What You Need To Know36127.2 User Summary Screen36327.2.1 User Add/Edit Screen36427.3 User Group Summary Screen36627.3.1 Group Add/Edit Screen36727.4 The User/Group Setting Screen36827.4.1 Default User Authentication Timeout Settings Edit Screens37027.4.2 User Aware Login Example37127.5 User /Group Technical Reference372	User/G	Group	
27.1.1 What You Can Do in this Chapter36127.1.2 What You Need To Know36127.2 User Summary Screen36327.2.1 User Add/Edit Screen36427.3 User Group Summary Screen36627.3.1 Group Add/Edit Screen36727.4 The User/Group Setting Screen36827.4.1 Default User Authentication Timeout Settings Edit Screens37027.4.2 User Aware Login Example37127.5 User /Group Technical Reference372		27.1 Overview	
27.1.2 What You Need To Know36127.2 User Summary Screen36327.2.1 User Add/Edit Screen36427.3 User Group Summary Screen36627.3.1 Group Add/Edit Screen36727.4 The User/Group Setting Screen36827.4.1 Default User Authentication Timeout Settings Edit Screens37027.4.2 User Aware Login Example37127.5 User /Group Technical Reference372		27.1.1 What You Can Do in this Chapter	361
27.2 User Summary Screen36327.2.1 User Add/Edit Screen36427.3 User Group Summary Screen36627.3.1 Group Add/Edit Screen36727.4 The User/Group Setting Screen36827.4.1 Default User Authentication Timeout Settings Edit Screens37027.4.2 User Aware Login Example37127.5 User /Group Technical Reference372		27.1.2 What You Need To Know	
27.2.1 User Add/Edit Screen		27.2 User Summary Screen	
27.3 User Group Summary Screen 366 27.3.1 Group Add/Edit Screen 367 27.4 The User/Group Setting Screen 368 27.4.1 Default User Authentication Timeout Settings Edit Screens 370 27.4.2 User Aware Login Example 371 27.5 User /Group Technical Reference 372		27.2.1 User Add/Edit Screen	
27.3.1 Group Add/Edit Screen36727.4 The User/Group Setting Screen36827.4.1 Default User Authentication Timeout Settings Edit Screens37027.4.2 User Aware Login Example37127.5 User /Group Technical Reference372		27.3 User Group Summary Screen	
 27.4 The User/Group Setting Screen		27.3.1 Group Add/Edit Screen	
27.4.1 Default User Authentication Timeout Settings Edit Screens		27.4 The User/Group Setting Screen	
27.4.2 User Aware Login Example		27.4.1 Default User Authentication Timeout Settings Edit Screens	
27.5 User /Group Technical Reference		27.4.2 User Aware Login Example	
		27.5 User /Group Technical Reference	

Chapter 28

Addresses	
28.1 Overview	
28.1.1 What You Can Do in this Chapter	
28.1.2 What You Need To Know	
28.2 Address Summary Screen	
28.2.1 IPv4 Address Add/Edit Screen	
28.2.2 IPv6 Address Add/Edit Screen	
28.3 Address Group Summary Screen	
28.3.1 Address Group Add/Edit Screen	
Chapter 29 Services	
29.1 Overview	
29.1.1 What You Can Do in this Chapter	
29.1.2 What You Need to Know	
29.2 The Service Summary Screen	
29.2.1 The Service Add/Edit Screen	
29.3 The Service Group Summary Screen	
29.3.1 The Service Group Add/Edit Screen	
Chapter 30 Schedules	
30.1 Overview	
30.1.1 What You Can Do in this Chapter	
30.1.2 What You Need to Know	
30.2 The Schedule Summary Screen	
30.2.1 The One-Time Schedule Add/Edit Screen	
30.2.2 The Recurring Schedule Add/Edit Screen	
Chapter 31 AAA Server	
31.1 Overview	
31.1.1 Directory Service (AD/LDAP)	
31.1.2 RADIUS Server	
31.1.3 ASAS	
31.1.4 What You Can Do in this Chapter	
31.1.5 What You Need To Know	
31.2 Active Directory or LDAP Server Summary	
31.2.1 Adding an Active Directory or LDAP Server	
31.3 RADIUS Server Summary	
31.3.1 Adding a RADIUS Server	

Chapte Auther	er 32 ntication Method	
	32.1 Overview	
	32.1.1 What You Can Do in this Chapter	
	32.1.2 Before You Begin	
	32.1.3 Example: Selecting a VPN Authentication Method	
	32.2 Authentication Method Objects	
	32.2.1 Creating an Authentication Method Object	
Chapte	er 33 cates	403
ocrain		
	33.1 Overview	
	33.1.1 What You Can Do in this Chapter	
	33.1.2 What You Need to Know	
	33.1.3 Verifying a Certificate	
	33.2 The My Certificates Screen	
	33.2.1 The My Certificates Add Screen	
	33.2.2 The My Certificates Edit Screen	
	33.2.3 The My Certificates Import Screen	
	33.3 The Trusted Certificates Screen	
	33.3.1 The Trusted Certificates Edit Screen	
	33.3.2 The Trusted Certificates Import Screen	
	33.4 Certificates Technical Reference	418
Chapte ISP Ac	er 34 ccounts	
	24.4 Overview	440
	34.1 What You Can Do in this Chapter	
	34.1.1 What You Can Do In this Chapter	
	34.2 ISP Account Summary	
Chapte	er 35	
SSL A	pplication	
	35.1 Overview	
	35.1.1 What You Can Do in this Chapter	
	35.1.2 What You Need to Know	
	35.1.3 Example: Specifying a Web Site for Access	
	35.2 The SSL Application Screen	
	35.2.1 Creating/Editing an SSL Application Object	
Chapte	er 36 v6	428
	36 1 Overview	

36.1.1 What You Can Do in this Chapter	
36.2 The DHCPv6 Request Screen	
36.2.1 DHCPv6 Request Add/Edit Screen	
36.3 The DHCPv6 Lease Screen	
36.3.1 DHCPv6 Lease Add/Edit Screen	
Chapter 37	
System	432
37.1 Overview	432
37.1.1 What You Can Do in this Chapter	
37.2 Host Name	
37.3 USB Storage	
37.4 Date and Time	
37.4.1 Pre-defined NTP Time Servers List	
37.4.2 Time Server Synchronization	
37.5 Console Port Speed	
37.6 DNS Overview	
37.6.1 DNS Server Address Assignment	
37.6.2 Configuring the DNS Screen	
37.6.3 Address Record	
37.6.4 PTR Record	
37.6.5 Adding an Address/PTR Record	
37.6.6 Domain Zone Forwarder	
37.6.7 Adding a Domain Zone Forwarder	
37.6.8 MX Record	
37.6.9 Adding a MX Record	
37.6.10 Adding a DNS Service Control Rule	
37.7 WWW Overview	
37.7.1 Service Access Limitations	
37.7.2 System Timeout	
37.7.3 HTTPS	
37.7.4 Configuring WWW Service Control	
37.7.5 Service Control Rules	
37.7.6 Customizing the WWW Login Page	
37.7.7 HTTPS Example	
37.8 SSH	
37.8.1 How SSH Works	
37.8.2 SSH Implementation on the ZyWALL	
37.8.3 Requirements for Using SSH	
37.8.4 Configuring SSH	
37.8.5 Secure Telnet Using SSH Examples	
37.9 Telnet	
37.9.1 Configuring Telnet	

37.10 FTP	
37.10.1 Configuring FTP	
37.11 SNMP	
37.11.1 Supported MIBs	
37.11.2 SNMP Traps	
37.11.3 Configuring SNMP	
37.12 Language Screen	
37.13 IPv6 Screen	472

Chapter 38 Log and Rep

og and Report	474
38.1 Overview	474
38.1.1 What You Can Do In this Chapter	
38.2 Email Daily Report	474
38.3 Log Setting Screens	
38.3.1 Log Setting Summary	
38.3.2 Edit System Log Settings	
38.3.3 Edit Log on USB Storage Setting	
38.3.4 Edit Remote Server Log Settings	
38.3.5 Log Category Settings Screen	

Chapter 39 File Manager

Manager	
39.1 Overview	
39.1.1 What You Can Do in this Chapter	
39.1.2 What you Need to Know	
39.2 The Configuration File Screen	
39.3 The Firmware Package Screen	
39.4 The Shell Script Screen	

Chapter 40

Diagnostics	
40.1 Overview	
40.1.1 What You Can Do in this Chapter	
40.2 The Diagnostic Screen	
40.2.1 The Diagnostics Files Screen	
40.3 The Packet Capture Screen	
40.3.1 The Packet Capture Files Screen	
40.4 Core Dump Screen	
40.4.1 Core Dump Files Screen	
40.5 The System Log Screen	

Chapter 41

41.1 Overview	507
41.1.1 What You Can Do in this Chapter	
41.2 The Routing Status Screen	
41.3 The SNAT Status Screen	
Chapter 42	
Reboot	514
42.1 Overview	514
42.1.1 What You Need To Know	514
42.2 The Reboot Screen	514
Chapter 43	
Shutdown	515
43.1 Overview	515
43.1.1 What You Need To Know	515
43.2 The Shutdown Screen	515
Chapter 44	
Troubleshooting	516
44.1 Resetting the ZyWALL	
44.2 Getting More Troubleshooting Help	
Appendix A Legal Information	
Index	

PART I User's Guide

Introduction

1.1 Overview

Note: This help covers the following ZyWALL models and refers to them all as "ZyWALL".

Features and interface names vary by model. Key feature differences between ZyWALL models are as follows. Other features are common to all models although features may vary slightly by model. See the specific product's datasheet for detailed specifications.

 Table 1
 Model-Specific Features

FEATURE	ZYWALL
Rack-mounting	110, 310, 1100
Wall-mounting	110
Port Role (see Section 7.2 on page 108)	110
Compact Flash Card Slot (not supported at the time of writing)	110

Here are some ZyWALL application scenarios.

IPv6 Routing

The ZyWALL supports IPv6 Ethernet, PPP, VLAN, and bridge routing. You may also create IPv6 policy routes and IPv6 objects. The ZyWALL can also route IPv6 packets through IPv4 networks using different tunneling methods.





VPN Connectivity

Set up VPN tunnels with other companies, branch offices, telecommuters, and business travelers to provide secure access to your network. You can also purchase the ZyWALL OTPv2 One-Time Password System for strong two-factor authentication for Web Configurator, Web access, SSL VPN, and ZyXEL IPSec VPN client user logins.





SSL VPN Network Access

SSL VPN lets remote users use their web browsers for a very easy-to-use VPN solution. A user just browses to the ZyWALL's web address and enters his user name and password to securely connect to the ZyWALL's network. Here full tunnel mode creates a virtual connection for a remote user and gives him a private IP address in the same subnet as the local network so he can access network resources in the same way as if he were part of the internal network.





User-Aware Access Control

Set up security policies to restrict access to sensitive information and shared resources based on the user who is trying to access it. In the following figure user **A** can access both the Internet and an internal file server. User **B** has a lower level of access and can only access the Internet. User **C** is not even logged in and cannot access either.



Load Balancing

Set up multiple connections to the Internet on the same port, or different ports, including cellular interfaces. In either case, you can balance the traffic loads between them.

Figure 5 Applications: Multiple WAN Interfaces



1.2 Management Overview

You can manage the ZyWALL in the following ways.

Web Configurator

The Web Configurator allows easy ZyWALL setup and management using an Internet browser. This User's Guide provides information about the Web Configurator.

								- 1	Provider
C Wine Device									
	<u> 1900</u>	R	P	n min si	4	Ś0			
		10000						_	
Autor Days .			ALC: N			-		-	
the law it	and interest of							-	
testing 1	0.0000		-						
the later large \$	100 (00 He HC R) - 50 (00 HC HC HC HC								
Annal States 1	And the second second second			-	79			-	
		1444	2.00			_		_	1111
hore- upting			4.1.74		-	train .	interest in the second	1	
Low Intelline	ADDRESS AND ADDRESS ADDRES		1.1	****		74-1610	control aut		
in Sele			P			100-0.8	NARAMEN,		
SO tain	1. (1. (1. (1. (1. (1. (1. (1. (1. (1. (1.1				100 to 10, pp .		
Greek upplicate	Anno (Anno) (2010)		0.0		-		mether ent_pt.		
faire of payments			· · · ·		_				_
bet Data			-	the line has					+ 4 5 4
Autors Reported			100	-		-	P 141-191	-	-
194 Barget			-	-		-	121999 000000		(Ref)
-	1.1		-				1002 1002	2429.088	1000
Burning Strapt			-	544		141	4810-5886	den .	-
and the second s	24 M		-	-		100		-	
Read Surger			1.2	- 2	1.2	-	COLUMN TWO IS NOT	-	2
and the second s	1.8.9			1.00	-	-	10.001100.00.000	-	-
take the same linese									
	0.5.00								
And in Fernand									
The second second	And comments								
-									
Colorest Sta									
Careto Sel	Desire	the second se							
		- 100							
1001									

Figure 6 Managing the ZyWALL: Web Configurator

Command-Line Interface (CLI)

The CLI allows you to use text-based commands to configure the ZyWALL. Access it using remote management (for example, SSH or Telnet) or via the physical or Web Configurator console port. See the Command Reference Guide for CLI details. The default settings for the console port are:

inas

SETTING	VALUE
Speed	115200 bps
Data Bits	8
Parity	None
Stop Bit	1
Flow Control	Off

1.3 Web Configurator

In order to use the Web Configurator, you must:

- Use one of the following web browser versions or later: Internet Explorer 7, Firefox 3.5, Chrome 9.0
- Allow pop-up windows (blocked by default in Windows XP Service Pack 2)
- · Enable JavaScripts, Java permissions, and cookies

The recommended screen resolution is 1024 x 768 pixels.

1.3.1 Web Configurator Access

- 1 Make sure your ZyWALL hardware is properly connected. See the Quick Start Guide.
- 2 In your browser go to http://192.168.1.1. By default, the ZyWALL automatically routes this request to its HTTPS server, and it is recommended to keep this setting. The **Login** screen appears.

	A CONTRACTOR OF
User Name:	
Password:	
One-Time Password:	(Optional)
(max. 63 alphanumeric, printab	le characters and no spaces)
	Login SSL VPN

3 Type the user name (default: "admin") and password (default: "1234").

If you have a OTP (One-Time Password) token generate a number and enter it in the **One-Time Password** field. The number is only good for one login. You must use the token to generate a new number the next time you log in.

- 4 Click Login. If you logged in using the default user name and password, the Update Admin Info screen appears. Otherwise, the dashboard appears.
- 5 Follow the directions in the Update Admin Info screen. If you change the default password, the Login screen appears after you click Apply. If you click Ignore, the Installation Setup Wizard opens if the ZyWALL is using its default configuration; otherwise the dashboard appears.

SHBOARD												
Sirtual Device											* 0	12 2
2	Rear Panel	PwR 575	en samme en				ŕ					
Device Information		_	×0 # X	📥 Top S IPv4	Firewall Rul	les that Blocked T	affic	_	_	_		1
System Name: <u>rywall-usg110</u> Model Name: ZyWALL 110				# From		То		Description	+	ilts		
Serial Number: \$122L23030003	80.82.00.40.46.04			h Top 5 IPv6	Firewall Rul	les that Blocked T	raffic				A 0	2 2
MAC Address Range: D0:02:02:09:A5:PE Firmware Version: V3.10/AAAA.01Preb7-	r38094 / 1.03 / 2013-01-24 06:27:52			# From		То		Description	+	lits		
System Status			* 0 # X	D The Latest	Alert Logs						A 0	0.0
System Uptime:	06:26:49			# Time		Priority	Category	Message	Source	De	stination	
Current Date/Time:	2013-01-29 / 09:45:18 GMT+00:00			1 2013-01-2	29 03:18:03	alert	file-manage	WARNING #conf				
VPN Status:	9			2 2013-01-2	29 03:17:57	alert	policy-route	Interface ontinen				
DHCP Table:	9			4 2013-01-0	9 03:17:57	alert	policy-route	interface war2 on				
Current Login User:	aumin (unimitiau / 00:29:39)			5 2013-01-0	29 03:17:57	alert	policy-route	interface wan1_pp_				
Boot Status:	≏ OK			0			_				1.1	
				W Interface S	tatus Summ	ary	1225	I water and the second			E an	
O System Resources			* 0 0 X	want	Status 100M/Evil	HA Status	Zone	172 23 30 25 / 255 255 1	255.0 0	P Assignment	Action	
CPU Usage					Davia		IN AN				(Nenew)	
0.48				warlz	Dawa		0.07	000010000		ande cilent	Renew	
Memory Usage				opt lan1	Down	n/a	LANI	192 168 1 1 / 255 255 2	550 9	osoiC tatlic	n/a	
20 %				lan2	Down	n/a	LAN2	192.168.2.1 / 255.255.2	55.0 5	tatic	n/a	
Flash Usage				ext-wlan	Down	n/a	WLAN	10.59 0 1 / 255 255 255	.0 5	tatic	n/a	
8 %		- 3		dmz	Down	n/a	DMZ	192.168.3.1 / 255.255.2	55.0 S	tatic	n/a	
USB Storage Usage											_	
0/0 MB												
Active Sessions			1.									
141/60000												
Extension Slot			* 0 ¢ X									

1.3.2 Web Configurator Screens Overview

The Web Configurator screen is divided into these parts (as illustrated on page 22):

- A title bar
- B navigation panel
- C main window



The title bar icons in the upper right corner provide the following functions.

LABEL	DESCRIPTION
Logout	Click this to log out of the Web Configurator.
Help	Click this to open the help page for the current screen.
About	Click this to display basic information about the ZyWALL.
Site Map	Click this to see an overview of links to the Web Configurator screens.
Object Reference	Click this to check which configuration items reference an object.
Console	Click this to open a Java-based console window from which you can run command line interface (CLI) commands. You will be prompted to enter your user name and password. See the Command Reference Guide for information about the commands.
CLI	Click this to open a popup window that displays the CLI commands sent by the Web Configurator to the ZyWALL.

 Table 3
 Title Bar: Web Configurator Icons

About

Click **About** to display basic information about the ZyWALL.

Figure 8 About

ZyXEL	ZyWALL
	Did you check <u>www.zyxel.com</u> today?
Boot Module: 1.03	
Current Version: V3.10(/	AAAA.0)Preb7-r38094
Released Date: 2013-01	-24 06:27:52
	ОК

Table 4 About

LABEL	DESCRIPTION
Boot Module	This shows the version number of the software that handles the booting process of the ZyWALL.
Current Version	This shows the firmware version of the ZyWALL.
Released Date	This shows the date (yyyy-mm-dd) and time (hh:mm:ss) when the firmware is released.
ОК	Click this to close the screen.

Site Map

Click **Site MAP** to see an overview of links to the Web Configurator screens. Click a screen's link to go to that screen.

Site Map			? >
Monitor			
System Status o Port Statistics interface Status o Traffic Statustics o Session Monitor o IGMP Statustics o DDNS Status o IP/MAC Binding o Login Users o Cellular Status o USB Storage	VPN Monitor o I <u>PSec</u> o <u>SSL</u> o <u>LZTP over IPSec</u>	Log	
Configuration			

Figure 9 Site Map

Object Reference

Click **Object Reference** to open the **Object Reference** screen. Select the type of object and the individual object and click **Refresh** to show which configuration settings reference the object.

Figure 10 Object Reference

Object Type:	Please select	t one	✓ Object Name:	Please select one	*
# Servic	e	Priority	Name	Description	
14 4 Pa	ge 1 of 1 🖡	Show 50	✓ items	No data	to display

The fields vary with the type of object. This table describes labels that can appear in this screen.

LABEL	DESCRIPTION
Object Name	This identifies the object for which the configuration settings that use it are displayed. Click the object's name to display the object's configuration screen in the main window.
#	This field is a sequential value, and it is not associated with any entry.
Service	This is the type of setting that references the selected object. Click a service's name to display the service's configuration screen in the main window.
Priority	If it is applicable, this field lists the referencing configuration item's position in its list, otherwise N/A displays.

 Table 5
 Object References

LABEL	DESCRIPTION
Name	This field identifies the configuration item that references the object.
Description	If the referencing configuration item has a description configured, it displays here.
Refresh	Click this to update the information in this screen.
Cancel	Click Cancel to close the screen.

Table 5 Object References (continued)

Console

Click **Console** to open a Java-based console window from which you can run CLI commands. You will be prompted to enter your user name and password. See the Command Reference Guide for information about the commands.





CLI Messages

Click **CLI** to look at the CLI commands sent by the Web Configurator. Open the pop-up window and then click some menus in the web configurator to dislay the corresponding commands.

Figure 12 CLI Messages

•	0	
CLI		
🎁 Clear		
[0] show port status		~
[1] show system uptime		
### CLI End		
[0] show port status		
[1] show system uptime		
### CLI End		
[0] show port status		
[1] show system uptime		
### CLI End		~
Response		

1.3.3 Navigation Panel

Use the navigation panel menu items to open status and configuration screens. Click the arrow in the middle of the right edge of the navigation panel to hide the panel or drag to resize it. The following sections introduce the ZyWALL's navigation panel menus and their screens.



Figure 13 Navigation Panel

Dashboard

The dashboard displays general device information, system status, system resource usage,, and interface status in widgets that you can re-arrange to suit your needs. See the Web Help for details on the dashboard.

Monitor Menu

The monitor menu screens display status and statistics information.

FOLDER OR LINK	ТАВ	FUNCTION
System Status		
Port Statistics		Displays packet statistics for each physical port.
Interface Status		Displays general interface information and packet statistics.
Traffic Statistics		Collect and display traffic statistics.
Session Monitor		Displays the status of all current sessions.
DDNS Status		Displays the status of the ZyWALL's DDNS domain names.
IP/MAC Binding		Lists the devices that have received an IP address from ZyWALL interfaces using IP/MAC binding.
Login Users		Lists the users currently logged into the ZyWALL.

 Table 6
 Monitor Menu Screens Summary

FOLDER OR LINK	ТАВ	FUNCTION
Cellular Status		Displays details about the ZyWALL's 3G connection status.
USB Storage		Displays details about USB device connected to the ZyWALL.
VPN Monitor		
IPSec		Displays and manages the active IPSec SAs.
SSL		Lists users currently logged into the VPN SSL client portal. You can also log out individual users and delete related session information.
L2TP over IPSec		Displays details about current L2TP sessions.
Log		Lists log entries.

 Table 6
 Monitor Menu Screens Summary (continued)

Configuration Menu

Use the configuration menu screens to configure the ZyWALL's features.

FOLDER OR LINK	ТАВ	FUNCTION
Quick Setup		Quickly configure WAN interfaces or VPN connections.
Network		
Interface	Port Role	Use this screen to set the ZyWALL's flexible ports as LAN1, WLAN, or DMZ.
	Ethernet	Manage Ethernet interfaces and virtual Ethernet interfaces.
	PPP	Create and manage PPPoE and PPTP interfaces.
	Cellular	Configure a cellular Internet connection for an installed 3G card.
	Tunnel	Configure tunneling between IPv4 and IPv6 networks.
	VLAN	Create and manage VLAN interfaces and virtual VLAN interfaces.
	Bridge	Create and manage bridges and virtual bridge interfaces.
	Trunk	Create and manage trunks (groups of interfaces) for load balancing.
Routing	Policy Route	Create and manage routing policies.
	Static Route	Create and manage IP static routing information.
	RIP	Configure device-level RIP settings.
	OSPF	Configure device-level OSPF settings, including areas and virtual links.
Zone		Configure zones used to define various policies.
DDNS	DDNS	Define and manage the ZyWALL's DDNS domain names.
NAT		Set up and manage port forwarding rules.
HTTP Redirect		Set up and manage HTTP redirection rules.
ALG		Configure SIP, H.323, and FTP pass-through settings.
IP/MAC Binding	Summary	Configure IP to MAC address bindings for devices connected to each supported interface.
	Exempt List	Configure ranges of IP addresses to which the ZyWALL does not apply IP/MAC binding.
DNS Inbound LB	DNS Load Balancing	Configure DNS Load Balancing.
Auth. Policy		Define rules to force user authentication.

Table 7 Configuration Menu Screens Summary

FOLDER OR LINK	ТАВ	FUNCTION
Firewall	Firewall	Create and manage level-3 traffic rules.
	Session Control	Limit the number of concurrent client NAT/firewall sessions.
VPN		
IPSec VPN	VPN Connection	Configure IPSec tunnels.
	VPN Gateway	Configure IKE tunnels.
	Concentrator	Combine IPSec VPN connections into a single secure network
	Configuration Provisioning	Set who can retrieve VPN rule settings from the ZyWALL using the ZyWALL IPSec VPN Client.
SSL VPN	Access Privilege	Configure SSL VPN access rights for users and groups.
	Global Setting	Configure the ZyWALL's SSL VPN settings that apply to all connections.
L2TP VPN	L2TP VPN	Configure L2TP over IPSec tunnels.
BWM	BWM	Enable and configure bandwidth management rules.
Device HA	General	Configure device HA global settings, and see the status of each interface monitored by device HA.
	Active-Passive Mode	Configure active-passive mode device HA.
Object		
User/Group	User	Create and manage users.
	Group	Create and manage groups of users.
	Setting	Manage default settings for all users, general settings for user sessions, and rules to force user authentication.
Address	Address	Create and manage host, range, and network (subnet) addresses.
	Address Group	Create and manage groups of addresses.
Service	Service	Create and manage TCP and UDP services.
	Service Group	Create and manage groups of services.
Schedule	Schedule	Create one-time and recurring schedules.
AAA Server	Active Directory	Configure the Active Directory settings.
	LDAP	Configure the LDAP settings.
	RADIUS	Configure the RADIUS settings.
Auth. Method	Authentication Method	Create and manage ways of authenticating users.
Certificate	My Certificates	Create and manage the ZyWALL's certificates.
	Trusted Certificates	Import and manage certificates from trusted sources.
ISP Account	ISP Account	Create and manage ISP account information for PPPoE/PPTP interfaces.
SSL Application		Create SSL web application objects.
DHCPv6	Request	Configure IPv6 DHCP request type and interface information.
	Lease	Configure IPv6 DHCP lease type and interface information.
System		
Host Name		Configure the system and domain name for the ZyWALL.
USB Storage	Settings	Configure the settings for the connected USB devices.
Date/Time		Configure the current date, time, and time zone in the ZyWALL.

 Table 7
 Configuration Menu Screens Summary (continued)

FOLDER OR LINK	ТАВ	FUNCTION
Console Speed		Set the console speed.
DNS		Configure the DNS server and address records for the ZyWALL.
WWW	Service Control	Configure HTTP, HTTPS, and general authentication.
	Login Page	Configure how the login and access user screens look.
SSH		Configure SSH server and SSH service settings.
TELNET		Configure telnet server settings for the ZyWALL.
FTP		Configure FTP server settings.
SNMP		Configure SNMP communities and services.
Language		Select the Web Configurator language.
IPv6		Enable IPv6 globally on the ZyWALL here.
Log & Report		
Email Daily Report		Configure where and how to send daily reports and what reports to send.
Log Settings		Configure the system log, e-mail logs, and remote syslog servers.

 Table 7
 Configuration Menu Screens Summary (continued)

Maintenance Menu

Use the maintenance menu screens to manage configuration and firmware files, run diagnostics, and reboot or shut down the ZyWALL.

FOLDER OR LINK	ТАВ	FUNCTION
File	Configuration File	Manage and upload configuration files for the ZyWALL.
Manager	Firmware Package	View the current firmware version and to upload firmware.
	Shell Script	Manage and run shell script files for the ZyWALL.
Diagnostics	Diagnostic	Collect diagnostic information.
	Packet Capture	Capture packets for analysis.
	Core Dump	Connect a USB device to the ZyWALL and save the ZyWALL operating system kernel to it here.
	System Log	Connect a USB device to the ZyWALL and archive the ZyWALL system logs to it here.
Packet	Routing Status	Check how the ZyWALL determines where to route a packet.
Explore	SNAT Status	View a clear picture on how the ZyWALL converts a packet's source IP address and check the related settings.
Reboot		Restart the ZyWALL.
Shutdown		Turn off the ZyWALL.

Table 8 Maintenance Menu Screens Summary

1.3.4 Tables and Lists

Web Configurator tables and lists are flexible with several options for how to display their entries.

Click a column heading to sort the table's entries according to that column's criteria.



С	onfigural	ion	
	📀 Add	🖉 Edit 🏢 Remove 📴 Object References	
	# 🤇	User Name 🔺	Description
	4	ad-users	External AD Users
	1	admin	Administration account

Click the down arrow next to a column heading for more options about how to display the entries. The options available vary depending on the type of fields in the column. Here are some examples of what you can do:

- · Sort in ascending or descending (reverse) alphabetical order
- · Select which columns to display
- Group entries by field
- Show entries in groups
- Filter by mathematical operators (<, >, or =) or searching for text

Figure 15 Common Table Column Options

Co	nfigurati	on						
	📀 Add	🕜 Edit 🍵 Remove 📴 Object R	efere	nces				
	#	User Name 🔺	- C	escription				
	4	ad-users	Az↓	Sort Ascending				
	1	admin	Z↓	Sort Descending				
	5	guest			-			
	2	ldap-users		Columns		V	# _h_	
	3	radius-users		Group By This Field		1	User Name	
	14 4 1	Page 1 of 1 > > Show		Show in Groups		V	Description	Displaying 1 - 5 of 5
				Filters				

Select a column heading cell's right border and drag to re-size the column.

Figure 16 Resizing a Table Column

Configuration	Configuration			
🗿 Add 📝	Edit 💼 Remo	ve 🔚 Object References		
#	User Name 🔺	Description		
4	ad-users	€xternal AD Users		
1	admin	Administration account		

Select a column heading and drag and drop it to change the column order. A green check mark displays next to the column's title when you drag the column to a valid new location.

Figure 17 Moving Columns

Configu	Configuration			
0	Add 📝 Edit 🍵 Remove 🖷 O	bject References		
#	User Name 🔺	Description		
4	ad-users	External AD Users		
1	admin Description	Administration account		
5	quest	Local User		

Use the icons and fields at the bottom of the table to navigate to different pages of entries and control how many entries display at a time.

Figure 18 Navigating Pages of Table Entries



The tables have icons for working with table entries. You can often use the [Shift] or [Ctrl] key to select multiple entries to remove, activate, or deactivate.

Figure 19 Common Table Icons

onfigu	uration			
O #	Add 📝 Edit 🍵 Remove	: 💡 Activate 💡 Inactivate	🍓 Connect 🚷 Disconnect [🛅 Object References
#	Status Name	VPN Gateway	Encapsulation Algorithm	Policy
1	9 😣		TRANSPORT 3DES/SHA 3DES	SME /
M		🕨 🗌 Show 50 🗸 items		Displaying 1 - 1 of :

Here are descriptions for the most common table icons.

Table 9 Common Table Icons

LABEL	DESCRIPTION
Add	Click this to create a new entry. For features where the entry's position in the numbered list is important (features where the ZyWALL applies the table's entries in order like the firewall for example), you can select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings. In some tables you can just click a table entry and edit it directly in the table. For those types of tables small red triangles display for table entries with changes that you have not yet applied.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Connect	To connect an entry, select it and click Connect .
Disconnect	To disconnect an entry, select it and click Disconnect .
Object References	Select an entry and click Object References to check which settings use the entry.
Move	To change an entry's position in a numbered list, select it and click Move to display a field to type a number for where you want to put that entry and press [ENTER] to move the entry to the number that you typed. For example, if you type 6, the entry you are moving becomes number 6 and the previous entry 6 (if there is one) gets pushed up (or down) one.

Working with Lists

When a list of available entries displays next to a list of selected entries, you can often just doubleclick an entry to move it from one list to the other. In some lists you can also use the [Shift] or [Ctrl] key to select multiple entries, and then use the arrow button to move them to the other list.

Installation Setup Wizard

2.1 Installation Setup Wizard Screens

When you log into the Web Configurator for the first time or when you reset the ZyWALL to its default configuration, the **Installation Setup Wizard** screen displays. This wizard helps you configure Internet connection settings and activate subscription services. This chapter provides information on configuring the Web Configurator's installation setup wizard. See the feature-specific chapters in this User's Guide for background information.

Tistallation Setup	Wizard	×
	Installation Setup Wizard	~
11	Internet Access > Internet Access Succeed > Device Registration	
	Welcome	
	The later steps will guide you to setup the Internet connection. - Connect to Internet - Device Registration	
/3/14/1	Click 'Next' to start the wizard; or 'Go to Dashboard' if you want to skip.	
	Go to Dashboard Next >	

Figure 20 Installation Setup Wizard

- Click the double arrow in the upper right corner to display or hide the help.
- Click **Go to Dashboard** to skip the installation setup wizard or click **Next** to start configuring for Internet access.

2.1.1 Internet Access Setup - WAN Interface

Use this screen to configure the WAN interface's type of encapsulation and method of IP address assignment.

The screens vary depending on the encapsulation type. Refer to information provided by your ISP to know what to enter in each field. Leave a field blank if you don't have that information.

Note: Enter the Internet access information exactly as your ISP gave it to you.

• **Encapsulation**: Choose the **Ethernet** option when the WAN port is used as a regular Ethernet. Otherwise, choose **PPPoE** or **PPTP** for a dial-up connection according to the information from your ISP.

- WAN Interface: This is the interface you are configuring for Internet access.
- Zone: This is the security zone to which this interface and Internet connection belong.
- **IP Address Assignment**: Select **Auto** if your ISP did not assign you a fixed IP address. Select **Static** if the ISP assigned a fixed IP address.

2.1.2 Internet Access: Ethernet

This screen is read-only if you set the previous screen's **IP Address Assignment** field to **Auto**. Use this screen to configure your IP address settings.

Note: Enter the Internet access information exactly as given to you by your ISP.

- Encapsulation: This displays the type of Internet connection you are configuring.
- First WAN Interface: This is the number of the interface that will connect with your ISP.
- Zone: This is the security zone to which this interface and Internet connection will belong.
- IP Address: Enter your (static) public IP address. Auto displays if you selected Auto as the IP Address Assignment in the previous screen.

The following fields display if you selected static IP address assignment.

- IP Subnet Mask: Enter the subnet mask for this WAN connection's IP address.
- **Gateway IP Address**: Enter the IP address of the router through which this WAN connection will send traffic (the default gateway).
- First / Second DNS Server: These fields display if you selected static IP address assignment. The Domain Name System (DNS) maps a domain name to an IP address and vice versa. Enter a DNS server's IP address(es). The DNS server is extremely important because without it, you must know the IP address of a computer before you can access it. The ZyWALL uses these (in the order you specify here) to resolve domain names for VPN, DDNS and the time server. Leave the field as 0.0.0.0 if you do not want to configure DNS servers.

2.1.3 Internet Access: PPPoE

Note: Enter the Internet access information exactly as given to you by your ISP.

2.1.3.1 ISP Parameters

- Type the PPPoE Service Name from your service provider. PPPoE uses a service name to identify and reach the PPPoE server. You can use alphanumeric and -_@\$./ characters, and it can be up to 64 characters long.
- Authentication Type Select an authentication protocol for outgoing connection requests. Options are:
 - CHAP/PAP Your ZyWALL accepts either CHAP or PAP when requested by the remote node.
 - CHAP Your ZyWALL accepts CHAP only.
 - **PAP** Your ZyWALL accepts PAP only.
 - MSCHAP Your ZyWALL accepts MSCHAP only.
 - MSCHAP-V2 Your ZyWALL accepts MSCHAP-V2 only.
- Type the **User Name** given to you by your ISP. You can use alphanumeric and -_@\$./ characters, and it can be up to 31 characters long.

- Type the **Password** associated with the user name. Use up to 64 ASCII characters except the [] and ?. This field can be blank.
- Select **Nailed-Up** if you do not want the connection to time out. Otherwise, type the **Idle Timeout** in seconds that elapses before the router automatically disconnects from the PPPoE server.

2.1.3.2 WAN IP Address Assignments

- WAN Interface: This is the name of the interface that will connect with your ISP.
- **Zone:** This is the security zone to which this interface and Internet connection will belong.
- **IP Address**: Enter your (static) public IP address. **Auto** displays if you selected **Auto** as the **IP Address Assignment** in the previous screen.
- First / Second DNS Server: These fields display if you selected static IP address assignment. The Domain Name System (DNS) maps a domain name to an IP address and vice versa. Enter a DNS server's IP address(es). The DNS server is extremely important because without it, you must know the IP address of a computer before you can access it. The ZyWALL uses these (in the order you specify here) to resolve domain names for VPN, DDNS and the time server. Leave the field as 0.0.0.0 if you do not want to configure DNS servers. If you do not configure a DNS server, you must know the IP address of a machine in order to access it.

2.1.4 Internet Access: PPTP

Note: Enter the Internet access information exactly as given to you by your ISP.

2.1.5 ISP Parameters

- Authentication Type Select an authentication protocol for outgoing calls. Options are:
 - CHAP/PAP Your ZyWALL accepts either CHAP or PAP when requested by the remote node.
 - CHAP Your ZyWALL accepts CHAP only.
 - **PAP** Your ZyWALL accepts PAP only.
 - MSCHAP Your ZyWALL accepts MSCHAP only.
 - MSCHAP-V2 Your ZyWALL accepts MSCHAP-V2 only.
- Type the **User Name** given to you by your ISP. You can use alphanumeric and -_@\$./ characters, and it can be up to 31 characters long.
- Type the **Password** associated with the user name. Use up to 64 ASCII characters except the [] and ?. This field can be blank. Re-type your password in the next field to confirm it.
- Select **Nailed-Up** if you do not want the connection to time out. Otherwise, type the **Idle Timeout** in seconds that elapses before the router automatically disconnects from the PPTP server.

2.1.5.1 PPTP Configuration

- **Base Interface**: This identifies the Ethernet interface you configure to connect with a modem or router.
- Type a Base IP Address (static) assigned to you by your ISP.
- Type the IP Subnet Mask assigned to you by your ISP (if given).
- Server IP: Type the IP address of the PPTP server.

• Type a **Connection ID** or connection name. It must follow the "c:id" and "n:name" format. For example, C:12 or N:My ISP. This field is optional and depends on the requirements of your broadband modem or router. You can use alphanumeric and -_: characters, and it can be up to 31 characters long.

2.1.5.2 WAN IP Address Assignments

- First WAN Interface: This is the connection type on the interface you are configuring to connect with your ISP.
- Zone This is the security zone to which this interface and Internet connection will belong.
- **IP Address**: Enter your (static) public IP address. Auto displays if you selected **Auto** as the **IP Address Assignment** in the previous screen.
- First / Second DNS Server: These fields display if you selected static IP address assignment. The Domain Name System (DNS) maps a domain name to an IP address and vice versa. Enter a DNS server's IP address(es). The DNS server is extremely important because without it, you must know the IP address of a computer before you can access it. The ZyWALL uses these (in the order you specify here) to resolve domain names for VPN, DDNS and the time server. Leave the field as 0.0.0.0 if you do not want to configure DNS servers.

2.1.6 Internet Access - Finish

You have set up your ZyWALL to access the Internet. A screen displays with your settings. If they are not correct, click **Back**.
Hardware Introduction

3.1 Default Zones, Interfaces, and Ports

The default configurations for zones, interfaces, and ports are as follows. References to interfaces may be generic rather than the specific name used in your model. For example, this guide may use "the WAN interface" rather than "wan1" or "wan2", "ge2" or" ge3".

An OPT (optional) Ethernet port can be configured as an additional WAN port, LAN, WLAN, or DMZ port.





1100

Note: Use an 8-wire Ethernet cable to run your Gigabit Ethernet at 1000 Mbps. Using a 4wire Ethernet cable limits your connection to 100 Mbps. Note that the connection speed also depends on what the Ethernet device at the other end can support.

3.2 Stopping the ZyWALL

Always use **Maintenance > Shutdown > Shutdown** or the shutdown command before you turn off the ZyWALL or remove the power. Not doing so can cause the firmware to become corrupt.

3.3 Rack-mounting

See Chapter 1 on page 18 for the ZyWALL models that can be rack mounted. Use the following steps to mount the ZyWALL on an EIA standard size, 19-inch rack or in a wiring closet with other equipment using a rack-mounting kit. Make sure the rack will safely support the combined weight of all the equipment it contains and that the position of the ZyWALL does not make the rack unstable or top-heavy. Take all necessary precautions to anchor the rack securely before installing the unit.

Note: Leave 10 cm of clearance at the sides and 20 cm in the rear.

Use a #2 Phillips screwdriver to install the screws.

Note: Failure to use the proper screws may damage the unit.

- 1 Align one bracket with the holes on one side of the ZyWALL and secure it with the included bracket screws (smaller than the rack-mounting screws).
- 2 Attach the other bracket in a similar fashion.



3 After attaching both mounting brackets, position the ZyWALL in the rack and up the bracket holes with the rack holes. Secure the ZyWALL to the rack with the rack-mounting screws.



3.4 Wall-mounting

See Chapter 1 on page 18 for the ZyWALL models that can be wall-mounted. Do the following to attach your ZyWALL to a wall.

1 Screw two screws with 6 mm ~ 8 mm (0.24" ~ 0.31") wide heads into the wall 150 mm apart (see the figure in step 2). Do not screw the screws all the way in to the wall; leave a small gap between the head of the screw and the wall.

The gap must be big enough for the screw heads to slide into the screw slots and the connection cables to run down the back of the ZyWALL.

- Note: Make sure the screws are securely fixed to the wall and strong enough to hold the weight of the ZyWALL with the connection cables.
- 2 Use the holes on the bottom of the ZyWALL to hang the ZyWALL on the screws.

3.5 Front Panel LEDs

This section introduces the ZyWALL's front panel LEDs.

Figure 21 ZyWALL Front Panel



310



The following tables describe the LEDs.

LED	COLOR	STATUS	DESCRIPTION
PWR		Off	The ZyWALL is turned off.
	Green	On	The ZyWALL is turned on.
	Red	On	There is a hardware component failure. Shut down the device, wait for a few minutes and then restart the device (see Section 3.2 on page 38). If the LED turns red again, then please contact your vendor.
SYS	Green	Off	The ZyWALL is not ready or has failed.
		On	The ZyWALL is ready and running.
		Blinking	The ZyWALL is booting.
	Red	On	The ZyWALL xd an error or has failed.
USB	Green	Off	No device is connected to the ZyWALL's USB port or the connected device is not supported by the ZyWALL.
		On	A 3G USB card or USB storage device is connected to the USB port.
	Orange	On	Connected to a 3G network through the connected 3G USB card.
P1, P2	Green	Off	There is no traffic on this port.
		Blinking	The ZyWALL is sending or receiving packets on this port.
	Orange	Off	There is no connection on this port.
		On	This port has a successful link.

Table 10Front Panel LEDs

3.5.1 Rear Panels



The following graphic shows the rear panel of the $\ensuremath{\mathsf{ZyWALL}}$.

Table 11 Rear Panel

LABEL	DESCRIPTION
Console	You can use the console port to manage the ZyWALL using CLI commands. You will be prompted to enter your user name and password. See the Command Reference Guide for more information about the CLI.
	When configuring using the console port, you need a computer equipped with communications software configured to the following parameters:
	Speed 115200 bps
	Data Bits 8
	Parity None
	Stop Bit 1
	Flow Control Off
CF Card Slot	This feature is not supported at the time of writing.
Power	Use the included power cord to connect the power socket to a power outlet. Turn the power switch on if your ZyWALL has a power switch.
Lock	Attach a lock-and-cable from the Kensington lock (the small, metal-reinforced, oval hole) to a permanent object, such as a pole, to secure the ZyWALL in place.
Fan	The fans are for cooling the ZyWALL. Make sure they are not obstructed to allow maximum ventilation.

Quick Setup Wizards

4.1 Quick Setup Overview

The Web Configurator's quick setup wizards help you configure Internet and VPN connection settings. This chapter provides information on configuring the quick setup screens in the Web Configurator. See the feature-specific chapters in this User's Guide for background information.

In the Web Configurator, click **Configuration > Quick Setup** to open the first **Quick Setup** screen.

Figure 22 Quick Setup

Quick Setup	×
WAN Interface	VPN Setup
WAN Quick Setting walks you through the steps of getting your device connected online.	For creating secure communications between nodes, VPN Quick Setting provides a simplified process to do that.

WAN Interface

Click this link to open a wizard to set up a WAN (Internet) connection. This wizard creates matching ISP account settings in the ZyWALL if you use PPPoE or PPTP. See Section 4.2 on page 42.

VPN SETUP

Use **VPN Setup** to configure a VPN (Virtual Private Network) rule for a secure connection to another computer or network. **Use VPN Settings for Configuration Provisioning** to set up a VPN rule that can be retrieved with the ZyWALL IPSec VPN Client. You only need to enter a user name, password and the IP address of the ZyWALL in the ZyWALL IPSec VPN Client to get all VPN settings automatically from the ZyWALL. See Section 4.3 on page 47.

4.2 WAN Interface Quick Setup

Click **WAN Interface** in the main **Quick Setup** screen to open the **WAN Interface Quick Setup Wizard Welcome** screen. Use these screens to configure an interface to connect to the Internet. Click **Next**.

Figure 2	Interface	Ouick	Setun	Wizard	
I Iguie Z.	IIIICHACE	QUICK	Jeiup	vvizaru	

Welcome	
The later steps will guide you to setup the Internet connection: - Choose Ethernet - Enter WAN Settings - WAN Configuration Summary	
Click "Next" to start.	
	Next >

4.2.1 Choose an Ethernet Interface

Select the Ethernet interface that you want to configure for a WAN connection and click Next.

Figure 24 Choose an Ethernet Interface

Ti Quick Setup		×
	WAN Interface	
	Choose Ethernet > Enter WAN Settings > WAN Configuration Summary	
II		
E	Ethernet	
R = 13	Ethernet Selection:	
E /		
/		
7		
1.000		
		< Back Next >

4.2.2 Select WAN Type

WAN Type Selection: Select the type of encapsulation this connection is to use. Choose **Ethernet** when the WAN port is used as a regular Ethernet.

Otherwise, choose **PPPoE** or **PPTP** for a dial-up connection according to the information from your ISP.

Figure 25 W/	AN Interface	Setup:	Step 2
--------------	--------------	--------	--------

.g		
WAN Interface Setup		
WAN Type Selection:	Ethernet 🗸	
	Back Next >	

The screens vary depending on what encapsulation type you use. Refer to information provided by your ISP to know what to enter in each field. Leave a field blank if you don't have that information.

Note: Enter the Internet access information exactly as your ISP gave it to you.

4.2.3 Configure WAN Settings

Use this screen to select whether the interface should use a fixed or dynamic IP address.

i igule zo	WAN IIIteria	Je Selup. Slep z		
Quick Setup				×
111	WAN Interface			~
	Choose Ethernet > Enter	WAN Settings > WAN Configuration Summary		
	Interface			
$B \to B$	WAN Interface:	ge1		
67 - 683	Zone:	WAN		
	IP Address Assignment:	Auto 👻		
/				
7				
			< Back	Next >

Figure 26 WAN Interface Setup: Step 2

- WAN Interface: This is the interface you are configuring for Internet access.
- Zone: This is the security zone to which this interface and Internet connection belong.
- **IP Address Assignment**: Select **Auto** If your ISP did not assign you a fixed IP address. Select **Static** if you have a fixed IP address.

4.2.4 WAN and ISP Connection Settings

Use this screen to configure the ISP and WAN interface settings. This screen is read-only if you set the **IP Address Assignment** to **Static**.

Note: Enter the Internet access information exactly as your ISP gave it to you.

ISP Parameters	
Encapsulation:	ртр
Authentication Type:	Chap/PAP 👻
User Name :	
Password:	
Retype to Confirm:	
Nailed-Up	
Idle timeout:	100 Seconds
PPTP Configuration	
Base Interface:	ge1
Base IP Address:	0.0.0.0
IP Subnet Mask:	255.255.255.0
Gateway IP Address:	(Optional)
Server IP:	0.0.0.0
Connection ID:	(Optional)
WAN Interface Setup	
WAN Interface:	ge1_ppp
Zone:	WAN
IP Address:	Auto
	-
	Back Next >

Figure 27 WAN and ISP Connection Settings: (PPTP Shown)

The following table describes the labels in this screen.

LABEL	DESCRIPTION
ISP Parameter	This section appears if the interface uses a PPPoE or PPTP Internet connection.
Encapsulation	This displays the type of Internet connection you are configuring.
Authentication TypeUse the drop-down list box to select an authentication protocol for outgoing ca Options are:	
	CHAP/PAP - Your ZyWALL accepts either CHAP or PAP when requested by this remote node.
	CHAP - Your ZyWALL accepts CHAP only.
	PAP - Your ZyWALL accepts PAP only.
	MSCHAP - Your ZyWALL accepts MSCHAP only.
	MSCHAP-V2 - Your ZyWALL accepts MSCHAP-V2 only.
User Name	Type the user name given to you by your ISP. You can use alphanumeric and@ $1.$
Password	Type the password associated with the user name above. Use up to 64 ASCII characters except the [] and ?. This field can be blank.
Retype to Confirm	Type your password again for confirmation.
Nailed-Up	Select Nailed-Up if you do not want the connection to time out.
Idle Timeout	Type the time in seconds that elapses before the router automatically disconnects from the PPPoE server. 0 means no timeout.
PPTP Configuration	This section only appears if the interface uses a PPPoE or PPTP Internet connection.
Base Interface	This displays the identity of the Ethernet interface you configure to connect with a modem or router.
Base IP Address	Type the (static) IP address assigned to you by your ISP.
IP Subnet Mask	Type the subnet mask assigned to you by your ISP (if given).

Table 12 WAN and ISP Connection Settings

LABEL	DESCRIPTION
Server IP	Type the IP address of the PPTP server.
Connection ID	Enter the connection ID or connection name in this field. It must follow the "c:id" and "n:name" format. For example, C:12 or N:My ISP. This field is optional and depends on the requirements of your DSL modem. You can use alphanumeric and: characters, and it can be up to 31 characters long.
WAN Interface Setup	
WAN Interface	This displays the identity of the interface you configure to connect with your ISP.
Zone	This field displays to which security zone this interface and Internet connection will belong.
IP Address	This field is read-only when the WAN interface uses a dynamic IP address. If your WAN interface uses a static IP address, enter it in this field.
First DNS Server Second DNS	These fields only display for an interface with a static IP address. Enter the DNS server IP address(es) in the field(s) to the right.
Server	Leave the field as 0.0.0.0 if you do not want to configure DNS servers. If you do not configure a DNS server, you must know the IP address of a machine in order to access it.
	DNS (Domain Name System) is for mapping a domain name to its corresponding IP address and vice versa. The DNS server is extremely important because without it, you must know the IP address of a computer before you can access it. The ZyWALL uses a system DNS server (in the order you specify here) to resolve domain names for VPN, DDNS and the time server.
Back	Click Back to return to the previous screen.
Next	Click Next to continue.

 Table 12
 WAN and ISP Connection Settings (continued)

4.2.5 Quick Setup Interface Wizard: Summary

This screen displays the WAN interface's settings.

Figure 28 Interface Wizard: Summary WAN (PPTP Shown)

Congratulations! The WAN settings have been successfully configured.				
WAN Interface Setup				
Encapsulation::	Ethernet			
WAN Interface::	ge1			
Zone::	WAN			
IP Address Assignment::	Auto			
IP Address:	0.0.0.0			
IP Subnet Mask:	0.0.0.0			
Gateway IP Address:	10.0.0.1			
First DNS Server:	10.0.0.2			
Second DNS Server:	10.0.0.3			
Endpoint Security	Close			

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Encapsulation	This displays what encapsulation this interface uses to connect to the Internet.
Service Name	This field only appears for a PPPoE interface. It displays the PPPoE service name specified in the ISP account.
Server IP	This field only appears for a PPTP interface. It displays the IP address of the PPTP server.
User Name	This is the user name given to you by your ISP.
Nailed-Up	If No displays the connection will not time out. Yes means the ZyWALL uses the idle timeout.
Idle Timeout	This is how many seconds the connection can be idle before the router automatically disconnects from the PPPoE server. 0 means no timeout.
Connection ID	If you specified a connection ID, it displays here.
WAN Interface	This identifies the interface you configure to connect with your ISP.
Zone	This field displays to which security zone this interface and Internet connection will belong.
IP Address Assignment	This field displays whether the WAN IP address is static or dynamic (Auto).
First DNS Server	If the IP Address Assignment is Static, these fields display the DNS server IP
Second DNS Server	address(es).
Close	Click Close to exit the wizard.

Table 13 Interface Wizard: Summary WAN

4.3 VPN Setup Wizard

Click VPN Setup in the main Quick Setup screen to open the VPN Setup Wizard Welcome screen.

Figure 29 VPN Setup Wizard



4.3.1 Welcome

Use wizards to create Virtual Private Network (VPN) rules. After you complete the wizard, the Phase 1 rule settings appear in the VPN > IPSec VPN > VPN Gateway screen and the Phase 2 rule settings appear in the VPN > IPSec VPN > VPN Connection screen.

- VPN Setup configures a VPN tunnel for a secure connection to another computer or network.
- VPN Settings for Configuration Provisioning sets up a VPN rule the ZyWALL IPSec VPN Client can retrieve. Just enter a user name, password and the IP address of the ZyWALL in the ZyWALL IPSec VPN Client to get the VPN settings automatically from the ZyWALL.



Ti VPN Setup Wizard		×
	VPN Setup Wizard	~
11.	Wizard Type > VPN Settings > Wizard Completed	
	Welcome	
/2 / - BE	VPN Settings	
	The later steps will guide you to setup the VPN. - Wizard Type - VIPK Settings - Wizard Completed	
10-20-00	VPN Settings for Configuration Provision	
	The later steps will guide you to setup the VPN. - Wizard Type - VPN Settings - Wizard Completed	
14		
1		
	Next >	
		_

4.3.2 VPN Setup Wizard: Wizard Type

Choose **Express** to create a VPN rule with the default phase 1 and phase 2 settings to connect to another ZLD-based ZyWALL using a pre-shared key.

Choose **Advanced** to change the default settings and/or use certificates instead of a pre-shared key to create a VPN rule to connect to another IPSec device.

Figure 31	VPN	Setup	Wizard:	Wizard	Туре
-----------	-----	-------	---------	--------	------



×

< Back

Next >

4.3.3 VPN Express Wizard - Scenario

Click the Express radio button as shown in Figure 31 on page 48 to display the following screen.

etup Wizard				
VPN Se	tup Wizard			
Wizard Typ	e > VPN Sett	ings > Wizard Co	mpleted	
	2			
Express S	ettings			
Scenario	l.	-		
Rule Na	me:	WIZ_VPN		
 Sit 	e-to-site			
Sit	e-to-site with Dyna	mic Peer		
🔘 Re	mote Access (Serve	er Role)		
🔘 Re	mote Access (Client	: Role)		

Rule Name: Type the name used to identify this VPN connection (and VPN gateway). You may use 1-31 alphanumeric characters, underscores (_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.

Select the scenario that best describes your intended VPN connection. The figure on the left of the screen changes to match the scenario you select.

- **Site-to-site** The remote IPSec device has a static IP address or a domain name. This ZyWALL can initiate the VPN tunnel.
- Site-to-site with Dynamic Peer The remote IPSec device has a dynamic IP address. Only the remote IPSec device can initiate the VPN tunnel.
- Remote Access (Server Role) Allow incoming connections from IPSec VPN clients. The clients have dynamic IP addresses and are also known as dial-in users. Only the clients can initiate the VPN tunnel.
- Remote Access (Client Role) Connect to an IPSec server. This ZyWALL is the client (dial-in user) and can initiate the VPN tunnel.

4.3.4 VPN Express Wizard - Configuration

Figure 33 VPN Express Wizard: Configuration

Secure Gateway:	Express Settings					
Secure Gateway:	Configuration					
Pre-Shared Key: Image: Constraint of the state of the st	Secure Gateway:		0	P/FQDN)		
Local Policy (IP/Mask) 0.0.0.0 / 255.255.255.0 Remote Policy (IP/Mask) 0.0.0.0 / 255.255.255.0	Pre-Shared Key:		0			
Remote Policy (IP/Mask) 0.0.0.0 / 255.255.0	Local Policy (IP/Mask)	0.0.0.0	1	255.255.255.0		
	Remote Policy (IP/Mask)	0.0.0.0	1	255.255.255.0		
					1	
Sack Next >					< Back	Next >

- Secure Gateway: Any displays in this field if it is not configurable for the chosen scenario. Otherwise, enter the WAN IP address or domain name of the remote IPSec device (secure gateway) to identify the remote IPSec router by its IP address or a domain name. Use 0.0.0.0 if the remote IPSec router has a dynamic WAN IP address.
- **Pre-Shared Key**: Type the password. Both ends of the VPN tunnel must use the same password. Use 8 to 31 case-sensitive ASCII characters or 8 to 31 pairs of hexadecimal ("0-9", "A-F") characters. Proceed a hexadecimal key with "0x". You will receive a PYLD_MALFORMED (payload malformed) packet if the same pre-shared key is not used on both ends.
- Local Policy (IP/Mask): Type the IP address of a computer on your network that can use the tunnel. You can also specify a subnet. This must match the remote IP address configured on the remote IPSec device.
- Remote Policy (IP/Mask): Any displays in this field if it is not configurable for the chosen scenario. Otherwise, type the IP address of a computer behind the remote IPSec device. You can also specify a subnet. This must match the local IP address configured on the remote IPSec device.

4.3.5 VPN Express Wizard - Summary

This screen provides a read-only summary of the VPN tunnel's configuration and commands that you can copy and paste into another ZLD-based ZyWALL's command line interface to configure it.

Elaura 24	VDN Express	Winord	Cummon	
Figure 34	VPIN EXPRESS	wizard:	Summary	/

Express Settings	•	
Summary		
Rule Name:	WIZ_VPN	
Secure Gateway:	1.2.3.4	
Pre-Shared Key:	shnr6bge45y4	
Local Policy:	192,168,2,1 / 255,255,255,0	
Remote Policy:	10.0.0.0 / 255.255.255.0	
Configuration for Secur	e Gateway	
## Then remove the for PLEASE REMOVE THIS L configure terminal isakmp policy WIZ_VPM ## It this device's wan, ## consider using DDW, ## the peer-ip lot.0.0.9 ## Use the correct inte ## Lose the correct inte ## local-ip interface wan	llowing line. INE 5 and changing re to a domain name. rface name in the and remove the "#". 1	
Click "Save" button to wri	te the VPN configuration to ZyWALL.	
		< Back Save

- Rule Name: Identifies the VPN gateway policy.
- Secure Gateway: IP address or domain name of the remote IPSec device. If this field displays Any, only the remote IPSec device can initiate the VPN connection.
- **Pre-Shared Key**: VPN tunnel password. It identifies a communicating party during a phase 1 IKE negotiation.
- Local Policy: IP address and subnet mask of the computers on the network behind your ZyWALL that can use the tunnel.
- **Remote Policy**: IP address and subnet mask of the computers on the network behind the remote IPSec device that can use the tunnel. If this field displays **Any**, only the remote IPSec device can initiate the VPN connection.
- Copy and paste the **Configuration for Secure Gateway** commands into another ZLD-based ZyWALL's command line interface to configure it to serve as the other end of this VPN tunnel. You can also use a text editor to save these commands as a shell script file with a ".zysh" filename extension. Use the file manager to run the script in order to configure the VPN connection. See the commands reference guide for details on the commands displayed in this list.

4.3.6 VPN Express Wizard - Finish

Now the rule is configured on the ZyWALL. The Phase 1 rule settings appear in the VPN > IPSec VPN > VPN Gateway screen and the Phase 2 rule settings appear in the VPN > IPSec VPN > VPN Connection screen.





Click **Close** to exit the wizard.

4.3.7 VPN Advanced Wizard - Scenario

Click the **Advanced** radio button as shown in Figure 31 on page 48 to display the following screen.

T VPN Setup Wizard	Portunale States and States with the second states and	×
	VPN Setup Wizard Vizard Type > VPN Settings > Wizard Completed Carbonal C	
Subcardi Sirkonity Vine	< Back Next	t>

Figure 36 VPN Advanced Wizard: Scenario

Rule Name: Type the name used to identify this VPN connection (and VPN gateway). You may use 1-31 alphanumeric characters, underscores (_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.

Select the scenario that best describes your intended VPN connection. The figure on the left of the screen changes to match the scenario you select.

- **Site-to-site** The remote IPSec device has a static IP address or a domain name. This ZyWALL can initiate the VPN tunnel.
- Site-to-site with Dynamic Peer The remote IPSec device has a dynamic IP address. Only the remote IPSec device can initiate the VPN tunnel.
- Remote Access (Server Role) Allow incoming connections from IPSec VPN clients. The clients have dynamic IP addresses and are also known as dial-in users. Only the clients can initiate the VPN tunnel.
- Remote Access (Client Role) Connect to an IPSec server. This ZyWALL is the client (dial-in user) and can initiate the VPN tunnel.

4.3.8 VPN Advanced Wizard - Phase 1 Settings

There are two phases to every IKE (Internet Key Exchange) negotiation – phase 1 (Authentication) and phase 2 (Key Exchange). A phase 1 exchange establishes an IKE SA (Security Association).

Advanced Settings		
Phase 1 Setting		
Secure Gateway:		
My Address (interface):	wan1	~
Negotiation Mode:	Main	~
Encryption Algorithm:	DES	×
Authentication Algorithm:	MD5	~
Key Group:	DH1	×
SA Life Time:	86400	(180 - 3000000 Seconds)
NAT Traversal		
📝 Dead Peer Detection (DPD)	
Authentication Method		
Pre-Shared Key		
Certificate	default	*
		< Back Next >

Figure 37 VPN Advanced Wizard: Phase 1 Settings

- Secure Gateway: Any displays in this field if it is not configurable for the chosen scenario. Otherwise, enter the WAN IP address or domain name of the remote IPSec device (secure gateway) to identify the remote IPSec device by its IP address or a domain name. Use 0.0.0.0 if the remote IPSec device has a dynamic WAN IP address.
- **My Address (interface)**: Select an interface from the drop-down list box to use on your ZyWALL.
- **Negotiation Mode**: Select **Main** for identity protection. Select **Aggressive** to allow more incoming connections from dynamic IP addresses to use separate passwords.

Note: Multiple SAs connecting through a secure gateway must have the same negotiation mode.

- Encryption Algorithm: 3DES and AES use encryption. The longer the key, the higher the security (this may affect throughput). Both sender and receiver must use the same secret key, which can be used to encrypt and decrypt the message or to generate and verify a message authentication code. The DES encryption algorithm uses a 56-bit key. Triple DES (3DES) is a variation on DES that uses a 168-bit key. As a result, 3DES is more secure than DES. It also requires more processing power, resulting in increased latency and decreased throughput.
 AES128 uses a 128-bit key and is faster than 3DES. AES192 uses a 192-bit key, and AES256 uses a 256-bit key.
- Authentication Algorithm: MD5 gives minimal security and SHA512 gives the highest security. MD5 (Message Digest 5) and SHA (Secure Hash Algorithm) are hash algorithms used to authenticate packet data. The stronger the algorithm the slower it is.
- **Key Group**: **DH5** is more secure than **DH1** or **DH2** (although it may affect throughput). DH1 (default) refers to Diffie-Hellman Group 1 a 768 bit random number. DH2 refers to Diffie-Hellman Group 2 a 1024 bit (1Kb) random number. DH5 refers to Diffie-Hellman Group 5 a 1536 bit random number.
- **SA Life Time**: Set how often the ZyWALL renegotiates the IKE SA. A short SA life time increases security, but renegotiation temporarily disconnects the VPN tunnel.
- **NAT Traversal**: Select this if the VPN tunnel must pass through NAT (there is a NAT router between the IPSec devices).
- Note: The remote IPSec device must also have NAT traversal enabled. See the help in the main IPSec VPN screens for more information.

- **Dead Peer Detection (DPD)** has the ZyWALL make sure the remote IPSec device is there before transmitting data through the IKE SA. If there has been no traffic for at least 15 seconds, the ZyWALL sends a message to the remote IPSec device. If it responds, the ZyWALL transmits the data. If it does not respond, the ZyWALL shuts down the IKE SA.
- Authentication Method: Select Pre-Shared Key to use a password or Certificate to use one of the ZyWALL's certificates.

4.3.9 VPN Advanced Wizard - Phase 2

Phase 2 in an IKE uses the SA that was established in phase 1 to negotiate SAs for IPSec.

ranceu oetungs				
nase 2 Setting				
Active Protocol:	ESP		~	
Encapsulation:	Tunnel		•	
Encryption Algorithm:	DES		~	
Authentication Algorithm:	SHA1		~	
5A Life Time:	86400	(1)	80 - 30000	00 Seconds)
Perfect Forward Secrecy (PFS):	None		*	
olicy Setting				
.ocal Policy (IP/Mask)	0.0.0.0	1	255.255	.255.0
Remote Policy (IP/Mask)	0.0.0	1	255.255	.255.0
operty				
Noted and the				

Figure 38 VPN Advanced Wizard: Step 4

- Active Protocol: ESP is compatible with NAT, AH is not.
- Encapsulation: Tunnel is compatible with NAT, Transport is not.
- Encryption Algorithm: 3DES and AES use encryption. The longer the AES key, the higher the security (this may affect throughput). Null uses no encryption.
- Authentication Algorithm: MD5 gives minimal security and SHA512 gives the highest security. MD5 (Message Digest 5) and SHA (Secure Hash Algorithm) are hash algorithms used to authenticate packet data. The stronger the algorithm the slower it is.
- **SA Life Time**: Set how often the ZyWALL renegotiates the IKE SA. A short SA life time increases security, but renegotiation temporarily disconnects the VPN tunnel.
- **Perfect Forward Secrecy (PFS)**: Disabling PFS allows faster IPSec setup, but is less secure. Select DH1, DH2 or DH5 to enable PFS. **DH5** is more secure than **DH1** or **DH2** (although it may affect throughput). DH1 refers to Diffie-Hellman Group 1 a 768 bit random number. DH2 refers to Diffie-Hellman Group 2 a 1024 bit (1Kb) random number. DH5 refers to Diffie-Hellman Group 5 a 1536 bit random number (more secure, yet slower).
- Local Policy (IP/Mask): Type the IP address of a computer on your network. You can also specify a subnet. This must match the remote IP address configured on the remote IPSec device.
- **Remote Policy (IP/Mask)**: Type the IP address of a computer behind the remote IPSec device. You can also specify a subnet. This must match the local IP address configured on the remote IPSec device.
- Nailed-Up: This displays for the site-to-site and remote access client role scenarios. Select this to have the ZyWALL automatically renegotiate the IPSec SA when the SA life time expires.

4.3.10 VPN Advanced Wizard - Summary

This is a read-only summary of the VPN tunnel settings.

Figure 39 VPN Advanced Wizard: Step 5

Express Settings		
Summary		
Rule Name:	WIZ_VPN	
Secure Gateway:	1.2.3.4	
Certificate:	default	
Local Policy:	0.0.0.0 / 255.255.255.0	
Remote Policy:	0.0.0.0 / 255.255.255.0	
Configuration for Secure Gal	eway	
## Check the peer-ip interfar ## Check the local-ip interfar ## Edit the <u>WIZ_VPN</u> LOCAL ## Then remove the followin PLEASE REMOVE THIS LINE configure terminal isakmp policy <u>WIZ_VPN</u> ## If this device's wan1 IP is ## consider using <u>DDNS</u> and ## the peer-ip listed here to peer-ip 10.0.0.9 ## Use the correct interface ## next command line and re	:e. address-object. g line. dynamic, changing a domain name. name in the move the "#". e VPN configuration to ZyWALL.	
		< Back Save

- Rule Name: Identifies the VPN connection (and the VPN gateway).
- Secure Gateway: IP address or domain name of the remote IPSec device.
- · Pre-Shared Key: VPN tunnel password.
- Certificate: The certificate the ZyWALL uses to identify itself when setting up the VPN tunnel.
- Local Policy: IP address and subnet mask of the computers on the network behind your ZyWALL that can use the tunnel.
- **Remote Policy**: IP address and subnet mask of the computers on the network behind the remote IPSec device that can use the tunnel.
- Copy and paste the **Configuration for Remote Gateway** commands into another ZLD-based ZyWALL's command line interface.
- Click Save to save the VPN rule.

4.3.11 VPN Advanced Wizard - Finish

Now the rule is configured on the ZyWALL. The Phase 1 rule settings appear in the VPN > IPSec VPN > VPN Gateway screen and the Phase 2 rule settings appear in the VPN > IPSec VPN > VPN Connection screen.

Figure 40 VPN Wizard: Finish

Rule Name:	WIZ_VPN
Secure Gateway:	1.2.3.4
My Address (interface):	wan1
Pre-Shared Key:	lkj581mjw777
hase 1	
Negotiation Mode:	main
Encryption Algorithm:	des
Authentication Algorithm:	md5
Key Group:	DH1
SA Life Time:	86400
NAT Traversal:	false
Dead Peer Detection (DPD):	true
hase 2	
Active Protocol:	esp
Encapsulation:	tunnel
Encryption Algorithm:	des
Authentication Algorithm:	sha
SA Life Time:	86400
Perfect Forward Secrecy:	None
Policy	
Local Policy:	0.0.0.0 / 255.255.255.0
Remote Policy:	0.0.0.0 / 255.255.255.0
Nailed-Up:	true

Click **Close** to exit the wizard.

4.4 VPN Settings for Configuration Provisioning Wizard: Wizard Type

Use VPN Settings for Configuration Provisioning to set up a VPN rule that can be retrieved with the ZyWALL IPSec VPN Client.

VPN rules for the ZyWALL IPSec VPN Client have certain restrictions. They must *not* contain the following settings:

- AH active protocol
- NULL encryption
- SHA512 authentication
- A subnet or range remote policy

Choose **Express** to create a VPN rule with the default phase 1 and phase 2 settings and to use a pre-shared key.

Choose **Advanced** to change the default settings and/or use certificates instead of a pre-shared key in the VPN rule.

Figure 41 VPN Settings for Configuration Provisioning Express Wizard: Wizard Type



4.4.1 Configuration Provisioning Express Wizard - VPN Settings

Click the **Express** radio button as shown in the previous screen to display the following screen.

VPN Setup Wizard			~
Wizard Type > VPN Sett	ings > Wizard Completed		
2	5		
Express Settings			
Scenario			
Rule Name:	WIZ_VPN_PROVISIONING		
Application Scenario:	Remote Access (Server Role)		
		< Back	Next >

Figure 42 VPN for Configuration Provisioning Express Wizard: Settings Scenario

Rule Name: Type the name used to identify this VPN connection (and VPN gateway). You may use 1-31 alphanumeric characters, underscores (_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.

Application Scenario: Only the **Remote Access (Server Role)** is allowed in this wizard. It allows incoming connections from the ZyWALL IPSec VPN Client.

4.4.2 Configuration Provisioning VPN Express Wizard - Configuration

Click **Next** to continue the wizard.

•	<u> </u>	<u> </u>	
			×
/PN Setup Wizard			~
Vizard Type > VPN Settin	gs > Wizard Completed		
1 2	3		
voress Settings			
Configuration			
comguration	1111		
Secure Gateway:	Any		
Pre-Shared Key:	12345678		
Local Policy (IP/Mask)	0.0.0.0 /	255.255.255.0	
Remote Policy (IP/Mask):	Anv		
			(Paula) (Neuta)
			< Back Next >

Figure 43 VPN for Configuration Provisioning Express Wizard: Configuration

- Secure Gateway: Any displays in this field because it is not configurable in this wizard. It allows incoming connections from the ZyWALL IPSec VPN Client.
- **Pre-Shared Key**: Type the password. Both ends of the VPN tunnel must use the same password. Use 8 to 31 case-sensitive ASCII characters or 8 to 31 pairs of hexadecimal ("0-9", "A-F") characters. Proceed a hexadecimal key with "0x". You will receive a PYLD_MALFORMED (payload malformed) packet if the same pre-shared key is not used on both ends.
- Local Policy (IP/Mask): Type the IP address of a computer on your network. You can also specify a subnet. This must match the remote IP address configured on the remote IPSec device.
- Remote Policy (IP/Mask): Any displays in this field because it is not configurable in this wizard.

4.4.3 VPN Settings for Configuration Provisioning Express Wizard -Summary

This screen has a read-only summary of the VPN tunnel's configuration and commands you can copy and paste into another ZLD-based ZyWALL's command line interface to configure it.

1			^
izard Type > VPN Settin 1 2	ngs > Wizard Completed		
xpress Settings			
Summary			
Rule Name:	WIZ_VPN_PROVISIONING		
Secure Gateway:	Any		
Pre-Shared Key:	12345678		=
Local Policy (IP/Mask):	0.0.0.0 / 255.255.255.0		
Remote Policy (IP/Mask):	Any		
Configuration for Secure G	ateway		
## Edit this shell script acco ## the comments before us ## Check the peer-ip interfi ## Check the local-ip interfi ## Check the WIZ_VDN_DPOPO	rding to ing it in the remote gateway. ace. VISIONING_LOCAL address-object.	*	
## Eait die WIZ_VPN_PROC ## Then remove the followi PLEASE REMOVE THIS LINE configure terminal isakmp policy WIZ_VPN_PRC ## If this device's wan1 IP ## consider using DDNS an ## the peer-ip listed here tr	ing line. DVISIONING is dynamic, 3 doanging 3 a domain name.		

Figure 44 VPN for Configuration Provisioning Express Wizard: Save

- Rule Name: Identifies the VPN gateway policy.
- Secure Gateway: Any displays in this field because it is not configurable in this wizard. It allows incoming connections from the ZyWALL IPSec VPN Client.
- **Pre-Shared Key**: VPN tunnel password. It identifies a communicating party during a phase 1 IKE negotiation.
- Local Policy: (Static) IP address and subnet mask of the computers on the network behind your ZyWALL that can be accessed using the tunnel.
- **Remote Policy**: **Any** displays in this field because it is not configurable in this wizard.
- The **Configuration for Secure Gateway** displays the configuration that the ZyWALL IPSec VPN Client will get from the ZyWALL.

4.4.4 VPN Settings for Configuration Provisioning Express Wizard - Finish

Now the rule is configured on the ZyWALL. The Phase 1 rule settings appear in the VPN > IPSec VPN > VPN Gateway screen and the Phase 2 rule settings appear in the VPN > IPSec VPN > VPN Connection screen. Enter the IP address of the ZyWALL in the ZyWALL IPSec VPN Client to get all these VPN settings automatically from the ZyWALL.

VPN Setup Wizard			
Wizard Type > VPN Settings	> Wizard Completed		
Express Settings			
Congratulations. The VPN Acc Summary	ess wizard is completed		
Rule Name:	WIZ_VPN_PROVISIONIN		
Secure Gateway:	Any		
Pre-Shared Key:	12345678		
Local Policy (IP/Mask):	0.0.0.0 / 255.255.255.0		
Remote Policy (IP/Mask):	Any		
Kemote Policy (JP/Mask): Now if you are doing first time device and activate trial service	Any installation of this device, you may click this <u>myZyXEL.com</u> lin of advanced security features.(You need to have internet a	ik and to register this access to register)	
		Close	

Figure 45 VPN for Configuration Provisioning Express Wizard: Finish

Click Close to exit the wizard.

4.4.5 VPN Settings for Configuration Provisioning Advanced Wizard - Scenario

Click the **Advanced** radio button as shown in the screen shown in Figure 41 on page 58 to display the following screen.

/PN Setup Wizard		
Izard Type > VPN Sett	ings > Wizard Completed	
xpress Settings		
Rule Name:	WIZ_VPN_PROVISIONING	
Application Scenario:	Remote Access (Server Role)	
		< Back Next >

Figure 46 VPN for Configuration Provisioning Advanced Wizard: Scenario Settings

Rule Name: Type the name used to identify this VPN connection (and VPN gateway). You may use 1-31 alphanumeric characters, underscores (_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.

Application Scenario: Only the **Remote Access (Server Role)** is allowed in this wizard. It allows incoming connections from the ZyWALL IPSec VPN Client.

Click Next to continue the wizard.

4.4.6 VPN Settings for Configuration Provisioning Advanced Wizard - Phase 1 Settings

There are two phases to every IKE (Internet Key Exchange) negotiation – phase 1 (Authentication) and phase 2 (Key Exchange). A phase 1 exchange establishes an IKE SA (Security Association).

Figure 47	VPN for	Configuration	Provisioning	Advanced	Wizard	Phase 1	l Settinas
i iguie 47		configuration	FIOVISIONING	Auvanceu	vvizaru.	Fliase	Jettings

Advanced Settings		
Phase 1 Setting		
Secure Gateway:	Any	
My Address (interface):	wan1	~
Negotiation Mode:	Main	~
Encryption Algorithm:	DES	~
Authentication Algorithm:	MD5	~
Key Group:	DH1	•
SA Life Time:	86400 (1	30 - 3000000 seconds)
Authentication Method		
Pre-Shared Key		
Certificate	default	~

- Secure Gateway: Any displays in this field because it is not configurable in this wizard. It allows incoming connections from the ZyWALL IPSec VPN Client.
- **My Address (interface)**: Select an interface from the drop-down list box to use on your ZyWALL.
- **Negotiation Mode**: Select **Main** for identity protection. Select **Aggressive** to allow more incoming connections from dynamic IP addresses to use separate passwords.
- Note: Multiple SAs connecting through a secure gateway must have the same negotiation mode.
- Encryption Algorithm: 3DES and AES use encryption. The longer the key, the higher the security (this may affect throughput). Both sender and receiver must know the same secret key, which can be used to encrypt and decrypt the message or to generate and verify a message authentication code. The DES encryption algorithm uses a 56-bit key. Triple DES (3DES) is a variation on DES that uses a 168-bit key. As a result, 3DES is more secure than DES. It also requires more processing power, resulting in increased latency and decreased throughput. AES128 uses a 128-bit key and is faster than 3DES. AES192 uses a 192-bit key and AES256 uses a 256-bit key.

- Authentication Algorithm: MD5 (Message Digest 5) and SHA (Secure Hash Algorithm) are hash algorithms used to authenticate packet data. MD5 gives minimal security. SHA1 gives higher security and SHA256 gives the highest security. The stronger the algorithm, the slower it is.
- **Key Group**: **DH5** is more secure than **DH1** or **DH2** (although it may affect throughput). DH1 (default) refers to Diffie-Hellman Group 1 a 768 bit random number. DH2 refers to Diffie-Hellman Group 2 a 1024 bit (1Kb) random number. **DH5** refers to Diffie-Hellman Group 5 a 1536 bit random number.
- **SA Life Time**: Set how often the ZyWALL renegotiates the IKE SA. A short SA life time increases security, but renegotiation temporarily disconnects the VPN tunnel.
- Authentication Method: Select Pre-Shared Key to use a password or Certificate to use one of the ZyWALL's certificates.

4.4.7 VPN Settings for Configuration Provisioning Advanced Wizard - Phase 2

Phase 2 in an IKE uses the SA that was established in phase 1 to negotiate SAs for IPSec.

dvanced Settings				
Phase 2 Setting				
Active Protocol:	ESP			
Encapsulation:	Tunnel		*	
Encryption Algorithm:	DES		*	
Authentication Algorithm:	SHA1		*	
SA Life Time:	86400	(180 -	3000000 se	conds
Perfect Forward Secrecy (PFS):	None		*	
Policy Setting				
Local Policy (IP/Mask)	0.0.0.0	/ 255	5.255.255.0)
Remote Policy (IP/Mask):	Any			

Figure 48 VPN for Configuration Provisioning Advanced Wizard: Phase 2

- Active Protocol: ESP is compatible with NAT. AH is not available in this wizard.
- Encapsulation: Tunnel is compatible with NAT, Transport is not.
- Encryption Algorithm: 3DES and AES use encryption. The longer the AES key, the higher the security (this may affect throughput). Null uses no encryption.
- Authentication Algorithm: MD5 (Message Digest 5) and SHA (Secure Hash Algorithm) are hash algorithms used to authenticate packet data. **MD5** gives minimal security. **SHA1** gives higher security and **SHA256** gives the highest security. The stronger the algorithm, the slower it is.
- **SA Life Time**: Set how often the ZyWALL renegotiates the IKE SA. A short SA life time increases security, but renegotiation temporarily disconnects the VPN tunnel.
- **Perfect Forward Secrecy (PFS):** Disabling PFS allows faster IPSec setup, but is less secure. Select DH1, DH2 or DH5 to enable PFS. **DH5** is more secure than **DH1** or **DH2** (although it may affect throughput). DH1 refers to Diffie-Hellman Group 1 a 768 bit random number. DH2 refers to Diffie-Hellman Group 2 a 1024 bit (1Kb) random number. DH5 refers to Diffie-Hellman Group 5 a 1536 bit random number (more secure, yet slower).
- Local Policy (IP/Mask): Type the IP address of a computer on your network. You can also specify a subnet. This must match the remote IP address configured on the remote IPSec device.

- **Remote Policy (IP/Mask)**: **Any** displays in this field because it is not configurable in this wizard.
- **Nailed-Up**: This displays for the site-to-site and remote access client role scenarios. Select this to have the ZyWALL automatically renegotiate the IPSec SA when the SA life time expires.

4.4.8 VPN Settings for Configuration Provisioning Advanced Wizard -Summary

This is a read-only summary of the VPN tunnel settings.

,	3	3		
Advanced Settings				* «
Rule Name:	WIZ VPN PROVISIONING			
Secure Gateway:	Any			
Pre-Shared Key:	12345678			
Local Policy (IP/Mask):	0.0.0.0 / 255.255.255.0			
Remote Policy (IP/Mask):	Any			
Configuration for Secure Gal	teway			
## the comments before usin ## Check the peer-ip interfac ## Check the local-ip interfac ## Edit the WIZ_VPN_PROVI: ## Then remove the following PLEASE REMOVE THIS LINE configure terminal isakmp policy WIZ_VPN_PROV ## If this device's wan1 IP is ## consider using DDNS and ## the peer-ip listed here to . peer-ip 172.23.26.249 ## Use the correct interface	g it in the remote gateway. ie. ie. SIONING_LOCAL address-object. J line. ISIONING dynamic, changing a domain name. name in the			E
Click "Save" button to write th	e VPN configuration to ZyWALL.			-
			< Back	Save

Figure 49 VPN for Configuration Provisioning Advanced Wizard: Summary

- Rule Name: Identifies the VPN connection (and the VPN gateway).
- Secure Gateway: Any displays in this field because it is not configurable in this wizard. It allows incoming connections from the ZyWALL IPSec VPN Client.
- · Pre-Shared Key: VPN tunnel password.
- Certificate: The certificate the ZyWALL uses to identify itself when setting up the VPN tunnel.
- Local Policy: IP address and subnet mask of the computers on the network behind your ZyWALL that can use the tunnel.
- Remote Policy: Any displays in this field because it is not configurable in this wizard.
- The **Configuration for Secure Gateway** displays the configuration that the ZyWALL IPSec VPN Client will get from the ZyWALL.
- Click Save to save the VPN rule.

4.4.9 VPN Settings for Configuration Provisioning Advanced Wizard- Finish

Now the rule is configured on the ZyWALL. The Phase 1 rule settings appear in the **VPN** > **IPSec VPN** > **VPN** Gateway screen and the Phase 2 rule settings appear in the **VPN** > **IPSec VPN** >

VPN Connection screen. Enter the IP address of the ZyWALL in the ZyWALL IPSec VPN Client to get all these VPN settings automatically from the ZyWALL.

PN Setup Wizard		
/izard Type > VPN Settings	> Wizard Completed	
dvanced Settings		
ongratulations. The VPN Acountry	cess wizard is completed	=
Rule Name:	WIZ_VPN_PROVISIONIN	
Secure Gateway:	Any	
My Address (interface):	wan1	
Pre-Shared Key:	12345678	
Phase 1		
Negotiation Mode:	main	
Encryption Algorithm:	des	
Authentication Algorithm:	md5	
Key Group:	DH1	
SA Life Time:	86400	
	false	

Figure 50 VPN for Configuration Provisioning Advanced Wizard: Finish

Click Close to exit the wizard.

Dashboard

5.1 Overview

Use the Dashboard screens to check status information about the ZyWALL.

5.1.1 What You Can Do in this Chapter

Use the Dashboard screens for the following.

- Use the main **Dashboard** screen (see Section 5.2 on page 67) to see the ZyWALL's general device information, system status, system resource usage, licensed service status, and interface status. You can also display other status screens for more information.
- Use the **VPN** status screen (see Section 5.2.4 on page 74) to look at the VPN tunnels that are currently established.
- Use the **DHCP Table** screen (see Section 5.2.5 on page 75) to look at the IP addresses currently assigned to DHCP clients and the IP addresses reserved for specific MAC addresses.
- Use the **Current Users** screen (see Section 5.2.6 on page 76) to look at a list of the users currently logged into the ZyWALL.

5.2 The Dashboard Screen

The **Dashboard** screen displays when you log into the ZyWALL or click **Dashboard** in the navigation panel. The dashboard displays general device information, system status, system resource usage, licensed service status, and interface status in widgets that you can re-arrange to suit your needs. You can also collapse, refresh, and close individual widgets.

DASHBOARD									A		e) Widget Set
		Rear Panel	175			orr - i i Pa Pa		er al la	C	D	E
Device Information	_		_	* 0 ¢ X	📥 Top 5 IPv4	4 Firewall Rul	les that Blocked T	raffic		_	* 0
System Name:	zywali-usq110 ZeMALL 110				# From		То		Description	Hits	
Model Name: Serial Number:	\$122L23030003										
MAC Address Range:	80:82:DC:69:A5:F	~ 80:82:DC:69:A6:04			Top 5 IPve	5 Firewall Rul	es that Blocked 1	raffic	-		+ 0
Firmware Version:	V3.10(AAAA.0)Preb	7-r38094 / 1.03 / 2013-01-24 05:27:52			+ From		10		Description	1165	
System Status				* 0 # X	The Lates	t Alert Logs					. 0
System Uptime:		06:26:49			# Time		Priority	Category	Message Sourc	e D	estination
Current Date/Time:		2013-01-29 / 09:45:18 GMT+00:00			1 2013-01-	-29 03:18:03	alert	file-manage	WARNING: #conf		
VPN Status:		9			2 2013-01-	-29 03:17:57	alert	policy-route	Trunk SYSTEM		
DHCP Table:		9			3 2013-01-	-29 03:17:57	alert	policy-route	Interface opt_ppp		
Current Login User:		admin (unlimited / 00:29:59)			4 2013-01-	-29 03:17:57	alert	policy-route	Interface wan2_pp		
Number of Login Users:		1			5 2013-01-	-29 03:17:57	aien	policy-route	Interface wan1_pp		
Boot Status:		ок			👷 Interface :	Status Summ	iary				* 8
o System Resources				A 0 0 X	Name	Status	HA Status	Zone	IP Addr/Netmask	IP Assignment	Action
CPU Usage					want	100M/Full	n/a	WAN	172 23 30 26 / 255 255 255 0	DHCP client	Renew
	0 %				wan2	Down	n/a	WAN	0.0.0.0/0.0.0.0	DHCP client	Renew
Memory Usage					opt	Down	n/a	OPT	0.0.0.0/0.0.0.0	Static	n/a
	20 %				lan1	Down	n/a	LAN1	192.168.1.1 / 255.255.255.0	Static	n/a
Elech Henne					lan2	Down	n/a	LAN2	192.168.2.1 / 255.255.255.0	Static	n/a
Plash Usage	8 %				ext-wian	Down	n/a	DM7	10.59.0.1 / 255 255 255 0	Static	n/a
	12										
USB Storage Usa	ge 0/0 MP		-								
	U/U MB										
- Active Sessions											
	141/6000	0									
III Extension Slot				* 0 ¢ X							
and a state of the			Status								
# Extension Slot	1,199/109										

Figure 51 Dashboard

The following table describes the labels in this screen.

Table 14 Dashboard

LABEL	DESCRIPTION
Widget Setting (A)	Use this link to open or close widgets by selecting/clearing the associated checkbox.
Up Arrow (B)	Click this to collapse a widget. It then becomes a down arrow. Click it again to enlarge the widget again.
Refresh Time Setting (C)	Set the interval for refreshing the information displayed in the widget.
Refresh Now (D)	Click this to update the widget's information immediately.
Close Widget (E)	Click this to close the widget. Use Widget Setting to re-open it.
Virtual Device	
Rear Panel	Click this to view details about the ZyWALL's rear panel. Hover your cursor over a connected interface or slot to display status details.
Front Panel	Click this to view details about the status of the ZyWALL's front panel LEDs and connections. See Section 3.5 on page 39 for LED descriptions. An unconnected interface or slot appears grayed out.
	The following front and rear panel labels display when you hover your cursor over a connected interface or slot.
Name	This field displays the name of each interface.
Slot	This field displays the name of each extension slot.

Table 14 Dashboard (continued)

LABEL	DESCRIPTION
Device	This field displays the name of the device connected to the USB port if one is connected.
Status	This field displays the current status of each interface or device installed in a slot. The possible values depend on what type of interface it is.
	Inactive - The Ethernet interface is disabled.
	Down - The Ethernet interface is enabled but not connected.
	Speed / Duplex - The Ethernet interface is enabled and connected. This field displays the port speed and duplex setting (Full or Half).
	For cellular (3G) interfaces, see Section 7.5 on page 132 the Web Help for the status that can appear.
Zone	This field displays the zone to which the interface is currently assigned.
IP Address/ Mask	This field displays the current IP address and subnet mask assigned to the interface. If the interface is a member of an active virtual router, this field displays the IP address it is currently using. This is either the static IP address of the interface (if it is the master) or the management IP address (if it is a backup).
Device Information	This identifies a device installed in one of the ZyWALL's extension slots or USB ports.
System Name	This field displays the name used to identify the ZyWALL on any network. Click the icon to open the screen where you can change it.
Model Name	This field displays the model name of this ZyWALL.
Serial Number	This field displays the serial number of this ZyWALL. The serial number is used for device tracking and control.
MAC Address Range	This field displays the MAC addresses used by the ZyWALL. Each physical port has one MAC address. The first MAC address is assigned to physical port 1, the second MAC address is assigned to physical port 2, and so on.
Firmware Version	This field displays the version number and date of the firmware the ZyWALL is currently running. Click the icon to open the screen where you can upload firmware.
System Status	
System Uptime	This field displays how long the ZyWALL has been running since it last restarted or was turned on.
Current Date/Time	This field displays the current date and time in the ZyWALL. The format is yyyy-mm-dd hh:mm:ss.
VPN Status	Click this to look at the VPN tunnels that are currently established. See Section 5.2.1 on page 72.
DHCP Table	Click this to look at the IP addresses currently assigned to the ZyWALL's DHCP clients and the IP addresses reserved for specific MAC addresses. See Section 5.2.5 on page 75.
Current Login User	This field displays the user name used to log in to the current session, the amount of reauthentication time remaining, and the amount of lease time remaining.
Number of Login Users	This field displays the number of users currently logged in to the ZyWALL. Click the icon to pop-open a list of the users who are currently logged in to the ZyWALL.

LABEL	DESCRIPTION
Boot Status	This field displays details about the ZyWALL's startup state.
	OK - The ZyWALL started up successfully.
	Firmware update OK - A firmware update was successful.
	Problematic configuration after firmware update - The application of the configuration failed after a firmware upgrade.
	System default configuration - The ZyWALL successfully applied the system default configuration. This occurs when the ZyWALL starts for the first time or you intentionally reset the ZyWALL to the system default settings.
	Fallback to lastgood configuration - The ZyWALL was unable to apply the startup- config.conf configuration file and fell back to the lastgood.conf configuration file.
	Fallback to system default configuration - The ZyWALL was unable to apply the lastgood.conf configuration file and fell back to the system default configuration file (system-default.conf).
	Booting in progress - The ZyWALL is still applying the system configuration.
System Resources	
CPU Usage	This field displays what percentage of the ZyWALL's processing capability is currently being used. Hover your cursor over this field to display the Show CPU Usage icon that takes you to a chart of the ZyWALL's recent CPU usage.
Memory Usage	This field displays what percentage of the ZyWALL's RAM is currently being used. Hover your cursor over this field to display the Show Memory Usage icon that takes you to a chart of the ZyWALL's recent memory usage.
Flash Usage	This field displays what percentage of the ZyWALL's onboard flash memory is currently being used.
USB Storage Usage	This field shows how much storage in the USB device connected to the ZyWALL is in use.
Active Sessions	This field shows how many sessions, established and non-established, that pass through/ from/to/within the ZyWALL. Hover your cursor over this field to display icons. Click the Detail icon to go to the Session Monitor screen to see details about the active sessions. Click the Show Active Sessions icon to display a chart of ZyWALL's recent session usage.
Extension Slot	This section of the screen displays the status of the extension card slot the USB ports.
Extension Slot	This field displays the name of each extension slot.
Device	This field displays the name of the device connected to the extension slot (or none if no device is detected).
	USB Flash Drive - Indicates a connected USB storage device and the drive's storage capacity.
Status	For cellular (3G) interfaces, see Section 6.10 on page 96 the Web Help for the status that can appear.
	Ready - A USB storage device connected to the ZyWALL is ready for the ZyWALL to use.
	Unused - The ZyWALL is unable to mount a USB storage device connected to the ZyWALL.
Interface Status Summary	If an Ethernet interface does not have any physical ports associated with it, its entry is displayed in light gray text. Click the Detail icon to go to a (more detailed) summary screen of interface statistics.
#	This shows how many interfaces there are.
Name	This field displays the name of each interface.

Table 14 Dashboard (continued)

Table 14 Dashboard (continued)

LABEL	DESCRIPTION
Status	This field displays the current status of each interface. The possible values depend on what type of interface it is.
	For Ethernet interfaces:
	Inactive - The Ethernet interface is disabled.
	Down - The Ethernet interface does not have any physical ports associated with it or the Ethernet interface is enabled but not connected.
	Speed / Duplex - The Ethernet interface is enabled and connected. This field displays the port speed and duplex setting (Full or Half).
	For PPP interfaces:
	Connected - The PPP interface is connected.
	Disconnected - The PPP interface is not connected.
	If the PPP interface is disabled, it does not appear in the list.
Zone	This field displays the zone to which the interface is currently assigned.
IP Addr/ Netmask	This field displays the current IP address and subnet mask assigned to the interface. If the IP address is 0.0.0/0.0.0.0, the interface is disabled or did not receive an IP address and subnet mask via DHCP.
	If this interface is a member of an active virtual router, this field displays the IP address it is currently using. This is either the static IP address of the interface (if it is the master) or the management IP address (if it is a backup).
Action	Use this field to get or to update the IP address for the interface.
	Click Renew to send a new DHCP request to a DHCP server.
	Click the Connect icon to have the ZyWALL try to connect a PPPoE/PPTP interface. If the interface cannot use one of these ways to get or to update its IP address, this field displays n/a .
	Click the Disconnect icon to stop a PPPoE/PPTP connection.
Top 5 Firewall Rules that blocked IPv4 (IPv6) Traffic	This section displays the most triggered five firewall rules that caused the ZyWALL to block .
#	This is the entry's rank in the list of the most commonly triggered firewall rules.
Priority	This is the position of the triggered firewall rule in the global rule list. The ordering of firewall rules is important as rules are applied in sequence.
From	This shows the zone packets came from that the triggered firewall rule.
То	This shows the zone packets went to that the triggered firewall rule.
Description	This field displays the descriptive name (if any) of the triggered firewall rule.
Hits	This field displays how many times the firewall rule was triggered.
Schedule	This field displays the schedule object of the triggered firewall rule.
User	This is the user name or user group name of the triggered firewall rule.
IPv4 (IPv6) Source	This displays the source IPv4 (IPv6) address object of the triggered firewall rule.
IPv4 (IPv6) Destination	This displays the destination IPv4 (IPv6) address object of the triggered firewall rule.
Service	This displays the service object of the triggered firewall rule.
Access	This field displays whether the triggered firewall rule denied (silently discarded) or rejected the passage of packets of the triggered firewall rule.

LABEL	DESCRIPTION
Logs	This field displays whether a log (and alert) was created for the triggered firewall rule.
The Latest Alert Logs	These fields display recent logs generated by the ZyWALL.
#	This is the entry's rank in the list of alert logs.
Time	This field displays the date and time the log was created.
Priority	This field displays the severity of the log.
Category	This field displays the type of log generated.
Message	This field displays the actual log message.
Source	This field displays the source address (if any) in the packet that generated the log.
Destination	This field displays the destination address (if any) in the packet that generated the log.
Protocol	This field displays the service protocol in the packet that generated the log.
Note	This field displays descriptive information (if any) of the log.

Table 14 Dashboard (continued)

5.2.1 The CPU Usage Screen

Use this screen to look at a chart of the ZyWALL's recent CPU usage. To access this screen, click **CPU Usage** in the dashboard.





The following table describes the labels in this screen.

Table 15	Dashboard	>	CPU	Usage
----------	-----------	---	-----	-------

LABEL	DESCRIPTION
	The y-axis represents the percentage of CPU usage.
	The x-axis shows the time period over which the CPU usage occurred
Refresh Interval	Enter how often you want this window to be automatically updated.
Refresh	Click this to update the information in the window right away.
5.2.2 The Memory Usage Screen

Use this screen to look at a chart of the ZyWALL's recent memory (RAM) usage. To access this screen, click **Memory Usage** in the dashboard.



The following table describes the labels in this screen.

LABEL	DESCRIPTION
	The y-axis represents the percentage of RAM usage.
	The x-axis shows the time period over which the RAM usage occurred
Refresh Interval	Enter how often you want this window to be automatically updated.
Refresh	Click this to update the information in the window right away.

Table 16 Dashboard > Memory Usage

5.2.3 The Active Sessions Screen

Use this screen to look at a chart of the ZyWALL's recent traffic session usage. To access this screen, click **Session Usage** in the dashboard.



The following table describes the labels in this screen.

LABEL	DESCRIPTION
Sessions	The y-axis represents the number of session.
	The x-axis shows the time period over which the session usage occurred
Refresh Interval	Enter how often you want this window to be automatically updated.
Refresh	Click this to update the information in the window right away.

 Table 17
 Dashboard > Session Usage

5.2.4 The VPN Status Screen

Use this screen to look at the VPN tunnels that are currently established. To access this screen, click **VPN Status** in **System Status** in the dashboard.

1	VPN Status				
#	Name 🔺	Encapsulation		Algorithm	
Refre	sh Interval:	5 minutes 🔻	Refresh		

Figure 55 Dashboard > System Status > VPN Status

The following table describes the labels in this screen.

LABEL	DESCRIPTION
#	This field is a sequential value, and it is not associated with a specific SA.
Name	This field displays the name of the IPSec SA.
Encapsulation	This field displays how the IPSec SA is encapsulated.
Algorithm	This field displays the encryption and authentication algorithms used in the SA.
Refresh Interval	Select how often you want this window to be updated automatically.
Refresh	Click this to update the information in the window right away.

 Table 18
 Dashboard > VPN Status

5.2.5 The DHCP Table Screen

Use this screen to look at the IP addresses currently assigned to DHCP clients and the IP addresses reserved for specific MAC addresses. To access this screen, click **DHCP Table** in **System Status** in the dashboard.

Figure 56 Dashboard > System Status > DHCP Table

0	HCP Table					
#	Interface 🔺	IP Address	Host Name	MAC Address	Description	Reserve
1	lan1	192.168.1.33	"twpc11746-01"	00:0f:fe:1e:4a:e0		

 Table 19
 Dashboard > DHCP Table

LABEL	DESCRIPTION
#	This field is a sequential value, and it is not associated with a specific entry.
Interface	This field identifies the interface that assigned an IP address to a DHCP client.
IP Address	This field displays the IP address currently assigned to a DHCP client or reserved for a specific MAC address. Click the column's heading cell to sort the table entries by IP address. Click the heading cell again to reverse the sort order.
Host Name	This field displays the name used to identify this device on the network (the computer name). The ZyWALL learns these from the DHCP client requests. "None" shows here for a static DHCP entry.
MAC Address	This field displays the MAC address to which the IP address is currently assigned or for which the IP address is reserved. Click the column's heading cell to sort the table entries by MAC address. Click the heading cell again to reverse the sort order.
Description	For a static DHCP entry, the host name or the description you configured shows here. This field is blank for dynamic DHCP entries.
Reserve	If this field is selected, this entry is a static DHCP entry. The IP address is reserved for the MAC address.
	If this field is clear, this entry is a dynamic DHCP entry. The IP address is assigned to a DHCP client.
	To create a static DHCP entry using an existing dynamic DHCP entry, select this field, and then click Apply .
	To remove a static DHCP entry, clear this field, and then click Apply .

5.2.6 The Number of Login Users Screen

Use this screen to look at a list of the users currently logged into the ZyWALL. Users who close their browsers without logging out are still shown as logged in here. To access this screen, click **Number of Login Users** in **System Status** in the dashboard or **Monitor > Login User**.

Figure 57 Dashboard > System Status > Number of Login Users

rren	t User List				
S F	orce Logout				
#	User ID	Reauth Lease T.	Туре	IP Address	User Info
0	a1	23:58:48 / 23:59:06	http/https	192.168.191.35	limited-admin
1	a2	23:59:21/23:59:21	http/https	192.168.191.35	user
2	a3	23:59:33 / 23:59:33	http/https	192.168.191.35	guest
3	admin	unlimited / 00:20:29	http/https	192.168.191.33	admin
4	admin	unlimited / 00:28:22	console	console	admin
5	admin	unlimited / 00:29:59	http/https	192.168.191.133	admin
6	test1	23:58:15 / 23:58:15	http/https	192.168.191.35	ext-user,ext-group-user(testg),ext-group-user(te

LABEL	DESCRIPTION
#	This field is a sequential value and is not associated with any entry.
User ID	This field displays the user name of each user who is currently logged in to the ZyWALL.
Reauth Lease T.	This field displays the amount of reauthentication time remaining and the amount of lease time remaining for each user. See Chapter 27 on page 361 for more information.
Туре	This field displays the way the user logged in to the ZyWALL.
IP address	This field displays the IP address of the computer used to log in to the ZyWALL.
User Info	This field displays the types of user accounts the ZyWALL uses. If the user type is ext-user (external user), this field will show its external-group information when you move your mouse over it. If the external user matches two external-group objects, both external-group object names will be shown.
Force Logout	Click this icon to end a user's session.

Table 20Dashboard > Number of Login Users

PART II Technical Reference

Monitor

6.1 Overview

Use the Monitor screens to check status and statistics information.

6.1.1 What You Can Do in this Chapter

Use the **Monitor** screens for the following.

- Use the **System Status > Port Statistics** screen (see Section 6.2 on page 80) to look at packet statistics for each physical port.
- Use the **System Status > Port Statistics > Graph View** screen (see Section 6.2 on page 80) to look at a line graph of packet statistics for each physical port.
- Use the **System Status** > Interface Status screen (Section 6.3 on page 82) to see all of the ZyWALL's interfaces and their packet statistics.
- Use the **System Status** > **Traffic Statistics** screen (see Section 6.4 on page 86) to start or stop data collection and view statistics.
- Use the **System Status** > **Session Monitor** screen (see Section 6.5 on page 89) to view sessions by user or service.
- Use the **System Status** > **DDNS Status** screen (see Section 6.6 on page 91) to view the status of the ZyWALL's DDNS domain names.
- Use the **System Status** > **IP/MAC Binding** screen (Section 6.7 on page 91) to view a list of devices that have received an IP address from ZyWALL interfaces with IP/MAC binding enabled.
- Use the **System Status** > **Login Users** screen (Section 6.8 on page 92) to look at a list of the users currently logged into the ZyWALL.
- Use the **System Status** > **Cellular Status** screen (Section 6.9 on page 93) to check your 3G connection status.
- Use the **System Status** > **USB Storage** screen (Section 6.10 on page 96) to view information about a connected USB storage device.
- Use the **VPN Monitor** > **IPSec** screen (Section 6.11 on page 97) to display and manage active IPSec SAs.
- Use the **VPN Monitor** > **SSL** screen (see Section 6.12 on page 99) to list the users currently logged into the VPN SSL client portal. You can also log out individual users and delete related session information.
- Use the VPN Monitor > L2TP over IPSec screen (see Section 6.13 on page 99) to display and manage the ZyWALL's connected L2TP VPN sessions.
- Use the **Log** (Section 6.14 on page 100) screen to view the ZyWALL's current log messages. You can change the way the log is displayed, you can e-mail the log, and you can also clear the log in this screen.

6.2 The Port Statistics Screen

Use this screen to look at packet statistics for each Gigabit Ethernet port. To access this screen, click **Monitor > System Status > Port Statistics**.

Figure 58 Monitor > System Status > Port Statistics

Rx B/s	Up Time
0	00:00:00
1758	46:07:52
0	00:00:00
0	00:00:00
0	00:00:00
	Rx B/s 0 1758 0 0

The following table describes the labels in this	screen.
--	---------

LABEL	DESCRIPTION
Poll Interval	Enter how often you want this window to be updated automatically, and click Set Interval .
Set Interval	Click this to set the Poll Interval the screen uses.
Stop	Click this to stop the window from updating automatically. You can start it again by setting the Poll Interval and clicking Set Interval .
Switch to Graphic View	Click this to display the port statistics as a line graph.
#	This field displays the port's number in the list.
Port	This field displays the physical port number.
Status	This field displays the current status of the physical port.
	Down - The physical port is not connected.
	Speed / Duplex - The physical port is connected. This field displays the port speed and duplex setting (Full or Half).
TxPkts	This field displays the number of packets transmitted from the ZyWALL on the physical port since it was last connected.
RxPkts	This field displays the number of packets received by the ZyWALL on the physical port since it was last connected.
Collisions	This field displays the number of collisions on the physical port since it was last connected.
Tx B/s	This field displays the transmission speed, in bytes per second, on the physical port in the one-second interval before the screen updated.
Rx B/s	This field displays the reception speed, in bytes per second, on the physical port in the one-second interval before the screen updated.
Up Time	This field displays how long the physical port has been connected.
System Up Time	This field displays how long the ZyWALL has been running since it last restarted or was turned on.

 Table 21
 Monitor > System Status > Port Statistics

6.2.1 The Port Statistics Graph Screen

Use this screen to look at a line graph of packet statistics for each physical port. To access this screen, click **Port Statistics** in the **Status** screen and then the **Switch to Graphic View Button**.



Figure 59 Monitor > System Status > Port Statistics > Switch to Graphic View

LABEL	DESCRIPTION
Refresh Interval	Enter how often you want this window to be automatically updated.
Refresh Now	Click this to update the information in the window right away.
Port Selection	Select the number of the physical port for which you want to display graphics.
Switch to Grid View	Click this to display the port statistics as a table.
bps	The y-axis represents the speed of transmission or reception.
time	The x-axis shows the time period over which the transmission or reception occurred
ТХ	This line represents traffic transmitted from the ZyWALL on the physical port since it was last connected.
RX	This line represents the traffic received by the ZyWALL on the physical port since it was last connected.
Last Update	This field displays the date and time the information in the window was last updated.
System Up Time	This field displays how long the ZyWALL has been running since it last restarted or was turned on.

 Table 22
 Monitor > System Status > Port Statistics > Switch to Graphic View

6.3 Interface Status Screen

This screen lists all of the ZyWALL's interfaces and gives packet statistics for them. Click **Monitor > System Status > Interface Status** to access this screen.

erface Summa	ary							
erface Status								
Name	Port	Status	HA St	Zone	IP Addr/Netmask	IP Assignment	Services	Action
ge1	P1	Down	n/a	LAN	192.168.1.1 / 255.255.255	.0 Static	DHCP server	n/a
ge1 ppp	P1	Inactive	n/a	n/a	0.0.0.0 / 0.0.0.0	Dynamic	n/a	n/a
🖻 ge2	P2	100M/Full	n/a	WAN	172.23.26.243 / 255.255.2	55.0 Static	n/a	n/a
ge2 ppp	P2	Inactive	n/a	n/a	0.0.0.0 / 0.0.0.0	Dynamic	n/a	n/a
⊒ <u>qe3</u>	P3	Down	n/a	WAN	0.0.0.0 / 0.0.0.0	DHCP client	n/a	Renew
ge3 ppp	P3	Inactive	n/a	n/a	0.0.0.0 / 0.0.0.0	Dynamic	n/a	n/a
🖻 <u>qe4</u>	P4	Down	n/a	DMZ	192.168.2.1 / 255.255.255	0 Static	n/a	n/a
ge4 ppp	P4	Inactive	n/a	n/a	0.0.0.0 / 0.0.0.0	Dynamic	n/a	n/a
🗏 <u>ge5</u>	P5	Down	n/a	DMZ	192.168.3.1 / 255.255.255	0 Static	n/a	n/a
ge5 ppp	P5	Inactive	n/a	n/a	0.0.0.0 / 0.0.0.0	Dynamic	n/a	n/a
ge6	P6	Down	n/a	WLAN	10.59.0.1 / 255.255.255.0	Static	DHCP server	n/a
ge6 ppp	P6	Inactive	n/a	n/a	0.0.0.0 / 0.0.0.0	Dynamic	n/a	n/a
🖃 <u>qe7</u>	P7	Down	n/a	n/a	0.0.0.0 / 0.0.0.0	Static	n/a	n/a
ge7 ppp	P7	Inactive	n/a	n/a	0.0.0.0 / 0.0.0.0	Dynamic	n/a	n/a
aux	aux	Inactive	n/a	n/a	0.0.0.0/0.0.0.0	Dynamic	n/a	n/a
J wlan-1	n/a	n/a	n/a	n/a	n/a / n/a	n/a	n/a	n/a
wlan-1-1	n/a	Down	n/a	LAN1	192.168.1.100 / 255.255.2	55.0 static	n/a	n/a
inel Interface Name	Status Stat Zone	IP Address		My Ad	idress R	emote Gatway Address	Mode	A
nnel Interface Name 16 Interface Si	Status Stat Zone tatus	IP Address		My Ad	ldress R	emote Gatway Address	Mode	A
nnel Interface Name 16 Interface Si Name	Status Stat Zone tatus Por	IP Address Status	Zone	My Ad	idress R	emote Gatway Address	Mode	A
nnel Interface Name 16 Interface Si Name 3 <u>ae1</u>	Status Stat Zone tatus Por P1	IP Address Status Down	Zone	My Ad IP Address	idress R	emote Gatway Address	Mode Services n/a,n/a	A Action n/a
nnel Interface Name 6 Interface SI Name a <u>ge1</u> <u>ge1 ppp</u>	Status Stat Zone tatus Por P1 P1	IP Address Status Down Inactive	Zone LAN LAN	My Ad IP Address :: ::	idress R	emote Gatway Address	Mode Services n/a,n/a n/a,n/a	A Action n/a n/a
Name 6 Interface SI Name a <u>ae1</u> <u>ae1</u> <u>ae1</u> <u>ae1</u> <u>ae2</u>	Status Stat Zone tatus Por P1 P1 P2	IP Address Status Down Inactive Inactive	Zone LAN LAN WAN	My Ad IP Address :: ::	idress R	emote Gatway Address	Mode Services n/a,n/a n/a,n/a n/a,n/a	A Action n/a n/a
Interface Name G Interface Sime G optimized Ge1 Ge2	Status Stat Zone tatus Por P1 P1 P2 P2 P2	IP Address Status Down Inactive Inactive	Zone LAN LAN WAN WAN	My Ad IP Address :: :: ::	idress R	emote Gatway Address	Mode Services n/a,n/a n/a,n/a n/a,n/a n/a,n/a	A Action n/a n/a n/a n/a
Anne Interface Si Anne 6 Interface Si Name 9 ge1 9 ge2 9 ge2 9 ge2 9 ge3	Status Stat Zone tatus Por P1 P1 P2 P2 P2 P3	IP Address Status Down Inactive Inactive Down	Zone LAN LAN WAN WAN	My Ad IP Address :: :: :: ::	idress R	emote Gatway Address	Mode	A Action n/a n/a n/a n/a n/a
Anne Interface Si Anne 6 Interface Si Name 9 ge1 9 ge2 9 ge2 9 ge3 9 ge3 9 ge3 9 ge3 9 ge3 9 ge3	Status Status Catus Catus Por P1 P1 P2 P2 P2 P3 P3 P3 P3	IP Address Status Down Inactive Inactive Down Inactive Dow	Zone LAN LAN WAN WAN WAN	My Ad IP Address :: :: :: :: ::	dress R	emote Gatway Address	Mode	A Action n/a n/a n/a n/a n/a n/a
Anne Interface Si Anne 6 Interface Si Name 9 <u>ge1</u> 9 <u>ge2</u> 9 <u>ge2</u> 9 <u>ge3</u> 9 <u>ge3</u> 9 <u>ge3</u> 9 <u>ge3</u> 9 <u>ge4</u>	Status Status Catus Catus Catus Por P1 P1 P2 P2 P2 P3 P3 P3 P3 P3 P3	IP Address Status Down Inactive Inactive Down Inactive I	Zone LAN LAN WAN WAN WAN WAN DMZ	My Address	dress R	emote Gatway Address	Mode	A Action n/a n/a n/a n/a n/a n/a n/a n/
Anne Interface Si Anne Ginterface Si Anne Ginterface Si Anne Get Dep Get Dep Ge	Status Status Catus Catus Catus Por P1 P1 P2 P2 P2 P2 P3 P3 P3 P4 P4	P Address Status Down Inactive Inactive Down Inactive	Zone LAN LAN WAN WAN WAN DMZ DMZ	My Address	dress R	emote Gatway Address	Mode	A Action n/a n/a n/a n/a n/a n/a n/a n/
Anne Interface Si Anne 6 Interface Si Anne 9 de1 1 de1 ppp 9 de2 1 de2 ppp 9 de3 1 de3 ppp 9 de4 1 de4 ppp 9 de4	Status Zone Status Por tatus P1 P1 P1 P2 P3 P4 P4 P5 P4	IP Address Status Down Inactive Inactive Down Inactive Inactiv	Zone LAN LAN WAN WAN WAN DMZ DMZ DMZ	My Address	idress R	emote Gatway Address	Mode	A Action n/a n/a n/a n/a n/a n/a n/a n/
Anne Interface SI Anne 6 Interface SI Name 9 Ge1 - Ge1 DDD 9 Ge2 - Ge2 DDD 9 Ge2 - Ge2 DDD 9 Ge3 - Ge3 DDD 9 Ge4 - Ge5 DDD - Ge5 DDD	Status Zone Status Por tatus P1 P1 P1 P2 P3 P4 P5 P5 P5	IP Address Status Down Inactive Inactive Down Inactive Do	Zone LAN LAN WAN WAN WAN DMZ DMZ DMZ	My Address	idress R	emote Gatway Address	Mode	A Action n/a n/a n/a n/a n/a n/a n/a n/
Name 6 Interface SI Name 9 ge1 9 ge2 9 ge2 9 ge2 9 ge3 9 ge4 9 ge4 9 ge5 9 ge5 9 ge5 9 ge6	Status Zone Status Por tatus P1 P1 P1 P2 P3 P4 P5 P5 P6	IP Address Status Down Inactive Inactive Down Inactive	Zone LAN LAN WAN WAN WAN DMZ DMZ DMZ DMZ WLAN	My Address	idress R	emote Gatway Address	Mode Services n/a,n/a	A Action Na
Name 6 Interface SI Name 9 ge1 9 ge2 9 ge2 9 ge2 9 ge3 9 ge3 9 ge4 9 ge4 9 ge5 9 ge5 9 ge6 9 g	Status Zone Status Por tatus P1 P1 P1 P2 P3 P4 P5 P5 P6 P6 P6	IP Address Status Down Inactive Inactive Down Inactive Inactive Down Inactive	Zone LAN LAN WAN WAN WAN DMZ DMZ DMZ DMZ WLAN	My Address	idress R	emote Gatway Address	Mode	A Action n/a
Name 6 Interface SI Name 9 ge1 9 ge2 9 ge2 9 ge2 9 ge2 9 ge3 9 ge4 9 ge4 9 ge5 9 ge6 1 ge6 ppp 9 ge6 1 ge6 ppp 9 ge6 1 ge6 ppp 9 ge7 9 g	Status Zone Status Por tatus P1 P1 P1 P2 P3 P4 P5 P5 P6 P6 P7	IP Address Status Down Inactive Inactive Down Inactive In	Zone LAN LAN WAN WAN WAN DMZ DMZ DMZ DMZ UMZ WLAN WLAN r/a	My Address	idress R	emote Gatway Address	Mode Services n/a,n/a	A A n/a
anel Interface Name 6 Interface 9 ge1 - ge1 ppp - ge2 ppp - ge3 ppp - ge4 ppp - ge5 ppp - ge6 ppp - ge6 ppp - ge7 ppp - ge7 ppp	Status Zone Status Por catus Por	IP Address Status Down Inactive Inactive Down Inactive Inactive Down Inactive Down Inactive Down Inactive Down Inactive Down Inactive Down Inactive Inac	Zone LAN LAN WAN WAN WAN DMZ DMZ DMZ DMZ DMZ WLAN WLAN n/a n/a	My Address	idress R	emote Gatway Address	Mode Services n/a,n/a	A Action n/a
Name Annel Interface SI Name Contraction of the second s	Status Zone Status Por tatus Por 2 P1 2 P2 2 P3 4 P5 5 P6 6 P7 77 P7	P Address Status Down Inactive Inactive Down Inactive	Zone LAN LAN WAN WAN WAN DMZ DMZ DMZ DMZ DMZ WLAN WLAN n/a n/a	My Address	idress R	emote Gatway Address	Mode	A n/a
Anel Interface Name 6 Interface Si Name 9 0e1 9 0e2 9 0e2 9 0e3 9 0e4 1 0 0e3 9 0e5 1 0e5	Status Zone Status Por tatus Por P1 P1 P2 P3 P3 P4 P4 P5 P6 P6 P7 P7 Status P7 Status P7 Status P7 Status P6 P6 P6 P7 P7 Status P7 P7 P7 Status P7 P7 P7	P Address Status Down Inactive Inactive Down Inactive	Zone LAN LAN WAN WAN WAN DMZ DMZ DMZ DMZ DMZ UMZ WLAN WLAN n/a	My Address	dress R	emote Gatway Address	Mode	A Action n/a n/a n/a n/a n/a n/a n/a n/a n/a n/
Anne Interface Si Name 6 Interface Si Name 9 001 000 9 002 000 9 002 000 9 002 000 9 003 9 004 9 005 9 000 9 005 9	Status Zone status Por tatus P1 P1 P1 P2 P3 P4 P4 P5 P6 P6 P7 P7 P7 ics Status	IP Address Status Down Inactive Inactive Down Inactive Down Inactive Down Inactive Down Inactive Down Inactive Down Inactive Inactive Inactive Down Inactive Down Inactive I	Zone LAN LAN WAN WAN WAN DMZ DMZ DMZ DMZ DMZ WLAN WLAN n/a n/a	My Adress	dress RxPkts	Tx B/s	Mode	A Action n/a n/a n/a n/a n/a n/a n/a n
Name G Interface SI Name G Interface SI Name G of Dep	Status Zone Stat Zone tatus Por P1 P1 P2 P2 P3 P4 P4 P4 P5 P6 P6 P7 P6 P7 P6 P6 P6 P6 P6 P6 P6 P6 P6 P7 P7 P7	IP Address Status Down Inactive Inactive Down Inactive	Zone LAN LAN WAN WAN WAN DMZ DMZ DMZ DMZ DMZ UMZ WLAN WLAN n/a	My Adress	dress RxPkts	Tx B/s 0	Mode Services n/a,n/a	A Action n/a n/a n/a n/a n/a n/a n/a n/a n/a n/
Interface Name 6 Interface 9 ge1 9 ge2 9 ge3 9 ge4 9 ge3 9 ge4 9 ge5 9 ge5 9 ge6 9 ge7 9 ge1 9 ge1 9 ge2	Status Zone Stat Zone tatus Por P1 P1 P2 P2 P3 P4 P4 P5 P6 P7 P6 P7 P6 P5 P7 P7 P7 P7 P6 P6 P7 P7 P7 P7 P6 P5 P6 P6 P7 P7 P7 P7	IP Address Status Down Inactive Inactive Down Inactive I	Zone LAN WAN WAN WAN DMZ DMZ DMZ DMZ DMZ WLAN WLAN n/a	My Ad	dress RxPkts 0 2286446	Tx B/s 0 21044	Mode Services n/a,n/a n/a,n/a	A Action n/a n/a n/a n/a n/a n/a n/a n/a n/a n/
nnel Interface Name fo Interface SI Name g ge1 g ge2 g ge3 g ge4	Status Zone Status Por tatus P1 P1 P1 P2 P3 P4 P4 P5 P6 P4 P5 P6 P7 P6 P5 P7 P5 P6 P5 P6 P5 P6 P5 P6 P5 P7 P7 P6 P6 P6 P5 P6 P6 P7	IP Address Status Down Inactive Inactive Inactive Down Inactive	Zone LAN WAN WAN WAN DMZ DMZ DMZ DMZ DMZ WLAN WLAN n/a n/a	My Ad	Idress R RxPkts 0 2286446 0	Tx B/s 0 21044 0	Mode Services n/a,n/a	A Action n/a n/a

0

0

0

n/a

0

0

0

n/a

⊕ ge7

aux

. ⊞∙ wlan-1

Down

Down

Inactive

n/a

6

4

0

n/a

0

0

0

n/a

Each field is described in the following table.

LABEL	DESCRIPTION
Interface Status	If an Ethernet interface does not have any physical ports associated with it, its entry is displayed in light gray text.
Expand/Close	Click this button to show or hide statistics for all the virtual interfaces on top of the Ethernet interfaces.
Name	This field displays the name of each interface. If there is an Expand icon (plus-sign) next to the name, click this to look at the status of virtual interfaces on top of this interface.
Port	This field displays the physical port number.
Status	This field displays the current status of each interface. The possible values depend on what type of interface it is.
	For Ethernet interfaces:
	Inactive - The Ethernet interface is disabled.
	Down - The Ethernet interface is enabled but not connected.
	Speed / Duplex - The Ethernet interface is enabled and connected. This field displays the port speed and duplex setting (Full or Half).
	For cellular (3G) interfaces, see Section 6.10 on page 96 the Web Help for the status that can appear.
	For virtual interfaces, this field always displays Up . If the virtual interface is disabled, it does not appear in the list.
	For VLAN and bridge interfaces, this field always displays Up . If the VLAN or bridge interface is disabled, it does not appear in the list.
	For PPP interfaces:
	Connected - The PPP interface is connected.
	Disconnected - The PPP interface is not connected.
	If the PPP interface is disabled, it does not appear in the list.
Zone	This field displays the zone to which the interface is assigned.
IP Addr/Netmask	This field displays the current IP address and subnet mask assigned to the interface. If the IP address and subnet mask are 0.0.0.0, the interface is disabled or did not receive an IP address and subnet mask via DHCP.
	If this interface is a member of an active virtual router, this field displays the IP address it is currently using. This is either the static IP address of the interface (if it is the master) or the management IP address (if it is a backup).
IP Assignment	This field displays how the interface gets its IP address.
	Static - This interface has a static IP address.
	DHCP Client - This interface gets its IP address from a DHCP server.
Services	This field lists which services the interface provides to the network. Examples include DHCP relay , DHCP server , DDNS , RIP , and OSPF . This field displays n/a if the interface does not provide any services to the network.
Action	Use this field to get or to update the IP address for the interface. Click Renew to send a new DHCP request to a DHCP server. Click Connect to try to connect a PPPoE/PPTP interface. If the interface cannot use one of these ways to get or to update its IP address, this field displays n/a .
Tunnel Interface Status	This displays the details of the ZyWALL's configured tunnel interfaces.
Name	This field displays the name of the interface.

 Table 23
 Monitor > System Status > Interface Status

LABEL	DESCRIPTION
Status	The activate (light bulb) icon is lit when the entry is active and dimmed when the entry is inactive.
Zone	This field displays the zone to which the interface is assigned.
IP Address	This is the IP address of the interface. If the interface is active (and connected), the ZyWALL tunnels local traffic sent to this IP address to the Remote Gateway Address .
My Address	This is the interface or IP address uses to identify itself to the remote gateway. The ZyWALL uses this as the source for the packets it tunnels to the remote gateway.
Remote Gateway Address	This is the IP address or domain name of the remote gateway to which this interface tunnels traffic.
Mode	This field displays the tunnel mode that you are using.
Action	This field lists which services the interface provides to the network. This field displays n/a if the interface does not provide any services to the network.
IPv6 Interface Status	This section displays the status of the IPv6 interface. If an Ethernet interface does not have any physical ports associated with it, its entry is displayed in light gray text.
Expand/Close	Click this button to show or hide statistics for all the virtual interfaces on top of the Ethernet interfaces.
Name	This field displays the name of each interface. If there is an Expand icon (plus-sign) next to the name, click this to look at the status of virtual interfaces on top of this interface.
Port	This field displays the physical port number.
Status	This field displays the current status of each interface. The possible values depend on what type of interface it is.
	For Ethernet interfaces:
	Inactive - The Ethernet interface is disabled.
	Down - The Ethernet interface is enabled but not connected.
	Speed / Duplex - The Ethernet interface is enabled and connected. This field displays the port speed and duplex setting (Full or Half).
	For cellular (3G) interfaces, see Section 6.9 on page 93 the Web Help for the status that can appear.
	For virtual interfaces, this field always displays Up . If the virtual interface is disabled, it does not appear in the list.
	For VLAN and bridge interfaces, this field always displays Up . If the VLAN or bridge interface is disabled, it does not appear in the list.
	For PPP interfaces:
	Connected - The PPP interface is connected.
	Disconnected - The PPP interface is not connected.
	If the PPP interface is disabled, it does not appear in the list.
HA Status	This field displays the status of the interface in the virtual router.
	Active - This interface is the master interface in the virtual router.
	Stand-By - This interface is a backup interface in the virtual router.
	Fault - This VRRP group is not functioning in the virtual router right now. For example, this might happen if the interface is down.
	n/a - Device HA is not active on the interface.
Zone	This field displays the zone to which the interface is assigned.

Table 23	Monitor	> System	Status >	Interface	Status	(continued)
			0.00.00		0.000	(001101000)

LABEL	DESCRIPTION
IP Address	This field displays the current IPv6 address assigned to the interface. If the IPv6 address is not displayed, the interface is disabled or did not receive an IPv6 address via DHCP.
	If this interface is a member of an active virtual router, this field displays the IP address it is currently using. This is either the static IP address of the interface (if it is the master) or the management IP address (if it is a backup).
IP Assignment	This field displays how the interface gets its IP address.
	Static - This interface has a static IP address.
	DHCP Client - This interface gets its IP address from a DHCP server.
Services	This field lists which services the interface provides to the network. Examples include DHCP relay , DHCP server , DDNS , RIP , and OSPF . This field displays n/a if the interface does not provide any services to the network.
Action	Use this field to get or to update the IP address for the interface. Click Renew to send a new DHCP request to a DHCP server. Click Connect to try to connect a PPPoE/PPTP interface. If the interface cannot use one of these ways to get or to update its IP address, this field displays n/a .
Interface Statistics	This table provides packet statistics for each interface.
Refresh	Click this button to update the information in the screen.
Expand/Close	Click this button to show or hide statistics for all the virtual interfaces on top of the Ethernet interfaces.
Name	This field displays the name of each interface. If there is a Expand icon (plus-sign) next to the name, click this to look at the statistics for virtual interfaces on top of this interface.
Status	This field displays the current status of the interface.
	Down - The interface is not connected.
	Speed / Duplex - The interface is connected. This field displays the port speed and duplex setting (Full or Half).
	This field displays Connected and the accumulated connection time (hh:mm:ss) when the PPP interface is connected.
TxPkts	This field displays the number of packets transmitted from the ZyWALL on the interface since it was last connected.
RxPkts	This field displays the number of packets received by the ZyWALL on the interface since it was last connected.
Tx B/s	This field displays the transmission speed, in bytes per second, on the interface in the one- second interval before the screen updated.
Rx B/s	This field displays the reception speed, in bytes per second, on the interface in the one- second interval before the screen updated.

 Table 23
 Monitor > System Status > Interface Status (continued)

6.4 The Traffic Statistics Screen

Click **Monitor > System Status > Traffic Statistics** to display the **Traffic Statistics** screen. This screen provides basic information about the following for example:

- Most-visited Web sites and the number of times each one was visited. This count may not be accurate in some cases because the ZyWALL counts HTTP GET packets. Please see Table 24 on page 87 for more information.
- · Most-used protocols or service ports and the amount of traffic on each one

· LAN IP with heaviest traffic and how much traffic has been sent to and from each one

You use the **Traffic Statistics** screen to tell the ZyWALL when to start and when to stop collecting information for these reports. You cannot schedule data collection; you have to start and stop it manually in the **Traffic Statistics** screen.

affic ata C	ollection				
A atist	ollect Statistics	sii Reset	nce 2009-08-26	i Wed 09:00:41 to	2009-08-26 Wed 09:00:42
Inter	face: w	an2		~	
Top:	S	ervice/Po	rt	¥ [Refresh Flush Data
#	Service/Port		Protocol	Direction	Amount
1	https(Port : 4-	43)	TCP	Ingress	125.156(KBytes)
2	https(Port: 4	43)	TCP	Egress	70.154(KBytes)
3	www(Port : I	30)	TCP	Ingress	24.015(KBytes)
4	others(Port :	1139)	TCP	Egress	8.329(KBytes)
5	others(Port :	1155)	TCP	Egress	5.837(KBytes)
6	others(Port :	1139)	TCP	Ingress	5.538(KBytes)
7	others(Port :	1155)	TCP	Ingress	3.930(KBytes)
8	www(Port : I	30)	TCP	Egress	2.656(KBytes)
9	others(Port :	1079)	TCP	Egress	2.337(KBytes)
10	loc-srv(Port :	135)	TCP	Egress	1.268(KBytes)
11	loc-srv(Port :	135)	TCP	Ingress	1.032(KBytes)
12	microsoft-ds(Port: 44	TCP	Ingress	656(Bytes)
13	others(Port :	1079)	TCP	Ingress	638(Bytes)
14	microsoft-ds(Port : 44	TCP	Egress	240(Bytes)
15	ntp(Port : 123	0	UDP	Egress	96(Bytes)
	atu/Daut : 400	0	LIDP	Ingress	96(Bytes)

Figure 61 Monitor > System Status > Traffic Statistics

There is a limit on the number of records shown in the report. Please see Table 25 on page 89 for more information. The following table describes the labels in this screen.

 Table 24
 Monitor > System Status > Traffic Statistics

LABEL	DESCRIPTION			
Data Collection				
Collect Statistics	Select this to have the ZyWALL collect data for the report. If the ZyWALL has already been collecting data, the collection period displays to the right. The progress is not tracked here real-time, but you can click the Refresh button to update it.			
Apply	Click Apply to save your changes back to the ZyWALL.			
Reset	Click Reset to return the screen to its last-saved settings.			
Statistics				
Interface	Select the interface from which to collect information. You can collect information from Ethernet, VLAN, bridge and PPPoE/PPTP interfaces.			

LABEL	DESCRIPTION
Traffic Type	Select the type of report to display. Choices are:
	Host IP Address/User - displays the IP addresses or users with the most traffic and how much traffic has been sent to and from each one.
	Service/Port - displays the most-used protocols or service ports and the amount of traffic for each one.
	Web Site Hits - displays the most-visited Web sites and how many times each one has been visited.
	Each type of report has different information in the report (below).
Refresh	Click this button to update the report display.
Flush Data	Click this button to discard all of the screen's statistics and update the report display.
	These fields are available when the Traffic Type is Host IP Address/User.
#	This field is the rank of each record. The IP addresses and users are sorted by the amount of traffic.
IP Address/User	This field displays the IP address or user in this record. The maximum number of IP addresses or users in this report is indicated in Table 25 on page 89.
Direction	This field indicates whether the IP address or user is sending or receiving traffic.
	Ingress- traffic is coming from the IP address or user to the ZyWALL.
	Egress - traffic is going from the ZyWALL to the IP address or user.
Amount	This field displays how much traffic was sent or received from the indicated IP address or user. If the Direction is Ingress , a red bar is displayed; if the Direction is Egress , a blue bar is displayed. The unit of measure is bytes, Kbytes, Mbytes or Gbytes, depending on the amount of traffic for the particular IP address or user. The count starts over at zero if the number of bytes passes the byte count limit. See Table 25 on page 89.
	These fields are available when the Traffic Type is Service/Port .
#	This field is the rank of each record. The protocols and service ports are sorted by the amount of traffic.
Service/Port	This field displays the service and port in this record. The maximum number of services and service ports in this report is indicated in Table 25 on page 89.
Protocol	This field indicates what protocol the service was using.
Direction	This field indicates whether the indicated protocol or service port is sending or receiving traffic.
	Ingress - traffic is coming into the router through the interface
	Egress - traffic is going out from the router through the interface
Amount	This field displays how much traffic was sent or received from the indicated service / port. If the Direction is Ingress , a red bar is displayed; if the Direction is Egress , a blue bar is displayed. The unit of measure is bytes, Kbytes, Mbytes, Gbytes, or Tbytes, depending on the amount of traffic for the particular protocol or service port. The count starts over at zero if the number of bytes passes the byte count limit. See Table 25 on page 89.
	These fields are available when the Traffic Type is Web Site Hits .
#	This field is the rank of each record. The domain names are sorted by the number of hits.
Web Site	This field displays the domain names most often visited. The ZyWALL counts each page viewed on a Web site as another hit. The maximum number of domain names in this report is indicated in Table 25 on page 89.
Hits	This field displays how many hits the Web site received. The ZyWALL counts hits by counting HTTP GET packets. Many Web sites have HTTP GET references to other Web sites, and the ZyWALL counts these as hits too. The count starts over at zero if the number of hits passes the hit count limit. See Table 25 on page 89.

Table 24 Monitor > System Status > Traffic Statistics (continued)

The following table displays the maximum number of records shown in the report, the byte count limit, and the hit count limit.

Table 25 Maximum Values for Reports

LABEL	DESCRIPTION
Maximum Number of Records	20
Byte Count Limit	2 ⁶⁴ bytes; this is just less than 17 million terabytes.
Hit Count Limit	2^{64} hits; this is over 1.8 x 10^{19} hits.

6.5 The Session Monitor Screen

The **Session Monitor** screen displays all established sessions that pass through the ZyWALL for debugging or statistical analysis. It is not possible to manage sessions in this screen. The following information is displayed.

- · User who started the session
- · Protocol or service port used
- Source address
- Destination address
- Number of bytes received (so far)
- Number of bytes transmitted (so far)
- Duration (so far)

You can look at all established sessions that passed through the ZyWALL by user, service, source IP address, or destination IP address. You can also filter the information by user, protocol / service or service group, source address, and/or destination address and view it by user.

Click Monitor > System Status > Session Monitor to display the following screen.

ession							
View:		all sessions	~	Refresh			
User:				Service:	1	any	
Source Addr	ess:			Destination Ac	Idress:		
Search							
User 🔺	Service	Source	Destination	Rx	Tx	Duration	
User 🔺 admin	Service HTTP	Source 192.168.1.33:4	Destination 72.14.203.102	Rx 48 Bytes	Tx 6144 Bytes	Duration s 7826	
User 🔺 admin admin	Service HTTP Any_UDP	Source 192.168.1.33:4 192.168.1.33:4	Destination 72.14.203.102 172.23.5.2:88	Rx 48 Bytes 1024 Bytes	Tx 6144 Bytes 1024 Bytes	Duration s 7826 s 234	
User 🔺 admin admin admin	Service HTTP Any_UDP Any_UDP	Source 192.168.1.33:4 192.168.1.33:4 192.168.1.33:4	Destination 72.14.203.102 172.23.5.2.88 172.23.5.2.88	Rx 48 Bytes 1024 Bytes 1024 Bytes	Tx 6144 Bytes 1024 Bytes 1024 Bytes	Duration s 7826 s 234 s 212	
User 🔺 admin admin admin admin	Service HTTP Any_UDP Any_UDP Any_UDP	Source 192.168.1.33:4 192.168.1.33:4 192.168.1.33:4 192.168.1.33:4	Destination 72.14.203.102 172.23.5.2:88 172.23.5.2:88 172.23.5.2:12:	Rx 48 Bytes 1024 Bytes 1024 Bytes 96 Bytes	Tx 6144 Bytes 1024 Bytes 1024 Bytes 96 Bytes	Duration s 7826 s 234 s 212 216 216	

Figure 62 Monitor > System Status > Session Monitor

Table 26	Monitor :	> 5	System	Status	>	Session Monitor
	mornior ,	~ `	Jystern	Julus	_	30331011 1010111101

LABEL	DESCRIPTION
View	Select how you want the established sessions that passed through the ZyWALL to be displayed. Choices are:
	sessions by users - display all active sessions grouped by user
	sessions by services - display all active sessions grouped by service or protocol
	sessions by source IP - display all active sessions grouped by source IP address
	sessions by destination IP - display all active sessions grouped by destination IP address
	all sessions - filter the active sessions by the User , Service , Source Address , and Destination Address , and display each session individually (sorted by user).
Refresh	Click this button to update the information on the screen. The screen also refreshes automatically when you open and close the screen.
	The User , Service , Source Address , and Destination Address fields display if you view all sessions. Select your desired filter criteria and click the Search button to filter the list of sessions.
User	This field displays when View is set to all sessions . Type the user whose sessions you want to view. It is not possible to type part of the user name or use wildcards in this field; you must enter the whole user name.
Service	This field displays when View is set to all sessions . Select the service or service group whose sessions you want to view. The ZyWALL identifies the service by comparing the protocol and destination port of each packet to the protocol and port of each services that is defined. (See Chapter 29 on page 380 for more information about services.)
Source	This field displays when View is set to all sessions . Type the source IP address whose sessions you want to view. You cannot include the source port.
Destination	This field displays when View is set to all sessions . Type the destination IP address whose sessions you want to view. You cannot include the destination port.
Search	This button displays when View is set to all sessions . Click this button to update the information on the screen using the filter criteria in the User , Service , Source Address , and Destination Address fields.
User	This field displays the user in each active session.
	If you are looking at the sessions by users (or all sessions) report, click + or - to display or hide details about a user's sessions.
Service	This field displays the protocol used in each active session.
	If you are looking at the sessions by services report, click + or - to display or hide details about a protocol's sessions.
Source	This field displays the source IP address and port in each active session.
	If you are looking at the sessions by source IP report, click + or - to display or hide details about a source IP address's sessions.
Destination	This field displays the destination IP address and port in each active session.
	If you are looking at the sessions by destination IP report, click + or - to display or hide details about a destination IP address's sessions.
Rx	This field displays the amount of information received by the source in the active session.
Тх	This field displays the amount of information transmitted by the source in the active session.
Duration	This field displays the length of the active session in seconds.

6.6 The DDNS Status Screen

The DDNS Status screen shows the status of the ZyWALL's DDNS domain names. Click **Monitor >** System Status > DDNS Status to open the following screen.

Figure 63 Monitor > System Status > DDNS Status

atus	5					
ONS	Status					
1	Undate					
#	Profile Name	Domain name	Effective IP	Last Update St	Last Update Time	
			change and	1,	No data t	I. I.

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Update	Click this to have the ZyWALL update the profile to the DDNS server. The ZyWALL attempts to resolve the IP address for the domain name.
Profile Name	This field displays the descriptive profile name for this entry.
Domain Name	This field displays each domain name the ZyWALL can route.
Effective IP	This is the (resolved) IP address of the domain name.
Last Update Status	This shows whether the last attempt to resolve the IP address for the domain name was successful or not. Updating means the ZyWALL is currently attempting to resolve the IP address for the domain name.
Last Update Time	This shows when the last attempt to resolve the IP address for the domain name occurred (in year-month-day hour:minute:second format).

Table 27 Monitor > System Status > DDNS Status

6.7 IP/MAC Binding Monitor

Click **Monitor > System Status > IP/MAC Binding** to open the **IP/MAC Binding Monitor** screen. This screen lists the devices that have received an IP address from ZyWALL interfaces with IP/MAC binding enabled and have ever established a session with the ZyWALL. Devices that have never established a session with the ZyWALL do not display in the list.

Figure 64 Monitor > System Status > IP/MAC Binding

nterface:	none		•		
# IP A	ddress	Host Na	ame	MAC Address	Last Access
14 4 14	Page 1 of 1 🕨	ÞI I	50 🔻 items		No data to disp

LABEL	DESCRIPTION
Interface	Select a ZyWALL interface that has IP/MAC binding enabled to show to which devices it has assigned an IP address.
#	This is the index number of an IP/MAC binding entry.
IP Address	This is the IP address that the ZyWALL assigned to a device.
Host Name	This field displays the name used to identify this device on the network (the computer name). The ZyWALL learns these from the DHCP client requests.
MAC Address	This field displays the MAC address to which the IP address is currently assigned.
Last Access	This is when the device last established a session with the ZyWALL through this interface.
Refresh	Click this button to update the information in the screen.

The following table describes the labels in this screen. **Table 28** Monitor > System Status > IP/MAC Binding

6.8 The Login Users Screen

Use this screen to look at a list of the users currently logged into the ZyWALL. To access this screen, click **Monitor > System Status > Login Users**.

Figure 65 Monitor > System Status > Login Users

S F	orce Logout				
#	User ID	Reauth Lease T.	Туре	IP Address	User Info
0	a1	23:58:48 / 23:59:06	http/https	192.168.191.35	limited-admin
1	a2	23:59:21/23:59:21	http/https	192.168.191.35	user
2	a3	23:59:33/23:59:33	http/https	192.168.191.35	guest
3	admin	unlimited / 00:20:29	http/https	192.168.191.33	admin
4	admin	unlimited / 00:28:22	console	console	admin
5	admin	unlimited / 00:29:59	http/https	192.168.191.133	admin
6	test1	23:58:15 / 23:58:15	http/https	192.168.191.35	ext-user.ext-aroup-user(testa).ext-aroup-user(te

LABEL	DESCRIPTION
#	This field is a sequential value and is not associated with any entry.
User ID	This field displays the user name of each user who is currently logged in to the ZyWALL.
Reauth Lease T.	This field displays the amount of reauthentication time remaining and the amount of lease time remaining for each user. See Chapter 27 on page 361.
Туре	This field displays the way the user logged in to the ZyWALL.
IP Address	This field displays the IP address of the computer used to log in to the ZyWALL.

 Table 29
 Monitor > System Status > Login Users

LABEL	DESCRIPTION
User Info	This field displays the types of user accounts the ZyWALL uses. If the user type is ext-user (external user), this field will show its external-group information when you move your mouse over it.
	If the external user matches two external-group objects, both external-group object names will be shown.
Force Logout	Select a user ID and click this icon to end a user's session.
Refresh	Click this button to update the information in the screen.

 Table 29
 Monitor > System Status > Login Users (continued)

6.9 Cellular Status Screen

This screen displays your 3G connection status. Click **Monitor > System Status > Cellular Status** to display this screen.

Figure 66 Monitor > System Status > Cellular Status

ulu	r Device Statu	s				
Re	efresh					
-						
9	More Information	(
# 🔺	Extension Slot	Connected Device	Status	Service Provider	Cellular System	Signal Quality
_	LISB 1	Huawei E220	Device ready	Chunghwa Telecom	WCDMA	Excellent
1	0301					

LABEL	DESCRIPTION
Refresh	Click this button to update the information in the screen.
More Information	Click this to display more information on your 3G, such as the signal strength, IMEA/ESN and IMSI. This is only available when the 3G device attached and activated on your ZyWALL. Refer to Section 6.9.1 on page 95.
#	This field is a sequential value, and it is not associated with any interface.
Extension Slot	This field displays where the entry's cellular card is located.
Connected Device	This field displays the model name of the cellular card.

 Table 30
 Monitor > System Status > Cellular Status

LABEL	DESCRIPTION
Status	No device - no 3G device is connected to the ZyWALL. No Service - no 3G network is available in the area; you cannot connect to the Internet. Limited Service - returned by the service provider in cases where the SIM card is expired, the user failed to pay for the service and so on; you cannot connect to the Internet. Device detected - displays when you connect a 3G device. Device detected - displays when you connect a 3G device. Device detected - displays when you connect a 3G device. Device detected - the ZyWALL's test of the 3G device failed. Probe device fail - the ZyWALL's test of the 3G device succeeded. Init device fail - the ZyWALL's test of the 3G device succeeded. Init device fail - the ZyWALL's check of whether or not the 3G device. Init device fail - the ZyWALL's check of whether or not the 3G device is locked failed. Device locked - the 3G device is locked. SIM ocked-PUK - the PUK is locked on the 3G device's SIM card. SIM locked-PUN - the PIN is locked on the 3G device's SIM card. SIM locked-PUN - the PIN is locked on the 3G device's SIM card. Unlock PUK fail - Your attempt to unlock a WCDMA 3G device's PUK failed because you entered an incorrect PUK. Unlock PIN fail - Your attempt to unlock a CDMA2000 3G device failed because you entered an incorrect PUK. Unlock device fail - Your attempt to unlock a CDMA2000 3G device failed because you entered an incorrect device code. Device unlocked - You entered the correct device code and unlocked a CDMA2000 3G device. Get dev-info fail - The ZyWALL succeeded in retrieving 3G device information. Get adev-info fail - The 3G device failed in network. Apply config - The ZyWALL is applying your configuration to the 3G device. Inactive - The 3G interface is disabled. Active - The 3G interface is disabled. Active - The 3G interface is disabled. Correct device - The connected 3G device is not compatible with the ZyWALL. Correct device - The connected 3G device is not successful. Set profile fail - Applying your ISP settings. PP fail - The ZyWALL successfully applie
Service Provider	can use the 3G connection. This displays the name of your network service provider. This shows Limited Service if the service provider has stopped service to the 3G SIM card. For
Cellular System	example if the bill has not been paid or the account has expired. This field displays what type of cellular network the 3G connection is using. The network type varies depending on the 3G card you inserted and could be UMTS , UMTS/HSDPA , GPRS or EDGE when you insert a GSM 3G card, or 1xRTT , EVDO Rev.0 or EVDO Rev.A when you insert a CDMA 3G card.
Signal Quality	This displays the strength of the signal. The signal strength mainly depends on the antenna output power and the distance between your ZyWALL and the service provider's base station.

 Table 30
 Monitor > System Status > Cellular Status (continued)

6.9.1 More Information

This screen displays more information on your 3G, such as the signal strength, IMEA/ESN and IMSI that helps identify your 3G device and SIM card. Click **Monitor > System Status > More Information** to display this screen.

Note: This screen is only available when the 3G device is attached to and activated on the ZyWALL.



Figure 67 Monitor > System Status > More Information

LABEL	DESCRIPTION
Extension Slot	This field displays where the entry's cellular card is located.
Service Provider	This displays the name of your network service provider. This shows Limited Service if the service provider has stopped service to the 3G SIM card. For example if the bill has not been paid or the account has expired.
Cellular System	This field displays what type of cellular network the 3G connection is using. The network type varies depending on the 3G card you inserted and could be UMTS, UMTS/HSDPA, GPRS or EDGE when you insert a GSM 3G card, or 1xRTT, EVDO Rev.0 or EVDO Rev.A when you insert a CDMA 3G card.
Signal Strength	This is the Signal Quality measured in dBm.
Signal Quality	This displays the strength of the signal. The signal strength mainly depends on the antenna output power and the distance between your ZyWALL and the service provider's base station.
Device Manufacturer	This shows the name of the company that produced the 3G device.
Device Model	This field displays the model name of the cellular card.
Device Firmware	This shows the software version of the 3G device.

 Table 31
 Monitor > System Status > More Information

LABEL	DESCRIPTION
Device IMEI/ESN	IMEI (International Mobile Equipment Identity) is a 15-digit code in decimal format that identifies the 3G device.
	ESN (Electronic Serial Number) is an 8-digit code in hexadecimal format that identifies the 3G device.
SIM Card IMSI	IMSI (International Mobile Subscriber Identity) is a 15-digit code that identifies the SIM card.

 Table 31
 Monitor > System Status > More Information (continued)

6.10 USB Storage Screen

This screen displays information about a connected USB storage device. Click **Monitor > System Status > USB Storage** to display this screen.

Figure 68 Monitor > System Status > USB Storage

Storage Information			
Information			
Device description:	N/A		
Usage:	N/A		
Filesystem:	N/A		
Speed:	N/A		
Status:	none		Use It
Detail:	none		

LABEL	DESCRIPTION
Device description	This is a basic description of the type of USB device.
Usage	This field displays how much of the USB storage device's capacity is currently being used out of its total capacity and what percentage that makes.
Filesystem	This field displays what file system the USB storage device is formatted with. This field displays Unknown if the file system of the USB storage device is not supported by the ZyWALL, such as NTFS.
Speed	This field displays the connection speed the USB storage device supports.

 Table 32
 Monitor > System Status > USB Storage

LABEL	DESCRIPTION
Status	Ready - you can have the ZyWALL use the USB storage device.
	Click Remove Now to stop the ZyWALL from using the USB storage device so you can remove it.
	Unused - the connected USB storage device was manually unmounted by using the Remove Now button or for some reason the ZyWALL cannot mount it.
	Click Use It to have the ZyWALL mount a connected USB storage device. This button is grayed out if the file system is not supported (unknown) by the ZyWALL.
	none - no USB storage device is connected.
Detail	This field displays any other information the ZyWALL retrieves from the USB storage device.
	Deactivated - the use of a USB storage device is disabled (turned off) on the ZyWALL.
	OutofSpace - the available disk space is less than the disk space full threshold (see Section 37.2 on page 433 for how to configure this threshold).
	Mounting - the ZyWALL is mounting the USB storage device.
	Removing - the ZyWALL is unmounting the USB storage device.
	none - the USB device is operating normally or not connected.

 Table 32
 Monitor > System Status > USB Storage (continued)

6.11 The IPSec Monitor Screen

You can use the **IPSec Monitor** screen to display and to manage active IPSec To access this screen, click **Monitor** > **VPN Monitor** > **IPSec**. The following screen appears. SAs. Click a column's heading cell to sort the table entries by that column's criteria. Click the heading cell again to reverse the sort order.

lame	e:							
olicy	/: T				Search			
65	Disconnect							
#	Name 🔺	Encapsulation	Policy	Algorithm	Up Time	Timeout	Inbound(Bytes)	Outbound(Bytes)
	Dame 1	of the Mill	Show 50	items				No data to disr

Figure 69 Monitor > VPN Monitor > IPSec

Each field is described in the following table.

LABEL	DESCRIPTION
Name	Enter the name of a IPSec SA here and click Search to find it (if it is associated). You can use a keyword or regular expression. Use up to 30 alphanumeric and $_+()!$ *^:? {}[]<>/ characters. See Section 6.11.1 on page 98 for more details.
Policy	Enter the IP address(es) or names of the local and remote policies for an IPSec SA and click Search to find it. You can use a keyword or regular expression. Use up to 30 alphanumeric and $_+()!$ *^:? {}[]<>/ characters. See Section 6.11.1 on page 98 for more details.
Search	Click this button to search for an IPSec SA that matches the information you specified above.
Disconnect	Select an IPSec SA and click this button to disconnect it.
Total Connection	This field displays the total number of associated IPSec SAs.
connection per page	Select how many entries you want to display on each page.
Page x of x	This is the number of the page of entries currently displayed and the total number of pages of entries. Type a page number to go to or use the arrows to navigate the pages of entries.
#	This field is a sequential value, and it is not associated with a specific SA.
Name	This field displays the name of the IPSec SA.
Encapsulation	This field displays how the IPSec SA is encapsulated.
Policy	This field displays the content of the local and remote policies for this IPSec SA. The IP addresses, not the address objects, are displayed.
Algorithm	This field displays the encryption and authentication algorithms used in the SA.
Up Time	This field displays how many seconds the IPSec SA has been active. This field displays N/A if the IPSec SA uses manual keys.
Timeout	This field displays how many seconds remain in the SA life time, before the ZyWALL automatically disconnects the IPSec SA. This field displays N/A if the IPSec SA uses manual keys.
Inbound (Bytes)	This field displays the amount of traffic that has gone through the IPSec SA from the remote IPSec router to the ZyWALL since the IPSec SA was established.
Outbound (Bytes)	This field displays the amount of traffic that has gone through the IPSec SA from the ZyWALL to the remote IPSec router since the IPSec SA was established.
Refresh	Click Refresh to update the information in the display.

 Table 33
 Monitor > VPN Monitor > IPSec

6.11.1 Regular Expressions in Searching IPSec SAs

A question mark (?) lets a single character in the VPN connection or policy name vary. For example, use "a?c" (without the quotation marks) to specify abc, acc and so on.

Wildcards (*) let multiple VPN connection or policy names match the pattern. For example, use "*abc" (without the quotation marks) to specify any VPN connection or policy name that ends with "abc". A VPN connection named "testabc" would match. There could be any number (of any type) of characters in front of the "abc" at the end and the VPN connection or policy name would still match. A VPN connection or policy name named "testacc" for example would not match.

A * in the middle of a VPN connection or policy name has the ZyWALL check the beginning and end and ignore the middle. For example, with "abc*123", any VPN connection or policy name starting with "abc" and ending in "123" matches, no matter how many characters are in between.

The whole VPN connection or policy name has to match if you do not use a question mark or asterisk.

6.12 The SSL Connection Monitor Screen

The ZyWALL keeps track of the users who are currently logged into the VPN SSL client Click **Monitor > VPN Monitor > SSL** to display the user list.

portal. Use this screen to do the following:

- View a list of active SSL VPN connections.
- Log out individual users and delete related session information.

Once a user logs out, the corresponding entry is removed from the Connection Monitor screen.

Figure 70 Monitor > VPN Monitor > SSL

-						
ren	Disconnect	.nection				
#	User 🔺	Access	Login Address	Connected Time	Inbound (Bytes)	Outbound (Bytes)
14	A Page 1	of 1 🕨 🕅 Show	w 50 🔻 items		1. 31432 333	No data to display

The following table describes the labels in this screen.

Table 34 Monitor > VPN Monitor > SSL

LABEL	DESCRIPTION
Disconnect	Select a connection and click this button to terminate the user's connection and delete corresponding session information from the ZyWALL.
#	This field displays the index number.
User	This field displays the account user name used to establish this SSL VPN connection.
Access	This field displays the name of the SSL VPN application the user is accessing.
Login Address	This field displays the IP address the user used to establish this SSL VPN connection.
Connected Time	This field displays the time this connection was established.
Inbound (Bytes)	This field displays the number of bytes received by the ZyWALL on this connection.
Outbound (Bytes)	This field displays the number of bytes transmitted by the ZyWALL on this connection.
Refresh	Click Refresh to update this screen.

6.13 The L2TP over IPSec Session Monitor Screen

Click **Monitor** > **VPN Monitor** > **L2TP over IPSec** to open the following screen. Use this screen to display and manage the ZyWALL's connected L2TP VPN sessions.

Figure 71 Monitor > VPN Monitor > L2TP over IPSec

rent L2TP Session				
Disconnect Refresh				
# A User Name	Hostname	Assigned IP	Public IP	

The following table describes the fields in this screen.

LABEL	DESCRIPTION
Disconnect	Select a connection and click this button to disconnect it.
#	This is the index number of a current L2TP VPN session.
User Name	This field displays the remote user's user name.
Hostname	This field displays the name of the computer that has this L2TP VPN connection with the ZyWALL.
Assigned IP	This field displays the IP address that the ZyWALL assigned for the remote user's computer to use within the L2TP VPN tunnel.
Public IP	This field displays the public IP address that the remote user is using to connect to the Internet.
Refresh	Click Refresh to update this screen.

 Table 35
 Monitor > VPN Monitor > L2TP over IPSec

6.14 Log Screen

Log messages are stored in two separate logs, one for regular log messages and one for debugging messages. In the regular log, you can look at all the log messages by selecting **All Logs**, or you can select a specific category of log messages (for example, firewall or user). You can also look at the debugging log by selecting **Debug Log**. All debugging messages have the same priority.

To access this screen, click **Monitor > Log**. The log is displayed in the following screen.

- Note: When a log reaches the maximum number of log messages, new log messages automatically overwrite existing log messages, starting with the oldest existing log message first.
- The maximum possible number of log messages in the ZyWALL varies by model.

Events that generate an alert (as well as a log message) display in red. Regular logs display in black. Click a column's heading cell to sort the table entries by that column's criteria. Click the heading cell again to reverse the sort order.

Figure 72 Monitor > Log

ew L	og			and the second			
Show	Filter						
ogs							
Displa	iy:	All Lo	gs	~			
E	mail Log Now 🍣 Ref	resh	of Cl	ear Log			
#	Time	Pri	Cat	Message	Source	Destination	Note
1	2013-01-11 06:11:51	no	User	Administrator admin from http/https has logged in ZyWALL	172.23.30.25	172.23.30.26	Account: admi
2	2013-01-11 06:09:14	no	User	Administrator admin from http/https has logged in ZyWALL	172.23.30.34	172.23.30.26	Account: admi
3	2013-01-11 03:23:22	по	User	Administrator admin from http/https has been logged out	172.23.30.34	172.23.30.26	Account: admi
4	2013-01-11 03:21:51	no	User	Administrator admin from http/https has been logged out	172.23.30.25	172.23.30.26	Account: admi
5	2013-01-11 03:20:30	info	Sy	NTP update has failed with server 1.pool.ntp.org[ret:5]			System
6	2013-01-11 02:34:10	no	User	Administrator admin from http/https has logged in ZyWALL	172.23.30.25	172.23.30.26	Account: admi
7	2013-01-11 02:29:39	no	User	Administrator admin from http/https has logged in ZyWALL	172.23.30.34	172.23.30.26	Account: admi
8	2013-01-10 10:28:25	no	User	Administrator admin from http/https has been logged out	172.23.30.34	172.23.30.26	Account: admi
9	2013-01-10 09:46:37	no	User	Administrator admin from http/https has logged in ZyWALL	172.23.30.34	172.23.30.26	Account: admi
10	2013-01-10 03:20:28	info	Sy	NTP update has failed with server 1.pool.ntp.org[ret:5]			System

Table 36 Monitor > Log

	DESCRIPTION
LADEL	DESCRIPTION
Show Filter / Hide Filter	Click this button to show or hide the filter settings.
	If the filter settings are hidden, the Display , Email Log Now , Refresh , and Clear Log fields are available.
	If the filter settings are shown, the Display , Priority , Source Address , Destination Address , Service , Keyword , and Search fields are available.
Display	Select the category of log message(s) you want to view. You can also view All Logs at one time, or you can view the Debug Log .
Priority	This displays when you show the filter. Select the priority of log messages to display. The log displays the log messages with this priority or higher. Choices are: any , emerg , alert , crit , error , warn , notice , and info , from highest priority to lowest priority. This field is read-only if the Category is Debug Log .
Source Address	This displays when you show the filter. Type the source IP address of the incoming packet that generated the log message. Do not include the port in this filter.
Destination Address	This displays when you show the filter. Type the IP address of the destination of the incoming packet when the log message was generated. Do not include the port in this filter.
Source Interface	This displays when you show the filter. Select the source interface of the packet that generated the log message.
Destination Interface	This displays when you show the filter. Select the destination interface of the packet that generated the log message.
Service	This displays when you show the filter. Select the service whose log messages you would like to see. The Web Configurator uses the protocol and destination port number(s) of the service to select which log messages you see.
Keyword	This displays when you show the filter. Type a keyword to look for in the Message , Source , Destination and Note fields. If a match is found in any field, the log message is displayed. You can use up to 63 alphanumeric characters and the underscore, as well as punctuation marks ()', :;?! +-*/= #\$% @ ; the period, double quotes, and brackets are not allowed.
Protocol	This displays when you show the filter. Select a service protocol whose log messages you would like to see.
Search	This displays when you show the filter. Click this button to update the log using the current filter settings.

LABEL	DESCRIPTION
Email Log Now	Click this button to send log message(s) to the Active e-mail address(es) specified in the Send Log To field on the Log Settings page (see Section 38.3.2 on page 478).
Clear Log	Click this button to clear the whole log, regardless of what is currently displayed on the screen.
#	This field is a sequential value, and it is not associated with a specific log message.
Time	This field displays the time the log message was recorded.
Priority	This field displays the priority of the log message. It has the same range of values as the Priority field above.
Category	This field displays the log that generated the log message. It is the same value used in the Display and (other) Category fields.
Message	This field displays the reason the log message was generated. The text "[count= x]", where x is a number, appears at the end of the Message field if log consolidation is turned on (see Log Consolidation in Table 196 on page 479). and multiple entries were aggregated to generate into this one.
Source	This field displays the source IP address and the port number in the event that generated the log message.
Destination	This field displays the destination IP address and the port number of the event that generated the log message.
Note	This field displays any additional information about the log message.

Table 36 Monitor > Log (continued)

The Web Configurator saves the filter settings if you leave the **View Log** screen and return to it later.

Interfaces

7.1 Interface Overview

Use the **Interface** screens to configure the ZyWALL's interfaces. You can also create interfaces on top of other interfaces.

- Ports are the physical ports to which you connect cables.
- **Interfaces** are used within the system operationally. You use them in configuring various features. An interface also describes a network that is directly connected to the ZyWALL. For example, You connect the LAN network to the LAN interface.
- Zones are groups of interfaces used to ease security policy configuration.

7.1.1 What You Can Do in this Chapter

- Use the **Port Role** screen (Section 7.2 on page 108) to create port groups and to assign physical ports and port groups to Ethernet interfaces.
- Use the **Ethernet** screens (Section 7.3 on page 109) to configure the Ethernet interfaces. Ethernet interfaces are the foundation for defining other interfaces and network policies. RIP and OSPF are also configured in these interfaces.
- Use the PPP screens (Section 7.4 on page 125) for PPPoE or PPTP Internet connections.
- Use the **Cellular** screens (Section 7.5 on page 132) to configure settings for interfaces for Internet connections through an installed 3G card.
- Use the **Tunnel** screens (Section 7.6 on page 140) to configure tunnel interfaces to be used in Generic Routing Encapsulation (GRE), IPv6 in IPv4, and 6to4 tunnels.
- Use the VLAN screens (Section 7.7 on page 147) to divide the physical network into multiple logical networks. VLAN interfaces receive and send tagged frames. The ZyWALL automatically adds or removes the tags as needed. Each VLAN can only be associated with one Ethernet interface.
- Use the **Bridge** screens (Section 7.8 on page 159) to combine two or more network segments into a single network.
- Use the **Virtual Interface** screen (Section 7.9.1 on page 171) to create virtual interfaces on top of Ethernet interfaces to tell the ZyWALL where to route packets. You can create virtual Ethernet interfaces, virtual VLAN interfaces, and virtual bridge interfaces.
- Use the Trunk screens (Chapter 8 on page 176) to configure load balancing.

7.1.2 What You Need to Know

Interface Characteristics

Interfaces generally have the following characteristics (although not all characteristics apply to each type of interface).

- An interface is a logical entity through which (layer-3) packets pass.
- An interface is bound to a physical port or another interface.
- Many interfaces can share the same physical port.
- An interface belongs to at most one zone.
- Many interfaces can belong to the same zone.
- Layer-3 virtualization (IP alias, for example) is a kind of interface.

Types of Interfaces

You can create several types of interfaces in the ZyWALL.

- Setting interfaces to the same port role forms a port group. Port groups create a hardware connection between physical ports at the layer-2 (data link, MAC address) level. Port groups are created when you use the Interface > Port Roles screen to set multiple physical ports to be part of the same interface.
- **Ethernet interfaces** are the foundation for defining other interfaces and network policies. RIP and OSPF are also configured in these interfaces.
- **Tunnel interfaces** send IPv4 or IPv6 packets from one network to a specific network through the Internet or a public network.
- VLAN interfaces receive and send tagged frames. The ZyWALL automatically adds or removes the tags as needed. Each VLAN can only be associated with one Ethernet interface.
- Bridge interfaces create a software connection between Ethernet or VLAN interfaces at the layer-2 (data link, MAC address) level. Unlike port groups, bridge interfaces can take advantage of some security features in the ZyWALL. You can also assign an IP address and subnet mask to the bridge.
- **PPP interfaces** support Point-to-Point Protocols (PPP). ISP accounts are required for PPPoE/PPTP interfaces.
- Cellular interfaces are for 3G WAN connections via a connected 3G device.
- Virtual interfaces provide additional routing information in the ZyWALL. There are three types: virtual Ethernet interfaces, virtual VLAN interfaces, and virtual bridge interfaces.
- Trunk interfaces manage load balancing between interfaces.

Port groups and trunks have a lot of characteristics that are specific to each type of interface. The other types of interfaces--Ethernet, PPP, cellular, VLAN, bridge, and virtual--have a lot of similar characteristics. These characteristics are listed in the following table and discussed in more detail below.

CHARACTERISTICS	ETHERNET	ETHERNET	PPP	PP CELLULAR		BRIDGE	VIRTUAL
Name*	wan1	lan1, lan2, dmz	ppp <i>x</i>	cellularx	vlan <i>x</i>	br <i>x</i>	* *
Configurable Zone	No	No	Yes	Yes Yes		Yes	No
IP Address Assignment							
Static IP address	Yes	Yes	Yes	Yes	Yes	Yes	Yes
DHCP client	Yes	No	Yes	Yes	Yes	Yes	No
Routing metric	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Interface Parameters							
Bandwidth restrictions	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Table 37 Ethernet, PPP, Cellular, VLAN, Bridge, and Virtual Interface Characteristics

CHARACTERISTICS	ETHERNET	ETHERNET	PPP	CELLULAR	VLAN	BRIDGE	VIRTUAL
Packet size (MTU)	Yes	Yes	Yes	Yes	Yes	Yes	No
DHCP							
DHCP server	No	Yes	No	No	Yes	Yes	No
DHCP relay	No	Yes	No	No	Yes	Yes	No
Connectivity Check	Yes	No	Yes	Yes	Yes	Yes	No

 Table 37
 Ethernet, PPP, Cellular, VLAN, Bridge, and Virtual Interface Characteristics (continued)

- * The format of interface names other than the Ethernet and ppp interface names is strict. Each name consists of 2-4 letters (interface type), followed by a number (*x*). For most interfaces, x is limited by the maximum number of the type of interface. For VLAN interfaces, x is defined by the number you enter in the VLAN name field. For example, Ethernet interface names are wan1, wan2, lan1, lan2, dmz; VLAN interfaces are vlan0, vlan1, vlan2, ...; and so on.

** - The names of virtual interfaces are derived from the interfaces on which they are created. For example, virtual interfaces created on Ethernet interface wan1 are called wan1:1, wan1:2, and so on. Virtual interfaces created on VLAN interface vlan2 are called vlan2:1, vlan2:2, and so on. You cannot specify the number after the colon(:) in the Web Configurator; it is a sequential number. You can specify the number after the colon if you use the CLI to set up a virtual interface.

Relationships Between Interfaces

In the ZyWALL, interfaces are usually created on top of other interfaces. Only Ethernet interfaces are created directly on top of the physical ports or port groups. The relationships between interfaces are explained in the following table.

INTERFACE	REQUIRED PORT / INTERFACE
port group	physical port
Ethernet interface	physical port
	port group
VLAN interface	Ethernet interface
bridge interface	Ethernet interface*
	VLAN interface*
PPP interface	Ethernet interface*
	VLAN interface*
	bridge interface
	WAN1, WAN2, OPT*
virtual interface	
(virtual Ethernet interface)	Ethernet interface*
(virtual VLAN interface)	VLAN interface*
(virtual bridge interface)	bridge interface
trunk	Ethernet interface
	Cellular interface
	VLAN interface
	bridge interface
	PPP interface

 Table 38
 Relationships Between Different Types of Interfaces

 You cannot set up a PPP interface, virtual Ethernet interface or virtual VLAN interface if the underlying interface is a member of a bridge. You also cannot add an Ethernet interface or VLAN interface to a bridge if the member interface has a virtual interface or PPP interface on top of it.

IPv6 Overview

IPv6 (Internet Protocol version 6), is designed to enhance IP address size and features. The increase in IPv6 address size to 128 bits (from the 32-bit IPv4 address) allows up to 3.4×10^{38} IP addresses.

IPv6 Addressing

The 128-bit IPv6 address is written as eight 16-bit hexadecimal blocks separated by colons (:). This is an example IPv6 address 2001:0db8:1a2b:0015:0000:1a2f:0000.

IPv6 addresses can be abbreviated in two ways:

- Leading zeros in a block can be omitted. So 2001:0db8:1a2b:0015:0000:0000:1a2f:0000 can be written as 2001:db8:1a2b:15:0:0:1a2f:0.
- Any number of consecutive blocks of zeros can be replaced by a double colon. A double colon can only appear once in an IPv6 address. So 2001:0db8:0000:0000:1a2f:0000:0000:0015 can be written as 2001:0db8::1a2f:0000:0000:0015, 2001:0db8:0000:0000:1a2f::0015, 2001:db8::1a2f:0:0:15 or 2001:db8:0:0:1a2f::15.

Prefix and Prefix Length

Similar to an IPv4 subnet mask, IPv6 uses an address prefix to represent the network address. An IPv6 prefix length specifies how many most significant bits (start from the left) in the address compose the network address. The prefix length is written as "/x" where x is a number. For example,

```
2001:db8:1a2b:15::1a2f:0/32
```

means that the first 32 bits (2001:db8) from the left is the network prefix.

Link-local Address

A link-local address uniquely identifies a device on the local network (the LAN). It is similar to a "private IP address" in IPv4. You can have the same link-local address on multiple interfaces on a device. A link-local unicast address has a predefined prefix of fe80::/10. The link-local unicast address format is as follows.

Table 39 Link-local Unicast Address Forn	nat
--	-----

1111 1110 10	0	Interface ID
10 bits	54 bits	64 bits

Subnet Masking

Stateless Autoconfiguration

With stateless autoconfiguration in IPv6, addresses can be uniquely and automatically generated. Unlike DHCPv6 (Dynamic Host Configuration Protocol version six) which is used in IPv6 stateful autoconfiguration, the owner and status of addresses don't need to be maintained by a DHCP server. Every IPv6 device is able to generate its own and unique IP address automatically when IPv6 is initiated on its interface. It combines the prefix and the interface ID (generated from its own Ethernet MAC address) to form a complete IPv6 address.

When IPv6 is enabled on a device, its interface automatically generates a link-local address (beginning with fe80).

When the ZyWALL's WAN interface is connected to an ISP with a router and the ZyWALL is set to automatically obtain an IPv6 network prefix from the router for the interface, it generates ¹another address which combines its interface ID and global and subnet information advertised from the router. This is a routable global IP address.

Prefix Delegation

Prefix delegation enables an IPv6 router (the ZyWALL) to use the IPv6 prefix (network address) received from the ISP (or a connected uplink router) for its LAN. The ZyWALL uses the received IPv6 prefix (for example, 2001:db2::/48) to generate its LAN IP address. Through sending Router Advertisements (RAs) regularly by multicast, the router passes the IPv6 prefix information to its LAN hosts. The hosts then can use the prefix to generate their IPv6 addresses.

IPv6 Router Advertisement

An IPv6 router sends router advertisement messages periodically to advertise its presence and other parameters to the hosts in the same network.

DHCPv6

The Dynamic Host Configuration Protocol for IPv6 (DHCPv6, RFC 3315) is a server-client protocol that allows a DHCP server to assign and pass IPv6 network addresses, prefixes and other configuration information to DHCP clients. DHCPv6 servers and clients exchange DHCP messages using UDP.

Each DHCP client and server has a unique DHCP Unique IDentifier (DUID), which is used for identification when they are exchanging DHCPv6 messages. The DUID is generated from the MAC address, time, vendor assigned ID and/or the vendor's private enterprise number registered with the IANA. It should not change over time even after you reboot the device.

Finding Out More

- See Section 7.10 on page 172 for background information on interfaces.
- See Chapter 8 on page 176 to configure load balancing using trunks.

^{1.} In IPv6, all network interfaces can be associated with several addresses.

7.1.3 What You Need to Do First

For IPv6 settings, go to the **Configuration** > **System** > **IPv6** screen to enable IPv6 support on the ZyWALL first.

7.2 Port Role Screen

To access this screen, click **Configuration > Network > Interface > Port Role**. Use the **Port Role** screen to set the ZyWALL's flexible ports as part of the **Ian1**, **Ian2**, **ext-wlan** or **dmz** interfaces. This creates a hardware connection between the physical ports at the layer-2 (data link, MAC address) level. This provides wire-speed throughput but no security.

See Section 1.1 on page 18 to see which ZyWALL models support port role.

Note the following if you are configuring from a computer connected to a **lan1**, **lan2**, **ext-wlan** or **dmz** port and change the port's role:

- A port's IP address varies as its role changes, make sure your computer's IP address is in the same subnet as the ZyWALL's **lan1**, **lan2**, **ext-wlan** or **dmz** IP address.
- Use the appropriate lan1, lan2, ext-wlan or dmz IP address to access the ZyWALL.

Figure 73 Configuration > Network > Interface > Port Role (ZyWALL 110)

Port Role	Ethernet	PPP	Cellular	Tunnel	VLAN	Bridge	e Ti	runk		
Configuratio	n									
			-WAN	11WAN2-	-OPT-		-IAN/	DM7-		
			P1	P2	P3	P4	P5	P6	P7	Physical Ports
			1.111	<u>}</u>				11111	11111	T Hybroar T onto
			opt (OPT	.)	۲					
			lan1 (LAI	N1)	0	۲	0	0	0	
			lan2 (LAI	N2)	0	0	۲	0	0	Interfaces
			ext-wlan	(WLAN)	0	0	0	۲	0	
			dmz (DM	IZ)	0	0	0	0	۲	

The physical Ethernet ports are shown at the top and the Ethernet interfaces and zones are shown at the bottom of the screen. Use the radio buttons to select for which interface (network) you want to use each physical port. For example, select a port's LAN radio button to use the port as part of the LAN interface. The port will use the ZyWALL's LAN IP address and MAC address.

When you assign more than one physical port to a network, you create a **port group**. Port groups have the following characteristics:

- There is a layer-2 Ethernet switch between physical ports in the port group. This provides wirespeed throughput but no security.
- It can increase the bandwidth between the port group and other interfaces.
- The port group uses a single MAC address.
Click **Apply** to save your changes and apply them to the ZyWALL.

Click **Reset** to change the port groups to their current configuration (last-saved values).

7.3 Ethernet Summary Screen

This screen lists every Ethernet interface and virtual interface created on top of Ethernet interfaces. If you enabled IPv6 in the **Configuration** > **System** > **IPv6** screen, you can also configure Ethernet interfaces used for your IPv6 networks on this screen. To access this screen, click **Configuration** > **Network** > **Interface** > **Ethernet**.

Unlike other types of interfaces, you cannot create new Ethernet interfaces nor can you delete any of them. If an Ethernet interface does not have any physical ports assigned to it, the Ethernet interface is effectively removed from the ZyWALL, but you can still configure it.

Ethernet interfaces are similar to other types of interfaces in many ways. They have an IP address, subnet mask, and gateway used to make routing decisions. They restrict the amount of bandwidth and packet size. They can provide DHCP services, and they can verify the gateway is available.

Use Ethernet interfaces to control which physical ports exchange routing information with other routers and how much information is exchanged through each one. The more routing information is exchanged, the more efficient the routers should be. However, the routers also generate more network traffic, and some routing protocols require a significant amount of configuration and management. The ZyWALL supports two routing protocols, RIP and OSPF. See Chapter 10 on page 197 for background information about these routing protocols.

	can incine	The Gradie	G macuvace and cheate with			
#	Status	Name	IP Address		Mask	
1	Q	wan1	DHCP 0.0.0.0		0.0.0.0	
2	@	wan2	DHCP 0.0.0.0		0.0.0	
3	Q	lan1	STATIC		255.255.255.0	
4	@	lan2	STATIC 192.168.2.1		255.255.255.0	
5	@	dmz	STATIC 192.168.3.1		255.255.255.0	
V6 C	onfiguration	n				
₩6 C	Configuration Edit 💡 Activ Status	n ate <table-cell> Inactivat Name</table-cell>	e 🕞 Object Reference	ess		
v6 C #	Configuration Edit 💡 Activ Status	n ate 🎯 Inactivat Name wan1	e P Addr	ess		
w6 C # 1	Edit O Activ Status	n ate 💮 Inactivate Name wan1 wan2	e De Object Reference IP Addr :: ::	ess		
w6 C # 1 2 3	Configuration Edit Or Activ Status Or Or O	n ate @ Inactivat Name wan1 wan2 lan1	e Carl Object Reference IP Addr :: :: ::	6 38		
* 1 2 3 4	Configuration Edit O Activ Status O O O O O O O	n ate @ inactivati Name wan1 wan2 lan1 lan2	e PAddr PAddr : : : : : :	655		
v6 C # 1 2 3 4 5	Edit O Activ Status 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Name Name Wan1 Wan2 Ian1 Ian2 dmz	P P Addr P Addr : : : : : : : :	ėss		

Figure 74 Configuration > Network > Interface > Ethernet

Each field is described in the following table.

LABEL	DESCRIPTION
Configuration / IPv6 Configuration	Use the Configuration section for IPv4 network settings. Use the IPv6 Configuration section for IPv6 network settings if you connect your ZyWALL to an IPv6 network. Both sections have similar fields as described below.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove a virtual interface, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an interface, select it and click Activate.
Inactivate	To turn off an interface, select it and click Inactivate.
Create Virtual Interface	To open the screen where you can create a virtual Ethernet interface, select an Ethernet interface and click Create Virtual Interface .
Object References	Select an entry and click Object Reference to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with any interface.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Name	This field displays the name of the interface.
IP Address	This field displays the current IP address of the interface. If the IP address is 0.0.0.0 (in the IPv4 network) or :: (in the IPv6 network), the interface does not have an IP address yet.
	In the IPv4 network, this screen also shows whether the IP address is a static IP address (STATIC) or dynamically assigned (DHCP). IP addresses are always static in virtual interfaces.
	In the IPv6 network, this screen also shows whether the IP address is a static IP address (STATIC), link-local IP address (LINK LOCAL), dynamically assigned (DHCP), or an IPv6 StateLess Address AutoConfiguration IP address (SLAAC). See Section 7.1.2 on page 103 for more information about IPv6.
Mask	This field displays the interface's subnet mask in dot decimal notation.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 40
 Configuration > Network > Interface > Ethernet

7.3.1 Ethernet Edit

The **Ethernet Edit** screen lets you configure IP address assignment, interface parameters, RIP settings, OSPF settings, DHCP settings, connectivity check, and MAC address settings. To access this screen, click an **Edit** icon in the **Ethernet Summary** screen. (See Section 7.3 on page 109.)

Note: If you create IP address objects based on an interface's IP address, subnet, or gateway, the ZyWALL automatically updates every rule or setting that uses the object whenever the interface's IP address settings change. For example, if you change the LAN's IP address, the ZyWALL automatically updates the corresponding interface-based, LAN subnet address object.

With RIP, you can use Ethernet interfaces to do the following things.

• Enable and disable RIP in the underlying physical port or port group.

- Select which direction(s) routing information is exchanged The ZyWALL can receive routing information, send routing information, or do both.
- Select which version of RIP to support in each direction The ZyWALL supports RIP-1, RIP-2, and both versions.
- Select the broadcasting method used by RIP-2 packets The ZyWALL can use subnet broadcasting or multicasting.

With OSPF, you can use Ethernet interfaces to do the following things.

- Enable and disable OSPF in the underlying physical port or port group.
- Select the area to which the interface belongs.
- Override the default link cost and authentication method for the selected area.
- Select in which direction(s) routing information is exchanged The ZyWALL can receive routing information, send routing information, or do both.
- Set the priority used to identify the DR or BDR if one does not exist.

4/Drv6 View * Inter Advanced Settings	
eneral Settings	
🗑 Bruble Interface	
ieneral IPv6 Setting	
C Grade Pv4	
aterface Properties	
Interface Type:	external
(hiterface Name:	Linew Construction
Port:	P1
Zone:	was -
MAC Address:	02:001AA:79:73.79
Description:	(brotqC)
P Address Assignment	
C Get Autometically	
Use Fixed 3 ^a Address	
P ADDESC	
Subvet Mate:	235.255.255.0
	(Optona)
(THE BL	00-120
Pv6 Address Assignment	
E Brabie Stateless Address Auto-configur	aton (SLAAC)
Link-Local Address:	r,ta
Pv6 Address Prefix Length:	(Optional)
Gateway:	(Optional)
Metic	0-10
Address from DHOPu6 Prefix Delegation	OM B
	A Descent Data Suffa Address
	ADDESS ADDESS
	🕺 4 Page 1 of 1 3 🕅 Show 10 💌 as . He data to deplay
OF Bod Setting	
DKPA:	(Bert In)
0.01	Distance water at the th
C OTTO as Mart	And an and a second
Contraction of DUITS	
Competence (Competence)	
Enable Rapid Convert	
DHCPv6 Request Options	All P
	 Name + Type Value
	[1] 4 : Page [1] of 1 : 3 3] : Show 50 wins the data to depiny
Rovter Advertisement Setting Brake Router Advertisement Advertiset Router Advertisement Advertised Hosts Get Hetwork Configure Advertised Hosts Get Other Configure Advertised Hosts Get Other Configure	N 4 Page of 1:3 3 ippen 50 unique No data tar diquiny Bito Frain DHDM6 Infrain DHDM6
Pvi Rovier Advertisement Setting Deale Rovier Advertisement Advertised Nosis Get Herino Configura Anomised resus Get Other Configura Rovier Inchemos: NDL	M 4 Page 6 1 3 3 1
Pvis Router Advertisement Setting Details Route Advertisement Advertsed Nexts Get Network Configura Advertsed Nexts Get Other Configuras Router Indexnos: HUL: No List: Lesson Index Table	M 4] Page of 1 : 3: 3: 3] Space No data to durkey Month Prop. DVD+6 In Prov. DVD+6 No No No Meduar W Station Station Station SHED (1280-1500, 0 is disabled) Stationed) Stationed Stationed
Prof. Rowher Advertisement Setting Padja Rowter Advertisement Advertised insiss Cell Internol Comfort Advertised insiss Cell Patri Colf Comfortune Router Indexense HEL: The Lints: Advertised insiss Advertised insis	N 4 Pege 1 of 1: 3: 9 pege 10 units No Add basis
Prof. Rooter Advertisement Setting Deble Rooter Advertisement Advertiser Network Configuration Rooter Indexence: Mills: High Lint: Advertiser Network: Mills: Advertiser Network: Advertiser Network: Advertiser Network: Mills: Advertiser Network: Mills: Advertiser Network: Mills:	N 4 Page of 5 : 3: 3: 3: 3: 9: 9 page 10: 10: 10: 10: 10: 10: 10: 10: 10: 10:
Prof. Rooter Adventisement Secting Deally, Rooter Adventisement Adventised Healt Gententook Configures Rooter Andresson Set Offer Configures Rooter Andresson HTLL	N 4 Page of 1 : 0 N i page No No Adds to deplay Boo Prived DeCP-4 Mode Mode No No No No 1000 (1200 - 1500, 0 + 0 - 0 + 0 + 0 + 0 + 0 + 0 + 0 + 0 +
Prof. Rowher Advertisement Setting Padja Rowter Advertisement Advertise trais Cel Interior & Configura Rowter Indexense: Bolter Cell Configura Rowter Indexense: HEL: The Lint: Advertised Prefix Table	N 4 Page of 1 : 3 ≥ N Page W re No data to deploy astor Franch DChV6 to franch DChV6 Made W 1400 (120-1500, 14 databated) 64 D-55, 0 ± databated) 64 D-55, 0 ± databated) 64 D-55, 0 ± databated) 7 M databashmath Cargo = N 4 7 M databashmath Cargo = N 4 9 M databashmath Cargo = N 4
Prof. Rooter Advertisement Setting Deble Route Advertisement Advertise Network Centenbook Configure Router Indexence: Router Indexence: Mill: Router Indexence: Advertised Prefix Fision Advertised Prefix Fision DeCPuid Prefix Decembook Prefix Fision DeCPuid Prefix	
Not Roader Adventisement Setting Stadie Route Adventisement Adventised institu Gentensky Configuration Route Information Route Information Route Information Route Information Route Information Route Information Adventised Information Adventised Information Decision	Add Control of 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
Vet Booker Advertisement Setting Debik Rover Advertisement Advertised insis Ger Interiod Compute Advertised insis Ger Netrodo Compute Rover Indefenses Netro Inter Inter Advertised Predix Table Advertised Predix Table Advertised Predix Table	N 4 Page of 1 : 2 ≥ 2 Page No No Model Add Social State Social State Model Social State Social State
Pris Rovber Advertisement Setting Pris Rovber Advertisement Advertisement Advertise vision Genetication Configure Rouber Profeseroe Rouber Profeseroe Rouber Profeseroe Rouber Profeseroe Advertised Profes Table Advertised Profes Table	N 4 Page of 1 : 0 M (1990) W/m No data to depiny
Prof. Rooter Advertisement Setting	
Ve Booker Advertisement Setting Dedie Rover Advertisement Advertiser version (Configure) Rover Interview (Configure)	N 4 Page [of 1 :] N 9] Page 10 w/m No data to deplay Month DrUCh 4 In France DrUCh Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the data to deplay Month DrUCh 4 Internet of the
Pris Rovber Advertisement Setting Pris Rovber Advertisement Advertised insis Certainol & Configure Rouber Indexense Rouber Indexense Rouber Indexense Rouber Indexense Advertised Prefix Table Adv	In de la la la dela de
Vie Rocker Adventisement Setting Subje Rocker Adventisement Adventise Institution Configurab Rocker Institution Adventised Intellis Configurab Rocker Institution Adventised Intellis Table Adventised Intellis Table Adventised Intellis Table	
Prof. Rooter Advertisement Setting	Add Page of 1 : b (0) Page To data to deploy Add France Called and address Called and address Add So (25, 0) is dealed) Called address Called address Add So (25, 0) is dealed) Called address Called address Add So (25, 0) is dealed) Called address Called address Add So (25, 0) is dealed) Called address Called address Add So (25, 0) is dealed) Called address Called address Add So (25, 0) is dealed) Called address Called address Add So (25, 0) is dealed) Called address Called address Add So (25, 0) is dealed) Called address Called address Add So (25, 0) is dealed) Called address Called address Add Page of (2, 1) (20, 0) (20, 0) Called address Called address Add Page of (2, 1) (20, 0) Called address Called address Called address Called address Called address Called address Called address Called address Called address
Pris Rovber Advertisement Setting Pris Rovber Advertisement Advertised insis Certainsis Configure Advertised insis Certainsis Configure Rouber Indexense Rouber Indexense Rouber Indexense Rouber Index Table Advertised Prefix Table Advertised Pr	Image:
Prof. Rooter Advertisement Setting	Image:
	Image:
Pris Rosker Advertisement Settlag Pris Rosker Advertisement Advertisement Advertised insis Certainsis Configure Router Indexense Router Indexense Router Indexense Router Indexense Router Index Index Index Index Indexense Advertised Prafile Table Advertised Prafile Advertised	Image:
Prof. Rooter Advertisement Setting Debits Rooter Advertisement Advertised insis Certainold Configure Rooter Indexes Cent Certainold Configure Rooter Indexes Cent Certainold Advertised Insis Certainold Advertised Ins	Image:
	Image: Inf is in the first of the information of the infor
Vei Rovter Advertisement Settlag Code Rovter Advertisement Advertisement insis Gentenink Company Rovter Professor Gentenink Company Rovter Professor Gentenink Company Rovter Professor Gentenink Advertised Profes Table Advertised Profes Advertised Profes Advertised Profes Advertised	Image:
An Rocker Advertisement Settlag Code Route Advertisement Advertage transit Cettlando Configure Router Indexerses Cottine Cettlando Configure Router Indexerses Router Indexerses Advertage Index Table Ad	Add
Prof. Rocker Advertisement Setting Prof. Rocker Advertisement Advertised insis Certainsis Compare Rocker Indexense insis Certainsis Compare Rocker Indexense insis Compare Rocker Indexense Rocker Index Index Advertised Index Table Advertised Index Table Advertised Index Index Compare Express Sanchurdit: Program Program Sanchurdit: Progra	Image: Image
Net Roster Advertisement Settlag Debit Roster Advertisement Advertised tests Certeined Configure Roster Profession Configure Roster Profession Configure Roster Profession Configure Roster Profession Advertised Profes Table Betable Concentry Ordet Ordet Profes Ordet Profes Ordet Profes Ordet Profes Ded Default Catenary Ordet Profes Betable Advertise Betable Betable Betable Betable Betable Betable Betable Beta	Image:
	Image:
Pris Rocker Advertisement Settlag Pris Rocker Advertisement Advertises insis Certainsis Configure Rocker Indexenses Rocker Indexenses Rocker Indexenses Rocker Indexenses Rocker Index Index Index Index Index Index Advertised Prisfis Table Advertised Prisfis Advertised Table Advertised Advertised Table Advertised Ad	Image:
Pris Rooter Advertisement Settlag Pris Rooter Advertisement Advertage vasion Configure Router Indexes vasion Configure Router Indexes vasion Configure Router Indexes Advertage vasion Configure Router Advertage vasion Advertage	Image:
	Image: Inf is in the properties of
Pris Rocker Advertisement Settlag Pris Rocker Advertisement Advertisement inits Certainsk Compan Rocker Indexense initis Certainsk Compan Rocker Indexense initia Certainsk Compan Rocker Indexense initia Certainsk Compan Rocker Indexense Rocker Index Index Company Advertised Practic Table Advertised Table Advertise	Image: Instance of the state of the sta
Pris Rooter Advertisement Settlag Pris Rooter Advertisement Advertage vasion Configuration Rooter Indexes vasion Configuration Rooter Indexes vasion Configuration Rooter Indexes vasion Configuration Rooter Indexes vasion Advertaged Index Table	Wind Prope of 5 : b if prove if wind The data to depairs Anson France DCP-04 Image: Compare to the data to depairs Image: Compare to the data to depairs Image: Compare to the data to depairs Image: Compare to the data to depairs Image: Compare to the data to depairs Image: Compare to the data to th
Profile Roder Advertisement Setting	Image: Inf is in the first of the inf
Pris Rocker Advertisement Settlag Pris Rocker Advertisement Advertisement insis Certeinen's Configure Rocker Indexense insis Certeinen's Configure Rocker Indexense Certeinen's Configure Rocker Indexense Certeinen's Configure Rocker Indexense Certeinen's Configure Advertised Prefile Tale Certeinen's Advertised Prefile Tale DecDrod Index Prefile Contact Transmitter Advertised Prefile Tale DecDrod Index Prefile Contact Transmitter Prefile Tale Profile Prefile Tale Profile Prefile Contact Transmitter	Image:
	Image: Inf i i i i i i i i i i i i i i i i i i i
Pris Rocker Advertisement Settlag Pris Rocker Advertisement Advertises insis Certeines & Configure Rocker Indexense insis Certeines & Configure Rocker Indexense insis Certeines & Configure Rocker Indexense insis Configure Rocker Index Index Configure Rocker Index Index Configure Advertised Prisfs Table Advertised Prisfs Advertised Prisfs Advertise Index Prisfs	Image: Info: Dechoic Info: Dechoic Info: Dechoic Image: Info:
Pris Rocker Advertisement Settlag Pris Rocker Advertisement Advertised resis Certensols Configure Rocker Indexes resis Certensols Configure Rocker Indexes resis Certensols Advertised Rocker Configure Rocker Indexes Advertised Rocker Configure Rocker Indexes Advertised Rocker Configure Rocker Indexes Advertised Rocker Configure Rocker Rocker Advertised Rocker Rock	Will Page of \$1 * 2 * 3 * 9 * 9000 Wint No data to deploy Asso France DCP-6 In france DCP-6 Will (100-5000, do stateded) Stateded
Profile Roder Advertisement Setting	Image: Image
Pris Rocker Advertisement Settlag Pris Rocker Advertisement Advertises insis Certeines & Configure Rocker Indexers insis Certeines & Configure Rocker Indexers Rocker Indexers Rocker Index Index Configure Rocker Index Index Configure Rocker Index Index Configure Rocker Index Index Configure Advertised Prisfs Table Advertised Prisfs Advertised Prisfs Advertised Brisfs Convecting Orde Orded Treace Condo Treace Brisfs Table Table Brisfs Table Table Brisfs Table Table Brisfs Advertised Prisfs	Image: Info: Image:
Pris Rocker Advertisement Settlag Pris Rocker Advertisement Adverse insis Certenion & Configure Rocker Indexes insis Certenion & Configure Advertised Prefix Fain DeCPod Prefix Rocker Indexes Advertised Prefix Fain DeCPod Prefix Brocket Rocker Index Index Prefix Concentry Ordel Order Prefix Configure Brocket Decement Concentry Conce Dece Defail Certain Conce Fail Telemon: Conce Fail Telemon: Concentry Brocket Brocket Brocket Brocket Brocket Concentry Brocket Brocket Concentry Brocket Brocket Concentry Brocket	Image:
	Image: Inf i i i i i i i i i i i i i i i i i i i
Pres Router Advertisement Settling	Image: of 1: 0 (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)

Figure 75 Configuration > Network > Interface > Ethernet > Edit (External Type)

constal Settings	
Enable Interface	
General IPv6 Setting	
E Dable Pré	
Interface Properties	
Interface Type:	Internal
Interface Name:	dra
Zane:	Dest .
HAC Address	00:00: AA: 79: 72: 78
Descriptions	(Dottore)
IP Address Assistment	
P Abbras	192.158.3.1
Subnet Hask:	255.255.255.0
IPv6 Address Assignment	
🗄 Brable Stateless Address Auto-configura	ton (SLAAC)
LPR COCH Address:	n/w
Setement	(Cettoral)
Metric	(0-13)
Address from DHCPv6 Prefix Delegation	0 AM (2)
	# Delegated Prefs Suffix Address Address
	() () Page () of 5 () () (Page (1) ure to data to diplay
DHCPv6 Setting	
DKPVE	Server w
CUE:	00-03-00-01-00-0014A-76-75-76
(Z) DUD as MAC Containing (DUD)	
E thable Rapid Conmit	
Information Red and Toron	ETABLETSK AND AND ADDRESS AND
DHCP-4 Lease Optime	
	# Name - Type Value
	N G Face 1 of 1 2 N Dear M Ave Maddate dealer
	An an other first and the second states of the second states in
Pv6 Router Advertisement Setting	
Advertised Hosts Get Network Configure	ton Fran DHONS
Advertised Hosts Get Other Configuration	n Fran DriChvis
Router Preference:	Međuni
MTU:	1499 (1200-1500, 0 is deather!)
Advertised Prefix Table	97 97-233, 0 8 disabled)
	4 Pvl AccessPeck Large -
	Mid American States and American
	te a construction in the state of the second state
Advertised Prefix from DHCPvS Prefix Delegation	Q AM 🔂 📲 🍟
	# Delegated Prefs Suffix Appress Address
	$\left \left\{ \begin{array}{ccc} 0 & \text{if } 1 & \text{if } 1 & 0 & 0 \end{array} \right & \text{Show} \overline{H} & \text{with} & \text{to dota to deploy} \end{array} \right.$
Interface Parameters	$[0, 0] = \operatorname{Page}\left[\underline{t}_{-} \mathrm{of} t : [0, 0]\right] = \operatorname{Stem}\left[\overline{H_{-}} \underline{w}\right] \mathrm{en} \qquad \text{for data to display}$
Interface Parameters	 A de la serie de
Interface Parameters Egent Darbeiths Jogen Endwiths With	B 4 Page 1 af 1 3 Bene 10 No deta to deploy SHETS Kbox B SHETS Open
Diferface Parameters Opens Bandwith Segres Bandwith: HTu:	Bit 4 Page 1 of 1 3 30 Den 10 No No Side175 Klan 8 Side175 Klan 8 Side175 Klan 8 Side175 Klan Side175 Klan Side175 Klan
biteface Parameters Egass Bandwith Joguss Bandwith: HTM: INCP Setting	(b) (c)
Ditrifice Parameters Egres Badwith Signs Endosth Miga DECP Setting DECP	No. 4: Page 1: af 1: 3: 3: 19 on 18: and 19 Statt N:
biterface Parameters Egres Endwith Hypers Endwith Highers Endwith DECP Prot End Addres Electronic	No. 4 Page I. of 1. 3 3 Point III No. 4 Statistic Hans III IIII IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII
biterface Parameters Egres Banhath Higes Banhath Hig. DECF Setting DFCP IP Pool Dark Address (Doloral) Frist 2016 Series (Doloral) Secold Di Series (Doloral)	20 4 Page II of 1 3 20 No. 61 No. 61 No. 61 20 10 10 8 10 10 10 10 20 20 20 20 20 20 20 20 20 20 20 20 20
biterfect Parameters Syme Innivitin Sryme Encholdti Mille DeCh Prot Dart Address (Deloval) Frist Did Sever (Deloval) Second Did Sever (Deloval) Second Did Sever (Deloval)	No. 4 Page 1 #1.3 30 Powe 88 Worket Strettin Race B Strettin Race B Strettin Race B Strettin Powe Powe DVCP Server Powe Powe Strettin Powe Powe
biorfect Parameters Egres Bandwith Signes Bandwith HTLL HTLL BRCF Setting DHOT Prot Dark Address (Debrad) Prot Dark Address (Debrad) Prot Dark Address (Debrad) Second DHS Server (Datwad) Herd DDS Server (Datwad) Herd DDS Server (Datwad)	B 4 Page 1 af 1 3 Brow 18 Ver 10 Ver 20 B B B B B B B B B B
biterface Parameters types landwith Hypes landwith Hypes landwith Hypes landwith Hypes landwith DeCh Prot Deta Address Debund) Frest Dels Server (Debund) Hype Dels Server (Debund) Hype Dels Server (Debund) Server (Debund)	Bit di Inger L af 1. 5 30 them 88 wire He dels to daster Strattin Hen 8 Bit di Din Hen 8 Bit di Din Hen 8 Dels films
biterface Parameters Egres Banhath Joges Banhath Hitti Diges Banhath Hitti Dig Diffi P Part Dars Address Datoral) Frei Dars Address Datoral) Frei Dars Marees Datoral) Frei Darsen Disarres (Datoral) The Data Server (Datoral) Default Val Server (Datoral) Default Val Server (Datoral)	20 4 Page 41 3 30 Power No No 20 10 10 10 10 10 10 10 20 10 10 10 10 10 10 20 10 10 10 10 10 20 20 20 20 20 20 20 20 20 20 20 20 20 20 20 20 20 20 20 20 20
biterface Parameters Egres Bandwith Segres Bandwith Mitter DECF Setting DECF Setting DeCh Prot Start Address (Datoval) Find DMS Server (Datoval) Secand DHS Server (Datoval) Mathed DMS Server (Datoval) Secand WHS Server (Datoval)	No. 4 Page If 1 3 30 or 88 Ver at the data to data t
biterfect Parameters Syme Innivitin Syme Encholdts Mills DeCh Prot Dart Address (Deloval) Prot Dart Address (Deloval) Prot Dart Address (Deloval) Second Dif Same (Deloval) Mild Dif Same (Deloval) Anst VIDI Same (Deloval) Second Dif Same (Deloval) Second Dif Same (Deloval) Second Dif Same (Deloval) Lesse Time	Bit 4: Page: aff 1: 3: 3: 3: 20m; 8: arg; arg; to data to deploy Bit 45: 5: 10m; 10m; 10m; 10m; 10m; 10m; 10m; 10m;
biterface Parameters tyres landwith: highest landwith: highest landwith: highest landwith: biter Prot Dets Prot De	Si di Ager L af 1. 3 Xi How 81 w re He data to dataler Si ett 15 Han Brets Brets Det Sine Brets Brets <
biterface Parameters Egres Banhath Highes Banhath High Drift P Poel Dari Address Dotorell Hird Ord Serve (Datorell) Hird Ord Serve (Datorel) Hird Ord Serve (Datorel) Hird Ord Serve (Datorel) Cefaul Robert (Datorel) Cefaul Robert (Datorel) Cefaul Robert (Datorel)	No. 0 Page If 1 3 30 10 wer No. 0 10 HOT Some No. 0 Page No. 0 10 10 HOT Some No. 0 Page 20 20 HOT Some No. 0 Page 20 20 Hot Some No. 0 Page 10
biterface Runniefers Egres Banhath Higes Banhath Hig. TRCF Setting DeCh IP Pool Dars Address (Datonal) Frist Did Server (Datonal) Second Did Server (Datonal) The d Did Server (Datonal) Second Vici Server (Datonal) Default Russer (Datonal) Default Russer (Datonal) Default Russer (Datonal) Default Russer (Datonal) Default Russer (Datonal)	Note: Post See: 200 Direct Server Post See: 200
biterface Pursuedras Syste Bencheth: Miller DeCh DeCh Priot Start Address (Datoral) Frant Did Server (Datoral) Secand Did Server (Datoral) Secand Did Server (Datoral) Secand Did Server (Datoral) Secand Wild	Note: Post See: 200 Dec: Post See: 200
biorface Parameters tignes landwith: singues landwith: singues landwith: singue biorface Processes Pro	Image:
bierface Parameters Egres Bandwith Signes Bandwith Inter- Paras Bandwith Inter- Para Bandwith Inter Defin Prot Data Address Doctore() Prot Data Address Doctore() Prot Data Address Doctore() Prot Data Staree (Datawa) Second 102 Gener (Datawa) Second 102	Image:
Distrifues Parameters Egres Bandwatts Egres Bandwatts Entry Entry Entry Entry Entry Pred Dars Address Datoral) Pred Dars Address Datoral) Pred Dars Address Datoral) Pred Dars Address Datoral) Entry	Image:
biterface Parameters types landwith jogess landwith jogess landwith types landwith proc Drich Proc Data Address Detoval) Facod Discore Detoval	Image:
biorface Parameters tyres Bandwith: bigress Band	Image: If 1 is 30 (How III) image: How III image: How III image: Image: Image: How III image: How IIII image: Image: Image: Image: How IIII image: Image: Image: Image: How IIII image: How IIII image: Image: Image: Image: How IIII image: How IIII image: How IIII image: Image: Image: Image: Image: How IIII image: How IIIII image: How IIIII image: How IIII image: How IIIIIIIIII image: How IIIII image: How IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII
biterface Parameters Egres Bandwith Signes Bandwith Signes Bandwith Signes Bandwith Signes DeCh Prot User Address (Debred) Fast Did Sinner (Debred) Fast Did Sinner (Debred) Fast Did Sinner (Debred) Fast Did Sinner (Debred) Cefeal Risure (Debred) Cefeal Ris	Image:
biterface Parameters Egres Bandwith Signes Bandwith Infig. biter Setting Drich P Pool Dari Address Datoral) Ansi 105 Serve (Datoral) Terd Dit Serve (Datoral) Terd Dit Serve (Datoral) Cellul Rischer (Datoral) Cellul Rischer (Datoral) Cellul Rischer (Datoral) Estended Coters Infield Rischer (Datoral) Estended Coters Biter Setting Intel setting RIP Setting RIP Setting Cellura Rischer Centors Stati Drich Rischer Centors Stati Drich Rischer Centors Stati Drich Rischer Centors Stati Drich Rischer RIP Setting Centors Stati Drich Rischer Centors Stati Drich Rischer Centors Stati Setting Centors Stati Seting Centors	Image:
biorface Parameters types landwith joges landwith joges landwith provide provide provide provide provide provide provide provide provide provide provide provide provide provide provide provide prov	Image:
biorfact Parameters tyres landwith: singues landwith: singues landwith: singue bord Drofn Prot Data Address (Dataval) Anst Data Same (Dataval)	Image: If is a low if is
bisroface Parameters Egree Bandwith Signes Bandwith Signes Bandwith Signes Bandwith Signes Bandwith Signes Bandwith Signes Bandwith Prof Unit Since (Datawa) Fast 2016 Same (Datawa) Samel 2016 Same	Image:
	Image:
biorface Parameters types Bandwith bigges Bandwith bi	Image:
biorface Parameters types Endwicht: strace provide the second	Image:
bisroface Parameters Egree Bandwith: Pagess Bandwith: HTLL Pagess Bandwith: HTLL Pagess Bandwith: HTLL Pagess Bandwith: HTLL Page Bandwith: HTLL	Image:

Figure 76 Configuration > Network > Interface > Ethernet > Edit (Internal Type)

ieneral Settings	
I Enable Interface	
ieneral IPv6 Setting	
Enable IPv6	
Interface Properties	assard V
Interface Name:	and and a second s
Ports	F3
Zonei	077
MAC Address	00:00:AA:79:72:64
Description:	(Optional)
P Address Assignment	
C Get Automatically	
IP Address	0000
Subriet Maski	0.000
Gatericayo	(Optional)
Metric:	0 (0-15)
Pv6 Address Assignment	
Line Enable Stateless Address Auto-con Unk-Local Address	nguration (SLAAG, J
IPv6 Address/Prefx Length:	(Defenal)
A CONTRACTOR OF A CONTRACTOR A	(observe)
HCPv6 Setting	
DHCPv6:	Server 💌
DUID:	00103-00101-00-001AA-179-72-04
DHCPv6 Lease Options	Add Barrow Could Advance
	# Name . Type Value
	Id d Page 1 of 1 b bl Show 50 with No data to deplay
	14 4
Pv6 Router Advertisement Setting	
Enable Router Advertisement	
Router Preference:	Medum Y
Advertised Prefix Table	O Add Balan 🖀 Addar
	# IPv6 Address Prefix Langth
	R d Page 1 of 1 D D Show an w terms No data to deplay
	II S Page 1 of 1 D D Show to the term to destiny
	[]
nterface Parameters Ecross Bandwidth	Image: The state is a state in the state in the state is a state in the state in the state is a state in the state in the state is a state in the state in the state is a state in the state in the state is a state in the state in the state is a state in the stat
nterface Parameters Egrass Bendniddhs	[0] Q Page E dt D Dean <u>to were</u> No data to deplay. 10405% star
nterface Parameters Egress Bardwith: Innactivity Chack	[0] 0] page 2 dt [0] Show 50 10 three No data to deplay. 10407% stips
nterface Parameters Egnes Bandwiths Canactivity Chack En Snible Connectivity Chack	[4] 4] Page E of t [] [] [Phone [50] [20 mm] Research to display. Integrat steps
Iterface Parameters Egress Bandnikh: Izmastiviky Chack Estable Connectivity Chack Chack Method:	Image E of t Image E of t Image E Image E Indepty Image E Image E Image E Image E Image E
Epress Bandwidts Epress Bandwidts Enable Connectivity Check Check Hethods Check Hethods	Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image
eterface Parameters Egrass Bendrukh Estate Connectivity Check Check Nethods Check Nethods Check Transoct	Id Id Page 2: of 1: Id Id Page 2: Id
Egrass Bendnolfth Egrass Bendnolfth Caractivity Chack Chack Method Chack Method Chack Francol Chack Francol	Image Image <td< td=""></td<>
Epres Baramoters Epres Bardwith: Canachting Clark Debk Method Check Method Check Fareod: Check Transot: Check Transot: Check Transot: @ Check Debuck Gateway	Image Image <td< td=""></td<>
Epress Bandwidth: Epress Bandwidth: Enable Connectivity Check Orleck Herbol: Orleck Herbol: Orleck Parcol: Orleck Transut: Orleck Transut: Orleck Transut: Orleck Debut Otlenwy © Orleck this address	Image Image <td< td=""></td<>
Epres Bendvidte Epres Bendvidte Datability Obeck Datability Obeck Deak Herdol Obek Herdol Obek Herdol Obek Herdol Obek Fail Tolerance ® Obeck Obeck Gateway © Obek Herdol Obek Herdol Bender Stationen Bender Bender Stationen Bender Stationen Bender St	Image Image <td< td=""></td<>
Epres Bandwith: Epres Bandwith: Senactivity Clark Deak (Nethod: Check Nethod: Check Nethod: Check Person: Check Person: @ Check Default Clarway @ Check District Clarway @ Check Banddress HCP Setting DeCh	Image Image <td< td=""></td<>
Aterface Parameters Egress Bendwith: Canachtivity Clack Chelk Mathod: Chelk Parodi: Chelk Parodi: Chelk Parodi: Chelk Parodi: Chelk Trianout: @ Chelk Datush Gateway @ Chelk Datush Gateway MCP Setting Dech: Papel Bet Ableyse (Detounds):	Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second state Image: Control in the second st
Atorface Parameters Epress Bandwidth Epress Bandwidth Enable Connectivity Check Check Natrods Check Natrods Check Parcods Check	Image Image Image Image Image Image Image 100000000 Image Image Image Image Image 100000000 Image Image Image Image 1000000000000000000000000000000000000
Epres Bed-obt: Epres Bed-obt: Enactivity Check Enact Consectively Check OneX Period: Check Period: Check Period: Check Period: Check Feat Tolernos: © Check Orbit. 2 Genery © Check Orbit. 2 Genery © Check Deht. 2 Genery Period: Dech Ethol Server (Doctorel): Prot Did Server (Doctorel):	Image Image <td< td=""></td<>
Egres Baramoters Egres Bandwith:	Image: Contract of the processing of the second
Alerfas Parameters Epress Bendvikhi	Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Control of a logic brane Non-Attack to directiony Image: Contre Non-Attack to ditera
	Image Image <td< td=""></td<>
Express Bandwiths Express Bandwiths Details Consectivity Check Check Hethods Check Hethods Check Introd. Check Fail Tolarancei © Check Dateback Observer Check Teal Dateback Observer DetCh Express Express DetCh Express Express (Optional): Prot Ditis Server (Optional): Prot UNIOS Server (Optional)	Image Image <td< td=""></td<>
Express Baramoters Express Bardwith Express Bardwith Express Bardwith Express Bardwith Check Hethod: Check Hethod: Check Parodi: Check Parodi: Check Parodi: Check Parodi: Check Talocincos @: Check Debut Cateway @: Check Talocincos PHCP Settion EMCP Set	Image Image <td< td=""></td<>
Alerface Parameters Egress Bandwidth Egress Bandwidth Estable Connectivity Check Coded Karlod Check Annod Check Annod Check Fail Colances @ Check Database Check Fail Colances @ Check Database Check Fail Colances Band Check Samer (Optional): Second DNS Samer (Optional): Dend Dist Samer (Optional): Second DNS Samer (Optional): Dend Dist Samer (Optional): Dend Dist Samer (Optional): Dend VIDIS Samer (Optional): Distribut Samer (Optional): Dend VIDIS	Image Image <td< td=""></td<>
	Image Image <td< td=""></td<>
Egres Barameters Egres Bandwith: Canactivity Clark Detection of Clark Method: Check Method: Check Method: Check Parodi: Check Transuct: Check	Image Image <td< td=""></td<>
Alerfans Parameters Egress Bendwith: Canactivity Clark Debis Consekt-Ity Otek Otek Method: Otek Method: Otek Perodi Otek Perodi Otek Tolence: ®: Otek Debis-R Getowy ®: Otek Debis-R Getowy Brod Start Alexes (Optional): Second DI/S Server (Optional): Second U/SI/S Server (Optional): Debis Rouder (Optional): Lass Time: Baended Options	Image Image
Alerface Parameters Egress Bandwidth: Egress Bandwidth: Denable Connective Oheck Oheck Natricel: Oheck Natricel: Oheck Natricel: Oheck Treact: Check Fail Tolerance: @ Oheck Utility Orden: @ Oheck Utility Otility Otility Otility Otility Otility Otility Otility @ Oheck Utility Otility Otilit	Image Image <td< td=""></td<>
Egres Bandwith: Searchivity Clark Deals (Constituty Chark Chark Method: Chark Method: Chark Method: Chark Immobili Chark Timobili Chark Timobili Chark Deals Orbital Cateway Chark Method: Chark Method: Cha	Image Image <td< td=""></td<>
	Image Image Image Image
Alerfas Parameters Egras Bandwith:	Image Image <td< td=""></td<>
Egres Bandwith: Egres Bandwith: Searchively Clack Check Herbol: Check Herbol: Check Parodi: Check Parodi: Check Parodi: Check Parodi: Check Parodi: Check Parodi: Check Parodi: Check Parodi: Parodi Start Address (Optional): Parod Did Server (Optional): Parod Did Server (Optional): Parod Did Server (Optional): Second Did Server (Optional): Parod Did Server (Optional): Second VI/US Server (Optional): Defund Paroder (Optional): Second VI/US Server (Optional): Defund Paroder (Optional	Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image
Express Bartanders Express Bardwiths Express Bardwiths Express Bardwiths Check Method: Check Method: Check Barod: Check Ba	Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image
Alerian Parameters Egress Bendvikh: Egress Bendvikh: Check Method: Check Method: Check Nethod: Check Parodi: Check Parodi: Check Trancot: Check Parodi: Check Trancot: Check Parodi: Check Trancot: Check Parodi: Check Trancot: Check Parodi: Check Parodi: Parod Bendve (Deformal): First Dick Server (Deformal): First UNIC Sarver (Deformal): Escond UNICS Server (Deformal): Escond UNIC	Image Image <td< td=""></td<>
Alerfas Parameters Egrass Bendenübis Densels Gonnectivity Check Doké Method Check Annodi Check Annodi Check Trinocoti Check Fill Tolannore P Code Default Gotomy P Poel Start Address (Optional): Becond DNS Sarver (Optional): Becond DNS Sarver (Optional): Becond DNS Sarver (Optional): Becond DNS Sarver (Optional): Becond VIDS Sarver (Optional): Betonided Options Betonided Options Default Acuters Ducch Table Ducch Table Ducch Table Dinke Emding Vibility Default Acuters Ducch Table Ducch Table Table Caddress Ducch Table Ducch Table Table Caddress Ducch Table Table Table Caddress Ducch Table Table Caddress Ducch Table Table Caddress Ducch Table Table Table Caddress Ducch Table Table Table Table Caddress Ducch Table Table Table Table Caddress Ducch Table Table Table Caddress Ducch Table	Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image Image
Express Bartanders Express Bardwidth: Express Bardwidth: Express Bardwidth: Check Herbod: Check Herbod: Check Herbod: Check Herbod: Check Transchi Derbol Start Aldress (Optional): Execut DIS Server (Optional):	Image Image Image </td
	Image Image Image </td
Alerfais Parameters Egress Bendwith: Egress Bendwith: Deschaft Occode-We Ofesk Oriek Method: Oriek Method: Oriek Parod: Defch: D	Image Image Image Image Image Image Image Image Image Image Image Image

Figure 77	Configuration	>	Network >	Interface >	Ethernet >	· Edit	(OPT)
-----------	---------------	---	-----------	-------------	------------	--------	-------

This screen's fields are described in the table below.

Tuble 41 Configurat	
LABEL	DESCRIPTION
IPv4/IPv6 View / IPv4 View / IPv6 View	Use this button to display both IPv4 and IPv6, IPv4-only, or IPv6-only configuration fields.
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
Create New Object	Click this button to create a DHCPv6 lease or DHCPv6 request object that you may use for the DHCPv6 settings in this screen.
General Settings	
Enable Interface	Select this to enable this interface. Clear this to disable this interface.
General IPv6 Setting	
Enable IPv6	Select this to enable IPv6 on this interface. Otherwise, clear this to disable it.
Interface Properties	
Interface Type	This field is configurable for the OPT interface only. Select to which type of network you will connect this interface. When you select internal or external the rest of the screen's options automatically adjust to correspond. The ZyWALL automatically adds default route and SNAT settings for traffic it routes from internal interfaces to external interfaces; for example LAN to WAN traffic.
	internal is for connecting to a local network. Other corresponding configuration options: DHCP server and DHCP relay. The ZyWALL automatically adds default SNAT settings for traffic flowing from this interface to an external interface.
	external is for connecting to an external network (like the Internet). The ZyWALL automatically adds this interface to the default WAN trunk.
	For general , the rest of the screen's options do not automatically adjust and you must manually configure a policy route to add routing and SNAT settings for the interface.
Interface Name	Specify a name for the interface. It can use alphanumeric characters, hyphens, and underscores, and it can be up to 11 characters long.
Port	This is the name of the Ethernet interface's physical port.
Zone	Select the zone to which this interface is to belong. You use zones to apply security settings such as firewall, remote management.
MAC Address	This field is read-only. This is the MAC address that the Ethernet interface uses.
Description	Enter a description of this interface. It is not used elsewhere. You can use alphanumeric and () +/:=?!*#@\$_%- characters, and it can be up to 60 characters long.
IP Address Assignment	These IP address fields configure an IPv4 IP address on the interface itself. If you change this IP address on the interface, you may also need to change a related address object for the network connected to the interface. For example, if you use this screen to change the IP address of your LAN interface, you should also change the corresponding LAN subnet address object.
Get Automatically	This option appears when Interface Type is external or general . Select this to make the interface a DHCP client and automatically get the IP address, subnet mask, and gateway address from a DHCP server.
	You should not select this if the interface is assigned to a VRRP group. See Chapter 26 on page 349.
Use Fixed IP Address	This option appears when Interface Type is external or general . Select this if you want to specify the IP address, subnet mask, and gateway manually.

Enter the IP address for this interface.

 Table 41
 Configuration > Network > Interface > Ethernet > Edit

IP Address

LABEL	DESCRIPTION						
Subnet Mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.						
Gateway	This option appears when Interface Type is external or general . Enter the IP address of the gateway. The ZyWALL sends packets to the gateway when it does not know how to route the packet to its destination. The gateway should be on the same network as the interface.						
Metric	This option appears when Interface Type is external or general . Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.						
IPv6 Address Assignment	These IP address fields configure an IPv6 IP address on the interface itself.						
Enable Stateless Address Auto- configuration (SLAAC)	Select this to enable IPv6 stateless auto-configuration on this interface. The interface will generate an IPv6 IP address itself from a prefix obtained from an IPv6 router in the network.						
Link-Local address	This displays the IPv6 link-local address and the network prefix that the ZyWALL generates itself for the interface.						
IPv6 Address/ Prefix Length	Enter the IPv6 address and the prefix length for this interface if you want to use a static IP address. This field is optional.						
	The prefix length indicates what the left-most part of the IP address is the same for all computers in the network, that is, the network address.						
Gateway	Enter the IPv6 address of the default outgoing gateway using colon (:) hexadecimal notation.						
Metric	Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.						
Address from DHCPv6 Prefix Delegation	Use this table to have the ZyWALL obtain an IPv6 prefix from the ISP or a connected uplink router for an internal network, such as the LAN or DMZ. You have to also enter a suffix address which is appended to the delegated prefix to form an address for this interface. See Prefix Delegation on page 107 for more information.						
	To use prefix delegation, you must:						
	 Create at least one DHCPv6 request object before configuring this table. The external interface must be a DHCPv6 client. You must configure the DHCPv6 request options using a DHCPv6 request object with the type of prefix-delegation. Assign the prefix delegation to an internal interface and enable router advertisement on that interface. 						
Add	Click this to create an entry.						
Edit	Select an entry and click this to change the settings.						
Remove	Select an entry and click this to delete it from this table.						
#	This field is a sequential value, and it is not associated with any entry.						
Delegated Prefix	Select the DHCPv6 request object to use from the drop-down list.						
Suffix Address	Enter the ending part of the IPv6 address, a slash (/), and the prefix length. The ZyWALL will append it to the delegated prefix.						
	For example, you got a delegated prefix of 2003:1234:5678/48. You want to configure an IP address of 2003:1234:5678:1111::1/128 for this interface, then enter ::1111:0:0:0:1/128 in this field.						

 Table 41
 Configuration > Network > Interface > Ethernet > Edit (continued)

LABEL	DESCRIPTION					
Address	This field displays the combined IPv6 IP address for this interface.					
	Note: This field displays the combined address after you click OK and reopen this screen.					
DHCPv6 Setting						
DUID	This field displays the DHCP Unique IDentifier (DUID) of the interface, which is unique and used for identification purposes when the interface is exchanging DHCPv6 messages with others. See DHCPv6 on page 107 for more information.					
DUID as MAC	Select this if you want the DUID is generated from the interface's default MAC address.					
Customized DUID	If you want to use a customized DUID, enter it here for the interface.					
Enable Rapid Commit	Select this to shorten the DHCPv6 message exchange process from four to two steps. This function helps reduce heavy network traffic load.					
	Note: Make sure you also enable this option in the DHCPv6 clients to make rapid commit work.					
Information Refresh Time	Enter the number of seconds a DHCPv6 client should wait before refreshing information retrieved from DHCPv6.					
Request Address	This field is available if you set this interface to DHCPv6 Client . Select this to get an IPv6 IP address for this interface from the DHCP server. Clear this to not get any IP address information through DHCPv6.					
DHCPv6 Request Options / DHCPv6 Lease Options	If this interface is a DHCPv6 client, use this section to configure DHCPv6 request settings that determine what additional information to get from the DHCPv6 server. If the interface is a DHCPv6 server, use this section to configure DHCPv6 lease settings that determine what additional information to offer to the DHCPv6 clients.					
Add	Click this to create an entry in this table. See Section 7.3.3 on page 123 for more information.					
Remove	Select an entry and click this to delete it from this table.					
Object Reference	Select an entry and click Object Reference to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.					
#	This field is a sequential value, and it is not associated with any entry.					
Name	This field displays the name of the DHCPv6 request or lease object.					
Туре	This field displays the type of the object.					
Value	This field displays the IPv6 prefix that the ZyWALL obtained from an uplink router (Server is selected) or will advertise to its clients (Client is selected).					
Interface	When Relay is selected, select this check box and an interface from the drop-down list if you want to use it as the relay server.					
Relay Server	When Relay is selected, select this check box and enter the IP address of a DHCPv6 server as the relay server.					
IPv6 Router Advertisement Setting						
Enable Router Advertisement	Select this to enable this interface to send router advertisement messages periodically. See IPv6 Router Advertisement on page 107 for more information.					
Advertised Hosts Get Network Configuration	Select this to have the ZyWALL indicate to hosts to obtain network settings (such as prefix and DNS settings) through DHCPv6.					
From DHCPv6	Clear this to have the ZyWALL indicate to hosts that DHCPv6 is not available and they should use the prefix in the router advertisement message.					

Table 41 Configuration > Network > Interface > Ethernet > Edit (continued)

LABEL	DESCRIPTION					
Advertised Hosts Get Other Configuration	Select this to have the ZyWALL indicate to hosts to obtain DNS information through DHCPv6.					
From DHCPv6	Clear this to have the ZyWALL indicate to hosts that DNS information is not available in this network.					
Router Preference	Select the router preference (Low , Medium or High) for the interface. The interface sends this preference in the router advertisements to tell hosts what preference they should use for the ZyWALL. This helps hosts to choose their default router especially when there are multiple IPv6 router in the network.					
	Note: Make sure the hosts also support router preference to make this function work.					
MTU	The Maximum Transmission Unit. Type the maximum size of each IPv6 data packet, in bytes, that can move through this interface. If a larger packet arrives, the ZyWALL discards the packet and sends an error message to the sender to inform this.					
Hop Limit	Enter the maximum number of network segments that a packet can cross before reaching the destination. When forwarding an IPv6 packet, IPv6 routers are required to decrease the Hop Limit by 1 and to discard the IPv6 packet when the Hop Limit is 0.					
Advertised Prefix Table	Configure this table only if you want the ZyWALL to advertise a fixed prefix to the network.					
Add	Click this to create an IPv6 prefix address.					
Edit	Select an entry in this table and click this to modify it.					
Remove	Select an entry in this table and click this to delete it.					
#	This field is a sequential value, and it is not associated with any entry.					
IPv6	Enter the IPv6 network prefix address and the prefix length.					
Prefix Length	The prefix length indicates what the left-most part of the IP address is the same for all computers in the network, that is, the network address.					
Advertised Prefix from DHCPv6 Prefix Delegation	This table is available when the Interface Type is internal . Use this table to configure the network prefix if you want to use a delegated prefix as the beginning part of the network prefix.					
Add	Click this to create an entry in this table.					
Edit	Select an entry in this table and click this to modify it.					
Remove	Select an entry in this table and click this to delete it.					
#	This field is a sequential value, and it is not associated with any entry.					
Delegated Prefix	Select the DHCPv6 request object to use for generating the network prefix for the network.					
Suffix Address	Enter the ending part of the IPv6 network address plus a slash (/) and the prefix length. The ZyWALL will append it to the selected delegated prefix. The combined address is the network prefix for the network.					
	For example, you got a delegated prefix of 2003:1234:5678/48. You want to divide it into 2003:1234:5678:1111/64 for this interface and 2003:1234:5678:2222/64 for another interface. You can use ::1111/64 and ::2222/64 for the suffix address respectively. But if you do not want to divide the delegated prefix into subnetworks, enter ::0/48 here, which keeps the same prefix length (/48) as the delegated prefix.					
Address	This is the final network prefix combined by the delegated prefix and the suffix.					
	Note: This field displays the combined address after you click OK and reopen this screen.					
Interface Parameters						

 Table 41
 Configuration > Network > Interface > Ethernet > Edit (continued)

LABEL	DESCRIPTION				
Egress Bandwidth	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can send through the interface to the network. Allowed values are 0 - 1048576.				
Ingress Bandwidth	This is reserved for future use.				
Danawidin	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can receive from the network through the interface. Allowed values are 0 - 1048576.				
MTU	Maximum Transmission Unit. Type the maximum size of each data packet, in bytes, that can move through this interface. If a larger packet arrives, the ZyWALL divides it into smaller fragments. Allowed values are 576 - 1500. Usually, this value is 1500.				
Connectivity Check	These fields appear when Interface Properties is External or General.				
	The interface can regularly check the connection to the gateway you specified to make sure it is still available. You specify how often the interface checks the connection, how long to wait for a response before the attempt is a failure, and how many consecutive failures are required before the ZyWALL stops routing to the gateway. The ZyWALL resumes routing to the gateway the first time the gateway passes the connectivity check.				
Enable Connectivity Check	Select this to turn on the connection check.				
Check Method	Select the method that the gateway allows.				
	Select icmp to have the ZyWALL regularly ping the gateway you specify to make sure it is still available.				
	Select tcp to have the ZyWALL regularly perform a TCP handshake with the gateway you specify to make sure it is still available.				
Check Period	Enter the number of seconds between connection check attempts.				
Check Timeout	Enter the number of seconds to wait for a response before the attempt is a failure.				
Check Fail Tolerance	Enter the number of consecutive failures before the ZyWALL stops routing through the gateway.				
Check Default Gateway	Select this to use the default gateway for the connectivity check.				
Check this address	Select this to specify a domain name or IP address for the connectivity check. Enter that domain name or IP address in the field next to it.				
Check Port	This field only displays when you set the Check Method to tcp . Specify the port number to use for a TCP connectivity check.				
DHCP Setting	This section appears when Interface Type is internal or general.				
DHCP	Select what type of DHCP service the ZyWALL provides to the network. Choices are:				
	None - the ZyWALL does not provide any DHCP services. There is already a DHCP server on the network.				
	DHCP Relay - the ZyWALL routes DHCP requests to one or more DHCP servers you specify. The DHCP server(s) may be on another network.				
	DHCP Server - the ZyWALL assigns IP addresses and provides subnet mask, gateway, and DNS server information to the network. The ZyWALL is the DHCP server for the network.				
	These fields appear if the ZyWALL is a DHCP Relay.				
Relay Server 1	Enter the IP address of a DHCP server for the network.				
Relay Server 2	This field is optional. Enter the IP address of another DHCP server for the network.				
	These fields appear if the ZyWALL is a DHCP Server.				

Table 41 Co	nfiguration >	Network >	Interface >	Ethernet >	Edit	(continued)
-------------	---------------	-----------	-------------	------------	------	-------------

LABEL DESCRIPTION		
IP Pool Start Address	Enter the IP address from which the ZyWALL begins allocating IP addresses. If you want to assign a static IP address to a specific computer, use the Static DHCP Table .	
	If this field is blank, the Pool Size must also be blank. In this case, the ZyWALL can assign every IP address allowed by the interface's IP address and subnet mask, except for the first address (network address), last address (broadcast address) and the interface's IP address.	
Pool Size	Enter the number of IP addresses to allocate. This number must be at least one and is limited by the interface's Subnet Mask . For example, if the Subnet Mask is 255.255.255.0 and IP Pool Start Address is 10.10.10.10, the ZyWALL can allocate 10.10.10.10 to 10.10.254, or 245 IP addresses.	
	If this field is blank, the IP Pool Start Address must also be blank. In this case, the ZyWALL can assign every IP address allowed by the interface's IP address and subnet mask, except for the first address (network address), last address (broadcast address) and the interface's IP address.	
First DNS Server, Second DNS Server	Specify the IP addresses up to three DNS servers for the DHCP clients to use. Use one of the following ways to specify these IP addresses.	
Third DNS Server	Custom Defined - enter a static IP address.	
	From ISP - select the DNS server that another interface received from its DHCP server.	
	ZyWALL - the DHCP clients use the IP address of this interface and the ZyWALL works as a DNS relay.	
First WINS Server, Second WINS Server	Type the IP address of the WINS (Windows Internet Naming Service) server that you want to send to the DHCP clients. The WINS server keeps a mapping table of the computer names on your network and the IP addresses that they are currently using.	
Default Router	If you set this interface to DHCP Server , you can select to use either the interface's IP address or another IP address as the default router. This default router will become the DHCP clients' default gateway.	
	To use another IP address as the default router, select Custom Defined and enter the IP address.	
Lease time	Specify how long each computer can use the information (especially the IP address) before it has to request the information again. Choices are:	
	infinite - select this if IP addresses never expire.	
	days, hours, and minutes - select this to enter how long IP addresses are valid.	
Extended	This table is available if you selected DHCP server .	
Options	Configure this table if you want to send more information to DHCP clients through DHCP packets.	
Add	Click this to create an entry in this table. See Section 7.3.4 on page 124.	
Edit	Select an entry in this table and click this to modify it.	
Remove	Select an entry in this table and click this to delete it.	
#	This field is a sequential value, and it is not associated with any entry.	
Name	This is the name of the DHCP option.	
Code	This is the code number of the DHCP option.	
Туре	This is the type of the set value for the DHCP option.	
Value	This is the value set for the DHCP option.	

 Table 41
 Configuration > Network > Interface > Ethernet > Edit (continued)

LABEL	DESCRIPTION
Enable IP/MAC Binding	Select this option to have this interface enforce links between specific IP addresses and specific MAC addresses. This stops anyone else from manually using a bound IP address on another device connected to this interface. Use this to make use only the intended users get to use specific IP addresses.
Enable Logs for IP/MAC Binding Violation	Select this option to have the ZyWALL generate a log if a device connected to this interface attempts to use an IP address that is bound to another device's MAC address.
Static DHCP Table	Configure a list of static IP addresses the ZyWALL assigns to computers connected to the interface. Otherwise, the ZyWALL assigns an IP address dynamically using the interface's IP Pool Start Address and Pool Size .
Add	Click this to create a new entry.
Edit	Select an entry and click this to be able to modify it.
Remove	Select an entry and click this to delete it.
#	This field is a sequential value, and it is not associated with a specific entry.
IP Address	Enter the IP address to assign to a device with this entry's MAC address.
MAC	Enter the MAC address to which to assign this entry's IP address.
Description	Enter a description to help identify this static DHCP entry. You can use alphanumeric and () +/ :=?!*#@ $_{e}^{-}$ characters, and it can be up to 60 characters long.
RIP Setting	See Section 10.2 on page 197 for more information about RIP.
Enable RIP	Select this to enable RIP in this interface.
Direction	This field is effective when RIP is enabled. Select the RIP direction from the drop-down list box.
	BiDir - This interface sends and receives routing information.
	In-Only - This interface receives routing information.
	Out-Only - This interface sends routing information.
Send Version	This field is effective when RIP is enabled. Select the RIP version(s) used for sending RIP packets. Choices are 1 , 2 , and 1 and 2 .
Receive Version	This field is effective when RIP is enabled. Select the RIP version(s) used for receiving RIP packets. Choices are 1 , 2 , and 1 and 2 .
V2-Broadcast	This field is effective when RIP is enabled. Select this to send RIP-2 packets using subnet broadcasting; otherwise, the ZyWALL uses multicasting.
OSPF Setting	See Section 10.3 on page 199 for more information about OSPF.
Area	Select the area in which this interface belongs. Select None to disable OSPF in this interface.
Priority	Enter the priority (between 0 and 255) of this interface when the area is looking for a Designated Router (DR) or Backup Designated Router (BDR). The highest-priority interface identifies the DR, and the second-highest-priority interface identifies the BDR. Set the priority to zero if the interface can not be the DR or BDR.
Link Cost	Enter the cost (between 1 and 65,535) to route packets through this interface.
Passive Interface	Select this to stop forwarding OSPF routing information from the selected interface. As a result, this interface only receives routing information.

Table 41 Configuration > Network > Interface > Ethernet > Edit (continued)

LABEL	DESCRIPTION
Authentication	Select an authentication method, or disable authentication. To exchange OSPF routing information with peer border routers, you must use the same authentication method that they use. Choices are:
	Same-as-Area - use the default authentication method in the area
	None - disable authentication
	Text - authenticate OSPF routing information using a plain-text password
	MD5 - authenticate OSPF routing information using MD5 encryption
Text Authentication Key	This field is available if the Authentication is Text . Type the password for text authentication. The key can consist of alphanumeric characters and the underscore, and it can be up to 16 characters long.
MD5 Authentication ID	This field is available if the Authentication is MD5 . Type the ID for MD5 authentication. The ID can be between 1 and 255.
MD5 Authentication Key	This field is available if the Authentication is MD5 . Type the password for MD5 authentication. The password can consist of alphanumeric characters and the underscore, and it can be up to 16 characters long.
MAC Address Setting	This section appears when Interface Properties is External or General . Have the interface use either the factory assigned default MAC address, a manually specified MAC address, or clone the MAC address of another device or computer.
Use Default MAC Address	Select this option to have the interface use the factory assigned default MAC address. By default, the ZyWALL uses the factory assigned MAC address to identify itself.
Overwrite Default MAC Address	Select this option to have the interface use a different MAC address. Either enter the MAC address in the fields or click Clone by host and enter the IP address of the device or computer whose MAC you are cloning. Once it is successfully configured, the address will be copied to the configuration file. It will not change unless you change the setting or upload a different configuration file.
Related Setting	
Configure PPPoE/PPTP	Click PPPoE/PPTP if this interface's Internet connection uses PPPoE or PPTP.
Configure VLAN	Click VLAN if you want to configure a VLAN interface for this Ethernet interface.
Configure WAN TRUNK	Click WAN TRUNK to go to a screen where you can set this interface to be part of a WAN trunk for load balancing.
Configure Policy Route	Click Policy Route to go to the policy route summary screen where you can manually associate traffic with this interface.
	You must manually configure a policy route to add routing and SNAT settings for an interface with the Interface Type set to general . You can also configure a policy route to override the default routing and SNAT behavior for an interface with an Interface Type of internal or external .
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

 Table 41
 Configuration > Network > Interface > Ethernet > Edit (continued)

7.3.2 Object References

When a configuration screen includes an **Object Reference** icon, select a configuration object and click **Object Reference** to open the **Object References** screen. This screen displays which configuration settings reference the selected object. The fields shown vary with the type of object.

Figure 78 Object References

#	Service 🔺	Priority	Name	Description	
1	Address	N/A	LAN1_SUBNET	N/A	
2	Zone: System Default	N/A	LAN1	N/A	
14	4 Page 1 of 1	🕨 🕅 🛛 She	ow 50 💉 items	Displaying 1	1 - 2 of 2

The following table describes labels that can appear in this screen.

LABEL	DESCRIPTION
Object Name	This identifies the object for which the configuration settings that use it are displayed. Click the object's name to display the object's configuration screen in the main window.
#	This field is a sequential value, and it is not associated with any entry.
Service	This is the type of setting that references the selected object. Click a service's name to display the service's configuration screen in the main window.
Priority	If it is applicable, this field lists the referencing configuration item's position in its list, otherwise N/A displays.
Name	This field identifies the configuration item that references the object.
Description	If the referencing configuration item has a description configured, it displays here.
Refresh	Click this to update the information in this screen.
Cancel	Click Cancel to close the screen.

Table 42 Object References

7.3.3 Add/Edit DHCPv6 Request/Release Options

When you configure an interface as a DHCPv6 server or client, you can additionally add DHCPv6 request or lease options which have the ZyWALL to add more information in the DHCPv6 packets. To open the screen, click **Configuration > Network > Interface > Ethernet > Edit**, select **DHCPv6 Server** or **DHCPv6 Client** in the **DHCPv6 Setting** section, and then click **Add** in the **DHCPv6 Request Options** or **DHCPv6 Lease Options** table.

Figure 79 Configuration > Network > Interface > Ethernet > Edit > Add DHCPv6 Request/Lease Options

Select a DHCPv6 request or lease object in the **Select one object** field and click **OK** to save it. Click **Cancel** to exit without saving the setting.

7.3.4 Add/Edit DHCP Extended Options

When you configure an interface as a DHCPv4 server, you can additionally add DHCP extended options which have the ZyWALL to add more information in the DHCP packets. The available fields vary depending on the DHCP option you select in this screen. To open the screen, click **Configuration > Network > Interface > Ethernet > Edit**, select **DHCP Server** in the **DHCP Setting** section, and then click **Add** or **Edit** in the **Extended Options** table.

Figure 80	Configuration >	Network >	Interface >	Ethernet >	Edit >	Add/Edit	Extended	Options
Contraction of the second s				A second s				

	User Defined		*	
Name:	User_Defined			
Code:			D	
Туре:	BOOLEAN	~		
Value:				

The following table describes labels that can appear in this screen.

LABEL	DESCRIPTION
Option	Select which DHCP option that you want to add in the DHCP packets sent through the interface. See Table 44 for more information.
Name	This field displays the name of the selected DHCP option. If you selected User Defined in the Option field, enter a descriptive name to identify the DHCP option. You can enter up to 16 characters ("a-z", "A-Z, "0-9", "-", and "_") with no spaces allowed. The first character must be alphabetical (a-z, A-Z).
Code	This field displays the code number of the selected DHCP option. If you selected User Defined in the Option field, enter a number for the option. This field is mandatory.
Туре	This is the type of the selected DHCP option. If you selected User Defined in the Option field, select an appropriate type for the value that you will enter in the next field. Only advanced users should configure User Defined . Misconfiguration could result in interface lockout.
Value	Enter the value for the selected DHCP option. For example, if you selected TFTP Server Name (66) and the type is TEXT , enter the DNS domain name of a TFTP server here. This field is mandatory.
First IP Address, Second IP Address, Third IP Address	If you selected Time Server (4) , NTP Server (41) , SIP Server (120) , CAPWAP AC (138) , or TFTP Server (150) , you have to enter at least one IP address of the corresponding servers in these fields. The servers should be listed in order of your preference.
First Enterprise ID, Second Enterprise ID	If you selected VIVC (124) or VIVS (125) , you have to enter at least one vendor's 32- bit enterprise number in these fields. An enterprise number is a unique number that identifies a company.
First Class, Second Class	If you selected VIVC (124) , enter the details of the hardware configuration of the host on which the client is running, or of industry consortium compliance.

Table 43	Configuration >	Network >	Interface	> Ethernet >	Edit >	Add/Edit Extended	d Options
----------	-----------------	-----------	-----------	--------------	--------	-------------------	-----------

LABEL	DESCRIPTION
First Information, Second Information	If you selected VIVS (125) , enter additional information for the corresponding enterprise number in these fields.
ОК	Click this to close this screen and update the settings to the previous Edit screen.
Cancel	Click Cancel to close the screen.

 Table 43
 Configuration > Network > Interface > Ethernet > Edit > Add/Edit Extended Options

The following table lists the available DHCP extended options (defined in RFCs) on the ZyWALL. See RFCs for more information.

OPTION NAME	CODE	DESCRIPTION
Time Offset	2	This option specifies the offset of the client's subnet in seconds from Coordinated Universal Time (UTC).
Time Server	4	This option specifies a list of Time servers available to the client.
NTP Server	42	This option specifies a list of the NTP servers available to the client by IP address.
TFTP Server Name	66	This option is used to identify a TFTP server when the "sname" field in the DHCP header has been used for DHCP options. The minimum length of the value is 1.
Bootfile	67	This option is used to identify a bootfile when the "file" field in the DHCP header has been used for DHCP options. The minimum length of the value is 1.
SIP Server	120	This option carries either an IPv4 address or a DNS domain name to be used by the SIP client to locate a SIP server.
VIVC	124	Vendor-Identifying Vendor Class option A DHCP client may use this option to unambiguously identify the vendor that manufactured the hardware on which the client is running, the software in use, or an industry consortium to which the vendor belongs.
VIVS	125	Vendor-Identifying Vendor-Specific option DHCP clients and servers may use this option to exchange vendor-specific information.
CAPWAP AC	138	CAPWAP Access Controller addresses option The Control And Provisioning of Wireless Access Points Protocol allows a Wireless Termination Point (WTP) to use DHCP to discover the Access Controllers to which it is to connect. This option carries a list of IPv4 addresses indicating one or more CAPWAP ACs available to the WTP.
TFTP Server	150	The option contains one or more IPv4 addresses that the client may use. The current use of this option is for downloading configuration from a VoIP server via TFTP; however, the option may be used for purposes other than contacting a VoIP configuration server.

 Table 44
 DHCP Extended Options

7.4 PPP Interfaces

Use PPPoE/PPTP interfaces to connect to your ISP. This way, you do not have to install or manage PPPoE/PPTP software on each computer in the network.

Figure 81 Example: PPPoE/PPTP Interfaces



PPPoE/PPTP interfaces are similar to other interfaces in some ways. They have an IP address, subnet mask, and gateway used to make routing decisions; they restrict bandwidth and packet size; and they can verify the gateway is available. There are two main differences between PPPoE/PPTP interfaces and other interfaces.

• You must also configure an ISP account object for the PPPoE/PPTP interface to use.

Each ISP account specifies the protocol (PPPoE or PPTP), as well as your ISP account information. If you change ISPs later, you only have to create a new ISP account, not a new PPPoE/PPTP interface. You should not have to change any network policies.

• You do not set up the subnet mask or gateway.

PPPoE/PPTP interfaces are interfaces between the ZyWALL and only one computer. Therefore, the subnet mask is always 255.255.255.255. In addition, the ZyWALL always treats the ISP as a gateway.

7.4.1 PPP Interface Summary

This screen lists every PPPoE/PPTP interface. To access this screen, click **Configuration > Network > Interface > PPP**.

# -	Status	Name	Base Interface	Account Profile
14	4 Page	1 of 1 🕨 🕅 Show 50	▼ items	No data to display
ste	m Default			
2	Edit 🧧 Activ	ate 🔞 Inactivate 🚷 Connect	🚱 Disconnect 📴 Object I	Reference
#	Status	Name	Base Interface	Account Profile
1	88	ge1_ppp	ge1	GE1_PPPoE_ACCOUNT
2	8 8	ge2_ppp	ge2	GE2_PPPoE_ACCOUNT
3	88	ge3_ppp	ge3	GE3_PPPoE_ACCOUNT
4	8 6	ge4_ppp	ge4	GE4_PPPoE_ACCOUNT
5	88	ge5_ppp	ge5	GE5_PPPoE_ACCOUNT
6	88	ge6_ppp	ge6	GE6_PPPoE_ACCOUNT
7	88	ge7_ppp	ge7	GE7_PPPoE_ACCOUNT
14	Page	1 of 1 > > Show 50	▼ items	Displaying 1 - 7 of 7

Figure 82 Configuration > Network > Interface > PPP

Each field is described in the table below.

LABEL	DESCRIPTION
User Configuration / System Default	The ZyWALL comes with the (non-removable) System Default PPP interfaces pre- configured. You can create (and delete) User Configuration PPP interfaces.
Add	Click this to create a new user-configured PPP interface.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove a user-configured PPP interface, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Connect	To connect an interface, select it and click Connect . You might use this in testing the interface or to manually establish the connection for a Dial-on-Demand PPPoE/PPTP interface.
Disconnect	To disconnect an interface, select it and click Disconnect . You might use this in testing the interface.
Object References	Select an entry and click Object Reference to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with any interface.
Status	The activate (light bulb) icon is lit when the entry is active and dimmed when the entry is inactive.
	The connect icon is lit when the interface is connected and dimmed when it is disconnected.
Name	This field displays the name of the interface.
Base Interface	This field displays the interface on the top of which the PPPoE/PPTP interface is.
Account Profile	This field displays the ISP account used by this PPPoE/PPTP interface.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 45	Configuration	>	Network >	Interface	>	PPP
	configuration	_	NCLWOIK /	michace	/	

7.4.2 PPP Interface Add or Edit

Note: You have to set up an ISP account before you create a PPPoE/PPTP interface.

This screen lets you configure a PPPoE or PPTP interface. If you enabled IPv6 in the **Configuration** > **System** > **IPv6** screen, you can also configure PPP interfaces used for your IPv6 networks on this screen. To access this screen, click the **Add** icon or an **Edit** icon in the PPP Interface screen.

eneral Settings		
Enable Interface		
eneral IPv6 Setting		
Enable IPv6		
iterface Properties		
Interface Name:	0	
Base Interface:	wan1	
Zone:	none 🗸 🚺	
Description:	(Optional)	
onnectivity		
Nailed-Up		
Dial-on-Demand		
SP Setting		
Account Profile:	×	
P Address Assignment	Nuclear Sector Sec	
Get Automatically	0.0.0.0	
USE FIXED IP Address		
Gatemay:	(Options))	
Metric:	(optional)	
	 (0,13) 	
Pv6 Address Assignment		
💟 Enable Stateless Address Auto-configura	ation (SLAAC)	
Metric:	(0-15)	
Address from DHCPv6 Prefix Delegation	S Add EE Edit T Remove	
	# Delegated Prefix Suffix Address Address	
	M G Dana 1 of 1 N Show 50 why No data to display	
HCPv6 Setting		
HCPv6 Setting DHCPv6:	Clent V	
HCPv6 Setting DHCPv6: DUID:	Client 💌	
HCPv6 Setting DHCPv6: DUID: IV DUID as MAC	Client 💌	
HCPv6 Setting DHCPv6: DUID: IV DUID as MAC Customized DUID:	Clent V	
HCPv6 Setting DHCPv6: DUID: V DUID as MAC Customized DUID: Enable Rapid Commit Reput Address	Clent V	
HCPv6 Setting DHCPv6: DUID: V DUID as MAC Customized DUID: Request Address DHCPv6 Request Options		
HCPv6 Setting DHCPv6: DUID: V DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options	Clent	
HCPv6 Setting DHCPv6: DUID: UID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options	Clent	
HCPv6 Setting DHCPv6: DUID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options	Clent Clent Add **********************************	
HCPv6 Setting DHCPv6: DUID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options tterface Parameters	Clent	
HCPv6 Setting DHCPv6: DUID: UDID: Enable Rapid Commit Request Address DHCPv6 Request Options terface Parameters Egress Bandwidth:	Client	
HCPv6 Setting DHCPv6: DUID: UDID: DUID as MAC Customized DUID: Customized DUID: Request Address DHCPv6 Request Options terface Parameters Egress Bandwidth: Ingress Bandwidth:	Client	
HCPv6 Setting DHCPv6: DUID: UDID as MAC Customized DUID: Peable Rapid Commit Request Address DHCPv6 Request Options terface Parameters Egress Bandwidth: Ingress Bandwidth: Ingress Bandwidth: MTU:	Client	
HCPv6 Setting DHCPv6: DUID: UDID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options DHCPv6 Request Options	Client	
HCPv6 Setting DHCPv6: DUID: UDID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options DHCPv6 Request Options Egress Bandwidth: Ingress Bandwidth: MTU: Dencetivity Check Enable Connectivity Check	Client	
HCPv6 Setting DHCPv6: DUID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options CHCPv6 Request Options Egress Bandwidth: Ingress Bandwidth: MTU: CHCPv6 Request Chcck Chcck Method:	Clent	
HCPv6 Setting DHCPv6: DUID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options ChCPv6 Request Options Egress Bandwidth: Ingress Bandwidth: Ingress Bandwidth: MTU: Dennectivity Check Check Method: Check Veriod: Check Period:	Clent	
HCPv6 Setting DHCPv6: DUID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options DHCPv6 Request Options Egress Bandwidth: Ingress	Clent	
HCPv6 Setting DHCPv6: DUID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options DHCPv6 Request Options Egress Bandwidth: Ingress Bandwidth: Ingress Bandwidth: Ingress Bandwidth: Egress Bandwidth: Check Method: Check Method: Check Timeout: Check Fail Tolerance:	Clent Clent Clent Add Image: Clent Image: C	
HCPv6 Setting DHCPv6: DUID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options DHCPv6 Request Options Egress Bandwidth: Ingress Bandwidth: Ingress Bandwidth: MTU: Denectivity Check Check Method: Check Period: Check Fail Tolerance: Check Fail Tolerance: Check Fail Tolerance: Check Chefault Gateway	Clent Clent Clent Add Image: Comparison of the second	
HCPv6 Setting DHCPv6: DUID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options tetrface Parameters Egress Bandwidth: Ingress Bandwidth: Ingress Bandwidth: Ingress Bandwidth: Ingress Bandwidth: Check Nethod: Check Method: Check Fail Tolerance: Check Fail Tolerance: Check Fail Tolerance: Check Inseutt: Check Fail Tolerance: Check Default Gateway Check this address	Clent Clent Clent Add	
HCPv6 Setting DHCPv6: DUID: DUID as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options terface Parameters Egress Bandwidth: Ingress Ban	Clent Clent <t< td=""><td></td></t<>	
HCPv6 Setting DHCPv6: DUID: UDUD as MAC Customized DUID: Enable Rapid Commit Request Address DHCPv6 Request Options terface Parameters Egress Bandwidth: Ingress Ban	Clent Clent <t< td=""><td></td></t<>	

Figure 83 Configuration > Network > Interface > PPP > Add

Each field is explained in the following table.

LABEL	DESCRIPTION
IPv4/IPv6 View / IPv4 View / IPv6 View	Use this button to display both IPv4 and IPv6, IPv4-only, or IPv6-only configuration fields.
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
Create New Object	Click this button to create an ISP Account or a DHCPv6 request object that you may use for the ISP or DHCPv6 settings in this screen.
General Settings	
Enable Interface	Select this to enable this interface. Clear this to disable this interface.
General IPv6 Setting	
Enable IPv6	Select this to enable IPv6 on this interface. Otherwise, clear this to disable it.
Interface Properties	
Interface Name	Specify a name for the interface. It can use alphanumeric characters, hyphens, and underscores, and it can be up to 11 characters long.
Base Interface	Select the interface upon which this PPP interface is built.
	Note: Multiple PPP interfaces can use the same base interface.
Zone	Select the zone to which this PPP interface belongs. The zone determines the security settings the ZyWALL uses for the interface.
Description	Enter a description of this interface. It is not used elsewhere. You can use alphanumeric and () +/:=?!*#@ $_{e}$ - characters, and it can be up to 60 characters long.
Connectivity	
Nailed-Up	Select this if the PPPoE/PPTP connection should always be up. Clear this to have the ZyWALL establish the PPPoE/PPTP connection only when there is traffic. You might use this option if a lot of traffic needs to go through the interface or it does not cost extra to keep the connection up all the time.
Dial-on-Demand	Select this to have the ZyWALL establish the PPPoE/PPTP connection only when there is traffic. You might use this option if there is little traffic through the interface or if it costs money to keep the connection available.
ISP Setting	
Account Profile	Select the ISP account that this PPPoE/PPTP interface uses. The drop-down box lists ISP accounts by name. Use Create new Object if you need to configure a new ISP account (see Chapter 34 on page 419 for details).
Protocol	This field is read-only. It displays the protocol specified in the ISP account.
User Name	This field is read-only. It displays the user name for the ISP account.
Service Name	This field is read-only. It displays the PPPoE service name specified in the ISP account. This field is blank if the ISP account uses PPTP.
IP Address Assignment	Click Show Advanced Settings to display more settings. Click Hide Advanced Settings to display fewer settings.
Get Automatically	Select this if this interface is a DHCP client. In this case, the DHCP server configures the IP address automatically. The subnet mask and gateway are always defined automatically in PPPoE/PPTP interfaces.
Use Fixed IP Address	Select this if you want to specify the IP address manually.

Table 46Configuration > Network > Interface > PPP > Add

LABEL	DESCRIPTION			
IP Address	This field is enabled if you select Use Fixed IP Address.			
	Enter the IP address for this interface.			
Metric	Enter the priority of the gateway (the ISP) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.			
IPv6 Address Assignment	These IP address fields configure an IPv6 IP address on the interface itself.			
Enable Stateless Address Auto- configuration (SLAAC)	Select this to enable IPv6 stateless auto-configuration on this interface. The interface will generate an IPv6 IP address itself from a prefix obtained from an IPv6 router in the network.			
Metric	Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.			
Address from DHCPv6 Prefix Delegation	Use this table to have the ZyWALL obtain an IPv6 prefix from the ISP or a connected uplink router for an internal network, such as the LAN or DMZ. You have to also enter a suffix address which is appended to the delegated prefix to form an address for this interface. See Prefix Delegation on page 107 for more information.			
	To use prefix delegation, you must:			
	Create at least one DHCPv6 request object before configuring this table.			
	 The external interface must be a DHCPv6 client. You must configure the DHCPv6 request options using a DHCPv6 request object with the type of prefix-delegation 			
	 Assign the prefix delegation to an internal interface and enable router advertisement on that interface. 			
Add	Click this to create an entry.			
Edit	Select an entry and click this to change the settings.			
Remove	Select an entry and click this to delete it from this table.			
#	This field is a sequential value, and it is not associated with any entry.			
Delegated Prefix	Select the DHCPv6 request object to use from the drop-down list.			
Suffix Address	Enter the ending part of the IPv6 address, a slash (/), and the prefix length. The ZyWALL will append it to the delegated prefix.			
	For example, you got a delegated prefix of 2003:1234:5678/48. You want to configure an IP address of 2003:1234:5678:1111::1/128 for this interface, then enter ::1111:0:0:0::1/128 in this field.			
Address	This field displays the combined IPv6 IP address for this interface.			
	Note: This field displays the combined address after you click OK and reopen this screen.			
DHCPv6 Setting				
DHCPv6	Select Client to obtain an IP address and DNS information from the service provider for the interface. Otherwise, select N/A to diable the function.			
DUID	This field displays the DHCP Unique IDentifier (DUID) of the interface, which is unique and used for identification purposes when the interface is exchanging DHCPv6 messages with others. See DHCPv6 on page 107 for more information.			
DUID as MAC	Select this if you want the DUID is generated from the interface's default MAC address.			
Customized DUID	If you want to use a customized DUID, enter it here for the interface.			

Table 46 Configuration > Network > Interface > PPP > Add (continued)

LABEL	DESCRIPTION			
Enable Rapid Commit	Select this to shorten the DHCPv6 message exchange process from four to two steps. This function helps reduce heavy network traffic load.			
	Note: Make sure you also enable this option in the DHCPv6 clients to make rapid commit work.			
Request Address	Select this to get an IPv6 IP address for this interface from the DHCP server. Clear this to not get any IP address information through DHCPv6.			
DHCPv6 Request Options	Use this section to configure DHCPv6 request settings that determine what additional information to get from the DHCPv6 server.			
Add	Click this to create an entry in this table. See Section 7.3.4 on page 124 for more information.			
Remove	Select an entry and click this to delete it from this table.			
Object Reference	Select an entry and click Object Reference to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.			
#	This field is a sequential value, and it is not associated with any entry.			
Name	This field displays the name of the DHCPv6 request object.			
Туре	This field displays the type of the object.			
Value	This field displays the IPv6 prefix that the ZyWALL will advertise to its clients.			
Interface Parameters				
Egress Bandwidth	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can send through the interface to the network. Allowed values are 0 - 1048576.			
Ingress Bandwidth	This is reserved for future use.			
Banawidin	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can receive from the network through the interface. Allowed values are 0 - 1048576.			
MTU	Maximum Transmission Unit. Type the maximum size of each data packet, in bytes, that can move through this interface. If a larger packet arrives, the ZyWALL divides it into smaller fragments. Allowed values are 576 - 1492. Usually, this value is 1492.			
Connectivity Check	The interface can regularly check the connection to the gateway you specified to make sure it is still available. You specify how often the interface checks the connection, how long to wait for a response before the attempt is a failure, and how many consecutive failures are required before the ZyWALL stops routing to the gateway. The ZyWALL resumes routing to the gateway the first time the gateway passes the connectivity check.			
Enable Connectivity Check	Select this to turn on the connection check.			
Check Method	Select the method that the gateway allows.			
	Select icmp to have the ZyWALL regularly ping the gateway you specify to make sure it is still available.			
	Select tcp to have the ZyWALL regularly perform a TCP handshake with the gateway you specify to make sure it is still available.			
Check Period	Enter the number of seconds between connection check attempts.			
Check Timeout	Enter the number of seconds to wait for a response before the attempt is a failure.			
Check Fail Tolerance	Enter the number of consecutive failures before the ZyWALL stops routing through the gateway.			
Check Default Gateway	Select this to use the default gateway for the connectivity check.			

 Table 46
 Configuration > Network > Interface > PPP > Add (continued)

LABEL	DESCRIPTION
Check this address	Select this to specify a domain name or IP address for the connectivity check. Enter that domain name or IP address in the field next to it.
Check Port	This field only displays when you set the Check Method to tcp . Specify the port number to use for a TCP connectivity check.
Related Setting	
Configure WAN TRUNK	Click WAN TRUNK to go to a screen where you can configure the interface as part of a WAN trunk for load balancing.
Policy Route	Click Policy Route to go to the screen where you can manually configure a policy route to associate traffic with this interface.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 46 Configuration > Network > Interface > PPP > Add (continued)

7.5 Cellular Configuration Screen (3G)

3G (Third Generation) is a digital, packet-switched wireless technology. Bandwidth usage is optimized as multiple users share the same channel and bandwidth is only allocated to users when they send data. It allows fast transfer of voice and non-voice data and provides broadband Internet access to mobile devices.

Note: The actual data rate you obtain varies depending on the 3G device you use, the signal strength to the service provider's base station, and so on.

You can configure how the ZyWALL's 3G device connects to a network (refer to Section 7.5.1 on page 134):

- You can set the 3G device to connect only to the home network, which is the network to which you are originally subscribed.
- You can set the 3G device to connect to other networks if the signal strength of the home network is too low or it is unavailable.

Aside from selecting the 3G network, the 3G card may also select an available 2.5G or 2.75G network automatically. See the following table for a comparison between 2G, 2.5G, 2.75G and 3G of wireless technologies.

	TYPE	MOBILE PHONE AND DATA STANDARDS			
NAWE		GSM-BASED	CDMA-BASED	SPEED	
2G	Circuit- switched	GSM (Global System for Mobile Communications), Personal Handy- phone System (PHS), etc.	Interim Standard 95 (IS-95), the first CDMA- based digital cellular standard pioneered by Qualcomm. The brand name for IS-95 is cdmaOne. IS-95 is also known as TIA-EIA-95.	Slow	
2.5G	Packet- switched	GPRS (General Packet Radio Services), High-Speed Circuit-Switched Data (HSCSD), etc.	CDMA2000 is a hybrid 2.5G / 3G protocol of mobile telecommunications standards that use CDMA, a multiple access scheme for digital radio.		
2.75G	Packet- switched	Enhanced Data rates for GSM Evolution (EDGE), Enhanced GPRS (EGPRS), etc.	CDMA2000 1xRTT (1 times Radio Transmission Technology) is the core CDMA2000 wireless air interface standard. It is also known as 1x, 1xRTT, or IS-2000 and considered to be a 2.5G or 2.75G technology.		
3G	Packet- switched	UMTS (Universal Mobile Telecommunications System), a third- generation (3G) wireless standard defined in ITU ^A specification, is sometimes marketed as 3GSM. The UMTS uses GSM infrastructures and W- CDMA (Wideband Code Division Multiple Access) as the air interface.	CDMA2000 EV-DO (Evolution-Data Optimized, originally 1x Evolution-Data Only), also referred to as EV-DO, EVDO, or just EV, is an evolution of CDMA2000 1xRTT and enables high-speed wireless connectivity. It is also denoted as IS-856 or High Data Rate (HDR).		
3.5G	Packet- switched	HSDPA (High-Speed Downlink Packet Access) is a mobile telephony protocol, used for UMTS-based 3G networks and allows for higher data transfer speeds.		Faet	
				Fast	

 Table 47
 2G, 2.5G, 2.75G, 3G and 3.5G Wireless Technologies

A. The International Telecommunication Union (ITU) is an international organization within which governments and the private sector V coordinate global telecom networks and services.

To change your 3G WAN settings, click Configuration > Network > Interface > Cellular.

Note: Install (or connect) a compatible 3G USB device to use a cellular connection.

Note: The WAN IP addresses of a ZyWALL with multiple WAN interfaces must be on different subnets.

Figure 84 Configuration > Network > Interface > Cellular

ellular Interface	Summary			
📀 Add 📝 Edit	: 👕 Remove 🢡 /	Activate 💡 Inactivat	e 🗟 Connect 🚱 Disconne	ct 🛛 📴 Object Reference
# 🔺 Status 🛛 I	Name	Extension Slot	Connected Device	ISP Settings
1 @	cellular1	USB 1	none	Device Profile 1
4 4 Page 1	of 1 🕨 🕅	Show 50 👻 items		Displaying 1 - 1 of 1

The following table describes the labels in this screen.

LABEL	DESCRIPTION		
Add	Click this to create a new cellular interface.		
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.		
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.		
Activate	To turn on an entry, select it and click Activate.		
Inactivate	To turn off an entry, select it and click Inactivate.		
Connect	To connect an interface, select it and click Connect . You might use this in testing the interface or to manually establish the connection.		
Disconnect	To disconnect an interface, select it and click Disconnect . You might use this in testing the interface.		
Object References	Select an entry and click Object Reference to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.		
#	This field is a sequential value, and it is not associated with any interface.		
Status	The activate (light bulb) icon is lit when the entry is active and dimmed when the entry is inactive.		
	The connect icon is lit when the interface is connected and dimmed when it is disconnected.		
Name	This field displays the name of the interface.		
Extension Slot	This field displays where the entry's cellular card is located.		
Connected Device	This field displays the name of the cellular card.		
ISP Settings	This field displays the profile of ISP settings that this cellular interface is set to use.		
Apply	Click Apply to save your changes back to the ZyWALL.		
Reset	Click Reset to return the screen to its last-saved settings.		

Table 48 Configuration > Network > Interface > Cellular

7.5.1 Cellular Add/Edit Screen

To change your 3G settings, click **Configuration > Network > Interface > Cellular > Add** (or **Edit**). In the pop-up window that displays, select the slot that contains the 3G device, then the following screen displays.

Hide Advanced Settings	
General Settings	
Enable Interface	
Interface Properties	
Interface Name:	cellular 1
Zone:	nne 🗴 🕄
Extension Slot	DC Card 1
Extension bloc	
Connected Device:	none
Description:	(Optional)
Connectivity	
📝 Nailed-Up	
Idle timeout:	0 seconds
ISP Settings	
Profile Selection:	Device Custom
	Profile 1
APN:	n/a
Dial String:	n/a
5IM Card Setting	
PIN Code:	
Retype to Confirm:	
Interface Parameters	
Egress Bandwidth:	1048576 Kbps
Ingress Bandwidth:	1048576 Kbps
MTU:	1492 Bytes
Check Period: Check Timeout: Check Fail Tolerance: Check Default Gateway	30 (5-30 seconds) 5 (1-10 seconds) 5 (1-10) 0.0.0 (200 seconds)
Check this address	(Domain Name or IP Address)
Related Setting	
Configure <u>WAN_TRUNK</u> Configure <u>Policy Route</u>	
P Address	
Get Automatically	0.0.0.0
Use Fixed IP Address	
IP Address Assignment:	
Metric:	0 (0-15)
Device Settings	
Network Selection:	auto 🗸
and the second	
sudget Setup	
Enable Budget Control	
ime suaget:	urs per month
Data Budget:	bytes Download/upload
Reset time and data budget counters	on: Last 🔽 day of each month
Actions when over budget	
Log: None 🗸	
New 3G connection:	Allow
Current 3G connection:	Keep 🗸
Actions when over 0 % of	time budget or 0 % of data budget
Log: None 🗸	

Collul ۷ЧЧ • • -

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
General Settings	
Enable Interface	Select this option to turn on this interface.
Interface Properties	
Interface Name	Select a name for the interface.
Zone	Select the zone to which you want the cellular interface to belong. The zone determines the security settings the ZyWALL uses for the interface.
Extension Slot	This is the USB slot that you are configuring for use with a 3G card.
Connected Device	This displays the manufacturer and model name of your 3G card if you inserted one in the ZyWALL. Otherwise, it displays none .
Description	Enter a description of this interface. It is not used elsewhere. You can use alphanumeric and () +/:=?!*#@\$_%- characters, and it can be up to 60 characters long.
Connectivity	
Nailed-Up	Select this if the connection should always be up. Clear this to have the ZyWALL to establish the connection only when there is traffic. You might not nail up the connection if there is little traffic through the interface or if it costs money to keep the connection available.
Idle timeout	This value specifies the time in seconds ($0 \sim 360$) that elapses before the ZyWALL automatically disconnects from the ISP's server. Zero disables the idle timeout.
ISP Settings	
Profile Selection	Select Device to use one of the 3G device's profiles of device settings. Then select the profile (use Profile 1 unless your ISP instructed you to do otherwise).
	Select Custom to configure your device settings yourself.
APN	This field is read-only if you selected Device in the profile selection. Select Custom in the profile selection to be able to manually input the APN (Access Point Name) provided by your service provider. This field applies with a GSM or HSDPA 3G card. Enter the APN from your service provider. Connections with different APNs may provide different services (such as Internet access or MMS (Multi-Media Messaging Service)) and charge method.
	You can enter up to 63 ASCII printable characters. Spaces are allowed.
Dial String	Enter the dial string if your ISP provides a string, which would include the APN, to initialize the 3G card.
	You can enter up to 63 ASCII printable characters. Spaces are allowed.
	This field is available only when you insert a GSM 3G card.
Authentication Type	The ZyWALL supports PAP (Password Authentication Protocol) and CHAP (Challenge Handshake Authentication Protocol). CHAP is more secure than PAP; however, PAP is readily available on more platforms.
	Use the drop-down list box to select an authentication protocol for outgoing calls. Options are:
	None: No authentication for outgoing calls.
	CHAP - Your ZyWALL accepts CHAP requests only.
	PAP - Your ZyWALL accepts PAP requests only.

 Table 49
 Configuration > Network > Interface > Cellular > Add

LABEL	DESCRIPTION
User Name	This field displays when you select an authentication type other than None . This field is read-only if you selected Device in the profile selection. If this field is configurable, enter the user name for this 3G card exactly as the service provider gave it to you.
	You can use 1 ~ 64 alphanumeric and $#:\$@$./$ characters. The first character must be alphanumeric or $@$./$. Spaces are not allowed.
Password	This field displays when you select an authentication type other than None . This field is read-only if you selected Device in the profile selection and the password is included in the 3G card's profile. If this field is configurable, enter the password for this SIM card exactly as the service provider gave it to you.
	You can use 0 ~ 63 alphanumeric and `~!@#\$%^&*()+={};:'<,>./ characters. Spaces are not allowed.
Retype to Confirm	This field displays when you select an authentication type other than None . This field is read-only if you selected Device in the profile selection and the password is included in the 3G card's profile. If this field is configurable, re-enter the password for this SIM card exactly as the service provider gave it to you.
SIM Card Setting	
PIN Code	This field displays with a GSM or HSDPA 3G card. A PIN (Personal Identification Number) code is a key to a 3G card. Without the PIN code, you cannot use the 3G card.
	Enter the 4-digit PIN code (0000 for example) provided by your ISP. If you enter the PIN code incorrectly, the 3G card may be blocked by your ISP and you cannot use the account to access the Internet.
	If your ISP disabled PIN code authentication, enter an arbitrary number.
Retype to Confirm	Type the PIN code again to confirm it.
Interface Parameters	
Egress Bandwidth	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can send through the interface to the network. Allowed values are 0 - 1048576. This setting is used in WAN load balancing and bandwidth management.
Ingress Bandwidth	This is reserved for future use.
Danawidin	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can receive from the network through the interface. Allowed values are 0 - 1048576.
MTU	Maximum Transmission Unit. Type the maximum size of each data packet, in bytes, that can move through this interface. If a larger packet arrives, the ZyWALL divides it into smaller fragments. Allowed values are 576 - 1492. Usually, this value is 1492.
Connectivity Check	The interface can regularly check the connection to the gateway you specified to make sure it is still available. You specify how often the interface checks the connection, how long to wait for a response before the attempt is a failure, and how many consecutive failures are required before the ZyWALL stops routing to the gateway. The ZyWALL resumes routing to the gateway the first time the gateway passes the connectivity check.
Enable Connectivity Check	Select this to turn on the connection check.
Check Method	Select the method that the gateway allows.
	Select icmp to have the ZyWALL regularly ping the gateway you specify to make sure it is still available.
	Select tcp to have the ZyWALL regularly perform a TCP handshake with the gateway you specify to make sure it is still available.

Table 49 Configuration > Network > Interface > Cellular > Add (continued)

LABEL	DESCRIPTION
Check Period	Enter the number of seconds between connection check attempts.
Check Timeout	Enter the number of seconds to wait for a response before the attempt is a failure.
Check Fail Tolerance	Enter the number of consecutive failures before the ZyWALL stops routing through the gateway.
Check Default Gateway	Select this to use the default gateway for the connectivity check.
Check this address	Select this to specify a domain name or IP address for the connectivity check. Enter that domain name or IP address in the field next to it.
Check Port	This field only displays when you set the Check Method to tcp . Specify the port number to use for a TCP connectivity check.
Related Setting	
Configure WAN TRUNK	Click WAN TRUNK to go to a screen where you can configure the interface as part of a WAN trunk for load balancing.
Configure Policy Route	Click Policy Route to go to the policy route summary screen where you can configure a policy route to override the default routing and SNAT behavior for the interface.
IP Address Assignment	
Get Automatically	Select this option If your ISP did not assign you a fixed IP address. This is the default selection.
Use Fixed IP Address	Select this option If the ISP assigned a fixed IP address.
IP Address Assignment	Enter the cellular interface's WAN IP address in this field if you selected Use Fixed IP Address.
Metric	Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.
Device Settings	
Network Selection	This field appears if you selected a 3G device that allows you to select the type of network to use. Select the type of 3G service for your 3G connection. If you are unsure what to select, check with your 3G service provider to find the 3G service available to you in your region.
	Select auto to have the card connect to an available network. Choose this option if you do not know what networks are available.
	You may want to manually specify the type of network to use if you are charged differently for different types of network or you only have one type of network available to you.
	Select GPRS / EDGE (GSM) only to have this interface only use a 2.5G or 2.75G network (respectively). If you only have a GSM network available to you, you may want to select this so the ZyWALL does not spend time looking for a WCDMA network.
	Select UMTS / HSDPA (WCDMA) only to have this interface only use a 3G or 3.5G network (respectively). You may want to do this if you want to make sure the interface does not use the GSM network.

Table 49	Configuration >	Network >	Interface >	Cellular	> Add	(continued)	
----------	-----------------	-----------	-------------	----------	-------	-------------	--

LABEL	DESCRIPTION
Network	Home network is the network to which you are originally subscribed.
Selection	Select Home to have the 3G device connect only to the home network. If the home network is down, the ZyWALL's 3G Internet connection is also unavailable.
	Select Auto (Default) to allow the 3G device to connect to a network to which you are not subscribed when necessary, for example when the home network is down or another 3G base station's signal is stronger. This is recommended if you need continuous Internet connectivity. If you select this, you may be charged using the rate of a different network.
Budget Setup	
Enable Budget Control	Select this to set a monthly limit for the user account of the installed 3G card. You can set a limit on the total traffic and/or call time. The ZyWALL takes the actions you specified when a limit is exceeded during the month.
Time Budget	Select this and specify the amount of time (in hours) that the 3G connection can be used within one month. If you change the value after you configure and enable budget control, the ZyWALL resets the statistics.
Data Budget	Select this and specify how much downstream and/or upstream data (in Mega bytes) can be transmitted via the 3G connection within one month.
	Select Download to set a limit on the downstream traffic (from the ISP to the ZyWALL).
	Select Upload to set a limit on the upstream traffic (from the ZyWALL to the ISP).
	Select Download/Upload to set a limit on the total traffic in both directions.
	If you change the value after you configure and enable budget control, the ZyWALL resets the statistics.
Reset time and data budget counters on	Select the date on which the ZyWALL resets the budget every month. If the date you selected is not available in a month, such as 30th or 31st, the ZyWALL resets the budget on the last day of the month.
Reset time and	This button is available only when you enable budget control in this screen.
counters	Click this button to reset the time and data budgets immediately. The count starts over with the 3G connection's full configured monthly time and data budgets. This does not affect the normal monthly budget restart; so if you configured the time and data budget counters to reset on the second day of the month and you use this button on the first, the time and data budget counters will still reset on the second.
Actions when over budget	Specify the actions the ZyWALL takes when the time or data limit is exceeded.
Log	Select None to not create a log, Log to create a log, or Log-alert to create an alert log. If you select Log or Log-alert you can also select recurring every to have the ZyWALL send a log or alert for this event periodically. Specify how often (from 1 to 65535 minutes) to send the log or alert.
New 3G connection	Select Allow to permit new 3G connections or Disallow to drop/block new 3G connections.
Current 3G connection	Select Keep to maintain an existing 3G connection or Drop to disconnect it. You cannot set New 3G connection to Allow and Current 3G connection to Drop at the same time.
	If you set New 3G connection to Disallow and Current 3G connection to Keep , the ZyWALL allows you to transmit data using the current connection, but you cannot build a new connection if the existing connection is disconnected.
Actions when over % of time budget or % of data budget	Specify the actions the ZyWALL takes when the specified percentage of time budget or data limit is exceeded. Enter a number from 1 to 99 in the percentage fields. If you change the value after you configure and enable budget control, the ZyWALL resets the statistics.

Table 49 Configuration > Network > Interface > Cellular > Add (continued)

LABEL	DESCRIPTION
Log	Select None to not create a log when the ZyWALL takes this action, Log to create a log, or Log-alert to create an alert log. If you select Log or Log-alert you can also select recurring every to have the ZyWALL send a log or alert for this event periodically. Specify how often (from 1 to 65535 minutes) to send the log or alert.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

 Table 49
 Configuration > Network > Interface > Cellular > Add (continued)

7.6 Tunnel Interfaces

The ZyWALL uses tunnel interfaces in Generic Routing Encapsulation (GRE), IPv6 in IPv4, and 6to4 tunnels.

GRE Tunneling

GRE tunnels encapsulate a wide variety of network layer protocol packet types inside IP tunnels. A GRE tunnel serves as a virtual point-to-point link between the ZyWALL and another router over an IPv4 network. At the time of writing, the ZyWALL only supports GRE tunneling in IPv4 networks.



Figure 86 GRE Tunnel Example

IPv6 Over IPv4 Tunnels

To route traffic between two IPv6 networks over an IPv4 network, an IPv6 over IPv4 tunnel has to be used.





On the ZyWALL, you can either set up a manual IPv6-in-IPv4 tunnel or an automatic 6to4 tunnel. The following describes each method:

IPv6-in-IPv4 Tunneling

Use this mode on the WAN of the ZyWALL if

• your ZyWALL has a public IPv4 IP address given from your ISP,

and

 you want to transmit your IPv6 packets to one and only one remote site whose LAN network is also an IPv6 network.

With this mode, the ZyWALL encapsulates IPv6 packets within IPv4 packets across the Internet. You must know the WAN IP address of the remote gateway device. This mode is normally used for a site-to-site application such as two branch offices.

Figure 88 IPv6-in-IPv4 Tunnel



In the ZyWALL, you must also manually configure a policy route for an IPv6-in-IPv4 tunnel to make the tunnel work.

6to4 Tunneling

This mode also enables IPv6 packets to cross IPv4 networks. Unlike IPv6-in-IPv4 tunneling, you do not need to configure a policy route for a 6to4 tunnel. Through your properly pre-configuring the destination router's IP address in the IP address assignments to hosts, the ZyWALL can automatically forward 6to4 packets to the destination they want to go. A 6to4 relay router is required to route 6to4 packets to a native IPv6 network if the packet's destination do not match your specified criteria.

In this mode, the ZyWALL should get a public IPv4 address for the WAN. The ZyWALL adds an IPv4 IP header to an IPv6 packet when transmitting the packet to the Internet. In reverse, the ZyWALL removes the IPv4 header from an IPv6 packet when receiving it from the Internet.

An IPv6 address using the 6to4 mode consists of an IPv4 address, the format is as the following:

2002: [a public IPv4 address in hexadecimal]::/48

For example,

A public IPv4 address is 202.156.30.41. The converted hexadecimal IP string is ca.9c.1E.29. The IPv6 address prefix becomes 2002:ca9c:1e29::/48.



7.6.1 Configuring a Tunnel

This screen lists the ZyWALL's configured tunnel interfaces. To access this screen, click **Network** > **Interface** > **Tunnel**.

Figure 90	Network	>	Interface	>	Tunnel

🗿 Ad	d 📝 Edit 👖	Remove 😡 Activate	💡 Inactivate 📴 Object	: Reference	
# :	Stat Name	IP Address	Tunnel Mode	My Address	Remote Gatway Addres
	💡 tunnel1	STATIC ::2/128	6to4 (2002::/16)	wan1 (0.0.0.0)	Automatic
	1090				

Each field is explained in the following table.

Table 50 Network > Interface > To	unnel
-----------------------------------	-------

LABEL	DESCRIPTION
Add	Click this to create a new GRE tunnel interface.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Object References	Select an entry and click Object Reference to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with any interface.

LABEL	DESCRIPTION
Status	The activate (light bulb) icon is lit when the entry is active and dimmed when the entry is inactive.
Name	This field displays the name of the interface.
IP Address	This is the IP address of the interface. If the interface is active (and connected), the ZyWALL tunnels local traffic sent to this IP address to the Remote Gateway Address .
Tunnel Mode	This is the tunnel mode of the interface (GRE , IPv6-in-IPv4 or 6to4). This field also displays the interface's IPv4 IP address and subnet mask if it is a GRE tunnel. Otherwise, it displays the interface's IPv6 IP address and prefix length.
My Address	This is the interface or IP address uses to identify itself to the remote gateway. The ZyWALL uses this as the source for the packets it tunnels to the remote gateway.
Remote Gateway Address	This is the IP address or domain name of the remote gateway to which this interface tunnels traffic.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to begin configuring this screen afresh.

Table 50Network > Interface > Tunnel (continued)

7.6.2 Tunnel Add or Edit Screen

This screen lets you configure a tunnel interface. Click **Configuration > Network > Interface > Tunnel > Add** (or **Edit**) to open the following screen.

Hide Advanced Settings					
General Settings					
📝 Enable					
Interface Properties					
Interface Name:	tunnel1				
Zone:	TUNNEL	▼ 1			
Tunnel Mode:	6to4	~			
IPv6 Address Assignment					
IPv6 Address/Prefix Length:	::2/128			(Optional)	
Metric:	2 (0-15)			211	
6to4 Tunnel Parameter					
6to4 Prefix:	2002::/16				
6to4 Prefix: Relay Router: 🏹 NDTE: traffic destinated to the n	2002::/16 192.88.99.1 on-6to4 prefix domain tunn	els to the relay router		(Optional)	
6to4 Prefix: Relay Router: NDTE: traffic destinated to the n Remote Gateway Prefix:	2002::/16 192.88.99.1 on-6to4 prefix domain tunn	iels to the relay router		(Optional)	
6to4 Prefix: Relay Router: NOTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings	2002::/16 192.88.99.1 on-6to4 prefix domain tunn	sels to the relay router		(Optional)	
6to4 Prefix: Relay Router: NDTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings My Address	2002::/16 192.88.99.1 on-6to4 prefix domain tunn	iels to the relay router		(Optional)	
6to4 Prefix: Relay Router: NOTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings My Address (a) Interface	2002::/16 192.88.99.1 on-6to4 prefix domain tunn wan1	sels to the relay router	client 0.0.0.0/0.0.0.0	(Optional)	
6to4 Prefix: Relay Router: NOTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings My Address Interface IP Address	2002::/16 192.88.99.1 on-6to4 prefix domain tunn wan1 0,0.0.0	vels to the relay router	client 0.0.0.0/0.0.0.0	(Optional)	
6to4 Prefix: Relay Router: NOTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings My Address Interface I P Address Remote Gatway Address:	2002::/16 192.88,99.1 on-6to4 prefix domain tunn wan1 0.0.0.0 Automatic	nels to the relay router	client 0.0.0.0/0.0.0.0	(Optional)	
6to4 Prefix: Relay Router: NOTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings My Address My Address Interface IP Address Remote Gatway Address: Interface Parameters	2002::/16 192.88.99.1 on-6to4 prefix domain tunn wan1 0.0.0.0 Automatic	vels to the relay router	client 0.0.0.0/0.0.0.0	(Optional)	
6to4 Prefix: Relay Router: NOTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings My Address My Address My Address P Address Remote Gatway Address: Interface Parameters Egress Bandwidth:	2002::/16 192.88.99.1 on-6to4 prefix domain tunn wan1 0.0.0.0 Automatic	Nels to the relay router	client 0.0.0.0/0.0.0.0	(Optional)	
6to4 Prefix: Relay Router: NOTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings My Address Interface Interface P Address Remote Gatway Address: Interface Parameters Egress Bandwidth: Ingress Bandwidth:	2002::/16 192.88.99.1 on-6to4 prefix domain tunn wan1 0.0.00 Automatic 1048576 1048576	bels to the relay router DHCP Kbps Kbps Kbps	client 0.0.0.0/0.0.0.0	(Optional)	
6to4 Prefix: Relay Router: NUTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings My Address My Address Interface IP Address Remote Gatway Address: Interface Parameters Egress Bandwidth: Ingress Bandwidth: MTU:	2002::/16 192.88.99.1 on-6to4 prefix domain tunn wan1 0,0.0.0 Automatic 1048576 1048576 1480	Kbps Kbps Bytes	client 0.0.0.0/0.0.0.0	(Optional)	
6to4 Prefix: Relay Router: NOTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings My Address My Address My Address P Address Remote Gatway Address: Interface Parameters Egress Bandwidth: Ingress Bandwidth: MTU: Related Setting	2002::/16 192.88,99.1 on-6to4 prefix domain tunn wan1 0.0.0.0 Automatic 1048576 1048576 1480	Kbps Kbps Bytes	client 0.0.0,0/0.0.0.0	(Optional)	
6to4 Prefix: Relay Router: NOTE: traffic destinated to the n Remote Gateway Prefix: Gateway Settings My Address Interface Interface P Address Remote Gatway Address: Interface Parameters Egress Bandwidth: Ingress Bandwidth: MTU: Related Setting Configure WAN_TRUNK	2002::/16 192.88.99.1 on-6to4 prefix domain tunn wan1 0.0.0.0 Automatic 1048576 1048576 1480	Kbps Kbps Bytes	client 0.0.0.0/0.0.0.0	(Optional)	

Figure 91 Network > Interface > Tunnel > Add/Edit

Each field is explained in the following table.

Table 51	Network >	Interface	> Tunnel	> Add/Edit

LABEL	DESCRIPTION
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
General Settings	
Enable	Select this to enable this interface. Clear this to disable this interface.
Interface Properties	
Interface Name	This field is read-only if you are editing an existing tunnel interface. Enter the name of the tunnel interface. The format is tunnel x , where x is 0 - 3. For example, tunnel0.
Zone	Use this field to select the zone to which this interface belongs. This controls what security settings the ZyWALL applies to this interface.
LABEL	DESCRIPTION
--------------------------------	--
Tunnel Mode	Select the tunneling protocol of the interface (GRE , IPv6-in-IPv4 or 6to4). See Section 7.6 on page 140 for more information.
IP Address Assignment	This section is available if you are configuring a GRE tunnel.
IP Address	Enter the IP address for this interface.
Subnet Mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.
Metric	Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.
IPv6 Address Assignment	This section is available if you are configuring an IPv6-in-IPv4 or a 6to4 tunnel.
IPv6 Address/ Prefix Length	Enter the IPv6 address and the prefix length for this interface if you want to use a static IP address. This field is optional.
	The prefix length indicates what the left-most part of the IP address is the same for all computers in the network, that is, the network address.
Metric	Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.
6to4 Tunnel Parameter	This section is available if you are configuring a 6to4 tunnel which encapsulates IPv6 to IPv4 packets.
6to4 Prefix	Enter the IPv6 prefix of a destination network. The ZyWALL forwards IPv6 packets to the hosts in the matched network.
	If you enter a prefix starting with 2002, the ZyWALL will forward the matched packets to the IPv4 IP address converted from the packets' destination address. The IPv4 IP address can be converted from the next 32 bits after the prefix you specified in this field. See 6to4 Tunneling on page 141 for an example. The ZyWALL forwards the unmatched packets to the specified Relay Router .
Relay Router	Enter the IPv4 address of a 6to4 relay router which helps forward packets between 6to4 networks and native IPv6 networks.
Remote Gateway Prefix	Enter the IPv4 network address and network bits of a remote 6to4 gateway, for example, 14.15.0.0/16.
	This field works if you enter a 6to4 Prefix not starting with 2002 (2003 for example). The ZyWALL forwards the matched packets to a remote gateway with the network address you specify here, and the bits converted after the 6to4 Prefix in the packets.
	For example, you configure the 6to4 prefix to 2003:A0B::/32 and the remote gateway prefix to 14.15.0.0/16. If a packet's destination is 2003:A0B:1011:5::8, the ZyWALL forwards the packet to 14.15.16.17, where the network address is 14.15.0.0 and the host address is the remain bits converted from 1011 after the packet's 6to4 prefix (2003:A0B).
Gateway Settings	
My Address	Specify the interface or IP address to use as the source address for the packets this interface tunnels to the remote gateway. The remote gateway sends traffic to this interface or IP address.
Remote Gateway Address	Enter the IP address or domain name of the remote gateway to which this interface tunnels traffic.
	Automatic displays in this field if you are configuring a 6to4 tunnel. It means the 6to4 tunnel will help forward packets to the corresponding remote gateway automatically by looking at the packet's destination address.

Table 51	Network	>	Interface	>	Tunnel	>	Add/Edit	(continued))
----------	---------	---	-----------	---	--------	---	----------	-------------	---

LABEL	DESCRIPTION
Interface Parameters	
Egress Bandwidth	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can send through the interface to the network. Allowed values are 0 - 1048576. This setting is used in WAN load balancing and bandwidth management.
Ingress Bandwidth	This is reserved for future use.
	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can receive from the network through the interface. Allowed values are 0 - 1048576.
MTU	Maximum Transmission Unit. Type the maximum size of each data packet, in bytes, that can move through this interface. If a larger packet arrives, the ZyWALL divides it into smaller fragments. Allowed values are 576 - 1500. Usually, this value is 1500.
Connectivity Check	This section is available if you are configuring a GRE tunnel.
	The interface can regularly check the connection to the gateway you specified to make sure it is still available. You specify how often the interface checks the connection, how long to wait for a response before the attempt is a failure, and how many consecutive failures are required before the ZyWALL stops routing to the gateway. The ZyWALL resumes routing to the gateway the first time the gateway passes the connectivity check.
Enable Connectivity Check	Select this to turn on the connection check.
Check Method	Select the method that the gateway allows.
	Select icmp to have the ZyWALL regularly ping the gateway you specify to make sure it is still available.
	Select tcp to have the ZyWALL regularly perform a TCP handshake with the gateway you specify to make sure it is still available.
Check Period	Enter the number of seconds between connection check attempts.
Check Timeout	Enter the number of seconds to wait for a response before the attempt is a failure.
Check Fail Tolerance	Enter the number of consecutive failures before the ZyWALL stops routing through the gateway.
Check Default Gateway	Select this to use the default gateway for the connectivity check.
Check this address	Select this to specify a domain name or IP address for the connectivity check. Enter that domain name or IP address in the field next to it.
Check Port	This field displays when you set the Check Method to tcp . Specify the port number to use for a TCP connectivity check.
Related Setting	
WAN TRUNK	Click this link to go to a screen where you can configure WAN trunk load balancing.
Policy Route	Click this link to go to the screen where you can manually configure a policy route to associate traffic with this interface.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 51 Network > Interface > Tunnel > Add/Edit (continued)

7.7 VLAN Interfaces

A Virtual Local Area Network (VLAN) divides a physical network into multiple logical networks. The standard is defined in IEEE 802.1q.

Figure 92 Example: Before VLAN



In this example, there are two physical networks and three departments **A**, **B**, and **C**. The physical networks are connected to hubs, and the hubs are connected to the router.

Alternatively, you can divide the physical networks into three VLANs.

Figure 93 Example: After VLAN



Each VLAN is a separate network with separate IP addresses, subnet masks, and gateways. Each VLAN also has a unique identification number (ID). The ID is a 12-bit value that is stored in the MAC header. The VLANs are connected to switches, and the switches are connected to the router. (If one switch has enough connections for the entire network, the network does not need switches **A** and **B**.)

- Traffic inside each VLAN is layer-2 communication (data link layer, MAC addresses). It is handled by the switches. As a result, the new switch is required to handle traffic inside VLAN 2. Traffic is only broadcast inside each VLAN, not each physical network.
- Traffic between VLANs (or between a VLAN and another type of network) is layer-3 communication (network layer, IP addresses). It is handled by the router.

This approach provides a few advantages.

- Increased performance In VLAN 2, the extra switch should route traffic inside the sales department faster than the router does. In addition, broadcasts are limited to smaller, more logical groups of users.
- Higher security If each computer has a separate physical connection to the switch, then broadcast traffic in each VLAN is never sent to computers in another VLAN.
- Better manageability You can align network policies more appropriately for users. For example, you can set different bandwidth limits for each VLAN. These rules are also independent of the physical network, so you can change the physical network without changing policies.

In this example, the new switch handles the following types of traffic:

- Inside VLAN 2.
- Between the router and VLAN 1.
- Between the router and VLAN 2.
- Between the router and VLAN 3.

VLAN Interfaces Overview

In the ZyWALL, each VLAN is called a VLAN interface. As a router, the ZyWALL routes traffic between VLAN interfaces, but it does not route traffic within a VLAN interface. All traffic for each VLAN interface can go through only one Ethernet interface, though each Ethernet interface can have one or more VLAN interfaces.

Note: Each VLAN interface is created on top of only one Ethernet interface.

Otherwise, VLAN interfaces are similar to other interfaces in many ways. They have an IP address, subnet mask, and gateway used to make routing decisions. They restrict bandwidth and packet size. They can provide DHCP services, and they can verify the gateway is available.

7.7.1 VLAN Summary Screen

This screen lists every VLAN interface and virtual interface created on top of VLAN interfaces. If you enabled IPv6 in the **Configuration** > **System** > **IPv6** screen, you can also configure VLAN interfaces used for your IPv6 networks on this screen. To access this screen, click **Configuration** > **Network** > **Interface** > **VLAN**.

#	Status	Name 🔺	Port/VID	IP Address		Mask	
M	Page	1 of 1 👂	Show 50 🗸 iter	ms			No data to displa
v6 C	onfiguratio	n					
1000 10		~ ~	-				
0	Add 📝 Edit	TRemove	Activate <table-cell> Inactivate 🕻</table-cell>	Object Reference			
© / #	Add 📝 Edit Status	Remove O	Activate <table-cell> Inactivate (Port/V</table-cell>	Diject Reference	IP Address		
⊘ ^µ #	Add 📝 Edit Status	Remove O	Activate <table-cell> Inactivate Port/V</table-cell>	Diject Reference	IP Address		

Figure 94 Configuration > Network > Interface > VLAN

Each field is explained in the following table.

LABEL	DESCRIPTION
Configuratio n / IPv6 Configuratio n	Use the Configuration section for IPv4 network settings. Use the IPv6 Configuration section for IPv6 network settings if you connect your ZyWALL to an IPv6 network. Both sections have similar fields as described below.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Create Virtual Interface	To open the screen where you can create a virtual interface, select an interface and click Create Virtual Interface .
Object References	Select an entry and click Object Reference to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with any interface.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Name	This field displays the name of the interface.
Port/VID	For VLAN interfaces, this field displays
	the Ethernet interface on which the VLAN interface is createdthe VLAN ID
	For virtual interfaces, this field is blank.
IP Address	This field displays the current IP address of the interface. If the IP address is 0.0.0.0, the interface does not have an IP address yet.
	This screen also shows whether the IP address is a static IP address (STATIC) or dynamically assigned (DHCP). IP addresses are always static in virtual interfaces.
Mask	This field displays the interface's subnet mask in dot decimal notation.

 Table 52
 Configuration > Network > Interface > VLAN

LABEL	DESCRIPTION
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 52 Configuration > Network > Interface > VLAN (continued)

7.7.2 VLAN Add/Edit

This screen lets you configure IP address assignment, interface bandwidth parameters, DHCP settings, and connectivity check for each VLAN interface. To access this screen, click the **Create Virtual Interface** icon in the **VLAN Summary** screen. The following screen appears.

coursed Settlines	
E Suble Snerbox	
Contract Contract of Contract	
Dude P-4	
interdiction Properties	
Interface Type:	great 2 E
biterface transit	det. 0
Zone:	KANG 🔤 🖪
RANZI	0-000
Description	Diteral
P Address Assignment	
C Get Automatically	
Use fixed IP Address	
Subret Redit	4.040
Galariey)	(Optional)
phater;	(in fi)
Pv6 Address Assignment	
[1] Entrie Statems Address Auto-perfigue	Harr (SLAAC)
Pró AdressPrefs Lergits	Colored
Gateway)	Cpteral
Hete	15-15
Address from DHON& Prefix Delegation	Q AM
	Employed Profer Suffix Address Addres
	If it Page () of () it If Ones 10 with the data to depice
INCPus SetSeg	
picks.	tere (t)
PE OUD as Had	
Calment (1.00)	
Endlik Rapid Commit	
Information Rathmath Times	4294667293 (840-4294667292)
DHOM Laser Options	
	# Tate - Tate Vite
	[1, 1] Page 1 of 1 > 1] Show 10 wire in data to deploy
Pv6 Rauter Advertisement Setting	
E Brable Router Advertagment	
Advertised mails bet fellows Cardigues Advertised mails bet Other Cardigues	Hen Frien DHCPv8
Router Preference	Medium (2)
HTLE	1400 (1200-2000)
Advertised Prefix Table	
	8 Pril Latere Park Length
	14 6 Page 1 of 1 3 31 Show 12 without In data to desire
Advertised Prefix from DHCPv6 Prefix Designation	0m 20 1
Advertised Prefix It un DHCPvG Prefix Delegation	Gade
duh erstand ih efst fran DecDrub Hrefe Deingebox	Quality Importantia > Subscription Approx. H 1 Page 3_1 M1 + H Page 4_1 M1 Approx.
Adversed Prefix free DrcPr-6 Prefix Designed	Qued good good good good good good good go
Adversed hefs fun Drch-d Hefs Despetor Repeties Personeters Epen Benderffic	0 +00 0 +00 +000 + 50% Aldress Aldres Aldres H + Ngg [] = 41 + H = 20m H = 300 H 100 H = 300 H In data to data to data DHEN 0 m 0 m 0 m
Advertised Profile Stars DeCPut Profile Despetition Interface Parameters Egress Restantify Targens Restantify Hitty	Q AM ■ ■ # Imposition 5 * Softwateres Addres H * Registration 5 * Softwateres * Softwateres
All-instand hele fast DecD-d hele December Representations Exprese Benchedite Exprese Benchedite Exprese Benchedite Exprese Benchedite	Image: Term Image: Term Sufficiency: Alares. If Image: Term Sufficiency: Alares. Image: Term Sufficiency: Sufficiency: Sufficiency: Image: Term Suffi
Adversed help help DrCh-d help Despite Advertises Presenters Spress Best-betty System Senductits (D): Second Second Constantion (Deck	Ø
Advanced help for DeCH4 help Designers Advances Fascaneters Types Bandwith: Types Bandwith: Types Bandwith: Types Bandwith: Types Bandwith: Decha Bandwith: Chash Bandwith:	Image: Tests Image: Tests Address Address If Program: Tests Image: Tests Address If Program: Tests Image: Tests Address If Program: Tests Image: Tests Address Image: Tests Operation Image: Tests Image: Tests Image: Tests Operation Image: Tests Image: Tests Image: Tests Operation Image: Tests Image: Tests
Adversed help: too DrOA help: Despeter: Topse bandet: Topse bandet: Topse bandet: Topse bandet: Topse bandet: Topse bandet: Despeter: Onde help: Onde help: Onde help: Onde help:	• Inspect tests • Sufficiential address • Sufficiential addres • Sufficiential address • S
Advanced Intis too DeCh4 India Despeter: Interface Presentations Speen Selendelly: types Selendelly: Speen Selendelly: Selende Conversion Oracle Oracle Interface Oracle Interface Oracl	• Inspect Parts • Instructures Autre. Autre. • Inspect Parts • Instructures Autre. • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts I
Advanced helds to a DeChild helds Designers Egens Bestindler Typess Bestindler Hitti Devels Develses Develse Conversion Ordel Orach Heldel Deck Themas Deck Themas Deck Themas	• Propuls for the * Buffer Address Address Address • Propuls for the * Buffer Address Address Address # + Propuls for the * Buffer Address Address Address # + Propuls for the * Buffer Address Address Address # + Propuls for the * Buffer Address Address Address # + Propuls for the * Buffer Address Address Address # + Propuls for the * Buffer Address Address Address # + Propuls for the * Buffer Address Address Address # # # # # # # # # # # # #
Advessed help to DOA help Beneficies Typese behaviot: Typese behaviot: Typ	Image: Tests Image: Tests Address Address Address If Program: Tests Image: Tests Address Address If Program: Tests Image: Tests Image: Tests Address Image: Tests Tests Image: Tests Image: Tests Image: Tests Image: Tests Tests Image: Tests Image: Tests Image: Tests Image: Tests Tests Image: Tests Image: Tests Image: Tests Image: Tests Tests Image: Tests Image: Tests Image: Tests
Advanced Intic two DeCHA Inde Despeters Types bandets Types bandets Types bandets Types bandets Types bandets Types bandets Dech Interio Dech Interio	• Inspect Parks • Sufficient Advance • Advance
Advanced helds to a DeOrd helds Despeter: Spres Standardt: Spres Standardt: Spre	• Program Harbs • Sufficie Aldress • Sufficie • Suff
Advanced helps to an DeChild helps beneficies frammeters Egress Bestindth: Typess Bestindth: Typess Bestindth: Typess Bestindth: Typess Bestindth: Typess Bestindth: Typess Bestindth: Debts Typess. Debts Typ	Image: Tests Bottle Address Address Address If Program Tests Bottle Address Address Address If Program Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Oper Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Descent Image: TS Address Tests Image: Tes
Advenuel hele ten DrOA hele Begeten Tyres behaldt Tyres behaldt Tyres behaldt Tyres behaldt Tyres behaldt Bekart Oesh Hele Oesh Hele Oesh Hele Oesh Hele Oesh Hele Oesh Hele Oesh Deal Genere Oesh Hele Oesh Deal Genere Tyres format Drof, Tyres Dearson	Image: Instrument Address Image: Instrument Address Image: Instrument Address Image: Instrument Image: Instrument
Advanced Info ton DicOv4 Inde Bagetee: Termine Sectors Termine Sectors	Image: Tests 1 Suffix Address Address If Program: Tests 1 Suffix Address Address If Program: Tests Image: Tests Image: Tests Image: Tests Tests Tests Image: Tests Image: Tests Tests Tests Image: Tests Image: Tests Tests Tests Tests Image: Tests Tests Tests Tests
Advanced hele to to DOVA heles Bangeles Strate Strategiest Strate Strategiest Strate Strategiest Strate Strategiest Strate Strategiest Strate Strategiest Strate Strategiest Strate Strategiest Strate Strategiest Strate Strategiest Strategies	Image: Tests Image: Tests Address Address Address If Progr: Tests Image: Tests Address Address If Progr: Tests Image: Tests Image: Tests Image: Tests Image: Tests Oper Image: Tests Image: Tests Image: Tests Image: Tests Oper Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests Image: Tests
Adversed help to DOA help a beneficies francesses Experise Services Types Behavions Types Behavions Types Behavions Beneficies (Deals Coals Areas Coals Areas DoA Park Istin Artens (Spanda) Tel Of Serve (Dearboo) Tel Of	Image: tests Image: tests Address Image: tests Image: tests Address Image: tests Image: tests Address Image: tests Image: tests Image: tests
Advenuel Info tra DCM-A Inde Begeter: Termina Francesco Termina Fran	Image: tests Image: tests<
Advanced herds than DeChild herds herdsize Preventions Spens Bankelstit Spens Bankelstit Spens Bankelstit Dechild Spense Dechild Spense	• English Static * Schröck Aldrein * Aldrein * Aldrein * Schröck Aldrein * Aldrein * Schröck * Schrö
Advented het fan DrOA hete Bergene Stratuer yn awer yn Yeren Benadet Uru Dea hetedio Oed fan Aren: Oed fan Aren: O	Image: Tests Image: Tests<
Adapted his to DOA hield Departure framework Types bandet: Types bandet: Types bandet: Types bandet: Types bandet: Types bandet: Types bandet: Types bandet: Dada hielderen: Bob Dada bandet: Bob Dada bandet:	Image: tests Image: tests<
Adapted Info to DOA Inde Bageton Termina Parameters Termina Info Termina Info Deal Para Deal	• Program Harbs • Sufficient Address • Address
Advention from the Control Advectory Provide Series Sectory (Sectory (Sectory (Sectory (Sectory (Sectory (Sectory (Sectory (Sectory (Sector) (Sect	Image: Tests
Advented hete ten DCA4 hete Bereface Fernanderen Terretario fernanderen Terretario fernanderen Terretario fernanderen Terretario fernanderen Terretario fernanderen Terretario fernanderen DCA5 heter DCA5 heter DCA	Image: Note: Image: Note:<
Adapted hits to DOA hield begins: Territor frauenters Territor frauenters Territor formation Territor	• Ingeni Hols • Sufficient Alex. • Ingeni Hols • Sufficient Alex. • Ingeni Hols • Sufficient • Alex. • Of Home • Sufficient • Alex. • Of Home • Annes Datewol • Alex. • Prime • Alex. • Alex. • Prime • Alex. • Alex. • Prime • Alex. • Alex. • Ingeni Molex. • Alex. • Alex. • Prime • Alex. • Alex.
Advanced Info to the DeChild Inde Server Server Info Types Server Info Types Server Info Types Server Info Server Server Info Server Info Ser	Image: Texts I
Adversed het ist DOA hetel Bereiter Types Behadet Types Behadet Types Behadet Types Behadet Types Behadet Types Behadet Bekadet Bekadet Dea hetel Dea hete	Image: Instrument of the Address Address Image: Instrument of the Address <td< td=""></td<>
Adversed hete ten DOAA hete begesen Stratus fananstern Types beskuldt Types beskuldt Types beskuldt Types beskuldt Types beskuldt Types beskuldt Types beskuldt Dada fanant Dada	Image: tests Image: tests <td< td=""></td<>
Advenuel Info to DOA Inde Begener: Territor Featureton Territor Featur	• Forget Hall • Softward Allere. • Softward Allere. • Softward Allere. • Forget Hall • Softward Allere. • Softward Allere. • Softward Alle
Advented het i te DOA hete Departer Types Behadet Use Peers Behadet Use Peers Adved to Cash Hete Cash	Image: State Note: A date: A d
Advented hele ten DOAA heles berefete presenter Terrete Bestellen Terrete Bestellen Terrete Bestellen Terrete Bestellen Terreter	Image: back
Abagese: http://www.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://www.http://www.http://www.http://wwww.http://wwww.http://www.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwww.http://wwwww.http://wwwww.http://wwwwwwwwwwwwwwwwwwwwwwwwwwwwwwwwww	• Forget fort • Soft Addres Addres • Forget fort • Soft Addres Addres • Forget fort • Soft Addres • Soft Addres • Forget fort • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Forget fort • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres • Soft Addres
Adapted his to DOA his Departure Types Balandit Types Balandit Types Balandit Types Balandit Types Balandit Types Balandit Deal Hypes Deal Hype	• Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspec Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspect Parts • Inspec Parts
Adapted his is DOA his Department Types based of Types base	Image: back
Adapted his is DOA hield Departer Territor frauences Territor frauences Territor frauences Territor frauences Territor frauences Territor frauences Cosh Hendi Oash Hendi Data Dhal Same Territor Dhale Same Territor Same Don Hendi Data Dhale Same Don Hendi Data Dhale Same Don Hendi Data Dhale Same Don Hendi Data Dhale Same Same Dhale Data Dhale Same Data Dhale Same Data Dhale Same Same Dhale Data Dhale Same Same Territor Data Dhale Same Same Dhale Data Dhale Same Same Territor Data Dhale Same Data Dhale Same Dat	Image: back
Adapted hit is DOA hield Departure	Image: back
Abereach Inte Inn DOA Intel Service Spreaderson Territor Spreade	Image: Instrument of the Subsection Address Image: Instrument of the Subsection Address Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the Subsection Image: Instrument of the S
Abarpad hit is DOA hid barries Stratus function Types bandutti Types bandutti Types bandutti Types bandutti Data function Data functi	• Project Hots • Softward Address Address Address In the Normal Market on the Normal

Figure 95 Configuration > Network > Interface > VLAN > Create Virtual Interface

Each field is explained in the following table.

LABEL	DESCRIPTION
IPv4/IPv6 View / IPv4 View / IPv6 View	Use this button to display both IPv4 and IPv6, IPv4-only, or IPv6-only configuration fields.
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
Create New Object	Click this button to create a DHCPv6 lease or DHCPv6 request object that you may use for the DHCPv6 settings in this screen.
General Settings	
Enable Interface	Select this to turn this interface on. Clear this to disable this interface.
General IPv6 Setting	
Enable IPv6	Select this to enable IPv6 on this interface. Otherwise, clear this to disable it.
Interface Properties	
Interface Type	Select one of the following option depending on the type of network to which the ZyWALL is connected or if you want to additionally manually configure some related settings.
	internal is for connecting to a local network. Other corresponding configuration options: DHCP server and DHCP relay. The ZyWALL automatically adds default SNAT settings for traffic flowing from this interface to an external interface.
	external is for connecting to an external network (like the Internet). The ZyWALL automatically adds this interface to the default WAN trunk.
	For general , the rest of the screen's options do not automatically adjust and you must manually configure a policy route to add routing and SNAT settings for the interface.
Interface Name	This field is read-only if you are editing an existing VLAN interface. Enter the number of the VLAN interface. You can use a number from 0~4094. For example, vlan0, vlan8, and so on. The total number of VLANs you can configure on the ZyWALL depends on the model.
Zone	Select the zone to which the VLAN interface belongs.
Base Port	Select the Ethernet interface on which the VLAN interface runs.
VLAN ID	Enter the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 1 - 4094. (0 and 4095 are reserved.)
Description	Enter a description of this interface. It is not used elsewhere. You can use alphanumeric and () +/:=?!*#@ $_{e}$ - characters, and it can be up to 60 characters long.
IP Address Assignment	
Get Automatically	Select this if this interface is a DHCP client. In this case, the DHCP server configures the IP address, subnet mask, and gateway automatically.
	You should not select this if the interface is assigned to a VRRP group.
Use Fixed IP Address	Select this if you want to specify the IP address, subnet mask, and gateway manually.
IP Address	This field is enabled if you select Use Fixed IP Address.
	Enter the IP address for this interface.
Subnet Mask	This field is enabled if you select Use Fixed IP Address.
	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network

 Table 53
 Configuration > Network > Interface > VLAN > Create Virtual Interface

LABEL	DESCRIPTION
Gateway	This field is enabled if you select Use Fixed IP Address.
	Enter the IP address of the gateway. The ZyWALL sends packets to the gateway when it does not know how to route the packet to its destination. The gateway should be on the same network as the interface.
Metric	Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.
IPv6 Address Assignment	These IP address fields configure an IPv6 IP address on the interface itself.
Enable Stateless Address Auto- configuration (SLAAC)	Select this to enable IPv6 stateless auto-configuration on this interface. The interface will generate an IPv6 IP address itself from a prefix obtained from an IPv6 router in the network.
Link-Local address	This displays the IPv6 link-local address and the network prefix that the ZyWALL generates itself for the interface.
IPv6 Address/ Prefix Length	Enter the IPv6 address and the prefix length for this interface if you want to configure a static IP address for this interface. This field is optional.
	The prefix length indicates what the left-most part of the IP address is the same for all computers in the network, that is, the network address.
Gateway	Enter the IPv6 address of the default outgoing gateway using colon (:) hexadecimal notation.
Metric	Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.
Address from DHCPv6 Prefix Delegation	Use this table to have the ZyWALL obtain an IPv6 prefix from the ISP or a connected uplink router for an internal network, such as the LAN or DMZ. You have to also enter a suffix address which is appended to the delegated prefix to form an address for this interface. See Prefix Delegation on page 107 for more information.
	To use prefix delegation, you must:
	 Create at least one DHCPv6 request object before configuring this table. The external interface must be a DHCPv6 client. You must configure the DHCPv6 request options using a DHCPv6 request object with the type of prefix-delegation. Assign the prefix delegation to an internal interface and enable router advertisement on that interface.
Add	Click this to create an entry.
Edit	Select an entry and click this to change the settings.
Remove	Select an entry and click this to delete it from this table.
#	This field is a sequential value, and it is not associated with any entry.
Delegated Prefix	Select the DHCPv6 request object to use from the drop-down list.
Suffix Address	Enter the ending part of the IPv6 address, a slash (/), and the prefix length. The ZyWALL will append it to the delegated prefix.
	For example, you got a delegated prefix of 2003:1234:5678/48. You want to configure an IP address of 2003:1234:5678:1111:1/128 for this interface, then enter ::1111:0:0:0:1/128 in this field.
Address	This field displays the combined IPv6 IP address for this interface.
	Note: This field displays the combined address after you click OK and reopen this screen.

Table 53 Configuration > Network > Interface > VLAN > Create Virtual Interface (continued)

LABEL	DESCRIPTION
DHCPv6 Setting	
DUID	This field displays the DHCP Unique IDentifier (DUID) of the interface, which is unique and used for identification purposes when the interface is exchanging DHCPv6 messages with others. See DHCPv6 on page 107 for more information.
DUID as MAC	Select this to have the DUID generated from the interface's default MAC address.
Customized DUID	If you want to use a customized DUID, enter it here for the interface.
Enable Rapid Commit	Select this to shorten the DHCPv6 message exchange process from four to two steps. This function helps reduce heavy network traffic load.
	Note: Make sure you also enable this option in the DHCPv6 clients to make rapid commit work.
Information Refresh Time	Enter the number of seconds a DHCPv6 client should wait before refreshing information retrieved from DHCPv6.
Request Address	This field is available if you set this interface to DHCPv6 Client . Select this to get an IPv6 IP address for this interface from the DHCP server. Clear this to not get any IP address information through DHCPv6.
DHCPv6 Request Options /	If this interface is a DHCPv6 client, use this section to configure DHCPv6 request settings that determine what additional information to get from the DHCPv6 server.
Options	If this interface is a DHCPv6 server, use this section to configure DHCPv6 lease settings that determine what to offer to the DHCPv6 clients.
Add	Click this to create an entry in this table. See Section 7.3.3 on page 123 for more information.
Remove	Select an entry and click this to change the settings.
Object Reference	Select an entry and click this to delete it from this table.
#	This field is a sequential value, and it is not associated with any entry.
Name	This field displays the name of the DHCPv6 request or lease object.
Туре	This field displays the type of the object.
Value	This field displays the IPv6 prefix that the ZyWALL obtained from an uplink router (Server is selected) or will advertise to its clients (Client is selected).
Interface	When Relay is selected, select this check box and an interface from the drop-down list if you want to use it as the relay server.
Relay Server	When Relay is selected, select this check box and enter the IP address of a DHCPv6 server as the relay server.
IPv6 Router Advertisement Setting	
Enable Router Advertisement	Select this to enable this interface to send router advertisement messages periodically. See IPv6 Router Advertisement on page 107 for more information.
Advertised Hosts Get Network Configuration	Select this to have the ZyWALL indicate to hosts to obtain network settings (such as prefix and DNS settings) through DHCPv6.
From DHCPv6	Clear this to have the ZyWALL indicate to hosts that DHCPv6 is not available and they should use the prefix in the router advertisement message.
Advertised Hosts Get Other Configuration	Select this to have the ZyWALL indicate to hosts to obtain DNS information through DHCPv6.
From DHCPv6	Clear this to have the ZyWALL indicate to hosts that DNS information is not available in this network.

 Table 53
 Configuration > Network > Interface > VLAN > Create Virtual Interface (continued)

LABEL	DESCRIPTION
Router Preference	Select the router preference (Low , Medium or High) for the interface. The interface sends this preference in the router advertisements to tell hosts what preference they should use for the ZyWALL. This helps hosts to choose their default router especially when there are multiple IPv6 router in the network.
	Note: Make sure the hosts also support router preference to make this function work.
MTU	The Maximum Transmission Unit. Type the maximum size of each IPv6 data packet, in bytes, that can move through this interface. If a larger packet arrives, the ZyWALL divides it into smaller fragments.
Hop Limit	Enter the maximum number of network segments that a packet can cross before reaching the destination. When forwarding an IPv6 packet, IPv6 routers are required to decrease the Hop Limit by 1 and to discard the IPv6 packet when the Hop Limit is 0.
Advertised Prefix Table	Configure this table only if you want the ZyWALL to advertise a fixed prefix to the network.
Add	Click this to create an IPv6 prefix address.
Edit	Select an entry in this table and click this to modify it.
Remove	Select an entry in this table and click this to delete it.
#	This field is a sequential value, and it is not associated with any entry.
IPv6	Enter the IPv6 network prefix address and the prefix length.
Address/ Prefix Length	The prefix length indicates what the left-most part of the IP address is the same for all computers in the network, that is, the network address.
Advertised Prefix from DHCPv6 Prefix Delegation	Use this table to configure the network prefix if you want to use a delegated prefix as the beginning part of the network prefix.
Add	Click this to create an entry in this table.
Edit	Select an entry in this table and click this to modify it.
Remove	Select an entry in this table and click this to delete it.
#	This field is a sequential value, and it is not associated with any entry.
Delegated Prefix	Select the DHCPv6 request object to use for generating the network prefix for the network.
Suffix Address	Enter the ending part of the IPv6 network address plus a slash (/) and the prefix length. The ZyWALL will append it to the selected delegated prefix. The combined address is the network prefix for the network.
	For example, you got a delegated prefix of 2003:1234:5678/48. You want to divide it into 2003:1234:5678:1111/64 for this interface and 2003:1234:5678:2222/64 for another interface. You can use ::1111/64 and ::2222/64 for the suffix address respectively. But if you do not want to divide the delegated prefix into subnetworks, enter ::0/48 here, which keeps the same prefix length (/48) as the delegated prefix.
Address	This is the final network prefix combined by the delegated prefix and the suffix.
	Note: This field displays the combined address after you click OK and reopen this screen.
Interface Parameters	
Egress Bandwidth	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can send through the interface to the network. Allowed values are 0 - 1048576.
Ingress	This is reserved for future use.
Danuwiuth	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can receive from the network through the interface. Allowed values are 0 - 1048576.

Table 53 Configuration > Network > Interface > VLAN > Create Virtual Interface (continued)

LABEL	DESCRIPTION
MTU	Maximum Transmission Unit. Type the maximum size of each data packet, in bytes, that can move through this interface. If a larger packet arrives, the ZyWALL divides it into smaller fragments. Allowed values are 576 - 1500. Usually, this value is 1500.
Connectivity Check	The ZyWALL can regularly check the connection to the gateway you specified to make sure it is still available. You specify how often to check the connection, how long to wait for a response before the attempt is a failure, and how many consecutive failures are required before the ZyWALL stops routing to the gateway. The ZyWALL resumes routing to the gateway the first time the gateway passes the connectivity check.
Enable Connectivity Check	Select this to turn on the connection check.
Check Method	Select the method that the gateway allows.
	Select icmp to have the ZyWALL regularly ping the gateway you specify to make sure it is still available.
	Select tcp to have the ZyWALL regularly perform a TCP handshake with the gateway you specify to make sure it is still available.
Check Period	Enter the number of seconds between connection check attempts.
Check Timeout	Enter the number of seconds to wait for a response before the attempt is a failure.
Check Fail Tolerance	Enter the number of consecutive failures before the ZyWALL stops routing through the gateway.
Check Default Gateway	Select this to use the default gateway for the connectivity check.
Check this address	Select this to specify a domain name or IP address for the connectivity check. Enter that domain name or IP address in the field next to it.
Check Port	This field only displays when you set the Check Method to tcp . Specify the port number to use for a TCP connectivity check.
DHCP Setting	The DHCP settings are available for the OPT, LAN and DMZ interfaces.
DHCP	Select what type of DHCP service the ZyWALL provides to the network. Choices are:
	None - the ZyWALL does not provide any DHCP services. There is already a DHCP server on the network.
	DHCP Relay - the ZyWALL routes DHCP requests to one or more DHCP servers you specify. The DHCP server(s) may be on another network.
	DHCP Server - the ZyWALL assigns IP addresses and provides subnet mask, gateway, and DNS server information to the network. The ZyWALL is the DHCP server for the network.
	These fields appear if the ZyWALL is a DHCP Relay.
Relay Server 1	Enter the IP address of a DHCP server for the network.
Relay Server 2	This field is optional. Enter the IP address of another DHCP server for the network.
	These fields appear if the ZyWALL is a DHCP Server.
IP Pool Start Address	Enter the IP address from which the ZyWALL begins allocating IP addresses. If you want to assign a static IP address to a specific computer, click Add Static DHCP .
	If this field is blank, the Pool Size must also be blank. In this case, the ZyWALL can assign every IP address allowed by the interface's IP address and subnet mask, except for the first address (network address), last address (broadcast address) and the interface's IP address.

 Table 53
 Configuration > Network > Interface > VLAN > Create Virtual Interface (continued)

LABEL	DESCRIPTION
Pool Size	Enter the number of IP addresses to allocate. This number must be at least one and is limited by the interface's Subnet Mask . For example, if the Subnet Mask is 255.255.255.0 and IP Pool Start Address is 10.10.10.10, the ZyWALL can allocate 10.10.10.10 to 10.10.10.254, or 245 IP addresses.
	If this field is blank, the IP Pool Start Address must also be blank. In this case, the ZyWALL can assign every IP address allowed by the interface's IP address and subnet mask, except for the first address (network address), last address (broadcast address) and the interface's IP address.
First DNS Server Second DNS	Specify the IP addresses up to three DNS servers for the DHCP clients to use. Use one of the following ways to specify these IP addresses.
Third DNS Server	Custom Defined - enter a static IP address.
	From ISP - select the DNS server that another interface received from its DHCP server.
	ZyWALL - the DHCP clients use the IP address of this interface and the ZyWALL works as a DNS relay.
First WINS Server, Second WINS Server	Type the IP address of the WINS (Windows Internet Naming Service) server that you want to send to the DHCP clients. The WINS server keeps a mapping table of the computer names on your network and the IP addresses that they are currently using.
Default Router	If you set this interface to DHCP Server , you can select to use either the interface's IP address or another IP address as the default router. This default router will become the DHCP clients' default gateway.
	To use another IP address as the default router, select Custom Defined and enter the IP address.
Lease time	Specify how long each computer can use the information (especially the IP address) before it has to request the information again. Choices are:
	infinite - select this if IP addresses never expire
	days, hours, and minutes - select this to enter how long IP addresses are valid.
Extended Options	This table is available if you selected DHCP server .
options	Configure this table if you want to send more information to DHCP clients through DHCP packets.
Add	Click this to create an entry in this table. See Section 7.3.4 on page 124.
Edit	Select an entry in this table and click this to modify it.
Remove	Select an entry in this table and click this to delete it.
#	This field is a sequential value, and it is not associated with any entry.
Name	This is the option's name.
Code	This is the option's code number.
Туре	This is the option's type.
Value	This is the option's value.
Enable IP/MAC Binding	Select this option to have the ZyWALL enforce links between specific IP addresses and specific MAC addresses for this VLAN. This stops anyone else from manually using a bound IP address on another device connected to this interface. Use this to make use only the intended users get to use specific IP addresses.
Enable Logs for IP/MAC Binding Violation	Select this option to have the ZyWALL generate a log if a device connected to this VLAN attempts to use an IP address that is bound to another device's MAC address.
Static DHCP Table	Configure a list of static IP addresses the ZyWALL assigns to computers connected to the interface. Otherwise, the ZyWALL assigns an IP address dynamically using the interface's IP Pool Start Address and Pool Size .

 Table 53
 Configuration > Network > Interface > VLAN > Create Virtual Interface (continued)

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Select an entry and click this to be able to modify it.
Remove	Select an entry and click this to delete it.
#	This field is a sequential value, and it is not associated with a specific entry.
IP Address	Enter the IP address to assign to a device with this entry's MAC address.
MAC Address	Enter the MAC address to which to assign this entry's IP address.
Description	Enter a description to help identify this static DHCP entry. You can use alphanumeric and () +/:=?!*#@ $_{e}^{-}$ characters, and it can be up to 60 characters long.
RIP Setting	See Section 10.2 on page 197 for more information about RIP.
Enable RIP	Select this to enable RIP on this interface.
Direction	This field is effective when RIP is enabled. Select the RIP direction from the drop-down list box.
	BiDir - This interface sends and receives routing information.
	In-Only - This interface receives routing information.
	Out-Only - This interface sends routing information.
Send Version	This field is effective when RIP is enabled. Select the RIP version(s) used for sending RIP packets. Choices are 1 , 2 , and 1 and 2 .
Receive Version	This field is effective when RIP is enabled. Select the RIP version(s) used for receiving RIP packets. Choices are 1 , 2 , and 1 and 2 .
V2-Broadcast	This field is effective when RIP is enabled. Select this to send RIP-2 packets using subnet broadcasting; otherwise, the ZyWALL uses multicasting.
OSPF Setting	See Section 10.3 on page 199 for more information about OSPF.
Area	Select the area in which this interface belongs. Select None to disable OSPF in this interface.
Priority	Enter the priority (between 0 and 255) of this interface when the area is looking for a Designated Router (DR) or Backup Designated Router (BDR). The highest-priority interface identifies the DR, and the second-highest-priority interface identifies the BDR. Set the priority to zero if the interface can not be the DR or BDR.
Link Cost	Enter the cost (between 1 and 65,535) to route packets through this interface.
Passive Interface	Select this to stop forwarding OSPF routing information from the selected interface. As a result, this interface only receives routing information.
Authentication	Select an authentication method, or disable authentication. To exchange OSPF routing information with peer border routers, you must use the same authentication method that they use. Choices are:
	Same-as-Area - use the default authentication method in the area
	None - disable authentication
	Text - authenticate OSPF routing information using a plain-text password
	MD5 - authenticate OSPF routing information using MD5 encryption
Text Authentication Key	This field is available if the Authentication is Text . Type the password for text authentication. The key can consist of alphanumeric characters and the underscore, and it can be up to 16 characters long.
MD5 Authentication ID	This field is available if the Authentication is MD5 . Type the ID for MD5 authentication. The ID can be between 1 and 255.

 Table 53
 Configuration > Network > Interface > VLAN > Create Virtual Interface (continued)

LABEL	DESCRIPTION
MD5 Authentication Key	This field is available if the Authentication is MD5 . Type the password for MD5 authentication. The password can consist of alphanumeric characters and the underscore, and it can be up to 16 characters long.
Related Setting	
Configure WAN TRUNK	Click WAN TRUNK to go to a screen where you can set this VLAN to be part of a WAN trunk for load balancing.
Configure Policy Route	Click Policy Route to go to the screen where you can manually configure a policy route to associate traffic with this VLAN.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 53 Configuration > Network > Interface > VLAN > Create Virtual Interface (continued)

7.8 Bridge Interfaces

This section introduces bridges and bridge interfaces and then explains the screens for bridge interfaces.

Bridge Overview

A bridge creates a connection between two or more network segments at the layer-2 (MAC address) level. In the following example, bridge X connects four network segments.



When the bridge receives a packet, the bridge records the source MAC address and the port on which it was received in a table. It also looks up the destination MAC address in the table. If the bridge knows on which port the destination MAC address is located, it sends the packet to that port. If the destination MAC address is not in the table, the bridge broadcasts the packet on every port (except the one on which it was received).

In the example above, computer A sends a packet to computer B. Bridge X records the source address 0A:0A:0A:0A:0A:0A:0A and port 2 in the table. It also looks up 0B:0B:0B:0B:0B:0B:0B in the table. There is no entry yet, so the bridge broadcasts the packet on ports 1, 3, and 4.

 Table 54
 Example: Bridge Table After Computer A Sends a Packet to Computer B

MAC ADDRESS	PORT
0A:0A:0A:0A:0A:0A	2

If computer B responds to computer A, bridge X records the source address OB:OB:OB:OB:OB:OB and port 4 in the table. It also looks up 0A:0A:0A:0A:0A:0A in the table and sends the packet to port 2 accordingly.

Table 55	Example:	Bridge Table	After Computer	B Responds	to Computer A
----------	----------	--------------	----------------	-------------------	---------------

MAC ADDRESS	PORT
0A:0A:0A:0A:0A	2
0B:0B:0B:0B:0B:0B	4

Bridge Interface Overview

A bridge interface creates a software bridge between the members of the bridge interface. It also becomes the ZyWALL's interface for the resulting network.

Unlike the device-wide bridge mode in ZyNOS-based ZyWALLs, this ZyWALL can bridge traffic between some interfaces while it routes traffic for other interfaces. The bridge interfaces also support more functions, like interface bandwidth parameters, DHCP settings, and connectivity check. To use the whole ZyWALL as a transparent bridge, add all of the ZyWALL's interfaces to a bridge interface.

A bridge interface may consist of the following members:

- Zero or one VLAN interfaces (and any associated virtual VLAN interfaces)
- · Any number of Ethernet interfaces (and any associated virtual Ethernet interfaces)

When you create a bridge interface, the ZyWALL removes the members' entries from the routing table and adds the bridge interface's entries to the routing table. For example, this table shows the routing table before and after you create bridge interface br0 (250.250.250.0/23) between lan1 and vlan1.

IP ADDRESS(ES)	DESTINATION
210.210.210.0/24	lan1
210.211.1.0/24	lan1:1
221.221.221.0/24	vlan0
222.222.222.0/24	vlan1
230.230.230.192/26	wan
241.241.241.241/32	dmz
242.242.242.242/32	dmz

Table 56 Example: Routing	Table Before and	After Br	idge Interface br0 Is Create	ed
IP ADDRESS(ES)	DESTINATION		IP ADDRESS(ES)	DESTINATION
210.210.210.0/24	lan1		221.221.221.0/24	vlan0
210.211.1.0/24	lan1:1		230.230.230.192/26	wan
221.221.221.0/24	vlan0		241.241.241.241/32	dmz
222.222.222.0/24	vlan1		242.242.242.242/32	dmz
230.230.230.192/26	wan		250.250.250.0/23	br0

In this example, virtual Ethernet interface lan1:1 is also removed from the routing table when lan1 is added to br0. Virtual interfaces are automatically added to or remove from a bridge interface when the underlying interface is added or removed.

7.8.1 Bridge Summary

This screen lists every bridge interface and virtual interface created on top of bridge interfaces. If you enabled IPv6 in the Configuration > System > IPv6 screen, you can also configure bridge interfaces used for your IPv6 network on this screen. To access this screen, click Configuration > Network > Interface > Bridge.

🙄 Ac	Statua Nama	e e Activate e Inactivate e	Iureate virtual Interface Member	
Ŧ	Status Name	IP Address	member	
K ·	Page 1 of 1	Show 50 vitems		No data to displa
#	Status Name	IP Address	Member	
M ·	Page 1 of 1	Show 50 🗸 items		No data to displa
K (Page 1 of 1	▷ 🕅 Show 50 💌 items		No data to dis

Figure 96 Configuration > Network > Interface > Bridge

Each field is described in the following table.

LABEL	DESCRIPTION
Configuration / IPv6 Configuration	Use the Configuration section for IPv4 network settings. Use the IPv6 Configuration section for IPv6 network settings if you connect your ZyWALL to an IPv6 network. Both sections have similar fields as described below.
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Create Virtual Interface	To open the screen where you can create a virtual interface, select an interface and click Create Virtual Interface .
Object References	Select an entry and click Object Reference to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with any interface.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Name	This field displays the name of the interface.
IP Address	This field displays the current IP address of the interface. If the IP address is 0.0.0.0, the interface does not have an IP address yet.
	This screen also shows whether the IP address is a static IP address (STATIC) or dynamically assigned (DHCP). IP addresses are always static in virtual interfaces.
Member	This field displays the Ethernet interfaces and VLAN interfaces in the bridge interface. It is blank for virtual interfaces.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 57
 Configuration > Network > Interface > Bridge

7.8.2 Bridge Add/Edit

This screen lets you configure IP address assignment, interface bandwidth parameters, DHCP settings, and connectivity check for each bridge interface. To access this screen, click the **Create Virtual Interface** icon in the **Bridge Summary** screen. The following screen appears.

Add Bridge	
vingening view * Impile Advanced Settings	El name can relat
General IPv6 Setting	
Interface Type:	general (m. 🖬
Interface Name:	· 0
Zurre:	LANG M
Descriptioni	(Optional)
Hember Configuration	
Available	Hember
lani	
lan2 dnz	
wian-1-1	•
In address further state	
C Get Automatically	
Use Fixed IP Address	
P Add test:	0.0.0
Subnet Mask:	0.0.0
Netro	0 (0-15)
IPv6 Address Assignment	aton (5.28C)
Link-Local Address:	nja
IPv6 Address,Prefix Langth:	Optional
Gateway:	Cotonal
Address from DHCPV6 Prefix	Chatel (2010)
Delegation	# Completifiers Suffix Address Addre
	1/ / Barr (d) / All Barr () Later Bridge to design
	1/ / Lund 7 a. 7 1. Li Rase 20 Res to 202 0 others
DHCPv6 Setting	
DUD	and M
DUED IN MAC	
Customened DUID)	
Enable Rapid Connit	
Information Refresh Time:	4294967295 (600-4294667295)
DHCPv6 Lease Options	QAM # / more to family and a family and a
	8 <u>Name + Type</u> Value
	(i ∈ Page 1 of 1 ≥ i) show 50 wire
MTU: Http://www.	1400 (1386-1500) 44 (1-280)
Advertised Pretix Table	
	· Provident and a second
	14 4 Page 3 of 1 2 21 Show 50 withms for data to deplay
Advertised Prefix Fran DHCPu6 Prefix Delegation	Over Mare Barren
	# Celegated Prefs Suffs Address Addre
	11 - 1 Page 3 of 2 P P Show 50 w/w No data to dayley
Interface Parameters	
Egress Bandwidths	1048576 K2ps
Ingress Bandwidth:	2048576 Kbps
RIU.	1300. Bytes
DHCP Setting	
DIO:	Did Sever v
Prot Start Address (Optional):	Catter Defrad
Second DNS Server (Optional):	Custon Defined
Third DNS Server (Optional):	Custom Defined
First W2NG Server (Optional):	
Second WINS Server (Optional):	
Lenie Time:	
and the second second	3 days 0 hours (Optional) emutes (Optional)
Extended Options	
and a second sec	A have Color Type Value
E Bruble P.MAC Briding	11 1 Page 1 of 1 P. P. Show 50 w terms . The data to deploy
E Bruble Logs for IP,MAC Binding Visiator	
State DHOP Table	O ANI (2019) Brown
	RAC Description
	14 4 Page 1 of 1 3 31, Share 50 w items the data to display
Connectivity Check	
Connectivity Check	
Connectivity Check	ang (s-s) seconds) (M)
Connectivity Dieck Dieckie Connectivity Dieck Diecki Herosci Dieck Timeouti Dieck Timeouti	Simp (b0) seconds) 5 (b0) seconds)
Connectivity Check Deals Correctivity Check Oreal Network Oreal Tamenat Oreal Fail Tairmone Check Fail Tairmone Check Fail Tairmone	XX (2-30 microse) (8) 5 (2+30 microse) (2-30 microse) 5 (2+30 microse) (2-30 microse)
Connectivity Oleck Deals Connectivity Oleck Oreal Nethod Unear Aread Oreal Treault Oreal Seful Tolenore @ Oreal Seful Tolenore @ Oreal Seful Tolenore @ Oreal Seful Tolenore	X0 Dr-Gr Manneng S (Dr Sh Manneh) S (Dr Sh County Topes on 10 Address
Connectivity Clanck Deals Connectivity Orack Oracl Martind Oracl Martind Oracl Torenct Oracl S foll Information Oracl S followed Deals D followed D beals D followed D followed D followed D beals D followed D followed	Image Device Recording Image 5 () build recording () 5 () build recording () 0.00.00 () () 0.00.00 () ()
Connectivity Check Contentivity Check Contentivity Check Contentivity Check Territorium Check Territor	Mail DrG. seconds Mail 5 (C. 10 memorit) 3 5 (C. 40) Conservance or 2P Address
Connectivity Check Connectivity Oreck Connectivity Oreck Oreck Trease: Oreck Trease: Oreck Trease: Oreck Trease: Oreck Trease: Oreck Trease: Connectivity Oreck Connectivity C	Mail Di-All standards All Di-All standards S Di-All standards S Di-All standards S Di-All standards Guidant Anne or 2 ⁿ Address

Figure 97 Configuration > Network > Interface > Bridge > Create Virtual Interface

Each field is described in the table below.

|--|

LABEL	DESCRIPTION
IPv4/IPv6 View / IPv4 View / IPv6 View	Use this button to display both IPv4 and IPv6, IPv4-only, or IPv6-only configuration fields.
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
Create New Object	Click this button to create a DHCPv6 lease or DHCPv6 request object that you may use for the DHCPv6 settings in this screen.
General Settings	
Enable Interface	Select this to enable this interface. Clear this to disable this interface.
General IPv6 Setting	
Enable IPv6	Select this to enable IPv6 on this interface. Otherwise, clear this to disable it.
Interface Properties	
Interface Type	Select one of the following option depending on the type of network to which the ZyWALL is connected or if you want to additionally manually configure some related settings.
	internal is for connecting to a local network. Other corresponding configuration options: DHCP server and DHCP relay. The ZyWALL automatically adds default SNAT settings for traffic flowing from this interface to an external interface.
	external is for connecting to an external network (like the Internet). The ZyWALL automatically adds this interface to the default WAN trunk.
	For general , the rest of the screen's options do not automatically adjust and you must manually configure a policy route to add routing and SNAT settings for the interface.
Interface Name	This field is read-only if you are editing the interface. Enter the name of the bridge interface. The format is brx , where x is 0 - 11. For example, br0, br3, and so on.
Zone	Select the zone to which the interface is to belong. You use zones to apply security settings such as firewall, remote management.
Description	Enter a description of this interface. It is not used elsewhere. You can use alphanumeric and () +/:=?!*#@ $^{-}$ characters, and it can be up to 60 characters long.
Member Configuration	
Available	This field displays Ethernet interfaces and VLAN interfaces that can become part of the bridge interface. An interface is not available in the following situations:
	There is a virtual interface on top of itIt is already used in a different bridge interface
	Select one, and click the >> arrow to add it to the bridge interface. Each bridge interface can only have one VLAN interface.
Member	This field displays the interfaces that are part of the bridge interface. Select one, and click the << arrow to remove it from the bridge interface.
IP Address Assignment	
Get Automatically	Select this if this interface is a DHCP client. In this case, the DHCP server configures the IP address, subnet mask, and gateway automatically.
Use Fixed IP Address	Select this if you want to specify the IP address, subnet mask, and gateway manually.

LABEL	DESCRIPTION
IP Address	This field is enabled if you select Use Fixed IP Address.
	Enter the IP address for this interface.
Subnet Mask	This field is enabled if you select Use Fixed IP Address.
	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.
Gateway	This field is enabled if you select Use Fixed IP Address.
	Enter the IP address of the gateway. The ZyWALL sends packets to the gateway when it does not know how to route the packet to its destination. The gateway should be on the same network as the interface.
Metric	Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.
IPv6 Address Assignment	These IP address fields configure an IPv6 IP address on the interface itself.
Enable Stateless Address Auto- configuration (SLAAC)	Select this to enable IPv6 stateless auto-configuration on this interface. The interface will generate an IPv6 IP address itself from a prefix obtained from an IPv6 router in the network.
Link-Local address	This displays the IPv6 link-local address and the network prefix that the ZyWALL generates itself for the interface.
IPv6 Address/ Prefix Length	Enter the IPv6 address and the prefix length for this interface if you want to use a static IP address. This field is optional.
	The prefix length indicates what the left-most part of the IP address is the same for all computers in the network, that is, the network address.
Gateway	Enter the IPv6 address of the default outgoing gateway using colon (:) hexadecimal notation.
Metric	Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.
Address from DHCPv6 Prefix Delegation	Use this table to have the ZyWALL obtain an IPv6 prefix from the ISP or a connected uplink router for an internal network, such as the LAN or DMZ. You have to also enter a suffix address which is appended to the delegated prefix to form an address for this interface. See Prefix Delegation on page 107 for more information.
	To use prefix delegation, you must:
	 Create at least one DHCPv6 request object before configuring this table. The external interface must be a DHCPv6 client. You must configure the DHCPv6 request options using a DHCPv6 request object with the type of prefix-delegation. Assign the prefix delegation to an internal interface and enable router advertisement on that interface.
Add	Click this to create an entry.
Edit	Select an entry and click this to change the settings.
Remove	Select an entry and click this to delete it from this table.
#	This field is a sequential value, and it is not associated with any entry.
Delegated Prefix	Select the DHCPv6 request object to use from the drop-down list.

 Table 58
 Configuration > Network > Interface > Bridge > Create Virtual Interface (continued)

LABEL	DESCRIPTION
Suffix Address	Enter the ending part of the IPv6 address, a slash (/), and the prefix length. The ZyWALL will append it to the delegated prefix.
	For example, you got a delegated prefix of 2003:1234:5678/48. You want to configure an IP address of 2003:1234:5678:1111::1/128 for this interface, then enter ::1111:0:0:0:1/128 in this field.
Address	This field displays the combined IPv6 IP address for this interface.
	Note: This field displays the combined address after you click OK and reopen this screen.
DHCPv6 Setting	
DUID	This field displays the DHCP Unique IDentifier (DUID) of the interface, which is unique and used for identification purposes when the interface is exchanging DHCPv6 messages with others. See DHCPv6 on page 107 for more information.
DUID as MAC	Select this if you want the DUID is generated from the interface's default MAC address.
Customized DUID	If you want to use a customized DUID, enter it here for the interface.
Enable Rapid Commit	Select this to shorten the DHCPv6 message exchange process from four to two steps. This function helps reduce heavy network traffic load.
	Note: Make sure you also enable this option in the DHCPv6 clients to make rapid commit work.
Information Refresh Time	Enter the number of seconds a DHCPv6 client should wait before refreshing information retrieved from DHCPv6.
Request Address	This field is available if you set this interface to DHCPv6 Client . Select this to get an IPv6 IP address for this interface from the DHCP server. Clear this to not get any IP address information through DHCPv6.
DHCPv6 Request Options / DHCPv6 Lease	If this interface is a DHCPv6 client, use this section to configure DHCPv6 request settings that determine what additional information to get from the DHCPv6 server.
Options	If the interface is a DHCPv6 server, use this section to configure DHCPv6 lease settings that determine what to offer to the DHCPv6 clients.
Add	Click this to create an entry in this table. See Section 7.3.3 on page 123 for more information.
Remove	Select an entry and click this to change the settings.
Object Reference	Select an entry and click this to delete it from this table.
#	This field is a sequential value, and it is not associated with any entry.
Name	This field displays the name of the DHCPv6 request or lease object.
Туре	This field displays the type of the object.
Value	This field displays the IPv6 prefix that the ZyWALL obtained from an uplink router (Server is selected) or will advertise to its clients (Client is selected).
Interface	When Relay is selected, select this check box and an interface from the drop-down list if you want to use it as the relay server.
Relay Server	When Relay is selected, select this check box and enter the IP address of a DHCPv6 server as the relay server.
IPv6 Router Advertisement Setting	
Enable Router Advertisement	Select this to enable this interface to send router advertisement messages periodically. See IPv6 Router Advertisement on page 107 for more information.

Table 58	Configuration >	Network >	Interface >	Bridge >	Create	Virtual	Interface	(continued))
								•	

LABEL	DESCRIPTION					
Advertised Hosts Get Network Configuration	Select this to have the ZyWALL indicate to hosts to obtain network settings (such as prefix and DNS settings) through DHCPv6.					
From DHCPv6	Clear this to have the ZyWALL indicate to hosts that DHCPv6 is not available and the should use the prefix in the router advertisement message.					
Advertised Hosts Get Other Configuration	Select this to have the ZyWALL indicate to hosts to obtain DNS information through DHCPv6.					
From DHCPv6	Clear this to have the ZyWALL indicate to hosts that DNS information is not available in this network.					
Router Preference	Select the router preference (Low , Medium or High) for the interface. The interface sends this preference in the router advertisements to tell hosts what preference they should use for the ZyWALL. This helps hosts to choose their default router especially when there are multiple IPv6 router in the network.					
	Note: Make sure the hosts also support router preference to make this function work.					
MTU	The Maximum Transmission Unit. Type the maximum size of each IPv6 data packet, in bytes, that can move through this interface. If a larger packet arrives, the ZyWALL divides it into smaller fragments.					
Hop Limit	Enter the maximum number of network segments that a packet can cross before reaching the destination. When forwarding an IPv6 packet, IPv6 routers are required to decrease the Hop Limit by 1 and to discard the IPv6 packet when the Hop Limit is 0.					
Advertised Prefix Table	Configure this table only if you want the ZyWALL to advertise a fixed prefix to the network.					
Add	Click this to create an IPv6 prefix address.					
Edit	Select an entry in this table and click this to modify it.					
Remove	Select an entry in this table and click this to delete it.					
#	This field is a sequential value, and it is not associated with any entry.					
IPv6	Enter the IPv6 network prefix address and the prefix length.					
Prefix Length	The prefix length indicates what the left-most part of the IP address is the same for all computers in the network, that is, the network address.					
Advertised Prefix from DHCPv6 Prefix Delegation	Use this table to configure the network prefix if you want to use a delegated prefix as the beginning part of the network prefix.					
Add	Click this to create an entry in this table.					
Edit	Select an entry in this table and click this to modify it.					
Remove	Select an entry in this table and click this to delete it.					
#	This field is a sequential value, and it is not associated with any entry.					
Delegated Prefix	Select the DHCPv6 request object to use for generating the network prefix for the network.					
Suffix Address	Enter the ending part of the IPv6 network address plus a slash (/) and the prefix length. The ZyWALL will append it to the selected delegated prefix. The combined address is the network prefix for the network.					
	For example, you got a delegated prefix of 2003:1234:5678/48. You want to divide it into 2003:1234:5678:1111/64 for this interface and 2003:1234:5678:2222/64 for another interface. You can use ::1111/64 and ::2222/64 for the suffix address respectively. But if you do not want to divide the delegated prefix into subnetworks, enter ::0/48 here, which keeps the same prefix length (/48) as the delegated prefix.					

Table 58	Configuration >	Network >	Interface >	Bridge >	Create	Virtual	Interface	(continued)
				· · · · ·				(· · · · · · /

LABEL	DESCRIPTION
Address	This is the final network prefix combined by the selected delegated prefix and the suffix.
	Note: This field displays the combined address after you click OK and reopen this screen.
Interface Parameters	
Egress Bandwidth	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can send through the interface to the network. Allowed values are 0 - 1048576.
Ingress Randwidth	This is reserved for future use.
Bandwidth	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can receive from the network through the interface. Allowed values are 0 - 1048576.
MTU	Maximum Transmission Unit. Type the maximum size of each data packet, in bytes, that can move through this interface. If a larger packet arrives, the ZyWALL divides it into smaller fragments. Allowed values are 576 - 1500. Usually, this value is 1500.
DHCP Setting	
DHCP	Select what type of DHCP service the ZyWALL provides to the network. Choices are:
	None - the ZyWALL does not provide any DHCP services. There is already a DHCP server on the network.
	DHCP Relay - the ZyWALL routes DHCP requests to one or more DHCP servers you specify. The DHCP server(s) may be on another network.
	DHCP Server - the ZyWALL assigns IP addresses and provides subnet mask, gateway, and DNS server information to the network. The ZyWALL is the DHCP server for the network.
	These fields appear if the ZyWALL is a DHCP Relay.
Relay Server 1	Enter the IP address of a DHCP server for the network.
Relay Server 2	This field is optional. Enter the IP address of another DHCP server for the network.
	These fields appear if the ZyWALL is a DHCP Server .
IP Pool Start Address	Enter the IP address from which the ZyWALL begins allocating IP addresses. If you want to assign a static IP address to a specific computer, click Add Static DHCP .
	If this field is blank, the Pool Size must also be blank. In this case, the ZyWALL can assign every IP address allowed by the interface's IP address and subnet mask, except for the first address (network address), last address (broadcast address) and the interface's IP address.
Pool Size	Enter the number of IP addresses to allocate. This number must be at least one and is limited by the interface's Subnet Mask . For example, if the Subnet Mask is 255.255.255.0 and IP Pool Start Address is 10.10.10.10, the ZyWALL can allocate 10.10.10.10 to 10.10.10.254, or 245 IP addresses.
	If this field is blank, the IP Pool Start Address must also be blank. In this case, the ZyWALL can assign every IP address allowed by the interface's IP address and subnet mask, except for the first address (network address), last address (broadcast address) and the interface's IP address.
First DNS Server Second DNS Server	Specify the IP addresses up to three DNS servers for the DHCP clients to use. Use one of the following ways to specify these IP addresses.
Third DNS Server	Custom Defined - enter a static IP address.
	From ISP - select the DNS server that another interface received from its DHCP server.
	ZyWALL - the DHCP clients use the IP address of this interface and the ZyWALL works as a DNS relay.

Table 58	Configuration	> Network >	Interface >	Bridge > Create	Virtual Interface	(continued)
	3			5		````

LABEL	DESCRIPTION
First WINS Server, Second WINS Server	Type the IP address of the WINS (Windows Internet Naming Service) server that you want to send to the DHCP clients. The WINS server keeps a mapping table of the computer names on your network and the IP addresses that they are currently using.
Default Router	If you set this interface to DHCP Server , you can select to use either the interface's IP address or another IP address as the default router. This default router will become the DHCP clients' default gateway.
	To use another IP address as the default router, select Custom Defined and enter the IP address.
Lease time	Specify how long each computer can use the information (especially the IP address) before it has to request the information again. Choices are:
	infinite - select this if IP addresses never expire
	days, hours, and minutes - select this to enter how long IP addresses are valid.
Extended	This table is available if you selected DHCP server .
Options	Configure this table if you want to send more information to DHCP clients through DHCP packets.
Add	Click this to create an entry in this table. See Section 7.3.4 on page 124.
Edit	Select an entry in this table and click this to modify it.
Remove	Select an entry in this table and click this to delete it.
#	This field is a sequential value, and it is not associated with any entry.
Name	This is the option's name.
Code	This is the option's code number.
Туре	This is the option's type.
Value	This is the option's value.
Enable IP/MAC Binding	Select this option to have this interface enforce links between specific IP addresses and specific MAC addresses. This stops anyone else from manually using a bound IP address on another device connected to this interface. Use this to make use only the intended users get to use specific IP addresses.
Enable Logs for IP/MAC Binding Violation	Select this option to have the ZyWALL generate a log if a device connected to this interface attempts to use an IP address that is bound to another device's MAC address.
Static DHCP Table	Configure a list of static IP addresses the ZyWALL assigns to computers connected to the interface. Otherwise, the ZyWALL assigns an IP address dynamically using the interface's IP Pool Start Address and Pool Size .
Add	Click this to create a new entry.
Edit	Select an entry and click this to be able to modify it.
Remove	Select an entry and click this to delete it.
#	This field is a sequential value, and it is not associated with a specific entry.
IP Address	Enter the IP address to assign to a device with this entry's MAC address.
MAC Address	Enter the MAC address to which to assign this entry's IP address.
Description	Enter a description to help identify this static DHCP entry. You can use alphanumeric and ()+/:=?!*#@ $_{-}$ characters, and it can be up to 60 characters long.
Connectivity Check	The interface can regularly check the connection to the gateway you specified to make sure it is still available. You specify how often the interface checks the connection, how long to wait for a response before the attempt is a failure, and how many consecutive failures are required before the ZyWALL stops routing to the gateway. The ZyWALL resumes routing to the gateway the first time the gateway passes the connectivity check.

Table 58	Configuration >	Network >	Interface >	Bridge >	Create	Virtual	Interface	(continued)
----------	-----------------	-----------	-------------	----------	--------	---------	-----------	-------------

LABEL	DESCRIPTION
Enable Connectivity Check	Select this to turn on the connection check.
Check Method	Select the method that the gateway allows.
	Select icmp to have the ZyWALL regularly ping the gateway you specify to make sure it is still available.
	Select tcp to have the ZyWALL regularly perform a TCP handshake with the gateway you specify to make sure it is still available.
Check Period	Enter the number of seconds between connection check attempts.
Check Timeout	Enter the number of seconds to wait for a response before the attempt is a failure.
Check Fail Tolerance	Enter the number of consecutive failures before the ZyWALL stops routing through the gateway.
Check Default Gateway	Select this to use the default gateway for the connectivity check.
Check this address	Select this to specify a domain name or IP address for the connectivity check. Enter that domain name or IP address in the field next to it.
Check Port	This field only displays when you set the Check Method to tcp . Specify the port number to use for a TCP connectivity check.
Related Setting	
Configure WAN TRUNK	Click WAN TRUNK to go to a screen where you can configure the interface as part of a WAN trunk for load balancing.
Configure Policy Route	Click Policy Route to go to the screen where you can manually configure a policy route to associate traffic with this bridge interface.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 58 Configuration > Network > Interface > Bridge > Create Virtual Interface (continued)

7.9 Virtual Interfaces

Use virtual interfaces to tell the ZyWALL where to route packets. Virtual interfaces can also be used in VPN gateways (see Chapter 20 on page 272) and VRRP groups (see Chapter 26 on page 349).

Virtual interfaces can be created on top of Ethernet interfaces, VLAN interfaces, or bridge interfaces. Virtual VLAN interfaces recognize and use the same VLAN ID. Otherwise, there is no difference between each type of virtual interface. Network policies (for example, firewall rules) that apply to the underlying interface automatically apply to the virtual interface as well.

Like other interfaces, virtual interfaces have an IP address, subnet mask, and gateway used to make routing decisions. However, you have to manually specify the IP address and subnet mask; virtual interfaces cannot be DHCP clients. Like other interfaces, you can restrict bandwidth through virtual interfaces, but you cannot change the MTU. The virtual interface uses the same MTU that the underlying interface uses. Unlike other interfaces, virtual interfaces do not provide DHCP services, and they do not verify that the gateway is available.

7.9.1 Virtual Interfaces Add/Edit

This screen lets you configure IP address assignment and interface parameters for virtual interfaces. To access this screen, click the **Create Virtual Interface** icon in the Ethernet, VLAN, or bridge interface summary screen.

Interface Name:	wan1:1	
Description:		(Optional)
P Address Assignment		
IP Address:	0.0.0.0	
Subnet Mask:	0.0.0	
Gateway:		(Optional)
Metric:	0 (015)	
interface Parameters		
Egress Bandwidth:	1048576	Kbps
Ingress Bandwidth:	1048576	Kbps

Figure 98 Configuration > Network > Interface > Create Virtual Interface

Each field is described in the table below.

Tuble of Configuration > Hothon 200 > Croate Virtual Internate
--

LABEL	DESCRIPTION
Interface Properties	
Interface Name	This field is read-only. It displays the name of the virtual interface, which is automatically derived from the underlying Ethernet interface, VLAN interface, or bridge interface.
Description	Enter a description of this interface. It is not used elsewhere. You can use alphanumeric and () +/:=?!*#@ $^-$ characters, and it can be up to 60 characters long.
IP Address Assignment	
IP Address	Enter the IP address for this interface.
Subnet Mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.
Gateway	Enter the IP address of the gateway. The ZyWALL sends packets to the gateway when it does not know how to route the packet to its destination. The gateway should be on the same network as the interface.
Metric	Enter the priority of the gateway (if any) on this interface. The ZyWALL decides which gateway to use based on this priority. The lower the number, the higher the priority. If two or more gateways have the same priority, the ZyWALL uses the one that was configured first.
Interface Parameters	
Egress Bandwidth	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can send through the interface to the network. Allowed values are 0 - 1048576.

LABEL	DESCRIPTION
Ingress Bandwidth	This is reserved for future use.
	Enter the maximum amount of traffic, in kilobits per second, the ZyWALL can receive from the network through the interface. Allowed values are 0 - 1048576.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

 Table 59
 Configuration > Network > Interface > Create Virtual Interface (continued)

7.10 Interface Technical Reference

Here is more detailed information about interfaces on the ZyWALL.

IP Address Assignment

Most interfaces have an IP address and a subnet mask. This information is used to create an entry in the routing table.





 Table 60
 Example: Routing Table Entries for Interfaces

IP ADDRESS(ES)	DESTINATION
100.100.1.1/16	lan1
200.200.200.1/24	wan1

For example, if the ZyWALL gets a packet with a destination address of 100.100.25.25, it routes the packet to interface lan1. If the ZyWALL gets a packet with a destination address of 200.200.200, it routes the packet to interface wan1.

In most interfaces, you can enter the IP address and subnet mask manually. In PPPoE/PPTP interfaces, however, the subnet mask is always 255.255.255.255 because it is a point-to-point interface. For these interfaces, you can only enter the IP address.

In many interfaces, you can also let the IP address and subnet mask be assigned by an external DHCP server on the network. In this case, the interface is a DHCP client. Virtual interfaces, however, cannot be DHCP clients. You have to assign the IP address and subnet mask manually.

In general, the IP address and subnet mask of each interface should not overlap, though it is possible for this to happen with DHCP clients.

In the example above, if the ZyWALL gets a packet with a destination address of 5.5.5.5, it might not find any entries in the routing table. In this case, the packet is dropped. However, if there is a default router to which the ZyWALL should send this packet, you can specify it as a gateway in one of the interfaces. For example, if there is a default router at 200.200.200.100, you can create a gateway at 200.200.200.100 on ge2. In this case, the ZyWALL creates the following entry in the routing table.

Table 61 Example: Routing Table Entry for a Gateway

IP ADDRESS(ES)	DESTINATION
0.0.0/0	200.200.200.100

The gateway is an optional setting for each interface. If there is more than one gateway, the ZyWALL uses the gateway with the lowest metric, or cost. If two or more gateways have the same metric, the ZyWALL uses the one that was set up first (the first entry in the routing table). In PPPoE/PPTP interfaces, the other computer is the gateway for the interface by default. In this case, you should specify the metric.

If the interface gets its IP address and subnet mask from a DHCP server, the DHCP server also specifies the gateway, if any.

Interface Parameters

The ZyWALL restricts the amount of traffic into and out of the ZyWALL through each interface.

- Egress bandwidth sets the amount of traffic the ZyWALL sends out through the interface to the network.
- Ingress bandwidth sets the amount of traffic the ZyWALL allows in through the interface from the network.²

If you set the bandwidth restrictions very high, you effectively remove the restrictions.

The ZyWALL also restricts the size of each data packet. The maximum number of bytes in each packet is called the maximum transmission unit (MTU). If a packet is larger than the MTU, the ZyWALL divides it into smaller fragments. Each fragment is sent separately, and the original packet is re-assembled later. The smaller the MTU, the more fragments sent, and the more work required to re-assemble packets correctly. On the other hand, some communication channels, such as Ethernet over ATM, might not be able to handle large data packets.

DHCP Settings

Dynamic Host Configuration Protocol (DHCP, RFC 2131, RFC 2132) provides a way to automatically set up and maintain IP addresses, subnet masks, gateways, and some network information (such as the IP addresses of DNS servers) on computers in the network. This reduces the amount of manual configuration you have to do and usually uses available IP addresses more efficiently.

In DHCP, every network has at least one DHCP server. When a computer (a DHCP client) joins the network, it submits a DHCP request. The DHCP servers get the request; assign an IP address; and provide the IP address, subnet mask, gateway, and available network information to the DHCP client. When the DHCP client leaves the network, the DHCP servers can assign its IP address to another DHCP client.

^{2.} At the time of writing, the ZyWALL does not support ingress bandwidth management.

In the ZyWALL, some interfaces can provide DHCP services to the network. In this case, the interface can be a DHCP relay or a DHCP server.

As a DHCP relay, the interface routes DHCP requests to DHCP servers on different networks. You can specify more than one DHCP server. If you do, the interface routes DHCP requests to all of them. It is possible for an interface to be a DHCP relay and a DHCP client simultaneously.

As a DHCP server, the interface provides the following information to DHCP clients.

• IP address - If the DHCP client's MAC address is in the ZyWALL's static DHCP table, the interface assigns the corresponding IP address. If not, the interface assigns IP addresses from a pool, defined by the starting address of the pool and the pool size.

START IP ADDRESS	POOL SIZE	RANGE OF ASSIGNED IP ADDRESS
50.50.50.33	5	50.50.50.33 - 50.50.50.37
75.75.75.1	200	75.75.75.1 - 75.75.75.200
99.99.1.1	1023	99.99.1.1 - 99.99.4.255
120.120.120.100	100	120.120.120.100 - 120.120.120.199

 Table 62
 Example: Assigning IP Addresses from a Pool

The ZyWALL cannot assign the first address (network address) or the last address (broadcast address) in the subnet defined by the interface's IP address and subnet mask. For example, in the first entry, if the subnet mask is 255.255.255.0, the ZyWALL cannot assign 50.50.50.0 or 50.50.255. If the subnet mask is 255.255.0.0, the ZyWALL cannot assign 50.50.0.0 or 50.50.255.255.0 therwise, it can assign every IP address in the range, except the interface's IP address.

If you do not specify the starting address or the pool size, the interface the maximum range of IP addresses allowed by the interface's IP address and subnet mask. For example, if the interface's IP address is 9.9.9.1 and subnet mask is 255.255.255.0, the starting IP address in the pool is 9.9.9.2, and the pool size is 253.

- Subnet mask The interface provides the same subnet mask you specify for the interface. See IP Address Assignment on page 172.
- Gateway The interface provides the same gateway you specify for the interface. See IP Address Assignment on page 172.
- DNS servers The interface provides IP addresses for up to three DNS servers that provide DNS services for DHCP clients. You can specify each IP address manually (for example, a company's own DNS server), or you can refer to DNS servers that other interfaces received from DHCP servers (for example, a DNS server at an ISP). These other interfaces have to be DHCP clients.

It is not possible for an interface to be the DHCP server and a DHCP client simultaneously.

WINS

WINS (Windows Internet Naming Service) is a Windows implementation of NetBIOS Name Server (NBNS) on Windows. It keeps track of NetBIOS computer names. It stores a mapping table of your network's computer names and IP addresses. The table is dynamically updated for IP addresses assigned by DHCP. This helps reduce broadcast traffic since computers can query the server instead of broadcasting a request for a computer name's IP address. In this way WINS is similar to DNS, although WINS does not use a hierarchy (unlike DNS). A network can have more than one WINS server. Samba can also serve as a WINS server.

PPPoE/PPTP Overview

Point-to-Point Protocol over Ethernet (PPPoE, RFC 2516) and Point-to-Point Tunneling Protocol (PPTP, RFC 2637) are usually used to connect two computers over phone lines or broadband connections. PPPoE is often used with cable modems and DSL connections. It provides the following advantages:

- The access and authentication method works with existing systems, including RADIUS.
- You can access one of several network services. This makes it easier for the service provider to offer the service
- PPPoE does not usually require any special configuration of the modem.

PPTP is used to set up virtual private networks (VPN) in unsecure TCP/IP environments. It sets up two sessions.

- 1 The first one runs on TCP port 1723. It is used to start and manage the second one.
- 2 The second one uses Generic Routing Encapsulation (GRE, RFC 2890) to transfer information between the computers.

PPTP is convenient and easy-to-use, but you have to make sure that firewalls support both PPTP sessions.

Trunk

8.1 Overview

Use trunks for WAN traffic load balancing to increase overall network throughput and reliability. Load balancing divides traffic loads between multiple interfaces. This allows you to improve quality of service and maximize bandwidth utilization for multiple ISP links.

Maybe you have two Internet connections with different bandwidths. You could set up a trunk that uses spillover or weighted round robin load balancing so time-sensitive traffic (like video) usually goes through the higher-bandwidth interface. For other traffic, you might want to use least load first load balancing to even out the distribution of the traffic load.

Suppose ISP A has better connections to Europe while ISP B has better connections to Australia. You could use policy routes and trunks to have traffic for your European branch office primarily use ISP A and traffic for your Australian branch office primarily use ISP B.

Or maybe one of the ZyWALL's interfaces is connected to an ISP that is also your Voice over IP (VoIP) service provider. You can use policy routing to send the VoIP traffic through a trunk with the interface connected to the VoIP service provider set to active and another interface (connected to another ISP) set to passive. This way VoIP traffic goes through the interface connected to the VoIP service provider set use the unterface connected to the VoIP traffic goes through the interface connected to the VoIP service provider set.

8.1.1 What You Can Do in this Chapter

- Use the **Trunk** summary screen (Section 8.2 on page 179) to configure link sticking and view the list of configured trunks and which load balancing algorithm each trunk uses.
- Use the **Add Trunk** screen (Section 8.2.1 on page 180) to configure the member interfaces for a trunk and the load balancing algorithm the trunk uses.
- Use the Add System Default screen (Section 8.2.2 on page 182) to configure the load balancing algorithm for the system default trunk.

8.1.2 What You Need to Know

- Add WAN interfaces to trunks to have multiple connections share the traffic load.
- If one WAN interface's connection goes down, the ZyWALL sends traffic through another member of the trunk.
- For example, you connect one WAN interface to one ISP and connect a second WAN interface to a second ISP. The ZyWALL balances the WAN traffic load between the connections. If one interface's connection goes down, the ZyWALL can automatically send its traffic through another interface.

You can also use trunks with policy routing to send specific traffic types through the best WAN interface for that type of traffic.

- If that interface's connection goes down, the ZyWALL can still send its traffic through another interface.
- You can define multiple trunks for the same physical interfaces.

Link Sticking

You can have the ZyWALL send each local computer's traffic that is going to the same destination through a single WAN interface for a specified period of time. This is useful when a server requires authentication. For example, the ZyWALL sends a user's traffic through one WAN IP address when he logs into a server B. If the user's subsequent sessions came from a different WAN IP address, the server would deny them. Here is an example.



Load Balancing Algorithms

The following sections describe the load balancing algorithms the ZyWALL can use to decide which interface the traffic (from the LAN) should use for a session³. The available bandwidth you configure on the ZyWALL refers to the actual bandwidth provided by the ISP and the measured bandwidth refers to the bandwidth an interface is currently using.

Least Load First

The least load first algorithm uses the current (or recent) outbound bandwidth utilization of each trunk member interface as the load balancing index(es) when making decisions about to which interface a new session is to be distributed. The outbound bandwidth utilization is defined as the measured outbound throughput over the available outbound bandwidth.

Here the ZyWALL has two WAN interfaces connected to the Internet. The configured available outbound bandwidths for WAN 1 and WAN 2 are 512K and 256K respectively.

^{3.} In the load balancing section, a session may refer to normal connection-oriented, UDP or SNMP2 traffic.





The outbound bandwidth utilization is used as the load balancing index. In this example, the measured (current) outbound throughput of WAN 1 is 412K and WAN 2 is 198K. The ZyWALL calculates the load balancing index as shown in the table below.

Since WAN 2 has a smaller load balancing index (meaning that it is less utilized than WAN 1), the ZyWALL will send the subsequent new session traffic through WAN 2.

	OUTBOUND	LOAD BALANCING INDEX		
INTERFACE	AVAILABLE (A)	MEASURED (M)	(M/A)	
WAN 1	512 K	412 K	0.8	
WAN 2	256 K	198 К	0.77	

Table 63	Least	Load	First	Exam	ple
	Loust	Load	1 11 31	Enann	pie

Weighted Round Robin

Round Robin scheduling services queues on a rotating basis and is activated only when an interface has more traffic than it can handle. A queue is given an amount of bandwidth irrespective of the incoming traffic on that interface. This queue then moves to the back of the list. The next queue is given an equal amount of bandwidth, and then moves to the end of the list; and so on, depending on the number of queues being used. This works in a looping fashion until a queue is empty.

The Weighted Round Robin (WRR) algorithm is best suited for situations when the bandwidths set for the two WAN interfaces are different. Similar to the Round Robin (RR) algorithm, the Weighted Round Robin (WRR) algorithm sets the ZyWALL to send traffic through each WAN interface in turn. In addition, the WAN interfaces are assigned weights. An interface with a larger weight gets more chances to transmit traffic than an interface with a smaller weight.

For example, in the figure below, the configured available bandwidth of WAN1 is 1M and WAN2 is 512K. You can set the ZyWALL to distribute the network traffic between the two interfaces by setting the weight of wan1 and wan2 to 2 and 1 respectively. The ZyWALL assigns the traffic of two sessions to wan1 and one session's traffic to wan2 in each round of 3 new sessions.

Figure 102 Weighted Round Robin Algorithm Example



Spillover

The spillover load balancing algorithm sends network traffic to the first interface in the trunk member list until the interface's maximum allowable load is reached, then sends the excess network traffic of new sessions to the next interface in the trunk member list. This continues as long as there are more member interfaces and traffic to be sent through them.

Suppose the first trunk member interface uses an unlimited access Internet connection and the second is billed by usage. Spillover load balancing only uses the second interface when the traffic load exceeds the threshold on the first interface. This fully utilizes the bandwidth of the first interface to reduce Internet usage fees and avoid overloading the interface.





8.2 The Trunk Summary Screen

Click **Configuration > Network > Interface > Trunk** to open the **Trunk** screen. This screen lists the configured trunks and the load balancing algorithm that each is configured to use.

Figure 104 Configuration > Network > Interface > Trunk

Port Grouping Ethernet PPP	Cellular Tunnel VLAN Bridge Trunk	
Show Advanced Settings		
Configuration		
Enable Link Sticking 1 Timeout:	300 (30-600 seconds) 🚺	
Disconnect Connections Before F	ialling Back 🚺	
Default WAN Trunk		
Default Trunk Selection SYSTEM_DEFAULT_WAN_TR 	UNK	
User Configured Trunk Please select one		
User Configuration		
💽 Add 📝 Edit 🍵 Remove 📠 C	bject Reference	
# Name	Algorithm	
14 4 Page 1 of 1 > >	Show 50 v items	No data to display
System Default		
📝 Edit 📴 Object Reference		
# Name	Algorithm	
1 SYSTEM_DEFAULT_WAN_T	RUNK IIf	
4 4 Page 1 of 1 ▶ ▶	Show 50 🗸 items	Displaying 1 - 1 of 1

The following table describes the items in this screen.

LABEL	DESCRIPTION
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
Enable Link Sticking	Enable link sticking to have the system route sessions from one source to the same destination through the same link for a period of time. This is useful for accessing server that are incompatible with a user's sessions coming from different links.
	For example, this is useful when a server requires authentication.
	This setting applies when you use load balancing and have multiple WAN interfaces set to active mode.
Timeout	Specify the time period during which sessions from one source to the same destination are to use the same link.
Disconnect Connections Before Falling Back	Select this to terminate existing connections on an interface which is set to passive mode when any interface set to active mode in the same trunk comes back up.
Enable Default SNAT	Select this to have the ZyWALL use the IP address of the outgoing interface as the source IP address of the packets it sends out through its WAN trunks. The ZyWALL automatically adds SNAT settings for traffic it routes from internal interfaces to external interfaces.
Default Trunk Selection	Select whether the ZyWALL is to use the default system WAN trunk or one of the user configured WAN trunks as the default trunk for routing traffic from internal interfaces to external interfaces.
User Configuration / System Default	The ZyWALL automatically adds all external interfaces into the pre-configured system default SYSTEM_DEFAULT_WAN_TRUNK . You cannot delete it. You can create your own User Configuration trunks and customize the algorithm, member interfaces and the active/passive mode.
Add	Click this to create a new user-configured trunk.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove a user-configured trunk, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with any interface.
Name	This field displays the label that you specified to identify the trunk.
Algorithm	This field displays the load balancing method the trunk is set to use.
Apply	Click this button to save your changes to the ZyWALL.
Reset	Click this button to return the screen to its last-saved settings.

Table 64 Configuration > Network > Interface > Trunk

8.2.1 Configuring a User-Defined Trunk

Click **Configuration > Network > Interface > Trunk**, in the **User Configuration** table click the **Add** (or **Edit**) icon to open the **following** screen. Use this screen to create or edit a WAN trunk entry.
.oad Balancing Algorithm:	Least Load First	~	
Load Balancing Index(es):	Outbound	~	
# Member	Mode	Egress Bandwidth	

Figure 105 Configuration > Network > Interface > Trunk > Add (or Edit)

Each field is described in the table below.

Table 65	Configuration >	Network >	Interface >	Trunk > Add ((or Edit)
----------	-----------------	-----------	-------------	---------------	-----------

LABEL	DESCRIPTION
Name	This is read-only if you are editing an existing trunk. When adding a new trunk, enter a descriptive name for this trunk. You may use 1-31 alphanumeric characters, underscores (_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Load Balancing	Select a load balancing method to use from the drop-down list box.
Algorithm	Select Weighted Round Robin to balance the traffic load between interfaces based on their respective weights. An interface with a larger weight gets more chances to transmit traffic than an interface with a smaller weight. For example, if the weight ratio of wan1 and wan2 interfaces is 2:1, the ZyWALL chooses wan1 for 2 sessions' traffic and wan2 for 1 session's traffic in each round of 3 new sessions.
	Select Least Load First to send new session traffic through the least utilized trunk member.
	Select Spillover to send network traffic through the first interface in the group member list until there is enough traffic that the second interface needs to be used (and so on).
Load Balancing	This field is available if you selected to use the Least Load First or Spillover method.
Index(es)	Select Outbound , Inbound , or Outbound + Inbound to set the traffic to which the ZyWALL applies the load balancing method. Outbound means the traffic traveling from an internal interface (ex. LAN) to an external interface (ex. WAN). Inbound means the opposite.
	The table lists the trunk's member interfaces. You can add, edit, remove, or move entries for user configured trunks.
Add	Click this to add a member interface to the trunk. Select an interface and click Add to add a new member interface after the selected member interface.
Edit	Select an entry and click Edit to modify the entry's settings.
Remove	To remove a member interface, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Move	To move an interface to a different number in the list, click the Move icon. In the field that appears, specify the number to which you want to move the interface.
#	This column displays the priorities of the group's interfaces. The order of the interfaces in the list is important since they are used in the order they are listed.

LABEL	DESCRIPTION
Member	Click this table cell and select an interface to be a group member.
	If you select an interface that is part of another Ethernet interface, the ZyWALL does not send traffic through the interface as part of the trunk. For example, if you have physical port 5 in the ge2 representative interface, you must select interface ge2 in order to send traffic through port 5 as part of the trunk. If you select interface ge5 as a member here, the ZyWALL will not send traffic through port 5 as part of the trunk.
Mode	Click this table cell and select Active to have the ZyWALL always attempt to use this connection.
	Select Passive to have the ZyWALL only use this connection when all of the connections set to active are down. You can only set one of a group's interfaces to passive mode.
Weight	This field displays with the weighted round robin load balancing algorithm. Specify the weight $(1 \sim 10)$ for the interface. The weights of the different member interfaces form a ratio. This ratio determines how much traffic the ZyWALL assigns to each member interface. The higher an interface's weight is (relative to the weights of the interfaces), the more sessions that interface should handle.
Ingress Bandwidth	This is reserved for future use.
	This field displays with the least load first load balancing algorithm. It displays the maximum number of kilobits of data the ZyWALL is to allow to come in through the interface per second.
	Note: You can configure the bandwidth of an interface in the corresponding interface edit screen.
Egress Bandwidth	This field displays with the least load first or spillover load balancing algorithm. It displays the maximum number of kilobits of data the ZyWALL is to send out through the interface per second.
	Note: You can configure the bandwidth of an interface in the corresponding interface edit screen.
Spillover	This field displays with the spillover load balancing algorithm. Specify the maximum bandwidth of traffic in kilobits per second (1~1048576) to send out through the interface before using another interface. When this spillover bandwidth limit is exceeded, the ZyWALL sends new session traffic through the next interface. The traffic of existing sessions still goes through the interface on which they started.
	The ZyWALL uses the group member interfaces in the order that they are listed.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 65 Configuration > Network > Interface > Trunk > Add (or Edit) (continued)

8.2.2 Configuring the System Default Trunk

In the **Configuration > Network > Interface > Trunk** screen and the **System Default** section, select the default trunk entry and click **Edit** to open the **following** screen. Use this screen to change the load balancing algorithm and view the bandwidth allocations for each member interface.

Note: The available bandwidth is allocated to each member interface equally and is not allowed to be changed for the default trunk.

Name	B:		SYSTEM_DEFAULT_WAN_T	RUNK	
load	Balancing Algorithm	1:	Least Load First	*	
#	Member	Mode	Ingress Bandwidth		Egress Bandwidth
1	wan1	Active	1048576 kbps		1048576 kbps
2	wan2	Active	1048576 kbps		1048576 kbps
3	wan1_ppp	Active	1048576 kbps		1048576 kbps
4	wan2_ppp	Active	1048576 kbps		1048576 kbps
14	4 Page 1 o	f1 🕨 🕅 S	how 50 🗸 items		Displaying 1 - 4 of 4

Figure 106 Configuration > Network > Interface > Trunk > Edit (System Default)

Each field is described in the table below.

LABEL	DESCRIPTION
Name	This field displays the name of the selected system default trunk.
Load Balancing	Select the load balancing method to use for the trunk.
Algorithm	Select Weighted Round Robin to balance the traffic load between interfaces based on their respective weights. An interface with a larger weight gets more chances to transmit traffic than an interface with a smaller weight. For example, if the weight ratio of wan1 and wan2 interfaces is 2:1, the ZyWALL chooses wan1 for 2 sessions' traffic and wan2 for 1 session's traffic in each round of 3 new sessions.
	Select Least Load First to send new session traffic through the least utilized trunk member.
	Select Spillover to send network traffic through the first interface in the group member list until there is enough traffic that the second interface needs to be used (and so on).
	The table lists the trunk's member interfaces. This table is read-only.
#	This column displays the priorities of the group's interfaces. The order of the interfaces in the list is important since they are used in the order they are listed.
Member	This column displays the name of the member interfaces.
Mode	This field displays Active if the ZyWALL always attempt to use this connection.
	This field displays Passive if the ZyWALL only use this connection when all of the connections set to active are down. Only one of a group's interfaces can be set to passive mode.
Weight	This field displays with the weighted round robin load balancing algorithm. Specify the weight $(1 \sim 10)$ for the interface. The weights of the different member interfaces form a ratio. s
Ingress Bandwidth	This is reserved for future use.
	This field displays with the least load first load balancing algorithm. It displays the maximum number of kilobits of data the ZyWALL is to allow to come in through the interface per second.
Egress Bandwidth	This field displays with the least load first or spillover load balancing algorithm. It displays the maximum number of kilobits of data the ZyWALL is to send out through the interface per second.

	a a b		• • • •			_				10			. 、
Table 66	Configuration	>	Network >	>	Interface >	•	Irunk >	>	Edit	(Si	ystem	Defaul	t)

LABEL	DESCRIPTION
Spillover	This field displays with the spillover load balancing algorithm. Specify the maximum bandwidth of traffic in kilobits per second (1~1048576) to send out through the interface before using another interface. When this spillover bandwidth limit is exceeded, the ZyWALL sends new session traffic through the next interface. The traffic of existing sessions still goes through the interface on which they started. The ZyWALL uses the group member interfaces in the order that they are listed.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 66Configuration > Network > Interface > Trunk > Edit (System Default) (continued)

Policy and Static Routes

9.1 Policy and Static Routes Overview

Use policy routes and static routes to override the ZyWALL's default routing behavior in order to send packets through the appropriate interface or VPN tunnel.

For example, the next figure shows a computer (**A**) connected to the ZyWALL's LAN interface. The ZyWALL routes most traffic from **A** to the Internet through the ZyWALL's default gateway (**R1**). You create one policy route to connect to services offered by your ISP behind router **R2**. You create another policy route to communicate with a separate network behind another router (**R3**) connected to the LAN.





Note: You can generally just use policy routes. You only need to use static routes if you have a large network with multiple routers where you use RIP or OSPF to propagate routing information to other routers.

9.1.1 What You Can Do in this Chapter

- Use the Policy Route screens (see Section 9.2 on page 187) to list and configure policy routes.
- Use the Static Route screens (see Section 9.3 on page 193) to list and configure static routes.

9.1.2 What You Need to Know

Policy Routing

Traditionally, routing is based on the destination address only and the ZyWALL takes the shortest path to forward a packet. IP Policy Routing (IPPR) provides a mechanism to override the default routing behavior and alter the packet forwarding based on the policy defined by the network administrator. Policy-based routing is applied to incoming packets on a per interface basis, prior to the normal routing.

How You Can Use Policy Routing

- Source-Based Routing Network administrators can use policy-based routing to direct traffic from different users through different connections.
- Bandwidth Shaping You can allocate bandwidth to traffic that matches routing policies and prioritize traffic. You can also use policy routes to manage other types of traffic (like ICMP traffic) and send traffic through VPN tunnels.
- Cost Savings IPPR allows organizations to distribute interactive traffic on high-bandwidth, highcost paths while using low-cost paths for batch traffic.
- Load Sharing Network administrators can use IPPR to distribute traffic among multiple paths.
- NAT The ZyWALL performs NAT by default for traffic going to or from the WAN interfaces. A
 routing policy's SNAT allows network administrators to have traffic received on a specified
 interface use a specified IP address as the source IP address.
- Note: The ZyWALL automatically uses SNAT for traffic it routes from internal interfaces to external interfaces. For example LAN to WAN traffic.

Static Routes

The ZyWALL usually uses the default gateway to route outbound traffic from computers on the LAN to the Internet. To have the ZyWALL send data to devices not reachable through the default gateway, use static routes. Configure static routes if you need to use RIP or OSPF to propagate the routing information to other routers. See Chapter 10 on page 197 for more on RIP and OSPF.

Policy Routes Versus Static Routes

- Policy routes are more flexible than static routes. You can select more criteria for the traffic to match and can also use schedules, NAT, and bandwidth management.
- Policy routes are only used within the ZyWALL itself. Static routes can be propagated to other routers using RIP or OSPF.
- Policy routes take priority over static routes. If you need to use a routing policy on the ZyWALL and propagate it to other routers, you could configure a policy route and an equivalent static route.

DiffServ

QoS is used to prioritize source-to-destination traffic flows. All packets in the same flow are given the same priority. CoS (class of service) is a way of managing traffic in a network by grouping similar types of traffic together and treating each type as a class. You can use CoS to give different priorities to different packet types. DiffServ (Differentiated Services) is a class of service (CoS) model that marks packets so that they receive specific per-hop treatment at DiffServ-compliant network devices along the route based on the application types and traffic flow. Packets are marked with DiffServ Code Points (DSCPs) indicating the level of service desired. This allows the intermediary DiffServ-compliant network devices to handle the packets differently depending on the code points without the need to negotiate paths or remember state information for every flow. In addition, applications do not have to request a particular service or give advanced notice of where the traffic is going.

DSCP Marking and Per-Hop Behavior

DiffServ defines a new DS (Differentiated Services) field to replace the Type of Service (TOS) field in the IP header. The DS field contains a 2-bit unused field and a 6-bit DSCP field which can define up to 64 service levels. The following figure illustrates the DS field.

DSCP (6 bits) Unused (2 bits)

DSCP is backward compatible with the three precedence bits in the ToS octet so that non-DiffServ compliant, ToS-enabled network device will not conflict with the DSCP mapping.

The DSCP value determines the forwarding behavior, the PHB (Per-Hop Behavior), that each packet gets across the DiffServ network. Based on the marking rule, different kinds of traffic can be marked for different kinds of forwarding. Resources can then be allocated according to the DSCP values and the configured policies.

Finding Out More

• See Section 9.4 on page 195 for more background information on policy routing.

9.2 Policy Route Screen

Click **Configuration > Network > Routing** to open the **Policy Route** screen. Use this screen to see the configured policy routes and turn policy routing based bandwidth management on or off.

A policy route defines the matching criteria and the action to take when a packet meets the criteria. The action is taken only when all the criteria are met. The criteria can include the user name, source address and incoming interface, destination address, schedule, IP protocol (ICMP, UDP, TCP, etc.) and port.

The actions that can be taken include:

- Routing the packet to a different gateway, outgoing interface, VPN tunnel, or trunk.
- Limiting the amount of bandwidth available and setting a priority for traffic.

IPPR follows the existing packet filtering facility of RAS in style and in implementation.

If you enabled IPv6 in the **Configuration** > **System** > **IPv6** screen, you can also configure policy routes used for your IPv6 networks on this screen.

Figure 108 Configuration > Network > Routing > Policy Route

O Ac	dd 📝 Edit	💼 Remove 🛛 🤤	Activate (🖗 Inactivate 🛓	Move						
# ^ :	Status User	Schedule	Incoming	Source	Destination	DSCP Code	Service	Source P	Next-Hop	DSCP Ma	SNAT
	a 18 - 6			PROFESSION NO.							
14 ×	nfiguration	of1 ▶	▶ Show	50 v items	5						No data to displa
76 Co	9 Page 1 onfiguration dd 28Edit	of 1 🕨	Activate (50 vitems	s Move						No data to displa
4 ≤ 76 Co ② Ac # ▲ !	nfiguration dd Zedit Status Use	r Schedule	Activate (50 v item Jinactivate d Source	s Move Dest	ination DS	CP Code	Service	Source Port	Next-Hop	No data to displa

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
IPv4 Configuration / IPv6 Configuration	Use the IPv4 Configuration section for IPv4 network settings. Use the IPv6 Configuration section for IPv6 network settings if you connect your ZyWALL to an IPv6 network. Both sections have similar fields as described below.
Use Policy Route to Override Direct Route	Select this to have the ZyWALL forward packets that match a policy route according to the policy route instead of sending the packets directly to a connected network.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Move	To change a rule's position in the numbered list, select the rule and click Move to display a field to type a number for where you want to put that rule and press [ENTER] to move the rule to the number that you typed.
	The ordering of your rules is important as they are applied in order of their numbering.
#	This is the number of an individual policy route.
Status	This icon is lit when the entry is active, red when the next hop's connection is down, and dimmed when the entry is inactive.
User	This is the name of the user (group) object from which the packets are sent. any means all users.
Schedule	This is the name of the schedule object. none means the route is active at all times if enabled.
Incoming	This is the interface on which the packets are received.
Source	This is the name of the source IP address (group) object. any means all IP addresses.
Destination	This is the name of the destination IP address (group) object. any means all IP addresses.

 Table 67
 Configuration > Network > Routing > Policy Route

LABEL	DESCRIPTION
DSCP Code	This is the DSCP value of incoming packets to which this policy route applies.
	any means all DSCP values or no DSCP marker.
	default means traffic with a DSCP value of 0. This is usually best effort traffic
	The " af " entries stand for Assured Forwarding. The number following the " af " identifies one of four classes and one of three drop preferences. See Assured Forwarding (AF) PHB for DiffServ on page 195 for more details.
Service	This is the name of the service object. any means all services.
Source Port	This is the name of a service object. The ZyWALL applies the policy route to the packets sent from the corresponding service port. any means all service ports.
Next-Hop	This is the next hop to which packets are directed. It helps forward packets to their destinations and can be a router, VPN tunnel, outgoing interface or trunk.
DSCP Marking	This is how the ZyWALL handles the DSCP value of the outgoing packets that match this route. If this field displays a DSCP value, the ZyWALL applies that DSCP value to the route's outgoing packets.
	preserve means the ZyWALL does not modify the DSCP value of the route's outgoing packets.
	default means the ZyWALL sets the DSCP value of the route's outgoing packets to 0.
	The " af " choices stand for Assured Forwarding. The number following the " af " identifies one of four classes and one of three drop preferences. See Assured Forwarding (AF) PHB for DiffServ on page 195 for more details.
SNAT	This is the source IP address that the route uses.
	It displays none if the ZyWALL does not perform NAT for this route.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 67
 Configuration > Network > Routing > Policy Route (continued)

9.2.1 Policy Route Edit Screen

Click **Configuration > Network > Routing** to open the **Policy Route** screen. Then click the **Add** or **Edit** icon in the **IPv4 Configuration** or **IPv6 Configuration** section. The **Add Policy Route** or **Policy Route Edit** screen opens. Use this screen to configure or edit a policy route. Both IPv4 and IPv6 policy route have similar settings except the **Address Translation** (SNAT) settings.

Add Policy Route			?
Show Advanced Settings ह Create new Obj	ect▼		
Configuration			
V Enable			
Description:		(Optional)	
Criteria			
User:	any	*	
Incoming:	any (Excluding ZyWALL)	~	
Source Address:	any	~	
Destination Address:	any	*	
DSCP Code:	any	~	
Schedule:	none	~	
Service:	any	*	
Next-Hop			
Туре:	Auto	~	
DSCP Marking	25	<i>0</i>	
DSCP Marking:	preserve	*	
Address Translation			
Source Network Address Translation:	outgoing-interface	*	
			•

Figure 109 Configuration > Network > Routing > Policy Route > Add/Edit (IPv4 Configuration)

		<u> </u>	
Add Policy Route			?
🔢 Show Advanced Settings 🔚 Create i	new Object -		
Configuration			
I Enable			
Description:		(Optional)	
Criteria			
User:	any	~	
Incoming:	any (Excluding ZyWALL)	*	
Source Address:	any	~	
Destination Address:	any	~	
	city .		
DSCP Code:	any		
Schedule:	none	~	
Service:	any	~	
Source Port:	any	~	
Next-Hop			
Туре:	Auto	¥	
DSCP Marking			
DSCP Marking:	preserve	v	
			OK Cancel

Figure 110 Configuration > Network > Routing > Policy Route > Add/Edit (IPv6 Configuration)

The following table describes the labels in this screen.

Table 68	Configuration	> Netwo	ork >	Routing	>	Policy	Route >	Add/Edit
----------	---------------	---------	-------	---------	---	--------	---------	----------

LABEL	DESCRIPTION
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
Create new Object	Use this to configure any new settings objects that you need to use in this screen.
Configuration	
Enable	Select this to activate the policy.
Description	Enter a descriptive name of up to 31 printable ASCII characters for the policy.
Criteria	
User	Select a user name or user group from which the packets are sent.
Incoming	Select where the packets are coming from; any, an interface, a tunnel, an SSL VPN, or the ZyWALL itself. For an interface, a tunnel, or an SSL VPN, you also need to select the individual interface, VPN tunnel, or SSL VPN connection.
Source Address	Select a source IP address object from which the packets are sent.
Destination Address	Select a destination IP address object to which the traffic is being sent. If the next hop is a dynamic VPN tunnel and you enable Auto Destination Address , the ZyWALL uses the local network of the peer router that initiated an incoming dynamic IPSec tunnel as the destination address of the policy instead of your configuration here.

LABEL	DESCRIPTION
DSCP Code	Select a DSCP code point value of incoming packets to which this policy route applies or select User Define to specify another DSCP code point. The lower the number the higher the priority with the exception of 0 which is usually given only best-effort treatment.
	any means all DSCP value or no DSCP marker.
	default means traffic with a DSCP value of 0. This is usually best effort traffic
	The " af " choices stand for Assured Forwarding. The number following the " af " identifies one of four classes and one of three drop preferences. See Assured Forwarding (AF) PHB for DiffServ on page 195 for more details.
User-Defined DSCP Code	Use this field to specify a custom DSCP code point.
Schedule	Select a schedule to control when the policy route is active. none means the route is active at all times if enabled.
Service	Select a service or service group to identify the type of traffic to which this policy route applies.
Source Port	Select a service or service group to identify the source port of packets to which the policy route applies.
Next-Hop	
Туре	Select Auto to have the ZyWALL use the routing table to find a next-hop and forward the matched packets automatically.
	Select Gateway to route the matched packets to the next-hop router or switch you specified in the Gateway field. You have to set up the next-hop router or switch as a HOST address object first.
	Select VPN Tunnel to route the matched packets via the specified VPN tunnel.
	Select Trunk to route the matched packets through the interfaces in the trunk group based on the load balancing algorithm.
	Select Interface to route the matched packets through the specified outgoing interface to a gateway (which is connected to the interface).
Gateway	This field displays when you select Gateway in the Type field. Select a HOST address object. The gateway is an immediate neighbor of your ZyWALL that will forward the packet to the destination. The gateway must be a router or switch on the same segment as your ZyWALL's interface(s).
VPN Tunnel	This field displays when you select VPN Tunnel in the Type field. Select a VPN tunnel through which the packets are sent to the remote network that is connected to the ZyWALL directly.
Auto Destination Address	This field displays when you select VPN Tunnel in the Type field. Select this to have the ZyWALL use the local network of the peer router that initiated an incoming dynamic IPSec tunnel as the destination address of the policy.
	Leave this cleared if you want to manually specify the destination address.
Trunk	This field displays when you select Trunk in the Type field. Select a trunk group to have the ZyWALL send the packets via the interfaces in the group.
Interface	This field displays when you select Interface in the Type field. Select an interface to have the ZyWALL send traffic that matches the policy route through the specified interface.
Auto-Disable	This field displays when you select Interface or Trunk in the Type field. Select this to have the ZyWALL automatically disable this policy route when the next hop's connection is down.

 Table 68
 Configuration > Network > Routing > Policy Route > Add/Edit (continued)

LABEL	DESCRIPTION					
DSCP Marking	Set how the ZyWALL handles the DSCP value of the outgoing packets that match this route.					
	Select one of the pre-defined DSCP values to apply or select User Define to specify another DSCP value. The " af " choices stand for Assured Forwarding. The number following the " af " identifies one of four classes and one of three drop preferences. See Assured Forwarding (AF) PHB for DiffServ on page 195 for more details.					
	Select preserve to have the ZyWALL keep the packets' original DSCP value.					
	Select default to have the ZyWALL set the DSCP value of the packets to 0.					
User-Defined DSCP Code	Use this field to specify a custom DSCP value.					
Address Translation	Use this section to configure NAT for the policy route. This section does not apply to policy routes that use a VPN tunnel as the next hop.					
Source Network	Select none to not use NAT for the route.					
	Select outgoing-interface to use the IP address of the outgoing interface as the source IP address of the packets that matches this route.					
	To use SNAT for a virtual interface that is in the same WAN trunk as the physical interface to which the virtual interface is bound, the virtual interface and physical interface must be in different subnets.					
	Otherwise, select a pre-defined address (group) to use as the source IP address(es) of the packets that match this route.					
	Use Create new Object if you need to configure a new address (group) to use as the source IP address(es) of the packets that match this route.					
ОК	Click OK to save your changes back to the ZyWALL.					
Cancel	Click Cancel to exit this screen without saving.					

 Table 68
 Configuration > Network > Routing > Policy Route > Add/Edit (continued)

9.3 IP Static Route Screen

Click **Configuration > Network > Routing > Static Route** to open the **Static Route** screen. This screen displays the configured static routes. Configure static routes to be able to use RIP or OSPF to propagate the routing information to other routers. If you enabled IPv6 in the **Configuration > System > IPv6** screen, you can also configure static routes used for your IPv6 networks on this screen.



Figure 111 Configuration > Network > Routing > Static Route

The following table describes the labels in this screen.

LABEL	DESCRIPTION
IPv4 Configuration / IPv6 Configuration	Use the IPv4 Configuration section for IPv4 network settings. Use the IPv6 Configuration section for IPv6 network settings if you connect your ZyWALL to an IPv6 network. Both sections have similar fields as described below.
Add	Click this to create a new static route.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
#	This is the number of an individual static route.
Destination	This is the destination IP address.
Subnet Mask	This is the IP subnet mask.
Prefix	This is the IPv6 prefix for the destination IP address.
Next-Hop	This is the IP address of the next-hop gateway or the interface through which the traffic is routed. The gateway is a router or switch on the same segment as your ZyWALL's interface(s). The gateway helps forward packets to their destinations.
Metric	This is the route's priority among the ZyWALL's routes. The smaller the number, the higher priority the route has.

Table 69	Configuration ~	>	Network	>	Routina	>	Static F	2011te
	conniguration 2	~	NELWOIK	~	Routing	_	Julie r	vuie

9.3.1 Static Route Add/Edit Screen

Select a static route index number and click **Add** or **Edit**. The screen shown next appears. Use this screen to configure the required information for a static route.

Figure 112 Cor	nfiguration > 1	Vetwork >	Routing	>	Static Route	>	Add	(IPv4	Configurat	ion)
A										

Static Route Setting		? ×
Destination IP:		
Subnet Mask:		
Gateway IP		
Interface	wan1 🗸	
Metric:	0	
	ок 🔤	Cancel

Figure 113 Configuration > Network > Routing > Static Route > Add (IPv6 Configuration)

Destination IP:			~~~~~		
Prefix Length:	64			(0-128)	
Gateway IP					
) Interface	wan1		¥		
letric:	0	(0-127)			

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Destination IP	This parameter specifies the IP network address of the final destination. Routing is always based on network number.
	If you need to specify a route to a single host, enter the specific IP address here and use a subnet mask of 255.255.255.255 (for IPv4) in the Subnet Mask field or a prefix of 128 (for IPv6) in the Prefix Length field to force the network number to be identical to the host ID.
	For IPv6, if you want to send all traffic to the gateway or interface specified in the Gateway IP or Interface field, enter :: in this field and 0 in the Prefix Length field.
Subnet Mask	Enter the IP subnet mask here.
Prefix Length	Enter the number of left-most digits in the destination IP address, which indicates the network prefix. Enter :: in the Destination IP field and 0 in this field if you want to send all traffic to the gateway or interface specified in the Gateway IP or Interface field.
Gateway IP	Select the radio button and enter the IP address of the next-hop gateway. The gateway is a router or switch on the same segment as your ZyWALL's interface(s). The gateway helps forward packets to their destinations.
Interface	Select the radio button and a predefined interface through which the traffic is sent.
Metric	Metric represents the "cost" of transmission for routing purposes. IP routing uses hop count as the measurement of cost, with a minimum of 1 for directly connected networks. Enter a number that approximates the cost for this link. The number need not be precise, but it must be 0~127. In practice, 2 or 3 is usually a good number.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 70 Configuration > Network > Routing > Static Route > Add

9.4 Policy Routing Technical Reference

Here is more detailed information about some of the features you can configure in policy routing.

NAT and SNAT

NAT (Network Address Translation - NAT, RFC 1631) is the translation of the IP address in a packet in one network to a different IP address in another network. Use SNAT (Source NAT) to change the source IP address in one network to a different IP address in another network.

Assured Forwarding (AF) PHB for DiffServ

Assured Forwarding (AF) behavior is defined in RFC 2597. The AF behavior group defines four AF classes. Inside each class, packets are given a high, medium or low drop precedence. The drop precedence determines the probability that routers in the network will drop packets when congestion occurs. If congestion occurs between classes, the traffic in the higher class (smaller numbered class) is generally given priority. Combining the classes and drop precedence produces

the following twelve DSCP encodings from AF11 through AF43. The decimal equivalent is listed in brackets.

	CLASS 1	CLASS 2	CLASS 3	CLASS 4
Low Drop Precedence	AF11 (10)	AF21 (18)	AF31 (26)	AF41 (34)
Medium Drop Precedence	AF12 (12)	AF22 (20)	AF32 (28)	AF42 (36)
High Drop Precedence	AF13 (14)	AF23 (22)	AF33 (30)	AF43 (38)

Table 71 Assured Forwarding (AF) Behavior Group

Maximize Bandwidth Usage

The maximize bandwidth usage option allows the ZyWALL to divide up any available bandwidth on the interface (including unallocated bandwidth and any allocated bandwidth that a policy route is not using) among the policy routes that require more bandwidth.

When you enable maximize bandwidth usage, the ZyWALL first makes sure that each policy route gets up to its bandwidth allotment. Next, the ZyWALL divides up an interface's available bandwidth (bandwidth that is unbudgeted or unused by the policy routes) depending on how many policy routes require more bandwidth and on their priority levels. When only one policy route requires more bandwidth, the ZyWALL gives the extra bandwidth to that policy route.

When multiple policy routes require more bandwidth, the ZyWALL gives the highest priority policy routes the available bandwidth first (as much as they require, if there is enough available bandwidth), and then to lower priority policy routes if there is still bandwidth available. The ZyWALL distributes the available bandwidth equally among policy routes with the same priority level.

Routing Protocols

10.1 Routing Protocols Overview

Routing protocols give the ZyWALL routing information about the network from other routers. The ZyWALL stores this routing information in the routing table it uses to make routing decisions. In turn, the ZyWALL can also use routing protocols to propagate routing information to other routers.

Routing protocols are usually only used in networks using multiple routers like campuses or large enterprises.

10.1.1 What You Can Do in this Chapter

- Use the **RIP** screen (see Section 10.2 on page 197) to configure the ZyWALL to use RIP to receive and/or send routing information.
- Use the **OSPF** screen (see Section 10.3 on page 199) to configure general OSPF settings and manage OSPF areas.
- Use the **OSPF Area Add/Edit** screen (see Section 10.3.2 on page 204) to create or edit an OSPF area.

10.1.2 What You Need to Know

The ZyWALL supports two standards, RIP and OSPF, for routing protocols. RIP and OSPF are compared here and discussed further in the rest of the chapter.

	RIP	OSPF
Network Size	Small (with up to 15 routers)	Large
Metric	Hop count	Bandwidth, hop count, throughput, round trip time and reliability.
Convergence	Slow	Fast

Table 72 RIP vs. OSPF

Finding Out More

See Section 10.4 on page 206 for background information on routing protocols.

10.2 The RIP Screen

RIP (Routing Information Protocol, RFC 1058 and RFC 1389) allows a device to exchange routing information with other routers. RIP is a vector-space routing protocol, and, like most such protocols, it uses hop count to decide which route is the shortest. Unfortunately, it also broadcasts

its routes asynchronously to the network and converges slowly. Therefore, RIP is more suitable for small networks (up to 15 routers).

- In the ZyWALL, you can configure two sets of RIP settings before you can use it in an interface.
- First, the **Authentication** field specifies how to verify that the routing information that is received is the same routing information that is sent. This is discussed in more detail in Authentication Types on page 207.
- Second, the ZyWALL can also **redistribute** routing information from non-RIP networks, specifically OSPF networks and static routes, to the RIP network. Costs might be calculated differently, however, so you use the **Metric** field to specify the cost in RIP terms.
- RIP uses UDP port 520.

Use the **RIP** screen to specify the authentication method and maintain the policies for redistribution.

Click Configuration > Network > Routing > RIP to open the following screen.

Authentication:	MD5	
MD5 Authentication ID:	(1255)	
MD5 Authentication Key:		
 A second sec second second sec		
edistribute		
Active OSPF		
Metric:	0 (0-16)	
Active Static Route		
Metric:	0 (0-16)	
	191497	

Figure 114 Configuration > Network > Routing > RIP

The following table describes the labels in this screen. **Table 73** Configuration > Network > Routing Protocol > RIP

LABEL	DESCRIPTION				
Authentication					
Authentication	Select the authentication method used in the RIP network. This authentication protects the integrity, but not the confidentiality, of routing updates.				
	None uses no authentication.				
	Text uses a plain text password that is sent over the network (not very secure).				
	MD5 uses an MD5 password and authentication ID (most secure).				
Text Authentication Key	This field is available if the Authentication is Text . Type the password for text authentication. The key can consist of alphanumeric characters and the underscore, and it can be up to 16 characters long.				
MD5 Authentication ID	This field is available if the Authentication is MD5 . Type the ID for MD5 authentication. The ID can be between 1 and 255.				
MD5 Authentication Key	This field is available if the Authentication is MD5 . Type the password for MD5 authentication. The password can consist of alphanumeric characters and the underscore, and it can be up to 16 characters long.				

LABEL	DESCRIPTION
Redistribute	
Active OSPF	Select this to use RIP to advertise routes that were learned through OSPF.
Metric	Type the cost for routes provided by OSPF. The metric represents the "cost" of transmission for routing purposes. RIP routing uses hop count as the measurement of cost, with 1 usually used for directly connected networks. The number does not have to be precise, but it must be between 0 and 16. In practice, 2 or 3 is usually used.
Active Static Route	Select this to use RIP to advertise routes that were learned through the static route configuration.
Metric	Type the cost for routes provided by the static route configuration. The metric represents the "cost" of transmission for routing purposes. RIP routing uses hop count as the measurement of cost, with 1 usually used for directly connected networks. The number does not have to be precise, but it must be between 0 and 16. In practice, 2 or 3 is usually used.
Apply	Click this button to save your changes to the ZyWALL.
Reset	Click this button to return the screen to its last-saved settings.

Table 73 Configuration > Network > Routing Protocol > RIP (continued)

10.3 The OSPF Screen

OSPF (Open Shortest Path First, RFC 2328) is a link-state protocol designed to distribute routing information within a group of networks, called an Autonomous System (AS). OSPF offers some advantages over vector-space routing protocols like RIP.

- OSPF supports variable-length subnet masks, which can be set up to use available IP addresses more efficiently.
- OSPF filters and summarizes routing information, which reduces the size of routing tables throughout the network.
- OSPF responds to changes in the network, such as the loss of a router, more quickly.
- OSPF considers several factors, including bandwidth, hop count, throughput, round trip time, and reliability, when it calculates the shortest path.
- OSPF converges more quickly than RIP.

Naturally, OSPF is also more complicated than RIP, so OSPF is usually more suitable for large networks.

OSPF uses IP protocol 89.

OSPF Areas

An OSPF Autonomous System (AS) is divided into one or more areas. Each area represents a group of adjacent networks and is identified by a 32-bit ID. In OSPF, this number may be expressed as an integer or as an IP address.

There are several types of areas.

• The backbone is the transit area that routes packets between other areas. All other areas are connected to the backbone.

- A normal area is a group of adjacent networks. A normal area has routing information about the OSPF AS, any networks outside the OSPF AS to which it is directly connected, and any networks outside the OSPF AS that provide routing information to any area in the OSPF AS.
- A stub area has routing information about the OSPF AS. It does not have any routing information about any networks outside the OSPF AS, including networks to which it is directly connected. It relies on a default route to send information outside the OSPF AS.
- A Not So Stubby Area (NSSA, RFC 1587) has routing information about the OSPF AS and networks outside the OSPF AS to which the NSSA is directly connected. It does not have any routing information about other networks outside the OSPF AS.

Each type of area is illustrated in the following figure.



Figure 115 OSPF: Types of Areas

This OSPF AS consists of four areas, areas 0-3. Area 0 is always the backbone. In this example, areas 1, 2, and 3 are all connected to it. Area 1 is a normal area. It has routing information about the OSPF AS and networks X and Y. Area 2 is a stub area. It has routing information about the OSPF AS, but it depends on a default route to send information to networks X and Y. Area 3 is a NSSA. It has routing information about the OSPF AS and network Y but not about network X.

OSPF Routers

Every router in the same area has the same routing information. They do this by exchanging Hello messages to confirm which neighbor (layer-3) devices exist, and then they exchange database descriptions (DDs) to create a synchronized link-state database. The link-state database contains records of router IDs, their associated links and path costs. The link-state database is then constantly updated through Link State Advertisements (LSA). Each router uses the link state database and the Dijkstra algorithm to compute the least cost paths to network destinations.

Like areas, each router has a unique 32-bit ID in the OSPF AS, and there are several types of routers. Each type is really just a different role, and it is possible for one router to play multiple roles at one time.

- An internal router (IR) only exchanges routing information with other routers in the same area.
- An Area Border Router (ABR) connects two or more areas. It is a member of all the areas to which it is connected, and it filters, summarizes, and exchanges routing information between them.

• An Autonomous System Boundary Router (ASBR) exchanges routing information with routers in networks outside the OSPF AS. This is called redistribution in OSPF.

SOURCE \ TYPE OF AREA	NORMAL	NSSA	STUB
Static routes	Yes	Yes	No
RIP	Yes	Yes	Yes

Table 74 OSPF: Redistribution from Other Sources to Each Type of Area

• A backbone router (BR) has at least one interface with area 0. By default, every router in area 0 is a backbone router, and so is every ABR.

Each type of router is illustrated in the following example.





In order to reduce the amount of traffic between routers, a group of routers that are directly connected to each other selects a designated router (DR) and a backup designated router (BDR). All of the routers only exchange information with the DR and the BDR, instead of exchanging information with all of the other routers in the group. The DR and BDR are selected by priority; if two routers have the same priority, the highest router ID is used.

The DR and BDR are selected in each group of routers that are directly connected to each other. If a router is directly connected to several groups, it might be a DR in one group, a BDR in another group, and neither in a third group all at the same time.

Virtual Links

In some OSPF AS, it is not possible for an area to be directly connected to the backbone. In this case, you can create a virtual link through an intermediate area to logically connect the area to the backbone. This is illustrated in the following example.

Figure 117 OSPF: Virtual Link



In this example, area 100 does not have a direct connection to the backbone. As a result, you should set up a virtual link on both ABR in area 10. The virtual link becomes the connection between area 100 and the backbone.

You cannot create a virtual link to a router in a different area.

OSPF Configuration

Follow these steps when you configure OSPF on the ZyWALL.

- 1 Enable OSPF.
- 2 Set up the OSPF areas.
- **3** Configure the appropriate interfaces. See Section 7.3.1 on page 110.
- 4 Set up virtual links, as needed.

10.3.1 Configuring the OSPF Screen

Use the first OSPF screen to specify the OSPF router the ZyWALL uses in the OSPF AS and maintain the policies for redistribution. In addition, it provides a summary of OSPF areas, allows you to remove them, and opens the **OSPF Add/Edit** screen to add or edit them.

Click **Configuration > Network > Routing > OSPF** to open the following screen.

eneral Settings					
OSPF Router ID (IP Add	lress):	Default	~		
edistribute					
Active RIP					
Type:		~	Metric:	(0-16777214)	
Active Static Route					
Type:		~	Metric:	(0-16777214)	
rea					
🗿 Add 📝 Edit 🍵	Remove				
# Area		Туре		Authentication	
14 4 Page 1	of 1 🕨 🕅	Show 50 🗸 it	ems		No data to displa

Figure 118 Configuration > Network > Routing > OSPF

The following table describes the labels in this screen. See Section 10.3.2 on page 204 for more information as well.

	5
LABEL	DESCRIPTION
OSPF Router ID	Select the 32-bit ID the ZyWALL uses in the OSPF AS.
	Default - the first available interface IP address is the ZyWALL's ID.
	User Defined - enter the ID (in IP address format) in the field that appears when you select User Define.
Redistribute	
Active RIP	Select this to advertise routes that were learned from RIP. The ZyWALL advertises routes learned from RIP to Normal and NSSA areas but not to Stub areas.
Туре	Select how OSPF calculates the cost associated with routing information from RIP. Choices are: Type 1 and Type 2 .
	Type 1 - cost = OSPF AS cost + external cost (Metric)
	Type 2 - cost = external cost (Metric); the OSPF AS cost is ignored.
Metric	Type the external cost for routes provided by RIP. The metric represents the "cost" of transmission for routing purposes. The way this is used depends on the Type field. This value is usually the average cost in the OSPF AS, and it can be between 1 and 16777214.
Active Static Route	Select this to advertise routes that were learned from static routes. The ZyWALL advertises routes learned from static routes to all types of areas.
Туре	Select how OSPF calculates the cost associated with routing information from static routes. Choices are: Type 1 and Type 2 .
	Type 1 - cost = OSPF AS cost + external cost (Metric)
	Type 2 - cost = external cost (Metric); the OSPF AS cost is ignored.
Metric	Type the external cost for routes provided by static routes. The metric represents the "cost" of transmission for routing purposes. The way this is used depends on the Type field. This value is usually the average cost in the OSPF AS, and it can be between 1 and 16777214.
Area	This section displays information about OSPF areas in the ZyWALL.

 Table 75
 Configuration > Network > Routing Protocol > OSPF

LABEL	DESCRIPTION
Add	Click this to create a new OSPF area.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
#	This field is a sequential value, and it is not associated with a specific area.
Area	This field displays the 32-bit ID for each area in IP address format.
Туре	This field displays the type of area. This type is different from the Type field above.
Authentication	This field displays the default authentication method in the area.
Apply	Click this button to save your changes to the ZyWALL.
Reset	Click this button to return the screen to its last-saved settings.

Table 75 Configuration > Network > Routing Protocol > OSPF (continued)

10.3.2 OSPF Area Add/Edit Screen

The **OSPF Area Add/Edit** screen allows you to create a new area or edit an existing one. To access this screen, go to the **OSPF** summary screen (see Section 10.3 on page 199), and click either the **Add** icon or an **Edit** icon.

	ID;			
Туре	4	Normal	*	
Auth	entication:	MD5	*	
MD5 Authentication ID:			(1-255)	
MD5	Authentication Key:			
0				
0	Add 🛃 Edit 🔟 Remo	ove		
*	Add 🧭 Edit 📕 Remo	ove	Authentication	
* 1	Add Ade Ldit Markeno Peer Router ID 1.4.5.8	ove	Authentication	

Figure 119 Configuration > Network > Routing > OSPF > Add

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Area ID	Type the unique, 32-bit identifier for the area in IP address format.
Туре	Select the type of OSPF area.
	Normal - This area is a normal area. It has routing information about the OSPF AS and about networks outside the OSPF AS.
	Stub - This area is an stub area. It has routing information about the OSPF AS but not about networks outside the OSPF AS. It depends on a default route to send information outside the OSPF AS.
	NSSA - This area is a Not So Stubby Area (NSSA), per RFC 1587. It has routing information about the OSPF AS and networks that are outside the OSPF AS and are directly connected to the NSSA. It does not have information about other networks outside the OSPF AS.
Authentication	Select the default authentication method used in the area. This authentication protects the integrity, but not the confidentiality, of routing updates.
	None uses no authentication.
	Text uses a plain text password that is sent over the network (not very secure).
	MD5 uses an MD5 password and authentication ID (most secure).
Text Authentication Key	This field is available if the Authentication is Text . Type the password for text authentication. The key can consist of alphanumeric characters and the underscore, and it can be up to 16 characters long.
MD5 Authentication ID	This field is available if the Authentication is MD5 . Type the default ID for MD5 authentication in the area. The ID can be between 1 and 255.
MD5 Authentication Key	This field is available if the Authentication is MD5 . Type the default password for MD5 authentication in the area. The password can consist of alphanumeric characters and the underscore, and it can be up to 16 characters long.
Virtual Link	This section is displayed if the Type is Normal . Create a virtual link if you want to connect a different area (that does not have a direct connection to the backbone) to the backbone. You should set up the virtual link on the ABR that is connected to the other area and on the ABR that is connected to the backbone.
Add	Click this to create a new virtual link.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
#	This field is a sequential value, and it is not associated with a specific area.
Peer Router ID	This is the 32-bit ID (in IP address format) of the other ABR in the virtual link.
Authentication	This is the authentication method the virtual link uses. This authentication protects the integrity, but not the confidentiality, of routing updates.
	None uses no authentication.
	Text uses a plain text password that is sent over the network (not very secure). Hover your cursor over this label to display the password.
	MD5 uses an MD5 password and authentication ID (most secure). Hover your cursor over this label to display the authentication ID and key.
	Same as Area has the virtual link also use the Authentication settings above.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 76	Configuration >	Network >	Routing	> OSPF	bbA <
	configuration >	NCLWOIR /	Routing	> 0011	∕ Auu

10.3.3 Virtual Link Add/Edit Screen

The **Virtual Link Add/Edit** screen allows you to create a new virtual link or edit an existing one. When the OSPF add or edit screen (see Section 10.3.2 on page 204) has the Type set to Normal, a Virtual Link table displays. Click either the **Add** icon or an entry and the **Edit** icon to display a screen like the following.

Peer Router ID:		
Authentication:	MD5	~
MD5 Authentication ID:		(1-255)
MD5 Authentication Key:		
	3	

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Peer Router ID	Enter the 32-bit ID (in IP address format) of the other ABR in the virtual link.
Authentication	Select the authentication method the virtual link uses. This authentication protects the integrity, but not the confidentiality, of routing updates.
	None uses no authentication.
	Text uses a plain text password that is sent over the network (not very secure).
	MD5 uses an MD5 password and authentication ID (most secure).
	Same as Area has the virtual link also use the Authentication settings above.
Text Authentication Key	This field is available if the Authentication is Text . Type the password for text authentication. The key can consist of alphanumeric characters and the underscore, and it can be up to 16 characters long.
MD5 Authentication ID	This field is available if the Authentication is MD5 . Type the default ID for MD5 authentication in the area. The ID can be between 1 and 255.
MD5 Authentication Key	This field is available if the Authentication is MD5 . Type the default password for MD5 authentication in the area. The password can consist of alphanumeric characters and the underscore, and it can be up to 16 characters long.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

10.4 Routing Protocol Technical Reference

Here is more detailed information about RIP and OSPF.

Authentication Types

Authentication is used to guarantee the integrity, but not the confidentiality, of routing updates. The transmitting router uses its key to encrypt the original message into a smaller message, and the smaller message is transmitted with the original message. The receiving router uses its key to encrypt the received message and then verifies that it matches the smaller message sent with it. If the received message is verified, then the receiving router accepts the updated routing information. The transmitting and receiving routers must have the same key.

The ZyWALL supports three types of authentication for RIP and OSPF routing protocols:

- None no authentication is used.
- **Text** authentication using a plain text password, and the (unencrypted) password is sent over the network. This method is usually used temporarily to prevent network problems.
- MD5 authentication using an MD5 password and authentication ID.

MD5 is an authentication method that produces a 128-bit checksum, called a message-digest, for each packet. It also includes an authentication ID, which can be set to any value between 1 and 255. The ZyWALL only accepts packets if these conditions are satisfied.

- The packet's authentication ID is the same as the authentication ID of the interface that received it.
- The packet's message-digest is the same as the one the ZyWALL calculates using the MD5 password.

For RIP, authentication is not available in RIP version 1. In RIP version 2, you can only select one authentication type for all interfaces. For OSPF, the ZyWALL supports a default authentication type by area. If you want to use this default in an interface or virtual link, you set the associated **Authentication Type** field to **Same as Area**. As a result, you only have to update the authentication information for the area to update the authentication type used by these interfaces and virtual links. Alternatively, you can override the default in any interface or virtual link by selecting a specific authentication method. Please see the respective interface sections for more information.

11

Zones

11.1 Zones Overview

Set up zones to configure network security and network policies in the ZyWALL. A zone is a group of interfaces and/or VPN tunnels. The ZyWALL uses zones instead of interfaces in many security and policy settings, such as firewall rules, Anti-X, and remote management.

Zones cannot overlap. Each Ethernet interface, VLAN interface, bridge interface, PPPoE/PPTP interface and VPN tunnel can be assigned to at most one zone. Virtual interfaces are automatically assigned to the same zone as the interface on which they run.





11.1.1 What You Can Do in this Chapter

Use the Zone screens (see Section 11.2 on page 209) to manage the ZyWALL's zones.

11.1.2 What You Need to Know

Effects of Zones on Different Types of Traffic

Zones effectively divide traffic into three types--intra-zone traffic, inter-zone traffic, and extra-zone traffic--which are affected differently by zone-based security and policy settings.

Intra-zone Traffic

- Intra-zone traffic is traffic between interfaces or VPN tunnels in the same zone. For example, in Figure 121 on page 208, traffic between VLAN 2 and the Ethernet is intra-zone traffic.
- In each zone, you can either allow or prohibit all intra-zone traffic. For example, in Figure 121 on page 208, you might allow intra-zone traffic in the LAN zone but prohibit it in the WAN zone.
- You can set up firewall rules to control intra-zone traffic (for example, DMZ-to-DMZ), but many other types of zone-based security and policy settings do not affect intra-zone traffic.

Inter-zone Traffic

Inter-zone traffic is traffic between interfaces or VPN tunnels in different zones. For example, in Figure 121 on page 208, traffic between VLAN 1 and the Internet is inter-zone traffic. This is the normal case when zone-based security and policy settings apply.

Extra-zone Traffic

- Extra-zone traffic is traffic to or from any interface or VPN tunnel that is not assigned to a zone. For example, in Figure 121 on page 208, traffic to or from computer **C** is extra-zone traffic.
- Some zone-based security and policy settings may apply to extra-zone traffic, especially if you can set the zone attribute in them to **Any** or **All**. See the specific feature for more information.

11.2 The Zone Screen

The **Zone** screen provides a summary of all zones. In addition, this screen allows you to add, edit, and remove zones. To access this screen, click **Configuration > Network > Zone**.

Figure 122	Configuration	>	Network	>	Zone
------------	---------------	---	---------	---	------

🕽 Add 📝 Edit 🍵 Remove 📴 Object Reference					
# *	Name	Block Intra-zone	Member		
A Page 1	of 1 🕨 🕅 Show 5	0 v items	N	o data to display	
em Default					
Edit 🖪 Objec	t Reference				
# *	Name	Block Intra-zone	Member		
# 🔺	Name LAN1	Block Intra-zone	Member lan1		
# • 1 2	LAN1 LAN2	Block Intra-zone no no	Member lan1 lan2		
# ▲ 1 2 3	Name LAN1 LAN2 WLAN	Block Intra-zone no no no	Member lan1 lan2 wlan-1-1		
# • 1 2 3 4	Name LAN1 LAN2 WLAN WAN	Block Intra-zone no no no yes	Member lan1 lan2 wlan-1-1 wan1,wan1_ppp		
# ▲ 1 2 3 4 5	Name LAN1 LAN2 WLAN WAN DMZ	Block Intra-zone no no no yes yes	Member lan1 lan2 wlan-1-1 wan1,wan1_ppp dmz		
# ▲ 1 2 3 4 5 6	Name LAN1 LAN2 WLAN WAN DMZ SSL_VPN	Block Intra-zone no no no yes yes yes	Member lan1 lan2 wlan-1-1 wan1,wan1_ppp dmz New		
# ▲ 1 2 3 4 5 6 7	Name LAN1 LAN2 WLAN WAN DMZ SSL_VPN IPSec_VPN	Block Intra-zone no no yes yes yes yes yes	Member lan1 lan2 wlan-1-1 wan1,wan1_ppp dmz New Default_L2TP_VPN_Connection		

The following table describes the labels in this screen.

LABEL	DESCRIPTION
User Configuration / System Default	The ZyWALL comes with pre-configured System Default zones that you cannot delete. You can create your own User Configuration zones
Add	Click this to create a new, user-configured zone.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove a user-configured trunk, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with any interface.
Name	This field displays the name of the zone.
Member	This field displays the names of the interfaces that belong to each zone.

Table 78 Configuration > Network > Zone

11.3 Zone Edit

The **Zone Edit** screen allows you to add or edit a zone. To access this screen, go to the **Zone** screen (see Section 11.2 on page 209), and click the **Add** icon or an **Edit** icon.

Figure 123 Network > Zone > Add

roup Members				
Name:				
lember List				
Available		-	Member	
=== Interface	2=== .			
ge1_ppp		1		
ge2_ppp	nder John der der John der John der John der "Sis			
ge3_ppp				
ge4_ppp	E	+		
ge5_ppp		+		
ge6_ppp				
ge7				
ge7_ppp				
ge8	-			

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Name	For a system default zone, the name is read only.
	For a user-configured zone, type the name used to refer to the zone. You may use 1-31 alphanumeric characters, underscores (_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Member List	Available lists the interfaces and VPN tunnels that do not belong to any zone. Select the interfaces and VPN tunnels that you want to add to the zone you are editing, and click the right arrow button to add them.
	Member lists the interfaces and VPN tunnels that belong to the zone. Select any interfaces that you want to remove from the zone, and click the left arrow button to remove them.
ОК	Click OK to save your customized settings and exit this screen.
Cancel	Click Cancel to exit this screen without saving.

Table 79 Network > Zone > Add/Edit

12

DDNS

12.1 DDNS Overview

Dynamic DNS (DDNS) services let you use a domain name with a dynamic IP address.

12.1.1 What You Can Do in this Chapter

- Use the **DDNS** screen (see Section 12.2 on page 213) to view a list of the configured DDNS domain names and their details.
- Use the **DDNS Add/Edit** screen (see Section 12.2.1 on page 214) to add a domain name to the ZyWALL or to edit the configuration of an existing domain name.

12.1.2 What You Need to Know

DNS maps a domain name to a corresponding IP address and vice versa. Similarly, dynamic DNS maps a domain name to a dynamic IP address. As a result, anyone can use the domain name to contact you (in NetMeeting, CU-SeeMe, etc.) or to access your FTP server or Web site, regardless of the current IP address.

Note: You must have a public WAN IP address to use Dynamic DNS.

You must set up a dynamic DNS account with a supported DNS service provider before you can use Dynamic DNS services with the ZyWALL. When registration is complete, the DNS service provider gives you a password or key. At the time of writing, the ZyWALL supports the following DNS service providers. See the listed websites for details about the DNS services offered by each.

PROVIDER	SERVICE TYPES SUPPORTED	WEBSITE
DynDNS	Dynamic DNS, Static DNS, and Custom DNS	www.dyndns.com
Dynu	Basic, Premium	www.dynu.com
No-IP	No-IP	www.no-ip.com
Peanut Hull	Peanut Hull	www.oray.cn
3322	3322 Dynamic DNS, 3322 Static DNS	www.3322.org

Table 80 DDNS Service Providers

Note: Record your DDNS account's user name, password, and domain name to use to configure the ZyWALL.

After, you configure the ZyWALL, it automatically sends updated IP addresses to the DDNS service provider, which helps redirect traffic accordingly.

12.2 The DDNS Screen

The **DDNS** screen provides a summary of all DDNS domain names and their configuration. In addition, this screen allows you to add new domain names, edit the configuration for existing domain names, and delete domain names. Click **Configuration > Network > DDNS** to open the following screen.

Figure 124 Configuration > Network > DDNS

0 A	Add 📝 E	dit 🎁 Remove 🧕	Activate 💿 Inactiv	ate		
# 🔺	Status	Profile Name	DDNS Type	Domain Name	Primary Interface/IP	Backup Interface/IP
1	Q	Example	DynDNS	example	wan1/from interface	none

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
#	This is the number of an individual DDNS profile.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Profile Name	This field displays the descriptive profile name for this entry.
DDNS Type	This field displays which DDNS service you are using.
Domain Name	This field displays each domain name the ZyWALL can route.
Primary Interface/IP	This field displays the interface to use for updating the IP address mapped to the domain name followed by how the ZyWALL determines the IP address for the domain name.
	from interface - The IP address comes from the specified interface.
	auto detected -The DDNS server checks the source IP address of the packets from the ZyWALL for the IP address to use for the domain name.
	custom - The IP address is static.
Backup Interface/IP	This field displays the alternate interface to use for updating the IP address mapped to the domain name followed by how the ZyWALL determines the IP address for the domain name. The ZyWALL uses the backup interface and IP address when the primary interface is disabled, its link is down or its connectivity check fails.
	from interface - The IP address comes from the specified interface.
	auto detected -The DDNS server checks the source IP address of the packets from the ZyWALL for the IP address to use for the domain name.
	custom - The IP address is static.

 Table 81
 Configuration > Network > DDNS

LABEL	DESCRIPTION
Apply	Click this button to save your changes to the ZyWALL.
Reset	Click this button to return the screen to its last-saved settings.

Table 81 Configuration > Network > DDNS (continued)

12.2.1 The Dynamic DNS Add/Edit Screen

The **DDNS Add/Edit** screen allows you to add a domain name to the ZyWALL or to edit the configuration of an existing domain name. Click **Configuration > Network > DDNS** and then an **Add** or **Edit** icon to open this screen.

Add Profile				1
Hide Advanced Settings				
Seneral Settings				
Fnable DDNS Profile				
Profile Name:		0		
DDNS Type:	DynDNS	v v		
DNE Assount				
Username:		0		
Password:				
Retype to Confirm:				
DNS Settings				
Domain Name:		0		_
Primary Binding Address				
Interface:	wan1	~		
IP Address:	Interface	~		
Backup Binding Address				
Tabarfran	none	~		
Interface:				
Enable Wildcard				
Enable Wildcard Mail Exchanger:			(Optional)	
Enable Wildcard Mail Exchanger: Backup Mail Exchanger			(Optional)	
Enable Wildcard Mail Exchanger: Backup Mail Exchanger			(Optional)	

Figure 125 Configuration > Network > DDNS > Add

The following table describes the labels in this screen.

Table 82 Configuration > Network > DDNS > Add		
LABEL	DESCRIPTION	
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.	
Enable DDNS Profile	Select this check box to use this DDNS entry.	
Profile Name	When you are adding a DDNS entry, type a descriptive name for this DDNS entry in the ZyWALL. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.	

This field is read-only when you are editing an entry. Select the type of DDNS service you are using.

DDNS Type

LABEL	DESCRIPTION
Username	Type the user name used when you registered your domain name. You can use up to 31 alphanumeric characters and the underscore. Spaces are not allowed.
	For a Dynu DDNS entry, this user name is the one you use for logging into the service, not the name recorded in your personal information in the Dynu website.
Password	Type the password provided by the DDNS provider. You can use up to 64 alphanumeric characters and the underscore. Spaces are not allowed.
Retype to Confirm	Type the password again to confirm it.
DDNS Settings	
Domain name	Type the domain name you registered. You can use up to 255 characters.
Primary Binding Address	Use these fields to set how the ZyWALL determines the IP address that is mapped to your domain name in the DDNS server. The ZyWALL uses the Backup Binding Address if the interface specified by these settings is not available.
Interface	Select the interface to use for updating the IP address mapped to the domain name. Select Any to let the domain name be used with any interface.
IP Address	The options available in this field vary by DDNS provider.
	Interface -The ZyWALL uses the IP address of the specified interface. This option appears when you select a specific interface in the Primary Binding Address Interface field.
	Auto - If the interface has a dynamic IP address, the DDNS server checks the source IP address of the packets from the ZyWALL for the IP address to use for the domain name. You may want to use this if there are one or more NAT routers between the ZyWALL and the DDNS server.
	Note: The ZyWALL may not determine the proper IP address if there is an HTTP proxy server between the ZyWALL and the DDNS server.
	Custom - If you have a static IP address, you can select this to use it for the domain name. The ZyWALL still sends the static IP address to the DDNS server.
Custom IP	This field is only available when the IP Address is Custom . Type the IP address to use for the domain name.
Backup Binding Address	Use these fields to set an alternate interface to map the domain name to when the interface specified by the Primary Binding Interface settings is not available.
Interface	Select the interface to use for updating the IP address mapped to the domain name. Select Any to let the domain name be used with any interface. Select None to not use a backup address.
IP Address	The options available in this field vary by DDNS provider.
	Interface -The ZyWALL uses the IP address of the specified interface. This option appears when you select a specific interface in the Backup Binding Address Interface field.
	Auto -The DDNS server checks the source IP address of the packets from the ZyWALL for the IP address to use for the domain name. You may want to use this if there are one or more NAT routers between the ZyWALL and the DDNS server.
	Note: The ZyWALL may not determine the proper IP address if there is an HTTP proxy server between the ZyWALL and the DDNS server.
	Custom - If you have a static IP address, you can select this to use it for the domain name. The ZyWALL still sends the static IP address to the DDNS server.
Custom IP	This field is only available when the IP Address is Custom . Type the IP address to use for the domain name.

 Table 82
 Configuration > Network > DDNS > Add (continued)

LABEL	DESCRIPTION
Enable Wildcard	This option is only available with a DynDNS account.
	Enable the wildcard feature to alias subdomains to be aliased to the same IP address as your (dynamic) domain name. This feature is useful if you want to be able to use, for example, www.yourhost.dyndns.org and still reach your hostname.
Mail Exchanger	This option is only available with a DynDNS account.
	DynDNS can route e-mail for your domain name to a mail server (called a mail exchanger). For example, DynDNS routes e-mail for john-doe@yourhost.dyndns.org to the host record specified as the mail exchanger.
	If you are using this service, type the host record of your mail server here. Otherwise leave the field blank.
	See www.dyndns.org for more information about mail exchangers.
Backup Mail Exchanger	This option is only available with a DynDNS account.
	Select this check box if you are using DynDNS's backup service for e-mail. With this service, DynDNS holds onto your e-mail if your mail server is not available. Once your mail server is available again, the DynDNS server delivers the mail to you. See www.dyndns.org for more information about this service.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 82Configuration > Network > DDNS > Add (continued)
NAT

13.1 NAT Overview

NAT (Network Address Translation - NAT, RFC 1631) is the translation of the IP address of a host in a packet. For example, the source address of an outgoing packet, used within one network is changed to a different IP address known within another network. Use Network Address Translation (NAT) to make computers on a private network behind the ZyWALL available outside the private network. If the ZyWALL has only one public IP address, you can make the computers in the private network available by using ports to forward packets to the appropriate private IP address.

Suppose you want to assign ports 21-25 to one FTP, Telnet and SMTP server (**A** in the example), port 80 to another (**B** in the example) and assign a default server IP address of 192.168.1.35 to a third (**C** in the example). You assign the LAN IP addresses and the ISP assigns the WAN IP address. The NAT network appears as a single host on the Internet.



Figure 126 Multiple Servers Behind NAT Example

13.1.1 What You Can Do in this Chapter

Use the **NAT** screens (see Section 13.2 on page 218) to view and manage the list of NAT rules and see their configuration details. You can also create new NAT rules and edit or delete existing ones.

13.1.2 What You Need to Know

NAT is also known as virtual server, port forwarding, or port translation.

• See Section 13.3 on page 221 for technical background information related to these screens.

13.2 The NAT Screen

The **NAT** summary screen provides a summary of all NAT rules and their configuration. In addition, this screen allows you to create new NAT rules and edit and delete existing NAT rules. To access this screen, login to the Web Configurator and click **Configuration > Network > NAT**. The following screen appears, providing a summary of the existing NAT rules.

Figure 127 Configuration > Network > NAT

N If yo	Note: ou want to config	gure SNAT, plase	go to <u>Policy</u>	Route.				
بر 🕥 بر	Add 📝 Edit 🎁	Remove 😡 Act	ivate 🖗 Ina	Original IR	Mannad ID	Brotocol	Original Bort	Mannad Bort
"	Q test	Many 1:1 NAT	swan1	EWLAN-1-1		L any	Onginal Port	mapped Port

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate .
Inactivate	To turn off an entry, select it and click Inactivate.
#	This field is a sequential value, and it is not associated with a specific entry.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Name	This field displays the name of the entry.
Mapping Type	This field displays what kind of NAT this entry performs: Virtual Server , 1:1 NAT , or Many 1:1 NAT .
Interface	This field displays the interface on which packets for the NAT entry are received.
Original IP	This field displays the original destination IP address (or address object) of traffic that matches this NAT entry. It displays any if there is no restriction on the original destination IP address.
Mapped IP	This field displays the new destination IP address for the packet.
Protocol	This field displays the service used by the packets for this NAT entry. It displays any if there is no restriction on the services.
Original Port	This field displays the original destination port(s) of packets for the NAT entry. This field is blank if there is no restriction on the original destination port.
Mapped Port	This field displays the new destination port(s) for the packet. This field is blank if there is no restriction on the original destination port.
Apply	Click this button to save your changes to the ZyWALL.
Reset	Click this button to return the screen to its last-saved settings.

Table 83 Configuration > Network > NAT

13.2.1 The NAT Add/Edit Screen

The **NAT Add/Edit** screen lets you create new NAT rules and edit existing ones. To open this window, open the **NAT** summary screen. (See Section 13.2 on page 218.) Then, click on an **Add** icon or **Edit** icon to open the following screen.

🔄 Create new Object 🔻				
General Settings				
💟 Enable Rule				
Rule Name:				
Mapping Type				
Classification:	🔘 Virtual Server	I:1 NAT	🔘 Many 1:1 NAT	
Mapping Rule				
Incoming Interface:	wan1	~		
Original IP:	User Defined	~		
User-Defined Original IP:		🔍 🕕 Address)		
Mapped IP:	User Defined	~		
User-Defined Mapped IP:		🔍 🕕 Address)		
Port Mapping Type:	any	~		
Related Settings				
Enable NAT Loopback				
Contigure <u>Firewall</u>				
			ОК	Cancel

Figure 128 Configuration > Network > NAT > Add

LABEL	DESCRIPTION
Create new Object	Use to configure any new settings objects that you need to use in this screen.
Enable Rule	Use this option to turn the NAT rule on or off.
Rule Name	Type in the name of the NAT rule. The name is used to refer to the NAT rule. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Classification	Select what kind of NAT this rule is to perform.
	Virtual Server - This makes computers on a private network behind the ZyWALL available to a public network outside the ZyWALL (like the Internet).
	1:1 NAT - If the private network server will initiate sessions to the outside clients, select this to have the ZyWALL translate the source IP address of the server's outgoing traffic to the same public IP address that the outside clients use to access the server.
	Many 1:1 NAT - If you have a range of private network servers that will initiate sessions to the outside clients and a range of public IP addresses, select this to have the ZyWALL translate the source IP address of each server's outgoing traffic to the same one of the public IP addresses that the outside clients use to access the server. The private and public ranges must have the same number of IP addresses.
	One many 1:1 NAT rule works like multiple 1:1 NAT rules, but it eases configuration effort since you only create one rule.

Table 84 Configuration > Network > NAT > Add

LABEL	DESCRIPTION
Incoming Interface	Select the interface on which packets for the NAT rule must be received. It can be an Ethernet, VLAN, bridge, or PPPoE/PPTP interface.
Original IP	Specify the destination IP address of the packets received by this NAT rule's specified incoming interface.
	any - Select this to use all of the incoming interface's IP addresses including dynamic addresses or those of any virtual interfaces built upon the selected incoming interface.
	User Defined - Select this to manually enter an IP address in the User Defined field. For example, you could enter a static public IP assigned by the ISP without having to create a virtual interface for it.
	Host address - select a host address object to use the IP address it specifies. The list also includes address objects based on interface IPs. So for example you could select an address object based on a WAN interface even if it has a dynamic IP address.
User Defined Original IP	This field is available if Original IP is User Defined . Type the destination IP address that this NAT rule supports.
Original IP Subnet/ Range	This field displays for Many 1:1 NAT. Select the destination IP address subnet or IP address range that this NAT rule supports. The original and mapped IP address subnets or ranges must have the same number of IP addresses.
Mapped IP	Select to which translated destination IP address this NAT rule forwards packets.
	User Defined - this NAT rule supports a specific IP address, specified in the User Defined field.
	HOST address - the drop-down box lists all the HOST address objects in the ZyWALL. If you select one of them, this NAT rule supports the IP address specified by the address object.
User Defined Original IP	This field is available if Mapped IP is User Defined . Type the translated destination IP address that this NAT rule supports.
Mapped IP Subnet/ Range	This field displays for Many 1:1 NAT . Select to which translated destination IP address subnet or IP address range this NAT rule forwards packets. The original and mapped IP address subnets or ranges must have the same number of IP addresses.
Port Mapping Type	Use the drop-down list box to select how many original destination ports this NAT rule supports for the selected destination IP address (Original IP). Choices are:
	Any - this NAT rule supports all the destination ports.
	Port - this NAT rule supports one destination port.
	Ports - this NAT rule supports a range of destination ports. You might use a range of destination ports for unknown services or when one server supports more than one service.
Protocol Type	This field is available if Mapping Type is Port or Ports . Select the protocol (TCP , UDP , or Any) used by the service requesting the connection.
Original Port	This field is available if Mapping Type is Port . Enter the original destination port this NAT rule supports.
Mapped Port	This field is available if Mapping Type is Port . Enter the translated destination port if this NAT rule forwards the packet.
Original Start Port	This field is available if Mapping Type is Ports . Enter the beginning of the range of original destination ports this NAT rule supports.
Original End Port	This field is available if Mapping Type is Ports . Enter the end of the range of original destination ports this NAT rule supports.
Mapped Start Port	This field is available if Mapping Type is Ports . Enter the beginning of the range of translated destination ports if this NAT rule forwards the packet.

Table 84 Configuration > Network > NAT > Add (continued)

LABEL	DESCRIPTION
Mapped End Port	This field is available if Mapping Type is Ports . Enter the end of the range of translated destination ports if this NAT rule forwards the packet. The original port range and the mapped port range must be the same size.
Enable NAT Loopback	Enable NAT loopback to allow users connected to any interface (instead of just the specified Incoming Interface) to use the NAT rule's specified Original IP address to access the Mapped IP device. For users connected to the same interface as the Mapped IP device, the ZyWALL uses that interface's IP address as the source address for the traffic it sends from the users to the Mapped IP device.
	For example, if you configure a NAT rule to forward traffic from the WAN to a LAN server, enabling NAT loopback allows users connected to other interfaces to also access the server. For LAN users, the ZyWALL uses the LAN interface's IP address as the source address for the traffic it sends to the LAN server. See NAT Loopback on page 221 for more details.
	If you do not enable NAT loopback, this NAT rule only applies to packets received on the rule's specified incoming interface.
Firewall	By default the firewall blocks incoming connections from external addresses. After you configure your NAT rule settings, click the Firewall link to configure a firewall rule to allow the NAT rule's traffic to come in.
	The ZyWALL checks NAT rules before it applies To-ZyWALL firewall rules, so To-ZyWALL firewall rules do not apply to traffic that is forwarded by NAT rules. The ZyWALL still checks other firewall rules according to the source IP address and mapped IP address.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to return to the NAT summary screen without creating the NAT rule (if it is new) or saving any changes (if it already exists).

 Table 84
 Configuration > Network > NAT > Add (continued)

13.3 NAT Technical Reference

Here is more detailed information about NAT on the ZyWALL.

NAT Loopback

Suppose an NAT 1:1 rule maps a public IP address to the private IP address of a LAN SMTP e-mail server to give WAN users access. NAT loopback allows other users to also use the rule's original IP to access the mail server.

For example, a LAN user's computer at IP address 192.168.1.89 queries a public DNS server to resolve the SMTP server's domain name (xxx.LAN-SMTP.com in this example) and gets the SMTP server's mapped public IP address of 1.1.1.1.



Figure 129 LAN Computer Queries a Public DNS Server

The LAN user's computer then sends traffic to IP address 1.1.1.1. NAT loopback uses the IP address of the ZyWALL's LAN interface (192.168.1.1) as the source address of the traffic going from the LAN users to the LAN SMTP server.





The LAN SMTP server replies to the ZyWALL'S LAN IP address and the ZyWALL changes the source address to 1.1.1.1 before sending it to the LAN user. The return traffic's source matches the original destination address (1.1.1.1). If the SMTP server replied directly to the LAN user without the traffic going through NAT, the source would not match the original destination address which would cause the LAN user's computer to shut down the session.





HTTP Redirect

14.1 Overview

HTTP redirect forwards the client's HTTP request (except HTTP traffic destined for the ZyWALL) to a web proxy server. In the following example, proxy server **A** is connected to the **DMZ** interface. When a client connected to the **LAN1** zone wants to open a web page, its HTTP request is redirected to proxy server **A** first. If proxy server **A** cannot find the web page in its cache, a policy route allows it to access the Internet to get them from a server. Proxy server **A** then forwards the response to the client.





14.1.1 What You Can Do in this Chapter

Use the **HTTP Redirect** screens (see Section 14.2 on page 225) to display and edit the HTTP redirect rules.

14.1.2 What You Need to Know

Web Proxy Server

A proxy server helps client devices make indirect requests to access the Internet or outside network resources/services. A proxy server can act as a firewall or an ALG (application layer gateway) between the private network and the Internet or other networks. It also keeps hackers from knowing internal IP addresses.

A client connects to a web proxy server each time he/she wants to access the Internet. The web proxy provides caching service to allow quick access and reduce network usage. The proxy checks its local cache for the requested web resource first. If it is not found, the proxy gets it from the specified server and forwards the response to the client.

HTTP Redirect, Firewall and Policy Route

With HTTP redirect, the relevant packet flow for HTTP traffic is:

- 1 Firewall
- 2 HTTP Redirect
- 3 Policy Route

Even if you set a policy route to the same incoming interface and service as a HTTP redirect rule, the ZyWALL checks the HTTP redirect rules first and forwards HTTP traffic to a proxy server if matched. You need to make sure there is no firewall rule(s) blocking the HTTP requests from the client to the proxy server.

You also need to manually configure a policy route to forward the HTTP traffic from the proxy server to the Internet. To make the example in Figure 132 on page 224 work, make sure you have the following settings.

For HTTP traffic between **lan1** and **dmz**:

- a from LAN1 to DMZ firewall rule (default) to allow HTTP requests from **lan1** to **dmz**. Responses to this request are allowed automatically.
- a HTTP redirect rule to forward HTTP traffic from lan1 to proxy server A.

For HTTP traffic between dmz and wan1:

- a from DMZ to WAN firewall rule (default) to allow HTTP requests from **dmz** to **wan1**. Responses to these requests are allowed automatically.
- a policy route to forward HTTP traffic from proxy server **A** to the Internet.

14.2 The HTTP Redirect Screen

To configure redirection of a HTTP request to a proxy server, click **Configuration > Network > HTTP Redirect**. This screen displays the summary of the HTTP redirect rules.

Note: You can configure up to one HTTP redirect rule for each (incoming) interface.

Figure 133 Configuration > Network > HTTP Redirect

0	Add 📝 🛙	idit 🍵 Remove	💡 Activate 🖗 Ina	octivate	
#	Status	Name 🔺	Interface	Proxy Server	Port
1	Q	example	lan2	192.168.3.6	80
14	🔹 Page	1 of 1 >	🕨 Show 50 🔹	items	Displaying 1 - 1 o

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
#	This field is a sequential value, and it is not associated with a specific entry.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Name	This is the descriptive name of a rule.
Interface	This is the interface on which the request must be received.
Proxy Server	This is the IP address of the proxy server.
Port	This is the service port number used by the proxy server.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 85
 Configuration > Network > HTTP Redirect

14.2.1 The HTTP Redirect Edit Screen

Click **Network > HTTP Redirect** to open the **HTTP Redirect** screen. Then click the **Add** or **Edit** icon to open the **HTTP Redirect Edit** screen where you can configure the rule.

Figure 134	Network > HITP Redir	ect > Edi
O Add HTTP Rec	lirect	? ×
📝 Enable Name:		
Interface:	wan1 🗸	
Proxy server:		
Port:		
	OK	Cancel

Figure 134 Network > HTTP Redirect > Edit

LABEL	DESCRIPTION
Enable	Use this option to turn the HTTP redirect rule on or off.
Name	Enter a name to identify this rule. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Interface	Select the interface on which the HTTP request must be received for the ZyWALL to forward it to the specified proxy server.
Proxy Server	Enter the IP address of the proxy server.
Port	Enter the port number that the proxy server uses.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

 Table 86
 Network > HTTP Redirect > Edit

15

ALG

15.1 ALG Overview

Application Layer Gateway (ALG) allows the following applications to operate properly through the ZyWALL's NAT.

- SIP Session Initiation Protocol (SIP) An application-layer protocol that can be used to create voice and multimedia sessions over Internet.
- H.323 A teleconferencing protocol suite that provides audio, data and video conferencing.
- FTP File Transfer Protocol an Internet file transfer service.

The following example shows SIP signaling (1) and audio (2) sessions between SIP clients **A** and **B** and the SIP server.

Figure 135 SIP ALG Example



The ALG feature is only needed for traffic that goes through the ZyWALL's NAT.

15.1.1 What You Can Do in this Chapter

Use the ALG screen (Section 15.2 on page 231) to set up SIP, H.323, and FTP ALG settings.

15.1.2 What You Need to Know

Application Layer Gateway (ALG), NAT and Firewall

The ZyWALL can function as an Application Layer Gateway (ALG) to allow certain NAT un-friendly applications (such as SIP) to operate properly through the ZyWALL'S NAT and firewall. The ZyWALL dynamically creates an implicit NAT session and firewall session for the application's traffic from the WAN to the LAN. The ALG on the ZyWALL supports all of the ZyWALL'S NAT mapping types.

FTP ALG

The FTP ALG allows TCP packets with a specified port destination to pass through. If the FTP server is located on the LAN, you must also configure NAT (port forwarding) and firewall rules if you want to allow access to the server from the WAN.

H.323 ALG

- The H.323 ALG supports peer-to-peer H.323 calls.
- The H.323 ALG handles H.323 calls that go through NAT or that the ZyWALL routes. You can also make other H.323 calls that do not go through NAT or routing. Examples would be calls between LAN IP addresses that are on the same subnet.
- The H.323 ALG allows calls to go out through NAT. For example, you could make a call from a private IP address on the LAN to a peer device on the WAN.
- The H.323 ALG operates on TCP packets with a specified port destination.
- The ZyWALL allows H.323 audio connections.
- The ZyWALL can also apply bandwidth management to traffic that goes through the H.323 ALG.

The following example shows H.323 signaling (1) and audio (2) sessions between H.323 devices A and B.

Figure 136 H.323 ALG Example



SIP ALG

- SIP phones can be in any zone (including LAN, DMZ, WAN), and the SIP server and SIP clients can be in the same network or different networks.
- There should be only one SIP server (total) on the ZyWALL's private networks. Any other SIP servers must be on the WAN. So for example you could have a Back-to-Back User Agent such as the IPPBX x6004 or an asterisk PBX on the DMZ or on the LAN but not on both.
- Using the SIP ALG allows you to use bandwidth management on SIP traffic.
- The SIP ALG handles SIP calls that go through NAT or that the ZyWALL routes. You can also make other SIP calls that do not go through NAT or routing. Examples would be calls between LAN IP addresses that are on the same subnet.
- The SIP ALG supports peer-to-peer SIP calls. The firewall (by default) allows peer to peer calls from the LAN zone to go to the WAN zone and blocks peer to peer calls from the WAN zone to the LAN zone.
- The SIP ALG allows UDP packets with a specified port destination to pass through.
- The ZyWALL allows SIP audio connections.
- You do not need to use TURN (Traversal Using Relay NAT) for VoIP devices behind the ZyWALL when you enable the SIP ALG.

Peer-to-Peer Calls and the ZyWALL

The ZyWALL ALG can allow peer-to-peer VoIP calls for both H.323 and SIP. You must configure the firewall and NAT (port forwarding) to allow incoming (peer-to-peer) calls from the WAN to a private IP address on the LAN (or DMZ).

VoIP Calls from the WAN with Multiple Outgoing Calls

When you configure the firewall and NAT (port forwarding) to allow calls from the WAN to a specific IP address on the LAN, you can also use policy routing to have H.323 (or SIP) calls from other LAN or DMZ IP addresses go out through a different WAN IP address. The policy routing lets the ZyWALL correctly forward the return traffic for the calls initiated from the LAN IP addresses.

For example, you configure the firewall and NAT to allow LAN IP address **A** to receive calls from the Internet through WAN IP address **1**. You also use a policy route to have LAN IP address **A** make calls out through WAN IP address **1**. Configure another policy route to have H.323 (or SIP) calls from LAN IP addresses **B** and **C** go out through WAN IP address **2**. Even though only LAN IP address **A** can receive incoming calls from the Internet, LAN IP addresses **B** and **C** can still make calls out to the Internet.



Figure 137 VoIP Calls from the WAN with Multiple Outgoing Calls

VoIP with Multiple WAN IP Addresses

With multiple WAN IP addresses on the ZyWALL, you can configure different firewall and NAT (port forwarding) rules to allow incoming calls from each WAN IP address to go to a specific IP address on the LAN (or DMZ). Use policy routing to have the H.323 (or SIP) calls from each of those LAN or DMZ IP addresses go out through the same WAN IP address that calls come in on. The policy routing lets the ZyWALL correctly forward the return traffic for the calls initiated from the LAN IP addresses.

For example, you configure firewall and NAT rules to allow LAN IP address **A** to receive calls through public WAN IP address **1**. You configure different firewall and port forwarding rules to allow LAN IP address **B** to receive calls through public WAN IP address **2**. You configure corresponding policy routes to have calls from LAN IP address **A** go out through WAN IP address **1** and calls from LAN IP address **B** go out through WAN IP address **2**.





• See Section 15.3 on page 233 for ALG background/technical information.

15.1.3 Before You Begin

You must also configure the firewall and enable NAT in the ZyWALL to allow sessions initiated from the WAN.

15.2 The ALG Screen

Click **Configuration > Network > ALG** to open the **ALG** screen. Use this screen to turn ALGs off or on, configure the port numbers to which they apply, and configure SIP ALG time outs.

Figure 139 Configuration > Network > ALG

Enable SIP ALG	
Enable SIP Transformations	
Enable Configure SIP Inactivity Timeout	
SIP Media Inactivity Timeout :	120 (seconds)
SIP Signaling Inactivity Timeout :	1800 (seconds)
SIP Signaling Port :	Add Train Train
	* Pon *
	1 3060
1.323 Settings	
Enable H.323 ALG	
Enable H.323 Transformations	
H.323 Signaling Port :	1720 (1025-65535)
Additional H.323 Signaling Port for Transformations :	(1025-65535) (Optional)
TP Settings	
Imable FTP ALG	
Enable FTP Transformations	
FTP Signaling Port :	21 (1-65535)
	(1-65535) (Ontional)
Additional FTP Signaling Port for Transformations :	()()

LABEL	DESCRIPTION
Enable SIP ALG	Turn on the SIP ALG to detect SIP traffic and help build SIP sessions through the ZyWALL's NAT.
Enable SIP Transformations	Select this to have the ZyWALL modify IP addresses and port numbers embedded in the SIP data payload.
	You do not need to use this if you have a SIP device or server that will modify IP addresses and port numbers embedded in the SIP data payload.
Enable Configure SIP Inactivity Timeout	Select this option to have the ZyWALL apply SIP media and signaling inactivity time out limits.
SIP Media Inactivity Timeout	Use this field to set how many seconds (1~86400) the ZyWALL will allow a SIP session to remain idle (without voice traffic) before dropping it.
	If no voice packets go through the SIP ALG before the timeout period expires, the ZyWALL deletes the audio session. You cannot hear anything and you will need to make a new call to continue your conversation.
SIP Signaling Inactivity Timeout	Most SIP clients have an "expire" mechanism indicating the lifetime of signaling sessions. The SIP user agent sends registration packets to the SIP server periodically and keeps the session alive in the ZyWALL.
	If the SIP client does not have this mechanism and makes no calls during the ZyWALL SIP timeout, the ZyWALL deletes the signaling session after the timeout period. Enter the SIP signaling session timeout value (1~86400).
SIP Signaling Port	If you are using a custom UDP port number (not 5060) for SIP traffic, enter it here.
Enable H.323 ALG	Turn on the H.323 ALG to detect H.323 traffic (used for audio communications) and help build H.323 sessions through the ZyWALL's NAT.
Enable H.323 Transformations	Select this to have the ZyWALL modify IP addresses and port numbers embedded in the H.323 data payload.
	You do not need to use this if you have a H.323 device or server that will modify IP addresses and port numbers embedded in the H.323 data payload.
H.323 Signaling Port	If you are using a custom TCP port number (not 1720) for H.323 traffic, enter it here.
Additional H.323 Signaling Port for Transformations	If you are also using H.323 on an additional TCP port number, enter it here.
Enable FTP ALG	Turn on the FTP ALG to detect FTP (File Transfer Program) traffic and help build FTP sessions through the ZyWALL's NAT.
Enable FTP Transformations	Select this option to have the ZyWALL modify IP addresses and port numbers embedded in the FTP data payload to match the ZyWALL's NAT environment.
	Clear this option if you have an FTP device or server that will modify IP addresses and port numbers embedded in the FTP data payload to match the ZyWALL's NAT environment.
FTP Signaling Port	If you are using a custom TCP port number (not 21) for FTP traffic, enter it here.
Additional FTP Signaling Port for Transformations	If you are also using FTP on an additional TCP port number, enter it here.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 87
 Configuration > Network > ALG

15.3 ALG Technical Reference

Here is more detailed information about the Application Layer Gateway.

ALG

Some applications cannot operate through NAT (are NAT un-friendly) because they embed IP addresses and port numbers in their packets' data payload. The ZyWALL examines and uses IP address and port number information embedded in the VoIP traffic's data stream. When a device behind the ZyWALL uses an application for which the ZyWALL has VoIP pass through enabled, the ZyWALL translates the device's private IP address inside the data stream to a public IP address. It also records session port numbers and allows the related sessions to go through the firewall so the application's traffic can come in from the WAN to the LAN.

ALG and Trunks

If you send your ALG-managed traffic through an interface trunk and all of the interfaces are set to active, you can configure routing policies to specify which interface the ALG-managed traffic uses.

You could also have a trunk with one interface set to active and a second interface set to passive. The ZyWALL does not automatically change ALG-managed connections to the second (passive) interface when the active interface's connection goes down. When the active interface's connection fails, the client needs to re-initialize the connection through the second interface (that was set to passive) in order to have the connection go through the second interface. VoIP clients usually re-register automatically at set intervals or the users can manually force them to re-register.

FTP

File Transfer Protocol (FTP) is an Internet file transfer service that operates on the Internet and over TCP/IP networks. A system running the FTP server accepts commands from a system running an FTP client. The service allows users to send commands to the server for uploading and downloading files.

H.323

H.323 is a standard teleconferencing protocol suite that provides audio, data and video conferencing. It allows for real-time point-to-point and multipoint communication between client computers over a packet-based network that does not provide a guaranteed quality of service. NetMeeting uses H.323.

SIP

The Session Initiation Protocol (SIP) is an application-layer control (signaling) protocol that handles the setting up, altering and tearing down of voice and multimedia sessions over the Internet. SIP is used in VoIP (Voice over IP), the sending of voice signals over the Internet Protocol.

SIP signaling is separate from the media for which it handles sessions. The media that is exchanged during the session can use a different path from that of the signaling. SIP handles telephone calls and can interface with traditional circuit-switched telephone networks.

RTP

When you make a VoIP call using H.323 or SIP, the RTP (Real time Transport Protocol) is used to handle voice data transfer. See RFC 1889 for details on RTP.

IP/MAC Binding

16.1 IP/MAC Binding Overview

IP address to MAC address binding helps ensure that only the intended devices get to use privileged IP addresses. The ZyWALL uses DHCP to assign IP addresses and records the MAC address it assigned to each IP address. The ZyWALL then checks incoming connection attempts against this list. A user cannot manually assign another IP to his computer and use it to connect to the ZyWALL.

Suppose you configure access privileges for IP address 192.168.1.27 and use static DHCP to assign it to Tim's computer's MAC address of 12:34:56:78:90:AB. IP/MAC binding drops traffic from any computer trying to use IP address 192.168.1.27 with another MAC address.



Figure 140 IP/MAC Binding Example

16.1.1 What You Can Do in this Chapter

- Use the **Summary** and **Edit** screens (Section 16.2 on page 236) to bind IP addresses to MAC addresses.
- Use the **Exempt List** screen (Section 16.3 on page 238) to configure ranges of IP addresses to which the ZyWALL does not apply IP/MAC binding.

16.1.2 What You Need to Know

DHCP

IP/MAC address bindings are based on the ZyWALL's dynamic and static DHCP entries.

Interfaces Used With IP/MAC Binding

IP/MAC address bindings are grouped by interface. You can use IP/MAC binding with Ethernet, bridge, VLAN interfaces. You can also enable or disable IP/MAC binding and logging in an interface's configuration screen.

16.2 IP/MAC Binding Summary

Click **Configuration > Network > IP/MAC Binding** to open the **IP/MAC Binding Summary** screen. This screen lists the total number of IP to MAC address bindings for devices connected to each supported interface.

1	idit 🧕 Acti	ivate 🕼 Inactivate		
#	Status	Interface 🔺	Number of Binding	
1	ø	br0	0	
2	ø	dmz	0	
3	ø	lan1	0	
4	ø	lan2	0	
5	P	vlan1	0	
6	ø	wan1	0	
7	P	wan2	0	
14	4 Page	1 of 1 Show 50 vitems		Displaying 1 - 7 of 3

Figure 141 Configuration > Network > IP/MAC Binding > Summary

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
#	This field is a sequential value, and it is not associated with a specific entry.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Interface	This is the name of an interface that supports IP/MAC binding.
Number of Binding	This field displays the interface's total number of IP/MAC bindings and IP addresses that the interface has assigned by DHCP.
Apply	Click Apply to save your changes back to the ZyWALL.

Table 88 Configuration > Network > IP/MAC Binding > Summary

16.2.1 IP/MAC Binding Edit

Click **Configuration > Network > IP/MAC Binding > Edit** to open the **IP/MAC Binding Edit** screen. Use this screen to configure an interface's IP to MAC address binding settings.

Figure 142 Configuration > Network > IP/MAC Binding > Edit

iterface Name:	dmz(192.168.3.1/255.255	5.255.0)		
Enable IP/MAC Bind	ing			
Enable Logs for IP/	MAC Binding Violation			
The second second for the second s				
tic DHCP Bindings				
a add Dit cala 🍵	Demous			
🕽 Add 📝 Edit 🣋	Remove			
 Add Edit 1 IP Address 	Remove MAC Addre	88	Description	
Add Z Edit IP Address	Remove MAC Addre	SS	Description No.d.	ata to display

The following table describes the labels in this screen.

LABEL	DESCRIPTION
IP/MAC Binding Settings	
Interface Name	This field displays the name of the interface within the ZyWALL and the interface's IP address and subnet mask.
Enable IP/MAC Binding	Select this option to have this interface enforce links between specific IP addresses and specific MAC addresses. This stops anyone else from manually using a bound IP address on another device connected to this interface. Use this to make use only the intended users get to use specific IP addresses.
Enable Logs for IP/ MAC Binding Violation	Select this option to have the ZyWALL generate a log if a device connected to this interface attempts to use an IP address not assigned by the ZyWALL.
Static DHCP Bindings	This table lists the bound IP and MAC addresses. The ZyWALL checks this table when it assigns IP addresses. If the computer's MAC address is in the table, the ZyWALL assigns the corresponding IP address. You can also access this table from the interface's edit screen.
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
#	This is the index number of the static DHCP entry.
IP Address	This is the IP address that the ZyWALL assigns to a device with the entry's MAC address.
MAC Address	This is the MAC address of the device to which the ZyWALL assigns the entry's IP address.
Description	This helps identify the entry.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 89 Configuration > Network > IP/MAC Binding > Edit

16.2.2 Static DHCP Edit

Click **Configuration > Network > IP/MAC Binding > Edit** to open the **IP/MAC Binding Edit** screen. Click the **Add** or **Edit** icon to open the following screen. Use this screen to configure an interface's IP to MAC address binding settings.

Interface Name:	dmz(192.168.3.1/255.255.255.0)	
IP Address:		
MAC Address:		
Description:		(Optional)

Figure 143 Configuration > Network > IP/MAC Binding > Edit > Add

The following table describes the labels in this screen.

	a a b					.				
Table 90	Configuration	>	Network	>	IP/MAC	Binding	>	Edit	>	Add

LABEL	DESCRIPTION
Interface Name	This field displays the name of the interface within the ZyWALL and the interface's IP address and subnet mask.
IP Address	Enter the IP address that the ZyWALL is to assign to a device with the entry's MAC address.
MAC Address	Enter the MAC address of the device to which the ZyWALL assigns the entry's IP address.
Description	Enter up to 64 printable ASCII characters to help identify the entry. For example, you may want to list the computer's owner.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

16.3 IP/MAC Binding Exempt List

Click Configuration > Network > IP/MAC Binding > Exempt List to open the IP/MAC Binding Exempt List screen. Use this screen to configure ranges of IP addresses to which the ZyWALL does not apply IP/MAC binding.

Figure 144 Configuration > Network > IP/MAC Binding > Exempt List

0	Add 📝 Edit 🎁 Remove			
#	Name	Start IP	End IP	
	example	192.168.2.12		
14	◀ Page 1 of 1 ▶ ▶ Sho	w 50 🗸 items		No data to displa

 Table 91
 Configuration > Network > IP/MAC Binding > Exempt List

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Click an entry or select it and click Edit to modify the entry's settings.

LABEL	DESCRIPTION
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
#	This is the index number of the IP/MAC binding list entry.
Name	Enter a name to help identify this entry.
Start IP	Enter the first IP address in a range of IP addresses for which the ZyWALL does not apply IP/ MAC binding.
End IP	Enter the last IP address in a range of IP addresses for which the ZyWALL does not apply IP/ MAC binding.
Add icon	Click the Add icon to add a new entry.
	Click the Remove icon to delete an entry. A window displays asking you to confirm that you want to delete it.
Apply	Click Apply to save your changes back to the ZyWALL.

Table 91	Configuration >	> Network >	IP/MAC Binding >	Exempt List	(continued)
----------	-----------------	-------------	------------------	-------------	-------------

Inbound Load Balancing

17.1 Inbound Load Balancing Overview

Inbound load balancing enables the ZyWALL to respond to a DNS query message with a different IP address for DNS name resolution. The ZyWALL checks which member interface has the least load and responds to the DNS query message with the interface's IP address.

In the following figure, an Internet host (**A**) sends a DNS query message to the DNS server (**D**) in order to resolve a domain name of www.example.com. DNS server **D** redirects it to the ZyWALL (**Z**)'s WAN1 with an IP address of 1.1.1.1. The ZyWALL receives the DNS query message and responds to it with the WAN2's IP address, 2.2.2.2, because the WAN2 has the least load at that moment.

Another Internet host (**B**) also sends a DNS query message to ask where www.example.com is. The ZyWALL responds to it with the WAN1's IP address, 1.1.1.1, since WAN1 has the least load this time.





17.1.1 What You Can Do in this Chapter

• Use the **Inbound LB** screen (see Section 17.2 on page 241) to view a list of the configured DNS load balancing rules.

• Use the Inbound LB Add/Edit screen (see Section 17.2.1 on page 242) to add or edit a DNS load balancing rule.

17.2 The Inbound LB Screen

The **Inbound LB** screen provides a summary of all DNS load balancing rules and the details. You can also use this screen to add, edit, or remove the rules. Click **Configuration > Network > Inbound LB** to open the following screen.

Note: After you finish the inbound load balancing settings, go to firewall and NAT screens to configure the corresponding rule and virtual server to allow the Internet users to access your internal servers.

Figure 146 Configuration > Network > Inbound LB

Enable DNS Lo	ad Balancing					
nfiguration						
🗿 Add 📝 Edit	💼 Remove 💡 Activate 🦷 Ir	activate 🔊 Move				
Status Priority (Query Domain Name	Query From Add Qu	ery From Zone	Load Balancing Member	Algorithm	
A Page	1 of 1 ▶ ▶ Show 50	✓ items				No data to displa

LABEL	DESCRIPTION
Global Setting	
Enable DNS Load Balancing	Select this to enable DNS load balancing.
Configuration	
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Move	To move an entry to a different number in the list, click the Move icon. In the field that appears, specify the number to which you want to move the entry.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Priority	This field displays the order in which the ZyWALL checks the member interfaces of this DNS load balancing rule.
Query Domain Name	This field displays the domain name for which the ZyWALL manages load balancing between the specified interfaces.

 Table 92
 Configuration > Network > Inbound LB

LABEL	DESCRIPTION
Query From Address	This field displays the source IP address of the DNS query messages to which the ZyWALL applies the DNS load balancing rule.
Query From Zone	The ZyWALL applies the DNS load balancing rule to the query messages received from this zone.
Load Balancing Member	This field displays the member interfaces which the ZyWALL manages for load balancing.
Algorithm	This field displays the load balancing method the ZyWALL uses for this DNS load balancing rule.
	Weighted Round Robin - Each member interface is assigned a weight. An interface with a larger weight gets more chances to transmit traffic than an interface with a smaller weight. For example, if the weight ratio of wan1 and wan2 interfaces is 2:1, the ZyWALL chooses wan1 for 2 sessions' traffic and wan2 for 1 session's traffic in each round of 3 new sessions.
	Least Connection - The ZyWALL chooses choose a member interface which is handling the least number of sessions.
	Least Load - Outbound - The ZyWALL chooses a member interface which is handling the least amount of outgoing traffic.
	Least Load - Inbound - The ZyWALL chooses a member interface which is handling the least amount of incoming traffic.
	Least Load - Total - The ZyWALL chooses a member interface which is handling the least amount of outgoing and incoming traffic.
Apply	Click this button to save your changes to the ZyWALL.
Reset	Click this button to return the screen to its last-saved settings.

 Table 92
 Configuration > Network > Inbound LB (continued)

17.2.1 The Inbound LB Add/Edit Screen

The **Add DNS Load Balancing** screen allows you to add a domain name for which the ZyWALL manages load balancing between the specified interfaces. You can configure the ZyWALL to apply DNS load balancing to some specific hosts only by configuring the **Query From** settings. Click **Configuration > Network > Inbound LB** and then the **Add** or **Edit** icon to open this screen.

Create new Object +				
General Settings				
Enable				
NS Setting				
Query Domain Name:				
Time to Live:	0	(0-2147483647 second	s, 0 is unchanged)	
uery From Setting				
IP Address:	any	~		
Zone:	any	~		
oad Balancing Member				
Load Balancing Algorithm:	Weighted Roun	nd Robin		
Failover IP Address:	0.0.0.0	(Optional)		
🗿 Add 📝 Edit 🍵 Remove				
# IP Address	Monitor	Interface	Weight	
4 4 Page 1 of 1 ▶ ▶	Show 50 🗸 items			No data to display
			ſ	OK Cancel

Figure 147 Configuration > Network > Inbound LB > Add

LABEL	DESCRIPTION
Create New Object	Use this to configure any new setting objects that you need to use in this screen.
General Settings	
Enable	Select this to enable this DNS load balancing rule.
DNS Setting	
Query Domain Name	Type up to 255 characters for a domain name for which you want the ZyWALL to manage DNS load balancing. You can use a wildcard (*) to let multiple domains match the name. For example, use *.example.com to specify any domain name that ends with "example.com" would match.
Time to Live	Enter the number of seconds the ZyWALL recommends DNS request hosts to keep the DNS entry in their caches before removing it. Enter 0 to have the ZyWALL not recommend this so the DNS request hosts will follow their DNS server's TTL setting.
Query From Setting	
IP Address	Enter the IP address of a computer or a DNS server which makes the DNS queries upon which to apply this rule.
	DNS servers process client queries using recursion or iteration:
	 In recursion, DNS servers make recursive queries on behalf of clients. So you have to configure this field to the DNS server's IP address when recursion is used.
	• In iteration, a client asks the DNS server and expects the best and immediate answer without the DNS server contacting other DNS servers. If the primary DNS server cannot provide the best answer, the client makes iteration queries to other configured DNS servers to resolve the name. You have to configure this field to the client's IP address when iteration is used.
Zone	Select the zone of DNS query messages upon which to apply this rule.
Load Balancing Member	

 Table 93
 Configuration > Network > Inbound LB > Add/Edit

LABEL	DESCRIPTION
Load Balancing	Select a load balancing method to use from the drop-down list box.
	Select Weighted Round Robin to balance the traffic load between interfaces based on their respective weights. An interface with a larger weight gets more chances to transmit traffic than an interface with a smaller weight. For example, if the weight ratio of wan1 and wan2 interfaces is 2:1, the ZyWALL chooses wan1 for 2 sessions' traffic and wan2 for every session's traffic in each round of 3 new sessions.
	Select Least Connection to have the ZyWALL choose the member interface which is handling the least number of sessions.
	Select Least Load - Outbound to have the ZyWALL choose the member interface which is handling the least amount of outgoing traffic.
	Select Least Load - Inbound to have the ZyWALL choose the member interface which is handling the least amount of incoming traffic.
	Select Least Load - Total to have the ZyWALL choose the member interface which is handling the least amount of outgoing and incoming traffic.
Failover IP Address	Enter an alternate IP address with which the ZyWALL will respond to a DNS query message when the load balancing algorithm cannot find any available interface.
Add	Click this to create a new member interface for this rule.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
#	This field displays the order in which the ZyWALL checks this rule's member interfaces.
IP Address	This field displays the IP address of the member interface.
Monitor Interface	This field displays the name of the member interface. The ZyWALL manages load balancing between the member interfaces.
Weight	This field is available if you selected Weighted Round Robin as the load balancing algorithm. This field displays the weight of the member interface. An interface with a larger weight gets more chances to transmit traffic than an interface with a smaller weight.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

 Table 93
 Configuration > Network > Inbound LB > Add/Edit (continued)

17.2.2 The Inbound LB Member Add/Edit Screen

The Add Load Balancing Member screen allows you to add a member interface for the DNS load balancing rule. Click Configuration > Network > Inbound LB > Add or Edit and then an Add or Edit icon to open this screen.

Add Load Balancing Member				? >
Load Balancing Member				
Member:	1			
Monitor Interface:	wan1		/ DHCP dient 0.0.0.0/0.0.0.0	
Weight:	1	(1-10)		
IP Address				
Same as Monitor Interface	0.0.0			
Custom	0.0.0			
				OK Cancel

Figure 148 Configuration > Network > Inbound LB > Add/Edit > Add

LABEL	DESCRIPTION
Member	The ZyWALL checks each member interface's loading in the order displayed here.
Monitor Interface	Select an interface to associate it with the DNS load balancing rule. This field also displays whether the IP address is a static IP address (Static), dynamically assigned (Dynamic) or obtained from a DHCP server (DHCP Client), as well as the IP address and subnet mask.
Weight	This field is available if you selected Weighted Round Robin for the load balancing algorithm.
	Specify the weight of the member interface. An interface with a larger weight gets more chances to transmit traffic than an interface with a smaller weight.
IP Address	
Same as Monitor Interface	Select this to send the IP address displayed in the Monitor Interface field to the DNS query senders.
Custom	Select this and enter another IP address to send to the DNS query senders.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

Table 94 Configuration > Network > Inbound LB > Add/Edit > Add/Edit

Authentication Policy

18.1 Overview

Use authentication policies to control who can access the network. After a user passes authentication the user's computer must meet the endpoint security object's Operating System (OS) option and security requirements to gain access.

In the following figure the ZyWALL's authentication policy requires endpoint security checking on local user **A**. **A** passes authentication and the endpoint security check and is given access. Local user **B** passes authentication but fails the endpoint security check and is denied access.



Figure 149 Authentication Policy Using Endpoint Security

18.1.1 What You Can Do in this Chapter

Use the **Configuration > Auth. Policy** screens (Section 18.2 on page 247) to create and manage authentication policies.

18.1.2 What You Need to Know

Authentication Policy and VPN

Authentication policies are applied based on a traffic flow's source and destination IP addresses. If VPN traffic matches an authentication policy's source and destination IP addresses, the user must pass authentication.

Multiple Endpoint Security Objects

You can set an authentication policy to use multiple endpoint security objects. This allows checking of computers with different OSs or security settings. When a client attempts to log in, the ZyWALL checks the client's computer against the endpoint security objects one-by-one. The client's computer must match one of the authentication policy's endpoint security objects in order to gain access.

Forced User Authentication

Instead of making users for which user-aware policies have been configured go to the ZyWALL **Login** screen manually, you can configure the ZyWALL to display the **Login** screen automatically whenever it routes HTTP traffic for anyone who has not logged in yet.

Note: This works with HTTP traffic only. The ZyWALL does display the **Login** screen when users attempt to send other kinds of traffic.

The ZyWALL does not automatically route the request that prompted the login, however, so users have to make this request again.

Finding Out More

• See Section 18.3 on page 251 for an example of using an authentication policy for user-aware access control.

18.2 Authentication Policy Screen

The **Authentication Policy** screen displays the authentication policies you have configured on the ZyWALL.

Click **Configuration > Auth. Policy** to display the screen.

Figure 150	Configuration	>	Auth.	Policy

Enable Authentication Po	licy					
ceptional Services						
📀 Add 🍵 Remove						
# Exceptional Services	*					
DNS						
📢 🖣 🛛 Page 🚺 of 1	▶ ▶ Show	50 🔻 items			1	Displaying 1 - 1 of 3
Contraction Policy Summer Contract Structure S	mary nove 🌚 Activate	🥡 Inactivate i	Move			
Statu: Priorit [,] Source	Destination	Schedule	Authentication	EPS	Description	
🌚 1 any	апу	none	force		test	
Defau any	any	none	unnecessary	n/a	n/a	
Id d Down to be a	b b Show	50 🔻 items			1	Displaying 1 - 2 of :
IN N Page I of I						

The following table gives an overview of the objects you can configure.

LABEL	DESCRIPTION				
Enable Authentication Policy	Select this to turn on the authentication policy feature.				
Exceptional Services	Use this table to list services that users can access without logging in. Click Add to change the list's membership. A screen appears. Available services appear on the left. Select any services you want users to be able to access without logging in and click the right arrow button to add them. The member services are the right. Select any service that you want to remove from the member list, and click the left arrow button to remove them. Keeping DNS as a member allows users' computers to resolve domain names into IP addresses. Figure 151 Configuration > Auth. Policy > Add Exceptional Service Figure 151 Configuration > Auth. Policy > Add Exceptional Service Image: Services List Available				
	In the table, select one or more entries and click Remove to delete it or them.				

Table 95 Configuration > Auth. Policy

LABEL	DESCRIPTION
Authentication Policy Summary	Use this table to manage the ZyWALL's list of authentication policies.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Move	To move an entry to a different number in the list, click the Move icon. In the field that appears, specify the number to which you want to move the interface.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Priority	This is the position of the authentication policy in the list. The priority is important as the policies are applied in order of priority. Default displays for the default authentication policy that the ZyWALL uses on traffic that does not match any exceptional service or other authentication policy. You can edit the default rule but not delete it.
Source	This displays the source address object to which this policy applies.
Destination	This displays the destination address object to which this policy applies.
Schedule	This field displays the schedule object that dictates when the policy applies. none means the policy is active at all times if enabled.
Authentication	This field displays the authentication requirement for users when their traffic matches this policy. This is n/a for the default policy.
	unnecessary - Users do not need to be authenticated.
	required - Users need to be authenticated. They must manually go to the login screen. The ZyWALL will not redirect them to the login screen.
	force - Users need to be authenticated. The ZyWALL automatically displays the login screen whenever it routes HTTP traffic for users who have not logged in yet.
Description	If the entry has a description configured, it displays here.
Apply	Click this button to save your changes to the ZyWALL.
Reset	Click this button to return the screen to its last-saved settings.

Table 95	Configuration >	Auth.	Policy	(continued)
14510 00	oornigaration 2	7.00.0111	101105	(containada)

18.2.1 Creating/Editing an Authentication Policy

Click **Configuration > Auth. Policy** and then the **Add** (or **Edit**) icon to open the **Endpoint Security Edit** screen. Use this screen to configure an authentication policy.

Auth. Policy Add			?
Create new Object -			
General Settings			
Enable Policy			
Description:		(Optional)	
User Authentication Pol	icy		
Source Address:	any	¥	
Destination Address:	any	¥	
Schedule:	none	*	
Authentication:	required	~	

Figure 152 Configuration > Auth. Policy > Add

The following table gives an overview of the objects you can configure.

LABEL	DESCRIPTION
Create new Object	Use to configure any new settings objects that you need to use in this screen.
Enable Policy	Select this check box to activate the authentication policy. This field is available for user- configured policies.
Description	Enter a descriptive name of up to 60 printable ASCII characters for the policy. Spaces are allowed. This field is available for user-configured policies.
User Authentication Policy	Use this section of the screen to determine which traffic requires (or does not require) the senders to be authenticated in order to be routed.
Source Address	Select a source address or address group for whom this policy applies. Select any if the policy is effective for every source. This is any and not configurable for the default policy.
Destination Address	Select a destination address or address group for whom this policy applies. Select any if the policy is effective for every destination. This is any and not configurable for the default policy.
Schedule	Select a schedule that defines when the policy applies. Otherwise, select none and the rule is always effective. This is none and not configurable for the default policy.
Authentication	Select the authentication requirement for users when their traffic matches this policy. unnecessary - Users do not need to be authenticated. required - Users need to be authenticated. They must manually go to the login screen. The ZyWALL will not redirect them to the login screen.
Log	This field is available for the default policy. Select whether to have the ZyWALL generate a log (log), log and alert (log alert) or not (no) for packets that match the default policy. See Chapter 38 on page 474 for more on logs.

 Table 96
 Configuration > Auth. Policy > Add

LABEL	DESCRIPTION
Force User Authentication	This field is available for user-configured policies that require authentication. Select this to have the ZyWALL automatically display the login screen when users who have not logged in yet try to send HTTP traffic.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

 Table 96
 Configuration > Auth. Policy > Add (continued)

18.3 User-aware Access Control Example

You can configure many policies and security settings for specific users or groups of users. Users can be authenticated locally by the ZyWALL or by an external (AD, RADIUS, or LDAP) authentication server.

In this example the users are authenticated by an external RADIUS server at 192.168.1.200. First, set up the user accounts and user groups in the ZyWALL. Then, set up user authentication using the RADIUS server. Finally, set up the policies in the table above.

18.3.1 Set Up User Accounts

Set up user accounts in the RADIUS server. This example uses the Web Configurator. If you can export user names from the RADIUS server to a text file, then you might configure a script to create the user accounts instead.

- 1 Click Configuration > Object > User/Group > User. Click the Add icon.
- 2 Enter the same user name that is used in the RADIUS server, and set the **User Type** to **ext-user** because this user account is authenticated by an external server. Click **OK**.

Figure 153 Configuration > Object > User/Group > User > Add

Add A User reductored	3		External RADUS-Users	? X
User Configuration				
User Name:	Leo			
User Type:	ext-user	*		
Description:	Local User			
Authentication Timeout Settings	Use Default Set	ettings	Use Manual Settings	
Lease Time:	1440		minutes	
Reauthentication Time:	1440		minutes	
				ancel

3 Repeat this process to set up the remaining user accounts.

18.3.2 Set Up User Groups

Set up the user groups and assign the users to the user groups.

- 1 Click Configuration > Object > User/Group > Group. Click the Add icon.
- 2 Enter the name of the group. In this example, it is "Finance". Then, select **User/Leo** and click the right arrow to move him to the **Member** list. This example only has one member in this group, so click **OK**. Of course you could add more members later.

Name:	Finance			
Description:			(Optional)	
1ember List				
Available		1	Member	
Andy	^		Leo	
Debbie				
Steven				
ad-users		•		
guest		+		
Idap-users				
radius-users				
wlan_user				
=== Group =	~			

Figure 154 Configuration > Object > User/Group > Group > Add

3 Repeat this process to set up the remaining user groups.

18.3.3 Set Up User Authentication Using the RADIUS Server

This step sets up user authentication using the RADIUS server. First, configure the settings for the RADIUS server. Then, set up the authentication method, and configure the ZyWALL to use the authentication method. Finally, force users to log into the ZyWALL before it routes traffic for them.

1 Click Configuration > Object > AAA Server > RADIUS. Double-click the radius entry. Configure the RADIUS server's address, authentication port (1812 if you were not told otherwise), and key. Select case-sensitive if the RADIUS server checks user name casing. Click Apply.
General Settings		
Name:	radius	
Description:		(Optional)
Server Settings		
Server Address:	192.168.1.200	(IP or FQDN)
Authentication Port:	1812	(1-65535)
Backup Server Address:		(IP or FQDN) (Optional)
Backup Authentication Port:		(1-65535) (Optional)
Timeout:	5	(1-300 seconds)
Case-sensitive User Names 🔋		
Server Authentication		
Key:	••••	
Jser Login Settings		
Group Membership Attribute:	Filter-Id(11)	▼ 11

Figure 155 Configuration > Object > AAA Server > RADIUS > Add

2 Click Configuration > Object > Auth. Method. Double-click the default entry. Click the Add icon. Select group radius because the ZyWALL should use the specified RADIUS server for authentication. Click OK.

ire 1	56 Configuration > Object > Auth. metho	d > Edit
dit Au	chentication Method default	2
eneral	Settings	
Name:	default	
A	d 🗾 Edit 🎁 Remove 📣 Move	
#	Method List	
1	group radius 💙	
2	local	

- OK Cancel
- 3 Click Configuration > Auth. Policy. In the Authentication Policy Summary section, click the Add icon.
- 4 Set up a default policy that forces every user to log into the ZyWALL before the ZyWALL routes traffic for them. Select Enable. Set the Authentication field to required, and make sure Force User Authentication is selected. Keep the rest of the default settings, and click OK.
 - Note: The users must log in at the Web Configurator login screen before they can use HTTP or MSN.

Schedule:

Authentication:

Force User Authentication []

igure 157 Confi	iguration > A	Auth. P	olicy > Add
Auth. Policy Edit			
🛅 Create new Object 🔹			
General Settings			
💟 Enable Policy			
Description:	default_policy	(Optio	onal)
User Authentication Po	icy		
Source Address:	any	~	N/A
Destination Address:	any	*	N/A

In the Auth. Policy screen, select Enable Authentication Policy and click Apply.

OK

Cancel

👻 N/A

*

	ii Se	ttings						
Er	hable	Authentic	ation Policy					
cept	iona	Service	5					
0	Add	💼 Remo	ve					
#	Ex	ceptional S	ervices 🔺					
	DN	IS	and the second se					
14	4]	Page 1] of 1 ▶ ▶	Show 50	✓ items		Dis	playing 1 - 1 of
ther	itica	Page 1 tion Polic	of 1	Show 50	items	Move	Dis	playing 1 - 1 of
ther C 4 Stat	Add Prio	Page 1 tion Polic	of 1 y Summary Remove @ Destination	Show 50 Activate Schedule	vitems	Move EPS	Dis	playing 1 - 1 of
then	Add Prio	Page 1 tion Polic 2 Edit Source any	of 1 y Summary Remove @ Destination any	Show 50 Activate G Schedule none	vitems	Move	Dis Description default_policy	playing 1 - 1 of
then () Stat	Add Prio 1 Def	Page 1 tion Polic 2 Edit Source any any	of 1 y Summary Remove @ Destination any any	Show 50 Activate Schedule none none	 items Inactivate Inactivate Inactivate Authentication force unnecessary 	Move EPS n/a	Dis Description default_policy n/a	playing 1 - 1 of

Figure 158 Configuration > Auth. Policy

none

required

When the users try to browse the web (or use any HTTP/HTTPS application), the Login screen appears. They have to log in using the user name and password in the RADIUS server.

18.3.4 User Group Authentication Using the RADIUS Server

The previous example showed how to have a RADIUS server authenticate individual user accounts. If the RADIUS server has different user groups distinguished by the value of a specific attribute, you can make a couple of slight changes in the configuration to have the RADIUS server authenticate groups of user accounts defined in the RADIUS server.

1 Click Configuration > Object > AAA Server > RADIUS. Double-click the radius entry. Besides configuring the RADIUS server's address, authentication port, and key; set the Group Membership Attribute field to the attribute that the ZyWALL is to check to determine to which group a user belongs. This example uses Class. This attribute's value is called a group identifier; it determines to which group a user belongs. In this example the values are Finance, Engineer, Sales, and Boss. Select case-sensitive if the RADIUS server checks user name casing.

Name:	radius	
Description:		(Optional)
erver Settings		
Server Address:	192.168.1.200	(IP or FQDN)
Authentication Port:	1812	(1-65535)
Backup Server Address:		(IP or FQDN) (Optional)
Backup Authentication Port:		(1-65535) (Optional)
Timeout:	5	(1-300 seconds)
Case-sensitive User Names		
Key:	•••••	
ser Login Settings		
Group Membership Attribute:	Class(25)	▶ 25

Figure 159 Configuration > Object > AAA Server > RADIUS > Add

Now you add ext-group-user user objects to identify groups based on the group identifier values.
 Set up one user account for each group of user accounts in the RADIUS server. Click Configuration
 > Object > User/Group > User. Click the Add icon.

Enter a user name and set the **User Type** to **ext-group-user**. In the **Group Identifier** field, enter Finance, Engineer, Sales, or Boss and set the Associated AAA Server Object to radius.

Figure 160 Configuration > Object > User/Group > User > Add

User Name:	Finance	
User Type:	ext-group-user	*
Group Identifier:	Finance	
Associated AAA Server Object:	radius	~
Description:	Local User	
Authentication Timeout Settings	Ose Default Settings	O Use Manual Settings
Lease Time:	1440	minutes
Reauthentication Time:	1440	minutes

3 Repeat this process to set up the remaining groups of user accounts.

19

Firewall

19.1 Overview

Use the firewall to block or allow services that use static port numbers. This example shows the ZyWALL's default firewall behavior for WAN to LAN traffic and how stateful inspection works. A LAN user can initiate a Telnet session from within the LAN zone and the firewall allows the response. However, the firewall blocks Telnet traffic initiated from the WAN zone and destined for the LAN zone. The firewall allows VPN traffic between any of the networks.

Figure 161 Default Firewall Action



19.1.1 What You Can Do in this Chapter

- Use the **Firewall** screens (Section 19.2 on page 259) to enable or disable the firewall and asymmetrical routes, and manage and configure firewall rules.
- Use the **Session Limit** screens (see Section 19.3 on page 264) to limit the number of concurrent NAT/firewall sessions a client can use.

19.1.2 What You Need to Know

Stateful Inspection

The ZyWALL has a stateful inspection firewall. The ZyWALL restricts access by screening data packets against defined access rules. It also inspects sessions. For example, traffic from one zone is not allowed unless it is initiated by a computer in another zone first.

Zones

A zone is a group of interfaces or VPN tunnels. Group the ZyWALL's interfaces into different zones based on your needs. You can configure firewall rules for data passing between zones or even between interfaces and/or VPN tunnels in a zone.

Example Firewall Behavior

Firewall rules are grouped based on the direction of travel of packets to which they apply. Here is example firewall behavior for traffic going through the ZyWALL in various directions. See the **Configuration > Firewall** screen for default firewall behavior.

Note: At the time of writing the ZyWALL's VPN and GRE tunnels support IPv4 traffic so IPv6 firewall rules do not apply to IPSec, SSL VPN, and GRE tunnel traffic.

FROM ZONE TO ZONE	BEHAVIOR
From any to ZyWALL	DHCP traffic from any interface to the ZyWALL is allowed.
	DHCPv6 and Default_Allow_ICMPv6_Group traffic from any interface to the ZyWALL is allowed.
From LAN to any (other than the ZyWALL)	Traffic from the LAN to any of the networks connected to the ZyWALL is allowed.
From DMZ to WAN	Traffic from the DMZ to the WAN is allowed.
From IPSec VPN to any (other than the ZyWALL)	Traffic from the IPSec VPN zone to any of the networks connected to the ZyWALL is allowed.
From SSL VPN to any (other than the ZyWALL)	Traffic from the SSL VPN zone to any of the networks connected to the ZyWALL is allowed.
From TUNNEL to any (other than the ZyWALL)	Traffic from the TUNNEL zone to any of the networks connected to the ZyWALL is allowed.
From LAN to ZyWALL	Traffic from the LAN to the ZyWALL itself is allowed.
From DMZ to ZyWALL	DNS and NetBIOS traffic from the DMZ to the ZyWALL itself is allowed.
From WAN to ZyWALL	The default services listed in To-ZyWALL Rules on page 257 are allowed from the WAN to the ZyWALL itself. All other WAN to ZyWALL traffic is dropped.
From IPSec VPN to ZyWALL	Traffic from the IPSec VPN zone to the ZyWALL itself is allowed.
From SSL VPN to ZyWALL	Traffic from the SSL VPN zone to the ZyWALL itself is allowed.
From TUNNEL to ZyWALL	Traffic from the TUNNEL zone to the ZyWALL itself is allowed.
From any to any	Traffic that does not match any firewall rule is dropped. This includes traffic from the DMZ or WAN to any of the networks behind the ZyWALL and traffic other than DNS and NetBIOS from the DMZ to the ZyWALL.
	This also includes traffic to or from interfaces or VPN tunnels that are not assigned to a zone (extra-zone traffic).

Table 97 Example Firewall Behavior

To-ZyWALL Rules

Rules with **ZyWALL** as the **To Zone** apply to traffic going to the ZyWALL itself. By default:

- The firewall allows only LAN, WLAN, or WAN computers to access or manage the ZyWALL.
- The ZyWALL allows DHCP traffic from any interface to the ZyWALL.
- The ZyWALL allows DHCPv6 and Default_Allow_ICMPv6_Group traffic from any interface to the ZyWALL.
- The ZyWALL drops most packets from the DMZ zone to the ZyWALL itself and generates a log except for DNS and NetBIOS traffic.
- The ZyWALL drops most packets from the WLAN zone to the ZyWALL itself and generates a log except for BOOTP_SERVER, HTTP, HTTPS, and DNS traffic.
- The ZyWALL drops most packets from the WAN zone to the ZyWALL itself and generates a log except for AH, ESP, GRE, HTTPS, IKE, NATT (NATT applies to IPv4 only), and VRRP traffic.

When you configure a firewall rule for packets destined for the ZyWALL itself, make sure it does not conflict with your service control rule. See Chapter 37 on page 432 for more information about service control (remote management). The ZyWALL checks the firewall rules before the service control rules for traffic destined for the ZyWALL.

A From Any To ZyWALL direction rule applies to traffic from an interface which is not in a zone.

Global Firewall Rules

Firewall rules with **from any** and/or **to any** as the packet direction are called global firewall rules. The global firewall rules are the only firewall rules that apply to an interface or VPN tunnel that is not included in a zone. The **from any** rules apply to traffic coming from the interface and the **to any** rules apply to traffic going to the interface.

Firewall Rule Criteria

The ZyWALL checks the schedule, user name (user's login name on the ZyWALL), source IP address, destination IP address and IP protocol type of network traffic against the firewall rules (in the order you list them). When the traffic matches a rule, the ZyWALL takes the action specified in the rule.

User Specific Firewall Rules

You can specify users or user groups in firewall rules. For example, to allow a specific user from any computer to access a zone by logging in to the ZyWALL, you can set up a rule based on the user name only. If you also apply a schedule to the firewall rule, the user can only access the network at the scheduled time. A user-aware firewall rule is activated whenever the user logs in to the ZyWALL and will be disabled after the user logs out of the ZyWALL.

Firewall and VPN Traffic

After you create a VPN tunnel and add it to a zone, you can set the firewall rules applied to VPN traffic. If you add a VPN tunnel to an existing zone (the LAN1 zone for example), you can configure a new LAN1 to LAN1 firewall rule. If you add the VPN tunnel to a new zone (the VPN zone for example), you can configure rules for VPN traffic between the VPN zone and other zones or **From VPN To-ZyWALL** rules for VPN traffic destined for the ZyWALL.

Session Limits

Accessing the ZyWALL or network resources through the ZyWALL requires a NAT session and corresponding firewall session. Peer to peer applications, such as file sharing applications, may use a large number of NAT sessions. A single client could use all of the available NAT sessions and prevent others from connecting to or through the ZyWALL. The ZyWALL lets you limit the number of concurrent NAT/firewall sessions a client can use.

Finding Out More

• See Section 19.4 on page 267 for an example of creating firewall rules as part of configuring user-aware access control.

19.2 The Firewall Screen

Asymmetrical Routes

If an alternate gateway on the LAN has an IP address in the same subnet as the ZyWALL's LAN IP address, return traffic may not go through the ZyWALL. This is called an asymmetrical or "triangle" route. This causes the ZyWALL to reset the connection, as the connection has not been acknowledged.

You can have the ZyWALL permit the use of asymmetrical route topology on the network (not reset the connection). However, allowing asymmetrical routes may let traffic from the WAN go directly to the LAN without passing through the ZyWALL. A better solution is to use virtual interfaces to put the ZyWALL and the backup gateway on separate subnets. Virtual interfaces allow you to partition your network into logical sections over the same interface. See the chapter about interfaces for more information.

By putting LAN 1 and the alternate gateway (**A** in the figure) in different subnets, all returning network traffic must pass through the ZyWALL to the LAN. The following steps and figure describe such a scenario.

- 1 A computer on the LAN1 initiates a connection by sending a SYN packet to a receiving server on the WAN.
- 2 The ZyWALL reroutes the packet to gateway **A**, which is in **Subnet 2**.
- **3** The reply from the WAN goes to the ZyWALL.
- 4 The ZyWALL then sends it to the computer on the LAN1 in Subnet 1.Figure 162 Using Virtual Interfaces to Avoid Asymmetrical Routes



19.2.1 Configuring the Firewall Screen

Click **Configuration > Firewall** to open the **Firewall** screen. Use this screen to enable or disable the firewall and asymmetrical routes, set a maximum number of sessions per host, and display the configured firewall rules. Specify from which zone packets come and to which zone packets travel to display only the rules specific to the selected direction. Note the following.

- Besides configuring the firewall, you also need to configure NAT rules to allow computers on the WAN to access LAN devices. See Chapter 13 on page 217 for more information.
- The ZyWALL applies NAT (Destination NAT) settings before applying the firewall rules. So for example, if you configure a NAT entry that sends WAN traffic to a LAN IP address, when you configure a corresponding firewall rule to allow the traffic, you need to set the LAN IP address as the destination.
- The ordering of your rules is very important as rules are applied in sequence.

m Zo	one: any		~	To Zone	e:	any	2	~	Refresh			
Add	d 🍞 Edit 🍵 F	Remove 🨡 Ac	tivate 😡 Ina	ctivate 🔐M	ove							
at	Priority -	From	То	Schedule	User		IPv4 Source	IPv4 Destin	Service	Access	Log	1
)	1	a LAN	any (Exclu	none	any		any	any	any	allow	no	
)	2	n DMZ	■ WAN	none	any		any	any	any	allow	no	
>	3	N WLAN	■ WAN	none	any		any	any	any	allow	no	
)	4	■ IPSec_VPN	any (Exclu	none	any		any	any	any	allow	no	
)	5	SSL_VPN	any (Exclu	none	any		any	any	any	allow	no	
2	6		any (Exclu	none	any		any	any	any	allow	no	
?	7	LAN	ZyWALL	none	any		any	any	any	allow	no	
2	8	■ DMZ	ZyWALL	none	any		any	any	■ Default	allow	no	
?	9	R WLAN	ZyWALL	none	any		any	any	■ Default	allow	no	
2	10	R WAN	ZyWALL	none	any		any	any	■ Default	allow	no	
?	11	RIPSec_VPN	ZyWALL	none	any		any	any	any	allow	no	
2	12	SSL_VPN	ZyWALL	none	any		any	any	any	allow	no	
}	13	a TUNNEL	ZyWALL	none	any		any	any	any	allow	no	
	Default	any	any	none	any		any	any	any	deny	log	
m zo	any			10 2010	e:	any			Refresh			
Add Add Add Add Add Add Add Add Add Add	d 2 Edit 1 F Priority A Ile Summary	Remove 🤪 Ac From al Route	tivate @ Ina. To	Ctivate MM Schedule	e: ove User	any	IPv6 Source	IPv6 Destin	Refresh	Access	s Log	
Add at 6 Ru All rom 2	d Celt from Priority A le Summary	Remove <table-cell> Ac From al Route</table-cell>	To	Ctivate MM Schedule To Zor	e: User he:	any any	IPv6 Source	IPv6 Destin	Refresh Refresh	Access	: Log	
Add Add Add All Com Z	d 2 Edit 1 Final Priority A le Summary	Remove @ Ac From al Route	tivate @ Ina To @ Activate 1 To	I o Zon ctivate MM Schedule To Zor Inactivate Schedule	e: User ne:	any any love	IPv6 Source	IPv6 Destin	Refresh Service Refresh	Access	Log	Log
Add at 6 Ru All om 2 All Statu	d 2 Edit 1 Friority A le Summary	Remove @ Ac From al Route Remove I From any	ttivate @ Ina To @ Activate @ To ZyWALL	To Zon Schedule To Zor Inactivate Schedule	e: User ne: User U	any any love lser any	IPv6 Source IPv6 So any	IPv6 Destin v	Refresh Service Refresh estinatio Servic	Access	Log Access allow	Log
) Add at 6 Ru 3 All oom 2 3 A Statu 9 9	d 2 Edit 1 Priority A le Summary ow Asymmetric Zone: any dd 2 Edit Priority A 1 2	Remove Q Ac From al Route Remove From any LAN1	Ctivate @ Ina To Activate @ To ZyWALL any (Exclu	To Zon Schedule To Zor Inactivate Schedule none ding none	e: User ne: Uuser M u a a	any any love lser iny iny	IPv6 Source	IPv6 Destin v urce IPv6 D any any	Refresh Service Refresh estinatio Servico « Defa any	Access	Log Access allow allow	Log no
Add at 6 Ru om 2 0 All 0 All 0 All 0 All 0 All 0 C 0 C	d 2 Edit 1 Priority A Ile Summary low Asymmetric Zone: any dd 2 Edit 1 2 3	Remove Q Ac From al Route From any aLAN1 a LAN2	Ctivate @ Ina To Activate @ To ZyWALL any (Exclute any (Exclute	To Zon Schedule To Zon Inactivate Schedule none ding none ding none	e: User ne: Uuser U a a a	any any love lser iny iny	IPv6 Source	IPv6 Destin v urce IPv6 D any any any	Refresh Service Refresh estinatio Servic « Defa any any	Access	Access allow allow allow	Log no no no
M 20 Add at 6 Ru 3 All oom 2 3 A Statu 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	d 2 Edit 1 Priority A low Asymmetric Zone: any dd 2 Edit 1 Final E Priority A 1 2 3 4	Remove Q Ac From al Route From any aLAN1 aLAN2 aDMZ	Ctivate @ Ina To Contract of the second Contract of the second of the se	To Zon Schedule To Zon Inactivate Schedule none ding none none	e: User ne: NM M a a a a	any any love lser iny iny iny	IPv6 Source	IPv6 Destin urce IPv6 D any any any any	Refresh Service Refresh estinatio Servic a Defa any any any any	Access	Access allow allow allow allow	Log no no no no no
Add at 6 Ru 6 Ru 7 All rom 2 2 All 7 All 7 All 8 5 tatu 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	d 2 Edit 1 Priority A la Summary low Asymmetric Zone: any dd 2 Edit 1 s Priority A 1 2 3 4 5	Remove @ Ac From al Route From any a LAN1 a LAN1 a LAN2 a DMZ a WLAN	Citivate @ Ina To Citivate @ Ina To Citivate Citivate To ZyWALL any (Exclur any (Exclur) any (Exclur any (Exclur) any	To Zon Schedule To Zor Inactivate Schedule none ding none ding none none none	e: User e: User u a a a a a a a	any any love lser iny iny iny iny	IPv6 Source	IPv6 Destin urce IPv6 D any any any any any any	Refresh Service Refresh estinatio Servic <u>a Defa</u> any any any any	Access	Access allow allow allow allow allow	Log no no no no no no no
M 20 Add at 6 Ru 6 Ru 7 All om 2 6 Ru 7 All 6 Ru 9 9 9 9 9 9 9 9 9 9 9 9 9	d det in the second sec	Remove P Ac From al Route From any aLAN1 aLAN2 aDMZ aWLAN aIPSec_VPI	Citivate @ Ina To Contract Con	To Zon Schedule To Zon Inactivate Schedule none ding none ding none none ding none none ding none	e: User ne: User U a a a a a a a	any	IPv6 Source IPv6 Source any any any any any any any any any	IPv6 Destin urce IPv6 D any any any any any any any any any	Refresh Service Refresh estinatio Servic « Defa any any any any any any	Access	Access allow allow allow allow allow allow	Log no no no no no no no no no
M 20 Add at 6 Ru 6 Ru 7 All rom 2 0 A 8 8 9 9 9 9 9 9 9 9 9 9 9 9 9	d det in any Priority A ale Summary iow Asymmetric Zone: any dd det fill s Priority A 1 2 3 4 5 6 7	Remove P Ac From al Route Remove P From any aLAN1 aLAN2 aDMZ aWLAN aIPSec_VPI aSSL_VPN	Citivate @ Ina To Activate @ Activate To ZyWALL any (Exclur any (Exclur any (Exclur any (Exclur any (Exclur any (Exclur any (Exclur any (Exclur any (Exclur	To Zon Schedule To Zon Inactivate Schedule none ding none ding none none ding none ding none ding none	e: User ne: Uuser Uusa a a a a a a a a a a a a a a a a a a	any	IPv6 Source IPv6 Source any any any any any any any any any any	IPv6 Destin urce IPv6 D any any any any any any any any	Refresh Service Refresh estinatio Servic a Defa any any any any any any any any	Access	Access allow allow allow allow allow allow allow allow	Log no no no no no no no no no no no no
M 20 Add at 6 Ru 6 Ru 9 All om 2 9 A 8 5tatu 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	d 2 Edit 1 Priority A le Summary ow Asymmetric Zone: any dd 2 Edit 9 Priority A 1 2 3 4 5 6 7 8	Remove @ Ac From al Route From any aLAN1 aLAN2 aDMZ aWLAN aIPSec_VPI aSSL_VPN aTUNNEL	Citivate @ Ina To Citivate @ Ina To Citivate Cit	To Zon Schedule To Zon Inactivate Schedule none ding none ding none ding none ding none ding none ding none	e: User ne: dN M 2 4 4 4 4 a a a a a a a a a a a a a a a	any	IPv6 Source IPv6 So any any any any any any any any any any	IPv6 Destin urce IPv6 D any any any any any any any any	Refresh Service Refresh estinatio servic a Defa any any any any any any any any any an	e aut_Alo	Access allow allow allow allow allow allow allow allow allow	Log no no no no no no no no no no no
Add at 6 Ru at 6 Ru at 6 Ru at 7 Al. 7 8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	d 2 Edit 1 Priority A le Summary ow Asymmetric Zone: any dd 2 Edit Priority A 1 2 3 4 5 6 7 8 9	Remove Acception Acceptication Acception Acception Acception Acception Acception Accep	Citivate @ Ina To To Activate Citivate	To Zon Schedule To Zon Inactivate Schedule none ding none ding none ding none ding none ding none ding none none	e: User ne: User User a a a a a a a a a a a a a a a a a a a	any	IPv6 Source	IPv6 Destin	Refresh Service estinatio servic a.Defa any any any any any any any any any an	Access	Log Access allow allow allow allow allow allow allow allow allow allow	Log no no no no no no no no no no no no no
M 20 Add at 6 Ru 3 All om 2 3 A 3 All om 2 3 A 3 All 3 Al	d 2 Edit 1 Priority A le Summary low Asymmetric Zone: any dd 2 Edit 1 1 2 3 4 5 6 7 8 9 10	Remove @ Ac From al Route From any a LAN1 a LAN2 a DMZ a WLAN a IPSec_VPI a SSL_VPN a SSL_VPN a TUNNEL a LAN1 a LAN2	Citivate @ Ina To To Activate Citivate	To Zon Schedule To Zon To Zon To Zon To Zon Schedule none ding none ding none ding none ding none ding none ding none ding none none	e: User ne: User User User U a a a a a a a a a a a a a a a a a a	any any any love love love love love love love love	IPv6 Source IPv6 So any any any any any any any any any any	IPv6 Destin	Refresh Service Refresh estinatio any any any any any any any any any any	Access	Log Log Access allow allow allow allow allow allow allow allow allow allow	Log no no no no no no no no no no
M 20 Add at 6 Ru All oom 2 2 A 3 All oom 2 2 A 3 Statu 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	d 2 Edit 1 Priority A le Summary low Asymmetric Zone: any dd 2 Edit 1 s Priority A 1 2 3 4 5 6 7 8 9 10 11	Remove @ Ac From al Route Remove 1 From any LAN1 LAN1 CAN2 CAN2 CAN2 CAN2 CAN2 CAN2 CAN2 CAN2	Citivate @ Ina To To Citivate @ Ina To Citivate @ Ina To Citivate @ Ina Citivate	To Zon Schedule To Zon Inactivate Schedule None ding none ding none ding none ding none ding none ding none none ding none none ding none none	e: User ne: User User U a a a a a a a a a a a a a a a a a a	any	IPv6 Source IPv6 So any any any any any any any any any any	Urce IPv6 Destin any any any any any any any any any any	Refresh Service Refresh estinatio servic sector a Defa any any any any any any any any any an	Access	Access allow allow allow allow allow allow allow allow allow allow allow allow allow	Log no no no no no no no no no no
Additional and a constraint of the constraint of	d det in any Priority A ale Summary iow Asymmetric Zone: any dd det in any second second second second second second second second second second second second second second se	Remove @ Ac From al Route Remove U From any aLAN1 aLAN2 aDMZ aWLAN aSS_VPN aSS_VPN aTUNNEL aLAN1 aLAN2 aDMZ aWLAN	Citivate @ Ina To To ZyWALL any (Exclur any (Exclur any (Exclur any (Exclur any (Exclur any (Exclur any (Exclur any (Exclur zyWALL ZyWALL ZyWALL ZyWALL ZyWALL	To Zon Schedule To Zon Inactivate Schedule none ding none ding none ding none ding none ding none ding none none none none none none	e: User ne: User U a a a a a a a a a a a a a a a a a a	any	IPv6 Source IPv6 Source any any any any any any any any any any	Urce IPv6 Destin any any any any any any any any any any	Refresh Service Refresh estinatio service a Defa any any any any any any any any any an	e ult_Allo	Log Access allow allow allow allow allow allow allow allow allow allow allow allow allow	Log no no no no no no no no no no
Add at 6 Ru 6 Ru 9 All or Z 9 All 0 All	d det in the second sec	Remove @ Ac From al Route Remove U From any aLAN1 aLAN2 a_DMZ a_WLAN a_IPSec_VPI a_SSL_VPN a_SSL_VPN a_IPSec_VPI a_SSL_VPN a_IDNZ a_LAN1 a_LAN2 a_DMZ a_UNNEL a_LAN2 a_DMZ a_UNNEL a_LAN2 a_UNNEL a_LAN2 a_UNZ a_UNNEL a_UNZ a UNZ a UNZ A A UNZ A A A A A A A A A A A A A A A A A A A	Citivate @ Ina To To Citivate	To Zon Schedule To Zon Inactivate Schedule none ding none ding none ding none ding none ding none ding none ding none none none none none none none none	e: User ne: User a a a a a a a a a a a a a a a a a a a	any	IPv6 Source IPv6 S	IPv6 Destin urce IPv6 D any any any any any any any any	Refresh Service Refresh estinatio service serv	e ult_Allo	Log Access allow	Log no no
Add at 6 Ru 6 Ru 9 All or Z 9 A 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	any and a second	Remove P Ac From al Route Remove From any aLAN1 aLAN2 aUAN aVLAN AVLAN AV	Citivate @ Ina To Citivate @ Ina To Activate @ Activate @ Activate @ To ZyWALL ZyWALL ZyWALL ZyWALL ZyWALL ZyWALL XWALL ZyWALL XWALL ZYWALL ZYWALL ZYWALL ZYWALL ZY	To Zon Schedule To Zon Inactivate Schedule none ding none ding none ding none ding none ding none ding none ding none none none none none none none none	e: User e: User a a a a a a a a a a a a a a a a a a a	any	IPv6 Source IPv6 Source any any any any any any any any	Urce IPv6 Destin urce IPv6 D any any any any any any any any	Refresh Service Refresh estinatio servic a Defa any any any any any any any any any an	e ut_Allo	Log Access allow allow allow allow allow allow allow allow allow allow allow allow allow allow allow	Log no no no no no no no no no no no no no
Add at 6 Ru 6 Ru 7 All 7 All 7 All 7 All 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	d 2 Edit 1 Priority A le Summary ow Asymmetric Zone: any dd 2 Edit 1 any edit 2 Edit Priority A 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15	Remove P Ac From al Route Remove From any aLAN1 aLAN1 aLAN2 aDMZ aVLAN aLAN1 aLAN2 aDMZ aUNNEL aLAN1 aLAN2 aVLAN aLAN1 aLAN2 aVLAN aLAN2 aVLAN aLAN2 aVLAN aLAN2 aVLAN aLAN2 aVLAN AVLAN AV	Citivate @ Ina To Citivate @ Ina To Activate @ Act	To Zon Ctivate MM Schedule To Zon To Zon To Zon To Zon Schedule To Zon Schedule Inone ding none ding none ding none ding none ding none inone none none none none none none	e: User ne: User a a a a a a a a a a a a a a a a a a a	any	IPv6 Source IPv6 Source any any any any any any any any	IPv6 Destin urce IPv6 D any any any any any any any any	Refresh Service Refresh estinatio servic a Defa any any any any any any any any any an	e ult_Allo ult_Allo ult_Allo ult_Allo	Log Access allow	Log no no no no no no no no no no
Add at 6 Ru 6 Ru 7 All 7 Om 2 2 A 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	d 2 Edit 1 Priority A le Summary ow Asymmetric Zone: any dd 2 Edit 1 S Priority A 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16	Remove Ac From al Route Remove From any aLAN1 aLAN1 aLAN2 aDMZ aWLAN aUPSec_VPI aLAN1 aLAN2 aWLAN aLAN2 aWLAN aLAN2 aWLAN aLAN2 aWLAN aLAN2 aWLAN aLAN2 aWLAN aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aWLAN aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 aLAN2 aLAN1 aLAN2 aLAN1 aLAN2 ALAN2	Citivate @ Ina To Citivate @ Ina To Citivate @ Ina To Citivate @ Ina To Citivate @ Ina Citivate C	To Zon Ctivate MM Schedule To Zon To Zon To Zon To Zon Schedule To Zon Schedule Inone Inon	e: User ne:	any	IPv6 Sources IPv6 Sources IPv6 So any any any any any any any any any any	IPv6 Destin Urce IPv6 D any	Refresh Refresh estinatio service a Defa any any any any any any any any any an	e ut_Allo ut_Allo ut_Allo ut_Allo	Log Access allow	Log no no no no no no no no no no
Add att 6 Ru 6 Ru 1 All rom 2 2 A 3 Attu 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	d 2 Edit 1 Priority A le Summary ow Asymmetric Zone: any d 2 Edit 1 Priority A Priority A 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 Default	Remove Ar	Citivate @ Ina To Citivate @ Ina To Citivate @ Ina To Citivate @ Ina To Citivate @ Ina Citivate C	To Zon Ctivate MM Schedule To Zon To	e: User ne: User User User U a a a a a a a a a a a a a a a a a a	any	IPv6 Source IPv6 So any any any any any any any any any any	IPv6 Destin IPv6 Destin Iurce IPv6 D any	Refresh Service Refresh estinatio any any any any any any any any any any	Access	Log Access allow al	Log no no no no no no no no no no

Figure 163 Configuration > Firewall

The following table describes the labels in this screen.

LABEL	DESCRIPTION
General Settings	
Enable Firewall	Select this check box to activate the firewall. The ZyWALL performs access control when the firewall is activated.
IPv4 / IPv6 Rule Summary	Separate firewall rules for IPv4 and IPv6 traffic appear when you enable the ZyWALL's global IPv6 option, otherwise the rules are just for IPv4 traffic.
Allow Asymmetrical Route	If an alternate gateway on the LAN has an IP address in the same subnet as the ZyWALL's LAN IP address, return traffic may not go through the ZyWALL. This is called an asymmetrical or "triangle" route. This causes the ZyWALL to reset the connection, as the connection has not been acknowledged.
	Select this check box to have the ZyWALL permit the use of asymmetrical route topology on the network (not reset the connection).
	Note: Allowing asymmetrical routes may let traffic from the WAN go directly to the LAN without passing through the ZyWALL. A better solution is to use virtual interfaces to put the ZyWALL and the backup gateway on separate subnets.
From Zone / To Zone	This is the direction of travel of packets. Select from which zone the packets come and to which zone they go.
	Firewall rules are grouped based on the direction of travel of packets to which they apply. For example, from LAN1 to LAN1 means packets traveling from a computer or subnet on the LAN to either another computer or subnet on the LAN1.
	From any displays all the firewall rules for traffic going to the selected To Zone .
	To any displays all the firewall rules for traffic coming from the selected From Zone.
	From any to any displays all of the firewall rules.
	To ZyWALL rules are for traffic that is destined for the ZyWALL and control which computers can manage the ZyWALL.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate .
Inactivate	To turn off an entry, select it and click Inactivate.
Move	To change a rule's position in the numbered list, select the rule and click Move to display a field to type a number for where you want to put that rule and press [ENTER] to move the rule to the number that you typed.
	The ordering of your rules is important as they are applied in order of their numbering.
The following read selected packet d	d-only fields summarize the rules you have created that apply to traffic traveling in the irection.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Priority	This is the position of your firewall rule in the global rule list (including all through-ZyWALL and to-ZyWALL rules). The ordering of your rules is important as rules are applied in sequence. Default displays for the default firewall behavior that the ZyWALL performs on traffic that does not match any other firewall rule.
From	This is the direction of travel of packets to which the firewall rule applies.
То	

 Table 98
 Configuration > Firewall

LABEL	DESCRIPTION
Schedule	This field tells you the schedule object that the rule uses. none means the rule is active at all times if enabled.
User	This is the user name or user group name to which this firewall rule applies.
IPv4 / IPv6 Source	This displays the IPv4 or IPv6 source address object to which this firewall rule applies.
IPv4 / IPv6 Destination	This displays the IPv4 or IPv6 destination address object to which this firewall rule applies.
Service	This displays the service object to which this firewall rule applies.
Access	This field displays whether the firewall silently discards packets (deny), discards packets and sends a TCP reset packet to the sender (reject) or permits the passage of packets (allow).
Log	This field shows you whether a log (and alert) is created when packets match this rule or not.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 98
 Configuration > Firewall (continued)

19.2.2 The Firewall Add/Edit Screen

In the Firewall screen, click the Edit or Add icon to display the Firewall Rule Edit screen.

Add Firewall Rule				?
🗐 Create new Object 👻				
🔽 Enable				
From:	any	~		
To:	any (Excluding ZyWALL)	~		
Description:		(Option	al)	
Schedule:	none	~		
User:	any	~		
Source:	any	~		
Destination:	any	~		
Service:	any	~		
Access:	allow	~		
Log:	no	~		
		-		
		0	К	Cancel

Figure 164 Configuration > Firewall > Add

The following table describes the labels in this screen.

Table 99Configuration > Firewall > Add

LABEL	DESCRIPTION
Create new Object	Use to configure any new settings objects that you need to use in this screen.
Enable	Select this check box to activate the firewall rule.

LABEL	DESCRIPTION
From	For through-ZyWALL rules, select the direction of travel of packets to which the rule applies.
То	any means all interfaces or VPN tunnels.
	ZyWALL means packets destined for the ZyWALL itself.
Description	Enter a descriptive name of up to 60 printable ASCII characters for the firewall rule. Spaces are allowed.
Schedule	Select a schedule that defines when the rule applies. Otherwise, select none and the rule is always effective.
User	This field is not available when you are configuring a to-ZyWALL rule.
	Select a user name or user group to which to apply the rule. The firewall rule is activated only when the specified user logs into the system and the rule will be disabled when the user logs out.
	Otherwise, select any and there is no need for user logging.
	Note: If you specified a source IP address (group) instead of any in the field below, the user's IP address should be within the IP address range.
Source	Select an IPv4 address or address group to apply an IPv4 rule to traffic coming from it. Select an IPv6 address or address group to apply an IPv6 rule to traffic coming from it. Select any to apply an IPv4 rule to all traffic coming from IPv4 addresses. Select any to apply an IPv6 rule to all traffic coming from IPv6 addresses.
Destination	Select an IPv4 address or address group to apply an IPv4 rule to traffic going to it. Select an IPv6 address or address group to apply an IPv6 rule to traffic going to it. Select any to apply an IPv4 rule to all traffic going to IPv4 addresses. Select any to apply an IPv6 rule to all traffic going to IPv6 addresses.
Service	Select a service or service group from the drop-down list box.
Access	Use the drop-down list box to select what the firewall is to do with packets that match this rule.
	Select deny to silently discard the packets without sending a TCP reset packet or an ICMP destination-unreachable message to the sender.
	Select reject to deny the packets and send a TCP reset packet to the sender. Any UDP packets are dropped without sending a response packet.
	Select allow to permit the passage of the packets.
Log	Select whether to have the ZyWALL generate a log (log), log and alert (log alert) or not (no) when the rule is matched. See Chapter 38 on page 474 for more on logs.
ОК	Click OK to save your customized settings and exit this screen.
Cancel	Click Cancel to exit this screen without saving.

Table 99 Configuration > Firewall > Add (continued)

19.3 The Session Limit Screen

Click **Configuration > Firewall > Session Limit** to display the **Firewall Session Limit** screen. Use this screen to limit the number of concurrent NAT/firewall sessions a client can use. You can apply a default limit for all users and individual limits for specific users, addresses, or both. The individual limit takes priority if you apply both.

Enable	e Session Li	mit				
Default Se	summary	lost: 0	0-8192, 0 is unlin	ited)		
O Add	📝 Edit	💼 Remove 💡 Activ	vate 🖗 Inactivate	Move		
Status	#	User	IPv4 Address	Description	Limit	
14.4	Page 1	of 1 🕨 🕅 Sho	w 50 💌 items		No data	to displa
P v6 Rule : Default Se	Summary ession per H	lost: 0] (0-8192, 0 is unlin	ited)		
Add	📝 Edit	👕 Remove 💡 Activ	vate 🖗 Inactivate	Move		
Status	#	User	IPv6 Address	Description	Limit	
14:4	Page 1	of 1 🕨 🔌 Sho	w 50 👻 items		No data	to displa

Figure 165 Configuration > Firewall > Session Limit

The following table describes the labels in this screen.

LABEL	DESCRIPTION
General Settings	
Enable Session limit	Select this check box to control the number of concurrent sessions hosts can have.
IPv4 / IPv6 Rule Summary	The IPv4 rules apply to IPv4 sessions. The IPv6 rules apply to IPv6 sessions.
Default Session per Host	Use this field to set a common limit to the number of concurrent NAT/firewall sessions each client computer can have.
	If only a few clients use peer to peer applications, you can raise this number to improve their performance. With heavy peer to peer application use, lower this number to ensure no single client uses too many of the available NAT sessions.
	Create rules below to apply other limits for specific users or addresses.
Rule Summary	This table lists the rules for limiting the number of concurrent sessions hosts can have.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Move	To change a rule's position in the numbered list, select the rule and click Move to display a field to type a number for where you want to put that rule and press [ENTER] to move the rule to the number that you typed.
	The ordering of your rules is important as they are applied in order of their numbering.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
#	This is the index number of a session limit rule. It is not associated with a specific rule.

 Table 100
 Configuration > Firewall > Session Limit

LABEL	DESCRIPTION
User	This is the user name or user group name to which this session limit rule applies.
IPv4 Address	This is the IPv4 address object to which this session limit rule applies.
IPv6 Address	This is the IPv6 address object to which this session limit rule applies.
Description	This is the information configured to help you identify the rule.
Limit	This is how many concurrent sessions this user or address is allowed to have.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 100 Configuration > Firewall > Session Limit (continued)

19.3.1 The Session Limit Add/Edit Screen

Click **Configuration > Firewall > Session Limit** and the **Add** or **Edit** icon to display the **Firewall Session Limit Edit** screen. Use this screen to configure rules that define a session limit for specific users or addresses.

Figure 166	Configuration	>	Firewall	>	Session	l imit	>	Edit
ligule 100	conniguration	/	Thewan	_	26221011		/	LUII

5			
O Ad	Id Session Limit		? 🗙
8 C	ireate new Object 👻		
	🗸 Enable Rule		
	Description:		(Optional)
ι	Jser:	any	~
4	Address:	any	~
2	Session Limit per Host:	0 (0-8192, 0 is u	unlimited)
			OK Cancel

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Create new Object	Use to configure any new settings objects that you need to use in this screen.
Enable Rule	Select this check box to turn on this session limit rule.
Description	Enter information to help you identify this rule. Use up to 60 printable ASCII characters. Spaces are allowed.
User	Select a user name or user group to which to apply the rule. The rule is activated only when the specified user logs into the system and the rule will be disabled when the user logs out.
	Otherwise, select any and there is no need for user logging.
	Note: If you specified an IP address (or address group) instead of any in the field below, the user's IP address should be within the IP address range.
Address	Select the IPv4 source address or address group to which this rule applies. Select any to apply the rule to all IPv4 source addresses.
IPv6 Address	Select the IPv6 source address or address group to which this rule applies. Select any to apply the rule to all IPv6 source addresses.

 Table 101
 Configuration > Firewall > Session Limit > Edit

LABEL	DESCRIPTION
Session Limit per Host	Use this field to set a limit to the number of concurrent NAT/firewall sessions this rule's users or addresses can have.
	For this rule's users and addresses, this setting overrides the Default Session per Host setting in the general Firewall Session Limit screen.
ОК	Click OK to save your customized settings and exit this screen.
Cancel	Click Cancel to exit this screen without saving.

Table 101 Configuration > Firewall > Session Limit > Edit (continued)

19.4 Firewall Rule Configuration Example

The following Internet firewall rule example allows Doom players from the WAN to IP addresses 192.168.1.10 through 192.168.1.15 (Dest_1) on the LAN1.

1 Click **Configuration > Firewall**. In the summary of IPv4 firewall rules click **Add** to configure a new first entry. The sequence (priority) of the rules is important since they are applied in order.

Figure 1	67 Firewall Example: Firewall Screen
Firewall	Session Limit
General S	ettings

eneral	Setting	IS										
Ena	able Fire	wall										
From Z	ow Asymi	metrical Rou any	ute	V Additional Constant	To Zone:	BI Mar	any	✓ R	efresh			
Status	s Priority	From From From From From From From From	m	To	Schedule	Use	er IPv4 Source	IPv4 Destinatio	Service	Access	Log	1
Q	1	٩١	AN1	any (Excluding	none	any	any any	any	any	allow	no	
0	2	4 L	AN2	any (Excluding	none	any	any	any	any	allow	no	
0	3	9.0	MZ	■ WAN	none	any	any	any	any	allow	no	

2 At the top of the screen, click **Create new Object > Address** to configure an address object. Configure it as follows and click **OK**.

Name:	Dest_1	
Address Type:	RANGE	*
Starting IP Address:	192.168.1.10	
End IP Address:	192.168.1.15	

Figure 168 Firewall Example: Create an Address Object

3 Click Create new Object > Service to configure a service object for Doom (UDP port 666). Configure it as follows and click OK.

Name:	Doom	
IP Protocol:	UDP	*
Starting Port:	666	(165535)
Ending Port:	666	(165535)

4 Select From WAN and To LAN1 and enter a name for the firewall rule.

Select **Dest_1** for the **Destination** and **Doom** as the **Service**. Enter a description and configure the rest of the screen as follows. Click \mathbf{OK} when you are done.

Figure 170	Firewall	Example:	Edit	a Firev	vall Rule
Add Firewall R	tule From Zone	: any		~	To Zonur
🔚 Create new Ol	oject 🔻				
📝 Enable					
From:		WAN		*	
To:		LAN1		~	
Description:		Doom-example	(0	ptional)	
Schedule:		none		~	
User:		any		~	
Source:		any		~	
Destination:		Dest_1		~	
Service:		Doom		~	
Access:		allow		~	
Log:		no		*	
	9 14	WLAT ZYWALL		ОК	Cancel

---. . . ---.. _

The firewall rule appears in the firewall rule summary. 5

Figure 171 Firewall Example: Doom Rule in Summary

Firewal	Ses	sion Limit								
Genera	I Setting	5								
V Er	nable Firev	vall								
Pv4 R	ule Sumn	nary								
	llow Asymr	netrical Rout	te							
From	Zone:	any		~	To Zor	ne: a	ny		*	Refresh
© #	Add 📝 🛙	Edit 📋 Re	move 😡 A	tivate 🖗	Inactivate	📣 Move				
State	Priority 🔺	From	То	Schedule	User	IPv4 Source	IPv4 Destir	Service	Access	Log
୍ଭ	1	WAN	■LAN1	none	any	any	■ Dest_1	¤ Doom	allow	по
@	2	WAN	ZyWALL	none	any	any	any	■Default_/	allow	по
9	3	WAN	ZyWALL	none	any	any	any	any	deny	log
0	4	WAN	any (Exclu	none	any	any	any	any	deny	log

19.5 Firewall Rule Example Applications

Suppose you decide to block LAN users from using IRC (Internet Relay Chat) through the Internet. To do this, you would configure a LAN to WAN firewall rule that blocks IRC traffic from any source IP address from going to any destination address. You do not need to specify a schedule since you need the firewall rule to always be in effect. The following figure shows the results of this rule.



Figure 172 Blocking All LAN to WAN IRC Traffic Example

Your firewall would have the following rules.

#	USER	SOURCE	DESTINATION	SCHEDULE	SERVICE	ACTION
1	Any	Any	Any	Any	IRC	Deny
2	Any	Any	Any	Any	Any	Allow

 Table 102
 Blocking All LAN to WAN IRC Traffic Example

• The first row blocks LAN access to the IRC service on the WAN.

• The second row is the firewall's default policy that allows all LAN1 to WAN traffic.

The ZyWALL applies the firewall rules in order. So for this example, when the ZyWALL receives traffic from the LAN, it checks it against the first rule. If the traffic matches (if it is IRC traffic) the firewall takes the action in the rule (drop) and stops checking the firewall rules. Any traffic that does not match the first firewall rule will match the second rule and the ZyWALL forwards it.

Now suppose you need to let the CEO use IRC. You configure a LAN1 to WAN firewall rule that allows IRC traffic from the IP address of the CEO's computer. You can also configure a LAN to WAN rule that allows IRC traffic from any computer through which the CEO logs into the ZyWALL with his/her user name. In order to make sure that the CEO's computer always uses the same IP address, make sure it either:

· Has a static IP address,

or

• You configure a static DHCP entry for it so the ZyWALL always assigns it the same IP address (see DHCP Settings on page 173 for information on DHCP).

Now you configure a LAN1 to WAN firewall rule that allows IRC traffic from the IP address of the CEO's computer (192.168.1.7 for example) to go to any destination address. You do not need to specify a schedule since you want the firewall rule to always be in effect. The following figure shows the results of your two custom rules.



Figure 173 Limited LAN to WAN IRC Traffic Example

Your firewall would have the following configuration.

#	USER	SOURCE	DESTINATION	SCHEDULE	SERVICE	ACTION
1	Any	192.168.1.7	Any	Any	IRC	Allow
2	Any	Any	Any	Any	IRC	Deny
3	Any	Any	Any	Any	Any	Allow

 Table 103
 Limited LAN1 to WAN IRC Traffic Example 1

- The first row allows the LAN1 computer at IP address 192.168.1.7 to access the IRC service on the WAN.
- The second row blocks LAN1 access to the IRC service on the WAN.
- The third row is the firewall's default policy of allowing all traffic from the LAN1 to go to the WAN.

Alternatively, you configure a LAN1 to WAN rule with the CEO's user name (say CEO) to allow IRC traffic from any source IP address to go to any destination address.

Your firewall would have the following configuration.

#	USER	SOURCE	DESTINATION	SCHEDULE	SERVICE	ACTION
1	CEO	Any	Any	Any	IRC	Allow
2	Any	Any	Any	Any	IRC	Deny
3	Any	Any	Any	Any	Any	Allow

Table 104 Limited LAN1 to WAN IRC Traffic Example 2

• The first row allows any LAN1 computer to access the IRC service on the WAN by logging into the ZyWALL with the CEO's user name.

- The second row blocks LAN1 access to the IRC service on the WAN.
- The third row is the firewall's default policy of allowing all traffic from the LAN1 to go to the WAN.

The rule for the CEO must come before the rule that blocks all LAN1 to WAN IRC traffic. If the rule that blocks all LAN1 to WAN IRC traffic came first, the CEO's IRC traffic would match that rule and the ZyWALL would drop it and not check any other firewall rules.

IPSec VPN

20.1 Virtual Private Networks (VPN) Overview

A virtual private network (VPN) provides secure communications between sites without the expense of leased site-to-site lines. A secure VPN is a combination of tunneling, encryption, authentication, access control and auditing. It is used to transport traffic over the Internet or any insecure network that uses TCP/IP for communication.

IPSec VPN

Internet Protocol Security (IPSec) VPN connects IPSec routers or remote users using IPSec client software. This standards-based VPN offers flexible solutions for secure data communications across a public network. IPSec is built around a number of standardized cryptographic techniques to provide confidentiality, data integrity and authentication at the IP layer. The ZyWALL can also combine multiple IPSec VPN connections into one secure network. Here local ZyWALL **X** uses an IPSec VPN tunnel to remote (peer) ZyWALL **Y** to connect the local (**A**) and remote (**B**) networks.





SSL VPN

SSL VPN uses remote users' web browsers to provide the easiest-to-use of the ZyWALL's VPN solutions. A user just browses to the ZyWALL's web address and enters his user name and password to securely connect to the ZyWALL's network. Remote users do not need to configure security settings. Here a user uses his browser to securely connect to network resources in the same way as if he were part of the internal network. See Chapter 21 on page 308 for more on SSL VPN.



L2TP VPN

L2TP VPN uses the L2TP and IPSec client software included in remote users' Android, iOS, or Windows operating systems for secure connections to the network behind the ZyWALL. The remote users do not need their own IPSec gateways or third-party VPN client software. For example, configure sales representatives' laptops, tablets, or smartphones to securely connect to the ZyWALL's network. See Chapter 24 on page 335 for more on L2TP over IPSec.



20.1.1 What You Can Do in this Chapter

- Use the **VPN Connection** screens (see Section 20.2 on page 276) to specify which IPSec VPN gateway an IPSec VPN connection policy uses, which devices behind the IPSec routers can use the VPN tunnel, and the IPSec SA settings (phase 2 settings). You can also activate or deactivate and connect or disconnect each VPN connection (each IPSec SA).
- Use the **VPN Gateway** screens (see Section 20.2.1 on page 277) to manage the ZyWALL's VPN gateways. A VPN gateway specifies the IPSec routers at either end of a VPN tunnel and the IKE SA settings (phase 1 settings). You can also activate and deactivate each VPN gateway.
- Use the **VPN Concentrator** screens (see Section 20.4 on page 292) to combine several IPSec VPN connections into a single secure network.
- Use the **Configuration Provisioning** screen (see Section 20.5 on page 294) to set who can retrieve VPN rule settings from the ZyWALL using the ZyWALL IPSec VPN Client.

20.1.2 What You Need to Know

An IPSec VPN tunnel is usually established in two phases. Each phase establishes a security association (SA), a contract indicating what security parameters the ZyWALL and the remote IPSec router will use. The first phase establishes an Internet Key Exchange (IKE) SA between the ZyWALL and remote IPSec router. The second phase uses the IKE SA to securely establish an IPSec SA through which the ZyWALL and remote IPSec router can send data between computers on the local network and remote network. This is illustrated in the following figure.



Figure 177 VPN: IKE SA and IPSec SA

In this example, a computer in network **A** is exchanging data with a computer in network **B**. Inside networks **A** and **B**, the data is transmitted the same way data is normally transmitted in the networks. Between routers **X** and **Y**, the data is protected by tunneling, encryption, authentication, and other security features of the IPSec SA. The IPSec SA is secure because routers **X** and **Y** established the IKE SA first.

Application Scenarios

The ZyWALL's application scenarios make it easier to configure your VPN connection settings.

SITE-TO-SITE	SITE-TO-SITE WITH DYNAMIC PEER	REMOTE ACCESS (SERVER ROLE)	REMOTE ACCESS (CLIENT ROLE)
Static IP	Dynamic IP		
Choose this if the remote IPSec router has a static IP address or a domain name. This ZyWALL can initiate the VPN tunnel. The remote IPSec router can also initiate the VPN tunnel if this ZyWALL has a static IP address or a domain name.	Choose this if the remote IPSec router has a dynamic IP address. You don't specify the remote IPSec router's address, but you specify the remote policy (the addresses of the devices behind the remote IPSec router). This ZyWALL must have a static IP address or a domain name. Only the remote IPSec router can initiate the VPN tunnel.	Choose this to allow incoming connections from IPSec VPN clients. The clients have dynamic IP addresses and are also known as dial-in users. You don't specify the addresses of the client IPSec routers or the remote policy. This creates a dynamic IPSec VPN rule that can let multiple clients connect. Only the clients can initiate the VPN tunnel.	Choose this to connect to an IPSec server. This ZyWALL is the client (dial-in user). Client role ZyWALLs initiate IPSec VPN connections to a server role ZyWALL. This ZyWALL can have a dynamic IP address. The IPSec server doesn't configure this ZyWALL's IP address or the addresses of the devices behind it. Only this ZyWALL can initiate the VPN tuppel

 Table 105
 IPSec VPN Application Scenarios

Finding Out More

• See Section 20.6 on page 296 for IPSec VPN background information.

20.1.3 Before You Begin

This section briefly explains the relationship between VPN tunnels and other features. It also gives some basic suggestions for troubleshooting.

You should set up the following features before you set up the VPN tunnel.

- In any VPN connection, you have to select address objects to specify the local policy and remote policy. You should set up the address objects first.
- In a VPN gateway, you can select an Ethernet interface, virtual Ethernet interface, VLAN interface, or virtual VLAN interface to specify what address the ZyWALL uses as its IP address when it establishes the IKE SA. You should set up the interface first. See Chapter 7 on page 103.
- In a VPN gateway, you can enable extended authentication. If the ZyWALL is in server mode, you should set up the authentication method (AAA server) first. The authentication method specifies how the ZyWALL authenticates the remote IPSec router. See Chapter 31 on page 390.
- In a VPN gateway, the ZyWALL and remote IPSec router can use certificates to authenticate each other. Make sure the ZyWALL and the remote IPSec router will trust each other's certificates. See Chapter 33 on page 403.

20.2 The VPN Connection Screen

Click **Configuration > VPN > IPSec VPN** to open the **VPN Connection** screen. The **VPN Connection** screen lists the VPN connection policies and their associated VPN gateway(s), and various settings. In addition, it also lets you activate or deactivate and connect or disconnect each VPN connection (each IPSec SA). Click a column's heading cell to sort the table entries by that column's criteria. Click the heading cell again to reverse the sort order.

Figure 178 Configuration > VPN > IPSec VPN > VPN Connection

ilobal Setting Use Policy Ro	oute to control dynam	nic IPSec rules			
Use Policy Ro	oute to control dynam	nic IPSec rules			
Ignore "Don'	Fragment" catting in				
	criagilienc securig i	ı packet header 🛐			
onfiguration					
			-		
O Add 🦉 E	dit 📷 Remove 🌘	Activate 🖞 Inactivat	e 🍓 Conn	ect 🔥 Discon	nect 📑 Object Referen
# Status	Name V	PN Gateway	Encapsula	Algorithm	Policy
1 😨 🚱	ct it		TUNNEL	DES/SHA1	LAN1_SUBNET/
2 💡 🚱	WIZ_VPN W	VIZ_VPN	TUNNEL	DES/SHA1	WIZ_VPN_LOCA
3 😡 🚱	WIZ_VPN_PROVIS V	VIZ_VPN_PROVISIONING	TUNNEL	DES/SHA1	WIZ_VPN_PROV
14 4 Page	1 of 1 > >	Show 50 vitems			Displaying 1 - 3 of 3

Each field is discussed in the following table. See Section 20.2.2 on page 283 and Section 20.2.1 on page 277 for more information.

LABEL	DESCRIPTION
Use Policy Route to control dynamic IPSec rules	Select this to be able to use policy routes to manually specify the destination addresses of dynamic IPSec rules. You must manually create these policy routes. The ZyWALL automatically obtains source and destination addresses for dynamic IPSec rules that do not match any of the policy routes.
	Clear this to have the ZyWALL automatically obtain source and destination addresses for all dynamic IPSec rules.
Ignore "Don't Fragment" setting in packet header	Select this to fragment packets larger than the MTU (Maximum Transmission Unit) that have the "don't" fragment" bit in the IP header turned on. When you clear this the ZyWALL drops packets larger than the MTU that have the "don't" fragment" bit in the header turned on.
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Connect	To connect an IPSec SA, select it and click Connect.
Disconnect	To disconnect an IPSec SA, select it and click Disconnect .
Object Reference	Select an entry and click Object Reference to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific connection.
Status	The activate (light bulb) icon is lit when the entry is active and dimmed when the entry is inactive.
	The connect icon is lit when the interface is connected and dimmed when it is disconnected.
Name	This field displays the name of the IPSec SA.
VPN Gateway	This field displays the associated VPN gateway(s). If there is no VPN gateway, this field displays "manual key".
Encapsulation	This field displays what encapsulation the IPSec SA uses.
Algorithm	This field displays what encryption and authentication methods, respectively, the IPSec SA uses.
Policy	This field displays the local policy and the remote policy, respectively.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 106 Configuration > VPN > IPSec VPN > VPN Connection

20.2.1 The VPN Connection Add/Edit (IKE) Screen

The **VPN Connection Add/Edit Gateway** screen allows you to create a new VPN connection policy or edit an existing one. To access this screen, go to the **Configuration > VPN Connection** screen (see Section 20.2 on page 276), and click either the **Add** icon or an **Edit** icon.

and the second		
THE R. LEW.		
Connection Name:	1	
Correctormane.		
Naled-Up		
Enable Replay Detection		
Enable NetBIOS broadcast over IPSe	E .	
Pos Adjustment		
Custom Size		
Auto		
/PN Gateway		
Application Scenario		
Ste-to-site		
Site-to-site with Dynamic Peer		
Remote Access (Server Role)		
Demote Arress (Clent Dale)		
VPN Gateway:	Disass salast one	× 0
Manual Key	Lineage select ore	
Manual Key		
My Address:		
Secure Galeman Address		
SDT:	[246 - 400F	
England Johnson Manda	and a second	12
to August Albert Provel		- 500
Active Protocoli		
Encryption Algorithm;		
Authentication Algorithms		<u>v</u>
Erroryption Kleys		
Authendication Keys		
olicy		
Local policy:	Please select one	
Remote policy:	Please select one	× 0
Policy Enforcement		
these 3 Catting		
mase z secong		
SA Life Time:	86400	180 - 3000000 Seconds)
Active Protocol:	ESP	*
Encapsulation:	Tunnel	×
Proposal	🔕 Add 🔡 Edit 🍟 Re	move
	# Encryption A	uthentication
	1 DES S	HA1
Perfect Forward Secrecy (PFS):	none	•
Jalatad Catting		
lelated Settings		
Related Settings Zone:	IPSec_VPN	× 6
Related Settings Zone:	IPSec_VPN	
Telated Settings Zone: connectivity Check	IPSec_VPN	
Related Settings Zone: Connectivity Check Check Romectivity Check	IPSec_VPN	
Related Settings Zone: ionnectivity Check I Enable Connectivity Check 1 Check Method:	IPSec_VPN	
Related Settings Zone: Connectivity Check © Enable Connectivity Check Check Method: Oreck Period:	IPSec_VPN komp 5(5-30 Seconds)	
Related Settings Zone: ::onnectivity Check I stable Connectivity Check Check Herbod: Check Firefod: Check Timeout:	IPSec_VPN kmp 5 (5-30 Seconds) 5 (1-10 Seconds)	■ ■
Related Settings Zone: Connectivity Check Check Method: Check Method: Check Period: Check Theout: Check Fail Tulerance:		
telated Settings Zone: Connectivity Check:	IPSec_VPN s (5-30 Seconds) s (1-10 Seconds) (1-10)	Domain Name or 19 Address)
Tone: Connectivity Check:	IFSec_VFN smp s (5-30 Seconds) s (1-10 Seconds) use in the Remote Policy	Doman Name or IP Address)
Related Settings Zone: Iconnectivity Check I stable Connectivity Oreck ■ Oreck Period: Oreck Period: Oreck Period: Oreck Period: I rug This Address III Deck the First and Last IP Addres III Oreck the First and Last IP Address III Oreck the First and Last IP Address	PFSec_VFN S (5-30 Seconds) S (1-10 Seconds) (0) (0) (0) (0) (0) (0) (0)	Im I Im Doman Name or IP Address)
telated Settings Zone: Connectivity Check Check Methods Check Period: Check Period: Check Period: Check Theoret: Prog This Address Check the First and Last IP Addr Log	IPSec_VPN S (5-30 Seconds) S (1-10 Seconds) G=10 ses in the Remote Policy	Coman Name or IP Address)
telated Settings Zone: Connectivity Check: © Enable Connectivity Check: Oted: Method: Oted: Method: Oted: Parlot: Oted: Final Tule ance: © Prog The Address @ Check the First and Last IP Addre] Log nbound/Outbound traffic INIT	IPSec_VPN icmp 5 (5-30 Seconds) 5 (1-10 Seconds) 10 (1-10 Seconds)	Conan Hane or IP Address)
telated Settings Zone: Connectivity Check Check Herlod: Check Period: Check Period: Check Field Tolerance: Prog The Address Check the First and Last IP Addre Log nbound/Outbound traffic NAT Outbound Traffic	IFSec_VFN Interp S (5-30 Seconds) S (1-10 Seconds) B Ss in the Remote Policy	Doman Name or IP Addrese)
Interest Settings Zone: Zone: Connectivity Check Check Method: Check Aferiod: Check Aferiod: Check Aferiod: Check Aferiod: Check Aferiod: Check Afer Interance: Prog This Address Check the First and Last IP Addre Log Notound/Contound traffic NAT Outbound Traffic NAT Cutbound Traffic NAT	PSec_VPN Kmp 5 (5-30 Seconds) 5 (1-10 Seconds) 6 -10 ses in the Remote Policy	Doman Name or IP Address)
telated Settings Zone: connectivity Check: © Enable Connectivity Check: © Enable Connectivity Check: © Check Netricol: © Check Netricol: © Prog The Address @ Check the First and Last IP Addre @ Check the First and Last IP Addre @ Docks the First and Last IP Addre @ Inog nbound/Outbound traffic NAT Outbound Traffic @ Source NAT Source:	PSec_VPN kmp 5 (5-30 Seconds) 5 (1-10 Seconds) 9 (1-10 Seconds) 1 (1-10 Se	Donan Hune or IP Address)
telated Settings Zone: Connectivity Check: © Inable Connectivity Check: © Check Nethod: © Check Partod: © Check Timeout: © Check Timeout: © Check Timeout: © Prog Tites Address © Check The First and Last IP Addre © Lines for First and Last IP Addre © Source: Log nbound/Outbound traffic INAT © Source: Destination:	PSec_VPN rump S (5-30 Seconds) S (1-30 Secon	Doman Name or IP Address)
Interest Settings Zone: Zone: Conectivity Check Check Nethod: Check Period: Check Peri	PSec_VPN Kmp S (5-30 Seconds) S (1-10 Seconds) G-10 Sss in the Remote Policy Please select one Please select one	
telated Settings Zone: Connectivity Check: Check Method: Check Period: Check Period: Check Period: Check Timeout: Chec	Pfleer_VPN kmp 5 (5-30 Seconda) 5 (1-10 Seconda) 6 -10 esin the Remote Policy Pflease select one Pflease select one Pflease select one	Doman Name or IP Address)
telated Settings Zone: Connectivity Check Check Nethod: Check Nethod: Check Period:	PSec_VPN imp 5 (5-30 Seconds) 5 (1-10 Seconds) 10 (1-10 Seconds) 1	Doman Hame or IP Address)
Interest Settings Zone: Zone: Connectivity Check Check Nethod: Check Period: Check Per	PfSec_VPN Kmp S (5-30 Seconds) S (1-10 Seconds) G (1-10 Seconds) S (1-10 Seconds) Pfloate select one	Domain Hame or IP Addrese)
Interest Settings Zone: Connectivity Check: Check Methods Check Period:	PfSec_VPN Kmp S (5-30 Seconds) S (1-10 Seconds) (-10) S (-10) Pficase select one	Doman Name or IP Address)
telated Settings Zone: Connectivity Check:	PfSec_VPN kmp S (5-30 Seconds) S (1-30 Secon	Doman Hane or IP Address)
Interest Settings Zone: Connectivity Check: Check Nethod: Check Nethod: Check Period:	PSec_VPN Kmp S (5-30 Seconds) S (1-10 Seconds) C Please select one	Image: State of the set of the se
Interest Settings Zone: Zone: Check Hethods Check Alehods Check Alericot: Chec	PSec_VPN Kmp S (5-30 Seconda) S (1-10 Seconda) S (1-10 Seconda) Phase select one Please select one	Doman Name or IP Address)
telated Settings Zone: connectivity Check:	PSec_VPN xmp s (5-30 Seconds) s (1-10 Seconds) s (1-10 Seconds) s (1-10 Seconds) Filease select one Please select one Pleas	Doman Hane or IP Address)
telated Settings Zone: Connectivity Check: Check Nethod: Check Period:	PSec_VPN Kmp S (5-30 Seconds) S (1-10 Seconds) Control Please select one Please select one Please se	

Figure 179 Configuration > VPN > IPSec VPN > VPN Connection > Edit (IKE)

Each field is described in the following table.

Table 107 Configuration > VPN > IPSec VPN > VPN Connection > Ed	lit
---	-----

LABEL	DESCRIPTION
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
Create new Object	Use to configure any new settings objects that you need to use in this screen.
General Settings	
Enable	Select this check box to activate this VPN connection.
Connection Name	Type the name used to identify this IPSec SA. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Nailed-Up	Select this if you want the ZyWALL to automatically renegotiate the IPSec SA when the SA life time expires.
Enable Replay Detection	Select this check box to detect and reject old or duplicate packets to protect against Denial-of-Service attacks.
Enable NetBIOS Broadcast over IPSec	Select this check box if you the ZyWALL to send NetBIOS (Network Basic Input/ Output System) packets through the IPSec SA.
	NetBIOS packets are TCP or UDP packets that enable a computer to connect to and communicate with a LAN. It may sometimes be necessary to allow NetBIOS packets to pass through IPSec SAs in order to allow local computers to find computers on the remote network and vice versa.
MSS Adjustment	Select Custom Size to set a specific number of bytes for the Maximum Segment Size (MSS) meaning the largest amount of data in a single TCP segment or IP datagram for this VPN connection.
	Select Auto to have the ZyWALL automatically set the MSS for this VPN connection.
VPN Gateway	
Application	Select the scenario that best describes your intended VPN connection.
Scenario	Site-to-site - Choose this if the remote IPSec router has a static IP address or a domain name. This ZyWALL can initiate the VPN tunnel.
	Site-to-site with Dynamic Peer - Choose this if the remote IPSec router has a dynamic IP address. Only the remote IPSec router can initiate the VPN tunnel.
	Remote Access (Server Role) - Choose this to allow incoming connections from IPSec VPN clients. The clients have dynamic IP addresses and are also known as dial- in users. Only the clients can initiate the VPN tunnel.
	Remote Access (Client Role) - Choose this to connect to an IPSec server. This ZyWALL is the client (dial-in user) and can initiate the VPN tunnel.
VPN Gateway	Select the VPN gateway this VPN connection is to use or select Create Object to add another VPN gateway for this VPN connection to use.
Manual Key	Select this option to configure a VPN connection policy that uses a manual key instead of IKE key management. This may be useful if you have problems with IKE key management. See Section 20.2.2 on page 283 for how to configure the manual key fields.
	Note: Only use manual key as a temporary solution, because it is not as secure as a regular IPSec SA.
Policy	
Local Policy	Select the address corresponding to the local network. Use Create new Object if you need to configure a new one.

LABEL	DESCRIPTION
Remote Policy	Select the address corresponding to the remote network. Use Create new Object if you need to configure a new one.
Policy Enforcement	Clear this to allow traffic with source and destination IP addresses that do not match the local and remote policy to use the VPN tunnel. Leave this cleared for free access between the local and remote networks.
	Selecting this restricts who can use the VPN tunnel. The ZyWALL drops traffic with source and destination IP addresses that do not match the local and remote policy.
Phase 2 Settings	
SA Life Time	Type the maximum number of seconds the IPSec SA can last. Shorter life times provide better security. The ZyWALL automatically negotiates a new IPSec SA before the current one expires, if there are users who are accessing remote resources.
Active Protocol	Select which protocol you want to use in the IPSec SA. Choices are:
	AH (RFC 2402) - provides integrity, authentication, sequence integrity (replay resistance), and non-repudiation but not encryption. If you select AH , you must select an Authentication algorithm.
	ESP (RFC 2406) - provides encryption and the same services offered by AH , but its authentication is weaker. If you select ESP , you must select an Encryption algorithm and Authentication algorithm.
	Both AH and ESP increase processing requirements and latency (delay).
	The ZyWALL and remote IPSec router must use the same active protocol.
Encapsulation	Select which type of encapsulation the IPSec SA uses. Choices are
	Tunnel - this mode encrypts the IP header information and the data.
	Transport - this mode only encrypts the data.
	The ZyWALL and remote IPSec router must use the same encapsulation.
Proposal	Use this section to manage the encryption algorithm and authentication algorithm pairs the ZyWALL accepts from the remote IPSec router for negotiating the IPSec SA.
Add	Click this to create a new entry.
Edit	Select an entry and click this to be able to modify it.
Remove	Select an entry and click this to delete it.
#	This field is a sequential value, and it is not associated with a specific proposal. The sequence of proposals should not affect performance significantly.
Encryption	This field is applicable when the Active Protocol is ESP . Select which key size and encryption algorithm to use in the IPSec SA. Choices are:
	NULL - no encryption key or algorithm
	DES - a 56-bit key with the DES encryption algorithm
	3DES - a 168-bit key with the DES encryption algorithm
	AES128 - a 128-bit key with the AES encryption algorithm
	AES192 - a 192-bit key with the AES encryption algorithm
	AES256 - a 256-bit key with the AES encryption algorithm
	The ZyWALL and the remote IPSec router must both have at least one proposal that uses use the same encryption and the same key.
	Longer keys are more secure, but require more processing power, resulting in increased latency and decreased throughput.

 Table 107
 Configuration > VPN > IPSec VPN > VPN Connection > Edit (continued)

LABEL	DESCRIPTION
Authentication	Select which hash algorithm to use to authenticate packet data in the IPSec SA. Choices are SHA1, SHA256, SHA512 and MD5. SHA is generally considered stronger than MD5, but it is also slower.
	The ZyWALL and the remote IPSec router must both have a proposal that uses the same authentication algorithm.
Perfect Forward Secrecy (PFS)	Select whether or not you want to enable Perfect Forward Secrecy (PFS) and, if you do, which Diffie-Hellman key group to use for encryption. Choices are:
	none - disable PFS
	DH1 - enable PFS and use a 768-bit random number
	DH2 - enable PFS and use a 1024-bit random number
	DH5 - enable PFS and use a 1536-bit random number
	PFS changes the root key that is used to generate encryption keys for each IPSec SA. The longer the key, the more secure the encryption, but also the longer it takes to encrypt and decrypt information. Both routers must use the same DH key group.
Related Settings	
Zone	Select the security zone into which to add this VPN connection policy. Any security rules or settings configured for the selected zone apply to this VPN connection policy.
Connectivity Check	The ZyWALL can regularly check the VPN connection to the gateway you specified to make sure it is still available.
Enable Connectivity Check	Select this to turn on the VPN connection check.
Check Method	Select how the ZyWALL checks the connection. The peer must be configured to respond to the method you select.
	Select icmp to have the ZyWALL regularly ping the address you specify to make sure traffic can still go through the connection. You may need to configure the peer to respond to pings.
	Select tcp to have the ZyWALL regularly perform a TCP handshake with the address you specify to make sure traffic can still go through the connection. You may need to configure the peer to accept the TCP connection.
Check Port	This field displays when you set the Check Method to tcp . Specify the port number to use for a TCP connectivity check.
Check Period	Enter the number of seconds between connection check attempts.
Check Timeout	Enter the number of seconds to wait for a response before the attempt is a failure.
Check Fail Tolerance	Enter the number of consecutive failures allowed before the ZyWALL disconnects the VPN tunnel. The ZyWALL resumes using the first peer gateway address when the VPN connection passes the connectivity check.
Check this Address	Select this to specify a domain name or IP address for the connectivity check. Enter that domain name or IP address in the field next to it.
Check the First and Last IP Address in the Remote Policy	Select this to have the ZyWALL check the connection to the first and last IP addresses in the connection's remote policy. Make sure one of these is the peer gateway's LAN IP address.
Log	Select this to have the ZyWALL generate a log every time it checks this VPN connection.
Inbound/Outbound traffic NAT	
Outbound Traffic	

Table 107Configuration > VPN > IPSec VPN > VPN Connection > Edit (continued)

LABEL	DESCRIPTION
Source NAT	This translation hides the source address of computers in the local network. It may also be necessary if you want the ZyWALL to route packets from computers outside the local network through the IPSec SA.
Source	Select the address object that represents the original source address (or select Create Object to configure a new one). This is the address object for the computer or network outside the local network. The size of the original source address range (Source) must be equal to the size of the translated source address range (SNAT).
Destination	Select the address object that represents the original destination address (or select Create Object to configure a new one). This is the address object for the remote network.
SNAT	Select the address object that represents the translated source address (or select Create Object to configure a new one). This is the address object for the local network. The size of the original source address range (Source) must be equal to the size of the translated source address range (SNAT).
Inbound Traffic	
Source NAT	This translation hides the source address of computers in the remote network.
Source	Select the address object that represents the original source address (or select Create Object to configure a new one). This is the address object for the remote network. The size of the original source address range (Source) must be equal to the size of the translated source address range (SNAT).
Destination	Select the address object that represents the original destination address (or select Create Object to configure a new one). This is the address object for the local network.
SNAT	Select the address object that represents the translated source address (or select Create Object to configure a new one). This is the address that hides the original source address. The size of the original source address range (Source) must be equal to the size of the translated source address range (SNAT).
Destination NAT	This translation forwards packets (for example, mail) from the remote network to a specific computer (for example, the mail server) in the local network.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Select an entry and click this to be able to modify it.
Remove	Select an entry and click this to delete it.
Move	To change an entry's position in the numbered list, select it and click Move to display a field to type a number for where you want to put that entry and press [ENTER] to move the entry to the number that you typed.
#	This field is a sequential value, and it is not associated with a specific NAT record. However, the order of records is the sequence in which conditions are checked and executed.
Original IP	Select the address object that represents the original destination address. This is the address object for the remote network.
Mapped IP	Select the address object that represents the desired destination address. For example, this is the address object for the mail server.
Protocol	Select the protocol required to use this translation. Choices are: TCP, UDP, or All.
Original Port Start / Original Port End	These fields are available if the protocol is TCP or UDP . Enter the original destination port or range of original destination ports. The size of the original port range must be the same size as the size of the mapped port range.
Mapped Port Start / Mapped Port End	These fields are available if the protocol is TCP or UDP . Enter the translated destination port or range of translated destination ports. The size of the original port range must be the same size as the size of the mapped port range.

 Table 107
 Configuration > VPN > IPSec VPN > VPN Connection > Edit (continued)

LABEL	DESCRIPTION
ОК	Click OK to save the changes.
Cancel	Click Cancel to discard all changes and return to the main VPN screen.

Table 107 Configuration > VPN > IPSec VPN > VPN Connection	n > Ed	dit (continued)
--	--------	-----------------

20.2.2 The VPN Connection Add/Edit Manual Key Screen

The VPN Connection Add/Edit Manual Key screen allows you to create a new VPN connection or edit an existing one using a manual key. This is useful if you have problems with IKE key management. To access this screen, go to the VPN Connection summary screen (see Section 20.2 on page 276), click either the Add icon or an existing manual key entry's Edit icon and click Show Advanced Settings. In the VPN Gateway section of the screen, select Manual Key.

Note: Only use manual key as a temporary solution, because it is not as secure as a regular IPSec SA.

Figure 180 Configuration > VPN > IPSec VPN > VPN Connection > Add > Manual Key

Add VPN Connection			?
Hide Advanced Settings 🔠 Create ne	w Object 🔻		
General Settings			
Enable			
Connection Name:			
Enable NetBIOS broadcast over IPS	ec		
MSS Adjustment			
Custom Size	0	🕕 🕕 00 - 1500 Bytes)	
Auto			
/PN Gateway			
Application Scenario			
Site-to-site			
Site-to-site with Dynamic Peer			
Remote Access (Server Role)			
Remote Access (Client Role)			
Manual Key			
Manual Key			
My Address:			
Secure Gateway Address:			
SPI:	(256 - 4095)		
Encapsulation Mode:	Tunnel	~	
Active Protocol:	ESP	*	
Encryption Algorithm:	DES	~	
Authentication Algorithm:	SHA1	~	
Encryption Key:			

This table describes labels specific to manual key configuration. See Section 20.2 on page 276 for descriptions of the other fields.

LABEL	DESCRIPTION
Manual Key	
My Address	Type the IP address of the ZyWALL in the IPSec SA.
Secure Gateway Address	Type the IP address of the remote IPSec router in the IPSec SA.
SPI	Type a unique SPI (Security Parameter Index) between 256 and 4095. The SPI is used to identify the ZyWALL during authentication.
	The ZyWALL and remote IPSec router must use the same SPI.
Encapsulation Mode	Select which type of encapsulation the IPSec SA uses. Choices are
Wode	Tunnel - this mode encrypts the IP header information and the data
	Transport - this mode only encrypts the data. You should only select this if the IPSec SA is used for communication between the ZyWALL and remote IPSec router.
	If you select Transport mode, the ZyWALL automatically switches to Tunnel mode if the IPSec SA is not used for communication between the ZyWALL and remote IPSec router. In this case, the ZyWALL generates a log message for this change.
	The ZyWALL and remote IPSec router must use the same encapsulation.
Active Protocol	Select which protocol you want to use in the IPSec SA. Choices are:
	AH (RFC 2402) - provides integrity, authentication, sequence integrity (replay resistance), and non-repudiation but not encryption. If you select AH , you must select an Authentication Algorithm .
	ESP (RFC 2406) - provides encryption and the same services offered by AH , but its authentication is weaker. If you select ESP , you must select an Encryption Algorithm and Authentication Algorithm .
	The ZyWALL and remote IPSec router must use the same protocol.
Encryption Algorithm	This field is applicable when the Active Protocol is ESP . Select which key size and encryption algorithm to use in the IPSec SA. Choices are:
	NULL - no encryption key or algorithm
	DES - a 56-bit key with the DES encryption algorithm
	3DES - a 168-bit key with the DES encryption algorithm
	AES128 - a 128-bit key with the AES encryption algorithm
	AES192 - a 192-bit key with the AES encryption algorithm
	AES256 - a 256-bit key with the AES encryption algorithm
	The ZyWALL and the remote IPSec router must use the same algorithm and key. Longer keys require more processing power, resulting in increased latency and decreased throughput.
Authentication Algorithm	Select which hash algorithm to use to authenticate packet data in the IPSec SA. Choices are SHA1 , SHA256 , SHA512 and MD5 . SHA is generally considered stronger than MD5 , but it is also slower.
	The ZyWALL and remote IPSec router must use the same algorithm.

Table 108 Configuration > VPN > IPSec VPN > VPN Connection > Add > Manual Key

LABEL	DESCRIPTION
Encryption Key	This field is applicable when you select an Encryption Algorithm . Enter the encryption key, which depends on the encryption algorithm.
	DES - type a unique key 8-32 characters long
	3DES - type a unique key 24-32 characters long
	AES128 - type a unique key 16-32 characters long
	AES192 - type a unique key 24-32 characters long
	AES256 - type a unique key 32 characters long
	You can use any alphanumeric characters or ,; $ ^{=}!@#$ %^&*()_+\{}':./<>=-".
	If you want to enter the key in hexadecimal, type "0x" at the beginning of the key. For example, "0x0123456789ABCDEF" is in hexadecimal format; in "0123456789ABCDEF" is in ASCII format. If you use hexadecimal, you must enter twice as many characters as listed above.
	The remote IPSec router must have the same encryption key.
	The ZyWALL ignores any characters above the minimum number of characters required by the algorithm. For example, if you enter 1234567890XYZ for a DES encryption key, the ZyWALL only uses 12345678. The ZyWALL still stores the longer key.
Authentication	Enter the authentication key. The length depends on the authentication algorithm.
i toy	MD5 - type a unique key 16-20 characters long
	SHA1 - type a unique key 20 characters long
	SHA256 - type a unique key 32 characters long
	SHA512 - type a unique key 64 characters long
	You can use any alphanumeric characters or ,; $ ` -!@#$%^&*()_+{}':./<>=-". If you want to enter the key in hexadecimal, type "0x" at the beginning of the key. For example, "0x0123456789ABCDEF" is in hexadecimal format; in "0123456789ABCDEF" is in ASCII format. If you use hexadecimal, you must enter twice as many characters as listed above.$
	The remote IPSec router must have the same authentication key.
	The ZyWALL ignores any characters above the minimum number of characters required by the algorithm. For example, if you enter 12345678901234567890 for a MD5 authentication key, the ZyWALL only uses 1234567890123456. The ZyWALL still stores the longer key.
ОК	Click OK to save your settings and exit this screen.
Cancel	Click Cancel to exit this screen without saving.

Table 108 Configuration > VPN > IPSec VPN > VPN Connection > Add > Manual Key (continued)

20.3 The VPN Gateway Screen

The **VPN Gateway** summary screen displays the IPSec VPN gateway policies in the ZyWALL, as well as the ZyWALL's address, remote IPSec router's address, and associated VPN connections for each one. In addition, it also lets you activate and deactivate each VPN gateway. To access this screen, click **Configuration > VPN > Network > IPSec VPN > VPN Gateway**. The following screen appears.

0	Add 📝 Edit 🎁 Remove 💡	Activate 🖗 Inactiva	ate 📴 Object Referen	ce
#	Name	My Address	Secure Gateway	VPN Connection
1	Oefault_L2TP_VPN_GW	■ wan1	0.0.0.0, 0.0.0.0	Default_L2TP_VPN_Conne
2	@ test	s wan1	0.0.0.0, 0.0.0.0	
14		Show 50 🗸 item:	s	Displaying 1 - 2 of

Figure 181 Configuration > VPN > IPSec VPN > VPN Gateway

Each field is discussed in the following table. See Section 20.3.1 on page 286 for more information.

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Object References	Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific VPN gateway.
Status	The activate (light bulb) icon is lit when the entry is active and dimmed when the entry is inactive.
Name	This field displays the name of the VPN gateway
My address	This field displays the interface or a domain name the ZyWALL uses for the VPN gateway.
Secure Gateway	This field displays the IP address(es) of the remote IPSec routers.
VPN Connection	This field displays VPN connections that use this VPN gateway.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 109 Configuration > VPN > IPSec VPN > VPN Gateway

20.3.1 The VPN Gateway Add/Edit Screen

The **VPN Gateway Add/Edit** screen allows you to create a new VPN gateway policy or edit an existing one. To access this screen, go to the **VPN Gateway summary** screen (see Section 20.3 on page 285), and click either the **Add** icon or an **Edit** icon.

Figure 182	Configuration >	VPN >	IPSec VPN >	VPN Gateway	$\prime > Fdit$
I Iguie Ioz	conniguration /				

Hide Advance Settings				
General Settings				
🔲 Enable				
VPN Gateway Name:] 🕖	
Gateway Settings				
My Address				
Interface	wan1		DHCP client 0.0.0.0/0.0.0.0	
🔘 Domain Name / IP				
Peer Gateway Address	22			
Static Address	Primary	0.0.0.0		
	Secondary	y 0.0.0.0		
Fall Back to Primary Peer Gal	eway when po	ssible	(60-96400 seconds)	
			(or or the secondar)	
	default	******	(See Mv Certificates)	
Local ID Type:	IP		×	
Content:	0.0.0.0			
Peer ID Type:	Any		v	
Content:				
Phase 1 Settings				
SA Life Time:	86400		(180 - 3000000 Seconds)	
Negotiation Mode:	Main		×	
Proposal	📀 Add	🔡 Edit	TRemove .	
	# En	cryption 🔺	Authentication	
	1 DE	ES	MD5	
Kev Group:	DH1		*	
NAT Traversal				
Dead Peer Detection (DPD)				
stended Authentication				
Enable Extended Authentication				
Server Mode	default		~	
🔘 Client Mode				
User Name:				
Passwordt				

Each field is described in the following table.	
---	--

LABEL	DESCRIPTION						
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.						
General Settings							
VPN Gateway Name	Type the name used to identify this VPN gateway. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.						
Gateway Settings							
My Address	Select how the IP address of the ZyWALL in the IKE SA is defined.						
	If you select Interface , select the Ethernet interface, VLAN interface, virtual Ethernet interface, virtual VLAN interface or PPPoE/PPTP interface. The IP address of the ZyWALL in the IKE SA is the IP address of the interface.						
	If you select Domain Name / IP , enter the domain name or the IP address of the ZyWALL. The IP address of the ZyWALL in the IKE SA is the specified IP address or the IP address corresponding to the domain name. 0.0.0.0 is not generally recommended as it has the ZyWALL accept IPSec requests destined for any interface address on the ZyWALL.						
Peer Gateway	Select how the IP address of the remote IPSec router in the IKE SA is defined.						
Address	Select Static Address to enter the domain name or the IP address of the remote IPSec router. You can provide a second IP address or domain name for the ZyWALL to try if it cannot establish an IKE SA with the first one.						
	Fall back to Primary Peer Gateway when possible: When you select this, if the connection to the primary address goes down and the ZyWALL changes to using the secondary connection, the ZyWALL will reconnect to the primary address when it becomes available again and stop using the secondary connection. Users will lose their VPN connection briefly while the ZyWALL changes back to the primary connection. To use this, the peer device at the secondary address cannot be set to use a nailed-up VPN connection. In the Fallback Check Interval field, set how often to check if the primary address is available.						
	Select Dynamic Address if the remote IPSec router has a dynamic IP address (and does not use DDNS).						
Authentication	Note: The ZyWALL and remote IPSec router must use the same authentication method to establish the IKE SA.						
Pre-Shared Key	Select this to have the ZyWALL and remote IPSec router use a pre-shared key (password) to identify each other when they negotiate the IKE SA. Type the pre-shared key in the field to the right. The pre-shared key can be:						
	 alphanumeric characters or ,;. `~!@#\$%^&*()_+\{}':./<>=-" pairs of hexadecimal (0-9, A-F) characters, preceded by "0x". 						
	Type "0x" at the beginning of a hexadecimal key. For example, "0x0123456789ABCDEF" is in hexadecimal format; "0123456789ABCDEF" is in ASCII format. If you use hexadecimal, you must enter twice as many characters since you need to enter pairs.						
	The ZyWALL and remote IPSec router must use the same pre-shared key.						

Table 110	Configuration	> VPN	>	IPSec	VPN	>	VPN Gateway	>	Edit
-----------	---------------	-------	---	-------	-----	---	-------------	---	------
LABEL	DESCRIPTION								
---------------	---	--	--	--					
Certificate	Select this to have the ZyWALL and remote IPSec router use certificates to authenticate each other when they negotiate the IKE SA. Then select the certificate the ZyWALL uses to identify itself to the remote IPsec router.								
	This certificate is one of the certificates in My Certificates . If this certificate is self- signed, import it into the remote IPsec router. If this certificate is signed by a CA, the remote IPsec router must trust that CA.								
	Note: The IPSec routers must trust each other's certificates.								
	The ZyWALL uses one of its Trusted Certificates to authenticate the remote IPSec router's certificate. The trusted certificate can be a self-signed certificate or that of a trusted CA that signed the remote IPSec router's certificate.								
Local ID Type	This field is read-only if the ZyWALL and remote IPSec router use certificates to identify each other. Select which type of identification is used to identify the ZyWALL during authentication. Choices are:								
	IP - the ZyWALL is identified by an IP address								
	DNS - the ZyWALL is identified by a domain name								
	E-mail - the ZyWALL is identified by the string specified in this field								
Content	This field is read-only if the ZyWALL and remote IPSec router use certificates to identify each other. Type the identity of the ZyWALL during authentication. The identity depends on the Local ID Type .								
	IP - type an IP address; if you type 0.0.0.0, the ZyWALL uses the IP address specified in the My Address field. This is not recommended in the following situations:								
	 There is a NAT router between the ZyWALL and remote IPSec router. You want the remote IPSec router to be able to distinguish between IPSec SA 								
	requests that come from IPSec routers with dynamic walk IP addresses.								
	DNS - type the fully qualified domain name (EODN). This value is only used for								
	identification and can be any string that matches the peer ID string.								
	E-mail - the ZyWALL is identified by the string you specify here; you can use up to 63 ASCII characters including spaces, although trailing spaces are truncated. This value is only used for identification and can be any string.								
Peer ID Type	Select which type of identification is used to identify the remote IPSec router during authentication. Choices are:								
	IP - the remote IPSec router is identified by an IP address								
	DNS - the remote IPSec router is identified by a domain name								
	E-mail - the remote IPSec router is identified by the string specified in this field								
	Any - the ZyWALL does not check the identity of the remote IPSec router								
	If the ZyWALL and remote IPSec router use certificates, there is one more choice.								
	Subject Name - the remote IPSec router is identified by the subject name in the certificate								

Table 110Configuration > VPN > IPSec VPN > VPN Gateway > Edit (continued)

LABEL	DESCRIPTION				
Content	his field is disabled if the Peer ID Type is Any . Type the identity of the remote IPSec outer during authentication. The identity depends on the Peer ID Type .				
	If the ZyWALL and remote IPSec router do not use certificates,				
	IP - type an IP address; see the note at the end of this description.				
	DNS - type the fully qualified domain name (FQDN). This value is only used for identification and can be any string that matches the peer ID string.				
	E-mail - the remote IPSec router is identified by the string you specify here; you can use up to 31 ASCII characters including spaces, although trailing spaces are truncated. This value is only used for identification and can be any string.				
	If the ZyWALL and remote IPSec router use certificates, type the following fields from the certificate used by the remote IPSec router.				
	IP - subject alternative name field; see the note at the end of this description.				
	DNS - subject alternative name field				
	E-mail - subject alternative name field				
	Subject Name - subject name (maximum 255 ASCII characters, including spaces)				
	Note: If Peer ID Type is IP , please read the rest of this section.				
	If you type 0.0.0.0, the ZyWALL uses the IP address specified in the Secure Gateway Address field. This is not recommended in the following situations:				
	 There is a NAT router between the ZyWALL and remote IPSec router. You want the remote IPSec router to be able to distinguish between IPSec SA requests that come from IPSec routers with dynamic WAN IP addresses. 				
	In these situations, use a different IP address, or use a different Peer ID Type .				
Phase 1 Settings					
SA Life Time (Seconds)	Type the maximum number of seconds the IKE SA can last. When this time has passed, the ZyWALL and remote IPSec router have to update the encryption and authentication keys and re-negotiate the IKE SA. This does not affect any existing IPSec SAs, however.				
Negotiation Mode	Select the negotiation mode to use to negotiate the IKE SA. Choices are				
Mode	$\ensuremath{\textbf{Main}}$ - this encrypts the ZyWALL's and remote IPSec router's identities but takes more time to establish the IKE SA				
	Aggressive - this is faster but does not encrypt the identities				
	The ZyWALL and the remote IPSec router must use the same negotiation mode.				
Proposal Use this section to manage the encryption algorithm and authentication algorithm ZyWALL accepts from the remote IPSec router for negotiating the IKE S					
Add	Click this to create a new entry.				
Edit	Select an entry and click this to be able to modify it.				
Remove	Select an entry and click this to delete it.				
#	This field is a sequential value, and it is not associated with a specific proposal. The sequence of proposals should not affect performance significantly.				

Table 110 Configuration > VPN > IPSec VPN > VPN Gateway > Edit (continued)

Table 110	Configuration >	VPN >	IPSec VPN >	VPN Gateway	/ > Edit ((continued)
	e e i i i gen a ti e i i i					

LABEL	DESCRIPTION			
Encryption	Select which key size and encryption algorithm to use in the IKE SA. Choices are:			
	DES - a 56-bit key with the DES encryption algorithm			
	3DES - a 168-bit key with the DES encryption algorithm			
	AES128 - a 128-bit key with the AES encryption algorithm			
	AES192 - a 192-bit key with the AES encryption algorithm			
	AES256 - a 256-bit key with the AES encryption algorithm			
	The ZyWALL and the remote IPSec router must use the same key size and encryption algorithm. Longer keys require more processing power, resulting in increased latency and decreased throughput.			
Authentication	Select which hash algorithm to use to authenticate packet data in the IPSec SA. Choices are SHA1, SHA256, SHA512 and MD5. SHA is generally considered stronger than MD5, but it is also slower.			
	The remote IPSec router must use the same authentication algorithm.			
Key Group	Select which Diffie-Hellman key group (DH <i>x</i>) you want to use for encryption keys. Choices are:			
	DH1 - use a 768-bit random number			
	DH2 - use a 1024-bit random number			
	DH5 - use a 1536-bit random number			
	The longer the key, the more secure the encryption, but also the longer it takes to encrypt and decrypt information. Both routers must use the same DH key group.			
NAT Traversal	Select this if any of these conditions are satisfied.			
	This IKE SA might be used to negotiate IPSec SAs that use ESP as the active protocol.			
	 There are one or more NAT routers between the ZyWALL and remote IPSec router, and these routers do not support IPSec pass-thru or a similar feature. 			
	The remote IPSec router must also enable NAT traversal, and the NAT routers have to forward packets with UDP port 500 and UDP 4500 headers unchanged.			
Dead Peer Detection (DPD) Select this check box if you want the ZyWALL to make sure the remote IPSec there before it transmits data through the IKE SA. The remote IPSec router support DPD. If there has been no traffic for at least 15 seconds, the ZyWA message to the remote IPSec router. If the remote IPSec router responds, t transmits the data. If the remote IPSec router does not respond, the ZyWA down the IKE SA.				
	If the remote IPSec router does not support DPD, see if you can use the VPN connection connectivity check (see Section 20.2.1 on page 277).			
More Settings/Less Settings	Click this button to show or hide the Extended Authentication fields.			
Extended Authentication	When multiple IPSec routers use the same VPN tunnel to connect to a single VPN tunnel (telecommuters sharing a tunnel for example), use extended authentication to enforce a user name and password check. This way even though they all know the VPN tunnel's security settings, each still has to provide a unique user name and password.			
Enable Extended Authentication	Select this if one of the routers (the ZyWALL or the remote IPSec router) verifies a user name and password from the other router using the local user database and/or an external server.			
Server Mode	Select this if the ZyWALL authenticates the user name and password from the remote IPSec router. You also have to select the authentication method, which specifies how the ZyWALL authenticates this information.			

LABEL	DESCRIPTION
Client Mode	Select this radio button if the ZyWALL provides a username and password to the remote IPSec router for authentication. You also have to provide the User Name and the Password .
User Name	This field is required if the ZyWALL is in Client Mode for extended authentication. Type the user name the ZyWALL sends to the remote IPSec router. The user name can be 1-31 ASCII characters. It is case-sensitive, but spaces are not allowed.
Password	This field is required if the ZyWALL is in Client Mode for extended authentication. Type the password the ZyWALL sends to the remote IPSec router. The password can be 1-31 ASCII characters. It is case-sensitive, but spaces are not allowed.
ОК	Click OK to save your settings and exit this screen.
Cancel	Click Cancel to exit this screen without saving.

|--|

20.4 VPN Concentrator

A VPN concentrator combines several IPSec VPN connections into one secure network.

Figure 183 VPN Topologies (Fully Meshed and Hub and Spoke)



In a fully-meshed VPN topology (**1** in the figure), there is a VPN connection between every pair of routers. In a hub-and-spoke VPN topology (**2** in the figure), there is a VPN connection between each spoke router (**B**, **C**, **D**, and **E**) and the hub router (**A**), which uses the VPN concentrator. The VPN concentrator routes VPN traffic between the spoke routers and itself.

A VPN concentrator reduces the number of VPN connections that you have to set up and maintain in the network. You might also be able to consolidate the policy routes in each spoke router, depending on the IP addresses and subnets of each spoke.

However a VPN concentrator is not for every situation. The hub router is a single failure point, so a VPN concentrator is not as appropriate if the connection between spoke routers cannot be down occasionally (maintenance, for example). There is also more burden on the hub router. It receives VPN traffic from one spoke, decrypts it, inspects it to find out to which spoke to route it, encrypts it, and sends it to the appropriate spoke. Therefore, a VPN concentrator is more suitable when there is a minimum amount of traffic between spoke routers.

20.4.1 VPN Concentrator Requirements and Suggestions

Consider the following when using the VPN concentrator.

- The local IP addresses configured in the VPN rules should not overlap.
- The concentrator must have at least one separate VPN rule for each spoke. In the local policy, specify the IP addresses of the networks with which the spoke is to be able to have a VPN tunnel. This may require you to use more than one VPN rule for each spoke.
- To have all Internet access from the spoke routers go through the VPN tunnel, set the VPN rules in the spoke routers to use 0.0.0.0 (any) as the remote IP address.
- Your firewall rules can still block VPN packets.

20.4.2 VPN Concentrator Screen

The **VPN Concentrator** summary screen displays the VPN concentrators in the ZyWALL. To access this screen, click **Configuration > VPN > IPSec VPN > Concentrator**.

ire 184 Configuration > VPN > IPSec VPN > Concentrator

N Con	inection	VPN Gateway	Concentrator	Configuration Provisioning	
onfigi	uration				
0	Add 📝 E	dit 🎁 Remove			
#	# Name Group Members				
1	Concent	rator-test	Default_L2T	2_VPN_Connection	
14	4 Page	1 of 1 🕨	Show 50 🗸 i	tems Displaying 1 - 1 of	

Each field is discussed in the following table. See Section 20.4.3 on page 293 for more information.

LABEL	DESCRIPTION		
Add	Click this to create a new entry.		
Edit	Select an entry and click this to be able to modify it.		
Remove	Select an entry and click this to delete it.		
#	This field is a sequential value, and it is not associated with a specific concentrator.		
Name	This field displays the name of the VPN concentrator.		
Group Members	These are the VPN connection policies that are part of the VPN concentrator.		

Table 111 Configuration > VPN > IPSec VPN > Concentrator

20.4.3 The VPN Concentrator Add/Edit Screen

Use the **VPN Concentrator Add/Edit** screen to create or edit a VPN concentrator. To access this screen, go to the **VPN Concentrator summary** screen (see Section 20.4 on page 292), and click either the **Add** icon or an **Edit** icon.

Available Default_L2TP_VPN_Connection	Member	1
	•	▲

Figure 185 Configuration > VPN > IPSec VPN > Concentrator > Edit

Each field is described in the following table.

Table 112 VPN > IPSec VPN > Concentrator > Edit

LABEL	DESCRIPTION
Name	Enter the name of the concentrator. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Member	Select the concentrator's IPSec VPN connection policies.
	Note: You must disable policy enforcement in each member. See Section 20.2.1 on page 277.
	IPSec VPN connection policies that do not belong to a VPN concentrator appear under Available . Select any VPN connection policies that you want to add to the VPN concentrator and click the right arrow button to add them.
	The VPN concentrator's member VPN connections appear under Member . Select any VPN connections that you want to remove from the VPN concentrator, and click the left arrow button to remove them.
ОК	Click OK to save your changes in the ZyWALL.
Cancel	Click Cancel to exit this screen without saving.

20.5 ZyWALL IPSec VPN Client Configuration Provisioning

Use the **Configuration > VPN > IPSec VPN > Configuration Provisioning** screen to configure who can retrieve VPN rule settings from the ZyWALL using the ZyWALL IPSec VPN Client. In the ZyWALL IPSec VPN Client, you just need to enter the IP address of the ZyWALL to get all the VPN rule settings automatically. You do not need to manually configure all rule settings in the ZyWALL IPSec VPN client.

VPN rules for the ZyWALL IPSec VPN Client have certain restrictions. They must *not* contain the following settings:

- AH active protocol
- NULL encryption
- SHA512 authentication
- A subnet or range remote policy

In the ZyWALL **Quick Setup** wizard, you can use the **VPN Settings for Configuration Provisioning** wizard to create a VPN rule that will not violate these restrictions.

Figure 186 Configuration > VPN > IPSec VPN > Configuration Provisioning

General Settings	General Settings					
Enable Configure	☑ Enable Configuration Provision					
Authentication						
Client Authentication	n Method:	default	~			
Configuration						
🕜 Add 🗾 Edit 🎁	🖥 Remove <table-cell> Activate 🌚</table-cell>	Inactivate 🔐 Move				
Sta Priority 🔺	VPN Connection		Allowed User			
· 💡 1	WIZ_VPN_PROVISIONING		Finance			
😡 rule-2		······	any			
@ 2	WIZ_VPN		Leo			
4	of 1 🕨 🕅 Show 5	50 🗸 items		Displaying 1 - 2 of 2		
		Apply	Reset			

Each field is discussed in the following table.

LABEL	DESCRIPTION
Enable Configuration Provisioning	Select this for users to be able to retrieve VPN rule settings using the ZyWALL IPSec VPN client.
Client Authentication Method	Choose how users should be authenticated. They can be authenticated using the local database on the ZyWALL or an external authentication database such as LDAP, Active Directory or RADIUS. default is a method you configured in Object > Auth Method. You may configure multiple methods there. If you choose the local database on the ZyWALL, then configure users using the Object > User/Group screen. If you choose LDAP, Active Directory or RADIUS authentication servers, then configure users on the respective server.
Configuration	When you add or edit a configuration provisioning entry, you are allowed to set the VPN Connection and Allowed User fields.
	Duplicate entries are not allowed. You cannot select the same VPN Connection and Allowed User pair in a new entry if the same pair exists in a previous entry.
	You can bind different rules to the same user, but the ZyWALL will only allow VPN rule setting retrieval for the first match found.
Add	Click Add to bind a configured VPN rule to a user or group. Only that user or group may then retrieve the specified VPN rule settings.
	If you click Add without selecting an entry in advance then the new entry appears as the first entry. Entry order is important as the ZyWALL searches entries in the order listed here to find a match. After a match is found, the ZyWALL stops searching. If you want to add an entry as number three for example, then first select entry 2 and click Add . To reorder an entry, use Move .
Edit	Select an existing entry and click Edit to change its settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate . Make sure that Enable Configuration Provisioning is also selected.
Inactivate	To turn off an entry, select it and click Inactivate.

 Table 113
 Configuration > VPN > IPSec VPN > Configuration Provisioning

LABEL	DESCRIPTION
Move	Use Move to reorder a selected entry. Select an entry, click Move , type the number where the entry should be moved, press <enter>, then click Apply.</enter>
Status	This icon shows if the entry is active (yellow) or not (gray). VPN rule settings can only be retrieved when the entry is activated (and Enable Configuration Provisioning is also selected).
Priority	Priority shows the order of the entry in the list. Entry order is important as the ZyWALL searches entries in the order listed here to find a match. After a match is found the ZyWALL stops searching.
VPN Connection	This field shows all configured VPN rules that match the rule criteria for the ZyWALL IPSec VPN client. Select a rule to bind to the associated user or group.
Allowed User	Select which user or group of users is allowed to retrieve the associated VPN rule settings using the ZyWALL IPSec VPN client. A user may belong to a number of groups. If entries are configured for different groups, the ZyWALL will allow VPN rule setting retrieval based on the first match found.
-	users of type admin of fimited-admin are not allowed.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 113 Configuration > VPN > IPSec VPN > Configuration Provisioning (continued)

20.6 IPSec VPN Background Information

Here is some more detailed IPSec VPN background information.

IKE SA Overview

The IKE SA provides a secure connection between the ZyWALL and remote IPSec router.

It takes several steps to establish an IKE SA. The negotiation mode determines how many. There are two negotiation modes--main mode and aggressive mode. Main mode provides better security, while aggressive mode is faster.

Note: Both routers must use the same negotiation mode.

These modes are discussed in more detail in Negotiation Mode on page 299. Main mode is used in various examples in the rest of this section.

IP Addresses of the ZyWALL and Remote IPSec Router

To set up an IKE SA, you have to specify the IP addresses of the ZyWALL and remote IPSec router. You can usually enter a static IP address or a domain name for either or both IP addresses. Sometimes, your ZyWALL might offer another alternative, such as using the IP address of a port or interface, as well.

You can also specify the IP address of the remote IPSec router as 0.0.0.0. This means that the remote IPSec router can have any IP address. In this case, only the remote IPSec router can initiate an IKE SA because the ZyWALL does not know the IP address of the remote IPSec router. This is often used for telecommuters.

IKE SA Proposal

The IKE SA proposal is used to identify the encryption algorithm, authentication algorithm, and Diffie-Hellman (DH) key group that the ZyWALL and remote IPSec router use in the IKE SA. In main mode, this is done in steps 1 and 2, as illustrated next.





The ZyWALL sends one or more proposals to the remote IPSec router. (In some devices, you can only set up one proposal.) Each proposal consists of an encryption algorithm, authentication algorithm, and DH key group that the ZyWALL wants to use in the IKE SA. The remote IPSec router selects an acceptable proposal and sends the accepted proposal back to the ZyWALL. If the remote IPSec router rejects all of the proposals, the ZyWALL and remote IPSec router cannot establish an IKE SA.

Note: Both routers must use the same encryption algorithm, authentication algorithm, and DH key group.

In most ZyWALLs, you can select one of the following encryption algorithms for each proposal. The algorithms are listed in order from weakest to strongest.

- Data Encryption Standard (DES) is a widely used method of data encryption. It applies a 56-bit key to each 64-bit block of data.
- Triple DES (3DES) is a variant of DES. It iterates three times with three separate keys, effectively tripling the strength of DES.
- Advanced Encryption Standard (AES) is a newer method of data encryption that also uses a secret key. AES applies a 128-bit key to 128-bit blocks of data. It is faster than 3DES.

Some ZyWALLs also offer stronger forms of AES that apply 192-bit or 256-bit keys to 128-bit blocks of data.

In most ZyWALLs, you can select one of the following authentication algorithms for each proposal. The algorithms are listed in order from weakest to strongest.

- MD5 (Message Digest 5) produces a 128-bit digest to authenticate packet data.
- SHA1 (Secure Hash Algorithm) produces a 160-bit digest to authenticate packet data.
- SHA256 (Secure Hash Algorithm) produces a 256-bit digest to authenticate packet data.
- SHA512 (Secure Hash Algorithm) produces a 512-bit digest to authenticate packet data.

See Diffie-Hellman (DH) Key Exchange on page 298 for more information about DH key groups.

Diffie-Hellman (DH) Key Exchange

The ZyWALL and the remote IPSec router use DH public-key cryptography to establish a shared secret. The shared secret is then used to generate encryption keys for the IKE SA and IPSec SA. In main mode, this is done in steps 3 and 4, as illustrated next.

Figure 188 IKE SA: Main Negotiation Mode, Steps 3 - 4: DH Key Exchange

Diffie-Hellman key exchange



DH public-key cryptography is based on DH key groups. Each key group is a fixed number of bits long. The longer the key, the more secure the encryption, but also the longer it takes to encrypt and decrypt information. For example, DH2 keys (1024 bits) are more secure than DH1 keys (768 bits), but DH2 keys take longer to encrypt and decrypt.

Authentication

Before the ZyWALL and remote IPSec router establish an IKE SA, they have to verify each other's identity. This process is based on pre-shared keys and router identities.

In main mode, the ZyWALL and remote IPSec router authenticate each other in steps 5 and 6, as illustrated below. The identities are also encrypted using the encryption algorithm and encryption key the ZyWALL and remote IPSec router selected in previous steps.

Figure 189 IKE SA: Main Negotiation Mode, Steps 5 - 6: Authentication (continued)



You have to create (and distribute) a pre-shared key. The ZyWALL and remote IPSec router use it in the authentication process, though it is not actually transmitted or exchanged.

Note: The ZyWALL and the remote IPSec router must use the same pre-shared key.

Router identity consists of ID type and content. The ID type can be domain name, IP address, or email address, and the content is a (properly-formatted) domain name, IP address, or e-mail address. The content is only used for identification. Any domain name or e-mail address that you enter does not have to actually exist. Similarly, any domain name or IP address that you enter does not have to correspond to the ZyWALL's or remote IPSec router's properties.

The ZyWALL and the remote IPSec router have their own identities, so both of them must store two sets of information, one for themselves and one for the other router. Local ID type and content refers to the ID type and content that applies to the router itself, and peer ID type and content refers to the ID type and content that applies to the other router.

Note: The ZyWALL's local and peer ID type and content must match the remote IPSec router's peer and local ID type and content, respectively.

For example, in Table 114 on page 299, the ZyWALL and the remote IPSec router authenticate each other successfully. In contrast, in Table 115 on page 299, the ZyWALL and the remote IPSec router cannot authenticate each other and, therefore, cannot establish an IKE SA.

ZYWALL	REMOTE IPSEC ROUTER
Local ID type: E-mail	Local ID type: IP
Local ID content: tom@yourcompany.com	Local ID content: 1.1.1.2
Peer ID type: IP	Peer ID type: E-mail
Peer ID content: 1.1.1.2	Peer ID content: tom@yourcompany.com

ZYWALL	REMOTE IPSEC ROUTER
Local ID type: E-mail	Local ID type: IP
Local ID content: tom@yourcompany.com	Local ID content: 1.1.1.2
Peer ID type: IP	Peer ID type: E-mail
Peer ID content: 1.1.1.20	Peer ID content: tom@yourcompany.com

It is also possible to configure the ZyWALL to ignore the identity of the remote IPSec router. In this case, you usually set the peer ID type to **Any**. This is less secure, so you should only use this if your ZyWALL provides another way to check the identity of the remote IPSec router (for example, extended authentication) or if you are troubleshooting a VPN tunnel.

Additional Topics for IKE SA

This section provides more information about IKE SA.

Negotiation Mode

There are two negotiation modes--main mode and aggressive mode. Main mode provides better security, while aggressive mode is faster.

Main mode takes six steps to establish an IKE SA.

Steps 1 - 2: The ZyWALL sends its proposals to the remote IPSec router. The remote IPSec router selects an acceptable proposal and sends it back to the ZyWALL.

Steps 3 - 4: The ZyWALL and the remote IPSec router exchange pre-shared keys for authentication and participate in a Diffie-Hellman key exchange, based on the accepted DH key group, to establish a shared secret.

Steps 5 - 6: Finally, the ZyWALL and the remote IPSec router generate an encryption key (from the shared secret), encrypt their identities, and exchange their encrypted identity information for authentication.

In contrast, aggressive mode only takes three steps to establish an IKE SA. Aggressive mode does not provide as much security because the identity of the ZyWALL and the identity of the remote IPSec router are not encrypted. It is usually used in remote-access situations, where the address of the initiator is not known by the responder and both parties want to use pre-shared keys for authentication. For example, the remote IPSec router may be a telecommuter who does not have a static IP address.

VPN, NAT, and NAT Traversal

In the following example, there is another router (A) between router X and router Y.

Figure 190 VPN/NAT Example



If router **A** does NAT, it might change the IP addresses, port numbers, or both. If router **X** and router **Y** try to establish a VPN tunnel, the authentication fails because it depends on this information. The routers cannot establish a VPN tunnel.

Most routers like router **A** now have an IPSec pass-thru feature. This feature helps router **A** recognize VPN packets and route them appropriately. If router **A** has this feature, router **X** and router **Y** can establish a VPN tunnel as long as the active protocol is ESP. (See Active Protocol on page 301 for more information about active protocols.)

If router A does not have an IPSec pass-thru or if the active protocol is AH, you can solve this problem by enabling NAT traversal. In NAT traversal, router **X** and router **Y** add an extra header to the IKE SA and IPSec SA packets. If you configure router **A** to forward these packets unchanged, router **X** and router **Y** can establish a VPN tunnel.

You have to do the following things to set up NAT traversal.

- Enable NAT traversal on the ZyWALL and remote IPSec router.
- Configure the NAT router to forward packets with the extra header unchanged. (See the field description for detailed information about the extra header.)

The extra header may be UDP port 500 or UDP port 4500, depending on the standard(s) the ZyWALL and remote IPSec router support.

Extended Authentication

Extended authentication is often used when multiple IPSec routers use the same VPN tunnel to connect to a single IPSec router. For example, this might be used with telecommuters.

In extended authentication, one of the routers (the ZyWALL or the remote IPSec router) provides a user name and password to the other router, which uses a local user database and/or an external server to verify the user name and password. If the user name or password is wrong, the routers do not establish an IKE SA.

You can set up the ZyWALL to provide a user name and password to the remote IPSec router, or you can set up the ZyWALL to check a user name and password that is provided by the remote IPSec router.

If you use extended authentication, it takes four more steps to establish an IKE SA. These steps occur at the end, regardless of the negotiation mode (steps 7-10 in main mode, steps 4-7 in aggressive mode).

Certificates

It is possible for the ZyWALL and remote IPSec router to authenticate each other with certificates. In this case, you do not have to set up the pre-shared key, local identity, or remote identity because the certificates provide this information instead.

- Instead of using the pre-shared key, the ZyWALL and remote IPSec router check the signatures on each other's certificates. Unlike pre-shared keys, the signatures do not have to match.
- The local and peer ID type and content come from the certificates.

Note: You must set up the certificates for the ZyWALL and remote IPSec router first.

IPSec SA Overview

Once the ZyWALL and remote IPSec router have established the IKE SA, they can securely negotiate an IPSec SA through which to send data between computers on the networks.

Note: The IPSec SA stays connected even if the underlying IKE SA is not available anymore.

This section introduces the key components of an IPSec SA.

Local Network and Remote Network

In an IPSec SA, the local network, the one(s) connected to the ZyWALL, may be called the local policy. Similarly, the remote network, the one(s) connected to the remote IPSec router, may be called the remote policy.

Active Protocol

The active protocol controls the format of each packet. It also specifies how much of each packet is protected by the encryption and authentication algorithms. IPSec VPN includes two active protocols, AH (Authentication Header, RFC 2402) and ESP (Encapsulating Security Payload, RFC 2406).

Note: The ZyWALL and remote IPSec router must use the same active protocol.

Usually, you should select ESP. AH does not support encryption, and ESP is more suitable with NAT.

Encapsulation

There are two ways to encapsulate packets. Usually, you should use tunnel mode because it is more secure. Transport mode is only used when the IPSec SA is used for communication between the ZyWALL and remote IPSec router (for example, for remote management), not between computers on the local and remote networks.

Note: The ZyWALL and remote IPSec router must use the same encapsulation.

These modes are illustrated below.

Figure 191 VPN: Transport ar	nd Tunnel M	ode Encapsi	ulation		
Original Packet	IP Header	TCP Header	Data]	
Transport Mode Packet	IP Header	AH/ESP Header	TCP Header	Data]
Tunnel Mode Packet	IP Header	AH/ESP Header	IP Header	TCP Header	Data

In tunnel mode, the ZyWALL uses the active protocol to encapsulate the entire IP packet. As a result, there are two IP headers:

- Outside header: The outside IP header contains the IP address of the ZyWALL or remote IPSec router, whichever is the destination.
- Inside header: The inside IP header contains the IP address of the computer behind the ZyWALL or remote IPSec router. The header for the active protocol (AH or ESP) appears between the IP headers.

In transport mode, the encapsulation depends on the active protocol. With AH, the ZyWALL includes part of the original IP header when it encapsulates the packet. With ESP, however, the ZyWALL does not include the IP header when it encapsulates the packet, so it is not possible to verify the integrity of the source IP address.

IPSec SA Proposal and Perfect Forward Secrecy

An IPSec SA proposal is similar to an IKE SA proposal (see IKE SA Proposal on page 297), except that you also have the choice whether or not the ZyWALL and remote IPSec router perform a new DH key exchange every time an IPSec SA is established. This is called Perfect Forward Secrecy (PFS).

If you enable PFS, the ZyWALL and remote IPSec router perform a DH key exchange every time an IPSec SA is established, changing the root key from which encryption keys are generated. As a result, if one encryption key is compromised, other encryption keys remain secure.

If you do not enable PFS, the ZyWALL and remote IPSec router use the same root key that was generated when the IKE SA was established to generate encryption keys.

The DH key exchange is time-consuming and may be unnecessary for data that does not require such security.

Additional Topics for IPSec SA

This section provides more information about IPSec SA in your ZyWALL.

IPSec SA using Manual Keys

You might set up an IPSec SA using manual keys when you want to establish a VPN tunnel quickly, for example, for troubleshooting. You should only do this as a temporary solution, however, because it is not as secure as a regular IPSec SA.

In IPSec SAs using manual keys, the ZyWALL and remote IPSec router do not establish an IKE SA. They only establish an IPSec SA. As a result, an IPSec SA using manual keys has some characteristics of IKE SA and some characteristics of IPSec SA. There are also some differences between IPSec SA using manual keys and other types of SA.

IPSec SA Proposal using Manual Keys

In an IPSec SA using manual keys, you can only specify one encryption algorithm and one authentication algorithm. You cannot specify several proposals. There is no DH key exchange, so you have to provide the encryption key and the authentication key the ZyWALL and remote IPSec router use.

Note: The ZyWALL and remote IPSec router must use the same encryption key and authentication key.

Authentication and the Security Parameter Index (SPI)

For authentication, the ZyWALL and remote IPSec router use the SPI, instead of pre-shared keys, ID type and content. The SPI is an identification number.

Note: The ZyWALL and remote IPSec router must use the same SPI.

NAT for Inbound and Outbound Traffic

The ZyWALL can translate the following types of network addresses in IPSec SA.

- Source address in outbound packets this translation is necessary if you want the ZyWALL to route packets from computers outside the local network through the IPSec SA.
- Source address in inbound packets this translation hides the source address of computers in the remote network.
- Destination address in inbound packets this translation is used if you want to forward packets (for example, mail) from the remote network to a specific computer (like the mail server) in the local network.

Each kind of translation is explained below. The following example is used to help explain each one.



Figure 192 VPN Example: NAT for Inbound and Outbound Traffic

Source Address in Outbound Packets (Outbound Traffic, Source NAT)

This translation lets the ZyWALL route packets from computers that are not part of the specified local network (local policy) through the IPSec SA. For example, in Figure 192 on page 304, you have to configure this kind of translation if you want computer **M** to establish a connection with any computer in the remote network (**B**). If you do not configure it, the remote IPSec router may not route messages for computer **M** through the IPSec SA because computer **M**'s IP address is not part of its local policy.

To set up this NAT, you have to specify the following information:

- Source the original source address; most likely, computer **M**'s network.
- Destination the original destination address; the remote network (B).
- SNAT the translated source address; the local network (A).

Source Address in Inbound Packets (Inbound Traffic, Source NAT)

You can set up this translation if you want to change the source address of computers in the remote network. To set up this NAT, you have to specify the following information:

- Source the original source address; the remote network (B).
- Destination the original destination address; the local network (A).
- SNAT the translated source address; a different IP address (range of addresses) to hide the original source address.

Destination Address in Inbound Packets (Inbound Traffic, Destination NAT)

You can set up this translation if you want the ZyWALL to forward some packets from the remote network to a specific computer in the local network. For example, in Figure 192 on page 304, you can configure this kind of translation if you want to forward mail from the remote network to the mail server in the local network (A).

You have to specify one or more rules when you set up this kind of NAT. The ZyWALL checks these rules similar to the way it checks rules for a firewall. The first part of these rules define the conditions in which the rule apply.

- Original IP the original destination address; the remote network (B).
- Protocol the protocol [TCP, UDP, or both] used by the service requesting the connection.
- Original Port the original destination port or range of destination ports; in Figure 192 on page 304, it might be port 25 for SMTP.

The second part of these rules controls the translation when the condition is satisfied.

- Mapped IP the translated destination address; in Figure 192 on page 304, the IP address of the mail server in the local network (**A**).
- Mapped Port the translated destination port or range of destination ports.

The original port range and the mapped port range must be the same size.

IPSec VPN Example

Here is an example of configuring a site-to-site IPSec VPN.



ZyWALL **X** uses 1.2.3.4 as its public address, and remote IPSec router **Y** uses 2.2.2.2. Create the VPN tunnel between the ZyWALL's LAN subnet (192.168.1.0/24) and the LAN subnet behind the peer IPSec router (172.16.1.0/24).

Set Up the VPN Gateway that Manages the IKE SA

In Configuration > VPN > IPSec VPN > VPN Gateway > Add, enable the VPN gateway and name it (VPN_GW_EXAMPLE here). Set My Address to Interface and select a WAN interface. Set Peer Gateway Address to Static Address and enter the remote IPSec router's public IP address (2.2.2.2 here) as the Primary. Set Authentication to Pre-Shared Key and enter 12345678. Click OK.

Add VPN Gateway	a.200		?
Show Advanced Settings			
General Settings			
🔲 Enable			
VPN Gateway Name:	VPN_GW_EX	AMPLE	
Gateway Settings			
My Address			
Interface	wan1	 Static 1.2.3.4/255.255.0.0 	
🔘 Domain Name / IP			
Peer Gateway Address			
Static Address	Primary	2.2.2.2	
	Secondary	0.0.0.0	
📃 Fall back to Primary Peer Gal	teway when possibl		
Fall Back Check Interval:	300	(60-86400 seconds)	
Dynamic Address			
Authentication			
Pre-Shared Key	12345678		
Certificate		✓ (See <u>My Certificates</u>)	
Phase 1 Settings			
SA Life Time:	86400	(180 - 3000000 Seconds)	

Set Up the VPN Connection that Manages the IPSec SA

- 1 In Configuration > VPN > IPSec VPN > VPN Connection > Add, click Create New Object > Address to create an address object for the remote network. Set the Address Type to SUBNET, the Network field to 172.16.1.0, and the Netmask to 255.255.255.0.
- 2 Enable the VPN connection and name it ("VPN_CONN_EXAMPLE"). Set VPN Gateway to Site-tosite and select the VPN gateway you configured (VPN_GW_EXAMPLE). Set Local Policy to LAN1_SUBNET and Remote Policy to VPN_REMOTE_SUBNET for the remote. Click OK.

Show Advanced Settings 🔚 Create ne	ew Object		Create Address		
			Name:	VPN_REMOTE_SUE	BNET
			Address Type:	SUBNET	
			Network:	172, 16, 1,0	
Connection Name:	VPN_CONN_EXA	MPLE	Notroalu		
VPN Gateway			Netmask:	255.255.255.0	
Application Scenario					
Site-to-site					
Site-to-site with Dynamic Peer					
Remote Access (Server Role)				=	
Remote Access (Client Role)				=	
VPN Gateway:	VPN_GW_EXAMP	YLE 👻 wa	an1 2.2.2.2 0.0.0.0		
Policy					
Local policy:	LAN1_SUBNET	✓ IN	TERFACE SUBNET, 192.168.1	.0/24	
Remote policy:	VPN_REMOTE_S	UBNET 👻 SU	JBNET, 172.16.1.0/24		
Phase 2 Setting					
SA Life Time:	86400	(180 - 300	00000 Seconds)		
Related Settings					
Zone:	IPSec_VPN	¥ 🚺	1		
Connectivity Check					
Enable Connectivity Check 👔					
et a lange t		~		-	

SSL VPN

21.1 Overview

Use SSL VPN to allow users to use a web browser for secure remote user login. The remote users do not need a VPN router or VPN client software.

21.1.1 What You Can Do in this Chapter

- Use the VPN > SSL VPN > Access Privilege screens (see Section 21.2 on page 309) to configure SSL access policies.
- Use the Click VPN > SSL VPN > Global Setting screen (see Section 21.3 on page 313) to set the IP address of the ZyWALL (or a gateway device) on your network for full tunnel mode access, enter access messages or upload a custom logo to be displayed on the remote user screen.

21.1.2 What You Need to Know

Full Tunnel Mode

In full tunnel mode, a virtual connection is created for remote users with private IP addresses in the same subnet as the local network. This allows them to access network resources in the same way as if they were part of the internal network.

Figure 194 Network Access Mode: Full Tunnel Mode



SSL Access Policy

An SSL access policy allows the ZyWALL to perform the following tasks:

- limit user access to specific applications or file sharing server on the network.
- allow user access to specific networks.
- assign private IP addresses and provide DNS/WINS server information to remote users to access internal networks.

SSL Access Policy Objects

The SSL access policies reference the following objects. If you update this information, in response to changes, the ZyWALL automatically propagates the changes through the SSL policies that use the object(s). When you delete an SSL policy, the objects are not removed.

OBJECT TYPE	OBJECT SCREEN	DESCRIPTION
User Accounts	User Account/ User Group	Configure a user account or user group to which you want to apply this SSL access policy.
Application	SSL Application	Configure an SSL application object to specify the type of application and the address of the local computer, server, or web site SSL users are to be able to access.
IP Pool	Address	Configure an address object that defines a range of private IP addresses to assign to user computers so they can access the internal network through a VPN connection.
Server Addresses	Address	Configure address objects for the IP addresses of the DNS and WINS servers that the ZyWALL sends to the VPN connection users.
VPN Network	Address	Configure an address object to specify which network segment users are allowed to access through a VPN connection.

Table 116 Objects

You cannot delete an object that is referenced by an SSL access policy. To delete the object, you must first unassociate the object from the SSL access policy.

Finding Out More

- See Section 21.4 on page 315 for an SSL VPN example.
- See Chapter 35 on page 422 for details on SSL application objects.

21.2 The SSL Access Privilege Screen

Click **VPN > SSL VPN** to open the **Access Privilege** screen. This screen lists the configured SSL access policies.

cess l	Privilege	Global Setting		
cess	Policy Su	immary		
A	dd 📝 E	dit 🎁 Remove 💡 Activa	ate 💡 Inactivate 📣 Move ा Objec	t References
# 🔺 :	Status	Name	User/Group	Access Policy Summary
1	8	New	radius-users	■New
14	Page	1 of 1 > > Show	v 50 🗸 items	Displaying 1 - 1 of
4	Page	1 of 1 🕨 🕅 Show	s 50 v items	Displaying 1 - 1 (

Figure 195 VPN > SSL VPN > Access Privilege

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Move	To move an entry to a different number in the list, click the Move icon. In the field that appears, specify the number to which you want to move the interface.
Object References	Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field displays the index number of the entry.
Status	This icon is lit when the entry is active and dimmed when the entry is inactive.
Name	This field displays the descriptive name of the SSL access policy for identification purposes.
User/Group	This field displays the user account or user group name(s) associated to an SSL access policy.
	This field displays up to three names.
Access Policy Summary	This field displays details about the SSL application object this policy uses including its name, type, and address.
Apply	Click Apply to save the settings.
Reset	Click Reset to discard all changes.

 Table 117
 VPN > SSL VPN > Access Privilege

21.2.1 The SSL Access Policy Add/Edit Screen

To create a new or edit an existing SSL access policy, click the **Add** or **Edit** icon in the **Access Privilege** screen.

Create new Object -		
Configuration		
Enable Policy		
Name:	New	
Zone:	SSL VPN	
Description:		(Ontional)
	new create	(optional)
User/Group		
Selectable User/Gr	oup Objects	Selected User/Group Objects
=== Obje	ect ===	
admin	_	
Idap-users		
radius-users	E	
ad-users		
	•	Sected Appleation Objects
etwork Extension (Op	tional) ension : to enter SSL VPN tunnel	
etwork Extension (Op Enable Network Exte Force all client traffic Assign IP Pool:	tional) ension t to enter SSL VPN tunnel Please select one	
etwork Extension (Op Enable Network Exte Force all client traffic Assign IP Pool: DNS Server 1:	tional) ension t to enter SSL VPN tunnel Please select one none	×
etwork Extension (Op Enable Network Extern Force all client traffic Assign IP Pool: DNS Server 1: DNS Server 2:	tional) ension t to enter SSL VPN tunnel Please select one none none	v v
etwork Extension (Op Enable Network Exte Force all client traffic Assign IP Pool: DNS Server 1: DNS Server 2: WINS Server 1:	tional) ension to enter SSL VPN tunnel Please select one none none none	× × × ×
etwork Extension (Op Enable Network Exte Force all client traffic Assign IP Pool: DNS Server 1: DNS Server 2: WINS Server 1: WINS Server 2:	tional) ension t to enter SSL VPN tunnel Please select one none none none none	

Figure 196 VPN > SSL VPN > Add/Edit

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Create new Object	Use to configure any new settings objects that you need to use in this screen.
Configuration	
Enable Policy	Select this option to activate this SSL access policy.

LABEL	DESCRIPTION
Name	Enter a descriptive name to identify this policy. You can enter up to 31 characters ("a-z", A-Z", "0-9") with no spaces allowed.
Zone	Select the zone to which to add this SSL access policy. You use zones to apply security settings such as firewall and remote management.
Description	Enter additional information about this SSL access policy. You can enter up to 60 characters ("0-9", "a-z", "A-Z", "-" and "_").
User/Group	The Selectable User/Group Objects list displays the name(s) of the user account and/ or user group(s) to which you have not applied an SSL access policy yet.
	To associate a user or user group to this SSL access policy, select a user account or user group and click the right arrow button to add to the Selected User/Group Objects list. You can select more than one name.
	To remove a user or user group, select the name(s) in the Selected User/Group Objects list and click the left arrow button.
	Note: Although you can select admin and limited-admin accounts in this screen, they are reserved for device configuration only. You cannot use them to access the SSL VPN portal.
SSL Application List (Optional)	The Selectable Application Objects list displays the name(s) of the SSL application(s) you can select for this SSL access policy.
	To associate an SSL application to this SSL access policy, select a name and click the right arrow button to add to the Selected Application Objects list. You can select more than one application.
	To remove an SSL application, select the name(s) in the Selected Application Objects list and click the left arrow button.
	Note: To allow access to shared files on a Windows 7 computer, within Windows 7 you must enable sharing on the folder and also go to the Network and Sharing Center 's Advanced sharing settings and turn on the current network profile's file and printer sharing.
Network Extension	(Optional)
Enable Network Extension	Select this option to create a VPN tunnel between the authenticated users and the internal network. This allows the users to access the resources on the network as if they were on the same local network. This includes access to resources not supported by SSL application objects. For example this lets users Telnet to the internal network even though the ZyWALL does not have SSL application objects for Telnet.
	Clear this option to disable this feature. Users can only access the applications as defined by the VPN tunnel's selected SSL application settings and the remote user computers are not made to be a part of the local network.
Force all client traffic to SSL VPN tunnel	Select this to send all traffic from the SSL VPN clients through the SSL VPN tunnel. This replaces the default gateway of the SSL VPN clients with the SSL VPN gateway.
Assign IP Pool	Define a separate pool of IP addresses to assign to the SSL users. Select it here.
	The SSL VPN IP pool cannot overlap with IP addresses on the ZyWALL's local networks (LAN and DMZ for example), the SSL user's network, or the networks you specify in the SSL VPN Network List .
DNS/WINS Server 12	Select the name of the DNS or WINS server whose information the ZyWALL sends to the remote users. This allows them to access devices on the local network using domain names instead of IP addresses.

Table 118 VPN > SSL VPN > Access Privilege > Add/Edit (continued)

LABEL	DESCRIPTION
Network List	To allow user access to local network(s), select a network name in the Selectable Address Objects list and click the right arrow button to add to the Selected Address Objects list. You can select more than one network. To block access to a network, select the network name in the Selected Address Objects list and click the network name in the Selected Address Objects
	list and click the left arrow button.
ОК	Click OK to save the changes and return to the main Access Privilege screen.
Cancel	Click Cancel to discard all changes and return to the main Access Privilege screen.

Table 118 VPN > SSL VPN > Access Privilege > Add/Edit (continued)

21.3 The SSL Global Setting Screen

Click **VPN** > **SSL VPN** and click the **Global Setting** tab to display the following screen. Use this screen to set the IP address of the ZyWALL (or a gateway device) on your network for full tunnel mode access, enter access messages or upload a custom logo to be displayed on the remote user screen.

Figure 197 VPN > SSL VPN > Global Set	ting
---------------------------------------	------

Access Privilege	Global Setting		
Network Exter	esion Local IP:	192 168 200 1	
55L VPN Login	Domain Name	192.100.200.1	
SSL VPN Login	Domain Name 1:		(Optional)
SSL VPN Login	Domain Name 2:	1	(Optional)
Message			
Login Message	eti	Welcome to SSL VPN	
Logout Messa	ge:	Goodbye to SSL VPN	
Update Client '	Virtual Desktop Lo	go g/ing), browse to the location of the file and	then click Unload.
File Path::	Select a file path	Browse	
ZyX	<u> </u>		≀eset Logo to default

The following table describes the labels in this screen.

Table 119	VPN >	> SSL	VPN >	Global	Setting
-----------	-------	-------	-------	--------	---------

LABEL	DESCRIPTION
Global Setting	
Network Extension Local IP	Specify the IP address of the ZyWALL (or a gateway device) for full tunnel mode SSL VPN access. Leave this field to the default settings unless it conflicts with another interface.
SSL VPN Login Dor	nain Name

LABEL	DESCRIPTION
SSL VPN Login Domain Name 1/ 2	Specify a full domain name for users to use for SSL VPN login. The domain name must be registered to one of the ZyWALL's IP addresses or be one of the ZyWALL's DDNS entries. You can specify up to two domain names so you could use one domain name for each of two WAN ports. For example, www.zyxel.com is a fully qualified domain name where "www" is the host.
	The ZyWALL displays the normal login screen without the button for logging into the Web Configurator.
Message	
Login Message	Specify a message to display on the screen when a user logs in and an SSL VPN connection is established successfully. You can enter up to 60 characters (0-9, a-z, A-Z, '()+,/:=?;!*#@ $_{-}^{-}$) with spaces allowed.
Logout Message	Specify a message to display on the screen when a user logs out and the SSL VPN connection is terminated successfully. You can enter up to 60 characters (0-9, a-z, A-Z, $'()+,/:=?;!*#@\$_\%-"$) with spaces allowed.
Update Client Virtual Desktop	You can upload a graphic logo to be displayed on the web browser on the remote user computer. The ZyXEL company logo is the default logo.
Logo	Specify the location and file name of the logo graphic or click Browse to locate it.
	Note: The logo graphic must be GIF, JPG, or PNG format. The graphic should use a resolution of 103 x 29 pixels to avoid distortion when displayed. The ZyWALL automatically resizes a graphic of a different resolution to 103 x 29 pixels. The file size must be 100 kilobytes or less. Transparent background is recommended.
Browse	Click Browse to locate the graphic file on your computer.
Upload	Click Upload to transfer the specified graphic file from your computer to the ZyWALL.
Reset Logo to Default	Click Reset Logo to Default to display the ZyXEL company logo on the remote user's web browser.
Apply	Click Apply to save the changes and/or start the logo file upload process.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 119
 VPN > SSL VPN > Global Setting (continued)

21.3.1 How to Upload a Custom Logo

Follow the steps below to upload a custom logo to display on the remote user SSL VPN screens.

- 1 Click **VPN** > **SSL VPN** and click the **Global Setting** tab to display the configuration screen.
- 2 Click **Browse** to locate the logo graphic. Make sure the file is in GIF, JPG, or PNG format.
- 3 Click **Apply** to start the file transfer process.
- 4 Log in as a user to verify that the new logo displays properly.

The following shows an example logo on the remote user screen.

X		Application File Sharing	Welcome SSL-Users I <u>Logout</u>	(* Add to Favorite (? Help English
	Application			
	# Туре	Name		
	1 🚯 Web Server	New		
	Page 1 or 1	V VI Show <u>50</u> vitems		Uispiaying 1 - 101 1

Figure 198 Example Logo Graphic Display

21.4 SSL VPN Example

This example uses SSL VPN to let remote users securely access the internal http://info website.

- 1 Click Configuration > VPN > SSL VPN > Access Privilege > Add and click Create New Object > Application to create an SSL application object. Set the Type to Web Application, the Server Type to Web Server, and the URL to http://info. Select Web Page Encryption to prevent users from saving the web content.
- 2 Enable the policy. Enter a descriptive name in the **Name** field ("SSL-Example" here). Select the users to which to give access (the Sales user group here). Select the SSL application object you created ("WebExample" here). Click **OK**.

				Object		
Configuration				Type:	Web Application	~
Enable Policy						
Name:	SSL-Example			Web Application		
Zone:	SSL_VPN		▼ 1	Server Type:	Web Server	~
Description:	New Create		(Optional)	Name:	WebExample	
				URL:	http://info	
User/Group				Entry Point:		(Optional
Selectable User/G	roup Objects		Selected User/Group Objects	Web Page Encryp	tion	
=== Obj	ect ===		=== 0bject ===			
admin			Sales			
Idap-users						
radius-users						
ad-users						
55L Application List (Optional)					
55L Application List (Selectable Applica	Optional) tion Objects		Selected Application Objects		_	
55L Application List (Selectable Applica	Optional) tion Objects		Selected Application Objects		-	
55L Application List (– Selectable Applica	Optional) tion Objects		Selected Application Objects		-	
55L Application List (– Selectable Applica	Optional) tion Objects	•	Selected Application Objects		-	
55L Application List (– Selectable Applica	Optional) tion Objects	•	Selected Application Objects		-	
55L Application List (- Selectable Applica	Optional) tion Objects	+	Selected Application Objects		-	
55L Application List (⊤ Selectable Applica	Optional) tion Objects	+	Selected Application Objects		-	
55L Application List (⊤ Selectable Applica	Optional) tion Objects	+	Selected Application Objects		-	
55L Application List (- Selectable Applica	Optional) tion Objects	•	Selected Application Objects			

3 Display the ZyWALL's login screen, enter your user account information (the user name and password), and click **SSL VPN** to establish an SSL VPN connection.

	(Optional)
e characters and no sp	oaces)
Login	
	le characters and no sp

4 Your computer starts establishing a secure connection to the ZyWALL after the login. This may take up to two minutes. If you get a message about needing Java, download and install it and restart your browser and re-login. If a certificate warning screen displays, click **OK**, **Yes** or **Continue**.

Welcome	×
Java Runtime Environment(JRE) 1.4 or higher must be installed and enabled in order for web application objects and the network extension feature to work.	
Please download the latest JRE at site: <u>http://www.java.com/download</u>	
You can use all of the SSL VPN features after installing the JRE and restarting your browser.	
NOTE: Without JRE, file sharing still works.	
OK	

5 The client portal screen displays after the connection is up. In this example, click the **Web Server** link to go to http://info.

		pplication	File Sharing	Welcon	ne Fred i <u>Loqout</u>	(Add to Favorite	<u>(5 н</u> е
						English	
ppli	cation						
ppli	cation						
ppli	cation						
ppli	cation Type	Name					
ppli # 1	Cation Type	Name WebE>	cample				

If the user account is not included in an SSL VPN access policy, the ZyWALL redirects the user to the user aware screen.

For more information on user portal screens, refer to Chapter 22 on page 318.

SSL User Screens

22.1 Overview

This chapter introduces the remote user SSL VPN screens. The following figure shows a network example where a remote user (A) logs into the ZyWALL from the Internet to access the web server (**WWW**) on the local network.

Figure 199 Network Example



22.1.1 What You Need to Know

The ZyWALL can use SSL VPN to provide secure connections to network resources such as applications, files, intranet sites or e-mail through a web-based interface and using Microsoft Outlook Web Access (OWA).

Network Resource Access Methods

As a remote user, you can access resources on the local network using one of the following methods.

· Using a supported web browser

Once you have successfully logged in through the ZyWALL, you can access intranet sites, webbased applications, or web-based e-mails using one of the supported web browsers.

• Using the ZyWALL SecuExtender client

Once you have successfully logged into the ZyWALL, if the SSL VPN access policy has network extension enabled the ZyWALL automatically loads the ZyWALL SecuExtender client program to your computer. With the ZyWALL SecuExtender, you can access network resources, remote desktops and manage files as if you were on the local network. See Chapter 23 on page 331 for more on the ZyWALL SecuExtender.

System Requirements

Here are the browser and computer system requirements for remote user access.

- Windows 7 (32 or 64-bit), Vista (32 or 64-bit), 2003 (32-bit), XP (32-bit), or 2000 (32-bit)
- Internet Explorer 7 and above or Firefox 1.5 and above

- Using RDP requires Internet Explorer
- Sun's Runtime Environment (JRE) version 1.6 or later installed and enabled.

Required Information

A remote user needs the following information from the network administrator to log in and access network resources.

- the domain name or IP address of the ZyWALL
- · the login account user name and password
- · if also required, the user name and/or password to access the network resource

Certificates

The remote user's computer establishes an HTTPS connection to the ZyWALL to access the login screen. If instructed by your network administrator, you must install or import a certificate (provided by the ZyWALL or your network administrator).

Finding Out More

See Chapter 21 on page 308 for how to configure SSL VPN on the ZyWALL.

22.2 Remote SSL User Login

This section shows you how to access and log into the network through the ZyWALL. Example screens for Internet Explorer are shown.

1 Open a web browser and enter the web site address or IP address of the ZyWALL. For example, "http://sslvpn.mycompany.com".

Figure 200 Enter the Address in a Web Browser



2 Click **OK** or **Yes** if a security screen displays.

rigure	9 ZU	I Login Security Screen
Security	Aler	t
ß	Infor char secu	mation you exchange with this site cannot be viewed or nged by others. However, there is a problem with the site's urity certificate.
		The security certificate was issued by a company you have not chosen to trust. View the certificate to determine whether you want to trust the certifying authority.
	0	The security certificate date is valid.
	⚠	The name on the security certificate is invalid or does not match the name of the site
	Doy	iou want to proceed?
	[Yes No View Certificate

Figure 201 Login Security Screen

3 A login screen displays. Enter the user name and password of your login account. If a token password is also required, enter it in the **One-Time Password** field. Click **SSL VPN** to log in and establish an SSL VPN connection to the network to access network resources.

Enter User Name/Password and User Name: Password:	l click to login.
User Name: Password:	
Password:	
One-Time Password:	(Optional)
(max. 63 alphanumeric, printable cha	acters and no spaces)

4 Your computer starts establishing a secure connection to the ZyWALL after a successful login. This may take up to two minutes. If you get a message about needing Java, download and install it and restart your browser and re-login. If a certificate warning screen displays, click **OK**, **Yes** or **Continue**.

Figure 203	Java Need	ed Message
------------	-----------	------------

Welcome	×
Java Runtime Environment(JRE) 1.4 or higher must be installed and enabled in order for web application objects and the network extension feature to work.	
Please download the latest JRE at site: <u>http://www.java.com/download</u>	
You can use all of the SSL VPN features after installing the JRE and restarting your browser.	
NOTE: Without JRE, file sharing still works.	
ОК	

5 The ZyWALL tries to install the SecuExtender client. As shown next, you may have to click some pop-ups to get your browser to allow the installation.

This website wants to install the following ad click here	on: 'ZyWALL SecuExtender ActiveX' from 'ZyXEL Communications Corporation'. If you trust the website and the add-on and want to install it,
ZyXEL	Welcome SSL-Users Lonout (* Add to Favorite (? Help Application File Sharing
	Application
	Welcome to SSL VPN You might need to click the yellow warning bar at the top of the browser window to show the installation. If you can not install ActiveX object, please contact with administrator. Note: Click the information bar and choose Install ActiveX Control to allow installation to proceed, or it will try to run with JRE. OK
	(C) Copyright 2011 by ZyXEL Communications Corp.

Figure 204 ActiveX Object Installation Blocked by Browser

Figure 205 SecuExtender Blocked by Internet Explorer

Internet	t Explorer - Security Warning	
Do you	u want to install this software?	
	Name: ZyWALL SecuExtender ActiveX	
	Publisher: ZyXEL Communications Corporation	
💙 Mo	ore options	istall
٧	While files from the Internet can be useful, this file type can potentially your computer. Only install software from publishers you trust. <u>What's I</u>	harm the risk?

6 The ZyWALL tries to run the "ssltun" application. You may need to click something to get your browser to allow this. In Internet Explorer, click **Run**.

Figure 206 SecuExtender Progress

Warning	- Security	X
The ap Do you	plication's digital signature has been verifie 1 want to run the application?	d. 🎸
Name: Publis	: ssltun her: ZyXEL Communications Corporation	
From:	https://172.23.31.19	
	1	Run Cancel
Û	The digital signature has been validated by a trusted source.	More Information

7 Click **Next** to use the setup wizard to install the SecuExtender client on your computer.

Figure 201 Secuex	lender Frogress
🔀 ZyWALL SecuExtender Se	etup 🔀
ZyXEL	Welcome to the ZyWALL SecuExtender Setup Wizard
	The Setup Wizard will install ZyWALL SecuExtender on your computer. Click "Next" to continue or "Cancel" to exit the Setup Wizard.
	< Back Next > Cancel

Figure 207 SecuExtender Progress

8 If a screen like the following displays, click **Continue Anyway** to finish installing the SecuExtender client on your computer.



9 The **Application** screen displays showing the list of resources available to you. See Figure 209 on page 323 for a screen example.

Note: Available resource links vary depending on the configuration your network administrator made.

22.3 The SSL VPN User Screens

This section describes the main elements in the remote user screens.

jure 209 Remo		
ZyXEL	Application File Sharing	Welcome SSL-Users I <u>Longout</u> (# <u>Add to Favorite</u> (7 <u>Help</u> English
	Application # Type Name 1 @ Web Server New 1 @ Page 1 of 1 >> Show 50 vitems	E Displaying 1 - 1 of 1
	6	
		(C) Convict 2011 by 2-VEL Computer aligns Con-

The following table describes the various parts of a remote user screen.

Table 120 Remote User Screen Overview

#	DESCRIPTION
1	Click on a menu tab to go to the Application or File Sharing screen.
2	Click this icon to log out and terminate the secure connection.
3	Click this icon to create a bookmark to the SSL VPN user screen in your web browser.
4	Click this icon to display the on-line help window.
5	Select your preferred language for the interface.
6	This part of the screen displays a list of the resources available to you.
	In the Application screen, click on a link to access or display the access method.
	In the File Sharing screen, click on a link to open a file or directory.

22.4 Bookmarking the ZyWALL

You can create a bookmark of the ZyWALL by clicking the **Add to Favorite** icon. This allows you to access the ZyWALL using the bookmark without having to enter the address every time.

- 1 In any remote user screen, click the **Add to Favorite** icon.
- 2 A screen displays. Accept the default name in the **Name** field or enter a descriptive name to identify this link.

3 Click **OK** to create a bookmark in your web browser.



22.5 Logging Out of the SSL VPN User Screens

To properly terminate a connection, click on the **Logout** icon in any remote user screen.

- 1 Click the **Logout** icon in any remote user screen.
- 2 A prompt window displays. Click **OK** to continue.



22.6 SSL User Application Screen

Use the **Application** tab's screen to access web-based applications (such as web sites and e-mail) on the network through the SSL VPN connection. Which applications you can access depends on the ZyWALL's configuration.

The **Name** field displays the descriptive name for an application. The **Type** field displays wether the application is a web site (**Web Server**) or web-based e-mail using Microsoft Outlook Web Access (**OWA**).

To access a web-based application, simply click a link in the **Application** screen to display the web screen in a separate browser window.
Figure 212 Application

ZyXEL		Application	File Sharing	Welcome SSL-Users <u>Loqout</u>	(† <u>Add to Favorite</u> English	(? <u>Help</u>
0	Application					
	# Type 1 3 Web Server	Name				
	If I Page 1 of 1	▶ ▶ <mark> Sh</mark> ow	50 v items		Displaying :	1-1of1

22.7 SSL User File Sharing

The **File Sharing** screen lets you access files on a file server through the SSL VPN connection. Use it to display and access shared files/folders on a file server.

You can also perform the following actions:

- · Access a folder.
- Open a file (if your web browser cannot open the file, you are prompted to download it).
- Save a file to your computer.
- Create a new folder.
- Rename a file or folder.
- Delete a file or folder.
- Upload a file.

Note: Available actions you can perform in the **File Sharing** screen vary depending on the rights granted to you on the file server.

22.7.1 The Main File Sharing Screen

The first **File Sharing** screen displays the name(s) of the shared folder(s) available. The following figure shows an example with one file share.

Figure 213 File Sharing



22.7.2 Opening a File or Folder

You can open a file if the file extension is recognized by the web browser and the associated application is installed on your computer.

- 1 Log in as a remote user and click the **File Sharing** tab.
- 2 Click on a file share icon.
- **3** If an access user name and password are required, a screen displays as shown in the following figure. Enter the account information and click **Login** to continue.

Figure 214 File Sharing: Enter Access User Name and Password

	Application	File Sharing		eut (* Add to Favorite English	(? Help
File Sharing					
	© File	Sharing Enter Us User Ni Passw (max. 31 alphan	er Name/Password and click to login. ame: word: Permission denied umeric, printable characters and no spaces) Login Cancel	×	

4 A list of files/folders displays. Double click a file to open it in a separate browser window or select a file and click **Download** to save it to your computer. You can also click a folder to access it.

For this example, click on a .doc file to open the Word document.

		English
	File Sharing	
	New Folder II Rename @ Unload T Remove Download & Un	
	# Type File Name Size	Modified Date
ñ	1 C Projects	2011-08-30 04:02:36
	2 Pictures	2010-04-01 00:59:35
0.0	3 🦳 Music	2010-04-01 00:59:35
	4 🦲 Templates	2010-04-01 00:59:35
	5 😋 Public	2010-04-01 00:59:35
	oo-derivatives.doc 159744	2008-04-10 15:50:16
	7 📄 spnp_map.jpg 368237	2011-07-13 02:54:03
	8 jre-6u26-windows-x64.exe 16920864	2011-07-11 09:56:19
	9 📄 install_sahi_v35_20110525.jar 4183870	2011-07-11 09:41:52
	4	Displaying 1 - 9 of 9
_	W I 및 フ・ G I ▼ oo-derivatives.doc [唯讀] [相] 福高 常用 描入 版面配置 参考資料 多件 収録 『 『 』 》 Bitstream Vera Sans ◆ 28 ◆ 中控 A 300 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	宮模式) - Microsoft Word 見 接視 「売」「算 律 ▲ ▲ ▲
	B <i>I</i> <u>U</u> • abe x, x' →	■ III (王·································
	導覽 ▼ × 按尋文件	Derivatives of

Figure 215 File Sharing: Open a Word File

22.7.3 Downloading a File

You are prompted to download a file which cannot be opened using a web browser.

Follow the on-screen instructions to download and save the file to your computer. Then launch the associated application to open the file.

22.7.4 Saving a File

After you have opened a file in a web browser, you can save a copy of the file by clicking **File > Save As** and following the on-screen instructions.

ormat Tools Table Go To Favorites Help
🕨 🕄 Search 👔 Favorites 🛞 Media 🎯 🎼 🖏 - 🎒 🗹 📄
cifs/172.23.37.110/Tmp/Share/PrintingProcess.doc 🗾 🔗 Go Links
1 • 1 • 1 • 2 • 1 • 3 • 1 • 4 • 1 • 5 • 1 • 6 • 1 • 7 • 1 • 8 • 1 • 9 • 1 • 10 • 1 • 11 • 1 •
>
Printing Process
cess Overview For Boxes 1. Digital output and proofing

Figure 216 File Sharing: Save a Word File

22.7.5 Creating a New Folder

To create a new folder in the file share location, click the **New Folder** icon.

Specify a descriptive name for the folder. You can enter up to 356 characters. Then click Add.

Note: Make sure the length of the folder name does not exceed the maximum allowed on the file server.

Figure 217 File Sharing: Create a New Folder

New Folder		? X
Folder Name:		
	OK Cancel	

22.7.6 Renaming a File or Folder

To rename a file or folder, select a file or folder and click the **Rename** icon.

ZyXEL			Application	File Sharing	Welcome ssi i <u>Loqout</u> (# <u>Add to F</u> English	avorite (? <u>Help</u>
	File S	haring				
	Projec	<u>ts</u>	\sim			
	1 60	New Folder	😰 Rename) 🗞 Upload 🍵	Remove 🔡 Download 🗲 Up		
	#	Туре 🔺	File Name	Size	Modified Date	
	1	<u> </u>	Projects		2011-08-30 04:02:36	
	2	<u> </u>	Pictures		2010-04-01 00:59:35	
	3	6	Music	an die ein die	2010-04-01 00:59:35	Alle Mar Mar Mar Mar Mar Mar Mar 201
	4	6	Templates		2010-04-01 00:59:35	
	5	6	Public		2010-04-01 00:59:35	
	6		oo-derivatives.doc	159744	2008-04-10 15:50:16	
	7		spnp_map.jpg	368237	2011-07-13 02:54:03	
	8		jre-6u26-windows-x64.exe	16920864	2011-07-11 09:56:19	
	9		install_sahi_v35_20110525.jar	4183870	2011-07-11 09:41:52	
	14	4 Page	1 of 1 > > Show 50) 💙 items	Di	splaying 1 - 9 of 9

Figure 218 File Sharing: Rename

A popup window displays. Specify the new name and/or file extension in the field provided. You can enter up to 356 characters. Then click **Apply**.

Note: Make sure the length of the name does not exceed the maximum allowed on the file server.

You may not be able to open a file if you change the file extension.

Figure 219 File Sharing: Rename

I Rename		? ×
New Name:	Pictures	
iplates.	OK	Cancel

22.7.7 Deleting a File or Folder

Click the **Delete** icon next to a file or folder to remove it.

22.7.8 Uploading a File

Follow the steps below to upload a file to the file server.

- 1 Log into the remote user screen and click the File Sharing tab.
- 2 Click **Upload** and specify the location and/or name of the file you want to upload. Or click **Browse** to locate it.
- 3 Click **OK** to send the file to the file server.
- 4 After the file is uploaded successfully, you should see the name of the file and a message in the screen.

Figure 220 File Sharing: File Upload

ZyXEL	Application File Sharing	Welcome ssl i <u>Loqout</u> (# <u>Add to Favorite</u> (? <u>Help</u> English
	File Sharing	
0	Projects	
	🇞 New Folder 🗉 Rename 👒 Upload 🍵 Remove 💾 Download 🗧 Up	
	# Type File Name Size	Modified Date
	1 Pr 🚱 Upload ? 🗙	2011-08-30 04:02:36
	2 Pi File Name: Select a file path Browse	2010-04-01 00:59:35
	3 C M	2010-04-01 00:59:35
		2010-04-01 00:59:35
	5 P	2010-04-01 00:59:35
	6 poo-derivatives.doc 159744	2008-04-10 15:50:16
	7 📄 spnp_map.jpg 368237	2011-07-13 02:54:03
	8 📄 jre-6u26-windows-x64.exe 16920864	2011-07-11 09:56:19
	9 📄 install_sahi_v35_20110525.jar 4183870	2011-07-11 09:41:52
	4 4 Page 1 of 1 ▶ ▶ Show 50 ▼ items	Displaying 1 - 9 of 9
		(C) Copyright 2011 by ZyXEL Communications Corp.

Note: Uploading a file with the same name and file extension replaces the existing file on the file server. No warning message is displayed.

ZyWALL SecuExtender

The ZyWALL automatically loads the ZyWALL SecuExtender client program to your computer after a successful login to an SSL VPN tunnel with network extension support enabled. The ZyWALL SecuExtender lets you:

- · Access servers, remote desktops and manage files as if you were on the local network.
- Use applications like e-mail, file transfer, and remote desktop programs directly without using a browser. For example, you can use Outlook for e-mail instead of the ZyWALL's web-based e-mail.
- Use applications, even proprietary applications, for which the ZyWALL does not offer SSL application objects.

The applications must be installed on your computer. For example, to use the VNC remote desktop program, you must have the VNC client installed on your computer.

23.1 The ZyWALL SecuExtender Icon

The ZyWALL SecuExtender icon color indicates the SSL VPN tunnel's connection status.

Figure 221 ZyWALL SecuExtender Icon



- Green: the SSL VPN tunnel is connected. You can connect to the SSL application and network resources. You can also use another application to access resources behind the ZyWALL.
- Gray: the SSL VPN tunnel's connection is suspended. This means the SSL VPN tunnel is connected, but the ZyWALL SecuExtender will not send any traffic through it until you right-click the icon and resume the connection.
- Red: the SSL VPN tunnel is not connected. You cannot connect to the SSL application and network resources.

23.2 Status

Right-click the ZyWALL SecuExtender icon in the system tray and select **Status** to open the **Status** screen. Use this screen to view the ZyWALL SecuExtender's connection status and activity statistics.

7 7 11 11 1 1 1	1	Y
ZyWALL SecuExten	der Status	
C		
SecuExtender IP Ad	draee:	10 59 0 1
DNS Server 1:	ur033.	0.0.0.0
DNS Server 2:		0.0.0.0
WINC Server 1:		0.0.0.0
WING Server 1:		0.0.0.0
Wilvs Server 2:	102.10	0.0.0.0
Network 1:	192.16	8.2.0/255.255.255.0
Network 2:		Not configured
Network 3:		Not configured
Network 4:		Not configured
Network 5:		Not configured
Network 6:		Not configured
Network 7:		Not configured
Network 8:		Not configured
Activity		
Connection Time	00:02:48	
Transmitted	18,186 Bytes	247 Packets
Received	124 Bytes	2 Packets
	_	

Figure 222	ZyWALL	SecuExtender	Status
------------	--------	--------------	--------

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Connection Status	
SecuExtender IP Address	This is the IP address the ZyWALL assigned to this remote user computer for an SSL VPN connection.
DNS Server 1/2	These are the IP addresses of the DNS server and backup DNS server for the SSL VPN connection.
	DNS (Domain Name System) maps a domain name to its corresponding IP address and vice versa. The DNS server is extremely important because without it, you must know the IP address of a computer before you can access it. Your computer uses the DNS server specified here to resolve domain names for resources you access through the SSL VPN connection.
WINS Server 1/2	These are the IP addresses of the WINS (Windows Internet Naming Service) and backup WINS servers for the SSL VPN connection. The WINS server keeps a mapping table of the computer names on your network and the IP addresses that they are currently using.
Network 1~8	These are the networks (including netmask) that you can access through the SSL VPN connection.
Activity	
Connected Time	This is how long the computer has been connected to the SSL VPN tunnel.
Transmitted	This is how many bytes and packets the computer has sent through the SSL VPN connection.
Received	This is how many bytes and packets the computer has received through the SSL VPN connection.

Table 121 ZyWALL SecuExtender Status

23.3 View Log

If you have problems with the ZyWALL SecuExtender, customer support may request you to provide information from the log. Right-click the ZyWALL SecuExtender icon in the system tray and select **Log** to open a notepad file of the ZyWALL SecuExtender's log.



```
******
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DETAIL] Build Datetime: Feb 24 2009/
10:25:07
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DEBUG]
                                                rasphone.pbk: C:\Documents and
Settings\11746\rasphone.pbk
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DEBUG]
                                                  SecuExtender.log:
C:\Documents and Settings\11746\SecuExtender.log
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DETAIL] Check Parameters
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DETAIL] Connect to 172.23.31.19:443/
10444
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DETAIL] Parameter is OK
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DETAIL] Checking System status...
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DETAIL] Checking service (first) ...
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DETAIL] SecuExtender Helper is running
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DETAIL] System is OK
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DEBUG]
                                                  Connect to 2887196435/443
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DETAIL] Handshake LoopCounter: 0
[ 2009/03/12 13:35:50 ] [SecuExtender Agent] [DETAIL] 611 bytes of handshake data
received
```

23.4 Suspend and Resume the Connection

When the ZyWALL SecuExtender icon in the system tray is green, you can right-click the icon and select **Suspend Connection** to keep the SSL VPN tunnel connected but not send any traffic through it until you right-click the icon and resume the connection.

23.5 Stop the Connection

Right-click the icon and select Stop Connection to disconnect the SSL VPN tunnel.

23.6 Uninstalling the ZyWALL SecuExtender

Do the following if you need to remove the ZyWALL SecuExtender.

- 1 Click start > All Programs > ZyXEL > ZyWALL SecuExtender > Uninstall ZyWALL SecuExtender.
- 2 In the confirmation screen, click Yes.

Figure 224 Uninstalling the ZyWALL SecuExtender Confirmation



3 Windows uninstalls the ZyWALL SecuExtender.

Figure 225 ZyWALL SecuExtender Uninstallation

ZyWALL SecuExtender
Please wait while Windows configures ZyWALL SecuExtender
Gathering required information
Cancel

L2TP VPN

24.1 Overview

L2TP VPN uses the L2TP and IPSec client software included in remote users' Android, iOS, or Windows operating systems for secure connections to the network behind the ZyWALL. The remote users do not need their own IPSec gateways or third-party VPN client software.





24.1.1 What You Can Do in this Chapter

• Use the L2TP VPN screen (see Section 24.2 on page 337) to configure the ZyWALL's L2TP VPN settings.

24.1.2 What You Need to Know

The Layer 2 Tunneling Protocol (L2TP) works at layer 2 (the data link layer) to tunnel network traffic between two peers over another network (like the Internet). In L2TP VPN, an IPSec VPN tunnel is established first and then an L2TP tunnel is built inside it. See Chapter 20 on page 272 for information on IPSec VPN.

IPSec Configuration Required for L2TP VPN

You must configure an IPSec VPN connection for L2TP VPN to use (see Chapter 20 on page 272 for details). The IPSec VPN connection must:

- Be enabled.
- Use transport mode.
- Not be a manual key VPN connection.
- Use Pre-Shared Key authentication.
- Use a VPN gateway with the **Secure Gateway** set to **0.0.0.0** if you need to allow L2TP VPN clients to connect from more than one IP address.

Using the Default L2TP VPN Connection

The **Default_L2TP_VPN_GW** gateway entry is pre-configured to be convenient to use for L2TP VPN. Edit it as follows:

- Set My Address to the WAN interface domain name or IP address you want to use.
- Replace the default Pre-Shared Key.

Create a host-type address object containing the **My Address** IP address configured in the **Default_L2TP_VPN_GW** and set the **Default_L2TP_VPN_Connection**'s **Local Policy** to use it.

Policy Route

Configure a policy route to let remote users access resources on a network behind the ZyWALL.

- Set the policy route's **Source Address** to the address object that you want to allow the remote users to access (**LAN_SUBNET** in the following figure).
- Set the **Destination Address** to the IP address pool that the ZyWALL assigns to the remote users (L2TP_POOL in the following figure).
- Set the next hop to be the VPN tunnel that you are using for L2TP.
- Figure 227 Policy Route for L2TP VPN



To manage the ZyWALL through the L2TP VPN tunnel, create a routing policy that sends the ZyWALL's return traffic back through the L2TP VPN tunnel.

- Set Incoming to ZyWALL.
- Set **Destination Address** to the L2TP address pool.
- Set the next hop to be the VPN tunnel that you are using for L2TP.

If some of the traffic from the L2TP clients needs to go to the Internet, create a policy route to send traffic from the L2TP tunnels out through a WAN trunk.

- Set Incoming to Tunnel and select your L2TP VPN connection.
- Set the Source Address to the L2TP address pool.
- Set the Next-Hop Type to Trunk and select the appropriate WAN trunk.

24.2 L2TP VPN Screen

Click Configuration > VPN > L2TP VPN to open the following screen. Use this screen to configure the ZyWALL's L2TP VPN settings.

Note: Disconnect any existing L2TP VPN sessions before modifying L2TP VPN settings. The remote users must make any needed matching configuration changes and reestablish the sessions using the new settings.

de Advanced Settings 🛛 🔠 Create nev	v Object 🔻		
eneral Settings			
Enable L2TP Over IPSec			
VPN Connection:	Please select one	······	0
IP Address Pool:	Please select one	~	0
Authentication Method:	default	*	
Authentication Server Certificate:	default	*	
Allowed User:	any	*	
Keep Alive Timer:	60 (1-180 secon	ds)	
First DNS Server (Optional):	Custom Defined	*	
Second DNS Server (Optional):	Custom Defined	*	
First WINS Server (Optional):			016
Second WINS Server (Optional):	1		

The following table describes the fields in this screen.

Table 122Configuration > VPN > L2TP VPN

LABEL	DESCRIPTION
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
Create new Object	Use to configure any new settings objects that you need to use in this screen.
Enable L2TP Over IPSec	Use this field to turn the ZyWALL's L2TP VPN function on or off.
VPN Connection	 Select the IPSec VPN connection the ZyWALL uses for L2TP VPN. All of the configured VPN connections display here, but the one you use must meet the requirements listed in IPSec Configuration Required for L2TP VPN on page 335. Note: Modifying this VPN connection (or the VPN gateway that it uses) disconnects any existing L2TP VPN sessions.
IP Address Pool	Select the pool of IP addresses that the ZyWALL uses to assign to the L2TP VPN clients. Use Create new Object if you need to configure a new pool of IP addresses.
Authentication Method	Select how the ZyWALL authenticates a remote user before allowing access to the L2TP VPN tunnel. The authentication method has the ZyWALL check a user's user name and password against the ZyWALL's local database, a remote LDAP, RADIUS, a Active Directory server, or more than one of these. See Chapter 32 on page 399 for how to create authentication method objects.

LABEL	DESCRIPTION
Authentication Server Certificate	Select the certificate to use to identify the ZyWALL for L2TP VPN connections. You must have certificates already configured in the My Certificates screen (Click My Certificates and see Chapter 33 on page 403 for details). The certificate is used with the EAP, PEAP, and MSCHAPv2 authentication protocols.
Allowed User	The remote user must log into the ZyWALL to use the L2TP VPN tunnel. Select a user or user group that can use the L2TP VPN tunnel. Use Create new Object if you need to configure a new user account. Otherwise, select any to allow any user with a valid account and password on the ZyWALL to log in.
Keep Alive Timer	The ZyWALL sends a Hello message after waiting this long without receiving any traffic from the remote user. The ZyWALL disconnects the VPN tunnel if the remote user does not respond.
First DNS Server, Second DNS Server	Specify the IP addresses of DNS servers to assign to the remote users. You can specify these IP addresses two ways. Custom Defined - enter a static IP address. From ISP - use the IP address of a DNS server that another interface received from its DHCP server.
First WINS Server, Second WINS Server	The WINS (Windows Internet Naming Service) server keeps a mapping table of the computer names on your network and the IP addresses that they are currently using. Type the IP addresses of up to two WINS servers to assign to the remote users. You can specify these IP addresses two ways.
Apply	Click Apply to save your changes in the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 122Configuration > VPN > L2TP VPN (continued)

Bandwidth Management

25.1 Overview

Bandwidth management provides a convenient way to manage the use of various services on the network. It manages general protocols (for example, HTTP and FTP) and applies traffic prioritization to enhance the performance of delay-sensitive applications like voice and video.

25.1.1 What You Can Do in this Chapter

Use the **BWM** screens (see Section 25.2 on page 343) to control bandwidth for services passing through the ZyWALL, and it identifies the conditions that refine this.

25.1.2 What You Need to Know

When you allow a service, you can restrict the bandwidth it uses. It controls TCP and UDP traffic. Use policy routes to manage other types of traffic (like ICMP).

Note: Bandwidth management in policy routes has priority over policy routes to manage the bandwidth of TCP and UDP traffic.

If you want to use a service, make sure both the firewall allow the service's packets to go through the ZyWALL.

Note: The ZyWALL checks firewall rules before it checks bandwidth management rules for traffic going through the ZyWALL.

Bandwidth management examines every TCP and UDP connection passing through the ZyWALL. Then, you can specify, by port, whether or not the ZyWALL continues to route the connection.

DiffServ and DSCP Marking

QoS is used to prioritize source-to-destination traffic flows. All packets in the same flow are given the same priority. CoS (class of service) is a way of managing traffic in a network by grouping similar types of traffic together and treating each type as a class. You can use CoS to give different priorities to different packet types.

DiffServ (Differentiated Services) is a class of service (CoS) model that marks packets so that they receive specific per-hop treatment at DiffServ-compliant network devices along the route based on the application types and traffic flow. Packets are marked with DiffServ Code Points (DSCPs) indicating the level of service desired. This allows the intermediary DiffServ-compliant network devices to handle the packets differently depending on the code points without the need to negotiate paths or remember state information for every flow. In addition, applications do not have to request a particular service or give advanced notice of where the traffic is going.

Connection and Packet Directions

Bandwidth management looks at the connection direction, that is from which interface the connection was initiated and to which interface the connection is going.

A connection has outbound and inbound packet flows. The ZyWALL controls the bandwidth of traffic of each flow as it is going out through an interface or VPN tunnel.

- The outbound traffic flows from the connection initiator to the connection responder.
- The inbound traffic flows from the connection responder to the connection initiator.

For example, a LAN1 to WAN connection is initiated from LAN1 and goes to the WAN.

- Outbound traffic goes from a LAN1 device to a WAN device. Bandwidth management is applied before sending the packets out a WAN interface on the ZyWALL.
- Inbound traffic comes back from the WAN device to the LAN1 device. Bandwidth management is applied before sending the traffic out a LAN1 interface.



Figure 229 LAN1 to WAN Connection and Packet Directions

Outbound and Inbound Bandwidth Limits

You can limit an application's outbound or inbound bandwidth. This limit keeps the traffic from using up too much of the out-going interface's bandwidth. This way you can make sure there is bandwidth for other applications. When you apply a bandwidth limit to outbound or inbound traffic, each member of the out-going zone can send up to the limit. Take a LAN1 to WAN policy for example.

- Outbound traffic is limited to 200 kbps. The connection initiator is on the LAN1 so outbound means the traffic traveling from the LAN1 to the WAN. Each of the WAN zone's two interfaces can send the limit of 200 kbps of traffic.
- Inbound traffic is limited to 500 kbs. The connection initiator is on the LAN1 so inbound means the traffic traveling from the WAN to the LAN1.



Figure 230 LAN1 to WAN, Outbound 200 kbps, Inbound 500 kbps

Bandwidth Management Priority

- The ZyWALL gives bandwidth to higher-priority traffic first, until it reaches its configured bandwidth rate.
- Then lower-priority traffic gets bandwidth.
- The ZyWALL uses a fairness-based (round-robin) scheduler to divide bandwidth among traffic flows with the same priority.
- The ZyWALL automatically treats traffic with bandwidth management disabled as priority 7 (the lowest priority).

Maximize Bandwidth Usage

Maximize bandwidth usage allows applications with maximize bandwidth usage enabled to "borrow" any unused bandwidth on the out-going interface.

After each application gets its configured bandwidth rate, the ZyWALL uses the fairness- based scheduler to divide any unused bandwidth on the out-going interface amongst applications that need more bandwidth and have maximize bandwidth usage enabled.

Unused bandwidth is divided equally. Higher priority traffic does not get a larger portion of the unused bandwidth.

Bandwidth Management Behavior

The following sections show how bandwidth management behaves with various settings. For example, you configure DMZ to WAN policies for FTP servers **A** and **B**. Each server tries to send 1000 kbps, but the WAN is set to a maximum outgoing speed of 1000 kbps. You configure policy A for server **A**'s traffic and policy B for server **B**'s traffic.





Configured Rate Effect

In the following table the configured rates total less than the available bandwidth and maximize bandwidth usage is disabled, both servers get their configured rate.

POLICY	CONFIGURED RATE	MAX. B. U.	PRIORITY	ACTUAL RATE
А	300 kbps	No	1	300 kbps
В	200 kbps	No	1	200 kbps

Priority Effect

Here the configured rates total more than the available bandwidth. Because server **A** has higher priority, it gets up to it's configured rate (800 kbps), leaving only 200 kbps for server **B**.

Table 124 Priority Effect

POLICY	CONFIGURED RATE	MAX. B. U.	PRIORITY	ACTUAL RATE
А	800 kbps	Yes	1	800 kbps
В	1000 kbps	Yes	2	200 kbps

Maximize Bandwidth Usage Effect

With maximize bandwidth usage enabled, after each server gets its configured rate, the rest of the available bandwidth is divided equally between the two. So server **A** gets its configured rate of 300 kbps and server **B** gets its configured rate of 200 kbps. Then the ZyWALL divides the remaining bandwidth (1000 - 500 = 500) equally between the two (500 / 2 = 250 kbps for each). The priority has no effect on how much of the unused bandwidth each server gets.

So server **A** gets its configured rate of 300 kbps plus 250 kbps for a total of 550 kbps. Server **B** gets its configured rate of 200 kbps plus 250 kbps for a total of 450 kbps.

POLICY	POLICY CONFIGURED RATE		PRIORITY	ACTUAL RATE
А	300 kbps	Yes	1	550 kbps
В	200 kbps	Yes	2	450 kbps

 Table 125
 Maximize Bandwidth Usage Effect

Priority and Over Allotment of Bandwidth Effect

Server **A** has a configured rate that equals the total amount of available bandwidth and a higher priority. You should regard extreme over allotment of traffic with different priorities (as shown here) as a configuration error. Even though the ZyWALL still attempts to let all traffic get through and not be lost, regardless of its priority, server **B** gets almost no bandwidth with this configuration.

POLICY	CONFIGURED RATE	MAX. B. U.	PRIORITY	ACTUAL RATE
А	1000 kbps	Yes	1	999 kbps
В	1000 kbps	Yes	2	1 kbps

Table 126 Priority and Over Allotment of Bandwidth Effect

Finding Out More

• See DSCP Marking and Per-Hop Behavior on page 187 for a description of DSCP marking.

25.2 The Bandwidth Management Screen

The Bandwidth management screens control the bandwidth allocation for TCP and UDP traffic. You can use source interface, destination interface, destination port, schedule, user, source, destination information, DSCP code and service type as criteria to create a sequence of specific conditions, similar to the sequence of rules used by firewalls, to specify how the ZyWALL handles the DSCP value and allocate bandwidth for the matching packets.

Click **Configuration > BWM** to open the following screen. This screen allows you to enable/disable bandwidth management and add, edit, and remove user-defined bandwidth management policies.

The default bandwidth management policy is the one with the priority of "default". It is the last policy the ZyWALL checks if traffic does not match any other bandwidth management policies you have configured. You cannot remove, activate, deactivate or move the default bandwidth management policy.

	g										
Enable BWN	1										
figuration											
🔵 Add 📝 Ed	lit 🎁 Ren	nove <table-cell> 🖓 A</table-cell>	ictivate 🖗 Ii	nactivate 🔊	Move						
Stat Prio I	Descripti	User	Schedule	Incoming In	Outgoing In	Source	Destinati	DSCP	Service	BWM In/Pri/O	DSCP Mark
def		any	none	any	any	any	any	any	Obj:any	no/7/no/7	preserve/p
	101 1 Mg	• 1. h. h	Show 50) vitems						Di	splaying 1 - 1 of 1
	107 T 104	• 1. h. h	Show 50	1 vitems	100.00		a deciencia	2003		Di	splaying 1 - 1 of

Figure 232 Configuration > Bandwidth Management

The following table describes the labels in this screen. See Section 25.2.1 on page 345 for more information as well.

LABEL	DESCRIPTION
Enable BWM	Select this check box to activate management bandwidth.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Select an entry and click this to be able to modify it.
Remove	Select an entry and click this to delete it.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
Move	To change an entry's position in the numbered list, select it and click Move to display a field to type a number for where you want to put that entry and press [ENTER] to move the entry to the number that you typed.
Status	The activate (light bulb) icon is lit when the entry is active and dimmed when the entry is inactive. The status icon is not available for the default bandwidth management policy.
Priority	This field displays a sequential value for each bandwidth management policy and it is not associated with a specific setting.
	This field displays default for the default bandwidth management policy.
Description	This field displays additional information about this policy.
User	This is the user name or user group to which the policy applies. If any displays, the policy applies to all users.
Schedule	This is the schedule that defines when the policy applies. none means the policy always applies.
Incoming Interface	This is the source interface of the traffic to which this policy applies.
Outgoing Interface	This is the destination interface of the traffic to which this policy applies.
Source	This is the source address or address group for whom this policy applies. If any displays, the policy is effective for every source.
Destination	This is the destination address or address group for whom this policy applies. If any displays, the policy is effective for every destination.
DSCP Code	These are the DSCP code point values of incoming and outgoing packets to which this policy applies. The lower the number the higher the priority with the exception of 0 which is usually given only best-effort treatment.
	any means all DSCP value or no DSCP marker.
	default means traffic with a DSCP value of 0. This is usually best effort traffic
	The " af " options stand for Assured Forwarding. The number following the " af " identifies one of four classes and one of three drop preferences. See Section 10.4 on page 206 for more details.
Service Type	App and the service name displays if you selected App Patrol Service for the service type. An App Patrol Service is a pre-defined service.
	Obj and the service name displays if you selected Service Object for the service type. A Service Object is a customized pre-defined service or another service. Mouse over the service object name to view the corresponding IP protocol number.

 Table 127
 Configuration > Bandwidth Management

LABEL	DESCRIPTION
BWM In/Pri/Out/Pri	This field shows the amount of bandwidth the traffic can use.
	In - This is how much inbound bandwidth, in kilobits per second, this policy allows the matching traffic to use. Inbound refers to the traffic the ZyWALL sends to a connection's initiator. If no displays here, this policy does not apply bandwidth management for the inbound traffic.
	Out - This is how much outgoing bandwidth, in kilobits per second, this policy allows the matching traffic to use. Outbound refers to the traffic the ZyWALL sends out from a connection's initiator. If no displays here, this policy does not apply bandwidth management for the outbound traffic.
	Pri - This is the priority for the incoming (the first Pri value) or outgoing (the second Pri value) traffic that matches this policy. The smaller the number, the higher the priority. Traffic with a higher priority is given bandwidth before traffic with a lower priority. The ZyWALL ignores this number if the incoming and outgoing limits are both set to 0. In this case the traffic is automatically treated as being set to the lowest priority (7) regardless of this field's configuration.
DSCP Marking	This is how the ZyWALL handles the DSCP value of the incoming and outgoing packets that match this policy.
	In - Inbound, the traffic the ZyWALL sends to a connection's initiator.
	Out - Outbound, the traffic the ZyWALL sends out from a connection's initiator.
	If this field displays a DSCP value, the ZyWALL applies that DSCP value to the route's outgoing packets.
	preserve means the ZyWALL does not modify the DSCP value of the route's outgoing packets.
	default means the ZyWALL sets the DSCP value of the route's outgoing packets to 0.
	The " af " choices stand for Assured Forwarding. The number following the " af " identifies one of four classes and one of three drop preferences. See Section 10.4 on page 206 for more details.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 127 Configuration > Bandwidth Management

25.2.1 The Bandwidth Management Add/Edit Screen

The **Configuration > Bandwidth Management Add/Edit** screen allows you to create a new condition or edit an existing one. To access this screen, go to the **Configuration > Bandwidth Management** screen (see Section 25.2 on page 343), and click either the **Add** icon or an **Edit** icon.

Figure 233 Configuration > Bandwidth Management > Edit (For the Default Policy)

O Add Policy					? ×
🔠 Create new Object 🗸					
Bandwidth Shaping					
Guaranteed Bandwidth	Inbound Priority:	7			
	Outbound Priority:	7			
				ОК	Cancel

Add Policy							?
Create new Object 🗸							
onfiguration							
🔽 Enable							
Description:			(Optional)				
Iriteria			14				
User:	any		*				
Schedule:	none		~				
Incoming Interface:	any		~				
Outgoing Interface:	any		*				
Source:	any		*				
Destination:	any		~				
DSCP Code:	any		~				
Service Object:	any		*				
4 M 10 - 201							
SCP Marking		F					
DSCP Marking	Inbound Mar	king:	preserve 💙				
	Outbound Ma	arking:	preserve 💙				
andwidth Chaning							
Guaranteed Bandwidth	Inhounds	0	liber (0 - disabled)	Priority	4	6	
		Desidualda	KDps (U: disabled)	Phoney.	4	-	
	Maximize	Danuwiuun	Usage	Maxingin;	U	KDDS	
	Outbound:	0	kbos (0 · disabled)	Priority:	4		
	Maximize	Bandwidth	Lisane	Maximum:	0	khor	
	·				×	raps	
lelated Setting							
Log	no		~				

Figure 234 Configuration > Bandwidth Management > Add/Edit

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Create new Object	Use to configure any new settings objects that you need to use in this screen.
Configuration	
Enable	Select this check box to turn on this policy.
Description	Enter a description of this policy. It is not used elsewhere. You can use alphanumeric and ()+/:=?!*#@ $_{e}$ - characters, and it can be up to 60 characters long.
Criteria	Use this section to configure the conditions of traffic to which this policy applies.
User	Select a user name or user group to which to apply the policy. Use Create new Object if you need to configure a new user account. Select any to apply the policy for every user.
Schedule	Select a schedule that defines when the policy applies or select Create Object to configure a new one (see Chapter 30 on page 386 for details). Otherwise, select none to make the policy always effective.
Incoming Interface	Select the source interface of the traffic to which this policy applies.

LABEL	DESCRIPTION
Outgoing Interface	Select the destination interface of the traffic to which this policy applies.
Source	Select a source address or address group for whom this policy applies. Use Create new Object if you need to configure a new one. Select any if the policy is effective for every source.
Destination	Select a destination address or address group for whom this policy applies. Use Create new Object if you need to configure a new one. Select any if the policy is effective for every destination.
DSCP Code	Select a DSCP code point value of incoming packets to which this policy
	route applies or select User Defined to specify another DSCP code point. The lower the number the higher the priority with the exception of 0 which is usually given only best-effort treatment.
	any means all DSCP value or no DSCP marker.
	default means traffic with a DSCP value of 0. This is usually best effort traffic
	The " af " choices stand for Assured Forwarding. The number following the " af " identifies one of four classes and one of three drop preferences.
	See Section 10.4 on page 206 for more details.
User-Defined DSCP Code	Use this field to specify a custom DSCP code point.
Service	This field is available if you selected Service Object as the service type.
	Select a service or service group to identify the type of traffic to which this policy applies. any means all services.
DSCP Marking	Set how the ZyWALL handles the DSCP value of the incoming and outgoing packets that match this policy. Inbound refers to the traffic the ZyWALL sends to a connection's initiator. Outbound refers to the traffic the ZyWALL sends out from a connection's initiator.
	Select one of the pre-defined DSCP values to apply or select User Defined to specify another DSCP value. The " af " choices stand for Assured Forwarding. The number following the " af " identifies one of four classes and one of three drop preferences. See Section 10.4 on page 206 for more details.
	Select preserve to have the ZyWALL keep the packets' original DSCP value.
	Select default to have the ZyWALL set the DSCP value of the packets to 0.
Bandwidth Shaping	Configure these fields to set the amount of bandwidth the matching traffic can use.
Inbound kbps	Type how much inbound bandwidth, in kilobits per second, this policy allows the traffic to use. Inbound refers to the traffic the ZyWALL sends to a connection's initiator.
	If you enter O here, this policy does not apply bandwidth management for the matching traffic that the ZyWALL sends to the initiator. Traffic with bandwidth management disabled (inbound and outbound are both set to 0) is automatically treated as the lowest priority (7).
	If the sum of the bandwidths for routes using the same next hop is higher than the actual transmission speed, lower priority traffic may not be sent if higher priority traffic uses all of the actual bandwidth.

Table 128Configuration > Bandwidth Management

LABEL	DESCRIPTION
Outbound kbps	Type how much outbound bandwidth, in kilobits per second, this policy allows the traffic to use. Outbound refers to the traffic the ZyWALL sends out from a connection's initiator.
	If you enter O here, this policy does not apply bandwidth management for the matching traffic that the ZyWALL sends out from the initiator. Traffic with bandwidth management disabled (inbound and outbound are both set to 0) is automatically treated as the lowest priority (7).
	If the sum of the bandwidths for routes using the same next hop is higher than the actual transmission speed, lower priority traffic may not be sent if higher priority traffic uses all of the actual bandwidth.
Priority	This field displays when the inbound or outbound bandwidth management is not set to 0. Enter a number between 1 and 7 to set the priority for traffic that matches this policy. The smaller the number, the higher the priority.
	Traffic with a higher priority is given bandwidth before traffic with a lower priority.
	The ZyWALL uses a fairness-based (round-robin) scheduler to divide bandwidth between traffic flows with the same priority.
	The number in this field is ignored if the incoming and outgoing limits are both set to 0. In this case the traffic is automatically treated as being set to the lowest priority (7) regardless of this field's configuration.
Maximize Bandwidth Usage	This field displays when the inbound or outbound bandwidth management is not set to 0. Enable maximize bandwidth usage to let the traffic matching this policy "borrow" any unused bandwidth on the out-going interface.
	After each application or type of traffic gets its configured bandwidth rate, the ZyWALL uses the fairness- based scheduler to divide any unused bandwidth on the out-going interface amongst applications and traffic types that need more bandwidth and have maximize bandwidth usage enabled.
Related Setting	
Log	Select whether to have the ZyWALL generate a log (log), log and alert (log alert) or neither (no) when any traffic matches this policy. See Chapter 38 on page 474 for more on logs.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

 Table 128
 Configuration > Bandwidth Management

Device HA

26.1 Overview

Device HA lets a backup ZyWALL (B) automatically take over if the master ZyWALL (A) fails.

Figure 235 Device HA Backup Taking Over for the Master



26.1.1 What You Can Do in this Chapter

- Use the **General** screen (Section 26.2 on page 350) to configure device HA global settings, and see the status of each interface monitored by device HA.
- Use the **Active-Passive Mode** screens (Section 26.3 on page 351) to use active-passive mode device HA. You can configure general active-passive mode device HA settings, view and manage the list of monitored interfaces, and synchronize backup ZyWALLs.

26.1.2 What You Need to Know

Active-Passive Mode

• Active-passive mode lets a backup ZyWALL take over if the master ZyWALL fails.

Management Access

You can configure a separate management IP address for each interface. You can use it to access the ZyWALL for management whether the ZyWALL is the master or a backup. The management IP address should be in the same subnet as the interface IP address.

Synchronization

Use synchronization to have a backup ZyWALL copy the master ZyWALL's configuration, and certificates.

Note: Only ZyWALLs of the same model and firmware version can synchronize.

Otherwise you must manually configure the master ZyWALL's settings on the backup (by editing copies of the configuration files in a text editor for example).

Finding Out More

• See Section 26.5 on page 356 for device HA background/technical information.

26.1.3 Before You Begin

• Configure a static IP address for each interface that you will have device HA monitor.

26.2 Device HA General

The **Configuration > Device HA General** screen lets you enable or disable device HA, and displays which device HA mode the ZyWALL is set to use along with a summary of the monitored interfaces.

Figure 236 Configuration > Device HA > General

General	Active-Passive M	ode				
General	Settings					
Enal	ble Device HA					
Device H	HA Mode:	Active-Passive Mode				
Monitore	d Interface Summ	ary				
# 1	Interface 🔺	Virtual Router IP/Netmask	Management IP / Netmask	Link Status	HA Status	
14 4	Page 1 of 1	▶ ▶ Show 50 vitems			No data t	o display
		Appl	ly Reset			

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Enable Device HA	Turn the ZyWALL's device HA feature on or off.
	Note: It is not recommended to use STP (Spanning Tree Protocol) with device HA.
Device HA Mode	This displays the ZyWALL is currently set to use active-passive mode device HA.
Monitored Interface Summary	This table shows the status of the interfaces that you selected for monitoring in the other device HA screens.
#	This is the entry's index number in the list.
Interface	These are the names of the interfaces that are monitored by device HA.
Virtual Router IP / Netmask	This is the interface's IP address and subnet mask. Whichever ZyWALL is the master uses this virtual router IP address and subnet mask.

Table 129Configuration > Device HA > General

LABEL	DESCRIPTION
Management IP / Netmask	This field displays the interface's management IP address and subnet mask. You can use this IP address and subnet mask to access the ZyWALL whether it is in master or backup mode.
Link Status	This tells whether the monitored interface's connection is down or up.
HA Status	The text before the slash shows whether the device is configured as the master or the backup role.
	This text after the slash displays the monitored interface's status in the virtual router.
	Active - This interface is up and using the virtual IP address and subnet mask.
	Stand-By - This interface is a backup interface in the virtual router. It is not using the virtual IP address and subnet mask.
	Fault - This interface is not functioning in the virtual router right now. In active-passive mode, if one of the master ZyWALL's interfaces loses its connection, the master ZyWALL forces all of its interfaces to the fault state so the backup ZyWALL can take over all of the master ZyWALL's functions.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 129
 Configuration > Device HA > General (continued)

26.3 The Active-Passive Mode Screen

Virtual Router

The master and backup ZyWALL form a single 'virtual router'. In the following example, master ZyWALL \bf{A} and backup ZyWALL \bf{B} form a virtual router.



Cluster ID

You can have multiple ZyWALL virtual routers on your network. Use a different cluster ID to identify each virtual router. In the following example, ZyWALLs **A** and **B** form a virtual router that uses cluster ID 1. ZyWALLs **C** and **D** form a virtual router that uses cluster ID 2.



Figure 238 Cluster IDs for Multiple Virtual Routers

Monitored Interfaces in Active-Passive Mode Device HA

You can select which interfaces device HA monitors. If a monitored interface on the ZyWALL loses its connection, device HA has the backup ZyWALL take over.

Enable monitoring for the same interfaces on the master and backup ZyWALLs. Each monitored interface must have a static IP address and be connected to the same subnet as the corresponding interface on the backup or master ZyWALL.

Virtual Router and Management IP Addresses

- If a backup takes over for the master, it uses the master's IP addresses. These IP addresses are know as the virtual router IP addresses.
- Each interface can also have a management IP address. You can connect to this IP address to manage the ZyWALL regardless of whether it is the master or the backup.

For example, ZyWALL **B** takes over **A**'s 192.168.1.1 LAN interface IP address. This is a virtual router IP address. ZyWALL **A** keeps it's LAN management IP address of 192.168.1.5 and ZyWALL **B** has its own LAN management IP address of 192.168.1.6. These do not change when ZyWALL **B** becomes the master.



Figure 239 Management IP Addresses

26.3.1 Configuring Active-Passive Mode Device HA

The **Device HA Active-Passive Mode** screen lets you configure general active-passive mode device HA settings, view and manage the list of monitored interfaces, and synchronize backup ZyWALLs. To access this screen, click **Configuration > Device HA > Active-Passive Mode**.

The following table describes the labels in this screen. See Section 26.4 on page 355 for more information as well.

LABEL	DESCRIPTION
Show Advanced Settings / Hide Advanced Settings	Click this button to display a greater or lesser number of configuration fields.
Device Role	Select the device HA role that the ZyWALL plays in the virtual router. Choices are:
	Master - This ZyWALL is the master ZyWALL in the virtual router. This ZyWALL uses the virtual IP address for each monitored interface.
	Note: Do not set this field to Master for two or more ZyWALLs in the same virtual router (same cluster ID).
	Backup - This ZyWALL is a backup ZyWALL in the virtual router. This ZyWALL does not use any of the virtual IP addresses.
Priority	This field is available for a backup ZyWALL. Type the priority of the backup ZyWALL. The backup ZyWALL with the highest value takes over the role of the master ZyWALL if the master ZyWALL becomes unavailable. The priority must be between 1 and 254. (The master interface has priority 255.)
Enable Preemption	This field is available for a backup ZyWALL. Select this if this ZyWALL should become the master ZyWALL if a lower-priority ZyWALL is the master when this one is enabled. (If the role is master, the ZyWALL preempts by default.)
Cluster Settings	
Cluster ID	Type the cluster ID number. A virtual router consists of a master ZyWALL and all of its backup ZyWALLs. If you have multiple ZyWALL virtual routers on your network, use a different cluster ID for each virtual router.
Authentication	Select the authentication method the virtual router uses. Every interface in a virtual router must use the same authentication method and password. Choices are:
	None - this virtual router does not use any authentication method.
	Text - this virtual router uses a plain text password for authentication. Type the password in the field next to the radio button. The password can consist of alphanumeric characters, the underscore, and some punctuation marks $(+-/*=:; .! @$ &%#~ ' \ ()), and it can be up to eight characters long.
	IP AH (MD5) - this virtual router uses an encrypted MD5 password for authentication. Type the password in the field next to the radio button. The password can consist of alphanumeric characters, the underscore, and some punctuation marks $(+-/*=:; .! @$ &%#~ ' \ ()), and it can be up to eight characters long.
	See Authentication Types on page 207 for more information about authentication methods.
Monitored Interface Summary	This table shows the status of the device HA settings and status of the ZyWALL's interfaces.
Edit	Select an entry and click this to be able to modify it.
Activate	To turn on an entry, select it and click Activate.

 Table 130
 Configuration > Device HA > Active-Passive Mode

LABEL	DESCRIPTION
Inactivate	To turn off an entry, select it and click Inactivate.
#	This is the entry's index number in the list.
Status	The activate (light bulb) icon is lit when the entry is active and dimmed when the entry is inactive.
Interface	This field identifies the interface. At the time of writing, Ethernet and bridge interfaces can be included in the active-passive mode virtual router. The member interfaces of any bridge interfaces do not display separately.
Virtual Router IP / Netmask	This is the master ZyWALL's (static) IP address and subnet mask for this interface. If a backup takes over for the master, it uses this IP address. These fields are blank if the interface is a DHCP client or has no IP settings.
Management IP / Netmask	This field displays the interface's management IP address and subnet mask. You can use this IP address and subnet mask to access the ZyWALL whether it is in master or backup mode.
Link Status	This tells whether the monitored interface's connection is down or up.
Synchronization	Use synchronization to have a backup ZyWALL copy the master ZyWALL's configuration, certificates.
	Every interface's management IP address must be in the same subnet as the interface's IP address (the virtual router IP address).
Server Address	If this ZyWALL is set to backup role, enter the IP address or Fully-Qualified Domain Name (FQDN) of the ZyWALL from which to get updated configuration. Usually, you should enter the IP address or FQDN of a virtual router on a secure network.
	If this ZyWALL is set to master role, this field displays the ZyWALL's IP addresses and/or Fully-Qualified Domain Names (FQDN) through which ZyWALLs in backup role can get updated configuration from this ZyWALL.
Sync. Now	This displays if the ZyWALL is set to use active-passive mode device HA, the ZyWALL is in the backup role and device HA is enabled. Click this to copy the specified ZyWALL's configuration.
Server Port	If this ZyWALL is set to the backup role, enter the port number to use for Secure FTP when synchronizing with the specified master ZyWALL.
	If this ZyWALL is set to master role, this field displays the ZyWALL's Secure FTP port number. Click the Configure link if you need to change the FTP port number.
	Every ZyWALL in the virtual router must use the same port number. If the master ZyWALL changes, you have to manually change this port number in the backups.
Password	Enter the password used for verification during synchronization. Every ZyWALL in the virtual router must use the same password.
	If you leave this field blank in the master ZyWALL, no backup ZyWALLs can synchronize from it.
	If you leave this field blank in a backup ZyWALL, it cannot synchronize from the master ZyWALL.
Retype to Confirm	Type the password again here to confirm it.
Auto Synchronize	Select this to get the updated configuration automatically from the specified ZyWALL according to the specified Interval . The first synchronization begins after the specified Interval ; the ZyWALL does not synchronize immediately.
Interval	When you select Auto Synchronize , set how often the ZyWALL synchronizes with the master.
Next Sync Time	This appears the next time and date (in hh:mm yyyy-mm-dd format) the ZyWALL will synchronize with the master.

 Table 130
 Configuration > Device HA > Active-Passive Mode (continued)

LABEL	DESCRIPTION
Apply	This appears when the ZyWALL is currently using active-passive mode device HA. Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 130 Configuration > Device HA > Active-Passive Mode (continued)

26.4 Configuring an Active-Passive Mode Monitored Interface

The **Device HA Active-Passive Mode Monitored Interface Edit** screen lets you enable or disable monitoring of an interface and set the interface's management IP address and subnet mask. To access this screen, click **Configuration > Device HA > Active-Passive Mode > Edit**.

If you configure device HA settings for an Ethernet interface and later add the Ethernet interface to a bridge, the ZyWALL retains the interface's device HA settings and uses them again if you later remove the interface from the bridge. If the bridge is later deleted or the interface is removed from it, Device HA will recover the interface's setting.

A bridge interface's device HA settings are not retained if you delete the bridge interface.

Enable Monitored Interface		<u> </u>
Interface Name:	ge1	
Virtual Router IP(VRIP)/Subnet Mask:	192.168.1.1/255.255.255.0	
Manage IP:		
Manage IP Subnet Mask:	255.255.255.0	
		~
		Cancel
		Garreet
ure 241 Configuration	> Device HA > Active	-Passive Mode >
ure 241 Configuration	> Device HA > Active	-Passive Mode >
ure 241 Configuration	> Device HA > Active	-Passive Mode >
ure 241 Configuration dit Monitored Interface	> Device HA > Active	-Passive Mode >
ure 241 Configuration dit Monitored Interface Enable Monitored Interface Interface Name:	van1	-Passive Mode >
ure 241 Configuration dit Monitored Interface Enable Monitored Interface Interface Name: Virtual Router IP(VRIP)/Subnet Mask:	<pre>wan1 192.168.1.1/255.255.255.0</pre>	-Passive Mode >
	<pre>> Device HA > Active wan1 192.168.1.1/255.255.0</pre>	-Passive Mode >
Ire 241 Configuration dit Monitored Interface Enable Monitored Interface Interface Name: Virtual Router IP(VRIP)/Subnet Mask: Manage IP: Manage IP Subnet Mask:	> Device HA > Active wan1 192.168.1.1/255.255.255.0 255.255.255.0	-Passive Mode >
ure 241 Configuration dit Monitored Interface Enable Monitored Interface Interface Name: Virtual Router IP(VRIP)/Subnet Mask: Manage IP: Manage IP Subnet Mask:	<pre>wan1 192.168.1.1/255.255.255.0 255.255.0</pre>	-Passive Mode >

Figure 240 Configuration > Device HA > Active-Passive Mode > Edit

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Enable Monitored Interface	Select this to have device HA monitor the status of this interface's connection.
Interface Name	This identifies the interface.
	Note: Do not connect the bridge interfaces on two ZyWALLs without device HA activated on both. Doing so could cause a broadcast storm.
	Either activate device HA before connecting the bridge interfaces or disable the bridge interfaces, connect the bridge interfaces, activate device HA, and finally reactivate the bridge interfaces.
Virtual Router IP (VRIP) / Subnet Mask	This is the interface's (static) IP address and subnet mask in the virtual router. Whichever ZyWALL is currently serving as the master uses this virtual router IP address and subnet mask. These fields are blank if the interface is a DHCP client or has no IP settings.
Manage IP	Enter the interface's IP address for management access. You can use this IP address to access the ZyWALL whether it is the master or a backup. This management IP address should be in the same subnet as the interface IP address.
Manage IP Subnet Mask	Enter the subnet mask of the interface's management IP address.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

 Table 131
 Configuration > Device HA > Active-Passive Mode > Edit

26.5 Device HA Technical Reference

Active-Passive Mode Device HA with Bridge Interfaces

Here are two ways to avoid a broadcast storm when you connect the bridge interfaces on two ZyWALLs.

First Option for Connecting the Bridge Interfaces on Two ZyWALLs

The first way is to activate device HA before connecting the bridge interfaces as shown in the following example.

1 Make sure the bridge interfaces of the master ZyWALL (A) and the backup ZyWALL (B) are not connected.



2 Configure the bridge interface on the master ZyWALL, set the bridge interface as a monitored interface, and activate device HA.



3 Configure the bridge interface on the backup ZyWALL, set the bridge interface as a monitored interface, and activate device HA.



4 Connect the ZyWALLs.



Second Option for Connecting the Bridge Interfaces on Two ZyWALLs

Another option is to disable the bridge interfaces, connect the bridge interfaces, activate device HA, and finally reactivate the bridge interfaces as shown in the following example.

1 In this case the ZyWALLs are already connected, but the bridge faces have not been configured yet. Configure a disabled bridge interface on the master ZyWALL but disable it. Then set the bridge interface as a monitored interface, and activate device HA.



2 Configure a corresponding disabled bridge interface on the backup ZyWALL. Then set the bridge interface as a monitored interface, and activate device HA.



3 Enable the bridge interface on the master ZyWALL and then on the backup ZyWALL.





Synchronization

During synchronization, the master ZyWALL sends the following information to the backup ZyWALL.

- Startup configuration file (startup-config.conf)
- Certificates (My Certificates, and Trusted Certificates)

Synchronization does not change the device HA settings in the backup ZyWALL.

Synchronization affects the entire device configuration. You can only configure one set of settings for synchronization, regardless of how many VRRP groups you might configure. The ZyWALL uses Secure FTP (on a port number you can change) to synchronize, but it is still recommended that the backup ZyWALL synchronize with a master ZyWALL on a secure network.

The backup ZyWALL gets the configuration from the master ZyWALL. The backup ZyWALL cannot become the master or be managed while it applies the new configuration. This usually takes two or three minutes or longer depending on the configuration complexity.

The following restrictions apply with active-passive mode.

• The master ZyWALL must have no inactive monitored interfaces.

• The backup ZyWALL cannot be the master. This refers to the actual role at the time of synchronization, not the role setting in the configuration screen.
User/Group

27.1 Overview

This chapter describes how to set up user accounts, user groups, and user settings for the ZyWALL. You can also set up rules that control when users have to log in to the ZyWALL before the ZyWALL routes traffic for them.

27.1.1 What You Can Do in this Chapter

- The User screen (see Section 27.2 on page 363) provides a summary of all user accounts.
- The **Group** screen (see Section 27.3 on page 366) provides a summary of all user groups. In addition, this screen allows you to add, edit, and remove user groups. User groups may consist of access users and other user groups. You cannot put admin users in user groups
- The **Setting** screen (see Section 27.4 on page 368) controls default settings, login settings, lockout settings, and other user settings for the ZyWALL. You can also use this screen to specify when users must log in to the ZyWALL before it routes traffic for them.

27.1.2 What You Need To Know

User Account

A user account defines the privileges of a user logged into the ZyWALL. User accounts are used in firewall rules, in addition to controlling access to configuration and services in the ZyWALL.

User Types

These are the types of user accounts the ZyWALL uses.

ТҮРЕ	ABILITIES	LOGIN METHOD(S)			
Admin Users					
admin	Change ZyWALL configuration (web, CLI)	WWW, TELNET, SSH, FTP, Console			
limited-admin	Look at ZyWALL configuration (web, CLI)	WWW, TELNET, SSH, Console			
	Perform basic diagnostics (CLI)				
Access Users					
user	Access network services	WWW, TELNET, SSH			
	Browse user-mode commands (CLI)				
guest	Access network services	WWW			
ext-user	External user account	WWW			
ext-group-user	External group user account	WWW			

 Table 132
 Types of User Accounts

Note: The default **admin** account is always authenticated locally, regardless of the authentication method setting. (See Chapter 32 on page 399 for more information about authentication methods.)

Ext-User Accounts

Set up an **ext-user** account if the user is authenticated by an external server and you want to set up specific policies for this user in the ZyWALL. If you do not want to set up policies for this user, you do not have to set up an **ext-user** account.

All **ext-user** users should be authenticated by an external server, such as AD, LDAP or RADIUS. If the ZyWALL tries to use the local database to authenticate an **ext-user**, the authentication attempt always fails. (This is related to AAA servers and authentication methods, which are discussed in Chapter 31 on page 390 and Chapter 32 on page 399, respectively.)

Note: If the ZyWALL tries to authenticate an **ext-user** using the local database, the attempt always fails.

Once an **ext-user** user has been authenticated, the ZyWALL tries to get the user type (see Table 132 on page 361) from the external server. If the external server does not have the information, the ZyWALL sets the user type for this session to **User**.

For the rest of the user attributes, such as reauthentication time, the ZyWALL checks the following places, in order.

- **1** User account in the remote server.
- 2 User account (Ext-User) in the ZyWALL.
- 3 Default user account for AD users (ad-users), LDAP users (Idap-users) or RADIUS users (radiususers) in the ZyWALL.

See Setting up User Attributes in an External Server on page 372 for a list of attributes and how to set up the attributes in an external server.

Ext-Group-User Accounts

Ext-Group-User accounts work are similar to ext-user accounts but allow you to group users by the value of the group membership attribute configured for the AD or LDAP server. See Section 31.2.1 on page 393 for more on the group membership attribute.

User Groups

User groups may consist of user accounts or other user groups. Use user groups when you want to create the same rule for several user accounts, instead of creating separate rules for each one.

Note: You cannot put access users and admin users in the same user group.

Note: You cannot put the default **admin** account into any user group.

The sequence of members in a user group is not important.

User Awareness

By default, users do not have to log into the ZyWALL to use the network services it provides. The ZyWALL automatically routes packets for everyone. If you want to restrict network services that certain users can use via the ZyWALL, you can require them to log in to the ZyWALL first. The ZyWALL is then 'aware' of the user who is logged in and you can create 'user-aware policies' that define what services they can use. See Section 27.4.2 on page 371 for a user-aware login example.

Finding Out More

• See Section 27.5 on page 372 for some information on users who use an external authentication server in order to log in.

27.2 User Summary Screen

The **User** screen provides a summary of all user accounts. To access this screen, login to the Web Configurator, and click **Configuration > Object > User/Group**.

Figure 242 (Configuration	> Ob	oject	>	User/Gro	up
--------------	---------------	------	-------	---	----------	----

figura	tion			
🗿 Add	📝 Edit 🍵 Remove 🕞 Object Refer	ence		
# *	User Name	User Type	Description	
1	admin	admin	Administration account	
2	Idap-users	ext-user	External LDAP Users	
3	radius-users	ext-user	External RADIUS Users	
4	ad-users	ext-user	External AD Users	
5	charlotte	admin	Local User	
6	steve	admin	Local User	
7	sharno	admin	Local User	
8	james	admin	Local User	
9	SSL-Users	user	SSL User	

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific user.
User Name	This field displays the user name of each user.

Table 133 Configuration > Object > User/Group

LABEL	DESCRIPTION
User Type	This field displays the types of user accounts the ZyWALL uses:
	admin - this user can look at and change the configuration of the ZyWALL
	 limited-admin - this user can look at the configuration of the ZyWALL but not to change it
	• user - this user has access to the ZyWALL's services and can also browse user-mode commands (CLI).
	 guest - this user has access to the ZyWALL's services but cannot look at the configuration
	• ext-user - this user account is maintained in a remote server, such as RADIUS or LDAP. See Ext-User Accounts on page 362 for more information about this type.
	• ext-group-user - this user account is maintained in a remote server, such as RADIUS or LDAP. See Ext-Group-User Accounts on page 362 for more information about this type.
Description	This field displays the description for each user.

Table 133 Configuration > Object > User/Group (continued)

27.2.1 User Add/Edit Screen

The User Add/Edit screen allows you to create a new user account or edit an existing one.

27.2.1.1 Rules for User Names

Enter a user name from 1 to 31 characters.

The user name can only contain the following characters:

- Alphanumeric A-z 0-9 (there is no unicode support)
- _ [underscores]
- - [dashes]

The first character must be alphabetical (A-Z a-z), an underscore (_), or a dash (-). Other limitations on user names are:

- User names are case-sensitive. If you enter a user 'bob' but use 'BOB' when connecting via CIFS or FTP, it will use the account settings used for 'BOB' not 'bob'.
- User names have to be different than user group names.
- Here are the reserved user names:

•	adm	•	admin	•	any	•	bin	•	daemon
•	debug	•	devicehaecived	•	ftp	•	games	•	halt
•	ldap-users	•	lp	•	mail	•	news	•	nobody
•	operator	•	radius-users	•	root	•	shutdown	•	sshd
•	sync	•	uucp	•	zyxel				

To access this screen, go to the **User** screen (see Section 27.2 on page 363), and click either the **Add** icon or an **Edit** icon.

Jser Name :		
Jser Type:	user 💌	
Password:]
Retype:		
Description:	Local User	
Authentication Timeout Settings	Use Default Settings	Use Manual Settings
Lease Time:	1440	minutes
Reauthentication Time:	1440	minutes

Figure 243 Configuration > User/Group > User > Add

The following table describes the labels in this screen.

LABEL	DESCRIPTION
User Name	Type the user name for this user account. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive. User names have to be different than user group names, and some words are reserved. See Section 27.2.1.1 on page 364.
User Type	This field displays the types of user accounts the ZyWALL uses:
	 admin - this user can look at and change the configuration of the ZyWALL limited-admin - this user can look at the configuration of the ZyWALL but not to change it
	 user - this user has access to the ZyWALL's services and can also browse user- mode commands (CLI).
	 guest - this user has access to the ZyWALL's services but cannot look at the configuration.
	• ext-user - this user account is maintained in a remote server, such as RADIUS or LDAP. See Ext-User Accounts on page 362 for more information about this type.
	 ext-group-user - this user account is maintained in a remote server, such as RADIUS or LDAP. See Ext-Group-User Accounts on page 362 for more information about this type.
Password	This field is not available if you select the ext-user or ext-group-user type.
	Enter the password of this user account. It can consist of 4 - 31 alphanumeric characters.
Retype	This field is not available if you select the ext-user or ext-group-user type.
Group Identifier	This field is available for a ext-group-user type user account.
	Specify the value of the AD or LDAP server's Group Membership Attribute that identifies the group to which this user belongs.
Associated AAA Server Object	This field is available for a ext-group-user type user account. Select the AAA server to use to authenticate this account's users.
Description	Enter the description of each user, if any. You can use up to 60 printable ASCII characters. Default descriptions are provided.
Authentication Timeout Settings	If you want the system to use default settings, select Use Default Settings . If you want to set authentication timeout to a value other than the default settings, select Use Manual Settings then fill your preferred values in the fields that follow.

Т	able 134	Configuration >	User/Group	>	User	>	Add

LABEL	DESCRIPTION
Lease Time	If you select Use Default Settings in the Authentication Timeout Settings field, the default lease time is shown.
	If you select Use Manual Settings , you need to enter the number of minutes this user has to renew the current session before the user is logged out. You can specify 1 to 1440 minutes. You can enter 0 to make the number of minutes unlimited. Admin users renew the session every time the main screen refreshes in the Web Configurator. Access users can renew the session by clicking the Renew button on their screen. If you allow access users to renew time automatically (see Section 27.4 on page 368), the users can select this check box on their screen as well. In this case, the session is automatically renewed before the lease time expires.
Reauthentication Time	If you select Use Default Settings in the Authentication Timeout Settings field, the default lease time is shown.
	If you select Use Manual Settings , you need to type the number of minutes this user can be logged into the ZyWALL in one session before the user has to log in again. You can specify 1 to 1440 minutes. You can enter 0 to make the number of minutes unlimited. Unlike Lease Time , the user has no opportunity to renew the session without logging out.
Configuration Validation	Use a user account from the group specified above to test if the configuration is correct. Enter the account's user name in the User Name field and click Test .
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

Table 134 Configuration > User/Group > User > Add (continued)

27.3 User Group Summary Screen

User groups consist of access users and other user groups. You cannot put admin users in user groups. The **Group** screen provides a summary of all user groups. In addition, this screen allows you to add, edit, and remove user groups. To access this screen, login to the Web Configurator, and click **Configuration > Object > User/Group > Group**.

Figure 244 Configuration > Object > User/Group > Group

onfiguration			
🗿 Add 📝 Edit 🍵 Remove	G Object References		
# Group Name -	Description	Member	
	Change So and Brand		KIN JAKA KA UKAMAN

The following table describes the labels in this screen. See Section 27.3.1 on page 367 for more information as well.

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Removing a group does not remove the user accounts in the group.

Table 135Configuration > Object > User/Group > Group

LABEL	DESCRIPTION
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific user group.
Group Name	This field displays the name of each user group.
Description	This field displays the description for each user group.
Member	This field lists the members in the user group. Each member is separated by a comma.

Table 135 Configuration > Object > User/Group > Group (continued)

27.3.1 Group Add/Edit Screen

The **Group Add/Edit** screen allows you to create a new user group or edit an existing one. To access this screen, go to the **Group** screen (see Section 27.3 on page 366), and click either the **Add** icon or an **Edit** icon.

gure 245	Configur	ation >	> User/G	roup >	> Group) >	Ad
Add Group						1	?
Configuration							
Name:			0				
Description:			(Optional)				
Member List							
Available === (ad-users guest Idap-users radius-users	Jbject ===	•	Member				
					ok	Cance	1

LABEL	DESCRIPTION
Name	Type the name for this user group. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive. User group names have to be different than user names.
Description	Enter the description of the user group, if any. You can use up to 60 characters, punctuation marks, and spaces.
Member List	The Member list displays the names of the users and user groups that have been added to the user group. The order of members is not important. Select users and groups from the Available list that you want to be members of this group and move them to the Member list. You can double-click a single entry to move it or use the [Shift] or [Ctrl] key to select multiple entries and use the arrow button to move them. Move any members you do not want included to the Available list.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

 Table 136
 Configuration > User/Group > Group > Add

27.4 The User/Group Setting Screen

The **Setting** screen controls default settings, login settings, lockout settings, and other user settings for the ZyWALL. You can also use this screen to specify when users must log in to the ZyWALL before it routes traffic for them.

To access this screen, login to the Web Configurator, and click **Configuration > Object > User/ Group > Setting**.

	neout settings			
📝 Edit				
# User Type	Lease	e Time	Reauthentication	Time
1 admin	1440		1440	
2 limited-admin	1440		1440	
3 user	1440		1440	
4 guest	1440	1440 1440		
5 ext-user	1440		1440	
6 ext-group-user	1440		1440	
-				
Allow renewing lease time au Enable user idle detection User idle timeout: Ser Logon Settings	itomatically	3	(1-60 minutes)	
iiscellaneous Settings Allow renewing lease time au Enable user idle detection User idle timeout: ser Logon Settings Use: time number of cimultane	itomatically	3	(1-60 minutes)	
iiscellaneous Settings Allow renewing lease time au Enable user idle detection User idle timeout: ser Logon Settings Limit the number of simultane Maximum pumber per admini	itomatically sous logons for admini	3	(1-60 minutes)	
iiscellaneous Settings Allow renewing lease time au Enable user idle detection User idle timeout: ser Logon Settings Limit the number of simultane Maximum number per admini	itomatically eous logons for admini stration account:	3 stration account	(1-60 minutes) (1-256)	
 Allow renewing lease time au Enable user idle detection User idle timeout: Ser Logon Settings Limit the number of simultan Maximum number per admini Limit the number of simultan 	itomatically eous logons for admini stration account: eous logons for access	3 stration account 1 account	(1-60 minutes) (1-256)	
iiscellaneous Settings Allow renewing lease time au Enable user idle detection User idle timeout: ser Logon Settings Limit the number of simultane Maximum number per admini Limit the number of simultane Maximum number per access	itomatically eous logons for admini stration account: eous logons for access : account:	3 stration account 1 account 1	(1-60 minutes) (1-256) (1-256)	
iiscellaneous Settings Allow renewing lease time at Enable user idle detection User idle timeout: ser Logon Settings Limit the number of simultane Maximum number per admini Limit the number of simultane Maximum number per access ser Lockout Settings	itomatically eous logons for admini stration account: eous logons for access : account:	3 stration account 1 account 1	(1-60 minutes) (1-256) (1-256)	
 Allow renewing lease time au Allow renewing lease time au Enable user idle detection User idle timeout: Ser Logon Settings Limit the number of simultan Maximum number per admini Limit the number of simultans Maximum number per access Ser Lockout Settings Enable logon retry limit 	itomatically eous logons for admini stration account: eous logons for access : account:	3 stration account 1 account 1	(1-60 minutes) (1-256) (1-256)	
iiscellaneous Settings Allow renewing lease time au Enable user idle detection User idle timeout: ser Logon Settings Limit the number of simultane Maximum number per admini Limit the number of simultane Maximum number per access ser Lockout Settings Imable logon retry limit Maximum retry count:	itomatically eous logons for admini stration account: eous logons for access : account:	3 stration account 1 account 1	(1-60 minutes) (1-256) (1-256) (1-299)	
 Cellaneous Settings Allow renewing lease time au Enable user idle detection User idle timeout: Ser Logon Settings Limit the number of simultane Maximum number per admini Limit the number of simultane Maximum number per access Ser Lockout Settings Enable logon retry limit Maximum retry count: Lockout period: 	itomatically eous logons for admini stration account: eous logons for access : account:	3 stration account 1 account 1 5 30	(1-60 minutes) (1-256) (1-256) (1-256) (1-65535 minutes)	

Figure 246 Configuration > Object > User/Group > Setting

 Table 137
 Configuration > Object > User/Group > Setting

LABEL	DESCRIPTION	
User Authentication Timeout Settings		
Default Authentication Timeout Settings	These authentication timeout settings are used by default when you create a new user account. They also control the settings for any existing user accounts that are set to use the default settings. You can still manually configure any user account's authentication timeout settings.	
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.	

LABEL	DESCRIPTION
#	This field is a sequential value, and it is not associated with a specific entry.
User Type	These are the kinds of user account the ZyWALL supports.
	 admin - this user can look at and change the configuration of the ZyWALL
	Iimited-admin - this user can look at the configuration of the ZyWALL but not to change it
	user - this user has access to the ZyWALL's services but cannot look at the configuration
	guest - this user has access to the ZyWALL's services but cannot look at the configuration
	• ext-user - this user account is maintained in a remote server, such as RADIUS or LDAP. See Ext-User Accounts on page 362 for more information about this type.
	• ext-group-user - this user account is maintained in a remote server, such as RADIUS or LDAP. See Ext-Group-User Accounts on page 362 for more information about this type.
Lease Time	This is the default lease time in minutes for each type of user account. It defines the number of minutes the user has to renew the current session before the user is logged out.
	Admin users renew the session every time the main screen refreshes in the Web Configurator. Access users can renew the session by clicking the Renew button on their screen. If you allow access users to renew time automatically (see Section 27.4 on page 368), the users can select this check box on their screen as well. In this case, the session is automatically renewed before the lease time expires.
Reauthentication Time	This is the default reauthentication time in minutes for each type of user account. It defines the number of minutes the user can be logged into the ZyWALL in one session before having to log in again. Unlike Lease Time , the user has no opportunity to renew the session without logging out.
Miscellaneous Settings	·
Allow renewing lease time automatically	Select this check box if access users can renew lease time automatically, as well as manually, simply by selecting the Updating lease time automatically check box on their screen.
Enable user idle detection	This is applicable for access users.
	Select this check box if you want the ZyWALL to monitor how long each access user is logged in and idle (in other words, there is no traffic for this access user). The ZyWALL automatically logs out the access user once the User idle timeout has been reached.
User idle timeout	This is applicable for access users.
	This field is effective when Enable user idle detection is checked. Type the number of minutes each access user can be logged in and idle before the ZyWALL automatically logs out the access user.
User Logon Settings	
Limit the number of simultaneous logons for administration account	Select this check box if you want to set a limit on the number of simultaneous logins by admin users. If you do not select this, admin users can login as many times as they want at the same time using the same or different IP addresses.
Maximum number per administration account	This field is effective when Limit for administration account is checked. Type the maximum number of simultaneous logins by each admin user.
Limit the number of simultaneous logons for access account	Select this check box if you want to set a limit on the number of simultaneous logins by non-admin users. If you do not select this, access users can login as many times as they want as long as they use different IP addresses.

Table 137	Configuration >	Object >	User/Group	> Setting	(continued)
	configuration >		USCI/ UI UUp	> Setting	(continucu)

LABEL	DESCRIPTION
Maximum number per access account	This field is effective when Limit for access account is checked. Type the maximum number of simultaneous logins by each access user.
User Lockout Settings	
Enable logon retry limit	Select this check box to set a limit on the number of times each user can login unsuccessfully (for example, wrong password) before the IP address is locked out for a specified amount of time.
Maximum retry count	This field is effective when Enable logon retry limit is checked. Type the maximum number of times each user can login unsuccessfully before the IP address is locked out for the specified lockout period . The number must be between 1 and 99.
Lockout period	This field is effective when Enable logon retry limit is checked. Type the number of minutes the user must wait to try to login again, if logon retry limit is enabled and the maximum retry count is reached. This number must be between 1 and 65,535 (about 45.5 days).
Apply	Click Apply to save the changes.
Reset	Click Reset to return the screen to its last-saved settings.

Table 137 Configuration > Object > User/Group > Setting (continued)

27.4.1 Default User Authentication Timeout Settings Edit Screens

The **Default Authentication Timeout Settings Edit** screen allows you to set the default authentication timeout settings for the selected type of user account. These default authentication timeout settings also control the settings for any existing user accounts that are set to use the default settings. You can still manually configure any user account's authentication timeout settings.

To access this screen, go to the **Configuration > Object > User/Group > Setting** screen (see Section 27.4 on page 368), and click one of the **Default Authentication Timeout Settings** section's **Edit** icons.

dit User Authentication	Timeout Settings	?
User Type:	admin	
Lease Time:	1440	(0-1440 minutes, 0 is unlimited)
Reauthentication Time:	1440	(0-1440 minutes, 0 is unlimited)
	1110	(011101111100) 010 011111100)

Figure 247 Configuration > Object > User/Group > Setting > Edit

The following table describes the labels in this screen.

LABEL	DESCRIPTION
User Type	This read-only field identifies the type of user account for which you are configuring the default settings.
	admin - this user can look at and change the configuration of the ZyWALL
	• limited-admin - this user can look at the configuration of the ZyWALL but not to change it.
	 user - this user has access to the ZyWALL's services but cannot look at the configuration.
	 guest - this user has access to the ZyWALL's services but cannot look at the configuration.
	• ext-user - this user account is maintained in a remote server, such as RADIUS or LDAP. See Ext-User Accounts on page 362 for more information about this type.
	• ext-group-user - this user account is maintained in a remote server, such as RADIUS or LDAP. See Ext-Group-User Accounts on page 362 for more information about this type.
Lease Time	Enter the number of minutes this type of user account has to renew the current session before the user is logged out. You can specify 1 to 1440 minutes. You can enter 0 to make the number of minutes unlimited.
	Admin users renew the session every time the main screen refreshes in the Web Configurator. Access users can renew the session by clicking the Renew button on their screen. If you allow access users to renew time automatically (see Section 27.4 on page 368), the users can select this check box on their screen as well. In this case, the session is automatically renewed before the lease time expires.
Reauthentication Time	Type the number of minutes this type of user account can be logged into the ZyWALL in one session before the user has to log in again. You can specify 1 to 1440 minutes. You can enter 0 to make the number of minutes unlimited. Unlike Lease Time , the user has no opportunity to renew the session without logging out.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

Table 138 Configuration > Object > User/Group > Setting > Edit

27.4.2 User Aware Login Example

Access users cannot use the Web Configurator to browse the configuration of the ZyWALL. Instead, after access users log into the ZyWALL, the following screen appears.

Figure 248 Web Configurator for Non-Admin Users

Click the logout button to terminate the You could renew your lease time by click For security reason you must login in aga	access session, ing the Renew button, in after 24 hours 0 minutes,
Jser-defined lease time (max 1440 minu	tes): 1440 Renew
Updating lease time automatically	
Remaining time before lease timeout (hh:mm:ss):	23:59:40
Remaining time before auth. timeout (hh:mm):	23:59

The following table describes the labels in this screen.

LABEL	DESCRIPTION
User-defined lease time (max minutes)	Access users can specify a lease time shorter than or equal to the one that you specified. The default value is the lease time that you specified.
Renew	Access users can click this button to reset the lease time, the amount of time remaining before the ZyWALL automatically logs them out. The ZyWALL sets this amount of time according to the
	 User-defined lease time field in this screen Lease time field in the User Add/Edit screen (see Section 27.2.1 on page 364) Lease time field in the Setting screen (see Section 27.4 on page 368)
Updating lease time automatically	This box appears if you checked the Allow renewing lease time automatically box in the Setting screen. (See Section 27.4 on page 368.) Access users can select this check box to reset the lease time automatically 30 seconds before it expires. Otherwise, access users have to click the Renew button to reset the lease time.
Remaining time before lease timeout	This field displays the amount of lease time that remains, though the user might be able to reset it.
Remaining time before auth. timeout	This field displays the amount of time that remains before the ZyWALL automatically logs the access user out, regardless of the lease time.

 Table 139
 Web Configurator for Non-Admin Users

27.5 User /Group Technical Reference

This section provides some information on users who use an external authentication server in order to log in.

Setting up User Attributes in an External Server

To set up user attributes, such as reauthentication time, in LDAP or RADIUS servers, use the following keywords in the user configuration file.

KEYWORD	CORRESPONDING ATTRIBUTE IN WEB CONFIGURATOR	
type	User Type. Possible Values: admin, limited-admin, user, guest.	
leaseTime	Lease Time. Possible Values: 1-1440 (minutes).	
reauthTime	Reauthentication Time. Possible Values: 1-1440 (minutes).	

Table 140 LDAP/RADIUS: Keywords for User Attributes

The following examples show you how you might set up user attributes in LDAP and RADIUS servers.

Figure 249 LDAP Example: Keywords for User Attributes

```
type: admin
leaseTime: 99
reauthTime: 199
```

Figure 250 RADIUS Example: Keywords for User Attributes

type=user;leaseTime=222;reauthTime=222

Creating a Large Number of Ext-User Accounts

If you plan to create a large number of **Ext-User** accounts, you might use CLI commands, instead of the Web Configurator, to create the accounts. Extract the user names from the LDAP or RADIUS server, and create a shell script that creates the user accounts. See Chapter 39 on page 488 for more information about shell scripts.

Addresses

28.1 Overview

Address objects can represent a single IP address or a range of IP addresses. Address groups are composed of address objects and other address groups.

28.1.1 What You Can Do in this Chapter

- The **Address** screen (Section 28.2 on page 374) provides a summary of all addresses in the ZyWALL. Use the **Address Add/Edit** screen to create a new address or edit an existing one.
- Use the Address Group summary screen (Section 28.3 on page 378) and the Address Group Add/Edit screen, to maintain address groups in the ZyWALL.

28.1.2 What You Need To Know

Address objects and address groups are used in dynamic routes, firewall rules, and VPN connection policies. Please see the respective sections for more information about how address objects and address groups are used in each one.

Address groups are composed of address objects and address groups. The sequence of members in the address group is not important.

28.2 Address Summary Screen

The address screens are used to create, maintain, and remove addresses. There are the types of address objects.

- HOST a host address is defined by an IP Address.
- RANGE a range address is defined by a Starting IP Address and an Ending IP Address.
- SUBNET a network address is defined by a Network IP address and Netmask subnet mask.

The **Address** screen provides a summary of all addresses in the ZyWALL. To access this screen, click **Configuration > Object > Address > Address**. Click a column's heading cell to sort the table entries by that column's criteria. Click the heading cell again to reverse the sort order.

0	Add 📝 Edit 🍵 Remove 🔂 Object	Reference		
#	Name 🔺	Туре	IPv4 Address	
	DMZ_SUBNET	INTERFACE SUBNET	dmz-192.168.3.0/24	
2	IP6to4-Relay	HOST	192.88.99.1	
3	LAN1_SUBNET	INTERFACE SUBNET	lan1-192.168.1.0/24	
4	LAN2_SUBNET	INTERFACE SUBNET	lan2-192.168.2.0/24	
5	WIZ_LAN_SUBNET	INTERFACE SUBNET	wan1-172.23.26.0/24	
6	WLAN-1-1_SUBNET	INTERFACE SUBNET	wlan-1-1-10.59.1.0/24	
14	✓ Page 1 of 1 ▶ ▶ She	now 50 🗸 items		Displaying 1 - 6 of 6
v6 A	Address Configuration			
0	Add 📝 Edit 🍵 Remove 📠 Object	Reference		
#	Name 🔺	Туре	IPv6 Address	
	IPv6-RemoteGW	HOST	2002::5	

Figure 251 Configuration > Object > Address > Address

The following table describes the labels in this screen. See Section 28.2.1 on page 376 for more information as well.

LABEL	DESCRIPTION			
IPv4 Address Configuration				
Add	Click this to create a new entry.			
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.			
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.			
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.			
#	This field is a sequential value, and it is not associated with a specific address.			
Name	This field displays the configured name of each address object.			
Туре	This field displays the type of each address object. "INTERFACE" means the object uses the settings of one of the ZyWALL's interfaces.			
IPv4 Address	This field displays the IPv4 addresses represented by each address object. If the object's settings are based on one of the ZyWALL's interfaces, the name of the interface displays first followed by the object's current address settings.			
IPv6 Address Configuration				
Add	Click this to create a new entry.			
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.			
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.			
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.			
#	This field is a sequential value, and it is not associated with a specific address.			
Name	This field displays the configured name of each address object.			

 Table 141
 Configuration > Object > Address > Address

LABEL	DESCRIPTION
Туре	This field displays the type of each address object. "INTERFACE" means the object uses the settings of one of the ZyWALL's interfaces.
IPv6 Address	This field displays the IPv6 addresses represented by each address object. If the object's settings are based on one of the ZyWALL's interfaces, the name of the interface displays first followed by the object's current address settings.

Table 141 Configuration > Object > Address > Address (continued)

28.2.1 IPv4 Address Add/Edit Screen

The **Configuration** > **IPv4 Address Add/Edit** screen allows you to create a new address or edit an existing one. To access this screen, go to the **Address** screen (see Section 28.2 on page 374), and click either the **Add** icon or an **Edit** icon in the **IPv4 Address Configuration** section.

Figure 252	IPv4 Address	Configuration	>	Add/Edit
------------	--------------	---------------	---	----------

dd Address Rule	staw 50 v	? >
Name:		
Address Type:	HOST	*
IP Address:	0.0.0.0	
	[Cancel

Table 142	IPv4 Address	Configuration	>	Add/Edit
-----------	--------------	---------------	---	----------

LABEL	DESCRIPTION
Name	Type the name used to refer to the address. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Address Type	Select the type of address you want to create. Choices are: HOST, RANGE, SUBNET, INTERFACE IP, INTERFACE SUBNET, and INTERFACE GATEWAY.
	Note: The ZyWALL automatically updates address objects that are based on an interface's IP address, subnet, or gateway if the interface's IP address settings change. For example, if you change 1's IP address, the ZyWALL automatically updates the corresponding interface-based, LAN subnet address object.
IP Address	This field is only available if the Address Type is HOST . This field cannot be blank. Enter the IP address that this address object represents.
Starting IP Address	This field is only available if the Address Type is RANGE . This field cannot be blank. Enter the beginning of the range of IP addresses that this address object represents.
Ending IP Address	This field is only available if the Address Type is RANGE . This field cannot be blank. Enter the end of the range of IP address that this address object represents.
Network	This field is only available if the Address Type is SUBNET , in which case this field cannot be blank. Enter the IP address of the network that this address object represents.
Netmask	This field is only available if the Address Type is SUBNET , in which case this field cannot be blank. Enter the subnet mask of the network that this address object represents. Use dotted decimal format.
Interface	If you selected INTERFACE IP , INTERFACE SUBNET , or INTERFACE GATEWAY as the Address Type , use this field to select the interface of the network that this address object represents.

LABEL DESCRIPTION	
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

 Table 142
 IPv4 Address Configuration > Add/Edit (continued)

28.2.2 IPv6 Address Add/Edit Screen

The **Configuration** > **IPv6 Address Add/Edit** screen allows you to create a new address or edit an existing one. To access this screen, go to the **Address** screen (see Section 28.2 on page 374), and click either the **Add** icon or an **Edit** icon in the **IPv6 Address Configuration** section.

Figure 253	IPv6 Address	Configuration	>	Add/Edit
------------	--------------	---------------	---	----------

Name:			
Object Type:	HOST	~	
IPv6 Address:			

LABEL	DESCRIPTION
Name	Type the name used to refer to the address. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Object Type	Select the type of address you want to create. Choices are: HOST, RANGE, SUBNET, INTERFACE IP, INTERFACE SUBNET, and INTERFACE GATEWAY.
	Note: The ZyWALL automatically updates address objects that are based on an interface's IP address, subnet, or gateway if the interface's IP address settings change. For example, if you change 1's IP address, the ZyWALL automatically updates the corresponding interface-based, LAN subnet address object.
IPv6 Address	This field is only available if the Address Type is HOST . This field cannot be blank. Enter the IP address that this address object represents.
IPv6 Starting Address	This field is only available if the Address Type is RANGE . This field cannot be blank. Enter the beginning of the range of IP addresses that this address object represents.
IPv6 Ending Address	This field is only available if the Address Type is RANGE . This field cannot be blank. Enter the end of the range of IP address that this address object represents.
IPv6 Address Prefix	This field is only available if the Address Type is SUBNET . This field cannot be blank. Enter the IPv6 address prefix that the ZyWALL uses for the LAN IPv6 address.
Interface	If you selected INTERFACE IP , INTERFACE SUBNET , or INTERFACE GATEWAY as the Address Type , use this field to select the interface of the network that this address object represents.
IPv6 Address Type	Select whether the IPv6 address is a link-local IP address (LINK LOCAL), static IP address (STATIC), an IPv6 StateLess Address Auto Configuration IP address (SLAAC), or is obtained from a DHCPv6 server (DHCPv6).
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

 Table 143
 IPv6 Address Configuration > Add/Edit

28.3 Address Group Summary Screen

The **Address Group** screen provides a summary of all address groups. To access this screen, click **Configuration > Object > Address > Address Group**. Click a column's heading cell to sort the table entries by that column's criteria. Click the heading cell again to reverse the sort order.

Figure 254 Configuration > Object > Address > Address Group

ddress	Address Group		
Pv4 Addre	ess Group Configuration		
🕜 Add	📝 Edit 🍵 Remove 📠 Object Reference		
# Na	ame 🔺	Description	
IA A	Page 1 of 1 > > Show 50 v	tems	No data to display
O Add	🖉 Edit 🍵 Remove 📠 Object Reference		
# Na	ame 🔺	Description	
14 4	Page 1 of 1 > > Show 50 v	tems	No data to display

The following table describes the labels in this screen. See Section 28.3.1 on page 379 for more information as well.

LABEL	DESCRIPTION		
IPv4 Address Grou	IPv4 Address Group Configuration		
Add	Click this to create a new entry.		
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.		
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.		
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.		
#	This field is a sequential value, and it is not associated with a specific address group.		
Name	This field displays the name of each address group.		
Description	This field displays the description of each address group, if any.		
IPv6 Address Group Configuration			
Add	Click this to create a new entry.		
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.		
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.		
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.		
#	This field is a sequential value, and it is not associated with a specific address group.		
Name	This field displays the name of each address group.		
Description	This field displays the description of each address group, if any.		

Table 144 Configuration > Object > Address > Address Group

28.3.1 Address Group Add/Edit Screen

The Address Group Add/Edit screen allows you to create a new address group or edit an existing one. To access this screen, go to the Address Group screen (see Section 28.3 on page 378), and click either the Add icon or an Edit icon in the IPv4 Address Group Configuration or IPv6 Address Group Configuration section.

Group Members			
Name: Description:	~~~~~		
1ember List			
- Available === Object === DMZ_SUBNET IP6to4-Relay LAN1_SUBNET LAN2_SUBNET WIZ_LAN_SUBNET WLAN-1-1_SUBNET	•	Member	

Figure 255 IPv4/IPv6 Address Group Configuration > Add

LABEL	DESCRIPTION
Name	Enter a name for the address group. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Description	This field displays the description of each address group, if any. You can use up to 60 characters, punctuation marks, and spaces.
Member List	The Member list displays the names of the address and address group objects that have been added to the address group. The order of members is not important.
	Select items from the Available list that you want to be members and move them to the Member list. You can double-click a single entry to move it or use the [Shift] or [Ctrl] key to select multiple entries and use the arrow button to move them.
	Move any members you do not want included to the Available list.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

Table 145	IPv4/IPv6	Address	Group	Configuration	> Ad	d
				3		

Services

29.1 Overview

Use service objects to define TCP applications, UDP applications, and ICMP messages. You can also create service groups to refer to multiple service objects in other features.

29.1.1 What You Can Do in this Chapter

- Use the **Service** screens (Section 29.2 on page 381) to view and configure the ZyWALL's list of services and their definitions.
- Use the **Service Group** screens (Section 29.2 on page 381) to view and configure the ZyWALL's list of service groups.

29.1.2 What You Need to Know

IP Protocols

IP protocols are based on the eight-bit protocol field in the IP header. This field represents the nextlevel protocol that is sent in this packet. This section discusses three of the most common IP protocols.

Computers use Transmission Control Protocol (TCP, IP protocol 6) and User Datagram Protocol (UDP, IP protocol 17) to exchange data with each other. TCP guarantees reliable delivery but is slower and more complex. Some uses are FTP, HTTP, SMTP, and TELNET. UDP is simpler and faster but is less reliable. Some uses are DHCP, DNS, RIP, and SNMP.

TCP creates connections between computers to exchange data. Once the connection is established, the computers exchange data. If data arrives out of sequence or is missing, TCP puts it in sequence or waits for the data to be re-transmitted. Then, the connection is terminated.

In contrast, computers use UDP to send short messages to each other. There is no guarantee that the messages arrive in sequence or that the messages arrive at all.

Both TCP and UDP use ports to identify the source and destination. Each port is a 16-bit number. Some port numbers have been standardized and are used by low-level system processes; many others have no particular meaning.

Unlike TCP and UDP, Internet Control Message Protocol (ICMP, IP protocol 1) is mainly used to send error messages or to investigate problems. For example, ICMP is used to send the response if a computer cannot be reached. Another use is ping. ICMP does not guarantee delivery, but networks often treat ICMP messages differently, sometimes looking at the message itself to decide where to send it.

Service Objects and Service Groups

Use service objects to define IP protocols.

- TCP applications
- UDP applications
- ICMP messages
- user-defined services (for other types of IP protocols)

These objects are used in policy routes, firewall rules.

Use service groups when you want to create the same rule for several services, instead of creating separate rules for each service. Service groups may consist of services and other service groups. The sequence of members in the service group is not important.

29.2 The Service Summary Screen

The **Service** summary screen provides a summary of all services and their definitions. In addition, this screen allows you to add, edit, and remove services.

To access this screen, log in to the Web Configurator, and click **Configuration > Object > Service** > **Service**. Click a column's heading cell to sort the table entries by that column's criteria. Click the heading cell again to reverse the sort order.

nfig	uration	
0	Add 📝 Edit 🍵 Remove 📑 Object Re	ferences
#	Name 🔺	Content
1	AH	Protocol=51
2	AIM	TCP=5190
3	AUTH	TCP=113
4	Any_TCP	TCP/1-65535
5	Any_UDP	UDP/1-65535
6	BGP	TCP=179
7	BOOTP_CLIENT	UDP=68
8	BOOTP_SERVER	UDP=67
9	CU_SEEME_TCP1	TCP=7648
10	CU_SEEME_TCP2	TCP=24032
11	CU_SEEME_UDP1	UDP=7648
12	CU_SEEME_UDP2	UDP=24032
13	DNS_TCP	TCP=53
14	DNS_UDP	UDP=53
15	ESP	Protocol=50
16	FINGER	TCP=79
17	FTP	TCP/20-21
18	H323	TCP=1720
19	нттр	TCP=80
20	HTTPS	TCP=443

Figure 256 Configuration > Object > Service > Service

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific service.
Name	This field displays the name of each service.
Content	This field displays a description of each service.

 Table 146
 Configuration > Object > Service > Service

29.2.1 The Service Add/Edit Screen

The **Service Add/Edit** screen allows you to create a new service or edit an existing one. To access this screen, go to the **Service** screen (see Section 29.2 on page 381), and click either the **Add** icon or an **Edit** icon.

тср 🗸
(165535)
(165535)

Figure 257 Configuration > Object > Service > Service > Edit

	Table 147	Configuration	> Object >	Service >	Service > Edit
--	-----------	---------------	------------	-----------	----------------

LABEL	DESCRIPTION
Name	Type the name used to refer to the service. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
IP Protocol	Select the protocol the service uses. Choices are: TCP, UDP, ICMP, ICMPv6, and User Defined.
Starting Port	This field appears if the IP Protocol is TCP or UDP . Specify the port number(s) used by
Ending Port	fields, the service uses the range of ports.
ІСМР Туре	This field appears if the IP Protocol is ICMP or ICMPv6.
	Select the ICMP message used by this service. This field displays the message text, not the message number.
IP Protocol	This field appears if the IP Protocol is User Defined.
Number	Enter the number of the next-level protocol (IP protocol). Allowed values are 1 - 255.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

29.3 The Service Group Summary Screen

The **Service Group** summary screen provides a summary of all service groups. In addition, this screen allows you to add, edit, and remove service groups.

To access this screen, log in to the Web Configurator, and click **Configuration > Object > Service** > **Service Group**.

nfig	uration		
0	Add 📝 Edit 🍵	Remove 📴 Object Reference	
#	Family 🔺	Name	Description
1	•	CU-SEEME	
2	•	DHCP	
3	•	DHCPv6	
4	•	DNS	
5	•	Default_Allow_DMZ_To_ZyWALL	System Default Allow From DMZ To ZyWALL
6	•	Default_Allow_WAN_To_ZyWALL	System Default Allow From WAN To ZyWALL
7		Default_Allow_WLAN_To_ZyWALL	System Default Allow From WLAN To ZyWALL
8	•	Default_Allow_any_to_ZyWALL	System Default Allow From any To ZyWALL
9		Default_Allow_v6_DMZ_To_ZyWALL	System Default Allow IPv6 From DMZ to ZyWALL
10		Default_Allow_v6_WAN_To_ZyWALL	System Default Allow IPv6 Form WAN To ZyWALL
11		Default_Allow_v6_WLAN_To_ZyWALL	System Default Allow IPv6 From WLAN To ZyWALL
12	•	IRC	
13		NetBIOS	
14	•	ROADRUNNER	
15		RTSP	
16		SNMP	
17		SNMP-TRAPS	
18		SSH	
19		Default_Allow_v6_any_to_ZyWALL	System Default Allow IPv6 From any To ZyWALL
20		Default_Allow_ICMPv6_Group	Default Allow icmpv6 to ZyWALL

Figure 258 Configuration > Object > Service > Service Group

The following table describes the labels in this screen. See Section 29.3.1 on page 384 for more information as well.

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific service group.

 Table 148
 Configuration > Object > Service > Service Group

LABEL	DESCRIPTION
Family	This field displays the Server Group supported type, which is according to your configurations in the Service Group Add/Edit screen.
	There are 3 types of families:
	 Supports IPv4 only Ref: Supports IPv6 only Ref: Supports both IPv4 and IPv6
Name	This field displays the name of each service group.
	By default, the ZyWALL uses services starting with "Default_Allow_" in the firewall rules to allow certain services to connect to the ZyWALL.
Description	This field displays the description of each service group, if any.

 Table 148
 Configuration > Object > Service > Service Group (continued)

29.3.1 The Service Group Add/Edit Screen

The **Service Group Add/Edit** screen allows you to create a new service group or edit an existing one. To access this screen, go to the **Service Group** screen (see Section 29.3 on page 383), and click either the **Add** icon or an **Edit** icon.

Configuration						
Name:						
Description:						
Member List						
Available			Member		1	
Any UDP	1					
Any_TCP						
АН						
AIM		•				
NEW_ICQ		+				
AUTH						
BGP						
BOOTP_CLIENT						
BOOTP_SERVER	~					

Figure 259 Configuration > Object > Service > Service Group > Edit

The following table describes the labels in this screen.

 Table 149
 Configuration > Object > Service > Service Group > Edit

LABEL	DESCRIPTION
Name	Enter the name of the service group. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Description	Enter a description of the service group, if any. You can use up to 60 printable ASCII characters.

LABEL	DESCRIPTION
Member List	The Member list displays the names of the service and service group objects that have been added to the service group. The order of members is not important.
	Select items from the Available list that you want to be members and move them to the Member list. You can double-click a single entry to move it or use the [Shift] or [Ctrl] key to select multiple entries and use the arrow button to move them.
	Move any members you do not want included to the Available list.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

 Table 149
 Configuration > Object > Service > Service Group > Edit (continued)

Schedules

30.1 Overview

Use schedules to set up one-time and recurring schedules for policy routes, firewall rules. The ZyWALL supports one-time and recurring schedules. One-time schedules are effective only once, while recurring schedules usually repeat. Both types of schedules are based on the current date and time in the ZyWALL.

Note: Schedules are based on the ZyWALL's current date and time.

30.1.1 What You Can Do in this Chapter

- Use the **Schedule** summary screen (Section 30.2 on page 387) to see a list of all schedules in the ZyWALL.
- Use the **One-Time Schedule Add/Edit** screen (Section 30.2.1 on page 388) to create or edit a one-time schedule.
- Use the **Recurring Schedule Add/Edit** screen (Section 30.2.2 on page 389) to create or edit a recurring schedule.

30.1.2 What You Need to Know

One-time Schedules

One-time schedules begin on a specific start date and time and end on a specific stop date and time. One-time schedules are useful for long holidays and vacation periods.

Recurring Schedules

Recurring schedules begin at a specific start time and end at a specific stop time on selected days of the week (Sunday, Monday, Tuesday, Wednesday, Thursday, Friday, and Saturday). Recurring schedules always begin and end in the same day. Recurring schedules are useful for defining the workday and off-work hours.

Finding Out More

• See Section 37.4 on page 434 for information about the ZyWALL's current date and time.

30.2 The Schedule Summary Screen

The **Schedule** summary screen provides a summary of all schedules in the ZyWALL. To access this screen, click **Configuration > Object > Schedule**.

Figure 260 Configuration > Object > Schedule

🔕 Add 📝 Edit 🍵 Rem	nove 🔚 Object References		
# Name	Start Day/Time	Stop Day/Time	
			25 135
4 Page 1 of 1	▶ ▶ Show 50 v items	No data	a to displa [.]
Image 1 of 1 curring Add Image Edit Edit Image	V Show 50 v items	No data	a to displa
Page 1 of 1 of 1 curring Add Add Rem Knme	Vi Show 50 vitems	No data Stop Time	a to displa

The following table describes the labels in this screen. See Section 30.2.1 on page 388 and Section 30.2.2 on page 389 for more information as well.

LABEL	DESCRIPTION
One Time	
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific schedule.
Name	This field displays the name of the schedule, which is used to refer to the schedule.
Start Day / Time	This field displays the date and time at which the schedule begins.
Stop Day / Time	This field displays the date and time at which the schedule ends.
Recurring	
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific schedule.
Name	This field displays the name of the schedule, which is used to refer to the schedule.
Start Time	This field displays the time at which the schedule begins.
Stop Time	This field displays the time at which the schedule ends.

Table 150 Configuration > Object > Schedule

30.2.1 The One-Time Schedule Add/Edit Screen

The **One-Time Schedule Add/Edit** screen allows you to define a one-time schedule or edit an existing one. To access this screen, go to the **Schedule** screen (see Section 30.2 on page 387), and click either the **Add** icon or an **Edit** icon in the **One Time** section.

Add Schedule One	: Time Rule ?	×
Configuration		
Name:		
Day Time		
StartDate:		
StartTime:		
StopDate:		
StopTime:		
)

Figure 261 Configuration > Object > Schedule > Edit (One Time)

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Configuration	
Name	Type the name used to refer to the one-time schedule. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Date Time	
StartDate	 Specify the year, month, and day when the schedule begins. Year - 1900 - 2999 Month - 1 - 12 Day - 1 - 31 (it is not possible to specify illegal dates, such as February 31.)
StartTime	 Specify the hour and minute when the schedule begins. Hour - 0 - 23 Minute - 0 - 59
StopDate	 Specify the year, month, and day when the schedule ends. Year - 1900 - 2999 Month - 1 - 12 Day - 1 - 31 (it is not possible to specify illegal dates, such as February 31.)
StopTime	 Specify the hour and minute when the schedule ends. Hour - 0 - 23 Minute - 0 - 59
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

Table 151 Configuration > Object > Schedule > Edit (One Time)

30.2.2 The Recurring Schedule Add/Edit Screen

The **Recurring Schedule Add/Edit** screen allows you to define a recurring schedule or edit an existing one. To access this screen, go to the **Schedule** screen (see Section 30.2 on page 387), and click either the **Add** icon or an **Edit** icon in the **Recurring** section.

au scheuule M	ecurring Rule				? ×
onfiguration					
Name:					
Day Time					
StartTime: StopTime:					
Weekly					
Week Days:	🔽 Monday 📝 Friday	TuesdaySaturday	✓ Wednesday✓ Sunday	🔽 Thursday	

The **Year**, **Month**, and **Day** columns are not used in recurring schedules and are disabled in this screen. The following table describes the remaining labels in this screen.

LABEL	DESCRIPTION
Configuration	
Name	Type the name used to refer to the recurring schedule. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Date Time	
StartTime	Specify the hour and minute when the schedule begins each day.
	• Hour - 0 - 23
	• Minute - 0 - 59
StopTime	Specify the hour and minute when the schedule ends each day.
	• Hour - 0 - 23
	• Minute - 0 - 59
Weekly	
Week Days	Select each day of the week the recurring schedule is effective.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

Table 152	Configuration >	Object >	Schedule >	Edit ((Recurring)
-----------	-----------------	----------	------------	--------	-------------

AAA Server

31.1 Overview

You can use a AAA (Authentication, Authorization, Accounting) server to provide access control to your network. The AAA server can be a Active Directory, LDAP, or RADIUS server. Use the **AAA Server** screens to create and manage objects that contain settings for using AAA servers. You use AAA server objects in configuring ext-group-user user objects and authentication method objects (see Chapter 32 on page 399).

31.1.1 Directory Service (AD/LDAP)

LDAP/AD allows a client (the ZyWALL) to connect to a server to retrieve information from a directory. A network example is shown next.



Figure 263 Example: Directory Service Client and Server

The following describes the user authentication procedure via an LDAP/AD server.

- 1 A user logs in with a user name and password pair.
- 2 The ZyWALL tries to bind (or log in) to the LDAP/AD server.
- **3** When the binding process is successful, the ZyWALL checks the user information in the directory against the user name and password pair.
- 4 If it matches, the user is allowed access. Otherwise, access is blocked.

31.1.2 RADIUS Server

RADIUS (Remote Authentication Dial-In User Service) authentication is a popular protocol used to authenticate users by means of an external server instead of (or in addition to) an internal device user database that is limited to the memory capacity of the device. In essence, RADIUS authentication allows you to validate a large number of users from a central location.





31.1.3 ASAS

ASAS (Authenex Strong Authentication System) is a RADIUS server that works with the One-Time Password (OTP) feature. Purchase a ZyWALL OTP package in order to use this feature. The package contains server software and physical OTP tokens (PIN generators). Do the following to use OTP. See the documentation included on the ASAS' CD for details.

- 1 Install the ASAS server software on a computer.
- 2 Create user accounts on the ZyWALL and in the ASAS server.
- 3 Import each token's database file (located on the included CD) into the server.
- 4 Assign users to OTP tokens (on the ASAS server).
- **5** Configure the ASAS as a RADIUS server in the ZyWALL's **Configuration > Object > AAA Server** screens.
- 6 Give the OTP tokens to (local or remote) users.

31.1.4 What You Can Do in this Chapter

- Use the **Configuration > Object > AAA Server > Active Directory** (or **LDAP**) screens (Section 31.2 on page 393) to configure Active Directory or LDAP server objects.
- Use the **Configuration > Object > AAA Server > RADIUS** screen (Section 31.3 on page 396) to configure the default external RADIUS server to use for user authentication.

31.1.5 What You Need To Know

AAA Servers Supported by the ZyWALL

The following lists the types of authentication server the ZyWALL supports.

Local user database

The ZyWALL uses the built-in local user database to authenticate administrative users logging into the ZyWALL's Web Configurator or network access users logging into the network through the ZyWALL. You can also use the local user database to authenticate VPN users.

Directory Service (LDAP/AD)

LDAP (Lightweight Directory Access Protocol)/AD (Active Directory) is a directory service that is both a directory and a protocol for controlling access to a network. The directory consists of a database specialized for fast information retrieval and filtering activities. You create and store user profile and login information on the external server.

RADIUS

RADIUS (Remote Authentication Dial-In User Service) authentication is a popular protocol used to authenticate users by means of an external or built-in RADIUS server. RADIUS authentication allows you to validate a large number of users from a central location.

Directory Structure

The directory entries are arranged in a hierarchical order much like a tree structure. Normally, the directory structure reflects the geographical or organizational boundaries. The following figure shows a basic directory structure branching from countries to organizations to organizational units to individuals.

Figure 265 Basic Directory Structure



Distinguished Name (DN)

A DN uniquely identifies an entry in a directory. A DN consists of attribute-value pairs separated by commas. The leftmost attribute is the Relative Distinguished Name (RDN). This provides a unique name for entries that have the same "parent DN" ("cn=domain1.com, ou=Sales, o=MyCompany" in the following examples).

cn=domain1.com, ou = Sales, o=MyCompany, c=US
cn=domain1.com, ou = Sales, o=MyCompany, c=JP

Base DN

A base DN specifies a directory. A base DN usually contains information such as the name of an organization, a domain name and/or country. For example, o=MyCompany, c=UK where o means organization and c means country.

Bind DN

A bind DN is used to authenticate with an LDAP/AD server. For example a bind DN of cn=zywallAdmin allows the ZyWALL to log into the LDAP/AD server using the user name of zywallAdmin. The bind DN is used in conjunction with a bind password. When a bind DN is not specified, the ZyWALL will try to log in as an anonymous user. If the bind password is incorrect, the login will fail.

31.2 Active Directory or LDAP Server Summary

Use the **Active Directory** or **LDAP** screen to manage the list of AD or LDAP servers the ZyWALL can use in authenticating users.

Click Configuration > Object > AAA Server > Active Directory (or LDAP) to display the Active Directory (or LDAP) screen.

Figure 266 Configuration > Object > AAA Server > Active Directory (or LDAP)

tive	Directory	LDAP	RADIUS			
) Ser	ver Summai	ry				
0	Add 📝 Edit	TRemo	ve 🔚 Object R	leferences		
#	Name			Server Address	Base DN	
	ad					
2	Test			1.2.3.4	1.1.1.1	
14	4 Page 1	of 1	▶ ▶ Show	50 🔽 items	1	Displaying 1 - 2 of 2

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field displays the index number.
Server Address	This is the address of the AD or LDAP server.
Base DN	This specifies a directory. For example, o=ZyXEL, c=US.

Table 153 Configuration > Object > AAA Server > Active Directory (or LDAP)

31.2.1 Adding an Active Directory or LDAP Server

Click **Object** > **AAA Server** > **Active Directory** (or **LDAP**) to display the **Active Directory** (or **LDAP**) screen. Click the **Add** icon or an **Edit** icon to display the following screen. Use this screen to create a new AD or LDAP entry or edit an existing one.

General Settings		
Name:	New	
Description:		(Optional)
Server Settings		
Server Address:		💭 💷 or FQDN)
Backup Server Address:		(IP or FQDN)(Optional)
Port:	389	(1-65535)
Base DN:		
Use SSL		
Search time limit:	5	(1-300 seconds)
Case-sensitive User Names 🔋		
erver Authentication		
Bind DN:		
Password:		
Retype to Confirm:		
Ison Login Cattings	L	
Legin Name Attributes		
Login Name Attribute:	SAMACCOUNTIVAME	
Alternative Login Name Attribute:		(Optional)
Group Membership Attribute:	memberOf	
omain Authentication for MSChap		
Enable		
User Name:		Must be a user who has rights to add a machine to the domain.
User Password:		
Retype to Confirm:		
Realm:		
Oligonation validation	the configured comes to	ildete ale sue actione
lisername:	the configured server to va	Tech
Oserhame.		

Figure 267 Configuration > Object > AAA Server > Active Directory (or LDAP) > Add

LABEL	DESCRIPTION
Name	Enter a descriptive name (up to 63 alphanumerical characters) for identification purposes.
Description	Enter the description of each server, if any. You can use up to 60 printable ASCII characters.
Server Address	Enter the address of the AD or LDAP server.
Backup Server Address	If the AD or LDAP server has a backup server, enter its address here.
Port	Specify the port number on the AD or LDAP server to which the ZyWALL sends authentication requests. Enter a number between 1 and 65535.
	This port number should be the same on all AD or LDAP server(s) in this group.

 Table 154
 Configuration > Object > AAA Server > Active Directory (or LDAP) > Add

LABEL	DESCRIPTION
Base DN	Specify the directory (up to 127 alphanumerical characters). For example, $o=ZyXEL$, $c=US$.
	This is only for LDAP.
Use SSL	Select Use SSL to establish a secure connection to the AD or LDAP server(s).
Search time limit	Specify the timeout period (between 1 and 300 seconds) before the ZyWALL disconnects from the AD or LDAP server. In this case, user authentication fails.
	Search timeout occurs when either the user information is not in the AD or LDAP server(s) or the AD or LDAP server(s) is down.
Case-sensitive User Names	Select this if the server checks the case of the usernames.
Bind DN	Specify the bind DN for logging into the AD or LDAP server. Enter up to 127 alphanumerical characters.
	For example, cn=zywallAdmin specifies zywallAdmin as the user name.
Password	If required, enter the password (up to 15 alphanumerical characters) for the ZyWALL to bind (or log in) to the AD or LDAP server.
Retype to Confirm	Retype your new password for confirmation.
Login Name Attribute	Enter the type of identifier the users are to use to log in. For example "name" or "e-mail address".
Alternative Login Name Attribute	If there is a second type of identifier that the users can use to log in, enter it here. For example "name" or "e-mail address".
Group Membership Attribute	An AD or LDAP server defines attributes for its accounts. Enter the name of the attribute that the ZyWALL is to check to determine to which group a user belongs. The value for this attribute is called a group identifier; it determines to which group a user belongs. You can add ext-group-user user objects to identify groups based on these group identifier values.
	For example you could have an attribute named "memberOf" with values like "sales", "RD", and "management". Then you could also create a ext-group-user user object for each group. One with "sales" as the group identifier, another for "RD" and a third for "management".
Domain	Select the Enable checkbox to enable domain authentication for MSChap.
Authentication for MSChap	This is only for Active Directory.
User Name	Enter the user name for the user who has rights to add a machine to the domain.
	This is only for Active Directory.
User Password	Enter the password for the associated user name.
	This is only for Active Directory.
Retype to Confirm	Retype your new password for confirmation.
	This is only for Active Directory.
Realm	Enter the realm FQDN.
	This is only for Active Directory.
Configuration Validation	Use a user account from the server specified above to test if the configuration is correct. Enter the account's user name in the Username field and click Test .
ОК	Click OK to save the changes.
Cancel	Click Cancel to discard the changes.

Table 154	Configuration :	> Object >	AAA Server	> Active I	Directory (d	or LDAP)	> Add	(continued)
-----------	-----------------	------------	------------	------------	--------------	----------	-------	-------------

31.3 RADIUS Server Summary

Use the **RADIUS** screen to manage the list of RADIUS servers the ZyWALL can use in authenticating users.

Click Configuration > Object > AAA Server > RADIUS to display the RADIUS screen.

Figure 268 Configuration > Object > AAA Server > RADIUS

tive Di	irectory LDAP RADIUS		
adius	Server Summary		
O F	Add 📝 Edit 🎁 Remove 📧 Object References		
#	Name	Server Address	
1	radius		

The following table describes the labels in this screen.

LABEL	DESCRIPTION			
Add	Click this to create a new entry.			
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.			
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.			
Object References	Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.			
#	This field displays the index number.			
Name	This is the name of the RADIUS server entry.			
Server Address	This is the address of the AD or LDAP server.			
Apply	Click Apply to save the changes.			
Reset	Click Reset to return the screen to its last-saved settings.			

 Table 155
 Configuration > Object > AAA Server > RADIUS

31.3.1 Adding a RADIUS Server

Click **Configuration > Object > AAA Server > RADIUS** to display the **RADIUS** screen. Click the **Add** icon or an **Edit** icon to display the following screen. Use this screen to create a new AD or LDAP entry or edit an existing one.
Name:	New	
Description:		(Optional)
ierver Settings		
Server Address:		In FQDN)
Authentication Port:	1812	(1-65535)
Backup Server Address:		(IP or FQDN) (Optional)
Backup Authentication Port:		(1-65535) (Optional)
Timeout:	5	(1-300 seconds)
Case-sensitive User Names		
Key:		
Jser Login Settings		
Group Membership Attribute:	Filter - Id(11)	▼ 11

Figure 269 Configuration > Object > AAA Server > RADIUS > Add

LABEL	DESCRIPTION
Name	Enter a descriptive name (up to 63 alphanumerical characters) for identification purposes.
Description	Enter the description of each server, if any. You can use up to 60 printable ASCII characters.
Server Address	Enter the address of the RADIUS server.
Authentication Port	Specify the port number on the RADIUS server to which the ZyWALL sends authentication requests. Enter a number between 1 and 65535.
Backup Server Address	If the RADIUS server has a backup server, enter its address here.
Backup Authentication Port	Specify the port number on the RADIUS server to which the ZyWALL sends authentication requests. Enter a number between 1 and 65535.
Timeout	Specify the timeout period (between 1 and 300 seconds) before the ZyWALL disconnects from the RADIUS server. In this case, user authentication fails.
	Search timeout occurs when either the user information is not in the RADIUS server or the RADIUS server is down.
Case-sensitive User Names	Select this if you want configure your username as case-sensitive.
Кеу	Enter a password (up to 15 alphanumeric characters) as the key to be shared between the external authentication server and the ZyWALL.
	The key is not sent over the network. This key must be the same on the external authentication server and the ZyWALL.

Table 156 Configuration > Object > AAA Server > RADIUS > Add

LABEL	DESCRIPTION
Group Membership Attribute	A RADIUS server defines attributes for its accounts. Select the name and number of the attribute that the ZyWALL is to check to determine to which group a user belongs. If it does not display, select user-defined and specify the attribute's number.
	This attribute's value is called a group identifier; it determines to which group a user belongs. You can add ext-group-user user objects to identify groups based on these group identifier values.
	For example you could have an attribute named "memberOf" with values like "sales", "RD", and "management". Then you could also create a ext-group-user user object for each group. One with "sales" as the group identifier, another for "RD" and a third for "management".
ОК	Click OK to save the changes.
Cancel	Click Cancel to discard the changes.

 Table 156
 Configuration > Object > AAA Server > RADIUS > Add (continued)

Authentication Method

32.1 Overview

Authentication method objects set how the ZyWALL authenticates wireless, HTTP/HTTPS clients, and peer IPSec routers (extended authentication) clients. Configure authentication method objects to have the ZyWALL use the local user database, and/or the authentication servers and authentication server groups specified by AAA server objects. By default, user accounts created and stored on the ZyWALL are authenticated locally.

32.1.1 What You Can Do in this Chapter

• Use the **Configuration > Object > Auth. Method** screens (Section 32.2 on page 400) to create and manage authentication method objects.

Finding Out More

32.1.2 Before You Begin

Configure AAA server objects (see Chapter 31 on page 390) before you configure authentication method objects.

32.1.3 Example: Selecting a VPN Authentication Method

After you set up an authentication method object in the **Auth. Method** screens, you can use it in the **VPN Gateway** screen to authenticate VPN users for establishing a VPN connection. Refer to the chapter on VPN for more information.

Follow the steps below to specify the authentication method for a VPN connection.

- 1 Access the Configuration > VPN > IPSec VPN > VPN Gateway > Edit screen.
- 2 Click Show Advance Setting and select Enable Extended Authentication.
- 3 Select Server Mode and select an authentication method object from the drop-down list box.
- 4 Click **OK** to save the settings.

Figure 270 Example: Using Authentication Method in VPN

📝 NAT Traversal			
👿 Dead Peer Detection (DPD)			
Extended Authentication			
Enable Extended Authentica	tion		
Server Mode	default	·	
Client Mode			
User Name:			
Password:			
			OK Cancel

32.2 Authentication Method Objects

Click **Configuration > Object > Auth. Method** to display the screen as shown.

Note: You can create up to 16 authentication method objects.

Figure 271	Configuration	> Object >	Auth. Method

uther	ntication Method		
onfig	uration		
0	Add 📝 Edit 🍵 Remove 🗓	Object References	
#	Method Name	Method List	
1	default	local	
14	↓ Page 1 of 1 ▶ ▶	Show 50 🗸 items	Displaying 1 - 1 of

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field displays the index number.
Method Name	This field displays a descriptive name for identification purposes.
Method List	This field displays the authentication method(s) for this entry.

 Table 157
 Configuration > Object > Auth. Method

32.2.1 Creating an Authentication Method Object

Follow the steps below to create an authentication method object.

1 Click Configuration > Object > Auth. Method.

- 2 Click Add.
- Specify a descriptive name for identification purposes in the Name field. You may use 1-31 3 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive. For example, "My_Device".
- 4 Click **Add** to insert an authentication method in the table.
- 5 Select a server object from the Method List drop-down list box.
- 6 You can add up to four server objects to the table. The ordering of the Method List column is important. The ZyWALL authenticates the users using the databases (in the local user database or the external authentication server) in the order they appear in this screen.

If two accounts with the same username exist on two authentication servers you specify, the ZyWALL does not continue the search on the second authentication server when you enter the username and password that doesn't match the one on the first authentication server.

Note: You can NOT select two server objects of the same type.

7 Click **OK** to save the settings or click **Cancel** to discard all changes and return to the previous screen.

Figure	272	Con	figura	ation	> Ok	ojec [.]	t > A	uth	Me	thoo	>	Ado
🔾 Add Au Genera	ithentio	ation № gs	lethod	1941115		1 (94)	lianc			? ×		
Name	: udd 🖃	Edit 1	Remov		0					1		
#	Method	List	TRUNET									
2	group	ldap										
							OK		Cano	el		

LABEL	DESCRIPTION
Name	Specify a descriptive name for identification purposes.
	You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive. For example, "My_Device".
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.

Table 158	Configuration >	Object >	Auth.	Method	> Add

LABEL	DESCRIPTION
Move	To change a method's position in the numbered list, select the method and click Move to display a field to type a number for where you want to put it and press [ENTER] to move the rule to the number that you typed.
	The ordering of your methods is important as ZyWALL authenticates the users using the authentication methods in the order they appear in this screen.
#	This field displays the index number.
Method List	Select a server object from the drop-down list box. You can create a server object in the AAA Server screen (see Chapter 31 on page 390 for more information).
	The ZyWALL authenticates the users using the databases (in the local user database or the external authentication server) in the order they appear in this screen.
	If two accounts with the same username exist on two authentication servers you specify, the ZyWALL does not continue the search on the second authentication server when you enter the username and password that doesn't match the one on the first authentication server.
ОК	Click OK to save the changes.
Cancel	Click Cancel to discard the changes.

Table 158 Configuration > Object > Auth. Method > Add (continued)

Certificates

33.1 Overview

The ZyWALL can use certificates (also called digital IDs) to authenticate users. Certificates are based on public-private key pairs. A certificate contains the certificate owner's identity and public key. Certificates provide a way to exchange public keys for use in authentication.

33.1.1 What You Can Do in this Chapter

- Use the **My Certificates** screens (see Section 33.2 on page 406 to Section 33.2.3 on page 412) to generate and export self-signed certificates or certification requests and import the CA-signed certificates.
- Use the **Trusted Certificates** screens (see Section 33.3 on page 413 to Section 33.3.2 on page 417) to save CA certificates and trusted remote host certificates to the ZyWALL. The ZyWALL trusts any valid certificate that you have imported as a trusted certificate. It also trusts any valid certificate signed by any of the certificates that you have imported as a trusted certificate.

33.1.2 What You Need to Know

When using public-key cryptology for authentication, each host has two keys. One key is public and can be made openly available. The other key is private and must be kept secure.

These keys work like a handwritten signature (in fact, certificates are often referred to as "digital signatures"). Only you can write your signature exactly as it should look. When people know what your signature looks like, they can verify whether something was signed by you, or by someone else. In the same way, your private key "writes" your digital signature and your public key allows people to verify whether data was signed by you, or by someone else. This process works as follows.

- 1 Tim wants to send a message to Jenny. He needs her to be sure that it comes from him, and that the message content has not been altered by anyone else along the way. Tim generates a public key pair (one public key and one private key).
- 2 Tim keeps the private key and makes the public key openly available. This means that anyone who receives a message seeming to come from Tim can read it and verify whether it is really from him or not.
- 3 Tim uses his private key to sign the message and sends it to Jenny.
- 4 Jenny receives the message and uses Tim's public key to verify it. Jenny knows that the message is from Tim, and that although other people may have been able to read the message, no-one can have altered it (because they cannot re-sign the message with Tim's private key).

5 Additionally, Jenny uses her own private key to sign a message and Tim uses Jenny's public key to verify the message.

The ZyWALL uses certificates based on public-key cryptology to authenticate users attempting to establish a connection, not to encrypt the data that you send after establishing a connection. The method used to secure the data that you send through an established connection depends on the type of connection. For example, a VPN tunnel might use the triple DES encryption algorithm.

The certification authority uses its private key to sign certificates. Anyone can then use the certification authority's public key to verify the certificates.

A certification path is the hierarchy of certification authority certificates that validate a certificate. The ZyWALL does not trust a certificate if any certificate on its path has expired or been revoked.

Certification authorities maintain directory servers with databases of valid and revoked certificates. A directory of certificates that have been revoked before the scheduled expiration is called a CRL (Certificate Revocation List). The ZyWALL can check a peer's certificate against a directory server's list of revoked certificates. The framework of servers, software, procedures and policies that handles keys is called PKI (public-key infrastructure).

Advantages of Certificates

Certificates offer the following benefits.

- The ZyWALL only has to store the certificates of the certification authorities that you decide to trust, no matter how many devices you need to authenticate.
- Key distribution is simple and very secure since you can freely distribute public keys and you never need to transmit private keys.

Self-signed Certificates

You can have the ZyWALL act as a certification authority and sign its own certificates.

Factory Default Certificate

The ZyWALL generates its own unique self-signed certificate when you first turn it on. This certificate is referred to in the GUI as the factory default certificate.

Certificate File Formats

Any certificate that you want to import has to be in one of these file formats:

- Binary X.509: This is an ITU-T recommendation that defines the formats for X.509 certificates.
- PEM (Base-64) encoded X.509: This Privacy Enhanced Mail format uses lowercase letters, uppercase letters and numerals to convert a binary X.509 certificate into a printable form.
- Binary PKCS#7: This is a standard that defines the general syntax for data (including digital signatures) that may be encrypted. A PKCS #7 file is used to transfer a public key certificate. The private key is not included. The ZyWALL currently allows the importation of a PKS#7 file that contains a single certificate.
- PEM (Base-64) encoded PKCS#7: This Privacy Enhanced Mail (PEM) format uses lowercase letters, uppercase letters and numerals to convert a binary PKCS#7 certificate into a printable form.

- Binary PKCS#12: This is a format for transferring public key and private key certificates. The private key in a PKCS #12 file is within a password-encrypted envelope. The file's password is not connected to your certificate's public or private passwords. Exporting a PKCS #12 file creates this and you must provide it to decrypt the contents when you import the file into the ZyWALL.
- Note: Be careful not to convert a binary file to text during the transfer process. It is easy for this to occur since many programs use text files by default.

Finding Out More

• See Section 33.4 on page 418 for certificate background information.

33.1.3 Verifying a Certificate

Before you import a trusted certificate into the ZyWALL, you should verify that you have the correct certificate. You can do this using the certificate's fingerprint. A certificate's fingerprint is a message digest calculated using the MD5 or SHA1 algorithm. The following procedure describes how to check a certificate's fingerprint to verify that you have the actual certificate.

- 1 Browse to where you have the certificate saved on your computer.
- 2 Make sure that the certificate has a ".cer" or ".crt" file name extension.

Figure 273 Remote Host Certificates

	Elondon-Office.cer
Remote Host Certificates	

3 Double-click the certificate's icon to open the **Certificate** window. Click the **Details** tab and scroll down to the **Thumbprint Algorithm** and **Thumbprint** fields.

	×
Field	Value
E Subject	Glenn
Public key	RSA (1024 Bits)
Key Usage	Digital Signature , Certificate Signing(
Subject Alternative Name	DNS Name=Glenn
Basic Constraints	Subject Type=CA, Path Length Cons
Thumbprint algorithm	sha1
Thumbprint	BUA7 2286 7960 FF92 52F4 6B4C A2 >

Figure 274 Certificate Details

Use a secure method to verify that the certificate owner has the same information in the 4 Thumbprint Algorithm and Thumbprint fields. The secure method may very based on your situation. Possible examples would be over the telephone or through an HTTPS connection.

33.2 The My Certificates Screen

Click Configuration > Object > Certificate > My Certificates to open the My Certificates screen. This is the ZyWALL's summary list of certificates and certification requests.

Ty Cer		musted C	eruncaces				
KI Sto	orage Space	in Use					
					1.113% used		
		2020 1					
1y Cer	rtificates Se	tting					
0	Add 📝 Edit	💼 Remo	ve 🔚 Object	Referer	ice		
#	Name 🔺	Туре	Subject		Issuer	Valid From	Valid To
1	default	SELF	CN=usg50_0	000AA7	CN=usg50_0000AA7	2009-12-24 14:44:05	2029-12-19 14:44:05
14	🔹 🛛 Page 🚺	of 1	🕨 🕅 Show	50	 items 		Displaying 1 - 1 of
	7 S. (2) 2						

Figure 275 Configuration & Object & Contificate & My Contificates

LABEL	DESCRIPTION
PKI Storage Space in Use	This bar displays the percentage of the ZyWALL's PKI storage space that is currently in use. When the storage space is almost full, you should consider deleting expired or unnecessary certificates before adding more certificates.
Add	Click this to go to the screen where you can have the ZyWALL generate a certificate or a certification request.
Edit	Double-click an entry or select it and click Edit to open a screen with an in-depth list of information about the certificate.
Remove	The ZyWALL keeps all of your certificates unless you specifically delete them. Uploading a new firmware or default configuration file does not delete your certificates. To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Subsequent certificates move up by one when you take this action.
Object References	You cannot delete certificates that any of the ZyWALL's features are configured to use. Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field displays the certificate index number. The certificates are listed in alphabetical order.
Name	This field displays the name used to identify this certificate. It is recommended that you give each certificate a unique name.
Туре	This field displays what kind of certificate this is.
	REQ represents a certification request and is not yet a valid certificate. Send a certification request to a certification authority, which then issues a certificate. Use the My Certificate Import screen to import the certificate and replace the request.
	SELF represents a self-signed certificate.
	CERT represents a certificate issued by a certification authority.
Subject	This field displays identifying information about the certificate's owner, such as CN (Common Name), OU (Organizational Unit or department), O (Organization or company) and C (Country). It is recommended that each certificate have unique subject information.
Issuer	This field displays identifying information about the certificate's issuing certification authority, such as a common name, organizational unit or department, organization or company and country. With self-signed certificates, this is the same information as in the Subject field.
Valid From	This field displays the date that the certificate becomes applicable.
Valid To	This field displays the date that the certificate expires. The text displays in red and includes an Expired! message if the certificate has expired.
Import	Click Import to open a screen where you can save a certificate to the ZyWALL.
Refresh	Click Refresh to display the current validity status of the certificates.

Table 159 Configuration > Object > Certificate > My Certificates

33.2.1 The My Certificates Add Screen

Click **Configuration > Object > Certificate > My Certificates** and then the **Add** icon to open the **My Certificates Add** screen. Use this screen to have the ZyWALL create a self-signed certificate, enroll a certificate with a certification authority or generate a certification request.

)
)
ts
) Is

Figure 276 Configuration > Object > Certificate > My Certificates > Add

LABEL	DESCRIPTION
Name	Type a name to identify this certificate. You can use up to 31 alphanumeric and ;'~ $!@#$ %^&()_+[]{}',.=- characters.
Subject Information	Use these fields to record information that identifies the owner of the certificate. You do not have to fill in every field, although you must specify a Host IP Address , Host Domain Name , or E-Mail . The certification authority may add fields (such as a serial number) to the subject information when it issues a certificate. It is recommended that each certificate have unique subject information.
	Select a radio button to identify the certificate's owner by IP address, domain name or e-mail address. Type the IP address (in dotted decimal notation), domain name or e-mail address in the field provided. The domain name or e-mail address is for identification purposes only and can be any string.
	A domain name can be up to 255 characters. You can use alphanumeric characters, the hyphen and periods.
	An e-mail address can be up to 63 characters. You can use alphanumeric characters, the hyphen, the @ symbol, periods and the underscore.
Organizational Unit	Identify the organizational unit or department to which the certificate owner belongs. You can use up to 31 characters. You can use alphanumeric characters, the hyphen and the underscore.
Organization	Identify the company or group to which the certificate owner belongs. You can use up to 31 characters. You can use alphanumeric characters, the hyphen and the underscore.
Town (City)	Identify the town or city where the certificate owner is located. You can use up to 31 characters. You can use alphanumeric characters, the hyphen and the underscore.
State, (Province)	Identify the state or province where the certificate owner is located. You can use up to 31 characters. You can use alphanumeric characters, the hyphen and the underscore.

LABEL	DESCRIPTION
Country	Identify the nation where the certificate owner is located. You can use up to 31 characters. You can use alphanumeric characters, the hyphen and the underscore.
Кеу Туре	Select RSA to use the Rivest, Shamir and Adleman public-key algorithm.
	Select DSA to use the Digital Signature Algorithm public-key algorithm.
Key Length	Select a number from the drop-down list box to determine how many bits the key should use (512 to 2048). The longer the key, the more secure it is. A longer key also uses more PKI storage space.
Enrollment Options	These radio buttons deal with how and when the certificate is to be generated.
Create a self-signed certificate	Select this to have the ZyWALL generate the certificate and act as the Certification Authority (CA) itself. This way you do not need to apply to a certification authority for certificates.
Create a certification request and save it locally for later manual enrollment	Select this to have the ZyWALL generate and store a request for a certificate. Use the My Certificate Details screen to view the certification request and copy it to send to the certification authority.
	Copy the certification request from the My Certificate Details screen (see Section 33.2.2 on page 409) and then send it to the certification authority.
ОК	Click OK to begin certificate or certification request generation.
Cancel	Click Cancel to quit and return to the My Certificates screen.

 Table 160
 Configuration > Object > Certificate > My Certificates > Add (continued)

If you configured the **My Certificate Create** screen to have the ZyWALL enroll a certificate and the certificate enrollment is not successful, you see a screen with a **Return** button that takes you back to the **My Certificate Create** screen. Click **Return** and check your information in the **My Certificate Create** screen. Make sure that the certification authority information is correct and that your Internet connection is working properly if you want the ZyWALL to enroll a certificate online.

33.2.2 The My Certificates Edit Screen

Click **Configuration > Object > Certificate > My Certificates** and then the **Edit** icon to open the **My Certificate Edit** screen. You can use this screen to view in-depth certificate information and change the certificate's name.

cold My Certificates		12
Name	example	
Name.	example	
ertification Path		
CN=example@example.com		
Refresh		
ertificate Information		
Туре:	Self-signed X.509 Certificate	
Version:	V3	
Serial Number:	1258090745	
Subject:	CN=example@example.com	
Issuer:	CN=example@example.com	
Signature Algorithm:	rsa-pkcs1-sha1	
Valid From:	2009-11-13 05:39:05 GMT	
Valid To:	2012-11-12 05:39:05 GMT	
Key Algorithm:	rsaEncryption (512 bits)	
Subject Alternative Name:	example@example.com	
Key Usage:	DigitalSignature, KeyEncipherment, KeyCertSign	
Basic Constraint:	Subject Type=CA, Path Length Constraint=1	
MD5 Fingerprint:	77;cd;59;cd;35;22;9a;57;8e;c4;b9;1b;1c;b2;e8;3b	
SHA1 Fingerprint:	a5:f3:d4:f0:b2:8d:53:b1:45:41:9e:ff:74:82:1e:e7:37:a0:b0:e3	
ertificate in PEM (Base-64)	Encoded Format	
BEGIN X509 CERTIFICATE- MIIBdjCCASCgAwiBAgIESvzw+ eGFtc5xJQGV4YW1 wbGLU/2991 NowHjEcMBoGA1UEAwwTZXh Export Certificate Only	TANBgkqhkiG9w0BAQUFADAeMRwwGgYDVQQDDBNI MBXDTASMTEXMEALINzkwNVoXDTEyMTEXMjALMzkw hbXbszUBleGFtcGxLmVvbrBMA0GCSqGSIb3DQEB Password:: Export Certificate with Private Key	
	ок	Cancel

Figure 277 Configuration > Object > Certificate > My Certificates > Edit

LABEL	DESCRIPTION
Name	This field displays the identifying name of this certificate. You can use up to 31 alphanumeric and ;' \sim !@#\$%^&()_+[]{}',.=- characters.
Certification Path	This field displays for a certificate, not a certification request.
	Click the Refresh button to have this read-only text box display the hierarchy of certification authorities that validate the certificate (and the certificate itself).
	If the issuing certification authority is one that you have imported as a trusted certification authority, it may be the only certification authority in the list (along with the certificate itself). If the certificate is a self-signed certificate, the certificate itself is the only one in the list. The ZyWALL does not trust the certificate and displays "Not trusted" in this field if any certificate on the path has expired or been revoked.
Refresh	Click Refresh to display the certification path.

LABEL	DESCRIPTION
Certificate Information	These read-only fields display detailed information about the certificate.
Туре	This field displays general information about the certificate. CA-signed means that a Certification Authority signed the certificate. Self-signed means that the certificate's owner signed the certificate (not a certification authority). "X.509" means that this certificate was created and signed according to the ITU-T X.509 recommendation that defines the formats for public-key certificates.
Version	This field displays the X.509 version number.
Serial Number	This field displays the certificate's identification number given by the certification authority or generated by the ZyWALL.
Subject	This field displays information that identifies the owner of the certificate, such as Common Name (CN), Organizational Unit (OU), Organization (O), State (ST), and Country (C).
Issuer	This field displays identifying information about the certificate's issuing certification authority, such as Common Name, Organizational Unit, Organization and Country.
	With self-signed certificates, this is the same as the Subject Name field.
	"none" displays for a certification request.
Signature Algorithm	This field displays the type of algorithm that was used to sign the certificate. The ZyWALL uses rsa-pkcs1-sha1 (RSA public-private key encryption algorithm and the SHA1 hash algorithm). Some certification authorities may use rsa-pkcs1-md5 (RSA public-private key encryption algorithm and the MD5 hash algorithm).
Valid From	This field displays the date that the certificate becomes applicable. "none" displays for a certification request.
Valid To	This field displays the date that the certificate expires. The text displays in red and includes an Expired! message if the certificate has expired. "none" displays for a certification request.
Key Algorithm	This field displays the type of algorithm that was used to generate the certificate's key pair (the ZyWALL uses RSA encryption) and the length of the key set in bits (1024 bits for example).
Subject Alternative Name	This field displays the certificate owner's IP address (IP), domain name (DNS) or e-mail address (EMAIL).
Key Usage	This field displays for what functions the certificate's key can be used. For example, "DigitalSignature" means that the key can be used to sign certificates and "KeyEncipherment" means that the key can be used to encrypt text.
Basic Constraint	This field displays general information about the certificate. For example, Subject Type=CA means that this is a certification authority's certificate and "Path Length Constraint=1" means that there can only be one certification authority in the certificate's path. This field does not display for a certification request.
MD5 Fingerprint	This is the certificate's message digest that the ZyWALL calculated using the MD5 algorithm.
SHA1 Fingerprint	This is the certificate's message digest that the ZyWALL calculated using the SHA1 algorithm.
Certificate in PEM (Base-64) Encoded Format	This read-only text box displays the certificate or certification request in Privacy Enhanced Mail (PEM) format. PEM uses lowercase letters, uppercase letters and numerals to convert a binary certificate into a printable form.
	You can copy and paste a certification request into a certification authority's web page, an e-mail that you send to the certification authority or a text editor and save the file on a management computer for later manual enrollment.
	You can copy and paste a certificate into an e-mail to send to friends or colleagues or you can copy and paste a certificate into a text editor and save the file on a management computer for later distribution (via floppy disk for example).

 Table 161
 Configuration > Object > Certificate > My Certificates > Edit (continued)

LABEL	DESCRIPTION
Export Certificate Only	Use this button to save a copy of the certificate without its private key. Click this button and then Save in the File Download screen. The Save As screen opens, browse to the location that you want to use and click Save .
Password	If you want to export the certificate with its private key, create a password and type it here. Make sure you keep this password in a safe place. You will need to use it if you import the certificate to another device.
Export Certificate with Private Key	Use this button to save a copy of the certificate with its private key. Type the certificate's password and click this button. Click Save in the File Download screen. The Save As screen opens, browse to the location that you want to use and click Save .
ОК	Click OK to save your changes back to the ZyWALL. You can only change the name.
Cancel	Click Cancel to quit and return to the My Certificates screen.

 Table 161
 Configuration > Object > Certificate > My Certificates > Edit (continued)

33.2.3 The My Certificates Import Screen

Click Configuration > Object > Certificate > My Certificates > Import to open the My Certificate Import screen. Follow the instructions in this screen to save an existing certificate to the ZyWALL.

Note: You can import a certificate that matches a corresponding certification request that was generated by the ZyWALL. You can also import a certificate in PKCS#12 format, including the certificate's public and private keys.

The certificate you import replaces the corresponding request in the My Certificates screen.

You must remove any spaces from the certificate's filename before you can import it.

Figure 278 Configuration > Object > Certificate > My Certificates > Import

 Binary PEM (B Binary PEM (B 	X.509 Base-64) encoded X.509 PKCS#7 Base-64) encoded PKCS#	# 7	
- Diridry	FRED#12		
or my certif	icate importation to be s	successful, a certification request correspor	nding to the imported
or my certif ertificate mu utomatically	icate importation to be s ust already exist on ZyW / be deleted.	successful, a certification request correspor /ALL. After the importation, the certificatio	nding to the imported n request will
or my certif ertificate mu utomatically ille Path::	icate importation to be s ist already exist on ZyW be deleted. Select a file path	successful, a certification request correspon /ALL. After the importation, the certificatio	nding to the imported n request will Browse
or my certif ertificate mu utomatically ile Path:: 'assword::	icate importation to be s ust already exist on ZyW v be deleted. Select a file path	ALL: After the importation, the certification (ALL: After the importation, the certification) (PKCS#12 only)	nding to the imported n request will Browse

LABEL	DESCRIPTION
File Path	Type in the location of the file you want to upload in this field or click Browse to find it.
	You cannot import a certificate with the same name as a certificate that is already in the ZyWALL.
Browse	Click Browse to find the certificate file you want to upload.
Password	This field only applies when you import a binary PKCS#12 format file. Type the file's password that was created when the PKCS #12 file was exported.
ОК	Click OK to save the certificate on the ZyWALL.
Cancel	Click Cancel to quit and return to the My Certificates screen.

 Table 162
 Configuration > Object > Certificate > My Certificates > Import

33.3 The Trusted Certificates Screen

Click **Configuration > Object > Certificate > Trusted Certificates** to open the **Trusted Certificates** screen. This screen displays a summary list of certificates that you have set the ZyWALL to accept as trusted. The ZyWALL also accepts any valid certificate signed by a certificate on this list as being trustworthy; thus you do not need to import any certificate that is signed by one of these certificates.

				1.426% use	d		
usteo	d Certificates	Setting					
	-						
E	idit 🔟 Remo	ve 🛛 📲 Object Referenci	95				
#	Name 🔺	Subject	Issuer		Valid From	Valid To	
1	MyCertificate	CN=mydevice@example	CN=my	device@example	2009-03-17 07:11	:25 GN 2012-03-16 07:1	1:25 GN
14	🖣 🛛 Page 🔒	of 1 🕨 🕨 Show	50	🕶 items		Displaying	1 - 1 of 1

Figure 279 Configuration > Object > Certificate > Trusted Certificates

Table 163	Configuration >	Object >	> Certificate >	Trusted	Certificates
-----------	-----------------	----------	-----------------	---------	--------------

LABEL	DESCRIPTION
PKI Storage Space in Use	This bar displays the percentage of the ZyWALL's PKI storage space that is currently in use. When the storage space is almost full, you should consider deleting expired or unnecessary certificates before adding more certificates.
Edit	Double-click an entry or select it and click Edit to open a screen with an in-depth list of information about the certificate.
Remove	The ZyWALL keeps all of your certificates unless you specifically delete them. Uploading a new firmware or default configuration file does not delete your certificates. To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Subsequent certificates move up by one when you take this action.

LABEL	DESCRIPTION
Object References	You cannot delete certificates that any of the ZyWALL's features are configured to use. Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field displays the certificate index number. The certificates are listed in alphabetical order.
Name	This field displays the name used to identify this certificate.
Subject	This field displays identifying information about the certificate's owner, such as CN (Common Name), OU (Organizational Unit or department), O (Organization or company) and C (Country). It is recommended that each certificate have unique subject information.
Issuer	This field displays identifying information about the certificate's issuing certification authority, such as a common name, organizational unit or department, organization or company and country. With self-signed certificates, this is the same information as in the Subject field.
Valid From	This field displays the date that the certificate becomes applicable.
Valid To	This field displays the date that the certificate expires. The text displays in red and includes an Expired! message if the certificate has expired.
Import	Click Import to open a screen where you can save the certificate of a certification authority that you trust, from your computer to the ZyWALL.
Refresh	Click this button to display the current validity status of the certificates.

	Table 163	Configuration >	Object >	Certificate >	Trusted	Certificates	(continued)
--	-----------	-----------------	----------	---------------	---------	--------------	-------------

33.3.1 The Trusted Certificates Edit Screen

Click **Configuration > Object > Certificate > Trusted Certificates** and then a certificate's **Edit** icon to open the **Trusted Certificates Edit** screen. Use this screen to view in-depth information about the certificate, change the certificate's name and set whether or not you want the ZyWALL to check a certification authority's list of revoked certificates before trusting a certificate issued by the certification authority.

Edit Trusted Certificates		?
Configuration		
Name:	MyCertificate.cer	
Certification Path		
CN=mvdevice@example.com		
[Pafrath]		
Kenesn		
Certificate Validation		
Enable X.509v3 CRL Distrib	ution Points and OCSP chedking	
OCSP Server		
URL:		
ID:		
Password:		
LDAP Server		
Address:	Port:	
ID:		
Password:		
ertificate Information		
Туре:	Self-signed X.509 Certificate	
Version:	V3	
Serial Number:	1237273885	
Subject:	CN=mydevice@example.com	
Issuer:	CN=mydevice@example.com	
Signature Algorithm:	rsa-pkcs1-sha1	
Valid From:	2009-03-17 07:11:25 GMT	
Valid To:	2012-03-16 07:11:25 GMT	
Key Algorithm:	rsaEncryption (512 bits)	
Subject Alternative Name:	mydevice@example.com	
Key Usage:	DigitalSignature, KeyEncipherment, KeyCertSign	
Basic Constraint:	Subject Type=CA, Path Length Constraint=1	
MD5 Fingerprint:	78:34(e2:25)e3:79:92:e2:b6:33:5f:a9:17:be:72:4f	
SHA1 Fingerprint:	ef:07:15:b5:e7:22:93:8b:1e:a7:e9:8f:cb:06:4c:04:f4:68:9c:e2	
antification in the second sec		
	F	
MIIBeTCCASOgAwIBAgIESb9 eWRldmlj2UBleGFtcGxlLmNvb MjVaMB8xHTAbBgNVBAMMFG	NHTANBgkqhkiG9w0BAQUFADARMR0wGwYDVQQDDBRt TAeFw0w0TaMTcwhtscMtyJaFw00MjaEwTYwItsEx TS2GV23W10GV4YW1ubGUV23WIFww0QY1KoZThvcN	
Export Certificate		

Figure 280 Configuration > Object > Certificate > Trusted Certificates > Edit

Table 164	Configuration >	Object >	Certificate >	Trusted	Certificates :	>	Edit
-----------	-----------------	----------	---------------	---------	----------------	---	------

LABEL	DESCRIPTION
Name	This field displays the identifying name of this certificate. You can change the name. You can use up to 31 alphanumeric and ;' \sim !@#\$%^&()_+[]{}',.=- characters.
Certification Path	Click the Refresh button to have this read-only text box display the end entity's certificate and a list of certification authority certificates that shows the hierarchy of certification authorities that validate the end entity's certificate. If the issuing certification authority is one that you have imported as a trusted certificate, it may be the only certification authority in the list (along with the end entity's own certificate). The ZyWALL does not trust the end entity's certificate and displays "Not trusted" in this field if any certificate on the path has expired or been revoked.
Refresh	Click Refresh to display the certification path.
Enable X.509v3 CRL Distribution Points and OCSP checking	Select this check box to turn on/off certificate revocation. When it is turned on, the ZyWALL validates a certificate by getting Certificate Revocation List (CRL) through HTTP or LDAP (can be configured after selecting the LDAP Server check box) and online responder (can be configured after selecting the OCSP Server check box).
OCSP Server	Select this check box if the directory server uses OCSP (Online Certificate Status Protocol).
URL	Type the protocol, IP address and path name of the OCSP server.
ID	The ZyWALL may need to authenticate itself in order to assess the OCSP server. Type the login name (up to 31 ASCII characters) from the entity maintaining the server (usually a certification authority).
Password	Type the password (up to 31 ASCII characters) from the entity maintaining the OCSP server (usually a certification authority).
LDAP Server	Select this check box if the directory server uses LDAP (Lightweight Directory Access Protocol). LDAP is a protocol over TCP that specifies how clients access directories of certificates and lists of revoked certificates.
Address	Type the IP address (in dotted decimal notation) of the directory server.
Port	Use this field to specify the LDAP server port number. You must use the same server port number that the directory server uses. 389 is the default server port number for LDAP.
ID	The ZyWALL may need to authenticate itself in order to assess the CRL directory server. Type the login name (up to 31 ASCII characters) from the entity maintaining the server (usually a certification authority).
Password	Type the password (up to 31 ASCII characters) from the entity maintaining the CRL directory server (usually a certification authority).
Certificate Information	These read-only fields display detailed information about the certificate.
Туре	This field displays general information about the certificate. CA-signed means that a Certification Authority signed the certificate. Self-signed means that the certificate's owner signed the certificate (not a certification authority). X.509 means that this certificate was created and signed according to the ITU-T X.509 recommendation that defines the formats for public-key certificates.
Version	This field displays the X.509 version number.
Serial Number	This field displays the certificate's identification number given by the certification authority.
Subject	This field displays information that identifies the owner of the certificate, such as Common Name (CN), Organizational Unit (OU), Organization (O) and Country (C).

LABEL	DESCRIPTION
Issuer	This field displays identifying information about the certificate's issuing certification authority, such as Common Name, Organizational Unit, Organization and Country.
	With self-signed certificates, this is the same information as in the Subject Name field.
Signature Algorithm	This field displays the type of algorithm that was used to sign the certificate. Some certification authorities use rsa-pkcs1-sha1 (RSA public-private key encryption algorithm and the SHA1 hash algorithm). Other certification authorities may use rsa-pkcs1-md5 (RSA public-private key encryption algorithm and the MD5 hash algorithm).
Valid From	This field displays the date that the certificate becomes applicable. The text displays in red and includes a Not Yet Valid! message if the certificate has not yet become applicable.
Valid To	This field displays the date that the certificate expires. The text displays in red and includes an Expiring! or Expired! message if the certificate is about to expire or has already expired.
Key Algorithm	This field displays the type of algorithm that was used to generate the certificate's key pair (the ZyWALL uses RSA encryption) and the length of the key set in bits (1024 bits for example).
Subject Alternative Name	This field displays the certificate's owner's IP address (IP), domain name (DNS) or e- mail address (EMAIL).
Key Usage	This field displays for what functions the certificate's key can be used. For example, "DigitalSignature" means that the key can be used to sign certificates and "KeyEncipherment" means that the key can be used to encrypt text.
Basic Constraint	This field displays general information about the certificate. For example, Subject Type=CA means that this is a certification authority's certificate and "Path Length Constraint=1" means that there can only be one certification authority in the certificate's path.
MD5 Fingerprint	This is the certificate's message digest that the ZyWALL calculated using the MD5 algorithm. You can use this value to verify with the certification authority (over the phone for example) that this is actually their certificate.
SHA1 Fingerprint	This is the certificate's message digest that the ZyWALL calculated using the SHA1 algorithm. You can use this value to verify with the certification authority (over the phone for example) that this is actually their certificate.
Certificate	This read-only text box displays the certificate or certification request in Privacy Enhanced Mail (PEM) format. PEM uses lowercase letters, uppercase letters and numerals to convert a binary certificate into a printable form.
	You can copy and paste the certificate into an e-mail to send to friends or colleagues or you can copy and paste the certificate into a text editor and save the file on a management computer for later distribution (via floppy disk for example).
Export Certificate	Click this button and then Save in the File Download screen. The Save As screen opens, browse to the location that you want to use and click Save .
ОК	Click OK to save your changes back to the ZyWALL. You can only change the name.
Cancel	Click Cancel to quit and return to the Trusted Certificates screen.

 Table 164
 Configuration > Object > Certificate > Trusted Certificates > Edit (continued)

33.3.2 The Trusted Certificates Import Screen

Click Configuration > Object > Certificate > Trusted Certificates > Import to open the Trusted Certificates Import screen. Follow the instructions in this screen to save a trusted certificate to the ZyWALL.

Note: You must remove any spaces from the certificate's filename before you can import the certificate.

Figure 281 Configuration > Object > Certificate > Trusted Certificates > Import

Diama and a		
the following	formats.	e cercificate file must be in one or a
 Binary PEM (I Binary PEM (I 	X.509 Jase-64) encoded X.509 PKC5#7 Jase-64) encoded PKC5#7	
File Path::	Select a file path	Browse

The following table describes the labels in this screen.

	soningulation > Object > Certificate > indiced Certificates > import
LABEL	DESCRIPTION
File Path	Type in the location of the file you want to upload in this field or click Browse to find it. You cannot import a certificate with the same name as a certificate that is already in the ZyWALL.
Browse	Click Browse to find the certificate file you want to upload.
OK	Click OK to save the certificate on the ZyWALL.
Cancel	Click Cancel to quit and return to the previous screen.

Table 165 Configuration > Object > Certificate > Trusted Certificates > Import

33.4 Certificates Technical Reference

OCSP

OCSP (Online Certificate Status Protocol) allows an application or device to check whether a certificate is valid. With OCSP the ZyWALL checks the status of individual certificates instead of downloading a Certificate Revocation List (CRL). OCSP has two main advantages over a CRL. The first is real-time status information. The second is a reduction in network traffic since the ZyWALL only gets information on the certificates that it needs to verify, not a huge list. When the ZyWALL requests certificate status information, the OCSP server returns a "expired", "current" or "unknown" response.

ISP Accounts

34.1 Overview

Use ISP accounts to manage Internet Service Provider (ISP) account information for PPPoE/PPTP interfaces. An ISP account is a profile of settings for Internet access using PPPoE or PPTP.

Finding Out More

• See Section 7.4 on page 125 for information about PPPoE/PPTP interfaces.

34.1.1 What You Can Do in this Chapter

Use the **Object** > **ISP Account** screens (Section 34.2 on page 419) to create and manage ISP accounts in the ZyWALL.

34.2 ISP Account Summary

This screen provides a summary of ISP accounts in the ZyWALL. To access this screen, click **Configuration > Object > ISP Account**.

Figure 282 Configuration > Object > ISP Account

iP Account			
onfiguration			
O Add Z Edit 💼 Remove 📑 Object	References		
# Profile Name A Protocol	Authentication Type	User Name	
1 some-ISP pppoe	chap-pap	test	
Id d Page 1 of 1 b bl Show	50 v items	Displayir	ng 1 - 1 of

The following table describes the labels in this screen. See the ISP Account Edit section below for more information as well.

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific entry.

 Table 166
 Configuration > Object > ISP Account

LABEL	DESCRIPTION
Profile Name	This field displays the profile name of the ISP account. This name is used to identify the ISP account.
Protocol	This field displays the protocol used by the ISP account.
Authentication Type	This field displays the authentication type used by the ISP account.
User Name	This field displays the user name of the ISP account.

 Table 166
 Configuration > Object > ISP Account (continued)

34.2.1 ISP Account Edit

The ISP Account Edit screen lets you add information about new accounts and edit information about existing accounts. To open this window, open the ISP Account screen. (See Section 34.2 on page 419.) Then, click on an Add icon or Edit icon to open the ISP Account Edit screen below.

au ISP Account Rule			() ×
Profile Name:			
Protocol:	рррое	*	
Authentication Type:	Chap/PAP	*	
User Name:			
Password:			
Retype to confirm:			
Service Name:		(Optional)	
Compression	🖲 On 🔘 Off		
Idle timeout:	0	(Seconds)	

LABEL	DESCRIPTION
Profile Name	This field is read-only if you are editing an existing account. Type in the profile name of the ISP account. The profile name is used to refer to the ISP account. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Protocol	This field is read-only if you are editing an existing account. Select the protocol used by the ISP account. Options are:
	pppoe - This ISP account uses the PPPoE protocol.
	pptp - This ISP account uses the PPTP protocol.

Table 167 Configuration > Object > ISP Account > Edit

LABEL	DESCRIPTION
Authentication Type	Use the drop-down list box to select an authentication protocol for outgoing calls. Options are:
	CHAP/PAP - Your ZyWALL accepts either CHAP or PAP when requested by this remote node.
	Chap - Your ZyWALL accepts CHAP only.
	PAP - Your ZyWALL accepts PAP only.
	MSCHAP - Your ZyWALL accepts MSCHAP only.
	MSCHAP-V2 - Your ZyWALL accepts MSCHAP-V2 only.
Encryption Method	This field is available if this ISP account uses the PPTP protocol. Use the drop-down list box to select the type of Microsoft Point-to-Point Encryption (MPPE). Options are:
	nomppe - This ISP account does not use MPPE.
	mppe-40 - This ISP account uses 40-bit MPPE.
	mppe-128 - This ISP account uses 128-bit MMPE.
User Name	Type the user name given to you by your ISP.
Password	Type the password associated with the user name above. The password can only consist of alphanumeric characters (A-Z, a-z, 0-9). This field can be blank.
Retype to Confirm	Type your password again to make sure that you have entered is correctly.
Server IP If this ISP account uses the PPPoE protocol, this field is not displayed.	
	If this ISP account uses the PPTP protocol, type the IP address of the PPTP server.
Connection ID	This field is available if this ISP account uses the PPTP protocol. Type your identification name for the PPTP server. This field can be blank.
Service Name	If this ISP account uses the PPPoE protocol, type the PPPoE service name to access. PPPoE uses the specified service name to identify and reach the PPPoE server. This field can be blank.
	If this ISP account uses the PPTP protocol, this field is not displayed.
Compression	Select On button to turn on stac compression, and select Off to turn off stac compression. Stac compression is a data compression technique capable of compressing data by a factor of about four.
Idle Timeout	This value specifies the number of seconds that must elapse without outbound traffic before the ZyWALL automatically disconnects from the PPPoE/PPTP server. This value must be an integer between 0 and 360. If this value is zero, this timeout is disabled.
ОК	Click OK to save your changes back to the ZyWALL. If there are no errors, the program returns to the ISP Account screen. If there are errors, a message box explains the error, and the program stays in the ISP Account Edit screen.
Cancel	Click Cancel to return to the ISP Account screen without creating the profile (if it is new) or saving any changes to the profile (if it already exists).

 Table 167
 Configuration > Object > ISP Account > Edit (continued)

SSL Application

35.1 Overview

You use SSL application objects in SSL VPN. Configure an SSL application object to specify the type of application and the address of the local computer, server, or web site SSL users are to be able to access. You can apply one or more SSL application objects in the **VPN** > **SSL VPN** screen for a user account/user group.

35.1.1 What You Can Do in this Chapter

- Use the **SSL Application** screen (Section 35.2 on page 424) to view the ZyWALL's configured SSL application objects.
- Use the SSL Application Edit screen to create or edit web-based application objects to allow remote users to access an application via standard web browsers (Section 35.2.1 on page 425).
- You can also use the **SSL Application Edit** screen to specify the name of a folder on a Linux or Windows file server which remote users can access using a standard web browser (Section 35.2.1 on page 425).

35.1.2 What You Need to Know

Application Types

You can configure the following SSL application on the ZyWALL.

Web-based

A web-based application allows remote users to access an intranet site using standard web browsers.

Remote User Screen Links

Available SSL application names are displayed as links in remote user screens. Depending on the application type, remote users can simply click the links or follow the steps in the pop-up dialog box to access.

Remote Desktop Connections

Use SSL VPN to allow remote users to manage LAN computers. Depending on the functions supported by the remote desktop software, they can install or remove software, run programs, change settings, and open, copy, create, and delete files. This is useful for troubleshooting, support, administration, and remote access to files and programs.

The LAN computer to be managed must have VNC (Virtual Network Computing) or RDP (Remote Desktop Protocol) server software installed. The remote user's computer does not use VNC or RDP client software. The ZyWALL works with the following remote desktop connection software:

RDP

• Windows Remote Desktop (supported in Internet Explorer)

VNC

- RealVNC
- TightVNC
- UltraVNC

For example, user A uses an SSL VPN connection to log into the ZyWALL. Then he manages LAN computer B which has RealVNC server software installed.

Figure 284 SSL-protected Remote Management



Weblinks

You can configure weblink SSL applications to allow remote users to access web sites.

35.1.3 Example: Specifying a Web Site for Access

This example shows you how to create a web-based application for an internal web site. The address of the web site is http://info with web page encryption.

- 1 Click Configuration > Object > SSL Application in the navigation panel.
- 2 Click the Add button and select Web Application in the Type field.

In the Server Type field, select Web Server.

Enter a descriptive name in the **Display Name** field. For example, "CompanyIntranet".

In the Address field, enter "http://info".

Select Web Page Encryption to prevent users from saving the web content.

Click **OK** to save the settings.

The configuration screen should look similar to the following figure.

Add SSL Application			?
Create new Object 🔻			
bject			
Туре:	Web Application	~	
Yeb Application			
Server Type:	Web Server	~	
Name:	WebExample		
URL:	http://info		
Entry Point:		(Optional)	
👿 Web Page Encryp	tion		
			OK Cancel

Figure 285 Example: SSL Application: Specifying a Web Site for Access

35.2 The SSL Application Screen

The main **SSL Application** screen displays a list of the configured SSL application objects. Click **Configuration > Object > SSL Application** in the navigation panel.

Figure 286 Configuration > Object > SSL Application

onrig	uration		
0	Add 📝 Edit 🍵	Remove 📴 Object References	
#	Name 🔺	Address	Туре
1	FileSharing_1	\\example\example_1	file-sharing
2	OWA-example	http://mail.example	owa
3	VNC_Server1	DMZ2_SUBNET:5900-5900	vnc
4	WebExample	http://info	web-server
5	Weblink-Example	http://example.com	weblink

LABEL	DESCRIPTION
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object Reference s to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field displays the index number.
Name	This field displays the name of the object.

Table 168 Configuration > Object > SSL Application

	oringulation > Object > 35L Application
LABEL	DESCRIPTION
Address	This field displays the IP address/URL of the application server or the location of a file share.
Туре	This field shows whether the object is a file-sharing, web-server, Outlook Web Access, Virtual Network Computing, or Remote Desktop Protocol SSL application.

 Table 168
 Configuration > Object > SSL Application

35.2.1 Creating/Editing an SSL Application Object

You can create a web-based application that allows remote users to access an application via standard web browsers. You can also create a file sharing application that specify the name of a folder on a file server (Linux or Windows) which remote users can access. Remote users can access files using a standard web browser and files are displayed as links on the screen.

To configure an SSL application, click the **Add** or **Edit** button in the **SSL Application** screen and select **Web Application** or **File Sharing** in the **Type** field. The screen differs depending on what object type you choose.

Note: If you are creating a file sharing SSL application, you must also configure the shared folder on the file server for remote access. Refer to the document that comes with your file server.

Figure 287	Configuration >	Ohiect >	SSL Application >	Add/Edit: Web	Application
rigule 201	connyuration >		JJL Application >	Add/Luit. WC	

O Add SSL Application		? ×
🔠 Create new Object 🗸		
Object		
Type:	Web Application	
Web Application		
Server Type:	Web Server 👻	
Name:	New 🐠	
URL:	Preview	
Entry Point:	(Optional)	
Web Page Encryption		
		OK Cancel

Add SSL Application		?
Create new Object 🗸		
0bject		
Type:	File Sharing	
File Sharing		
Name:	New	
Shared Path:		

Figure 288 Configuration > Object > SSL Application > Add/Edit: File Sharing

LABEL	DESCRIPTION
Create new Object	Use this to configure any new settings objects that you need to use in this screen.
Object	
Туре	Select Web Application or File Sharing from the drop-down list box.
Web Application	
Server Type	This field only appears when you choose Web Application as the object type.
	Specify the type of service for this SSL application.
	Select Web Server to allow access to the specified web site hosted on the local network.
	Select OWA (Outlook Web Access) to allow users to access e-mails, contacts, calenders via Microsoft Outlook-like interface using supported web browsers. The ZyWALL supports one OWA object.
	Select VNC to allow users to manage LAN computers that have Virtual Network Computing remote desktop server software installed.
	Select RDP to allow users to manage LAN computers that have Remote Desktop Protocol remote desktop server software installed.
	Select Weblink to create a link to a web site that you expect the SSL VPN users to commonly use.
Name	Enter a descriptive name to identify this object. You can enter up to 31 characters ("0-9", "a-z", "A-Z", "-" and "_"). Spaces are not allowed.
URL	This field only appears when you choose Web Application as the object type.
	This field displays if the Server Type is set to Web Server, OWA, or Weblink.
	Enter the Fully-Qualified Domain Name (FQDN) or IP address of the application server.
	Note: You must enter the "http://" or "https://" prefix.
	Remote users are restricted to access only files in this directory. For example, if you enter "\remote\" in this field, remote users can only access files in the "remote" directory.
	If a link contains a file that is not within this domain, then remote users cannot access it.

 Table 169
 Configuration > Object > SSL Application > Add/Edit: Web Application

LABEL	DESCRIPTION
Preview	This field only appears when you choose Web Application as the object type.
	This field displays if the Server Type is set to Web Server, OWA or Weblink.
	Click Preview to access the URL you specified in a new IE web browser.
Entry Point	This field only appears when you choose Web Application as the object type.
	This field displays if the Server Type is set to Web Server or OWA.
	This field is optional. You only need to configure this field if you need to specify the name of the directory or file on the local server as the home page or home directory on the user screen.
Web Page	This field only appears when you choose Web Application as the object type.
Encryption	Select this option to prevent users from saving the web content.
Shared Path	This field only appears when you choose File Sharing as the object type.
	Specify the IP address, domain name or NetBIOS name (computer name) of the file server and the name of the share to which you want to allow user access. Enter the path in one of the following formats.
	"\\ <ip address="">\<share name="">"</share></ip>
	"\\ <domain name="">\<share name="">"</share></domain>
	"\\ <computer name="">\<share name="">"</share></computer>
	For example, if you enter "\\my-server\Tmp", this allows remote users to access all files and/or folders in the "\Tmp" share on the "my-server" computer.
ОК	Click OK to save the changes and return to the main SSL Application Configuration screen.
Cancel	Click Cancel to discard the changes and return to the main SSL Application Configuration screen.

 Table 169
 Configuration > Object > SSL Application > Add/Edit: Web Application (continued)

36

DHCPv6

36.1 Overview

This chapter describes how to configure DHCPv6 request type and lease type objects.

36.1.1 What You Can Do in this Chapter

- The **Request** screen (see Section 27.2 on page 363) allows you to configure DHCPv6 request type objects.
- The Lease screen (see Section 27.3 on page 366) allows you to configure DHCPv6 lease type objects.

36.2 The DHCPv6 Request Screen

The **Request** screen allows you to add, edit, and remove DHCPv6 request type objects. To access this screen, login to the Web Configurator, and click **Configuration > Object > DHCPv6 > Request**.

Figure 289 Configuration > Object > DHCPv6 > Request

🕽 Add 🍞 Edit 🍵 Remove	Cobject Reference			
Name i	Ture	Interface	Value	

Table 170	Config	uration > Object > DHCPv6 > Request
LABEL		DESCRIPTION

LABEL	DESCRIPTION
Configuration	
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific object.
Name	This field displays the name of each request object.

Table ITO Coning	
LABEL	DESCRIPTION
Туре	This field displays the request type of each request object.
Interface	This field displays the interface used for each request object.
Value	This field displays the value for each request object.

Table 170 Configuration > Object > DHCPv6 > Request (continued)

36.2.1 DHCPv6 Request Add/Edit Screen

The Request Add/Edit screen allows you to create a new request object or edit an existing one.

To access this screen, go to the **Request** screen (see Section 27.2 on page 363), and click either the **Add** icon or an **Edit** icon.

	_	
Prefix Delegation	~	
none	~	
	Prefix Delegation none	Prefix Delegation

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Name	Type the name for this request object. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Request Type	Select the request type for this request object. You can choose from Prefix Delegation , DNS Server , NTP Server , or SIP Server .
Interface	Select the interface for this request object.
OK	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

Table 171 Configuration > DHCPv6 > Request > Add

36.3 The DHCPv6 Lease Screen

The **Lease** screen allows you to add, edit, and remove DHCPv6 lease type objects. To access this screen, login to the Web Configurator, and click **Configuration > Object > DHCPv6 > Lease**.

Figure 291 Con	figuration >	Object >	DHCPv6 >	Lease
----------------	--------------	----------	----------	-------

ucst LCGsc and			
ifiguration			
Add L Edit Remove	Object Reference		

LABEL	DESCRIPTION
Configuration	
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so.
Object References	Select an entry and click Object References to open a screen that shows which settings use the entry. See Section 7.3.2 on page 122 for an example.
#	This field is a sequential value, and it is not associated with a specific object.
Name	This field displays the name of each lease object.
Туре	This field displays the request type of each lease object.
Interface	This field displays the interface used for each lease object.
Value	This field displays the value for each lease object.

 Table 172
 Configuration > Object > DHCPv6 > Lease

36.3.1 DHCPv6 Lease Add/Edit Screen

The Lease Add/Edit screen allows you to create a new lease object or edit an existing one.

To access this screen, go to the Lease screen (see Section 36.3 on page 429), and click either the Add icon or an Edit icon.

Lease Type: Prefix Delegation Interface: DUID:	Name:			
Interface: none 🔽	Lease Type:	Prefix Delegation	~	
DUID:	Interface:	none	~	
	DUID:			
Prefix:	Prefix:]0	

 \sim ٩dd

LABEL	DESCRIPTION
Name	Type the name for this lease object. You may use 1-31 alphanumeric characters, underscores(_), or dashes (-), but the first character cannot be a number. This value is case-sensitive.
Lease Type	Select the lease type for this lease object. You can choose from Prefix Delegation , DNS Server , Address , Address Pool , NTP Server , or SIP Server .
Interface	Select the interface for this lease object.
DUID	If you select Prefix Delegation or Address in the Lease Type field , enter the DUID of the interface.
Prefix	If you select Prefix Delegation or Address in the Lease Type field , enter the IPv6 prefix of the interface.
DNS Server	If you select DNS Server in the Lease Type field , select a request object or User Defined in the DNS Server field and enter the IP address of the DNS server in the User Defined Address field below.
Starting IP Address	If you select Address Pool in the Lease Type field , enter the first of the contiguous addresses in the IP address pool.
End IP Address	If you select Address Pool in the Lease Type field , enter the last of the contiguous addresses in the IP address pool.
NTP Server	If you select NTP Server in the Lease Type field , select a request object or User Defined in the NTP Server field and enter the IP address of the NTP server in the User Defined Address field below.
SIP Server	If you select SIP Server in the Lease Type field , select a request object or User Defined in the SIP field and enter the IP address of the SIP server in the User Defined Address field below.
User Defined Address	If you select DNS Server , NTP Server , or SIP Server as your lease type, you must enter the IP address of the server your selected.
ОК	Click OK to save your changes back to the ZyWALL.
Cancel	Click Cancel to exit this screen without saving your changes.

 Table 173
 Configuration > DHCPv6 > Lease > Add

37

System

37.1 Overview

Use the system screens to configure general ZyWALL settings.

37.1.1 What You Can Do in this Chapter

- Use the **System > Host Name** screen (see Section 37.2 on page 433) to configure a unique name for the ZyWALL in your network.
- Use the **System > USB Storage** screen (see Section 37.3 on page 433) to configure the settings for the connected USB devices.
- Use the System > Date/Time screen (see Section 37.4 on page 434) to configure the date and time for the ZyWALL.
- Use the System > Console Speed screen (see Section 37.5 on page 438) to configure the console port speed when you connect to the ZyWALL via the console port using a terminal emulation program.
- Use the System > DNS screen (see Section 37.6 on page 439) to configure the DNS (Domain Name System) server used for mapping a domain name to its corresponding IP address and vice versa.
- Use the **System > WWW** screens (see Section 37.7 on page 445) to configure settings for HTTP or HTTPS access to the ZyWALL and how the login and access user screens look.
- Use the System > SSH screen (see Section 37.8 on page 461) to configure SSH (Secure SHeII) used to securely access the ZyWALL's command line interface. You can specify which zones allow SSH access and from which IP address the access can come.
- Use the System > TELNET screen (see Section 37.9 on page 465) to configure Telnet to access the ZyWALL's command line interface. Specify which zones allow Telnet access and from which IP address the access can come.
- Use the System > FTP screen (see Section 37.10 on page 467) to specify from which zones FTP can be used to access the ZyWALL. You can also specify from which IP addresses the access can come. You can upload and download the ZyWALL's firmware and configuration files using FTP. Please also see Chapter 39 on page 488 for more information about firmware and configuration files.
- Your ZyWALL can act as an SNMP agent, which allows a manager station to manage and monitor the ZyWALL through the network. Use the System > SNMP screen (see Section 37.11 on page 468) to configure SNMP settings, including from which zones SNMP can be used to access the ZyWALL. You can also specify from which IP addresses the access can come.
- Use the **System > Language** screen (see Section 37.12 on page 472) to set a language for the ZyWALL's Web Configurator screens.
- Use the System > IPv6 screen (see Section 37.13 on page 472) to enable or disable IPv6 support on the ZyWALL.

Note: See each section for related background information and term definitions.
37.2 Host Name

A host name is the unique name by which a device is known on a network. Click **Configuration > System > Host Name** to open the **Host Name** screen.

Figure 293 Configuration > System > Host Name

(Online al)
ile: (Optional)
(optional)

The following table describes the labels in this screen.

LABEL	DESCRIPTION
System Name	Enter a descriptive name to identify your ZyWALL device. This name can be up to 64 alphanumeric characters long. Spaces are not allowed, but dashes (-) underscores (_) and periods (.) are accepted.
Domain Name	Enter the domain name (if you know it) here. This name is propagated to DHCP clients connected to interfaces with the DHCP server enabled. This name can be up to 254 alphanumeric characters long. Spaces are not allowed, but dashes "-" are accepted.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 174
 Configuration > System > Host Name

37.3 USB Storage

The ZyWALL can use a connected USB device to store the system log and other diagnostic information. Use this screen to turn on this feature and set a disk full warning limit.

Note: Only connect one USB device. It must allow writing (it cannot be read-only) and use the FAT16, FAT32, EXT2, or EXT3 file system.

Click **Configuration > System > USB Storage** to open the screen as shown next.

Figure 294 Configuration > System > USB Storage

Settings			
Capaval			
Gelleral			
Activate USB storage service			
Diele Eully services such as a service a success in large theory		100	MP
Disk full warning when remaining space is less than:		100	
	(
	Apply	Reset	J

The following table describes the labels in this screen.

Table 175	Configuration	>	System	>	USB	Storage

LABEL	DESCRIPTION
Activate USB storage service	Select this if you want to use the connected USB device(s).
Disk full warning when remaining space is less than	Set a number and select a unit (MB or %) to have the ZyWALL send a warning message when the remaining USB storage space is less than the value you set here.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

37.4 Date and Time

For effective scheduling and logging, the ZyWALL system time must be accurate. The ZyWALL's Real Time Chip (RTC) keeps track of the time and date. There is also a software mechanism to set the time manually or get the current time and date from an external server.

To change your ZyWALL's time based on your local time zone and date, click **Configuration > System > Date/Time**. The screen displays as shown. You can manually set the ZyWALL's time and date or have the ZyWALL get the date and time from a time server.

Current Time:	06:26:09 GMT+00	:00							
Current Date:	2009-11-13								
me and Date Setup									
🔘 Manual									
New Time (hh:mm:ss):	06 : 25 :	39							
New Date (yyyy-mm-dd):	2009-11-13								
-									
Get from Time Server									
 Get from Time Server Time Server Address*: *Optional. There is a pre- 	0.pool.ntp.org -defined NTP time set	rver list.		Syn	c. Now				
 Get from Time Server Time Server Address*: *Optional. There is a pre- me Zone Setup Time Zone: 	0.pool.ntp.org -defined NTP time set (GMT 00:00) Greet	rver list. nwich Mean Tim	e : Dublin, F	Syna Edinburgh, Li 🗸	c. Now				
 Get from Time Server Time Server Address*: *Optional. There is a pre- me Zone Setup Time Zone: Enable Daylight Saving 	0.pool.ntp.org -defined NTP time set (GMT 00:00) Greet	rver list. nwich Mean Tim	e : Dublin, E	dinburgh, Li 🗸	c. Now				
 Get from Time Server Time Server Address*: *Optional. There is a pre- me Zone Setup Time Zone: Enable Daylight Saving Start Date: 	0.pool.ntp.org -defined NTP time set (GMT 00:00) Green First	rver list. nwich Mean Tim	e : Dublin, E	Edinburgh, Li v	c. Now	at	12	: 0	00
 Get from Time Server Time Server Address*: *Optional. There is a pre- me Zone Setup Time Zone: Enable Daylight Saving Start Date: End Date: 	0.pool.ntp.org -defined NTP time set (GMT 00:00) Greet First	rver list. nwich Mean Tim Monday	e : Dublin, E	Edinburgh, Li v January January	c. Now	at	12]: [C]: [C	00
 Get from Time Server Time Server Address*: *Optional. There is a pre- Time Zone Setup Time Zone: Enable Daylight Saving Start Date: End Date: Offset: 	0.pool.ntp.org -defined NTP time set (GMT 00:00) Greet First First 1 hours	rver list, nwich Mean Tim r Monday r Monday	e : Dublin, E	Edinburgh, Li v January January	c. Now	at at	12	: C : C	00
 Get from Time Server Time Server Address*: *Optional. There is a pre- me Zone Setup Time Zone: Enable Daylight Saving Start Date: End Date: Offset: 	0.pool.ntp.org -defined NTP time set (GMT 00:00) Green First	rver list. nwich Mean Tim Monday	e : Dublin, E	Edinburgh, Li 🗸 January January	c. Now	at at	12]: [C]: [C	00
 Get from Time Server Time Server Address*: *Optional. There is a pre- me Zone Setup Time Zone: Enable Daylight Saving Start Date: End Date: Offset: 	0.pool.ntp.org -defined NTP time set (GMT 00:00) Greet First First 1 hours	rver list, nwich Mean Tim Monday Monday	e : Dublin, E	Edinburgh, Li v January January	c. Now	at at	12] : [C] : [C	00

ч ті **D** - +

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Current Time and Date	
Current Time	This field displays the present time of your ZyWALL.
Current Date	This field displays the present date of your ZyWALL.
Time and Date Setup	
Manual	Select this radio button to enter the time and date manually. If you configure a new time and date, time zone and daylight saving at the same time, the time zone and daylight saving will affect the new time and date you entered. When you enter the time settings manually, the ZyWALL uses the new setting once you click Apply .
New Time (hh-mm- ss)	This field displays the last updated time from the time server or the last time configured manually. When you set Time and Date Setup to Manual , enter the new time in this field and then click Apply .
New Date (yyyy-mm-dd)	This field displays the last updated date from the time server or the last date configured manually. When you set Time and Date Setup to Manual , enter the new date in this field and then click Apply .

Table 176 Configuration > System > Date and Time

LABEL	DESCRIPTION
Get from Time Server	Select this radio button to have the ZyWALL get the time and date from the time server you specify below. The ZyWALL requests time and date settings from the time server under the following circumstances.
	 When the ZyWALL starts up. When you click Apply or Synchronize Now in this screen. 24-hour intervals after starting up.
Time Server Address	Enter the IP address or URL of your time server. Check with your ISP/network administrator if you are unsure of this information.
Sync. Now	Click this button to have the ZyWALL get the time and date from a time server (see the Time Server Address field). This also saves your changes (except the daylight saving settings).
Time Zone Setup	
Time Zone	Choose the time zone of your location. This will set the time difference between your time zone and Greenwich Mean Time (GMT).
Enable Daylight Saving	Daylight saving is a period from late spring to early fall when many countries set their clocks ahead of normal local time by one hour to give more daytime light in the evening.
	Select this option if you use Daylight Saving Time.
Start Date	Configure the day and time when Daylight Saving Time starts if you selected Enable Daylight Saving . The at field uses the 24 hour format. Here are a couple of examples:
	Daylight Saving Time starts in most parts of the United States on the second Sunday of March. Each time zone in the United States starts using Daylight Saving Time at 2 A.M. local time. So in the United States you would select Second , Sunday , March and type 2 in the at field.
	Daylight Saving Time starts in the European Union on the last Sunday of March. All of the time zones in the European Union start using Daylight Saving Time at the same moment (1 A.M. GMT or UTC). So in the European Union you would select Last , Sunday , March . The time you type in the at field depends on your time zone. In Germany for instance, you would type 2 because Germany's time zone is one hour ahead of GMT or UTC (GMT+1).
End Date	Configure the day and time when Daylight Saving Time ends if you selected Enable Daylight Saving . The at field uses the 24 hour format. Here are a couple of examples:
	Daylight Saving Time ends in the United States on the first Sunday of November. Each time zone in the United States stops using Daylight Saving Time at 2 A.M. local time. So in the United States you would select First , Sunday , November and type 2 in the at field.
	Daylight Saving Time ends in the European Union on the last Sunday of October. All of the time zones in the European Union stop using Daylight Saving Time at the same moment (1 A.M. GMT or UTC). So in the European Union you would select Last , Sunday , October . The time you type in the at field depends on your time zone. In Germany for instance, you would type 2 because Germany's time zone is one hour ahead of GMT or UTC (GMT+1).
Offset	Specify how much the clock changes when daylight saving begins and ends.
	Enter a number from 1 to 5.5 (by 0.5 increments).
	For example, if you set this field to 3.5, a log occurred at 6 P.M. in local official time will appear as if it had occurred at 10:30 P.M.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 176 Configuration > System > Date and Time (continued)

37.4.1 Pre-defined NTP Time Servers List

When you turn on the ZyWALL for the first time, the date and time start at 2003-01-01 00:00:00. The ZyWALL then attempts to synchronize with one of the following pre-defined list of Network Time Protocol (NTP) time servers.

The ZyWALL continues to use the following pre-defined list of NTP time servers if you do not specify a time server or it cannot synchronize with the time server you specified.

Table 177Default Time Servers

0.pool.ntp.org
1.pool.ntp.org
2.pool.ntp.org

When the ZyWALL uses the pre-defined list of NTP time servers, it randomly selects one server and tries to synchronize with it. If the synchronization fails, then the ZyWALL goes through the rest of the list in order from the first one tried until either it is successful or all the pre-defined NTP time servers have been tried.

37.4.2 Time Server Synchronization

Click the **Synchronize Now** button to get the time and date from the time server you specified in the **Time Server Address** field.

When the Please Wait... screen appears, you may have to wait up to one minute.



The **Current Time** and **Current Date** fields will display the appropriate settings if the synchronization is successful.

If the synchronization was not successful, a log displays in the **View Log** screen. Try re-configuring the **Date/Time** screen.

To manually set the ZyWALL date and time.

- 1 Click System > Date/Time.
- 2 Select Manual under Time and Date Setup.
- **3** Enter the ZyWALL's time in the **New Time** field.
- 4 Enter the ZyWALL's date in the **New Date** field.
- 5 Under Time Zone Setup, select your Time Zone from the list.
- 6 As an option you can select the **Enable Daylight Saving** check box to adjust the ZyWALL clock for daylight savings.

7 Click Apply.

To get the ZyWALL date and time from a time server

- 1 Click System > Date/Time.
- 2 Select Get from Time Server under Time and Date Setup.
- 3 Under Time Zone Setup, select your Time Zone from the list.
- 4 As an option you can select the **Enable Daylight Saving** check box to adjust the ZyWALL clock for daylight savings.
- 5 Under Time and Date Setup, enter a Time Server Address (Table 177 on page 437).
- 6 Click Apply.

37.5 Console Port Speed

This section shows you how to set the console port speed when you connect to the ZyWALL via the console port using a terminal emulation program. See Table 2 on page 21 for default console port settings.

Click Configuration > System > Console Speed to open the Console Speed screen.

Figure 297 Conf	iguration > 3	System > Cor	sole Speed
Console Speed		-	-
General Settings			
Console Port Speed:	115200	~	
		Apply Re	eset

The following table describes the labels in this screen.

 Table 178
 Configuration > System > Console Speed

LABEL	DESCRIPTION
Console Port Speed	Use the drop-down list box to change the speed of the console port. Your ZyWALL supports 9600, 19200, 38400, 57600, and 115200 bps (default) for the console port.
	The Console Port Speed applies to a console port connection using terminal emulation software and NOT the Console in the ZyWALL Web Configurator Status screen.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

37.6 DNS Overview

DNS (Domain Name System) is for mapping a domain name to its corresponding IP address and vice versa. The DNS server is extremely important because without it, you must know the IP address of a machine before you can access it.

37.6.1 DNS Server Address Assignment

The ZyWALL can get the DNS server addresses in the following ways.

- The ISP tells you the DNS server addresses, usually in the form of an information sheet, when you sign up. If your ISP gives you DNS server addresses, manually enter them in the DNS server fields.
- If your ISP dynamically assigns the DNS server IP addresses (along with the ZyWALL's WAN IP address), set the DNS server fields to get the DNS server address from the ISP.
- You can manually enter the IP addresses of other DNS servers.

37.6.2 Configuring the DNS Screen

Click **Configuration > System > DNS** to change your ZyWALL's DNS settings. Use the **DNS** screen to configure the ZyWALL to use a DNS server to resolve domain names for ZyWALL system features like VPN, DDNS and the time server. You can also configure the ZyWALL to accept or discard DNS queries. Use the **Network > Interface** screens to configure the DNS server information that the ZyWALL sends to the specified DHCP client devices.

ddress/P1	TR Record				
🗿 Add	📝 Edit 🍵 Remove	a.			
# 🔺	FQDN	IP A	Address		
14 4 1	Page 1 of 1	🕨 Show 50 🗸 i	items		No data to display
omain Zor	ne Forwarder				
🗿 Add	📝 Edit 🍵 Remove	a 📣 Move			
# 🔺 Dom	nain Zone Type		DNS Server	Query via	
	Default		10.5.5.1	wap2	
				TUTL	
14 4 1	Page 1 of 1 >	▶ Show 50 ▼ i	items		Displaying 1 - 1 of 1
X Record	Page 1 of 1 • • (for My FQDN)	▶ Show 50 V	items		Displaying 1 - 1 of 1
X Record	Page 1 of 1 (for My FQDN) Edit TRemove Domain Name	▶ Show 50 v i	items = GDN		Displaying 1 - 1 of 1
I4 4 X Record Image: Constraint of the second of	Page 1 of 1 (for My FQDN) Edit TRemove Domain Name Page 1 of 1	Image: Show 50 Image: Show Im	FQDN items		Displaying 1 - 1 of 1 No data to display
X Record	Page 1 of 1 (for My FQDN) Edit TRemove Domain Name Page 1 of 1 htrol	Image: Show 50 Image: Show Im	FGDN tems		Displaying 1 - 1 of 1 No data to display
X Record Add Add Add Add Add Crvice Cor	Page 1 of 1 (for My FQDN) Edit Transver Page 1 of 1 htrol Edit Transver Page 2 Remove	Image: Show 50 Image: Show Im	FQDN Items		Displaying 1 - 1 of 1 No data to display
X Record Add X Add Add Add X Add X Add X Add X Add X Add	Page 1 of 1 (for My FQDN) Charles Edit Remove Domain Name Page 1 of 1 Charles Remove Remov	Show 50	FQDN	Action	Displaying 1 - 1 of 1 No data to display
X Record Add Add Add Add Add Add Add A	Page 1 of 1 (for My FQDN) Consin Name Page 1 of 1 trol Edit TREMOVE Remove Remo	 Show 50 v IP# Show 50 v Address ALL 	FQDN Items	Action	Displaying 1 - 1 of 1 No data to display

Figure 298 Configuration > System > DNS

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Address/PTR Record	This record specifies the mapping of a Fully-Qualified Domain Name (FQDN) to an IP address. An FQDN consists of a host and domain name. For example, www.zyxel.com.tw is a fully qualified domain name, where "www" is the host, "zyxel" is the third-level domain, "com" is the second-level domain, and "tw" is the top level domain.
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.
#	This is the index number of the address/PTR record.
FQDN	This is a host's fully qualified domain name.
IP Address	This is the IP address of a host.
Domain Zone Forwarder	This specifies a DNS server's IP address. The ZyWALL can query the DNS server to resolve domain zones for features like VPN, DDNS and the time server.
	When the ZyWALL needs to resolve a domain zone, it checks it against the domain zone forwarder entries in the order that they appear in this list.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.
Move	To change an entry's position in the numbered list, select the method and click Move to display a field to type a number for where you want to put it and press [ENTER] to move the rule to the number that you typed.
#	This is the index number of the domain zone forwarder record. The ordering of your rules is important as rules are applied in sequence.
	A hyphen (-) displays for the default domain zone forwarder record. The default record is not configurable. The ZyWALL uses this default record if the domain zone that needs to be resolved does not match any of the other domain zone forwarder records.
Domain Zone	A domain zone is a fully qualified domain name without the host. For example, zyxel.com.tw is the domain zone for the www.zyxel.com.tw fully qualified domain name.
	A "*" means all domain zones.
Туре	This displays whether the DNS server IP address is assigned by the ISP dynamically through a specified interface or configured manually (User-Defined).
DNS Server	This is the IP address of a DNS server. This field displays N/A if you have the ZyWALL get a DNS server IP address from the ISP dynamically but the specified interface is not active.
Query Via	This is the interface through which the ZyWALL sends DNS queries to the entry's DNS server. If the ZyWALL connects through a VPN tunnel, tunnel displays.
MX Record (for My FQDN)	A MX (Mail eXchange) record identifies a mail server that handles the mail for a particular domain.
Add	Click this to create a new entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.

 Table 179
 Configuration > System > DNS

LABEL	DESCRIPTION
#	This is the index number of the MX record.
Domain Name	This is the domain name where the mail is destined for.
IP/FQDN	This is the IP address or Fully-Qualified Domain Name (FQDN) of a mail server that handles the mail for the domain specified in the field above.
Service Control	This specifies from which computers and zones you can send DNS queries to the ZyWALL.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.
Move	To change an entry's position in the numbered list, select the method and click Move to display a field to type a number for where you want to put it and press [ENTER] to move the rule to the number that you typed.
#	This the index number of the service control rule. The ordering of your rules is important as rules are applied in sequence.
	The entry with a hyphen (-) instead of a number is the ZyWALL's (non-configurable) default policy. The ZyWALL applies this to traffic that does not match any other configured rule. It is not an editable rule. To apply other behavior, configure a rule that traffic will match so the ZyWALL will not have to use the default policy.
Zone	This is the zone on the ZyWALL the user is allowed or denied to access.
Address	This is the object name of the IP address(es) with which the computer is allowed or denied to send DNS queries.
Action	This displays whether the ZyWALL accepts DNS queries from the computer with the IP address specified above through the specified zone (Accept) or discards them (Deny).

 Table 179
 Configuration > System > DNS (continued)

37.6.3 Address Record

An address record contains the mapping of a Fully-Qualified Domain Name (FQDN) to an IP address. An FQDN consists of a host and domain name. For example, www.zyxel.com is a fully qualified domain name, where "www" is the host, "zyxel" is the second-level domain, and "com" is the top level domain. mail.myZyXEL.com.tw is also a FQDN, where "mail" is the host, "myZyXEL" is the third-level domain, "com" is the second-level domain, and "tw" is the top level domain.

The ZyWALL allows you to configure address records about the ZyWALL itself or another device. This way you can keep a record of DNS names and addresses that people on your network may use frequently. If the ZyWALL receives a DNS query for an FQDN for which the ZyWALL has an address record, the ZyWALL can send the IP address in a DNS response without having to query a DNS name server.

37.6.4 PTR Record

A PTR (pointer) record is also called a reverse record or a reverse lookup record. It is a mapping of an IP address to a domain name.

37.6.5 Adding an Address/PTR Record

Click the Add icon in the Address/PTR Record table to add an address/PTR record.

Figure 299 Configuration > System > DNS > Address/PTR Record Edit

Add Address/PT	R Record	? ×
FQDN: IP Address:		
Default	OK /	Cancel

The following table describes the labels in this screen.

Table 180 Configuration > System > DNS > Address/PTR Record Edit

LABEL	DESCRIPTION
FQDN	Type a Fully-Qualified Domain Name (FQDN) of a server. An FQDN starts with a host name and continues all the way up to the top-level domain name. For example, www.zyxel.com.tw is a fully qualified domain name, where "www" is the host, "zyxel" is the third-level domain, "com" is the second-level domain, and "tw" is the top level domain. Underscores are not allowed. Use "*." as a prefix in the FQDN for a wildcard domain name (for example, *.example.com).
IP Address	Enter the IP address of the host in dotted decimal notation.
ОК	Click OK to save your customized settings and exit this screen.
Cancel	Click Cancel to exit this screen without saving

37.6.6 Domain Zone Forwarder

A domain zone forwarder contains a DNS server's IP address. The ZyWALL can query the DNS server to resolve domain zones for features like VPN, DDNS and the time server. A domain zone is a fully qualified domain name without the host. For example, zyxel.com.tw is the domain zone for the www.zyxel.com.tw fully qualified domain name.

37.6.7 Adding a Domain Zone Forwarder

Click the **Add** icon in the **Domain Zone Forwarder** table to add a domain zone forwarder record.

Domain Zone:		
DNS Server		
ONS Server(s) from ISP	aux 💌	
First DNS Server:	N/A	
Second DNS Server:	N/A	
Third DNS Server:	N/A	
Public DNS Server		
Query via:	auto 📉 😽	
Private DNS Server		2

Figure 300 Configuration > System > DNS > Domain Zone Forwarder Add

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Domain Zone	A domain zone is a fully qualified domain name without the host. For example, zyxel.com.tw is the domain zone for the www.zyxel.com.tw fully qualified domain name. For example, whenever the ZyWALL receives needs to resolve a zyxel.com.tw domain name, it can send a query to the recorded name server IP address. Enter * if all domain zones are served by the specified DNS server(s).
DNS Server	Select DNS Server(s) from ISP if your ISP dynamically assigns DNS server information. You also need to select an interface through which the ISP provides the DNS server IP address(es). The interface should be activated and set to be a DHCP client. The fields below display the (read-only) DNS server IP address(es) that the ISP assigns. N/A displays for any DNS server IP address fields for which the ISP does not assign an IP address.
	Select Public DNS Server if you have the IP address of a DNS server. Enter the DNS server's IP address in the field to the right. The ZyWALL must be able to connect to the DNS server without using a VPN tunnel. The DNS server could be on the Internet or one of the ZyWALL's local networks. You cannot use 0.0.0.0. Use the Query via field to select the interface through which the ZyWALL sends DNS queries to a DNS server.
	Select Private DNS Server if you have the IP address of a DNS server to which the ZyWALL connects through a VPN tunnel. Enter the DNS server's IP address in the field to the right. You cannot use 0.0.0.0.
ОК	Click OK to save your customized settings and exit this screen.
Cancel	Click Cancel to exit this screen without saving

Table 181	Configuration	> System >	DNS >	Domain	Zone For	warder Add
	3					

37.6.8 MX Record

A MX (Mail eXchange) record indicates which host is responsible for the mail for a particular domain, that is, controls where mail is sent for that domain. If you do not configure proper MX records for your domain or other domain, external e-mail from other mail servers will not be able to be delivered to your mail server and vice versa. Each host or domain can have only one MX record, that is, one domain is mapping to one host.

37.6.9 Adding a MX Record

Click the **Add** icon in the **MX Record** table to add a MX record.

Figure 301 Configuration > System > DNS > MX Record Add

O Add MX Record Show 5	0 🗸 items 🕐 🗙
Domain Name: IP Address/FQDN:	
Default	OK Cancel

The following table describes the labels in this screen.

Table 182Configuration > System > DNS > MX Record Add

LABEL	DESCRIPTION
Domain Name	Enter the domain name where the mail is destined for.
IP Address/FQDN	Enter the IP address or Fully-Qualified Domain Name (FQDN) of a mail server that handles the mail for the domain specified in the field above.
ОК	Click OK to save your customized settings and exit this screen.
Cancel	Click Cancel to exit this screen without saving

37.6.10 Adding a DNS Service Control Rule

Click the Add icon in the Service Control table to add a service control rule.

Figure 302 Configuration > System > DNS > Service Control Rule Add

Add Service Control	anu 50 and itoma	? 🗙
🛅 Create new Object	•	
Address Object:	ALL	*
Zone:	ALL	*
Action:	Accept	*
e		
		OK Cancel
Edit 📕 Remove 🚽 Mo		

The following table describes the labels in this screen.

 Table 183
 Configuration > System > DNS > Service Control Rule Add

LABEL	DESCRIPTION
Create new Object	Use this to configure any new settings objects that you need to use in this screen.
Address Object	Select ALL to allow or deny any computer to send DNS queries to the ZyWALL.
	Select a predefined address object to just allow or deny the computer with the IP address that you specified to send DNS queries to the ZyWALL.
Zone	Select ALL to allow or prevent DNS queries through any zones.
	Select a predefined zone on which a DNS query to the ZyWALL is allowed or denied.
Action	Select Accept to have the ZyWALL allow the DNS queries from the specified computer.
	Select Deny to have the ZyWALL reject the DNS queries from the specified computer.
ОК	Click OK to save your customized settings and exit this screen.
Cancel	Click Cancel to exit this screen without saving

37.7 WWW Overview

The following figure shows secure and insecure management of the ZyWALL coming in from the WAN. HTTPS and SSH access are secure. HTTP and Telnet access are not secure.

- Note: To allow the ZyWALL to be accessed from a specified computer using a service, make sure you do not have a service control rule or to-ZyWALL firewall rule to block that traffic.
- See To-ZyWALL Rules on page 257 for more on To-ZyWALL firewall rules.

To stop a service from accessing the ZyWALL, clear **Enable** in the corresponding service screen.

37.7.1 Service Access Limitations

A service cannot be used to access the ZyWALL when:

- 1 You have disabled that service in the corresponding screen.
- 2 The allowed IP address (address object) in the **Service Control** table does not match the client IP address (the ZyWALL disallows the session).
- **3** The IP address (address object) in the **Service Control** table is not in the allowed zone or the action is set to **Deny**.
- 4 There is a firewall rule that blocks it.

37.7.2 System Timeout

There is a lease timeout for administrators. The ZyWALL automatically logs you out if the management session remains idle for longer than this timeout period. The management session does not time out when a statistics screen is polling.

Each user is also forced to log in the ZyWALL for authentication again when the reauthentication time expires.

You can change the timeout settings in the User/Group screens.

37.7.3 HTTPS

You can set the ZyWALL to use HTTP or HTTPS (HTTPS adds security) for Web Configurator sessions. Specify which zones allow Web Configurator access and from which IP address the access can come.

HTTPS (HyperText Transfer Protocol over Secure Socket Layer, or HTTP over SSL) is a web protocol that encrypts and decrypts web pages. Secure Socket Layer (SSL) is an application-level protocol that enables secure transactions of data by ensuring confidentiality (an unauthorized party cannot read the transferred data), authentication (one party can identify the other party) and data integrity (you know if data has been changed).

It relies upon certificates, public keys, and private keys (see Chapter 33 on page 403 for more information).

HTTPS on the ZyWALL is used so that you can securely access the ZyWALL using the Web Configurator. The SSL protocol specifies that the HTTPS server (the ZyWALL) must always authenticate itself to the HTTPS client (the computer which requests the HTTPS connection with the ZyWALL), whereas the HTTPS client only should authenticate itself when the HTTPS server requires it to do so (select **Authenticate Client Certificates** in the **WWW** screen). **Authenticate Client Certificates** is optional and if selected means the HTTPS client must send the ZyWALL a certificate. You must apply for a certificate for the browser from a CA that is a trusted CA on the ZyWALL.

Please refer to the following figure.

- 1 HTTPS connection requests from an SSL-aware web browser go to port 443 (by default) on the ZyWALL's web server.
- 2 HTTP connection requests from a web browser go to port 80 (by default) on the ZyWALL's web server.

Figure 303 HTTP/HTTPS Implementation



Note: If you disable **HTTP** in the **WWW** screen, then the ZyWALL blocks all HTTP connection attempts.

37.7.4 Configuring WWW Service Control

Click **Configuration > System > WWW** to open the **WWW** screen. Use this screen to specify from which zones you can access the ZyWALL using HTTP or HTTPS. You can also specify which IP addresses the access can come from.

Note: Admin Service Control deals with management access (to the Web Configurator). User Service Control deals with user access to the ZyWALL (logging into SSL VPN for example).

TTPS			
The second s			
Server Berti	442		
Server Port:	(0		
Authenticate Client Certificates	(See Irusted CAs)		
Server Certificate:	default 🗸		
📝 Redirect HTTP to HTTPS			
dmin Service Control			
🔕 Add 📝 Edit 🍵 Remove 🎒 M	ove		
# 🔺 Zone	Address	Action	
- ALL	ALL	accept	
4 4 Page 1 of 1 ▶ ▶ 9	50 🗸 items		Displaying 1 - 1 of
ser Service Control			
🔘 Add 📝 Edit 🏢 Remove 🔐 M	ove		
# . 7one	Address	Action	
- All	All	accent	
	Thomas 50 and items	uooopi	Displaying 1 - 1 of
TP Enable			Upplaying x x o
▼ Enable Server Port: dmin Service Control	80		
TP Enable Server Port: dmin Service Control Add	80 0V6		
TP Tenable Server Port: dmin Service Control Add Healt Remove → M # ▲ Zone	80 Address	Action	copicying a la c
 TP ✓ Enable Server Port: dmin Service Control ⊘ Add	80 Address ALL	Action accept	
TTP	80 Address ALL Show 50 v items	Action accept	Displaying 1 - 1 of
TTP Tenable Server Port: dmin Service Control Add C Edit Remove M M Zone - ALL ALL ALL ALL Service Control	80 Address ALL Show 50 v items	Action accept	Displaying 1 - 1 of
TTP Tenable Server Port: dmin Service Control Add Edit Remove M M Add Pedit Remove Service Control Add Edit Remove M M Service Control Add Edit Remove M M	80 Address ALL Show 50 v items	Action accept	Displaying 1 - 1 of
TTP Tenable Server Port: dmin Service Control Add Edit Remove M Cone ALL Page 1 of 1 Page 1 Ser Service Control Add Edit Remove M Cone Control Control Cone Cone Cone Cone Cone Cone Cone Cone	80 Address ALL Show 50 v items ove Address	Action accept Action	Displaying 1 - 1 of
TTP Tenable Server Port: dmin Service Control Add Edit Remove M M Zone - ALL ALL ALL ALL ALL ALL Cone - AL Cone - A	80 Address ALL Show 50 v items ove Address ALL	Action accept Action accept	Displaying 1 - 1 of
TTP Tenable Server Port: dmin Service Control Add Edit Remove M M Cone ALL A Page 1 of 1 Add Edit Remove M M Cone ALL A Page 1 of 1 Add A Page 1 A Ad	80 Address ALL Show 50 v items ALL Show 50 v items	Action accept Action accept	Displaying 1 - 1 of Displaying 1 - 1 of Displaying 1 - 1 of
TTP	80 80 Address ALL Show 50 v items ALL Show 50 v items	Action accept Action accept	Displaying 1 - 1 of Displaying 1 - 1 of
TTP Tenable Server Port: dmin Service Control Add Edit Remove M M Cone ALL A Page 1 of 1 Page 1 of 1 Page Add Page 1 of 1 Page 1 Page 1 Of 1 Page	80 Address ALL Show 50 v items Address ALL Show 50 v items	Action accept Action accept	Displaying 1 - 1 of Displaying 1 - 1 of

Figure 304 Configuration > System > WWW > Service Control

The following table describes the labels in this screen.

Table 184	Configuration	> System	> WWW	> Service Control	ol
-----------	---------------	----------	-------	-------------------	----

LABEL	DESCRIPTION
HTTPS	
Enable	Select the check box to allow or disallow the computer with the IP address that matches the IP address(es) in the Service Control table to access the ZyWALL Web Configurator using secure HTTPs connections.
Server Port	The HTTPS server listens on port 443 by default. If you change the HTTPS server port to a different number on the ZyWALL, for example 8443, then you must notify people who need to access the ZyWALL Web Configurator to use "https://ZyWALL IP Address: 8443 " as the URL.

LABEL	DESCRIPTION
Authenticate Client Certificates	Select Authenticate Client Certificates (optional) to require the SSL client to authenticate itself to the ZyWALL by sending the ZyWALL a certificate. To do that the SSL client must have a CA-signed certificate from a CA that has been imported as a trusted CA on the ZyWALL (see Section 37.7.7.5 on page 456 on importing certificates for details).
Server Certificate	Select a certificate the HTTPS server (the ZyWALL) uses to authenticate itself to the HTTPS client. You must have certificates already configured in the My Certificates screen.
Redirect HTTP to HTTPS	To allow only secure Web Configurator access, select this to redirect all HTTP connection requests to the HTTPS server.
Admin/User Service Control	Admin Service Control specifies from which zones an administrator can use HTTPS to manage the ZyWALL (using the Web Configurator). You can also specify the IP addresses from which the administrators can manage the ZyWALL.
	User Service Control specifies from which zones a user can use HTTPS to log into the ZyWALL (to log into SSL VPN for example). You can also specify the IP addresses from which the users can access the ZyWALL.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.
Move	To change an entry's position in the numbered list, select the method and click Move to display a field to type a number for where you want to put it and press [ENTER] to move the rule to the number that you typed.
#	This is the index number of the service control rule.
	The entry with a hyphen (-) instead of a number is the ZyWALL's (non-configurable) default policy. The ZyWALL applies this to traffic that does not match any other configured rule. It is not an editable rule. To apply other behavior, configure a rule that traffic will match so the ZyWALL will not have to use the default policy.
Zone	This is the zone on the ZyWALL the user is allowed or denied to access.
Address	This is the object name of the IP address(es) with which the computer is allowed or denied to access.
Action	This displays whether the computer with the IP address specified above can access the ZyWALL zone(s) configured in the Zone field (Accept) or not (Deny).
HTTP	
Enable	Select the check box to allow or disallow the computer with the IP address that matches the IP address(es) in the Service Control table to access the ZyWALL Web Configurator using HTTP connections.
Server Port	You may change the server port number for a service if needed, however you must use the same port number in order to use that service to access the ZyWALL.
Admin/User Service Control	Admin Service Control specifies from which zones an administrator can use HTTP to manage the ZyWALL (using the Web Configurator). You can also specify the IP addresses from which the administrators can manage the ZyWALL.
	User Service Control specifies from which zones a user can use HTTP to log into the ZyWALL (to log into SSL VPN for example). You can also specify the IP addresses from which the users can access the ZyWALL.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry.

 Table 184
 Configuration > System > WWW > Service Control (continued)

LABEL	DESCRIPTION
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.
Move	To change an entry's position in the numbered list, select the method and click Move to display a field to type a number for where you want to put it and press [ENTER] to move the rule to the number that you typed.
#	This is the index number of the service control rule.
	The entry with a hyphen (-) instead of a number is the ZyWALL's (non-configurable) default policy. The ZyWALL applies this to traffic that does not match any other configured rule. It is not an editable rule. To apply other behavior, configure a rule that traffic will match so the ZyWALL will not have to use the default policy.
Zone	This is the zone on the ZyWALL the user is allowed or denied to access.
Address	This is the object name of the IP address(es) with which the computer is allowed or denied to access.
Action	This displays whether the computer with the IP address specified above can access the ZyWALL zone(s) configured in the Zone field (Accept) or not (Deny).
Authentication	
Client Authentication Method	Select a method the HTTPS or HTTP server uses to authenticate a client.
	You must have configured the authentication methods in the Auth. method screen.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 184
 Configuration > System > WWW > Service Control (continued)

37.7.5 Service Control Rules

Click Add or Edit in the Service Control table in a WWW, SSH, Telnet, FTP or SNMP screen to add a service control rule.

|--|

🛅 Create new Object 🔻			
Address Object:	ALL	~	
Zone:	ALL	~	
Action:	Accept	*	
		ОК	Cancel
I LEAD NO NE LON			

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Create new Object	Use this to configure any new settings objects that you need to use in this screen.
Address Object	Select ALL to allow or deny any computer to communicate with the ZyWALL using this service.
	Select a predefined address object to just allow or deny the computer with the IP address that you specified to access the ZyWALL using this service.
Zone	Select ALL to allow or prevent any ZyWALL zones from being accessed using this service.
	Select a predefined ZyWALL zone on which a incoming service is allowed or denied.
Action	Select Accept to allow the user to access the ZyWALL from the specified computers.
	Select Deny to block the user's access to the ZyWALL from the specified computers.
ОК	Click OK to save your customized settings and exit this screen.
Cancel	Click Cancel to exit this screen without saving

 Table 185
 Configuration > System > Service Control Rule > Edit

37.7.6 Customizing the WWW Login Page

Click **Configuration > System > WWW > Login Page** to open the **Login Page** screen. Use this screen to customize the Web Configurator login screen. You can also customize the page that displays after an access user logs into the Web Configurator to access network services like the Internet. See Chapter 27 on page 361 for more on access user accounts.

ect Type			
Use Default Login P:	age		
Use Customized Log	gin Page		
Logo Filo			
To upload a logo file ((*,oif/ong/ipg), browse to the location of the file and then click Upload.		
File Path:	Select a file path Browse Upload		
ustomized Login	Page	My Device	Enter user Name/Password and cick to login.
Title:	My Device		Liser Name
TitleColor:	#378ec9 (CSS color code)	and the second	
Message Color:	black Color (CSS color code)		Password:
Note Message:			One-Time Password: (Optional)
Background (Picture	re maxium size: 438 x 337 px)		(max. 63 alphanumeric, printable characters and no spaces)
Picture	Select a file path Browse		Error Message
Color	#36b9d2 Color • (CSS color code)		
ustomized Acces	ss Page		
Title:	is Page You now have logged in.		
ustomized Acces Title: Message Color:	ss Page You now have logged in. black Color 🗙 (CSS color code)		
ustomized Acces Title: Message Color: Note Message:	Vau now have logged in. black (CSS color code) none	×	You now have logged in.
ustomized Acces Title: Message Color: Note Message: Background (Picture	s Page You now have logged in. black Color (CSS color code) none maxium size: 438 x 337 px)	×	You now have logged in. Click the logout button to terminate the access session.
ustomized Acces Title: Message Color: Note Message: Background (Picture Picture	s Page You now have logged in. black Color (CSS color code) none e maxum size: 438 x 337 px) Select a file path Drowse Liptosci	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after
ustomized Acces Title: Message Color: Note Message: Background (Picture Picture © Color	s Page You now have logged in. black Color * mone e maxtum size: 438 x 337 px) Select a file path Browseuw # 36b9d2 Color *	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew
ustomized Acces Title: Message Color: Note Message: Background (Picture Picture @ Color	s Page You now have logged in. black Color (CSS color code) none e maxum size: 438 x 337 px) Select a file path Browseuu #3669d2 Color * (CSS color code)	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew Remaining time before lease 23:03:39
ustomized Acces Title: Message Color: Note Message: Background (Picture Picture Color	ss Page You now have logged in. black Color * mone e maxium size: 436 x 337 px) Select a file path Browserve #36b9d2 Color *	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew Remaining time before lease 23:03:39 timeout (hh:mm:s): Bernamin time before auth
ustomized Acces Title: Message Color: Note Message: Background (Picture Picture Color	ss Page You now have logged in. black Color * mone reaxium size: 438 x 337 px) Select a file path Browsenn =36b9d2 Color *	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew Remaining time before lease 23:03:39 timeout (hh:mm:s): Remaining time before auth. timeout (hh:mm): timeout (hh:mm):
ustomized Acces Title: Message Color: Note Message: Background (Picture Picture @ Color	ss Page You now have logged in. black Color (CSS color code) none re maxim size: 436 x 337 px) Select a file path Browser, Libbod =36b9d2 Color (CSS color code)	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew Remaining time before lease 23:03:39 timeout (hhirmm:s): Remaining time before auth. timeout (hhirmm): none
ustomized Acces Title: Message Color: Note Message: Background (Picture © Picture © Color	se Page You now have logged in. black Color * Inone e maxium size: 436 x 337 px) Select a file path Bromseur #36b9d2 Color *	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew Remaining time before lease 23:03:39 timeout (hharmns): Remaining time before auth. timeout (hharmn): none
ustomized Acces Title: Message Color: Note Message: Background (Picture © Picture © Color	ss Page You now have logged in. black color none e maxium size: 436 x 337 px) Select a file path Browssur #36b9d2 Color (CSS color code)	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew Remaining time before lease 23:03:39 timeout (hhirmmss): Remaining time before auth. timeout (hhirmm): none
ustomized Acces Title: Message Color: Note Message: Background (Picture Picture Color	se Page	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew Remaining time before lease 23:03:39 timeout (hhirmmiss): Remaining time before auth. timeout (hhirmm): none
ustomized Acces Title: Message Color: Note Message: Background (Picture Picture Color	ss Page You now have logged in. black Color * (CSS color code) none re maxim size: 438 x 337 px) Select a file path Browser #3699d2 Color * (CSS color code)	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew Remaining time before lease 23:03:39 timeout (hharmiss): Remaining time before auth. timeout (hharmis):
ustomized Acces Title: Message Color: Note Message: Background (Picture ◎ Picture ◎ Color	se Page	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew Remaining time before lease 23:03:39 timeout (hharmiss): Remaining time before auth. timeout (hharmis): Imout his intervention the second
Title: Message Color: Note Message: Background (Pictur. Picture © Color	se Page	×	You now have logged in. Click the logout button to terminate the access session. You could renew your lease time by clicking the Renew button. For security reason you must login in again after User-defined lease time (max Renew Remaining time before lease 23:03:39 timeout (hharmiss): Remaining time before auth. timeout (hharmis): Imout hisming: none Logout

Figure 306 Configuration > System > WWW > Login Page

The following figures identify the parts you can customize in the login and access pages.

Logo Title	age Customization	_
My Device	Enter User Name/Password and click to login.	
PROPERTY.	User Name: Password:	Message (color of all text)
	One-Time Password: (Optional)	
	(max. 31 alphanumeric, printable characters and no spaces)	
	Error Message	- Background
	Login_ SSL VPN Ld	
	Note: 1. Turn on Javascript and Cookie setting in your web browser. 2. Turn off Popup Window Blocking in your web browser.	
	This is the note you can configure.	 Note Message (last line of text)

Figure 308 Access Page Customization



You can specify colors in one of the following ways:

- Click **Color** to display a screen of web-safe colors from which to choose.
- Enter the name of the desired color.

- Enter a pound sign (#) followed by the six-digit hexadecimal number that represents the desired color. For example, use "#000000" for black.
- Enter "rgb" followed by red, green, and blue values in parenthesis and separate by commas. For example, use "rgb(0,0,0)" for black.

Your desired color should display in the preview screen on the right after you click in another field, click **Apply**, or press [ENTER]. If your desired color does not display, your browser may not support it. Try selecting another color.

The following table describes the labels in the screen.

LABEL	DESCRIPTION
Select Type	Select whether the Web Configurator uses the default login screen or one that you customize in the rest of this screen.
Logo File	You can upload a graphic logo to be displayed on the upper left corner of the Web Configurator login screen and access page.
	Specify the location and file name of the logo graphic or click Browse to locate it.
	Note: Use a GIF, JPG, or PNG of 100 kilobytes or less.
	Click Upload to transfer the specified graphic file from your computer to the ZyWALL.
Customized Login Page	Use this section to set how the Web Configurator login screen looks.
Title	Enter the title for the top of the screen. Use up to 64 printable ASCII characters. Spaces are allowed.
Title Color	Specify the color of the screen's title text.
Message Color	Specify the color of the screen's text.
Note Message	Enter a note to display at the bottom of the screen. Use up to 64 printable ASCII characters. Spaces are allowed.
Background	Set how the screen background looks.
	To use a graphic, select Picture and upload a graphic. Specify the location and file name of the logo graphic or click Browse to locate it. The picture's size cannot be over 438 x 337 pixels.
	Note: Use a GIF, JPG, or PNG of 100 kilobytes or less.
	To use a color, select Color and specify the color.
Customized Access Page	Use this section to customize the page that displays after an access user logs into the Web Configurator to access network services like the Internet.
Title	Enter the title for the top of the screen. Use up to 64 printable ASCII characters. Spaces are allowed.
Message Color	Specify the color of the screen's text.
Note Message	Enter a note to display below the title. Use up to 64 printable ASCII characters. Spaces are allowed.

 Table 186
 Configuration > System > WWW > Login Page

LABEL	DESCRIPTION
Background	Set how the window's background looks.
	To use a graphic, select Picture and upload a graphic. Specify the location and file name of the logo graphic or click Browse to locate it. The picture's size cannot be over 438 x 337 pixels.
	Note: Use a GIF, JPG, or PNG of 100 kilobytes or less.
	To use a color, select Color and specify the color.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 186Configuration > System > WWW > Login Page

37.7.7 HTTPS Example

If you haven't changed the default HTTPS port on the ZyWALL, then in your browser enter "https:// ZyWALL IP Address/" as the web site address where "ZyWALL IP Address" is the IP address or domain name of the ZyWALL you wish to access.

37.7.7.1 Internet Explorer Warning Messages

When you attempt to access the ZyWALL HTTPS server, you will see the error message shown in the following screen.

	Figure 309	Security	Alert	Dialog	Box ((Internet	Explorer)
--	------------	----------	-------	--------	-------	-----------	----------	---



Select **Continue to this website** to proceed to the Web Configurator login screen. Otherwise, select **Click here to close this webpage** to block the access.

37.7.7.2 Mozilla Firefox Warning Messages

When you attempt to access the ZyWALL HTTPS server, a **The Connection is Untrusted** screen appears as shown in the following screen. Click **Technical Details** if you want to verify more information about the certificate from the ZyWALL.

Select I Understand the Risks and then click Add Exception to add the ZyWALL to the security exception list. Click Confirm Security Exception.



	This Connection is Untrusted
	You have asked Firefox to connect securely to 172.10.26.9 , but we can't confirm that your connection is secure.
	Normally, when you try to connect securely, sites will present trusted identification to prove that you are going to the right place. However, this site's identity can't be verified.
	What Should I Do?
	If you usually connect to this site without problems, this error could mean that someone is trying to impersonate the site, and you shouldn't continue.
	Get me out of here!
•	Technical Details
•	I Understand the Risks
Figure 31	1 Security Certificate 2 (Firefox)

Add Security Exception
You are about to override how Firefox identifies this site. Legitimate banks, stores, and other public sites will not ask you to do this.
Location: https://172.10.26.9/redirect.cgi?arip=172.10.26.9 Get Certificate
Certificate Status This site attempts to identify itself with invalid information. <u>View</u> Wrong Site
Certificate belongs to a different site, which could indicate an identity theft. Unknown Identity
Certificate is not trusted, because it hasn't been verified by a recognized authority.
Permanently store this exception
Confirm Security Exception Cancel

37.7.7.3 Avoiding Browser Warning Messages

Here are the main reasons your browser displays warnings about the ZyWALL's HTTPS server certificate and what you can do to avoid seeing the warnings:

- The issuing certificate authority of the ZyWALL's HTTPS server certificate is not one of the browser's trusted certificate authorities. The issuing certificate authority of the ZyWALL's factory default certificate is the ZyWALL itself since the certificate is a self-signed certificate.
- For the browser to trust a self-signed certificate, import the self-signed certificate into your operating system as a trusted certificate.
- To have the browser trust the certificates issued by a certificate authority, import the certificate authority's certificate into your operating system as a trusted certificate.

37.7.7.4 Login Screen

After you accept the certificate, the ZyWALL login screen appears. The lock displayed in the bottom of the browser status bar denotes a secure connection.



Figure 312 Login Screen (Internet Explorer)

37.7.7.5 Enrolling and Importing SSL Client Certificates

The SSL client needs a certificate if Authenticate Client Certificates is selected on the ZyWALL.

You must have imported at least one trusted CA to the ZyWALL in order for the **Authenticate Client Certificates** to be active (see the Certificates chapter for details).

Apply for a certificate from a Certification Authority (CA) that is trusted by the ZyWALL (see the ZyWALL's **Trusted CA** Web Configurator screen).



Figure 313 ZyWALL Trusted CA Screen

The CA sends you a package containing the CA's trusted certificate(s), your personal certificate(s) and a password to install the personal certificate(s).

37.7.7.5.1 Installing the CA's Certificate

1 Double click the CA's trusted certificate to produce a screen similar to the one shown next.

Figure 314 CA Certificate Example

🚞 Certifica	te Inform	nation			
This certificate	is intend	ed to:			
•Ensures the	e identity o	of a remote	computer		
 Proves you Ensures sol 	r identity t ftware can	to a remote the from soft	computer ware publis	her	
•Protects so	ftware fro	m alteration	after publi	cation	
Protects e- Ollows data	mail messa to be sign	iges and with the	current tim	•	-
Issued to:	CSO-CA				
Issued by:	CSO-CA				
Valid from	8/30/200:	3 to 8/30/2	2005		

2 Click Install Certificate and follow the wizard as shown earlier in this appendix.

37.7.7.5.2 Installing Your Personal Certificate(s)

You need a password in advance. The CA may issue the password or you may have to specify it during the enrollment. Double-click the personal certificate given to you by the CA to produce a screen similar to the one shown next

1 Click **Next** to begin the wizard.

Certificate Import Wizard		×
	Welcome to the Certificate Import Wizard This wizard helps you copy certificates, certificate trust lists, and certificate revocation lists from your disk to a certificate store. A certificate, which is issued by a certification authority, is a confirmation of your identity and contains information used to protect data or to establish secure network connections. A certificate store is the system area where certificates are kept. To continue, click Next.	
	< Back Next > Cancel	

Figure 315 Personal Certificate Import Wizard 1

2 The file name and path of the certificate you double-clicked should automatically appear in the **File name** text box. Click **Browse** if you wish to import a different certificate.

ifi	icate Import Wizard		
ile	e to Import		
	Specify the file you want to import.		
	Eile name:		
			Browse
	Note: More than one certificate can be	stored in a single file in th	e following formats
		case and all of sea they have all of	o roiowing roinides
	Personal Information Exchange- PKC	5#12 (.PFX,.P12)	is rollowing rolling.
	Personal Information Exchange- PKC Cryptographic Message Syntax Stand	5 #12 (.PFX,.P12) lard- PKCS #7 Certificate	s (.P7B)
	Personal Information Exchange- PKC Cryptographic Message Syntax Stand Microsoft Serialized Certificate Store (5 #12 (.PFX,.P12) lard- PKCS #7 Certificate (.SST)	is (.P78)
	Personal Information Exchange- PKC Cryptographic Message Syntax Stanc Microsoft Serialized Certificate Store (5 #12 (.PFX,.P12) dard- PKC5 #7 Certificate (.SST)	is (.P78)
	Personal Information Exchange- PKC Cryptographic Message Syntax Stand Microsoft Serialized Certificate Store (5 #12 (.PFX,.P12) Jard- PKCS #7 Certificate (.SST)	s (.P78)
	Personal Information Exchange- PKCS Cryptographic Message Syntax Stand Microsoft Serialized Certificate Store (5 #12 (.PFX,.P12) dard- PKCS #7 Certificate (.SST)	s (.P7B)
	Personal Information Exchange- PKCS Cryptographic Message Syntax Stanc Microsoft Serialized Certificate Store (5 #12 (.PFX,.P12) Jard- PKCS #7 Certificate (.SST)	s (.P78)

3 Enter the password given to you by the CA.

rtificate Import Wizard	
Password	
To maintain security, the private key was protected with a password.	
Type the password for the private key.	
Password:	
Enable strong private key protection. You will be	
prompted every time the private key is used by an application if you enable this option.	
Mark the private key as exportable	

Figure 317 Personal Certificate Import Wizard 3

4 Have the wizard determine where the certificate should be saved on your computer or select **Place all certificates in the following store** and choose a different location.

Figure 318 Personal Certificate Import Wizard 4

ertificate Store		
Certificate stores are system areas w	here certificates are kept.	
Windows can automatically select a ce	ertificate store, or you can specify a location	for
 Automatically select the certific 	cate store based on the type of certificate	
C Place all certificates in the follo	wing store	
Certificate store:		
	Browse,	a [

5 Click **Finish** to complete the wizard and begin the import process.

Certificate Import Wizard			×
	Completing the Wizard	Certificate Import	
<u> </u>	Certificate Store Selected Content File Name	Automatically determined by t PFX D:\Projects_2003-10\CPE2\cp	
	•		
	<u> </u>	Finish Cancel	

Figure 319 Personal Certificate Import Wizard 5

You should see the following screen when the certificate is correctly installed on your computer.
 Figure 320 Personal Certificate Import Wizard 6

i iguic c		i thr
Certificat	te Import Wizard	X
٩	The import was success	ful.
	OK	

37.7.7.6 Using a Certificate When Accessing the ZyWALL Example

Use the following procedure to access the ZyWALL via HTTPS.

1 Enter 'https://ZyWALL IP Address/ in your browser's web address field.

Figure	321	Acce	ss the	e Zy	WALL	Via	HTTPS				
🕘 ab	out:l	olank	- Micr	oso	ft Int	erne	et Explor	er			
Eile	Edit	⊻iew	Favor	ites	Tools	Help)				
] 🗲 Ba	ack 🔻	→ - (8	a	Q Sear	rch (😹 Favorites	GHistory	₿•	d i	500
] A <u>d</u> dr	ress	🔊 https:	//192.1	68.1.	1						

2 When **Authenticate Client Certificates** is selected on the ZyWALL, the following screen asks you to select a personal certificate to send to the ZyWALL. This screen displays even if you only have a single certificate as in the example.

- Client Authentication
 ? ×

 Identification
 Identification

 Select the certificate to use when connecting.
 Identification.

 testtls
 Identification.

 More Info...
 View Certificate...

 OK
 Cancel
- Figure 322 SSL Client Authentication

3 You next see the Web Configurator login screen.

Figure 323 Secure Web Configurator Login Screen

Password:		
One-Time Password:	(Optional)	
(max. 31 alphanumeric, printable cha	racters and no spaces)	
	Login SSL VPN	
🐧 Note:		
1. Turn on Javascript and Cookie setti	ng in your web browser.	
3. Turn on Java Runtime Environment	(JRE) in your web browser.	
		🖃 💋 Int

37.8 SSH

You can use SSH (Secure SHeII) to securely access the ZyWALL's command line interface. Specify which zones allow SSH access and from which IP address the access can come.

SSH is a secure communication protocol that combines authentication and data encryption to provide secure encrypted communication between two hosts over an unsecured network. In the following figure, computer A on the Internet uses SSH to securely connect to the WAN port of the ZyWALL for a management session.





37.8.1 How SSH Works

The following figure is an example of how a secure connection is established between two remote hosts using SSH v1.



Figure 325 How SSH v1 Works Example

1 Host Identification

The SSH client sends a connection request to the SSH server. The server identifies itself with a host key. The client encrypts a randomly generated session key with the host key and server key and sends the result back to the server.

The client automatically saves any new server public keys. In subsequent connections, the server public key is checked against the saved version on the client computer.

2 Encryption Method

Once the identification is verified, both the client and server must agree on the type of encryption method to use.

3 Authentication and Data Transmission

After the identification is verified and data encryption activated, a secure tunnel is established between the client and the server. The client then sends its authentication information (user name and password) to the server to log in to the server.

37.8.2 SSH Implementation on the ZyWALL

Your ZyWALL supports SSH versions 1 and 2 using RSA authentication and four encryption methods (AES, 3DES, Archfour, and Blowfish). The SSH server is implemented on the ZyWALL for management using port 22 (by default).

37.8.3 Requirements for Using SSH

You must install an SSH client program on a client computer (Windows or Linux operating system) that is used to connect to the ZyWALL over SSH.

37.8.4 Configuring SSH

Click **Configuration > System > SSH** to change your ZyWALL's Secure Shell settings. Use this screen to specify from which zones SSH can be used to manage the ZyWALL. You can also specify from which IP addresses the access can come.

Version 1 Server Port:	22	_		
Server Certificate:	default	~		
🗿 Add 📝 Edit 🍵	Remove 🎳 Move	4		
# 🔺	Zone	Address	Action	
	ALL	ALL	Accept	
	STATES AND A STATE AND A	Land Land Land Land Land Land Land Land	D:	1

Figure 326 Configuration > System > SSH

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Enable	Select the check box to allow or disallow the computer with the IP address that matches the IP address(es) in the Service Control table to access the ZyWALL CLI using this service.
Version 1	Select the check box to have the ZyWALL use both SSH version 1 and version 2 protocols. If you clear the check box, the ZyWALL uses only SSH version 2 protocol.
Server Port	You may change the server port number for a service if needed, however you must use the same port number in order to use that service for remote management.
Server Certificate	Select the certificate whose corresponding private key is to be used to identify the ZyWALL for SSH connections. You must have certificates already configured in the My Certificates screen (Click My Certificates and see Chapter 33 on page 403 for details).
Service Control	This specifies from which computers you can access which ZyWALL zones.

Table 187 Configuration > System > SSH

LABEL	DESCRIPTION
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry. Refer to Table 185 on page 450 for details on the screen that opens.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.
Move	To change an entry's position in the numbered list, select the method and click Move to display a field to type a number for where you want to put it and press [ENTER] to move the rule to the number that you typed.
#	This the index number of the service control rule.
Zone	This is the zone on the ZyWALL the user is allowed or denied to access.
Address	This is the object name of the IP address(es) with which the computer is allowed or denied to access.
Action	This displays whether the computer with the IP address specified above can access the ZyWALL zone(s) configured in the Zone field (Accept) or not (Deny).
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 187
 Configuration > System > SSH (continued)

37.8.5 Secure Telnet Using SSH Examples

This section shows two examples using a command interface and a graphical interface SSH client program to remotely access the ZyWALL. The configuration and connection steps are similar for most SSH client programs. Refer to your SSH client program user's guide.

37.8.5.1 Example 1: Microsoft Windows

This section describes how to access the ZyWALL using the Secure Shell Client program.

- 1 Launch the SSH client and specify the connection information (IP address, port number) for the ZyWALL.
- 2 Configure the SSH client to accept connection using SSH version 1.
- 3 A window displays prompting you to store the host key in you computer. Click **Yes** to continue.

Figure 327 SSH Example 1: Store Host Key

Host Identifi	cation	×
i	You are connecting to the host "192.168.1.1" for the first tim The host has provided you its identification, a host public key The fingerprint of the host public key is: "xevac-bycor-kubyz-dipah-ravut-fyduz-kazuk-goler-cavom-hil	e. 'ot-sexox''
	You can save the host key to the local database by clicking You can continue without saving the host key by clicking No You can also cancel the connection by clicking Cancel. Do you want to save the new host key to the local database	Yes. ?
	Yes No Cancel Help	

Enter the password to log in to the ZyWALL. The CLI screen displays next.

37.8.5.2 Example 2: Linux

This section describes how to access the ZyWALL using the OpenSSH client program that comes with most Linux distributions.

1 Test whether the SSH service is available on the ZyWALL.

Enter "telnet 192.168.1.1 22" at a terminal prompt and press [ENTER]. The computer attempts to connect to port 22 on the ZyWALL (using the default IP address of 192.168.1.1).

A message displays indicating the SSH protocol version supported by the ZyWALL.

Figure 328 SSH Example 2: Test

```
$ telnet 192.168.1.1 22
Trying 192.168.1.1...
Connected to 192.168.1.1.
Escape character is '^]'.
SSH-1.5-1.0.0
```

2 Enter "ssh -1 192.168.1.1". This command forces your computer to connect to the ZyWALL using SSH version 1. If this is the first time you are connecting to the ZyWALL using SSH, a message displays prompting you to save the host information of the ZyWALL. Type "yes" and press [ENTER].

Then enter the password to log in to the ZyWALL.

Figure 329 SSH Example 2: Log in

```
$ ssh -1 192.168.1.1
The authenticity of host '192.168.1.1 (192.168.1.1)' can't be established.
RSA1 key fingerprint is 21:6c:07:25:7e:f4:75:80:ec:af:bd:d4:3d:80:53:d1.
Are you sure you want to continue connecting (yes/no)? yes
Warning: Permanently added '192.168.1.1' (RSA1) to the list of known hosts.
Administrator@192.168.1.1's password:
```

3 The CLI screen displays next.

37.9 Telnet

You can use Telnet to access the ZyWALL's command line interface. Specify which zones allow Telnet access and from which IP address the access can come.

37.9.1 Configuring Telnet

Click **Configuration > System > TELNET** to configure your ZyWALL for remote Telnet access. Use this screen to specify from which zones Telnet can be used to manage the ZyWALL. You can also specify from which IP addresses the access can come.

Figure 330 Configuration > System > TELNET

Enable Server Port	23			
vice Control				
🔇 Add 📝 I	Edit 🍵 Remove 📣 Move			
# 🔺	Zone	Address	Action	
2	ALL	ALL	Accept	
📢 🔄 Page	e 1 of 1 🕨 🕨 Show 50	·▼ items	Disp	olaying 1 - 1 of 1

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Enable	Select the check box to allow or disallow the computer with the IP address that matches the IP address(es) in the Service Control table to access the ZyWALL CLI using this service.
Server Port	You may change the server port number for a service if needed, however you must use the same port number in order to use that service for remote management.
Service Control	This specifies from which computers you can access which ZyWALL zones.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry. Refer to Table 185 on page 450 for details on the screen that opens.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.
Move	To change an entry's position in the numbered list, select the method and click Move to display a field to type a number for where you want to put it and press [ENTER] to move the rule to the number that you typed.
#	This the index number of the service control rule.
	The entry with a hyphen (-) instead of a number is the ZyWALL's (non-configurable) default policy. The ZyWALL applies this to traffic that does not match any other configured rule. It is not an editable rule. To apply other behavior, configure a rule that traffic will match so the ZyWALL will not have to use the default policy.
Zone	This is the zone on the ZyWALL the user is allowed or denied to access.
Address	This is the object name of the IP address(es) with which the computer is allowed or denied to access.
Action	This displays whether the computer with the IP address specified above can access the ZyWALL zone(s) configured in the Zone field (Accept) or not (Deny).
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 188
 Configuration > System > TELNET

37.10 FTP

You can upload and download the ZyWALL's firmware and configuration files using FTP. To use this feature, your computer must have an FTP client. Please see Chapter 39 on page 488 for more information about firmware and configuration files.

37.10.1 Configuring FTP

To change your ZyWALL's FTP settings, click **Configuration > System > FTP** tab. The screen appears as shown. Use this screen to specify from which zones FTP can be used to access the ZyWALL. You can also specify from which IP addresses the access can come.

Figure 331 Configuration > System > FTP

Enable TIS required				
Server Port:	21			
rvice Control				
Add Z Edit	Temove Move	Address	Action	
	ALL	ALL	Accept	
4 4 Page 1	of 1 > > Show 50	🗸 items	Dis	playing 1 - 1 of

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Enable	Select the check box to allow or disallow the computer with the IP address that matches the IP address(es) in the Service Control table to access the ZyWALL using this service.
TLS required	Select the check box to use FTP over TLS (Transport Layer Security) to encrypt communication.
	This implements TLS as a security mechanism to secure FTP clients and/or servers.
Server Port	You may change the server port number for a service if needed, however you must use the same port number in order to use that service for remote management.
Server Certificate	Select the certificate whose corresponding private key is to be used to identify the ZyWALL for FTP connections. You must have certificates already configured in the My Certificates screen (Click My Certificates and see Chapter 33 on page 403 for details).
Service Control	This specifies from which computers you can access which ZyWALL zones.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry. Refer to Table 185 on page 450 for details on the screen that opens.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.

LABEL	DESCRIPTION
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.
Move	To change an entry's position in the numbered list, select the method and click Move to display a field to type a number for where you want to put it and press [ENTER] to move the rule to the number that you typed.
#	This the index number of the service control rule.
	The entry with a hyphen (-) instead of a number is the ZyWALL's (non-configurable) default policy. The ZyWALL applies this to traffic that does not match any other configured rule. It is not an editable rule. To apply other behavior, configure a rule that traffic will match so the ZyWALL will not have to use the default policy.
Zone	This is the zone on the ZyWALL the user is allowed or denied to access.
Address	This is the object name of the IP address(es) with which the computer is allowed or denied to access.
Action	This displays whether the computer with the IP address specified above can access the ZyWALL zone(s) configured in the Zone field (Accept) or not (Deny).
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 189
 Configuration > System > FTP (continued)

37.11 SNMP

Simple Network Management Protocol is a protocol used for exchanging management information between network devices. Your ZyWALL supports SNMP agent functionality, which allows a manager station to manage and monitor the ZyWALL through the network. The ZyWALL supports SNMP version one (SNMPv1) and version two (SNMPv2c). The next figure illustrates an SNMP management operation.




An SNMP managed network consists of two main types of component: agents and a manager.

An agent is a management software module that resides in a managed device (the ZyWALL). An agent translates the local management information from the managed device into a form compatible with SNMP. The manager is the console through which network administrators perform network management functions. It executes applications that control and monitor managed devices.

The managed devices contain object variables/managed objects that define each piece of information to be collected about a device. Examples of variables include such as number of packets received, node port status etc. A Management Information Base (MIB) is a collection of managed objects. SNMP allows a manager and agents to communicate for the purpose of accessing these objects.

SNMP itself is a simple request/response protocol based on the manager/agent model. The manager issues a request and the agent returns responses using the following protocol operations:

- Get Allows the manager to retrieve an object variable from the agent.
- GetNext Allows the manager to retrieve the next object variable from a table or list within an agent. In SNMPv1, when a manager wants to retrieve all elements of a table from an agent, it initiates a Get operation, followed by a series of GetNext operations.
- Set Allows the manager to set values for object variables within an agent.
- Trap Used by the agent to inform the manager of some events.

37.11.1 Supported MIBs

The ZyWALL supports MIB II that is defined in RFC-1213 and RFC-1215. The ZyWALL also supports private MIBs (zywall.mib and zyxel-zywall-ZLD-Common.mib) to collect information about CPU and memory usage and VPN total throughput. The focus of the MIBs is to let administrators collect

statistical data and monitor status and performance. You can download the ZyWALL's MIBs from www.zyxel.com.

37.11.2 SNMP Traps

The ZyWALL will send traps to the SNMP manager when any one of the following events occurs.

OBJECT LABEL	OBJECT ID	DESCRIPTION
Cold Start	1.3.6.1.6.3.1.1.5.1	This trap is sent when the ZyWALL is turned on or an agent restarts.
linkDown	1.3.6.1.6.3.1.1.5.3	This trap is sent when the Ethernet link is down.
linkUp	1.3.6.1.6.3.1.1.5.4	This trap is sent when the Ethernet link is up.
authenticationFailure	1.3.6.1.6.3.1.1.5.5	This trap is sent when an SNMP request comes from non-authenticated hosts.
vpnTunnelDisconnected	1.3.6.1.4.1.890.1.6.22.2.3	This trap is sent when an IPSec VPN tunnel is disconnected.
vpnTunnelName	1.3.6.1.4.1.890.1.6.22.2.2.1.1	This trap is sent along with the vpnTunnelDisconnected trap. This trap carries the disconnected tunnel's IPSec SA name.
vpnIKEName	1.3.6.1.4.1.890.1.6.22.2.2.1.2	This trap is sent along with the vpnTunnelDisconnected trap. This trap carries the disconnected tunnel's IKE SA name.
vpnTunnelSPI	1.3.6.1.4.1.890.1.6.22.2.2.1.3	This trap is sent along with the vpnTunnelDisconnected trap. This trap carries the security parameter index (SPI) of the disconnected VPN tunnel.

Table 190 SNMP Traps

37.11.3 Configuring SNMP

To change your ZyWALL's SNMP settings, click **Configuration > System > SNMP** tab. The screen appears as shown. Use this screen to configure your SNMP settings, including from which zones SNMP can be used to access the ZyWALL. You can also specify from which IP addresses the access can come.

Server Port:	161			
Get Community:	public			
Set Community:	private			
Trap:				
Community:		(Optional)		
Destination:		(Optional)		
	Zone	Address	Action	
	ALL	ALL	Accept	
	the second se			

Figure 333 Configuration > System > SNMP

LABEL	DESCRIPTION
Enable	Select the check box to allow or disallow the computer with the IP address that matches the IP address(es) in the Service Control table to access the ZyWALL using this service.
Server Port	You may change the server port number for a service if needed, however you must use the same port number in order to use that service for remote management.
Get Community	Enter the Get Community , which is the password for the incoming Get and GetNext requests from the management station. The default is public and allows all requests.
Set Community	Enter the Set community , which is the password for incoming Set requests from the management station. The default is private and allows all requests.
Тгар	
Community	Type the trap community, which is the password sent with each trap to the SNMP manager. The default is public and allows all requests.
Destination	Type the IP address of the station to send your SNMP traps to.
Service Control	This specifies from which computers you can access which ZyWALL zones.
Add	Click this to create a new entry. Select an entry and click Add to create a new entry after the selected entry. Refer to Table 185 on page 450 for details on the screen that opens.
Edit	Double-click an entry or select it and click Edit to be able to modify the entry's settings.
Remove	To remove an entry, select it and click Remove . The ZyWALL confirms you want to remove it before doing so. Note that subsequent entries move up by one when you take this action.
Move	To change an entry's position in the numbered list, select the method and click Move to display a field to type a number for where you want to put it and press [ENTER] to move the rule to the number that you typed.

Table 191	Configuration	> S'	vstem	>	SNMP
	connigaration	~)	yStorr	~	

LABEL	DESCRIPTION
#	This the index number of the service control rule.
	The entry with a hyphen (-) instead of a number is the ZyWALL's (non-configurable) default policy. The ZyWALL applies this to traffic that does not match any other configured rule. It is not an editable rule. To apply other behavior, configure a rule that traffic will match so the ZyWALL will not have to use the default policy.
Zone	This is the zone on the ZyWALL the user is allowed or denied to access.
Address	This is the object name of the IP address(es) with which the computer is allowed or denied to access.
Action	This displays whether the computer with the IP address specified above can access the ZyWALL zone(s) configured in the Zone field (Accept) or not (Deny).
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 191 Configuration > System > SNMP (continued)

37.12 Language Screen

Click **Configuration > System > Language** to open the following screen. Use this screen to select a display language for the ZyWALL's Web Configurator screens.

Figure 334 Configuration > System > Language

Language Setting		
Language Setting		
Language Setting:	English	*
		Apply

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Language Setting	Select a display language for the ZyWALL's Web Configurator screens. You also need to open a new browser session to display the screens in the new language.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 192
 Configuration > System > Language

37.13 IPv6 Screen

Click **Configuration > System > IPv6** to open the following screen. Use this screen to enable IPv6 support for the ZyWALL's Web Configurator screens. See the IPv6 Overview on page 106 for more information about IPv6.

Figure 335 Configuration > System > IPv6

IPv6	
Global Setting	
Enable IPv6	
	Apply
	Apply Reset

LABEL	DESCRIPTION
Enable IPv6	Select this to have the ZyWALL support IPv6 and make IPv6 settings be available on the screens that the functions support, such as the Configuration > Network > Interface > Ethernet , VLAN , and Bridge screens. The ZyWALL discards all IPv6 packets if you clear this check box.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

 Table 193
 Configuration > System > IPv6

Log and Report

38.1 Overview

Use these screens to configure daily reporting and log settings.

38.1.1 What You Can Do In this Chapter

- Use the **Email Daily Report** screen (Section 38.2 on page 474) to configure where and how to send daily reports and what reports to send.
- Use the Log Setting screens (Section 38.3 on page 476) to specify settings for recording log messages and alerts, e-mailing them, storing them on a connected USB storage device, and sending them to remote syslog servers.

38.2 Email Daily Report

Use the **Email Daily Report** screen to start or stop data collection and view various statistics about traffic passing through your ZyWALL.

Note: Data collection may decrease the ZyWALL's traffic throughput rate.

Click **Configuration > Log & Report > Email Daily Report** to display the following screen. Configure this screen to have the ZyWALL e-mail you system statistics every day.

General Settings			oport
📝 Enable Email Daily Report			
Email Settings			
Mail Server:	vvv	(Outgoing SMTP Server Name or IP	Address)
Mail Subject:		Append system name	Append date time
Mail From:	vvv@qq.mm	(Email Address)	
Mail To:	xxx@33.bb	(Email Address)	
		(Email Address)	
SMTP Authentication			
User Name :			
Password:			
Retype to Confirm:			
Send Report Now			
Schedule			
Time For Sending Report:	0 (hours)	0 (minutes)	
2 1	(load by	- (innoteby	
Report Items			
System Resource Usage			
CPU Usage			
Memory Usage Session Usage			
Port Usage			
Interface Traffic Statistics			
Reset counters after sending	eport successfully		
Reset All Counters			
		Apply Reset	
		Apply Reset	

Figure 336 Configuration > Log & Report > Email Daily Report

LABEL	DESCRIPTION
Enable Email Daily Report	Select this to send reports by e-mail every day.
Mail Server	Type the name or IP address of the outgoing SMTP server.
Mail Subject	Type the subject line for the outgoing e-mail. Select Append system name to add the ZyWALL's system name to the subject. Select Append date time to add the ZyWALL's system date and time to the subject.
Mail From	Type the e-mail address from which the outgoing e-mail is delivered. This address is used in replies.
Mail To	Type the e-mail address (or addresses) to which the outgoing e-mail is delivered.
SMTP Authentication	Select this check box if it is necessary to provide a user name and password to the SMTP server.
User Name	This box is effective when you select the SMTP Authentication check box. Type the user name to provide to the SMTP server when the log is e-mailed.

Table 194	Configuration >	> Log & Report >	Email Daily Report
-----------	-----------------	------------------	--------------------

LABEL	DESCRIPTION
Password	This box is effective when you select the SMTP Authentication check box. Type the password to provide to the SMTP server when the log is e-mailed.
Retype to Confirm	Type the password again to make sure that you have entered is correctly.
Send Report Now	Click this button to have the ZyWALL send the daily e-mail report immediately.
Time for sending report	Select the time of day (hours and minutes) when the log is e-mailed. Use 24-hour notation.
Report Items	Select the information to include in the report. Select Reset counters after sending report successfully if you only want to see statistics for a 24 hour period.
Reset All Counters	Click this to discard all report data and start all of the counters over at zero.
Apply	Click Apply to save your changes back to the ZyWALL.
Reset	Click Reset to return the screen to its last-saved settings.

Table 194 Configuration > Log & Report > Email Daily Report (continued)

38.3 Log Setting Screens

The **Log Setting** screens control log messages and alerts. A log message stores the information for viewing or regular e-mailing later, and an alert is e-mailed immediately. Usually, alerts are used for events that require more serious attention, such as system errors and attacks.

The ZyWALL provides a system log and supports e-mail profiles and remote syslog servers. View the system log in the **MONITOR > Log** screen. Use the e-mail profiles to mail log messages to the specific destinations. You can also have the ZyWALL store system logs on a connected USB storage device. The other four logs are stored on specified syslog servers.

The **Log Setting** screens control what information the ZyWALL saves in each log. You can also specify which log messages to e-mail for the system log, and where and how often to e-mail them. These screens also set for which events to generate alerts and where to email the alerts.

The first **Log Setting** screen provides a settings summary. Use the **Edit** screens to configure settings such as log categories, e-mail addresses, and server names for any log. Use the **Log Category Settings** screen to edit what information is included in the system log, USB storage, e-mail profiles, and remote servers.

38.3.1 Log Setting Summary

To access this screen, click Configuration > Log & Report > Log Setting.

2	Edit 💡 Acti	vate 🎯 Inactivate			
#	Status	Name	Log Format	Summary	
1	B	System Log	Internal	E-mail Server 1 Mail Server: Mail Subject: Send From: Send Log to: Send Alert to: Schedule: Send log when full.	
2	Ф	System Log	Internal	E-mail Server 2 Mail Server. Mail Subject: Send From: Send Log to: Send Alert to: Schedule: Send log when full.	
3	ଜ	USB Storage	Internal	USB Status: none	
4	P	Remote Server 1	VRPT/Syslog	Server Address: Log Facility: Local 1	
5	9	Remote Server 2	VRPT/Syslog	Server Address: Log Facility: Local 1	
6	9	Remote Server 3	VRPT/Syslog	Server Address: Log Facility: Local 1	
7	9	Remote Server 4	VRPT/Syslog	Server Address: Log Facility: Local 1	
14	A Page	1 of 1	Show 50 🗸 items		Displaying 1 - 7 of

Figure 337 Configuration > Log & Report > Log Setting

LABEL	DESCRIPTION
Edit	Double-click an entry or select it and click Edit to open a screen where you can modify it.
Activate	To turn on an entry, select it and click Activate.
Inactivate	To turn off an entry, select it and click Inactivate.
#	This field is a sequential value, and it is not associated with a specific log.
Name	This field displays the type of log setting entry (system log, logs stored on a USB storage device connected to the ZyWALL, or one of the remote servers).
Log Format	This field displays the format of the log.
	Internal - system log; you can view the log on the View Log tab.
	VRPT/Syslog - ZyXEL's Vantage Report, syslog-compatible format.
	CEF/Syslog - Common Event Format, syslog-compatible format.
Summary	This field is a summary of the settings for each log. Please see Section 38.3.2 on page 478 for more information.
Log Category Settings	Click this button to open the Log Category Settings Edit screen.
Apply	Click this button to save your changes (activate and deactivate logs) and make them take effect.

	a a 				a
Table 195	Configuration	> Loc	I & Report	> L(oa Settina

38.3.2 Edit System Log Settings

The **Log Settings Edit** screen controls the detailed settings for each log in the system log (which includes the e-mail profiles). Go to the **Log Settings Summary** screen (see Section 38.3.1 on page 476), and click the system log **Edit** icon.

ail Server 1			
Active			
Mail Server:		(Outgoing SMTP Server Name or IP	Address)
Mail Subject:			
Send From:		(E-Mail Address)	
Send Log to:		(E-Mai Address)	
Send Alerts to:		(E-Mail Address)	
Sending Log:	When Full	*	
Day for Sending Log:		v	
Time for Sending Logs			
SMTP Authentication			
User Name :			
Password:			
Retype to Confirm:			
ail Server 2			
Active			
Mail Server:		(Outgoing SMTP Server Name or IP	Address)
Mail Subject:			* 1
Send From:		(E-Mail Address)	
Send Log to:		(E-Mai Address)	
Send Alerts to:		(E-Mail Address)	
Sending Log:	When Full	×	
Day for Sending Loss	Tanto	×	
Time for Sending Logi			
SMTP Authentication		100	
User Name :			
Password:			
Retype to Confirm:			
ive Log and Alert			
🖞 System Log 🔹 🧰 E-mail Server 1 🕫	E-mail Server 2 -		
Loo Category	System Log	E-mail Server 1	E-mail Server 2
Account		2 U	
Auth. Policy	0.00		00 E3
Bulker Sacara	0.00		
Jue-n service	0.0		10 E
eww	0.00		
Cellular	0.0	100 EE	12 E
Connectivity Check	0.00		10 E
Daily Report	0.0		
Default		(2) (2)	12 E
Device KA	0.00		
DHCP		121 E1	
File Manager	0.0		
Frewal	0.00		
Force Authentication	0.00		2 11
KE	0.00		
Intround Load Relancing	0.00		100 ED
Interface	0 . 0		
eltertace	0.00		
#-MAC Binding	0 . 0		
PSec .	0.00	12 EL	12 E3
L2TP Over IPSec	0 . 0	(2) ED	
NAT	0.0		
PKI	0.0	2	R D
Policy Route	0.00		
Port Grouping	0.0		12 FT
Routing Protocol	0.0.0		
Sessions Limit	0.0.0		
SSL VPN	0.00		
System	0.00		
URB Storage	0.00		
	0.0	2 1	
	0.0		N 12
i Żyśłi	0 . 0		
< < Page 1 of 1 ≥ ≥ S	how 50 🛩 items		Displaying 1 - 30 of 30
Concolidation			
s meschanation			
Active	10	10.000	

Figure 338 Configuration > Log & Report > Log Setting > Edit (System Log)

Table 196	Configuration >	· Log 8	Report >	Log Setting	> Edit	(System	Log)
-----------	-----------------	---------	----------	-------------	--------	---------	------

LABEL	DESCRIPTION
E-Mail Server 1/2	
Active	Select this to send log messages and alerts according to the information in this section. You specify what kinds of log messages are included in log information and what kinds of log messages are included in alerts in the Active Log and Alert section.
Mail Server	Type the name or IP address of the outgoing SMTP server.
Mail Subject	Type the subject line for the outgoing e-mail.
Send From	Type the e-mail address from which the outgoing e-mail is delivered. This address is used in replies.
Send Log To	Type the e-mail address to which the outgoing e-mail is delivered.
Send Alerts To	Type the e-mail address to which alerts are delivered.
Sending Log	Select how often log information is e-mailed. Choices are: When Full, Hourly and When Full, Daily and When Full, and Weekly and When Full.
Day for Sending Log	This field is available if the log is e-mailed weekly. Select the day of the week the log is e-mailed.
Time for Sending Log	This field is available if the log is e-mailed weekly or daily. Select the time of day (hours and minutes) when the log is e-mailed. Use 24-hour notation.
SMTP Authentication	Select this check box if it is necessary to provide a user name and password to the SMTP server.
User Name	This box is effective when you select the SMTP Authentication check box. Type the user name to provide to the SMTP server when the log is e-mailed.
Password	This box is effective when you select the SMTP Authentication check box. Type the password to provide to the SMTP server when the log is e-mailed.
Retype to Confirm	Type the password again to make sure that you have entered is correctly.
Active Log and Alert	
System Log	Use the System Log drop-down list to change the log settings for all of the log categories.
	disable all logs (red X) - do not log any information for any category for the system log or e-mail any logs to e-mail server 1 or 2.
	enable normal logs (green check mark) - create log messages and alerts for all categories for the system log. If e-mail server 1 or 2 also has normal logs enabled, the ZyWALL will e-mail logs to them.
	enable normal logs and debug logs (yellow check mark) - create log messages, alerts, and debugging information for all categories. The ZyWALL does not e-mail debugging information, even if this setting is selected.
E-mail Server 1	Use the E-Mail Server 1 drop-down list to change the settings for e-mailing logs to e-mail server 1 for all log categories.
	Using the System Log drop-down list to disable all logs overrides your e-mail server 1 settings.
	enable normal logs (green check mark) - e-mail log messages for all categories to e-mail server 1.
	enable alert logs (red exclamation point) - e-mail alerts for all categories to e-mail server 1.

LABEL	DESCRIPTION
E-mail Server 2	Use the E-Mail Server 2 drop-down list to change the settings for e-mailing logs to e-mail server 2 for all log categories.
	Using the System Log drop-down list to disable all logs overrides your e-mail server 2 settings.
	enable normal logs (green check mark) - e-mail log messages for all categories to e-mail server 2.
	enable alert logs (red exclamation point) - e-mail alerts for all categories to e-mail server 2.
#	This field is a sequential value, and it is not associated with a specific address.
Log Category	This field displays each category of messages. It is the same value used in the Display and Category fields in the View Log tab. The Default category includes debugging messages generated by open source software.
System log	Select which events you want to log by Log Category. There are three choices:
	disable all logs (red X) - do not log any information from this category
	enable normal logs (green check mark) - create log messages and alerts from this category
	enable normal logs and debug logs (yellow check mark) - create log messages, alerts, and debugging information from this category; the ZyWALL does not e-mail debugging information, however, even if this setting is selected.
E-mail Server 1	Select whether each category of events should be included in the log messages when it is e-mailed (green check mark) and/or in alerts (red exclamation point) for the e-mail settings specified in E-Mail Server 1 . The ZyWALL does not e-mail debugging information, even if it is recorded in the System log .
E-mail Server 2	Select whether each category of events should be included in log messages when it is e-mailed (green check mark) and/or in alerts (red exclamation point) for the e-mail settings specified in E-Mail Server 2 . The ZyWALL does not e-mail debugging information, even if it is recorded in the System log .
Log Consolidation	
Active	Select this to activate log consolidation. Log consolidation aggregates multiple log messages that arrive within the specified Log Consolidation Interval . In the View Log tab, the text "[count= x]", where x is the number of original log messages, is appended at the end of the Message field, when multiple log messages were aggregated.
Log Consolidation Interval	Type how often, in seconds, to consolidate log information. If the same log message appears multiple times, it is aggregated into one log message with the text "[count= x]", where x is the number of original log messages, appended at the end of the Message field.
ОК	Click this to save your changes and return to the previous screen.
Cancel	Click this to return to the previous screen without saving your changes.

 Table 196
 Configuration > Log & Report > Log Setting > Edit (System Log) (continued)

38.3.3 Edit Log on USB Storage Setting

The **Edit Log on USB Storage Setting** screen controls the detailed settings for saving logs to a connected USB storage device. Go to the **Log Setting Summary** screen (see Section 38.3.1 on page 476), and click the USB storage **Edit** icon.

	unlicate logs to USB storage (if ready)			
tive		The file list of daily archives is in MAINTENANCE	-	
	Selection -	>Diagnostic->System Log		
-		S	electi	ion
#	Log Category		0	
1	Account	۲	0	0
2	Auth. Policy	۲	0	0
3	Built-in Service	۲	0	Ø
4	BWM	۲	0	0
5	Cellular	۲	0	0
6	Connectivity Check	۲	0	0
7	Daily Report	۲	0	0
8	Default	۲	0	0
9	Device HA	۲	0	O
10	DHCP	۲	0	0
11	File Manager	۲	0	0
12	Firewall	0	0	0
13	Force Authentication	۲	0	0
14	IKE	٢	0	0
15	Inbound Load Balancing	۲	0	0
16	Interface	۲	0	0
17	IP-MAC Binding	۲	0	0
18	IPSec	۲	0	0
19	L2TP Over IPSec	۲	0	0
20	NAT	۲	0	0
21	РКІ	۲	0	0
22	Policy Route	۲	0	0
23	Port Grouping	۲	0	0
24	Routing Protocol	۲	0	0
25	Sessions Limit	٢	0	0
26	SSL VPN	0	0	0
27	System	0	0	0
28	USB Storage	0	0	0
29	User	0	0	0
30	ZySH	0	0	0
14	✓ Page 1 of 1 ▶ ▶ Show 50	▼ items		Displaying 1 - 30 of 30

Figure 339 Configuration > Log & Report > Log Setting > Edit (USB Storage)

LABEL	DESCRIPTION
Duplicate logs to USB storage (if ready)	Select this to have the ZyWALL save a copy of its system logs to a connected USB storage device. Use the Active Log section to specify what kinds of messages to include.
Active Log	
Selection	Use the Selection drop-down list to change the log settings for all of the log categories.
	disable all logs (red X) - do not send the remote server logs for any log category.
	enable normal logs (green check mark) - send the remote server log messages and alerts for all log categories.
	enable normal logs and debug logs (yellow check mark) - send the remote server log messages, alerts, and debugging information for all log categories.
#	This field is a sequential value, and it is not associated with a specific entry.
Log Category	This field displays each category of messages. The Default category includes debugging messages generated by open source software.
Selection	Select what information you want to log from each Log Category (except All Logs; see below). Choices are:
	disable all logs (red X) - do not log any information from this category
	enable normal logs (green check mark) - log regular information and alerts from this category
	enable normal logs and debug logs (yellow check mark) - log regular information, alerts, and debugging information from this category
ОК	Click this to save your changes and return to the previous screen.
Cancel	Click this to return to the previous screen without saving your changes.

 Table 197
 Configuration > Log & Report > Log Setting > Edit (USB Storage)

38.3.4 Edit Remote Server Log Settings

The **Log Settings Edit** screen controls the detailed settings for each log in the remote server (syslog). Go to the **Log Settings Summary** screen (see Section 38.3.1 on page 476), and click a remote server **Edit** icon.

g se	ttings for Remote Server				
A	lictive				
Lo	og Format:	VRPT/Syslog	_		
Se	erver Address:	The second second	(Server Name or IP A	ddres	ss)
Lo	og Facility:	Local 1 💌			
tive	Log				
D s	Selection -				
#	Log Category		S	elect	ion Ø
	Account		0	0	0
	Auth. Policy		۲	0	0
	Built-in Service		0	0	0
	BWM		0	0	0
	Cellular		٢	0	0
	Connectivity Check			0	0
	Daily Report			0	0
	Default		0	0	
	Device HA		0	0	0
0	DHCP			0	0
1	File Manager			0	0
2	Firewall		0	0	0
2	Force Authentication		۲	O	0
2	Force Authentication		۲	0	0
4	IKE		۲	0	0
5	Inbound Load Balancing		۲	0	0
6	Interface		۲	0	0
7	Interface Statistics		۲	0	0
В	IP-MAC Binding		۲	0	0
9	IPSec		۲	0	0
D	L2TP Over IPSec		۲	0	0
1	NAT		۲	0	0
2	РКІ		۲	0	0
3	Policy Route		۲	0	0
4	Port Grouping		۲	0	0
5	Routing Protocol		۲	0	0
6	Sessions Limit		۲	0	0
7	SSL VPN		۲	0	0
в	System		۲	0	0
9	System Monitoring		٢	0	0
D	Traffic Log		0	0	0
1	USB Storage			0	0
2	User			0	0
3	ZySH			0	
1.4	d Bass I of the bill Ch	au 50 - itama	•	0	Displaying 1 - 33 of 33

Figure 340 Configuration > Log & Report > Log Setting > Edit (Remote Server)

LABEL	DESCRIPTION
Log Settings for Remote Server	
Active	Select this check box to send log information according to the information in this section. You specify what kinds of messages are included in log information in the Active Log section.
Log Format	This field displays the format of the log information. It is read-only.
	VRPT/Syslog - ZyXEL's Vantage Report, syslog-compatible format.
	CEF/Syslog - Common Event Format, syslog-compatible format.
Server Address	Type the server name or the IP address of the syslog server to which to send log information.
Log Facility	Select a log facility. The log facility allows you to log the messages to different files in the syslog server. Please see the documentation for your syslog program for more information.
Active Log	
Selection	Use the Selection drop-down list to change the log settings for all of the log categories.
	disable all logs (red X) - do not send the remote server logs for any log category.
	enable normal logs (green check mark) - send the remote server log messages and alerts for all log categories.
	enable normal logs and debug logs (yellow check mark) - send the remote server log messages, alerts, and debugging information for all log categories.
#	This field is a sequential value, and it is not associated with a specific address.
Log Category	This field displays each category of messages. It is the same value used in the Display and Category fields in the View Log tab. The Default category includes debugging messages generated by open source software.
Selection	Select what information you want to log from each Log Category (except All Logs; see below). Choices are:
	disable all logs (red X) - do not log any information from this category
	enable normal logs (green check mark) - log regular information and alerts from this category
	enable normal logs and debug logs (yellow check mark) - log regular information, alerts, and debugging information from this category
ОК	Click this to save your changes and return to the previous screen.
Cancel	Click this to return to the previous screen without saving your changes.

Table 198 Configuration > Log & Report > Log Setting > Edit (Remote Server)

38.3.5 Log Category Settings Screen

The **Log Category Settings** screen allows you to view and to edit what information is included in the system log, USB storage, e-mail profiles, and remote servers at the same time. It does not let you change other log settings (for example, where and how often log information is e-mailed or remote server names). To access this screen, go to the **Log Settings Summary** screen (see Section 38.3.1 on page 476), and click the **Log Category Settings** button.

Figure 341 Log Category Settings

35	System Log 🗸 🗋	USB Sto	orage	• 🖂	E-mail (Serve	r 1-	🔀 E-mai	il Server 2	2• ■ R	emote Se	erver 1	• 🗉	Rem	ote Sei	ver 2	·	Remo	te Se	rver 3	1	Rem	ote Server
	Log Category	S	/stem	Log	USE	Stor	age	E-mail S E-l	Server 1 Mail	E-mail S E-I	Gerver 2 Mail	Remo	ite Se Sysloj	rver 1 g	Remo	te Ser Syslog	ver 2	Remo	te Se Syslog	rver 3	Remo	te Se Syslo	erver 4 g
	Account	0	0	0	0	0	0					0	0	0	0	0	0	0	0	0		0	0
	Auth, Policy	0	•	0		0	0		1000			•	0	0	•	0	0		0	0	•	0	0
	Built-in Service			0		0	0						0	0	•	0	0		0	0	•	0	0
	BWM	- 0		0	۲	0	0	V		V			0	0	•	0	0	۲	0	0		0	0
	Cellular	0	0	0	0	0	0			V		0	0	0	0	0	0	0	0	0	0	0	0
	Connectivity C	0		0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	•	0	0
	Daily Report		۲	0		0	0			V			0	0	•	0	0	۲	0	0	0	0	0
	Default	0	۲	0	۲	0	0					•	0	0	۲	0	0	۲	0	0	•	0	0
	Device HA	0	0	۲	۲	0	0	V				۲	0	0	۲	0	0	۲	0	0	۲	0	0
	DHCD	0	۲	0	۲	0	0	V				۲	0	0	۲	0	0	۲	0	0	۲	0	0
	Eile Massacc	0	۲	0	۲	0	0	V				۲	0	0	۲	0	0	۲	0	0	۲	0	0
	File Manager	0	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	۲	0	0
	Firewall	0	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	۲	0	0
	Force Authent	u ()	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	۲	0	0
	IKE	0	۲	0	۲	0	0	V		V		0	0	0	۲	0	0	۲	0	0	۲	0	0
	Inbound Load	0	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	۲	0	0
	Interface	0	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	۲	0	0
	Interface Stati		۲	0	0	0	0					۲	0	0	۲	0	0	۲	0	0	۲	0	0
i.	IP-MAC Bindin	9 0	۲	0	۲	0	0	V	1000	V		0	0	0	۲	0	0	۲	0	0	۲	0	0
<u>(</u>	IPSec	0	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	۲	0	0
2	IPSEC	0	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	0	0	0
	L2TP Over IPS	···· ©	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	۲	0	0
	NAT	0	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	۲	0	0
	PKI	0	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	0	0	۲	0	0
	Policy Route	0	۲	0	0	0	0	V		1		۲	0	0	۲	0	0	۲	0	0	۲	0	0
e.	Port Grouping	0	۲	0	۲	0	0	V		V		۲	0	O	۲	0	0	۲	0	0	۲	0	0
2	Routing Protoc	ol O	0	0	۲	0	0	V				۲	0	0	۲	0	0	۲	0	0	۲	0	0
	Sessions Limit	t ©	۲	0	۲	0	0	V		V		۲	0	O	۲	0	0	۲	0	0	۲	0	0
	SSL VPN	0	0	0	0	0	0	V		1		0	0	0	۲	0	0	۲	0	0	۲	0	0
	System	0	۲	0	۲	0	0	V				۲	0	0	۲	0	0	۲	0	0	۲	0	0
E.	System Monito	· O	0	0	O	0	0					۲	0	0	۲	0	0	۲	0	0	۲	0	0
	Traffic Log	0	0	0	0	0	0					۲	0	0	۲	0	0	۲	0	0	۲	0	0
	USB Storage	0	۲	0	۲	0	0	V		1		۲	0	0	۲	0	0	۲	0	0	۲	0	0
	User	0	۲	0	۲	0	0	V		V		۲	0	0	۲	0	0	۲	O	0	۲	0	0
	ZySH	0	0	0	۲	0	0	V		V		۲	O	0	۲	0	O	۲	0	0	۲	0	Ø
1	Page 1	of 1	Þ Þ	l sh	ow 50)	item	าร												Dis	olayin	g 1 -	33 of 33

This screen provides a different view and a different way of indicating which messages are included in each log and each alert. Please see Section 38.3.2 on page 478, where this process is discussed. (The **Default** category includes debugging messages generated by open source software.)

LABEL	DESCRIPTION					
System Log	Use the System Log drop-down list to change the log settings for all of the log categories.					
	disable all logs (red X) - do not log any information for any category for the system log or e-mail any logs to e-mail server 1 or 2.					
	enable normal logs (green check mark) - create log messages and alerts for all categories for the system log. If e-mail server 1 or 2 also has normal logs enabled, the ZyWALL will e-mail logs to them.					
	enable normal logs and debug logs (yellow check mark) - create log messages, alerts, and debugging information for all categories. The ZyWALL does not e-mail debugging information, even if this setting is selected.					
USB Storage	Use the USB Storage drop-down list to change the log settings for saving logs to a connected USB storage device.					
	disable all logs (red X) - do not log any information for any category to a connected USB storage device.					
	enable normal logs (green check mark) - create log messages and alerts for all categories and save them to a connected USB storage device.					
	enable normal logs and debug logs (yellow check mark) - create log messages, alerts, and debugging information for all categories and save them to a connected USB storage device.					
E-mail Server 1	Use the E-Mail Server 1 drop-down list to change the settings for e-mailing logs to e-mail server 1 for all log categories.					
	Using the System Log drop-down list to disable all logs overrides your e-mail server 1 settings.					
	enable normal logs (green check mark) - e-mail log messages for all categories to e-mail server 1.					
	enable alert logs (red exclamation point) - e-mail alerts for all categories to e-mail server 1.					
E-mail Server 2	Use the E-Mail Server 2 drop-down list to change the settings for e-mailing logs to e-mail server 2 for all log categories.					
	Using the System Log drop-down list to disable all logs overrides your e-mail server 2 settings.					
	enable normal logs (green check mark) - e-mail log messages for all categories to e-mail server 2.					
	enable alert logs (red exclamation point) - e-mail alerts for all categories to e-mail server 2.					
Remote Server 1~4	For each remote server, use the Selection drop-down list to change the log settings for all of the log categories.					
	disable all logs (red X) - do not send the remote server logs for any log category.					
	enable normal logs (green check mark) - send the remote server log messages and alerts for all log categories.					
	enable normal logs and debug logs (yellow check mark) - send the remote server log messages, alerts, and debugging information for all log categories.					
#	This field is a sequential value, and it is not associated with a specific address.					
Log Category	This field displays each category of messages. It is the same value used in the Display and Category fields in the View Log tab. The Default category includes debugging messages generated by open source software.					

 Table 199
 Configuration > Log & Report > Log Setting > Log Category Settings

LABEL	DESCRIPTION
System Log	Select which events you want to log by Log Category. There are three choices:
	disable all logs (red X) - do not log any information from this category
	enable normal logs (green check mark) - create log messages and alerts from this category
	enable normal logs and debug logs (yellow check mark) - create log messages, alerts, and debugging information from this category; the ZyWALL does not e-mail debugging information, however, even if this setting is selected.
USB Storage	Select which event log categories to save to a connected USB storage device. There are three choices:
	disable all logs (red X) - do not log any information from this category
	enable normal logs (green check mark) - save log messages and alerts from this category
	enable normal logs and debug logs (yellow check mark) - save log messages, alerts, and debugging information from this category.
E-mail Server 1 E-mail	Select whether each category of events should be included in the log messages when it is e- mailed (green check mark) and/or in alerts (red exclamation point) for the e-mail settings specified in E-Mail Server 1 . The ZyWALL does not e-mail debugging information, even if it is recorded in the System log .
E-mail Server 2 E-mail	Select whether each category of events should be included in log messages when it is e- mailed (green check mark) and/or in alerts (red exclamation point) for the e-mail settings specified in E-Mail Server 2 . The ZyWALL does not e-mail debugging information, even if it is recorded in the System log .
Remote Server 1~4	For each remote server, select what information you want to log from each Log Category (except All Logs; see below). Choices are:
	disable all logs (red X) - do not log any information from this category
	enable normal logs (green check mark) - log regular information and alerts from this category
	enable normal logs and debug logs (yellow check mark) - log regular information, alerts, and debugging information from this category
ОК	Click this to save your changes and return to the previous screen.
Cancel	Click this to return to the previous screen without saving your changes.

 Table 199
 Configuration > Log & Report > Log Setting > Log Category Settings (continued)

File Manager

39.1 Overview

Configuration files define the ZyWALL's settings. Shell scripts are files of commands that you can store on the ZyWALL and run when you need them. You can apply a configuration file or run a shell script without the ZyWALL restarting. You can store multiple configuration files and shell script files on the ZyWALL. You can edit configuration files or shell scripts in a text editor and upload them to the ZyWALL. Configuration files use a .conf extension and shell scripts use a .zysh extension.

39.1.1 What You Can Do in this Chapter

- Use the **Configuration File** screen (see Section 39.2 on page 490) to store and name configuration files. You can also download configuration files from the ZyWALL to your computer and upload configuration files from your computer to the ZyWALL.
- Use the **Firmware Package** screen (see Section 39.3 on page 494) to check your current firmware version and upload firmware to the ZyWALL.
- Use the **Shell Script** screen (see Section 39.4 on page 496) to store, name, download, upload and run shell script files.

39.1.2 What you Need to Know

Configuration Files and Shell Scripts

When you apply a configuration file, the ZyWALL uses the factory default settings for any features that the configuration file does not include. When you run a shell script, the ZyWALL only applies the commands that it contains. Other settings do not change.

These files have the same syntax, which is also identical to the way you run CLI commands manually. An example is shown below.

Figure 342 Configuration File / Shell Script: Example

```
# enter configuration mode
configure terminal
# change administrator password
username admin password 4321 user-type admin
# configure ge3
interface ge3
ip address 172.23.37.240 255.255.255.0
ip gateway 172.23.37.254 metric 1
exit
# create address objects for remote management / to-ZyWALL firewall rules
# use the address group in case we want to open up remote management later
address-object TW_SUBNET 172.23.37.0/24
object-group address TW TEAM
address-object TW_SUBNET
exit
# enable Telnet access (not enabled by default, unlike other services)
ip telnet server
# open WAN-to-ZyWALL firewall for TW_TEAM for remote management
firewall WAN ZyWALL insert 4
sourceip TW TEAM
service TELNET
action allow
exit
write
```

While configuration files and shell scripts have the same syntax, the ZyWALL applies configuration files differently than it runs shell scripts. This is explained below.

Table 200	Configuration	Files and	Shell	Scripts	in the	ZyWALL
-----------	---------------	-----------	-------	---------	--------	--------

Configuration Files (.conf)	Shell Scripts (.zysh)
 Resets to default configuration. Goes into CLI Configuration mode. Runs the commands in the configuration file. 	Goes into CLI Privilege mode.Runs the commands in the shell script.

You have to run the example in Figure 342 on page 489 as a shell script because the first command is run in **Privilege** mode. If you remove the first command, you have to run the example as a configuration file because the rest of the commands are executed in **Configuration** mode.

Comments in Configuration Files or Shell Scripts

In a configuration file or shell script, use "#" or "!" as the first character of a command line to have the ZyWALL treat the line as a comment.

Your configuration files or shell scripts can use "exit" or a command line consisting of a single "!" to have the ZyWALL exit sub command mode.

Note: "exit" or "!'" must follow sub commands if it is to make the ZyWALL exit sub command mode.

1

Line 3 in the following example exits sub command mode.

interface ge1 ip address dhcp !

Lines 1 and 3 in the following example are comments and line 4 exits sub command mode.

```
interface ge1
# this interface is a DHCP client
!
```

Lines 1 and 2 are comments. Line 5 exits sub command mode.

```
! this is from Joe
# on 2008/04/05
interface ge1
ip address dhcp
!
```

Errors in Configuration Files or Shell Scripts

When you apply a configuration file or run a shell script, the ZyWALL processes the file line-by-line. The ZyWALL checks the first line and applies the line if no errors are detected. Then it continues with the next line. If the ZyWALL finds an error, it stops applying the configuration file or shell script and generates a log.

You can change the way a configuration file or shell script is applied. Include setenv stop-onerror off in the configuration file or shell script. The ZyWALL ignores any errors in the configuration file or shell script and applies all of the valid commands. The ZyWALL still generates a log for any errors.

39.2 The Configuration File Screen

Click **Maintenance > File Manager > Configuration File** to open the **Configuration File** screen. Use the **Configuration File** screen to store, run, and name configuration files. You can also download configuration files from the ZyWALL to your computer and upload configuration files from your computer to the ZyWALL.

Once your ZyWALL is configured and functioning properly, it is highly recommended that you back up your configuration file before making further configuration changes. The backup configuration file will be useful in case you need to return to your previous settings.

Configuration File Flow at Restart

- If there is not a **startup-config.conf** when you restart the ZyWALL (whether through a management interface or by physically turning the power off and back on), the ZyWALL uses the **system-default.conf** configuration file with the ZyWALL's default settings.
- If there is a **startup-config.conf**, the ZyWALL checks it for errors and applies it. If there are no errors, the ZyWALL uses it and copies it to the **lastgood.conf** configuration file as a back up file. If there is an error, the ZyWALL generates a log and copies the **startup-config.conf** configuration file to the **startup-config-bad.conf** configuration file and tries the existing **lastgood.conf** configuration file. If there isn't a **lastgood.conf** configuration file or it also has an error, the ZyWALL applies the **system-default.conf** configuration file.
- You can change the way the **startup-config.conf** file is applied. Include the setenv-startup stop-on-error off command. The ZyWALL ignores any errors in the **startup-config.conf** file and applies all of the valid commands. The ZyWALL still generates a log for any errors.

۲.R	ename 🎁 Remove 💾 Download 👔 Copy	> Apply		
#	File Name	Size	Last Modified	
	lastgood.conf	16567	2012-06-14 14:26:05	
2	startup-config.conf	16567	2012-06-14 14:30:27	
6	system-default.conf	14278	2012-06-05 23:16:28	
ŧ.	htm-default.conf	20	2012-06-05 23:16:28	
5	startup-config-bad.conf	15493	2011-10-27 19:28:28	
5	startup-config-back.conf	15566	1970-01-01 08:00:12	
14	Page 1 of 1 P Show 50	▼ items		Displaying 1 - 6 of

Figure 343 Maintenance > File Manager > Configuration File

Do not turn off the ZyWALL while configuration file upload is in progress.

Table 201	Maintenance :	> File Manager	> Configuration File
	munitoriunec /		> 00111901011110

LABEL	DESCRIPTION
Rename	Use this button to change the label of a configuration file on the ZyWALL. You can only rename manually saved configuration files. You cannot rename the lastgood.conf , system-default.conf and startup-config.conf files.
	You cannot rename a configuration file to the name of another configuration file in the ZyWALL.
	Click a configuration file's row to select it and click Rename to open the Rename File screen.
	Figure 344 Maintenance > File Manager > Configuration File > Rename
	E Rename ? X
	Source file : htm-default.conf Target file :
	OK Cancel
	Specify the new name for the configuration file. Use up to 25 characters (including a-zA-Z0-9;'~!@# $\%^{0}_{,.}=$).
	Click OK to save the duplicate or click Cancel to close the screen without saving a duplicate of the configuration file.
Remove	Click a configuration file's row to select it and click Remove to delete it from the ZyWALL. You can only delete manually saved configuration files. You cannot delete the system- default.conf , startup-config.conf and lastgood.conf files.
	A pop-up window asks you to confirm that you want to delete the configuration file. Click OK to delete the configuration file or click Cancel to close the screen without deleting the configuration file.
Download	Click a configuration file's row to select it and click Download to save the configuration to your computer.
Сору	Use this button to save a duplicate of a configuration file on the ZyWALL.
	Click a configuration file's row to select it and click Copy to open the Copy File screen.
	Figure 345 Maintenance > File Manager > Configuration File > Copy
	Copy File
	Source file : htm-default.conf
	Target file :
	OK Cancel
	Specify a name for the duplicate configuration file. Use up to 25 characters (including a-zA-ZO-9; '~ $!@#$ %^&()_+[]{}',.=-).
	Click OK to save the duplicate or click Cancel to close the screen without saving a duplicate of the configuration file.

LABEL	DESCRIPTION
Apply	Use this button to have the ZyWALL use a specific configuration file.
	Click a configuration file's row to select it and click Apply to have the ZyWALL use that configuration file. The ZyWALL does not have to restart in order to use a different configuration file, although you will need to wait for a few minutes while the system reconfigures.
	The following screen gives you options for what the ZyWALL is to do if it encounters an error in the configuration file.
	Figure 346 Maintenance > File Manager > Configuration File > Apply
	> Apply Configuration File
	Apply Configuration File
	File Name: system-default.conf
	If applying the configuration file encounters an error: Immediately stop applying the configuration file
	Immediately stop applying the configuration file and roll back to the previous configuration
	Ignore errors and finish applying the configuration file
	Ignore errors and finish applying the configuration file and then roll back to the previous configuration
	OK Cancel
	Immediately stop applying the configuration file - this is not recommended because it would leave the rest of the configuration blank. If the interfaces were not configured before the first error, the console port may be the only way to access the device.
	Immediately stop applying the configuration file and roll back to the previous configuration - this gets the ZyWALL started with a fully valid configuration file as quickly as possible.
	Ignore errors and finish applying the configuration file - this applies the valid parts of the configuration file and generates error logs for all of the configuration file's errors. This lets the ZyWALL apply most of your configuration and you can refer to the logs for what to fix.
	Ignore errors and finish applying the configuration file and then roll back to the previous configuration - this applies the valid parts of the configuration file, generates error logs for all of the configuration file's errors, and starts the ZyWALL with a fully valid configuration file.
	Click OK to have the ZyWALL start applying the configuration file or click Cancel to close the screen
#	This column displays the number for each configuration file entry. This field is a sequential value, and it is not associated with a specific address. The total number of configuration files that you can save depends on the sizes of the configuration files and the available flash storage space.

 Table 201
 Maintenance > File Manager > Configuration File (continued)

LABEL	DESCRIPTION
File Name	This column displays the label that identifies a configuration file.
	You cannot delete the following configuration files or change their file names.
	The system-default.conf file contains the ZyWALL's default settings. Select this file and click Apply to reset all of the ZyWALL settings to the factory defaults. This configuration file is included when you upload a firmware package.
	The startup-config.conf file is the configuration file that the ZyWALL is currently using. If you make and save changes during your management session, the changes are applied to this configuration file. The ZyWALL applies configuration changes made in the Web Configurator to the configuration file when you click Apply or OK . It applies configuration changes made via commands when you use the write command.
	The lastgood.conf is the most recently used (valid) configuration file that was saved when the device last restarted. If you upload and apply a configuration file with an error, you can apply lastgood.conf to return to a valid configuration.
Size	This column displays the size (in KB) of a configuration file.
Last Modified	This column displays the date and time that the individual configuration files were last changed or saved.
Upload Configuration	The bottom part of the screen allows you to upload a new or previously saved configuration file from your computer to your ZyWALL
File	You cannot upload a configuration file named system-default.conf or lastgood.conf.
	If you upload startup-config.conf , it will replace the current configuration and immediately apply the new settings.
File Path	Type in the location of the file you want to upload in this field or click Browse to find it.
Browse	Click Browse to find the .conf file you want to upload. The configuration file must use a ".conf" filename extension. You will receive an error message if you try to upload a fie of a different format. Remember that you must decompress compressed (.zip) files before you can upload them.
Upload	Click Upload to begin the upload process. This process may take up to two minutes.

 Table 201
 Maintenance > File Manager > Configuration File (continued)

39.3 The Firmware Package Screen

Click **Maintenance > File Manager > Firmware Package** to open the **Firmware Package** screen. Use the **Firmware Package** screen to check your current firmware version and upload firmware to the ZyWALL.

Note: The Web Configurator is the recommended method for uploading firmware. You only need to use the command line interface if you need to recover the firmware. See the CLI Reference Guide for how to determine if you need to recover the firmware and how to recover it.

Find the firmware package at www.zyxel.com in a file that (usually) uses the system model name with a .bin extension, for example, "zywall.bin".

The firmware update can take up to five minutes. Do not turn off or reset the ZyWALL while the firmware update is in progress!

Configuration File	Firmware Package Shell Script
Version	
Boot Module:	1.03
Current Version:	V3.10(AAAA.0)Preb7-r38094
Released Date:	2013-01-24 06:27:52
Upload File	
To upload firmware	, browse to the location of the file (*.bin) and then click Upload.
File Path:	Browse) Upload

Figure 347 Maintenance > File Manager > Firmware Package

The following table describes the labels in this screen.

 Table 202
 Maintenance > File Manager > Firmware Package

LABEL	DESCRIPTION
Boot Module	This is the version of the boot module that is currently on the ZyWALL.
Current Version	This is the firmware version and the date created.
Released Date	This is the date that the version of the firmware was created.
File Path	Type in the location of the file you want to upload in this field or click Browse to find it.
Browse	Click Browse to find the .bin file you want to upload. Remember that you must decompress compressed (.zip) files before you can upload them.
Upload	Click Upload to begin the upload process. This process may take up to two minutes.

After you see the **Firmware Upload in Process** screen, wait two minutes before logging into the ZyWALL again.

Figure 348 Firmware Upload In Process

Please	wait
Firmwa	are upload is in progress
	and apload to in programmin
io.	-
1	

Note: The ZyWALL automatically reboots after a successful upload.

The ZyWALL automatically restarts causing a temporary network disconnect. In some operating systems, you may see the following icon on your desktop.

Figure 349 Network

Local Area Connection	
) 10:44

After five minutes, log in again and check your new firmware version in the **Dashboard** screen.

If the upload was not successful, the following message appears in the status bar at the bottom of the screen.

Figure 350	Firmware I	Jpload Erro
Error Message	×	
errno: -42007 errmsg: Firmwa	e content error!	
	ж	

39.4 The Shell Script Screen

Use shell script files to have the ZyWALL use commands that you specify. Use a text editor to create the shell script files. They must use a ".zysh" filename extension.

Click **Maintenance** > **File Manager** > **Shell Script** to open the **Shell Script** screen. Use the **Shell Script** screen to store, name, download, upload and run shell script files. You can store multiple shell script files on the ZyWALL at the same time.

Note: You should include write commands in your scripts. If you do not use the write command, the changes will be lost when the ZyWALL restarts. You could use multiple write commands in a long script.

Figure 351 Maintenance > File Manager > Shell Script

I	Rename 💼 Remove 🔡	Download 🌓 Copy	> Apply	
#	File Name	Size	Last Modified	
14	4 Page 1 of 1 🕨	▶ Show 50	r items	No data to display
oload	l Shell Script			

Each field is described in the following table.

	DESCRIPTION
Rename	Use this button to change the label of a shell script file on the ZyWALL.
	You cannot rename a shell script to the name of another shell script in the ZVWALL
	Click a shell script's row to select it and click Dename to open the Dename File screen
	Firme 250 Maintenance – File Manager – Chall Carint – Danama
	Figure 352 Maintenance > File Manager > Shell Script > Rename Rename ? X
	Source file : wiz-VPN-2.zysh Target file : OK Cancel
	Specify the new name for the shell script file. Use up to 25 characters (including a-zA-Z0-9;'~!@#\$%^&()_+[]{}',.=-).
	Click OK to save the duplicate or click Cancel to close the screen without saving a duplicate of the configuration file.
Remove	Click a shell script file's row to select it and click Remove to delete the shell script file from the ZyWALL.
	A pop-up window asks you to confirm that you want to delete the shell script file. Click OK to delete the shell script file or click Cancel to close the screen without deleting the shell script file.
Download	Click a shell script file's row to select it and click Download to save the configuration to your computer.
Сору	Use this button to save a duplicate of a shell script file on the ZyWALL.
	Click a shell script file's row to select it and click Copy to open the Copy File screen.
	Figure 353 Maintenance > File Manager > Shell Script > Copy Copy File Source file : wiz-VPN-2.zysh Target file : OK Cancel
	Specify a name for the duplicate file. Use up to 25 characters (including a-zA-Z0-9; '~!@# $\%^{0}_{,.=}$).
	Click OK to save the duplicate or click Cancel to close the screen without saving a duplicate of the configuration file.
Apply	Use this button to have the ZyWALL use a specific shell script file.
	Click a shell script file's row to select it and click Apply to have the ZyWALL use that shell script file. You may need to wait awhile for the ZyWALL to finish applying the commands.
#	This column displays the number for each shell script file entry.
File Name	This column displays the label that identifies a shell script file.
Size	This column displays the size (in KB) of a shell script file.
Last Modified	This column displays the date and time that the individual shell script files were last changed or saved.

 Table 203
 Maintenance > File Manager > Shell Script

LABEL	DESCRIPTION
Upload Shell Script	The bottom part of the screen allows you to upload a new or previously saved shell script file from your computer to your ZyWALL.
File Path	Type in the location of the file you want to upload in this field or click Browse to find it.
Browse	Click Browse to find the .zysh file you want to upload.
Upload	Click Upload to begin the upload process. This process may take up to several minutes.

 Table 203
 Maintenance > File Manager > Shell Script (continued)

Diagnostics

40.1 Overview

Use the diagnostics screens for troubleshooting.

40.1.1 What You Can Do in this Chapter

- Use the **Diagnostics** screen (see Section 40.2 on page 499) to generate a file containing the ZyWALL's configuration and diagnostic information if you need to provide it to customer support during troubleshooting.
- Use the **Packet Capture** screens (see Section 40.3 on page 501) to capture packets going through the ZyWALL.
- Use the **Core Dump** screens (see Section 40.4 on page 504) to have the ZyWALL save a process's core dump to an attached USB storage device if the process terminates abnormally (crashes) so you can send the file to customer support for troubleshooting.
- Use the **System Log** screens (see Section 40.5 on page 505) to download files of system logs from a connected USB storage device to your computer.

40.2 The Diagnostic Screen

The **Diagnostic** screen provides an easy way for you to generate a file containing the ZyWALL's configuration and diagnostic information. You may need to send this file to customer support for troubleshooting.

Click Maintenance > Diagnostics to open the Diagnostic screen.

Figure 354 Maintenance > Diagnostics

Diagnostics	Packet Capture	Core Dump	System Log		111	
Collect	Files					
Diagnostic In	formation Collector					
Filename:	diaginfo-20	120614.tar.bz2				
Last Modified	2012-06-14	17:48:20				
Size:	1999 KB					
Copy the	diagnostic file to USB	storane (if readv)			
				(100)		Now Download
				Appl	y Collect	Now Download

Table 204 Maintenance >	Diagnostics
LABEL	DESCRIPTION
Filename	This is the name of the most recently created diagnostic file.
Last modified	This is the date and time that the last diagnostic file was created. The format is yyyy-mm-dd hh:mm:ss.
Size	This is the size of the most recently created diagnostic file.
Copy the diagnostic file to USB storage (if ready)	Select this to have the ZyWALL create an extra copy of the diagnostic file to a connected USB storage device.
Apply	Click Apply to save your changes.
Collect Now	Click this to have the ZyWALL create a new diagnostic file. Wait while information is collected. Diagnostic Information Collecting Status Please wait, collecting Cancel Cancel Diagnostic Information Collecting Status Done the collection. CK
Download	Click this to save the most recent diagnostic file to a computer.

Table 204	Maintenance	> Diagnostics
-----------	-------------	---------------

40.2.1 The Diagnostics Files Screen

Click Maintenance > Diagnostics > Files to open the diagnostic files screen. This screen lists the files of diagnostic information the ZyWALL has collected and stored in a connected USB storage device. You may need to send these files to customer support for troubleshooting.

Figure 355 Maintenance > Diagnostics > Files

Collect Files	torage		
d archives in USB	torage		
a archives in oob			
	welood		
Kentove 🕒 Do			
# File Name		Size Last Modif	fied

Table 205 Maintenance > Diagnostics > F	Files
---	-------

LABEL	DESCRIPTION
Remove	Select files and click Remove to delete them from the ZyWALL. Use the [Shift] and/or [Ctrl] key to select multiple files. A pop-up window asks you to confirm that you want to delete.
Download	Click a file to select it and click Download to save it to your computer.

LABEL	DESCRIPTION		
#	This column displays the number for each file entry. The total number of files that you can save depends on the file sizes and the available storage space.		
File Name	This column displays the label that identifies the file.		
Size	This column displays the size (in bytes) of a file.		
Last Modified	This column displays the date and time that the individual files were saved.		

Table 205 Maintenance > Diagnostics > Files (continued)

40.3 The Packet Capture Screen

Use this screen to capture network traffic going through the ZyWALL's interfaces. Studying these packet captures may help you identify network problems. Click **Maintenance > Diagnostics > Packet Capture** to open the packet capture screen.

Note: New capture files overwrite existing files of the same name. Change the **File Suffix** field's setting to avoid this.

Diagnostics	Packet Capture	Core Dump	System Log		
Capture	Files				
Interfaces					
Available	Interfaces		Capture Int	erfaces	
ge1		*			
ge2					
ge3					
ge4					
ge5		-			
205					
ilter					
IP Version:		any		~	
Protocol Ty	pe:	any		~	
Host IP:		any	any		
Host Port:		0	0 ((
Continu	ously capture and overw	vrite old ones	(B)		
Save da	ta to USB storage (servi	ce deactivated)	10)		
Captured Pa	acket Files:	10		MB	
Split thresh	old:	2		MB	
Duration:		0	0 (0: unlimited))
File Suffix:		-pad	-packet-capture		
Number Of	Bytes To Capture (Per Pa	acket): 1500		Bytes	

Figure 356 Maintenance > Diagnostics > Packet Capture

LABEL	DESCRIPTION		
Interfaces	Enabled interfaces (except for virtual interfaces) appear under Available Interfaces . Select interfaces for which to capture packets and click the right arrow button to move them to the Capture Interfaces list. Use the [Shift] and/or [Ctrl] key to select multiple objects.		
IP Version	Select the version of IP for which to capture packets. Select any to capture packets for all IP versions.		
Protocol Type	Select the protocol of traffic for which to capture packets. Select any to capture packets for all types of traffic.		
Host IP	Select a host IP address object for which to capture packets. Select any to capture packets for all hosts. Select User Defined to be able to enter an IP address.		
Host Port	This field is configurable when you set the IP Type to any , tcp , or udp . Specify the port number of traffic to capture.		
Continuously capture and overwrite old ones	Select this to have the ZyWALL keep capturing traffic and overwriting old packet capture entries when the available storage space runs out.		
Save data to onboard storage only	Select this to have the ZyWALL only store packet capture entries on the ZyWALL. The available storage size is displayed as well.		
	Note: The ZyWALLL reserves some onboard storage space as a buffer.		
Save data to USB storage	Select this to have the ZyWALL store packet capture entries only on a USB storage device connected to the ZyWALL if the ZyWALL allows this.		
	Status:		
	Unused - the connected USB storage device was manually unmounted by using the Remove Now button or for some reason the ZyWALL cannot mount it.		
	none - no USB storage device is connected.		
	service deactivated - USB storage feature is disabled (in Configuration > Object > USB Storage), so the ZyWALL cannot use a connected USB device to store system logs and other diagnostic information.		
	available - you can have the ZyWALL use the USB storage device. The available storage capacity also displays.		
	Note: The ZyWALL reserves some USB storage space as a buffer.		
Captured Packet Files	When saving packet captures only to the ZyWALL's onboard storage, specify a maximum limit in megabytes for the total combined size of all the capture files on the ZyWALL.		
	When saving packet captures to a connected USB storage device, specify a maximum limit in megabytes for each capture file.		
	Note: If you have existing capture files and have not selected the Continuously capture and overwrite old ones option, you may need to set this size larger or delete existing capture files.		
	The valid range depends on the available onboard/USB storage size. The ZyWALL stops the capture and generates the capture file when either the file reaches this size or the time period specified in the Duration field expires.		
Split threshold	Specify a maximum size limit in megabytes for individual packet capture files. After a packet capture file reaches this size, the ZyWALL starts another packet capture file.		

 Table 206
 Maintenance > Diagnostics > Packet Capture

LABEL	DESCRIPTION
Duration	Set a time limit in seconds for the capture. The ZyWALL stops the capture and generates the capture file when either this period of time has passed or the file reaches the size specified in the File Size field. 0 means there is no time limit.
File Suffix	Specify text to add to the end of the file name (before the dot and filename extension) to help you identify the packet capture files. Modifying the file suffix also avoids making new capture files that overwrite existing files of the same name.
	The file name format is "interface name-file suffix.cap", for example "vlan2-packet-capture.cap".
Number Of Bytes To Capture (Per Packet)	Specify the maximum number of bytes to capture per packet. The ZyWALL automatically truncates packets that exceed this size. As a result, when you view the packet capture files in a packet analyzer, the actual size of the packets may be larger than the size of captured packets.
Capture	Click this button to have the ZyWALL capture packets according to the settings configured in this screen.
	You can configure the ZyWALL while a packet capture is in progress although you cannot modify the packet capture settings.
	The ZyWALL's throughput or performance may be affected while a packet capture is in progress.
	After the ZyWALL finishes the capture it saves a separate capture file for each selected interface. The total number of packet capture files that you can save depends on the file sizes and the available flash storage space. Once the flash storage space is full, adding more packet captures will fail.
Stop	Click this button to stop a currently running packet capture and generate a separate capture file for each selected interface.
Reset	Click this button to return the screen to its last-saved settings.

 Table 206
 Maintenance > Diagnostics > Packet Capture (continued)

40.3.1 The Packet Capture Files Screen

Click **Maintenance > Diagnostics > Packet Capture > Files** to open the packet capture files screen. This screen lists the files of packet captures stored on the ZyWALL or a connected USB storage device. You can download the files to your computer where you can study them using a packet analyzer (also known as a network or protocol analyzer) such as Wireshark.

Figure 357 Maintenance > Diagnostics > Packet Capture > Files

gnostics Packet Captur	re Core Dump System Log		
apture Files			
ptured Packet Files			
👕 Remove 🔚 Download			
# File Name		Size	Last Modified
1 wan1packet-capture.00000.cap		367548	2010-11-24 01:50:31
2 wan1packet-capture.	bxt	81	2010-11-24 01:50:31
🕅 🗐 🛛 Page 1 of 1	Show 50 v items		Displaying 1 -
ptured Packet Files in USB	storage		
# File Name		Size	Last Modified
			12 A. 1

LABEL	DESCRIPTION
Remove	Select files and click Remove to delete them from the ZyWALL or the connected USB storage device. Use the [Shift] and/or [Ctrl] key to select multiple files. A pop-up window asks you to confirm that you want to delete.
Download	Click a file to select it and click Download to save it to your computer.
#	This column displays the number for each packet capture file entry. The total number of packet capture files that you can save depends on the file sizes and the available flash storage space.
File Name	This column displays the label that identifies the file. The file name format is interface name-file suffix.cap.
Size	This column displays the size (in bytes) of a configuration file.
Last Modified	This column displays the date and time that the individual files were saved.

Table 207 Maintenance > Diagnostics > Packet Capture > Files

40.4 Core Dump Screen

Use the **Core Dump** screen to have the ZyWALL save a process's core dump to an attached USB storage device if the process terminates abnormally (crashes). You may need to send this file to customer support for troubleshooting.

Click Maintenance > Diagnostics > Core Dump to open the following screen.

Figure 358 Maintenance > Diagnostics > Core Dump

Diagnostics	Packet Capture	Core Dump	System Log
Settings	Files		
Core Dump			
Save co	re dump to USB stora	ge (if ready)	
	A	oply Rese	:t

 Table 208
 Maintenance > Diagnostics > Core Dump

LABEL	DESCRIPTION
Save core dump to USB storage (if ready)	Select this to have the ZyWALL save a process's core dump to an attached USB storage device if the process terminates abnormally (crashes). If you clear this option the ZyWALL only saves
Apply	Click Apply to save the changes.
Reset	Click Reset to return the screen to its last-saved settings.
40.4.1 Core Dump Files Screen

Click **Maintenance > Diagnostics > Core Dump > Files** to open the core dump files screen. This screen lists the core dump files stored on the ZyWALL or a connected USB storage device. You may need to send these files to customer support for troubleshooting.

Figure 359 Maintenance > Diagnostics > Core Dump > Files

iagno	stics	Packet Capture	Core Dump	System Log	A DECEMBER OF
Setti	ings	Files			
ore	dump f	iles in system spac	e		
1	Remove	🔚 Download			
#	File	lame		Size	Last Modified
1	2010	-07-06-07-21-06-sle	ep.core.zip	27069	2010-07-06 07:21:06
2	2010	-07-06-07-22-50-ssl	hipsecpm.core.zip	179528	2010-07-06 07:22:52
14	4 P	age 1 of 1 🕨	Show 50 🗸	items	Displaying 1 - 1 of
ore (dump f	iles in USB storage	2		
#	File	lame		Size	Last Modified
1	2010	-07-06-07-22-50-ssl	hipsecpm.core.zip	179528	2010-07-06 07:22:53
14			Chow 50	itome	No data ta dirah

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Remove	Select files and click Remove to delete them from the ZyWALL. Use the [Shift] and/or [Ctrl] key to select multiple files. A pop-up window asks you to confirm that you want to delete.
Download	Click a file to select it and click Download to save it to your computer.
#	This column displays the number for each packet capture file entry. The total number of packet capture files that you can save depends on the file sizes and the available flash storage space.
File Name	This column displays the label that identifies the file.
Size	This column displays the size (in bytes) of a file.
Last Modified	This column displays the date and time that the individual files were saved.

Table 209 Maintenance > Diagnostics > Core Dump > Files

40.5 The System Log Screen

Click **Maintenance > Diagnostics > System Log** to open the system log files screen. This screen lists the files of system logs stored on a connected USB storage device. The files are in comma separated value (csv) format. You can download them to your computer and open them in a tool like Microsoft's Excel.

Figure 360 Maintenance > Diagnostics > System Log

st Modified

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Remove	Select files and click Remove to delete them from the ZyWALL. Use the [Shift] and/or [Ctrl] key to select multiple files. A pop-up window asks you to confirm that you want to delete.
Download	Click a file to select it and click Download to save it to your computer.
#	This column displays the number for each file entry. The total number of files that you can save depends on the file sizes and the available storage space.
File Name	This column displays the label that identifies the file.
Size	This column displays the size (in bytes) of a file.
Last Modified	This column displays the date and time that the individual files were saved.

Table 210	Mainte	nance	>	Diagnostics	>	System	Log

Packet Flow Explore

41.1 Overview

Use this to get a clear picture on how the ZyWALL determines where to forward a packet and how to change the source IP address of the packet according to your current settings. This function provides you a summary of all your routing and SNAT settings and helps troubleshoot any related problems.

41.1.1 What You Can Do in this Chapter

- Use the **Routing Status** screen (see Section 41.2 on page 507) to view the overall routing flow and each routing function's settings.
- Use the **SNAT Status** screen (see Section 41.3 on page 511) to view the overall source IP address conversion (SNAT) flow and each SNAT function's settings.

41.2 The Routing Status Screen

The **Routing Status** screen allows you to view the current routing flow and quickly link to specific routing settings. Click a function box in the **Routing Flow** section, the related routes (activated) will display in the **Routing Table** section. To access this screen, click **Maintenance** > **Packet Flow Explore**.

The order of the routing flow may vary depending on whether you:

- select use policy route to override direct route in the CONFIGURATION > Network > Routing > Policy Route screen.
- use policy routes to control 1-1 NAT by using the policy control-virtual-server-rules activate command.
- select use policy routes to control dynamic IPSec rules in the CONFIGURATION > VPN > IPSec VPN > VPN Connection screen.

Note: Once a packet matches the criteria of a routing rule, the ZyWALL takes the corresponding action and does not perform any further flow checking.

•	Direct Policy Route	1-1 SNAT SiteToSite VPN	Dynamic VPN Static- Dynamic Route	Default WAN Trunk	Out		
xuting Table							
	ote:	route. C - directly Connected C	- OSPE derived R - RIP derived	G - selected Gateway I - rejec	t B - Black hole I - Loon		
ags:	ote: : A - Activated route, S - Static Destination	route, C - directly Connected C Gateway) - OSPF derived, R - RIP derived	l, G - selected Gateway ! - rejec Metric	tt, B - Black hole, L - Loop. Flags	Persist	
No ags:	ote: A - Activated route, S - Static Destination 127.0.0.0/8	route, C - directly Connected C Gateway 0.0.0.0	D - OSPF derived, R - RIP derived Interface Io	l, G - selected Gateway I - rejec Metric 0	tt, B - Black hole, L - Loop. Flags ACG	Persist	
ng Ng ags:	ote: A - Activated route, S - Static Destination 127.0.0.0/8 192.168.1.0/24	route, C - directly Connected C Gateway 0.0.0.0 0.0.0 0	D - OSPF derived, R - RIP derived Interface Io Rian1	l, G - selected Gateway ! - rejec Metric 0	tt, B - Black hole, L - Loop. Flags ACG ACG	Persist -	
NC lags:	te: A - Activated route, S - Static Destination 127.0.0.0/8 192.168 1.0/24 192.168.2.0/24	route, C - directly Connected C Gateway 0.0.0.0 0.0.0 0.0.0	OSPF derived, R - RIP derived Interface Io alan1 alan2	I, G - selected Gateway ! - rejec Metric 0 0	tt, B - Black hole, L - Loop. Flags ACG ACG ACG	Persist - -	









In Direct Policy Route SiteToSite	Dynamic VPN Static- Dynami Route	c Default WAN Trunk	Main Route → Out				
Note: If you want to configure NAT, please go to <u>NAT</u> .							
# NAT Rule	Source	Destination	Outgoing	Gateway			
[K] < <p>4 Page 1 of 1 ▷ ▷] Show 50 v items</p>				No data to display			







Figure 365 Maintenance > Packet Flow Explore > Routing Status (Dynamic VPN)







Routing] Flow							
In 📫	Direct Policy Policy I-1 SNAT SiteToSite	Dynamic VPN VPN VPN VPN VPN VPN VPN VPN	Out					
Routing	j table							
N If yo	🚺 Note: If you want to configure Default WAN Trunk, please go to <u>Trunk</u> .							
#	Source	Destination	Trunk					
1	any	any	SYSTEM_DEFAULT_WAN_TRUNK					
M	Page 1 of 1 > M Show 50 v items		Displaying 1 - 1 of 1					



Routin	Routing How								
In 📫	Direct Policy Route 1-1 SNAT	iteToSite VPN VPN VPN	Static- Dynamic Route	Default WAN Trunk					
Routing	Table								
Slag	lote: :: A - Activated route, S - Static route, C - directly C	onnected O - OSPF derive	ed, R - RIP derived,	G - selected Gateway ! - reject, B - Black hole, I	- Loop.				
	Destination	Gateway	Interface	Metric	Flags	Persist			
1	127.0.0.0/8	0.0.0.0	lo	0	ACG				
2	192.168.1.0/24	0.0.0.0	s lan1	0	ACG				
3	192.168.2.0/24	0.0.0.0	Ian2	0	ACG				
4	192.168.3.0/24	0.0.0.0	a dmz	0	ACG				
\mathbb{M}	Page 1 of 1 > > > Show 50 v iter	ns					Displaying 1 - 4 of 4		

The following table describes the labels in this screen.

 Table 211
 Maintenance > Packet Flow Explore > Routing Status

LABEL	DESCRIPTION
Routing Flow	This section shows you the flow of how the ZyWALL determines where to route a packet. Click a function box to display the related settings in the Routing Table section.
Routing Table	This section shows the corresponding settings according to the function box you click in the Routing Flow section.
The following field Routing Flow set	is are available if you click Direct Route , Static-Dynamic Route , or Main Route in the ction.
#	This field is a sequential value, and it is not associated with any entry.
Destination	This is the destination IP address of a route.
Gateway	This is the IP address of the next-hop gateway or the interface through which the traffic is routed.
Interface	This is the name of an interface associated with the route.
Metric	This is the route's priority among the displayed routes.
Flags	This indicates additional information for the route. The possible flags are:
	 A - this route is currently activated S - this is a static route C - this is a direct connected route O - this is a dynamic route learned through OSPF R - this is a dynamic route learned through RIP G - the route is to a gateway (router) in the same network. ! - this is a route which forces a route lookup to fail. B - this is a route which discards packets. L - this is a recursive route.
Persist	This is the remaining time of a dynamically learned route. The ZyWALL removes the route after this time period is counted down to zero.
The following field	Is are available if you click Policy Route in the Routing Flow section.
#	This field is a sequential value, and it is not associated with any entry.
PR #	This is the number of an activated policy route. If you have configured a schedule for the route, this screen only displays the route at the scheduled time.
Incoming	This is the interface on which the packets are received.
Source	This is the source IP address(es) from which the packets are sent.
Destination	This is the destination IP address(es) to which the packets are transmitted.
Service	This is the name of the service object. any means all services.
DSCP Code	This is the DSCP value of incoming packets to which this policy route applies. See Section 9.2 on page 187 for more information.
Next Hop Type	This is the type of the next hop to which packets are directed.
Next Hop Info	 This is the main route if the next hop type is Auto. This is the interface name and gateway IP address if the next hop type is Interface / GW. This is the tunnel name if the next hop type is VPN Tunnel. This is the trunk name if the next hop type is Trunk.
The following field	is are available if you click 1-1 SNAT in the Routing Flow section.
#	This field is a sequential value, and it is not associated with any entry.
NAT Rule	This is the name of an activated 1:1 or Many 1:1 NAT rule in the NAT table.
Source	This is the original source IP address(es). any means any IP address.
Destination	This is the original destination IP address(es). any means any IP address.

LABEL	DESCRIPTION				
Outgoing	This is the name of an interface which transmits packets out of the ZyWALL.				
Gateway This is the IP address of the gateway in the same network of the outgoing inte					
The following field	is are available if you click SiteToSite VPN or Dynamic VPN in the Routing Flow section.				
#	This field is a sequential value, and it is not associated with any entry.				
Source	This is the IP address(es) of the local VPN network.				
Destination	This is the IP address(es) for the remote VPN network.				
VPN Tunnel	This is the name of the VPN tunnel.				
The following field	is are available if you click Default WAN Trunk in the Routing Flow section.				
#	This field is a sequential value, and it is not associated with any entry.				
Source	This is the source IP address(es) from which the packets are sent. any means any IP address.				
Destination	This is the destination IP address(es) to which the packets are transmitted. any means any IP address.				
Trunk	This is the name of the WAN trunk through which the matched packets are transmitted.				

 Table 211
 Maintenance > Packet Flow Explore > Routing Status (continued)

41.3 The SNAT Status Screen

The **SNAT Status** screen allows you to view and quickly link to specific source NAT (SNAT) settings. Click a function box in the **SNAT Flow** section, the related SNAT rules (activated) will display in the **SNAT Table** section. To access this screen, click **Maintenance** > **Packet Flow Explore** > **SNAT Status**.

The order of the SNAT flow may vary depending on whether you:

- select use default SNAT in the CONFIGURATION > Network > Interface > Trunk screen.
- use policy routes to control 1-1 NAT by using the policy control-virtual-server-rules activate command.

Note: Once a packet matches the criteria of an SNAT rule, the ZyWALL takes the corresponding action and does not perform any further flow checking.

SNAT Flow						
In Default SNAT Default Out						
🐚 Note:						
If you want to configure Policy Route SNAT, please go to Policy Rout	<u>e</u> .					
# PR#	Outgoing	SNAT				
[K] 4 Page 1 of 1 ▶ [K] Show 50 ▼ items		No data to display				

Figure 369 Maintenance > Packet Flow Explore > SNAT Status (Policy Route SNAT)







Figure 371 Maintenance > Packet Flow Explore > SNAT Status (Loopback SNAT)



SNAT Flow		
In Policy Route SNAT > 1-1 SNAT > Loopback > Default SNAT > SNAT >	Out	
Note: If you want to configure Default SNAT, please go to <u>Trunk</u> .		
# Incoming	Outgoing	SNAT
1 <u>Ginternal Interface</u>	External Interface	Outgoing Interface IP
[1] 4 Page 1 of 1 ▷ □] Show 50 v tems		Displaying 1 - 1 of 1

The following table describes the labels in this screen.

LABEL	DESCRIPTION
SNAT Flow	This section shows you the flow of how the ZyWALL changes the source IP address for a packet according to the rules you have configured in the ZyWALL. Click a function box to display the related settings in the SNAT Table section.
SNAT Table	The table fields in this section vary depending on the function box you select in the SNAT Flow section.
The following field	ds are available if you click Policy Route SNAT in the SNAT Flow section.
#	This field is a sequential value, and it is not associated with any entry.
PR #	This is the number of an activated policy route which uses SNAT.
Outgoing	This is the outgoing interface that the route uses to transmit packets.
SNAT	This is the source IP address(es) that the SNAT rule uses finally.
The following field	ds are available if you click 1-1 SNAT in the SNAT Flow section.
#	This field is a sequential value, and it is not associated with any entry.
NAT Rule	This is the name of an activated NAT rule which uses SNAT.
Source	This is the original source IP address(es).

Table 212 Maintenance > Packet Flow Explore > SNAT Status

LABEL	DESCRIPTION	
Destination	This is the original destination IP address(es).	
Outgoing	This is the outgoing interface that the SNAT rule uses to transmit packets.	
SNAT	This is the source IP address(es) that the SNAT rule uses finally.	
The following field	ds are available if you click Loopback SNAT in the SNAT Flow section.	
#	This field is a sequential value, and it is not associated with any entry.	
NAT Rule	This is the name of an activated NAT rule which uses SNAT and enables NAT loopback.	
Source	This is the original source IP address(es). any means any IP address.	
Destination	This is the original destination IP address(es). any means any IP address.	
SNAT	This indicates which source IP address the SNAT rule uses finally. For example, Outgoing Interface IP means that the ZyWALL uses the IP address of the outgoing interface as the source IP address for the matched packets it sends out through this rule.	
The following field	ds are available if you click Default SNAT in the SNAT Flow section.	
#	This field is a sequential value, and it is not associated with any entry.	
Incoming	This indicates internal interface(s) on which the packets are received.	
Outgoing	This indicates external interface(s) from which the packets are transmitted.	
SNAT	This indicates which source IP address the SNAT rule uses finally. For example, Outgoing Interface IP means that the ZyWALL uses the IP address of the outgoing interface as the source IP address for the matched packets it sends out through this rule.	

 Table 212
 Maintenance > Packet Flow Explore > SNAT Status (continued)

42

Reboot

42.1 Overview

Use this to restart the device (for example, if the device begins behaving erratically). See also Section on page 32 for information on different ways to start and stop the ZyWALL.

42.1.1 What You Need To Know

If you applied changes in the Web configurator, these were saved automatically and do not change when you reboot. If you made changes in the CLI, however, you have to use the write command to save the configuration before you reboot. Otherwise, the changes are lost when you reboot.

Reboot is different to reset; (see Section 44.1 on page 524) reset returns the device to its default configuration.

42.2 The Reboot Screen

The **Reboot** screen allows remote users to restart the device. To access this screen, click **Maintenance** > **Reboot**.

Figure 373 Maintenance > Reboot

Reboot	
Reboot	
Click the appears.	Reboot button to reboot the device. Please wait a few minutes until the login screen If the login screen does not appear, type the IP address of the device in your Web browser.
	Reboot

Click the **Reboot** button to restart the ZyWALL. Wait a few minutes until the login screen appears. If the login screen does not appear, type the IP address of the device in your Web browser.

You can also use the CLI command reboot to restart the ZyWALL.

Shutdown

43.1 Overview

Use this to shutdown the device in preparation for disconnecting the power. See also Section on page 32 for information on different ways to start and stop the ZyWALL.

Always use the Maintenance > Shutdown > Shutdown screen or the "shutdown" command before you turn off the ZyWALL or remove the power. Not doing so can cause the firmware to become corrupt.

43.1.1 What You Need To Know

Shutdown writes all cached data to the local storage and stops the system processes.

43.2 The Shutdown Screen

To access this screen, click Maintenance > Shutdown.

Figure 374 Maintenance > Shutdown



Click the **Shutdown** button to shut down the ZyWALL. Wait for the device to shut down before you manually turn off or remove the power. It does not turn off the power.

You can also use the CLI command shutdown to shutdown the ZyWALL.

Troubleshooting

This chapter offers some suggestions to solve problems you might encounter.

- You can also refer to the logs (see Chapter 6 on page 100).
- For the order in which the ZyWALL applies its features and checks, see Chapter 41 on page 507.

None of the LEDs turn on.

Make sure that you have the power cord connected to the ZyWALL and plugged in to an appropriate power source. Make sure you have the ZyWALL turned on. Check all cable connections.

If the LEDs still do not turn on, you may have a hardware problem. In this case, you should contact your local vendor.

Cannot access the ZyWALL from the LAN.

- Check the cable connection between the ZyWALL and your computer or switch.
- Ping the ZyWALL from a LAN computer. Make sure your computer's Ethernet card is installed and functioning properly. Also make sure that its IP address is in the same subnet as the ZyWALL's.
- In the computer, click **Start**, **(AII) Programs**, **Accessories** and then **Command Prompt**. In the **Command Prompt** window, type "ping" followed by the ZyWALL's LAN IP address (192.168.1.1 is the default) and then press [ENTER]. The ZyWALL should reply.
- If you've forgotten the ZyWALL's password, use the RESET button. Press the button in for about 5 seconds (or until the PWR LED starts to blink), then release it. It returns the ZyWALL to the factory defaults (password is 1234, LAN IP address 192.168.1.1 etc.; see your User's Guide for details).

I cannot access the Internet.

- Check the ZyWALL's connection to the Ethernet jack with Internet access. Make sure the Internet gateway device (such as a DSL modem) is working properly.
- Check the WAN interface's status in the **Dashboard**. Use the installation setup wizard again and make sure that you enter the correct settings. Use the same case as provided by your ISP.

I configured security settings but the ZyWALL is not applying them for certain interfaces.

Many security settings are usually applied to zones. Make sure you assign the interfaces to the appropriate zones. When you create an interface, there is no security applied on it until you assign it to a zone.

The ZyWALL is not applying the custom policy route I configured.

The ZyWALL checks the policy routes in the order that they are listed. So make sure that your custom policy route comes before any other routes that the traffic would also match.

The ZyWALL is not applying the custom firewall rule I configured.

The ZyWALL checks the firewall rules in the order that they are listed. So make sure that your custom firewall rule comes before any other rules that the traffic would also match.

I cannot enter the interface name I want.

The format of interface names other than the Ethernet interface names is very strict. Each name consists of 2-4 letters (interface type), followed by a number (x, limited by the maximum number of each type of interface). For example, VLAN interfaces are vlan0, vlan1, vlan2, ...; and so on.

• The names of virtual interfaces are derived from the interfaces on which they are created. For example, virtual interfaces created on Ethernet interface wan1 are called wan1:1, wan1:2, and so on. Virtual interfaces created on VLAN interface vlan2 are called vlan2:1, vlan2:2, and so on. You cannot specify the number after the colon(:) in the Web Configurator; it is a sequential number. You can specify the number after the colon if you use the CLI to set up a virtual interface.

I cannot set up a PPP interface, virtual Ethernet interface or virtual VLAN interface on an Ethernet interface.

You cannot set up a PPP interface, virtual Ethernet interface or virtual VLAN interface if the underlying interface is a member of a bridge. You also cannot add an Ethernet interface or VLAN interface to a bridge if the member interface has a virtual interface or PPP interface on top of it.

My rules and settings that apply to a particular interface no longer work.

The interface's IP address may have changed. To avoid this create an IP address object based on the interface. This way the ZyWALL automatically updates every rule or setting that uses the object whenever the interface's IP address settings change. For example, if you change LAN1's IP address, the ZyWALL automatically updates the corresponding interface-based, LAN1 subnet address object.

I cannot set up a PPP interface.

You have to set up an ISP account before you create a PPPoE or PPTP interface.

The data rates through my cellular connection are no-where near the rates I expected.

The actual cellular data rate you obtain varies depending on the cellular device you use, the signal strength to the service provider's base station, and so on.

I created a cellular interface but cannot connect through it.

- Make sure you have a compatible 3G device installed or connected. See www.zyxel.com for details.
- Make sure you have the cellular interface enabled.
- Make sure the cellular interface has the correct user name, password, and PIN code configured with the correct casing.
- If the ZyWALL has multiple WAN interfaces, make sure their IP addresses are on different subnets.

I cannot configure a particular VLAN interface on top of an Ethernet interface even though I have it configured it on top of another Ethernet interface.

Each VLAN interface is created on top of only one Ethernet interface.

The ZyWALL is not applying an interface's configured ingress bandwidth limit.

At the time of writing, the ZyWALL does not support ingress bandwidth management.

The ZyWALL is not scanning some zipped files.

The ZyWALL cannot unzip password protected ZIP files or a ZIP file within another ZIP file. There are also limits to the number of ZIP files that the ZyWALL can concurrently unzip.

The ZyWALL is deleting some zipped files.

The ZyWALL cannot unzip password protected ZIP files or a ZIP file within another ZIP file. There are also limits to the number of ZIP files that the ZyWALL can concurrently unzip.

The ZyWALL routes and applies SNAT for traffic from some interfaces but not from others.

The ZyWALL automatically uses SNAT for traffic it routes from internal interfaces to external interfaces. For example LAN to WAN traffic. You must manually configure a policy route to add routing and SNAT settings for an interface with the **Interface Type** set to **General**. You can also configure a policy route to override the default routing and SNAT behavior for an interface with the **Interface Type** set to **Internal** or **External**.

I cannot get Dynamic DNS to work.

- You must have a public WAN IP address to use Dynamic DNS.
- Make sure you recorded your DDNS account's user name, password, and domain name and have entered them properly in the ZyWALL.
- You may need to configure the DDNS entry's IP Address setting to **Auto** if the interface has a dynamic IP address or there are one or more NAT routers between the ZyWALL and the DDNS server.
- The ZyWALL may not determine the proper IP address if there is an HTTP proxy server between the ZyWALL and the DDNS server.

I cannot create a second HTTP redirect rule for an incoming interface.

You can configure up to one HTTP redirect rule for each (incoming) interface.

The ZyWALL keeps resetting the connection.

If an alternate gateway on the LAN has an IP address in the same subnet as the ZyWALL's LAN IP address, return traffic may not go through the ZyWALL. This is called an asymmetrical or "triangle" route. This causes the ZyWALL to reset the connection, as the connection has not been acknowledged.

You can set the ZyWALL's firewall to permit the use of asymmetrical route topology on the network (so it does not reset the connection) although this is not recommended since allowing asymmetrical routes may let traffic from the WAN go directly to the LAN without passing through the ZyWALL. A better solution is to use virtual interfaces to put the ZyWALL and the backup gateway on separate

subnets. See Asymmetrical Routes on page 259 and the chapter about interfaces for more information.

I cannot set up an IPSec VPN tunnel to another device.

If the IPSec tunnel does not build properly, the problem is likely a configuration error at one of the IPSec routers. Log into both ZyXEL IPSec routers and check the settings in each field methodically and slowly. Make sure both the ZyWALL and remote IPSec router have the same security settings for the VPN tunnel. It may help to display the settings for both routers side-by-side.

Here are some general suggestions. See also Chapter 20 on page 272.

- The system log can often help to identify a configuration problem.
- If you enable NAT traversal, the remote IPSec device must also have NAT traversal enabled.
- The ZyWALL and remote IPSec router must use the same authentication method to establish the IKE SA.
- Both routers must use the same negotiation mode.
- Both routers must use the same encryption algorithm, authentication algorithm, and DH key group.
- When using manual keys, the ZyWALL and remote IPSec router must use the same encryption key and authentication key.
- When using pre-shared keys, the ZyWALL and the remote IPSec router must use the same preshared key.
- The ZyWALL's local and peer ID type and content must match the remote IPSec router's peer and local ID type and content, respectively.
- The ZyWALL and remote IPSec router must use the same active protocol.
- The ZyWALL and remote IPSec router must use the same encapsulation.
- The ZyWALL and remote IPSec router must use the same SPI.
- If the sites are/were previously connected using a leased line or ISDN router, physically disconnect these devices from the network before testing your new VPN connection. The old route may have been learnt by RIP and would take priority over the new VPN connection.
- To test whether or not a tunnel is working, ping from a computer at one site to a computer at the other.

Before doing so, ensure that both computers have Internet access (via the IPSec routers).

• It is also helpful to have a way to look at the packets that are being sent and received by the ZyWALL and remote IPSec router (for example, by using a packet sniffer).

Check the configuration for the following ZyWALL features.

- The ZyWALL does not put IPSec SAs in the routing table. You must create a policy route for each VPN tunnel. See Chapter 9 on page 185.
- Make sure the To-ZyWALL firewall rules allow IPSec VPN traffic to the ZyWALL. IKE uses UDP port 500, AH uses IP protocol 51, and ESP uses IP protocol 50.
- The ZyWALL supports UDP port 500 and UDP port 4500 for NAT traversal. If you enable this, make sure the To-ZyWALL firewall rules allow UDP port 4500 too.

- Make sure regular firewall rules allow traffic between the VPN tunnel and the rest of the network. Regular firewall rules check packets the ZyWALL sends before the ZyWALL encrypts them and check packets the ZyWALL receives after the ZyWALL decrypts them. This depends on the zone to which you assign the VPN tunnel and the zone from which and to which traffic may be routed.
- If you set up a VPN tunnel across the Internet, make sure your ISP supports AH or ESP (whichever you are using).
- If you have the ZyWALL and remote IPSec router use certificates to authenticate each other, You must set up the certificates for the ZyWALL and remote IPSec router first and make sure they trust each other's certificates. If the ZyWALL's certificate is self-signed, import it into the remote IPsec router. If it is signed by a CA, make sure the remote IPsec router trusts that CA. The ZyWALL uses one of its **Trusted Certificates** to authenticate the remote IPSec router's certificate. The trusted certificate can be the remote IPSec router's self-signed certificate or that of a trusted CA that signed the remote IPSec router's certificate.
- Multiple SAs connecting through a secure gateway must have the same negotiation mode.

The VPN connection is up but VPN traffic cannot be transmitted through the VPN tunnel.

If you have the **Configuration > VPN > IPSec VPN > VPN Connection** screen's **Use Policy Route to control dynamic IPSec rules option** enabled, check the routing policies to see if they are sending traffic elsewhere instead of through the VPN tunnels.

I uploaded a logo to show in the SSL VPN user screens but it does not display properly.

The logo graphic must be GIF, JPG, or PNG format. The graphic should use a resolution of 103 x 29 pixels to avoid distortion when displayed. The ZyWALL automatically resizes a graphic of a different resolution to 103 x 29 pixels. The file size must be 100 kilobytes or less. Transparent background is recommended.

I logged into the SSL VPN but cannot see some of the resource links.

Available resource links vary depending on the SSL application object's configuration.

I changed the LAN IP address and can no longer access the Internet.

The ZyWALL automatically updates address objects based on an interface's IP address, subnet, or gateway if the interface's IP address settings change. However, you need to manually edit any address objects for your LAN that are not based on the interface.

I cannot get the RADIUS server to authenticate the ZyWALL's default admin account.

The default **admin** account is always authenticated locally, regardless of the authentication method setting. (See Chapter 31 on page 390 for more information about authentication methods.)

The ZyWALL fails to authentication the ext-user user accounts I configured.

An external server such as AD, LDAP or RADIUS must authenticate the ext-user accounts. If the ZyWALL tries to use the local database to authenticate an **ext-user**, the authentication attempt will always fail. (This is related to AAA servers and authentication methods, which are discussed in Chapter 31 on page 390 and Chapter 32 on page 399, respectively.)

I cannot add the admin users to a user group with access users.

You cannot put access users and admin users in the same user group.

I cannot add the default admin account to a user group.

You cannot put the default **admin** account into any user group.

The schedule I configured is not being applied at the configured times.

Make sure the ZyWALL's current date and time are correct.

I cannot get a certificate to import into the ZyWALL.

- 1 For **My Certificates**, you can import a certificate that matches a corresponding certification request that was generated by the ZyWALL. You can also import a certificate in PKCS#12 format, including the certificate's public and private keys.
- 2 You must remove any spaces from the certificate's filename before you can import the certificate.
- **3** Any certificate that you want to import has to be in one of these file formats:
 - Binary X.509: This is an ITU-T recommendation that defines the formats for X.509 certificates.
 - PEM (Base-64) encoded X.509: This Privacy Enhanced Mail format uses lowercase letters, uppercase letters and numerals to convert a binary X.509 certificate into a printable form.
 - Binary PKCS#7: This is a standard that defines the general syntax for data (including digital signatures) that may be encrypted. A PKCS #7 file is used to transfer a public key certificate. The private key is not included. The ZyWALL currently allows the importation of a PKS#7 file that contains a single certificate.

- PEM (Base-64) encoded PKCS#7: This Privacy Enhanced Mail (PEM) format uses lowercase letters, uppercase letters and numerals to convert a binary PKCS#7 certificate into a printable form.
- Binary PKCS#12: This is a format for transferring public key and private key certificates. The private key in a PKCS #12 file is within a password-encrypted envelope. The file's password is not connected to your certificate's public or private passwords. Exporting a PKCS #12 file creates this and you must provide it to decrypt the contents when you import the file into the ZyWALL.
- Note: Be careful not to convert a binary file to text during the transfer process. It is easy for this to occur since many programs use text files by default.

I cannot access the ZyWALL from a computer connected to the Internet.

Check the service control rules and to-ZyWALL firewall rules.

I uploaded a logo to display on the upper left corner of the Web Configurator login screen and access page but it does not display properly.

Make sure the logo file is a GIF, JPG, or PNG of 100 kilobytes or less.

I uploaded a logo to use as the screen or window background but it does not display properly.

Make sure the logo file is a GIF, JPG, or PNG of 100 kilobytes or less.

The ZyWALL's traffic throughput rate decreased after I started collecting traffic statistics.

Data collection may decrease the ZyWALL's traffic throughput rate.

I can only see newer logs. Older logs are missing.

When a log reaches the maximum number of log messages, new log messages automatically overwrite existing log messages, starting with the oldest existing log message first.

The commands in my configuration file or shell script are not working properly.

• In a configuration file or shell script, use "#" or "!" as the first character of a command line to have the ZyWALL treat the line as a comment.

- Your configuration files or shell scripts can use "exit" or a command line consisting of a single "!" to have the ZyWALL exit sub command mode.
- Include write commands in your scripts. Otherwise the changes will be lost when the ZyWALL restarts. You could use multiple write commands in a long script.

Note: "exit" or "!'" must follow sub commands if it is to make the ZyWALL exit sub command mode.

See Chapter 39 on page 488 for more on configuration files and shell scripts.

I cannot get the firmware uploaded using the commands.

The Web Configurator is the recommended method for uploading firmware. You only need to use the command line interface if you need to recover the firmware. See the CLI Reference Guide for how to determine if you need to recover the firmware and how to recover it.

My packet capture captured less than I wanted or failed.

The packet capture screen's **File Size** sets a maximum size limit for the total combined size of all the capture files on the ZyWALL, including any existing capture files and any new capture files you generate. If you have existing capture files you may need to set this size larger or delete existing capture files.

The ZyWALL stops the capture and generates the capture file when either the capture files reach the **File Size** or the time period specified in the **Duration** field expires.

My earlier packet capture files are missing.

New capture files overwrite existing files of the same name. Change the **File Suffix** field's setting to avoid this.

44.1 Resetting the ZyWALL

If you cannot access the ZyWALL by any method, try restarting it by turning the power off and then on again. If you still cannot access the ZyWALL by any method or you forget the administrator password(s), you can reset the ZyWALL to its factory-default settings. Any configuration files or shell scripts that you saved on the ZyWALL should still be available afterwards.

Use the following procedure to reset the ZyWALL to its factory-default settings. This overwrites the settings in the startup-config.conf file with the settings in the system-default.conf file.

Note: This procedure removes the current configuration.

If you want to reboot the device without changing the current configuration, see Chapter 42 on page 514.

- 1 Make sure the **SYS** LED is on and not blinking.
- 2 Press the **RESET** button and hold it until the **SYS** LED begins to blink. (This usually takes about five seconds.)
- **3** Release the **RESET** button, and wait for the ZyWALL to restart.

You should be able to access the ZyWALL using the default settings.

44.2 Getting More Troubleshooting Help

Search for support information for your model at www.zyxel.com for more troubleshooting suggestions.

Legal Information

Copyright

Copyright © 2014 by ZyXEL Communications Corporation.

The contents of this publication may not be reproduced in any part or as a whole, transcribed, stored in a retrieval system, translated into any language, or transmitted in any form or by any means, electronic, mechanical, magnetic, optical, chemical, photocopying, manual, or otherwise, without the prior written permission of ZyXEL Communications Corporation.

Published by ZyXEL Communications Corporation. All rights reserved.

Disclaimer

ZyXEL does not assume any liability arising out of the application or use of any products, or software described herein. Neither does it convey any license under its patent rights nor the patent rights of others. ZyXEL further reserves the right to make changes in any products described herein without notice. This publication is subject to change without notice.

Certifications (Class B) ZyWALL 110 Federal Communications Commission (FCC) Interference Statement

The device complies with Part 15 of FCC rules. Operation is subject to the following two conditions:

- This device may not cause harmful interference.
- This device must accept any interference received, including interference that may cause undesired operations.

This device has been tested and found to comply with the limits for a Class B digital device pursuant to Part 15 of the FCC Rules. These limits are designed to provide reasonable protection against harmful interference in a residential installation. This device generates, uses, and can radiate radio frequency energy, and if not installed and used in accordance with the instructions, may cause harmful interference to radio communications. However, there is no guarantee that interference will not occur in a particular installation.

If this device does cause harmful interference to radio/television reception, which can be determined by turning the device off and on, the user is encouraged to try to correct the interference by one or more of the following measures:

- 1 Reorient or relocate the receiving antenna.
- Increase the separation between the equipment and the receiver.
- Connect the equipment into an outlet on a circuit different from that to which the receiver is connected.
- Consult the dealer or an experienced radio/TV technician for help.

Notices

2

3

Changes or modifications not expressly approved by the party responsible for compliance could void the user's authority to operate the equipment.

This device is designed for the WLAN 2.4 GHz and/or 5 GHz networks throughout the EC region and Switzerland, with restrictions in France.

Ce produit est conçu pour les bandes de fréquences 2,4 GHz et/ou 5 GHz conformément à la législation Européenne. En France métropolitaine, suivant les décisions n°03-908 et 03-909 de l'ARCEP, la puissance d'émission ne devra pas dépasser 10 mW (10 dB) dans le cadre d'une installation WiFi en extérieur pour les fréquences comprises entre 2454 MHz et 2483,5 MHz.

This Class B digital apparatus complies with Canadian ICES-003.

Cet appareil numérique de la classe B est conforme à la norme NMB-003 du Canada.

Certifications (Class A) ZyWALL 310, 1100 Federal Communications Commission (FCC) Interference Statement

This device complies with Part 15 of FCC rules. Operation is subject to the following two conditions:

This device may not cause harmful interference.

This device must accept any interference received, including interference that may cause undesired operations.

FCC Warning

This device has been tested and found to comply with the limits for a Class A digital switch, pursuant to Part 15 of the FCC Rules. These limits are designed to provide reasonable protection against harmful interference in a commercial environment. This device generates, uses, and can radiate radio frequency energy and, if not installed and used in accordance with the instruction manual, may cause harmful interference to radio communications. Operation of this device in a residential area is likely to cause harmful interference in which case the user will be required to correct the interference at his own expense.

CE Mark Warning:

This is a class A product. In a domestic environment this product may cause radio interference in which case the user may be required to take adequate measures.

Taiwanese BSMI (Bureau of Standards, Metrology and Inspection) A Warning:

警告使用者 這是甲類的資訊產品,在居住的環境使用時, 可能造成射頻干擾,在這種情況下, 使用者會被要求採取某些適當的對策。

Notices

Changes or modifications not expressly approved by the party responsible for compliance could void the user's authority to operate the equipment.

Cet appareil numérique de la classe A est conforme à la norme NMB-003 du Canada.

CLASS 1 LASER PRODUCT (for products with mini-GBIC slots or laser products, such as fiber-optic transceiver and GPON products) APPAREIL À LASER DE CLASS 1 (for products with mini-GBIC slots or laser products, such as fiber-optic transceiver and GPON products) PRODUCT COMPLIES WITH 21 CFR 1040.10 AND 1040.11. (for products with mini-GBIC slots or laser products, such as fiber-optic transceiver and GPON products)

PRODUIT CONFORME SELON 21 CFR 1040.10 ET 1040.11. (for products with mini-GBIC slots or laser products, such as fiber-optic transceiver and GPON products)

Viewing Certifications

Go to http://www.zyxel.com to view this product's documentation and certifications.

ZyXEL Limited Warranty

ZyXEL warrants to the original end user (purchaser) that this product is free from any defects in material or workmanship for a specific period (the Warranty Period) from the date of purchase. The Warranty Period varies by region. Check with your vendor and/or the authorized ZyXEL local distributor for details about the Warranty Period of this product. During the warranty period, and upon proof of purchase, should the product have indications of failure due to faulty workmanship and/or materials, ZyXEL will, at its discretion, repair or replace the defective products or components without charge for either parts or labor, and to whatever extent it shall deem necessary to restore the product or components to proper operating condition. Any replacement will consist of a new or re-manufactured functionally equivalent product of equal or higher value, and will be solely at the discretion of ZyXEL. This warranty shall not apply if the product has been modified, misused, tampered with, damaged by an act of God, or subjected to abnormal working conditions.

Note

Repair or replacement, as provided under this warranty, is the exclusive remedy of the purchaser. This warranty is in lieu of all other warranties, express or implied, including any implied warranty of merchantability or fitness for a particular use or purpose. ZyXEL shall in no event be held liable for indirect or consequential damages of any kind to the purchaser.

To obtain the services of this warranty, contact your vendor. You may also refer to the warranty policy for the region in which you bought the device at http://www.zyxel.com/web/support_warranty_info.php.

Registration

Register your product online to receive e-mail notices of firmware upgrades and information at www.zyxel.com for global products, or at www.us.zyxel.com for North American products.

Open Source Licenses

This product contains in part some free software distributed under GPL license terms and/or GPL like licenses. Open source licenses are provided with the firmware package. You can download the latest firmware at www.zyxel.com. To obtain the source code covered under those Licenses, please contact support@zyxel.com.tw to get it.

Safety Warnings

- Do NOT use this product near water, for example, in a wet basement or near a swimming pool.
- Do NOT expose your device to dampness, dust or corrosive liquids.
- Do NOT store things on the device.
- Do NOT install, use, or service this device during a thunderstorm. There is a remote risk of electric shock from lightning.
- Connect ONLY suitable accessories to the device. Do NOT open the device or unit. Opening or removing covers can expose you to dangerous high voltage points or other risks. ONLY qualified service personnel should service or disassemble this device. Please contact your vendor for further information.
- Make sure to connect the cables to the correct ports.
- Place connecting cables carefully so that no one will step on them or stumble over them
- Always disconnect all cables from this device before servicing or disassembling Use ONLY an appropriate power adaptor or cord for your device. Connect it to the right supply voltage (for example, 110V AC in North America or 230V AC in Europe).
- Do NOT remove the plug and connect it to a power outlet by itself; always attach the plug to the power adaptor first before connecting it to a power outlet
- Do NOT allow anything to rest on the power adaptor or cord and do NOT place the product where anyone can walk on the power adaptor or cord.
- Do NOT use the device if the power adaptor or cord is damaged as it might cause electrocution.
- If the power adaptor or cord is damaged, remove it from the device and the power source. Do NOT attempt to repair the power adaptor or cord. Contact your local vendor to order a new one.
- Do not use the device outside, and make sure all the connections are indoors. There is a remote risk of electric shock from lightning.

- CAUTION: RISK OF EXPLOSION IF BATTERY (on the motherboard) IS REPLACED BY AN INCORRECT TYPE. DISPOSE OF USED BATTERIES ACCORDING TO THE INSTRUCTIONS. Dispose them at the applicable collection point for the recycling of electrical and electronic equipment. For detailed information about recycling of this product, please contact your local city office, your household waste disposal service or the store where you purchased the product.
- Do NOT obstruct the device ventilation slots, as insufficient airflow may harm your device.

Your product is marked with this symbol, which is known as the WEEE mark. WEEE stands for Waste Electronics and Electrical Equipment. It means that used electrical and electronic products should not be mixed with general waste. Used electrical and electronic equipment should be treated separately.

"INFORMAZIONI AGLI UTENTI"

Ai sensi dell'art. 13 del Decreto Legislativo 25 luglio 2005, n.151

"Attuazione delle Direttive 2002/95/CE, 2002/96/CE e 2003/108/CE, relative alla riduzione dell'uso di sostanze pericolose nelle apparecchiature elettriche ed elettroniche, nonché allo smaltimento dei rifiuti"

Il simbolo del cassonetto barrato riportato sull'apparecchiatura o sulla sua confezione indica che il prodotto alla fine della propria vita utile deve essere raccolto separatamente dagli altri rifiuti.

La raccolta differenziata della presente apparecchiatura giunta a fine vita è organizzata e gestita dal produttore. L'utente che vorrà disfarsi della presente apparecchiatura dovrà quindi contattare il

produttore e seguire il sistema che questo ha adottato per consentire la raccolta separata dell'apparecchiatura giunta a fine vita.

L'adeguata raccolta differenziata per l'avvio successivo dell'apparecchiatura dismessa al riciclaggio, al trattamento e allo smaltimento ambientalmente compatibile contribuisce ad evitare possibili effetti negativi sull'ambiente e sulla salute e favorisce il reimpiego e/o riciclo dei materiali di cui è composta l'apparecchiatura.

Lo smaltimento abusivo del prodotto da parte del detentore comporta l'applicazione delle sanzioni amministrative previste dalla normativa vigente."

Environmental Product Declaration



Index

Symbols

Numbers

3322 Dynamic DNS 212
3DES 297
3G see also cellular 132
6in4 tunneling 140
6to4 tunneling 141

Α

AAA Base DN 392 Bind DN 393, 395 directory structure 392 Distinguished Name, see DN DN 392, 393, 395 password 395 port 394, 397 search time limit 395 SSL 395 AAA server 390 AD 392 and users 362 directory service 390 LDAP 390, 392 local user database 391 RADIUS 390, 392, 396 RADIUS group 396 see also RADIUS access 21 Access Point Name, see APN access users 361, 363 custom page 450 forcing login 247 idle timeout 369

logging in 247 multiple logins 370 see also users 361 Web Configurator 371 access users, see also force user authentication policies account user 361, 428 accounting server 390 Active Directory, see AD active protocol 301 AH 301 and encapsulation 302 ESP 301 active sessions 70, 73, 89 AD 390, 392, 393, 395 directory structure 392 Distinguished Name, see DN password 395 port 394, 397 search time limit 395 SSL 395 address groups 374 and firewall 250 and FTP 468 and SNMP 472 and SSH 464 and Telnet 466 and WWW 450 address objects 374 and firewall 250 and FTP 468 and NAT 193, 220 and policy routes 192 and SNMP 472 and SSH 464 and Telnet 466 and VPN connections 276 and WWW 450 HOST 374 RANGE 374 SUBNET 374 types of 374

address record 441 admin user troubleshooting 522 admin users 361 multiple logins 369 see also users 361 Advanced Encryption Standard, see AES AES 297 AF 195 AH 280, 301 and transport mode 302 alerts 479, 480, 482, 484, 485, 486 ALG 228, 233 and firewall 228, 230 and NAT 228, 230 and policy routes 230, 233 and trunks 233 FTP 228, 229 H.323 228, 229, 233 peer-to-peer calls 230 RTP 234 see also VoIP pass through 228 SIP 228, 229 APN 136 Application Layer Gateway, see ALG ASAS (Authenex Strong Authentication System) 391 asymmetrical routes 259 allowing through the firewall 262 vs virtual interfaces 259 attacks Denial of Service (DoS) 279 Authenex Strong Authentication System (ASAS) 391 authentication in IPSec 281 LDAP/AD 392 server 390 authentication algorithms 207, 297 and active protocol 297 and routing protocols 207 MD5 207, 297 SHA1 297 text 207 Authentication Header, see AH authentication method objects 399 and users 362

and WWW 449 create 400 example 399 authentication policy 246 exceptional services 248 authentication type 45, 421 Authentication, Authorization, Accounting servers, see AAA server authorization server 390

В

backing up configuration files 490 bandwidth egress 137, 146 ingress 137, 146 bandwidth limit troubleshooting 518 bandwidth management maximize bandwidth usage 196, 342, 343 Base DN 392 Bind DN 393, 395 bookmarks 323 boot module 495 bridge interfaces 104, 160 and virtual interfaces of members 160 basic characteristics 104 effect on routing table 160 member interfaces 160 virtual 170 bridges 159

С

CA and certificates **404** CA (Certificate Authority), see certificates capturing packets **501** card SIM **137** CEF (Common Event Format) **477**, **484** cellular **132** APN **136** interfaces **104**

signal quality 94, 95 SIM card 137 status 96 system 94, 95 troubleshooting 518 certificate troubleshooting 522 Certificate Authority (CA) see certificates Certificate Revocation List (CRL) 404 vs OCSP 418 certificates 403 advantages of 404 and CA 404 and FTP 467 and HTTPS 446 and IKE SA 301 and SSH 463 and synchronization (device HA) 359 and VPN gateways 276 and WWW 448 certification path 404, 410, 416 expired 404 factory-default 404 file formats 404 fingerprints 411, 417 importing 407 in IPSec 289 not used for encryption 404 revoked 404 self-signed 404, 409 serial number 411, 416 storage space 407, 413 thumbprint algorithms 405 thumbprints 405 used for authentication 404 verifying fingerprints 405 certification requests 409 certifications 526 notices 526, 527 viewing 527 Challenge Handshake Authentication Protocol (CHAP) 421 CHAP (Challenge Handshake Authentication Protocol) 421 CHAP/PAP 421 CLI 21, 25 button 25

messages 25 popup window 25 Reference Guide 2 client 331 cluster ID 351 commands 21 sent by Web Configurator 25 Common Event Format (CEF) 477, 484 compression (stac) 421 computer names 120, 157, 169, 174, 338 configuration information 499, 504 web-based SSL application example 423 configuration file troubleshooting 523 configuration files 488 at restart 491 backing up 490 downloading 492 downloading with FTP 467 editing 488 how applied 489 lastgood.conf 491, 494 managing 490 startup-config.conf 494 startup-config-bad.conf 491 syntax 489 system-default.conf 494 uploading 494 uploading with FTP 467 use without restart 488 connection troubleshooting 519 connection monitor (in SSL) 99 connectivity check 119, 131, 137, 146, 156, 169, 281 console port speed 438 cookies 21 copyright 526 CPU usage 70, 72 current date/time 69, 434 and schedules 386 daylight savings 436 setting manually 437 time server 438 current user list 99 custom

access user page 450 login page 450

D

Data Encryption Standard, see DES date 434 daylight savings 436 DDNS 212 backup mail exchanger 216 mail exchanger 216 service providers 212 troubleshooting 519 Dead Peer Detection, see DPD default firewall behavior 256 Default_L2TP_VPN_Connection 336 Default_L2TP_VPN_GW 336 Denial of Service (Dos) attacks 279 DES 297 device access troubleshooting 516 device HA 349 active-passive mode 349, 351 cluster ID 351 copying configuration 349 device role 353 HA status 351 legacy mode 349 management access 349 management IP address 349 modes 349 monitored interfaces 352, 355 password 354 synchronization 349, 359 synchronization password 354 synchronization port number 354 virtual router 351 virtual router and management IP addresses 352 device High Availability see device HA 349 DHCP 173, 433 and DNS servers 174 and domain name 433 and interfaces 174 client list 75

pool 174 static DHCP 174 DHCP Unique IDentifier 107 DHCPv6 428 DHCP Unique IDentifier 107 DHCPv6 Request 428 diagnostics 499, 504 Diffie-Hellman key group 298 DiffServ 195 Digital Signature Algorithm public-key algorithm, see DSA direct routes 188 directory 390 directory service 390 file structure 392 disclaimer 526 Distinguished Name (DN) 392, 393, 395 DN 392, 393, 395 DNS 439 address records 441 domain name forwarders 442 domain name to IP address 441 IP address to domain name 441 L2TP VPN 338 Mail eXchange (MX) records 443 pointer (PTR) records 441 DNS inbound LB 240 DNS servers 46, 439, 442 and interfaces 174 documentation related 2 domain name 433 Domain Name System, see DNS DPD 291 DSA 409 DSCP 189, 192, 345, 510 DUID 107 Dynamic Domain Name System, see DDNS Dynamic Host Configuration Protocol, see DHCP. dynamic peers in IPSec 279 DynDNS 212 DynDNS see also DDNS 212 Dynu 212

Ε

egress bandwidth 137, 146 e-mail daily statistics report 474 Encapsulating Security Payload, see ESP encapsulation and active protocol 302 IPSec 280 transport mode 302 tunnel mode 302 VPN 302 encryption IPSec 280 RSA 411 encryption algorithms 297 3DES 297 AES 297 and active protocol 297 DES 297 encryption method 421 end-point security multiple objects 247 enforcing policies in IPSec 280 ESP 280, 301 and transport mode 302 Ethernet interfaces 104 and OSPF 111 and RIP 110 and routing protocols 109 basic characteristics 104 virtual 170 exceptional services 248 extended authentication and VPN gateways 276 IKE SA 301 ext-user troubleshooting 522

F

FCC interference statement 526 file extensions configuration files 488 shell scripts 488

file manager 488 file sharing SSL application create 425 Firefox 21 firewall 256 actions 264 and address groups 250 and address objects 250 and ALG 228, 230 and H.323 (ALG) 229 and HTTP redirect 225 and IPSec SA 258 and IPSec VPN 521 and logs 250, 264 and NAT 260 and schedules 250, 264, 344, 346 and service groups 264 and service objects 381 and services 264 and SIP (ALG) 229 and user groups 264, 266 and users 264, 266 and VoIP pass through 230 and zones 256, 262 asymmetrical routes 259, 262 global rules 258 priority 262 rule criteria 258 see also to-ZyWALL firewall 256 session limits 258, 264 to-ZyWALL, see to-ZyWALL firewall triangle routes 259, 262 troubleshooting 517 firmware and restart 494 boot module, see boot module current version 69, 495 getting updated 494 uploading 494, 495 uploading with FTP 467 firmware upload troubleshooting 524 flash usage 70 forcing login 247 FQDN 441 FTP 467 additional signaling port 232 ALG 228

and address groups 468 and address objects 468 and certificates 467 and zones 468 signaling port 232 with Transport Layer Security (TLS) 467 full tunnel mode 308, 312 Fully-Qualified Domain Name, see FQDN

G

Generic Routing Encapsulation, see GRE. global SSL setting 313 user portal logo 314 GRE 175 GSM 137 Guide CLI Reference 2 Quick Start 2

Η

H.323 233 additional signaling port 232 ALG 228, 233 and firewall 229 and RTP 234 signaling port 232 HA status see device HA 351 HSDPA 137 HTTP over SSL, see HTTPS redirect to HTTPS 448 vs HTTPS 446 HTTP redirect 224 and firewall 225 and interfaces 227 and policy routes 225 packet flow 225 troubleshooting 519 HTTPS 445 and certificates 446 authenticating clients 446 avoiding warning messages 455 example 454 vs HTTP 446 with Internet Explorer 454 with Netscape Navigator 454 hub-and-spoke VPN, see VPN concentrator HyperText Transfer Protocol over Secure Socket Layer, see HTTPS

I

ICMP 380 IEEE 802.1q VLAN IEEE 802.1q. See VLAN. IKE SA aggressive mode 296, 299, 300 and certificates 301 and RADIUS 301 and to-ZyWALL firewall 520 authentication algorithms 297 content 299 Dead Peer Detection (DPD) 291 Diffie-Hellman key group 298 encryption algorithms 297 extended authentication 301 ID type 299 IP address, remote IPSec router 296 IP address, ZyXEL device 296 local identity 299 main mode 296, 299 NAT traversal 300 negotiation mode 296 password 301 peer identity 299 pre-shared key 298 proposal 297 see also VPN user name 301 inbound LB algorithm least connection 242 least load 242 weighted round robin 242 inbound load balancing 240 time to live 243 incoming bandwidth 137, 146 ingress bandwidth 137, 146 interface

status 70, 84, 85 troubleshooting 517 interfaces 103 and DNS servers 174 and HTTP redirect 227 and layer-3 virtualization 104 and NAT 220 and physical ports 104 and policy routes 192 and static routes 195 and VPN gateways 276 and zones 104 as DHCP relays 174 as DHCP servers 174, 433 backup, see trunks bandwidth management 173, 182, 183 bridge, see also bridge interfaces. cellular 104 DHCP clients 172 Ethernet, see also Ethernet interfaces. gateway 173 general characteristics 103 IP address 172 metric 173 MTU 173 overlapping IP address and subnet mask 172 port groups, see also port groups. PPPoE/PPTP, see also PPPoE/PPTP interfaces. prerequisites 105 relationships between 105 static DHCP 174 subnet mask 172 trunks, see also trunks. Tunnel, see also Tunnel interfaces. types 104 virtual, see also virtual interfaces. VLAN, see also VLAN interfaces. Internet access troubleshooting 516, 521 Internet Control Message Protocol, see ICMP Internet Explorer 21 Internet Protocol Security, see IPSec Internet Protocol version 6, see IPv6 IP policy routing, see policy routes IP pool 312 IP protocols 380 and service objects 381 ICMP, see ICMP

TCP, see TCP UDP, see UDP IP static routes, see static routes IP/MAC binding 235 exempt list 238 monitor 91 static DHCP 237 IPSec 272 active protocol 280 AH 280 and certificates 276 authentication 281 basic troubleshooting 520 certificates 289 connections 276 connectivity check 281 Default_L2TP_VPN_Connection 336 Default_L2TP_VPN_GW 336 encapsulation 280 encryption 280 ESP 280 established in two phases 274 L2TP VPN 335 local network 272 local policy 279 manual key 279 NetBIOS 279 peer 272 Perfect Forward Secrecy 281 PFS 281 phase 2 settings 280 policy enforcement 280 remote access 279 remote IPSec router 272 remote network 272 remote policy 280 replay detection 279 SA life time 280 SA monitor 97 SA see also IPSec SA 301 see also VPN site-to-site with dynamic peer 279 static site-to-site 279 transport encapsulation 280 tunnel encapsulation 280 VPN gateway 276 **IPSec SA** active protocol 301 and firewall 258, 521

and to-ZyWALL firewall 520 authentication algorithms 297 authentication key (manual keys) 303 destination NAT for inbound traffic 304 encapsulation 302 encryption algorithms 297 encryption key (manual keys) 303 local policy 301 manual keys 303 NAT for inbound traffic 303 NAT for outbound traffic 303 Perfect Forward Secrecy (PFS) 302 proposal 302 remote policy 301 search by name 98 search by policy 98 Security Parameter Index (SPI) (manual keys) 303 see also IPSec see also VPN source NAT for inbound traffic 304 source NAT for outbound traffic 304 status 97 transport mode 302 tunnel mode 302 when IKE SA is disconnected 301 **IPSec VPN** troubleshooting 520 IPv6 106 addressing 106 link-local address 106 prefix 106 prefix delegation 107 prefix length 106 stateless autoconfiguration 107 IPv6 tunnelings 6in4 tunneling 140 6to4 tunneling 141 IPv6-in-IPv4 tunneling 140 **ISP** account CHAP 421 CHAP/PAP 421 MPPE **421** MSCHAP 421 MSCHAP-V2 421 PAP 421 ISP accounts 419 and PPPoE/PPTP interfaces 126, 419 authentication type 421

encryption method **421** stac compression **421**

J

Java permissions 21 JavaScripts 21

Κ

key pairs 403

L

L2TP VPN 335 Default_L2TP_VPN_Connection 336 Default L2TP VPN GW 336 DNS 338 IPSec configuration 335 policy routes 336 session monitor 99 WINS 338 lastgood.conf 491, 494 Layer 2 Tunneling Protocol Virtual Private Network, see L2TP VPN 335 LDAP 390 and users 362 Base DN 392 Bind DN 393, 395 directory 390 directory structure 392 Distinguished Name, see DN DN 392, 393, 395 password 395 port 394, 397 search time limit 395 SSL 395 user attributes 372 least connection algorithm 242 least load algorithm 242 least load first load balancing 177 LED troubleshooting 516

Lightweight Directory Access Protocol, see LDAP load balancing 176 algorithms 177, 181, 183 DNS inbound 240 least load first 177 round robin 178 see also trunks 176 session-oriented 177 spillover 179 weighted round robin 178 local user database 391 log troubleshooting 523 log messages categories 480, 482, 484, 485, 486 debugging 100 regular 100 types of 100 logged in users 76 login custom page 450 SSL user 319 logo troubleshooting 523 logo in SSL 314 logout SSL user 324 Web Configurator 23 logs and firewall 250, 264 e-mail profiles 476 e-mailing log messages 102, 479 formats 477 log consolidation 480 settings 476 syslog servers 476 system 476 types of 476

Μ

MAC address and VLAN 147 Ethernet interface 115 range 69 management access

troubleshooting 523 management access and device HA 349 Management Information Base (MIB) 469 manual key IPSec 279 MD5 297 memory usage 70, 73 Message Digest 5, see MD5 messages CLI 25 metrics, see reports Microsoft Challenge-Handshake Authentication Protocol (MSCHAP) 421 **Challenge-Handshake Authentication Protocol** Version 2 (MSCHAP-V2) 421 Point-to-Point Encryption (MPPE) 421 model name 69 monitor 99 SA 97 monitored interfaces 352 device HA 355 mounting wall 39 MPPE (Microsoft Point-to-Point Encryption) 421 MSCHAP (Microsoft Challenge-Handshake Authentication Protocol) 421 MSCHAP-V2 (Microsoft Challenge-Handshake Authentication Protocol Version 2) 421 MTU 137, 146 My Certificates, see also certificates 406

Ν

NAT 195, 217 ALG, see ALG and address objects 193 and address objects (HOST) 220 and ALG 228, 230 and firewall 260 and interfaces 220 and policy routes 186, 193 and to-ZyWALL firewall 221 and VoIP pass through 230 and VPN 300 loopback 221 port forwarding, see NAT

port translation, see NAT traversal 300 NBNS 120, 157, 169, 174, 312 **NetBIOS** Broadcast over IPSec 279 Name Server, see NBNS. NetBIOS Name Server, see NBNS NetMeeting 233 see also H.323 Netscape Navigator 21 network access mode 19 full tunnel 308 Network Address Translation, see NAT network list, see SSL 313 Network Time Protocol (NTP) 437 No-IP 212 NSSA 200

0

objects 309 AAA server 390 addresses and address groups 374 authentication method 399 certificates 403 schedules 386 services and service groups 380 SSL application 422 users, user groups 361, 428 One-Time Password (OTP) 391 Online Certificate Status Protocol (OCSP) 418 vs CRL 418 Open Shortest Path First, see OSPF OSPF 199 and Ethernet interfaces 111 and RIP 201 and static routes 201 and to-ZyWALL firewall 199 area 0 200 areas, see OSPF areas authentication method 111 autonomous system (AS) 199 backbone 200 configuration steps 202 direction 111 link cost 111

priority 111 redistribute 201 redistribute type (cost) 203 routers, see OSPF routers virtual links 201 vs RIP 197, 199 OSPF areas 199, 200 and Ethernet interfaces 111 backbone 199 Not So Stubby Area (NSSA) 200 stub areas 200 types of 199 OSPF routers 200 area border (ABR) 200 autonomous system boundary (ASBR) 201 backbone (BR) 201 backup designated (BDR) 201 designated (DR) 201 internal (IR) 200 link state advertisements priority 201 types of 200 other documentation 2 OTP (One-Time Password) 391 outgoing bandwidth 137, 146

Ρ

packet statistics 80, 81 packet capture 501 files 500, 503, 505 troubleshooting 524 packet captures downloading files 500, 504, 505, 506 PAP (Password Authentication Protocol) 421 Password Authentication Protocol (PAP) 421 Peanut Hull 212 Peer-to-peer (P2P) calls 230 Perfect Forward Secrecy (PFS) 281 Diffie-Hellman key group 302 Personal Identification Number code, see PIN code PFS (Perfect Forward Secrecy) 281, 302 physical ports packet statistics 80, 81

PIN code 137 PIN generator 391 pointer record 441 Point-to-Point Protocol over Ethernet, see PPPoE. Point-to-Point Tunneling Protocol, see PPTP policy enforcement in IPSec 280 policy route troubleshooting 517 policy routes 186 actions 187 and address objects 192 and ALG 230, 233 and HTTP redirect 225 and interfaces 192 and NAT 186 and schedules 192, 344, 346 and service objects 381 and trunks 176, 192 and user groups 191, 344, 346 and users 191, 344, 346 and VoIP pass through 230 and VPN connections 192, 520 benefits 186 criteria 187 L2TP VPN 336 overriding direct routes 188 pop-up windows 21 port forwarding, see NAT port groups 104 port roles 108 and Ethernet interfaces 108 and physical ports 108 port translation, see NAT power off 515 PPP 175 troubleshooting 518 **PPP** interfaces subnet mask 172 PPPoE 175 and RADIUS 175 TCP port 1723 175 PPPoE/PPTP interfaces 104, 125 and ISP accounts 126, 419 basic characteristics 104 gateway 126 subnet mask 126 PPTP 175

and GRE 175 as VPN 175 prefix delegation 107 problems 516 product registration 527 proxy servers 224 web, see web proxy servers PTR record 441 Public-Key Infrastructure (PKI) 404 public-private key pairs 403

Q

QoS 186, 339 Quick Start Guide 2

R

RADIUS 390, 392 advantages 390 and IKE SA 301 and PPPoE 175 and users 362 user attributes 373 **RADIUS** server troubleshooting 521 RDP 422 Real-time Transport Protocol, see RTP RealVNC 422 reboot 514 vs reset 514 Reference Guide, CLI 2 registration product 527 related documentation 2 Relative Distinguished Name (RDN) 392, 393, 395 remote access IPSec 279 Remote Authentication Dial-In User Service, see RADIUS remote desktop connections 422 Remote Desktop Protocol see RDP remote management

FTP, see FTP see also service control 445 Telnet 465 to-ZyWALL firewall 257 WWW, see WWW remote network 272 remote user screen links 422 replay detection 279 reports collecting data 87 daily 474 daily e-mail 474 specifications 89 traffic statistics 86 reset 524 vs reboot 514 RESET button 524 RFC 1058 (RIP) 197 1389 (RIP) 197 1587 (OSPF areas) 200 1631 (NAT) 195 1889 (RTP) 234 2131 (DHCP) 173 2132 (DHCP) 173 2328 (OSPF) 199 2402 (AH) 280, 301 2406 (ESP) 280, 301 2516 (PPPoE) 175 2637 (PPTP) 175 2890 (GRE) 175 3261 (SIP) 233 RIP 197 and Ethernet interfaces 110 and OSPF 198 and static routes 198 and to-ZyWALL firewall 198 authentication 198 direction 111 redistribute 198 RIP-2 broadcasting methods 111 versions 111 vs OSPF 197 Rivest, Shamir and Adleman public-key algorithm (RSA) 409 round robin 178 routing troubleshooting 519

Routing Information Protocol, see RIP routing protocols 197 and authentication algorithms 207 and Ethernet interfaces 109 RSA 409, 411, 417 RTP 234 see also ALG 234

S

schedule troubleshooting 522 schedules 386 and current date/time 386 and firewall 250, 264, 344, 346 and policy routes 192, 344, 346 one-time 386 recurring 386 types of 386 screen resolution 21 SecuExtender 331 Secure Hash Algorithm, see SHA1 Secure Socket Layer, see SSL security associations, see IPSec security settings troubleshooting 517 serial number 69 service control 445 and to-ZyWALL firewall 445 and users 445 limitations 445 timeouts 445 service groups 381 and firewall 264 service objects 380 and firewall 381 and IP protocols 381 and policy routes 381 services 380 and firewall 264 Session Initiation Protocol, see SIP session limits 258, 264 session monitor (L2TP VPN) 99 sessions 89 sessions usage 70, 73
SHA1 297 shell script troubleshooting 523 shell scripts 488 and users 373 downloading 497 editing 496 how applied 489 managing 496 syntax 489 uploading 498 shutdown 515 signal quality 94, 95 SIM card 137 Simple Network Management Protocol, see SNMP Simple Traversal of UDP through NAT, see STUN SIP 229, 233 ALG 228 and firewall 229 and RTP 234 media inactivity timeout 232 signaling inactivity timeout 232 signaling port 232 SNAT 195 troubleshooting 519 SNMP 468, 469 agents 469 and address groups 472 and address objects 472 and zones 472 Get 469 GetNext 469 Manager 469 managers 469 MIB 469 network components 469 Set 469 Trap 469 traps 470 versions 468 Source Network Address Translation, see SNAT spillover (for load balancing) 179 SSH 461 and address groups 464 and address objects 464 and certificates 463 and zones 464 client requirements 463

encryption methods 463 for secure Telnet 464 how connection is established 462 versions 463 with Linux 465 with Microsoft Windows 464 SSL 308, 312, 445 access policy 308 and AAA 395 and AD 395 and LDAP 395 certificates 319 client 331 client virtual desktop logo 314 computer names 312 connection monitor 99 full tunnel mode 312 global setting 313 IP pool 312 network list 313 remote user login 319 remote user logout 324 SecuExtender 331 see also SSL VPN 308 troubleshooting 521 user application screens 324 user file sharing 325 user screen bookmarks 323 user screens 318, 322 user screens access methods 318 user screens certificates 319 user screens login 319 user screens logout 324 user screens required information 319 user screens system requirements 318 WINS 312 SSL application object 422 file sharing application 425 remote user screen links 422 summary 424 types 422 web-based 422, 425 web-based example 423 SSL policy add 310 edit 310 objects used 309 SSL VPN 308 access policy 308

full tunnel mode 308 network access mode 19 remote desktop connections 422 see also SSL 308 troubleshooting 521 weblink 423 stac compression 421 startup-config.conf 494 and synchronization (device HA) 359 if errors 491 missing at restart 491 present at restart 491 startup-config-bad.conf 491 static DHCP 237 static routes 186 and interfaces 195 and OSPF 201 and RIP 198 metric 195 statistics daily e-mail report 474 traffic 86 status 67 stub area 200 STUN 229 and ALG 229 supported browsers 21 synchronization 349 information synchronized 359 password 354 port number 354 restrictions 359 syslog 477, 484 syslog servers, see also logs system log, see logs system name 69, 433 system reports, see reports system uptime 69 system-default.conf 494

Т

TCP 380 connections 380 port numbers 380 Telnet 465 and address groups 466 and address objects 466 and zones 466 with SSH 464 throughput rate troubleshooting 523 TightVNC 422 time 434 time servers (default) 437 token 391 to-ZyWALL firewall 257 and NAT 221 and NAT traversal (VPN) 520 and OSPF 199 and remote management 257 and RIP 198 and service control 445 and VPN 520 global rules 257 see also firewall 256 traffic statistics 86 Transmission Control Protocol, see TCP transport encapsulation 280 Transport Layer Security (TLS) 467 triangle routes 259 allowing through the firewall 262 vs virtual interfaces 259 Triple Data Encryption Standard, see 3DES troubleshooting 499, 504, 516 admin user 522 bandwidth limit 518 cellular 518 certificate 522 configuration file 523 connection resets 519 DDNS 519 device access 516 ext-user 522 firewall 517 firmware upload 524 HTTP redirect 519 interface 517 Internet access 516, 521 IPSec VPN 520 LEDs 516 logo 523 logs 523

management access 523 packet capture 524 policy route 517 PPP 518 RADIUS server 521 routing 519 schedules 522 security settings 517 shell scripts 523 SNAT 519 SSL 521 SSL VPN 521 throughput rate 523 VLAN 518 VPN 521 zipped files 518 trunks 104, 176 and ALG 233 and policy routes 176, 192 member interface mode 182, 183 member interfaces 182, 183 see also load balancing 176 Trusted Certificates, see also certificates 413 tunnel encapsulation 280 Tunnel interfaces 104

U

UDP 380 messages 380 port numbers 380 UltraVNC 422 upgrading firmware 494 uploading configuration files 494 firmware 494 shell scripts 496 usage CPU 70, 72 flash 70 memory 70, 73 onboard flash 70 sessions 70, 73 user authentication 361 external 362

local user database 391 user awareness 363 User Datagram Protocol, see UDP user group objects 361, 428 user groups 361, 362, 428 and firewall 264, 266 and policy routes 191, 344, 346 user name rules 364 user objects 361, 428 user portal links 422 logo 314 see SSL user screens 318, 322 user sessions, see sessions user SSL screens 318, 322 access methods 318 bookmarks 323 certificates 319 login 319 logout 324 required information 319 system requirements 318 user-aware 251 users 361, 428 access, see also access users admin (type) 361 admin, see also admin users and AAA servers 362 and authentication method objects 362 and firewall 264, 266 and LDAP 362 and policy routes 191, 344, 346 and RADIUS 362 and service control 445 and shell scripts 373 attributes for Ext-User 362 attributes for LDAP 372 attributes for RADIUS 373 attributes in AAA servers 372 currently logged in 69, 76 default lease time 369, 371 default reauthentication time 369, 371 default type for Ext-User 362 ext-group-user (type) 361 Ext-User (type) 362 ext-user (type) 361 groups, see user groups

Guest (type) 361 lease time 366 limited-admin (type) 361 lockout 370 reauthentication time 366 types of 361 user (type) 361 user names 364

V

Vantage Report (VRPT) 477, 484 virtual interfaces 104, 170 basic characteristics 104 not DHCP clients 172 types of 170 vs asymmetrical routes 259 vs triangle routes 259 Virtual Local Area Network, see VLAN. Virtual Local Area Network. See VLAN. Virtual Network Computing see VNC Virtual Private Network, see VPN virtual router 351 VLAN 140, 147 advantages 148 and MAC address 147 ID 147 troubleshooting 518 VLAN interfaces 104, 148 and Ethernet interfaces 148, 518 basic characteristics 104 virtual 170 VoIP pass through 233 and firewall 230 and NAT 230 and policy routes 230 see also ALG 228 VPN 272 active protocol 301 and NAT 300 and the firewall 258 basic troubleshooting 520 hub-and-spoke, see VPN concentrator IKE SA, see IKE SA IPSec 272

IPSec SA proposal 297 security associations (SA) 274 see also IKE SA see also IPSec 272 see also IPSec SA status 74 troubleshooting 521 VPN concentrator 292 advantages 292 and IPSec SA policy enforcement 294 disadvantages 292 **VPN** connections and address objects 276 and policy routes 192, 520 VPN gateways and certificates 276 and extended authentication 276 and interfaces 276 and to-ZyWALL firewall 520 VRPT (Vantage Report) 477, 484

W

wall-mounting 39 warranty 527 note 527 Web Configurator 20 access 21 access users 371 requirements 21 supported browsers 21 web proxy servers 225 see also HTTP redirect web-based SSL application 422 configuration example 423 create 425 weblink 423 weighted round robin (for load balancing) 178 weighted round robin algorithm 242 Windows Internet Naming Service, see WINS Windows Internet Naming Service, see WINS. Windows Remote Desktop 422 WINS 120, 157, 169, 174, 312 in L2TP VPN 338

WINS server 120, 338 Wizard Setup 33, 42 WWW 446 and address groups 450 and address objects 450 and authentication method objects 449 and certificates 448 and zones 450 see also HTTP, HTTPS 446

Ζ

zipped files troubleshooting 518 zones 208 and firewall 256, 262 and FTP 468 and interfaces 208 and SNMP 472 and SSH 464 and Telnet 466 and VPN 208 and WWW 450 extra-zone traffic 209 inter-zone traffic 209 intra-zone traffic 209 types of traffic 208